

ANTIOXIDANTS IN BIOLOGY AND MEDICINE Essentials, Advances, and

Clinical Applications

Yunbo Li



ANTIOXIDANTS IN BIOLOGY AND MEDICINE: ESSENTIALS, ADVANCES, AND CLINICAL APPLICATIONS

No part of this digital document may be reproduced, stored in a retrieval system or transmitted in any form or by any means. The publisher has taken reasonable care in the preparation of this digital document, but makes no expressed or implied warranty of any kind and assumes no responsibility for any errors or omissions. No liability is assumed for incidental or consequential damages in connection with or arising out of information contained herein. This digital document is sold with the clear understanding that the publisher is not engaged in rendering legal, medical or any other professional services.

ANTIOXIDANTS IN BIOLOGY AND MEDICINE: ESSENTIALS, ADVANCES, AND CLINICAL APPLICATIONS

YUNBO LI



Nova Science Publishers, Inc. New York Copyright © 2011 by Nova Science Publishers, Inc.

All rights reserved. No part of this book may be reproduced, stored in a retrieval system or transmitted in any form or by any means: electronic, electrostatic, magnetic, tape, mechanical photocopying, recording or otherwise without the written permission of the Publisher.

For permission to use material from this book please contact us: Telephone 631-231-7269; Fax 631-231-8175 Web Site: http://www.novapublishers.com

NOTICE TO THE READER

The Publisher has taken reasonable care in the preparation of this book, but makes no expressed or implied warranty of any kind and assumes no responsibility for any errors or omissions. No liability is assumed for incidental or consequential damages in connection with or arising out of information contained in this book. The Publisher shall not be liable for any special, consequential, or exemplary damages resulting, in whole or in part, from the readers' use of, or reliance upon, this material. Any parts of this book based on government reports are so indicated and copyright is claimed for those parts to the extent applicable to compilations of such works.

Independent verification should be sought for any data, advice or recommendations contained in this book. In addition, no responsibility is assumed by the publisher for any injury and/or damage to persons or property arising from any methods, products, instructions, ideas or otherwise contained in this publication.

This publication is designed to provide accurate and authoritative information with regard to the subject matter covered herein. It is sold with the clear understanding that the Publisher is not engaged in rendering legal or any other professional services. If legal or any other expert assistance is required, the services of a competent person should be sought. FROM A DECLARATION OF PARTICIPANTS JOINTLY ADOPTED BY A COMMITTEE OF THE AMERICAN BAR ASSOCIATION AND A COMMITTEE OF PUBLISHERS.

Additional color graphics may be available in the e-book version of this book.

Library of Congress Cataloging-in-Publication Data

Antioxidants in biology and medicine: essentials, advances, and clinical applications / author, Yunbo Li. p. ; cm. Includes bibliographical references and index. ISBN 978-1-61209-466-3 (eBook) 1. Antioxidants. 2. Active oxygen. I. Li, Yunbo. [DNLM: 1. Antioxidants. QV 325] QP535.01A585 2010 616.07'1--dc22 2010041342

Published by Nova Science Publishers, Inc. / New York

Contents

Preface		vii
SECTION I	INTRODUCTION TO OXIDATIVE STRESS AND ANTIOXIDANTS	1
Chapter 1	Introduction to Oxidative Stress	3
Chapter 2	Introduction to Antioxidants	21
SECTION II	ANTIOXIDANTS SYNTHESIZED BY CELLS	31
Chapter 3	Superoxide Dismutase	33
Chapter 4	Catalase	47
Chapter 5	Glutathione and its Synthesizing Enzymes	59
Chapter 6	Glutathione Peroxidase	73
Chapter 7	Glutathione Reductase	85
Chapter 8	Glutathione S-Transferase	91
Chapter 9	Glutaredoxin	103
Chapter 10	Thioredoxin	111
Chapter 11	Peroxiredoxin	123
Chapter 12	Thioredoxin Reductase	135
Chapter 13	Sulfiredoxin	143
Chapter 14	Methionine Sulfoxide Reductase	149
Chapter 15	Heme Oxygenase	159
Chapter 16	NAD(P)H:Quinone Oxidoreductase	179
Chapter 17	Paraoxonase	197
Chapter 18	Metal Ion Sequestering Proteins	217
Chapter 19	Other Non-Protein Antioxidants Synthesized by Cells	239

SECTION III	ANTIOXIDANTS DERIVED FROM THE DIET	263
Chapter 20	Vitamin C	265
Chapter 21	Vitamin E	281
Chapter 22	Carotenoids	297
Chapter 23	Phenolic Compounds	315
SECTION IV	Synthetic Antioxidants	337
Chapter 24	Antioxidant Enzyme Mimetics	339
Chapter 25	Glutathione Precursors	353
Chapter 26	Spin Traps	367
Chapter 27	Nanomaterials	377
Index		389

Preface

Molecular oxygen is essential for aerobic life, yet incomplete reduction or excitation of molecular oxygen during aerobic metabolisms results in the inevitable formation of reactive oxygen species (ROS). ROS are capable of causing damage to cellular constituents, including proteins, lipids, and nucleic acids. As such, aerobic organisms, including mammals have evolved a series of complementary antioxidant enzymes and non-enzymatic antioxidant molecules to control ROS and thereby protect critical biomolecules from oxidative damage. In addition, many diet-derived compounds have antioxidant activities and thus also contribute to the mammalian antioxidant defenses. Under physiological conditions, ROS and antioxidant defenses are generally in balance. However, increased formation of ROS or compromised antioxidant defenses may disrupt this balance, leading to oxidative stress that may eventually contribute to the development of many diseases.

The last two to three decades have witnessed the remarkable advances in our knowledge about ROS and antioxidants in biology and medicine. It has been increasingly recognized that ROS play an important role in diverse disease processes, including cardiovascular disorders, diabetes, neurodegeneration, and cancer. In line with this notion, accumulating evidence also demonstrates the effectiveness of antioxidant-based strategies for the intervention of an increasing number of disease processes that involve an oxidative stress mechanism. In this context, many synthetic antioxidant compounds have been developed, which can also be used to augment antioxidant defenses to protect against oxidative stress-associated disease pathophysiology. New knowledge in antioxidant biology and medicine will certainly further enhance our ability to develop more effective mechanistically-based strategies to combat human diseases that involve an oxidative stress component.

This book covers both the fundamentals and the most recent advances in antioxidant research in biology and medicine. The aim of the book is to provide a critical evaluation of the various endogenous and exogenous antioxidants with an emphasis on their role in health and disease. To this end, the book focuses on discussing solid scientific evidence obtained from basic research in mammalian models of human diseases as well as studies conducted directly in human subjects. The book is divided into four sections, as listed below.

- Section I: Introduction to Oxidative Stress and Antioxidants
- Section II: Antioxidants Synthesized by Cells
- Section III: Antioxidants Derived from the Diet
- Section IV: Synthetic Antioxidants

To set a stage for the subsequent discussion on how individual antioxidants function to protect against oxidative stress and disease processes, Section I introduces the basic knowledge on ROS and oxidative stress. It also provides an overview of the different kinds of antioxidants, the molecular regulation of antioxidant genes, and the antioxidant-based strategies for disease intervention. Section I has two chapters.

Section II consists of seventeen chapters devoted to the discussion of each of the major endogenous antioxidants synthesized by cells with respect to their basic biology and involvement in health and disease. Each chapter covers knowledge in three major areas: (1) basic biochemistry and molecular regulation, (2) animal studies on the role in disease processes, and (3) human studies and clinical applications.

Section III contains four chapters that cover the major antioxidants derived from the diet: vitamin C, vitamin E, carotenoids, and phenolic compounds. The biochemical properties and role in health and disease (including both experimental and clinical evidence) for each of these diet-derived antioxidants are discussed.

Section IV consists of four chapters on synthetic antioxidants, including antioxidant enzyme mimetics, glutathione precursors, spin traps, and nanomaterials. These chapters describe the chemical and biochemical properties of each of the above four classes of synthetic antioxidants with an emphasis on their protective effects in disease processes in both experimental animals and human subjects.

It is hoped that this book by integrating knowledge on antioxidants from essentials to advances, and from basic research to clinical applications will provide the reader a unique approach to understanding the rapidly evolving field of antioxidants in biology and medicine.

> Yunbo Li, MD, PhD Blacksburg, Virginia January, 2011

SECTION I INTRODUCTION TO OXIDATIVE STRESS AND ANTIOXIDANTS

Utilization of molecular oxygen by aerobic organisms, including mammals leads to the inevitable formation of reactive oxygen and nitrogen species (ROS/RNS) from various metabolic processes. ROS/RNS have the potential to damage cellular constituents. As such, aerobic organisms have evolved a number of antioxidants to protect critical biomolecules from ROS/RNS-mediated damage. Under physiological conditions, formation of ROS/RNS and functioning of antioxidants are generally in balance. However, increased ROS/RNS production or compromised antioxidant defenses may disrupt this balance, leading to potentially detrimental conditions, such as oxidative stress injury. Understanding of the basic biology of ROS/RNS and oxidative stress is crucial for developing antioxidant-based strategies for the intervention of diseases involving an ROS/RNS-dependent mechanism. Section I introduces the basics related to ROS/RNS and oxidative stress, and provides an overview of antioxidants and their gene regulation. This sets a stage for the subsequent chapters devoted to the discussion of individual antioxidants in biology and medicine. Section I consists of two chapters.

Chapter 1

Introduction to Oxidative Stress

Abstract

Reactive oxygen and nitrogen species (ROS/RNS) are collective terms that refer to a number of oxygen- and nitrogen-containing reactive species. ROS/RNS are produced from various cellular sources. They also exist as environmental pollutants. Although ROS/RNS fulfill important physiological functions, uncontrolled formation of ROS/RNS leads to detrimental conditions, such as oxidative stress injury. This chapter introduces the common types of ROS/RNS and summarizes the role of oxidative stress in disease development.

1. Historical Overview

Molecular oxygen is essential for aerobic life, yet incomplete reduction or excitation of molecular oxygen during aerobic metabolisms generates oxygen-containing reactive species that pose a serious threat to aerobic organisms. Although more than two centuries ago J. Priestley stated that "oxygen might not be so proper for use in the usual healthy state of the body...", insight into the mechanisms of oxygen toxicity was provided in an article published in 1954 by R. Gerschman and coworkers, hypothesizing that oxygen poisoning and radiation injury may have at least one common basis of action, possibly through the formation of oxidizing free radicals [1]. Another important event in the investigations of oxygen toxicity was the finding by J.M. McCord and I. Fridovich in 1969 of an enzymatic function of a protein containing both copper and zinc, which then was known alternatively as erythrocuprein [2]. The function of this enzyme (Cu,Zn superoxide dismutase) is catalysis of dismutation of superoxide (a free radical) to produce hydrogen peroxide and molecular oxygen. The discovery of Cu,Zn superoxide dismutase has triggered extensive investigations into the free radical mechanisms of oxygen toxicity. A large body of evidence over the last several decades demonstrates a critical role for oxygen-derived free radicals and related reactive species in the pathophysiology of a wide variety of diseases. While free radicals and related reactive species cause deleterious effects in biological systems, it is important to note that these reactive species may also play important physiological roles, such as antimicrobial activity and participation in cell signal transduction [3, 4].

2. Free Radicals and Reactive Oxygen and Nitrogen Species (ROS/RNS)

2.1. Definitions

2.1.1. Free Radicals

Free radical is a commonly encountered term in biology and medicine. It refers to any chemical species capable of independent existence that contains one or more unpaired electrons. An unpaired electron refers to the one that occupies an atomic or molecular orbital by itself. For example, the superoxide (O_2^{-}) noted above is a free radical. The superscript dot indicates the unpaired electron whereas the superscript minus sign indicates that the free radical is negatively charged. Both anionic and cationic free radicals exist in biological systems. In general, free radicals are reactive species. However, as described below, reactive species are not necessarily free radicals.

2.1.2. ROS/RNS

Reactive oxygen species (ROS) is a term also frequently used in biology and medicine. This term can be simply defined as oxygen-containing reactive species. It is a collective term to include superoxide (O_2 ⁻), hydrogen peroxide (H_2O_2), hydroxyl radical (OH), singlet oxygen (1O_2), peroxyl radical (LOO⁻), alkoxyl radical (LO⁻), lipid hydroperoxide (LOOH), peroxynitrite (ONOO⁻), hypochlorous acid (HOCl), and ozone (O₃). Similarly, the term reactive nitrogen species (RNS) has been coined to include nitric oxide ('NO), peroxynitrite, nitrogen dioxide ('NO₂), and other oxides of nitrogen or nitrogen-containing reactive species [5]. It is of note that peroxynitrite is classified as both an ROS and an RNS in the literature.

Among the ROS listed above, some contain unpaired electrons, and thus belong to free radicals. They are also called oxygen free radicals. Examples of oxygen free radicals include superoxide, hydroxyl radical, peroxyl radical, and alkoxyl radical. Some ROS do not contain any unpaired electrons, and thus they are not free radicals. Examples of the non-radical ROS include hydrogen peroxide, peroxynitrite, hypochlorous acid, and ozone. Singlet oxygen can exist in two states: the delta state $({}^{1}\Delta_{g})$ and sigma state $({}^{1}\Sigma_{g}^{+})$. $({}^{1}\Sigma_{g}^{+}){}^{1}O_{2}$ is a free radical because it contains two unpaired electrons. On the other hand, $({}^{1}\Delta_{g}){}^{1}O_{2}$ is a non-radical. Due to its highly unstable nature, $({}^{1}\Sigma_{g}^{+}){}^{1}O_{2}$ is generally regarded to lack biological significance. As such, in biology and medicine, singlet oxygen usually refers to the ${}^{1}\Delta_{g}$ state and is considered a non-radical species.

2.2. Sources of ROS/RNS

Utilization of molecular oxygen by aerobic organisms leads to the formation of ROS/RNS. ROS/RNS are generated from various endogenous cellular sources, such as NAD(P)H oxidases and mitochondria. ROS/RNS are also produced from a variety of exogenous sources, including physical agents, xenobiotics, and biologic agents (Table 1-1).

2.2.1. Endogenous Sources

The various endogenous sources of ROS/RNS in mammals are described below.

Source	Description
Endogenous Sources	NAD(P)H oxidases
	Mitochondria (electron transport chain and oxidases)
	Xanthine oxidoreductase
	Cytochrome P450 enzymes
	Uncoupled nitric oxide synthases
	Peroxisomes
Exogenous Sources	Physical agents and particulate matter
	Xenobiotics
	Biologic agents (bacteria and viruses)

Table 1-1. Sources of reactive oxygen and nitrogen species in biology and medicine

2.2.1.1. NAD(P)H OXIDASES

The NAD(P)H oxidase in phagocytic cells produces large amounts of superoxide during respiratory burst. Phagocytic NAD(P)H oxidase is perhaps the best characterized cellular source of ROS. This multicomponent enzyme consists of the catalytic subunit $gp91^{phox}$ (also referred to as NOX2), together with the regulatory subunits $p22^{phox}$, $p47^{phox}$, $p40^{phox}$, $p67^{phox}$, and the small GTPase Rac. For many years, this enzyme was thought to exist as a single isoform only in phagocytes and its function restricted to antimicrobial action. This notion changed with the discovery of multiple NOX forms that represent an enzyme family containing at least five members, namely, NOX1, NOX2, NOX3, NOX4, and NOX5. NOX stands for <u>N</u>AD(P)H <u>ox</u>idase. The NOX enzymes are expressed in various types of cells, and represent an important source of ROS under a wide variety of physiological and pathophysiological conditions [6].

2.2.1.2. MITOCHONDRIA

Another important cellular source of ROS is mitochondria. In mammalian cells, mitochondrial respiration usually accounts for 80-90% of the oxygen consumed by the cells. Under physiological conditions, both complexes I and III of the mitochondrial electron transport chain are involved in the univalent reduction of molecular oxygen to superoxide [7]. It is estimated that about 0.1% of the oxygen utilized by mitochondria is converted to superoxide. The superoxide formed in mitochondria undergoes dismutation to yield hydrogen peroxide either spontaneously or catalyzed by manganese superoxide dismutase in mitochondrial matrix. The formation of ROS by mitochondrial electron transport chain is increased under various pathophysiological conditions, including tissue ischemia-reperfusion [8]. Certain xenobiotics, such as quinone compounds undergo redox cycling catalyzed by mitochondrial electron transport chain complexes, leading to increased ROS formation. ROS formed through this mechanism are believed to contribute to the cardiotoxicity of doxorubicin, a quinone-containing anticancer drug [9]. In addition to mitochondrial electron transport chain, NOX4 is recently identified to be present in mitochondria, and may contribute to ROS formation in this organelle [10]. Monoamine oxidase associated with mitochondria is also known to produce hydrogen peroxide. Hence, ROS formed in mitochondria may result from two general pathways: mitochondrial electron transport chain complexes and oxidases present in mitochondria. Besides ROS, nitric oxide is also generated in mitochondria via the action of mitochondrial nitric oxide synthase. Uncontrolled production of mitochondria-derived ROS/RNS has been implicated in diverse pathophysiological processes, such as tissue ischemia-reperfusion injury and neurodegeneration [11].

2.2.1.3. OTHER ENDOGENOUS SOURCES

Other cellular sources of ROS include xanthine oxidoreductase, cytochrome P450 enzymes, uncoupled nitric oxide synthases, and peroxisomes. Xanthine oxidoreductase has two interconvertible forms; xanthine dehydrogenase and xanthine oxidase. Both forms catalyze the conversion of hypoxanthine to xanthine, and xanthine to uric acid. Both forms also catalyze one- and two-electron reduction of molecular oxygen to form superoxide and hydrogen peroxide, respectively, with xanthine oxidase being more active in generating the ROS. Xanthine oxidoreductase-derived ROS are implicated in pathophysiological processes, such as tissue ischemia-reperfusion injury. Cytochrome P450 enzymes, particularly P4502E1 can generate significant amounts of ROS when being induced by xenobiotics [12]. The ROS formed from P4502E1 contribute to alcoholic liver injury. Nitric oxide synthases normally catalyze the formation of nitric oxide. Under certain conditions, such as substrate deficiency or oxidative modifications, these enzymes can catalyze one-electron reduction of molecular oxygen to form superoxide. The organelle peroxisome contains enzymes, especially the flavoprotein dehydrogenases that produce significant amounts of hydrogen peroxide. Peroxisome also contains large quantities of catalase, which decomposes hydrogen peroxide (see Chapter 4 for discussion of catalase).

2.2.2. Exogenous Sources

2.2.2.1. PHYSICAL AGENTS AND PARTICULATE MATTER

Ionizing radiation causes homolysis of water to form hydroxyl radical, which contributes to radiation-induced tissue injury. Hydroxyl radical is also formed from ultraviolet light-induced homolysis of hydrogen peroxide (Section 2.3.3). Exposure of biological systems to ionizing radiation and ultraviolet light may result in the formation of other ROS/RNS as well. In addition to ionizing radiation and ultraviolet light, exposure to asbestos fibers or silica dusts leads to the formation of ROS/RNS in biological systems. The metal ions associated with asbestos fibers participate in Fenton- or Fenton-type reaction to yield ROS (see Section 2.3.1 for description of Fenton reaction). Similarly, the surface redox chemical properties of silica dusts play a role in ROS formation. On the other hand, both asbestos fibers and silica dusts can provoke inflammatory responses, resulting in production of ROS/RNS. Various types of nanomaterials, including silicon, gold, and copper nanoparticles as well as CdTe quantum dots have been shown to induce ROS/RNS formation in biological systems though the detailed mechanisms remain to be elucidated [13]. In contrast, cerium oxide nanoparticles and fullerene-derived nanomaterials exert antioxidant effects via scavenging ROS in biological systems (see Chapter 27 for discussion of nanomaterials).

2.2.2.2. XENOBIOTICS

As described in Section 2.2.1.2 above, certain xenobiotics, including drugs are able to cause increased ROS/RNS levels in cells and tissues, leading to oxidative injury. There exist four general mechanisms for xenobiotic-augmented ROS/RNS levels in biological systems, as listed below.

- Redox cycling of xenobiotics, such as quinone compounds
- Interference with mitochondrial electron transport chain, causing increased electron leakage and thereby the augmented formation of superoxide
- Induction of ROS/RNS-producing enzymes, such as P4502E1
- Inhibition of cellular antioxidants

In addition, some ROS/RNS, such as ozone and nitrogen oxides exist as air pollutants. Exposure to these ROS/RNS air pollutants via inhalation may cause pulmonary inflammation and oxidative injury.

2.2.2.3. BIOLOGIC AGENTS

Biologic agents, including bacteria and viruses may cause increased levels of ROS/RNS in target cells and tissues. As outlined below, there are four potential mechanisms for the increased ROS/RNS levels caused by biologic agents.

- Biologic agents induce activation of inflammatory cells, such as phagocytic cells, resulting in ROS formation from NAD(P)H oxidase.
- Biologic agents can cause activation of other ROS/RNS-producing enzymes, such as inducible nitric oxide synthase to generate large amounts of nitric oxide. As described below, nitric oxide reacts with superoxide, generating the highly reactive species peroxynitrite (Section 2.3.4).
- Certain bacteria contain ROS-producing enzymes that may release significant amounts of ROS.
- Some bacteria-derived toxins may downregulate tissue antioxidants. Viral infections may also decrease the expression of certain antioxidants in host cells via modulating cell signal transduction [14, 15].

2.3. Chemical Reactivity of Major Types of ROS/RNS

As noted above, ROS/RNS are generated from various endogenous and exogenous sources. The effects of ROS/RNS in biological systems are determined by multiple factors, among which the chemical reactivity of these species plays an important part in determining their biological effects. This section summarizes the major chemical reactivity for each of the ROS/RNS commonly encountered in biology and medicine.

2.3.1. Superoxide (O_2^{-})

Superoxide is formed from one-electron reduction of molecular oxygen with NAD(P)H oxidases, mitochondria, and xanthine oxidoreductase as major cellular sources (see Section 2.2 for discussion of sources of ROS/RNS). Due to its lower reduction potential, superoxide often acts as a reducing agent rather than an oxidizing species. As such, the ability of superoxide to directly oxidize biomolecules is much limited. Nevertheless, superoxide is an important ROS that can result in cell and tissue injury [16]. The biological damaging potential of superoxide is related to several chemical properties of this free radical species, which are summarized below.

- Although it generally acts as a reducing agent, superoxide does oxidize the ironsulfur clusters in several enzymes, such as aconitase. Oxidation of the iron-sulfur clusters by superoxide in these enzymes leads to the release of iron from the enzymes and the subsequent enzyme inactivation.
- Superoxide reacts with nitric oxide ('NO) at an almost diffusion-limited rate to form the highly damaging species peroxynitrite (ONOO'; Section 2.3.4) (Reaction 1).

$$O_2 + NO \longrightarrow ONOO^-$$
 (1)

 In the presence of transition metal ions (e.g., iron) superoxide reacts with hydrogen peroxide, producing hydroxyl radical, an extremely reactive species (Section 2.3.3). The reaction of superoxide and hydrogen peroxide in the presence of iron to produce hydroxyl radical is referred to as iron-catalyzed Haber-Weiss reaction (Reaction 2).

$$H_2O_2 + O_2 \stackrel{\cdot}{\longrightarrow} OH + OH \stackrel{\cdot}{\to} O_2$$
(2)

In the absence of iron this reaction proceeds slowly. The presence of iron dramatically accelerates the reaction to produce hydroxyl radical. The iron-catalyzed Haber-Weiss reaction can be written in two sub-reactions (Reactions 3 and 4). Reaction 4 is also referred to as Fenton reaction.

$$O_2 \stackrel{\cdot}{\cdot} + Fe^{3+} \longrightarrow O_2 + Fe^{2+}$$
(3)

$$Fe^{2+} + H_2O_2 \longrightarrow Fe^{3+} + OH + OH^-$$
 (4)

Superoxide undergoes spontaneous dismutation to form hydrogen peroxide and molecular oxygen, and this dismutation reaction is catalyzed by superoxide dismutase (SOD) with a reaction rate constant being over three orders of magnitude greater than that of spontaneous dismutation (Reaction 5) (see Chapter 3 for discussion of SOD).

$$O_2 \stackrel{\cdot}{\cdot} + O_2 \stackrel{\cdot}{\cdot} + 2H^+ \xrightarrow{\text{SOD}} H_2O_2 + O_2$$
(5)

2.3.2. Hydrogen Peroxide (H_2O_2)

Hydrogen peroxide is formed through either spontaneous or SOD-catalyzed dismutation of superoxide. As such, the cellular sources for superoxide formation are also the ones for hydrogen peroxide. Similar to superoxide, hydrogen peroxide is not a potent oxidant, and the direct reactions between hydrogen peroxide and most biomolecules are much limited. However, some enzymes, such as protein phosphatases can be directly inhibited by hydrogen peroxide through oxidation of the sulfhydryl (SH) groups of the enzymes [17]. Although it is not a potent oxidant, hydrogen peroxide at high levels can lead to cell and tissue injury. As summarized below, several mechanisms may account for the damaging effects of hydrogen peroxide in biological systems.

 Hydrogen peroxide reacts with the sulfhydryl groups of certain enzymes, leading to enzyme inactivation. • Reaction of hydrogen peroxide with iron produces hydroxyl radical, an extremely potent ROS. The Fenton reaction is believed to be an important mechanism by which hydrogen peroxide mediates oxidative damage. Other metal ions, such as cuprous ion (Cu¹⁺⁾ also catalyze the formation of hydroxyl radical from hydrogen peroxide via a similar reaction called Fenton-type reaction (Reaction 6).

$$Cu^{1+} + H_2O_2 \longrightarrow Cu^{2+} + OH + OH^-$$
(6)

 Reaction of hydrogen peroxide with chloride (Cl⁻) generates hypochlorous acid (HOCl), another potent oxidant (Section 2.3.8). This reaction is catalyzed by myeloperoxidase (MPO) found in phagocytic cells (Reaction 7). Hypochlorous acid is involved in the killing of invading microorganisms as well as tissue injury associated with phagocytic cell activation [18].

$$H_2O_2 + C\Gamma \xrightarrow{MPO} HOCl + OH^-$$
 (7)

Hydrogen peroxide is decomposed to water by three major types of enzymes in mammals, which are listed below.

- Catalase (Chapter 4)
- Glutathione peroxidase (Chapter 6)
- Peroxiredoxin (Chapter 11)

2.3.3. Hydroxyl Radical ('OH)

т. · · п. 1· /·

One-electron reduction of hydrogen peroxide forms hydroxyl radical. One source of hydroxyl radical formation in biological systems is the reaction of hydrogen peroxide with transition metal ions through Fenton or Fenton-type reactions (Reactions 4 and 6). As outlined below, there are several other potential sources of hydroxyl radical formation, which may also have biological relevance.

 Ionizing radiation induces homolytic cleavage of water to form hydroxyl radical and hydrogen atom (H⁻) (Reaction 8). Ultraviolet (UV) light-induced homolytic cleavage of hydrogen peroxide also yields hydroxyl radicals (Reaction 9). The hydroxyl radicals formed mediate ionizing radiation- and UV light-induced injury.

$$H_2O \xrightarrow{\text{Ionizing Radiation}} H^{\cdot} + OH$$
(8)

$$H_2O_2 \xrightarrow{\text{UV Light}} 2^{\circ}OH$$
 (9)

• Reaction of hypochlorous acid (HOCl) with superoxide produces hydroxyl radical (Reaction 10). This reaction may be particularly relevant to sustained activation of inflammatory cells (e.g., phagocytic cells) during which both hypochlorous acid and superoxide are formed in large quantities.

$$HOCl + O_2^{\bullet} \longrightarrow O_2 + Cl^{\bullet} + OH$$
(10)

 Peroxynitrite (ONOO⁻) becomes protonated to form peroxynitrous acid (ONOOH) (Reaction 11). Decomposition of peroxynitrous acid yields hydroxyl radical and nitrogen dioxide radical ('NO₂) (Reaction 12) [19].

$$ONOO^- + H^+ \longrightarrow ONOOH$$
 (11)

$$ONOOH \longrightarrow OH + NO_2$$
(12)

• Reaction of electronically excited nitric dioxide ('NO₂*) with water is reported to form hydroxyl radical (Reaction 13).

$$NO_2^* + H_2O \longrightarrow OH + HONO$$
(13)

HONO is nitrous acid. The above reaction is considered a chemical mechanism of atmospheric hydroxyl radical production [20].

Hydroxyl radical is known as the most reactive ROS. Due to its extremely high oxidizing capacity, hydroxyl radical reacts with the first biomolecule that it bumps into at a reaction rate constant very near the diffusion limit. Thus, hydroxyl radical reacts non-selectively with both less vital biomolecules (e.g., cellular antioxidants) and vital biomolecules (e.g., DNA).

2.3.4. Peroxynitrite (ONOO⁻)

Peroxynitrite is formed from the reaction of superoxide and nitric oxide (Reaction 1). This reaction occurs at an almost diffusion-limited rate. Thus, peroxynitrite forms if a biological system simultaneously generates both superoxide and nitric oxide. As indicated above, peroxynitrite is negatively charged and undergoes protonation to form peroxynitrous acid (ONOOH) (Reaction 11).

Peroxynitrite is a potent oxidant capable of causing biological damage. Although it is not a free radical, peroxynitrite is much more reactive than its parent molecules, superoxide and nitric oxide. In this regard, as described below, formation of peroxynitrite is believed to mediate the detrimental effects of high levels of nitric oxide formed during inflammation (Section 2.3.5). Peroxynitrite reacts directly or indirectly with most critical biomolecules, including lipids, proteins, and nucleic acids. Several chemical mechanisms contribute to the biological damaging effects of peroxynitrite, which are outlined below [19].

- Peroxynitrite and its protonated form (peroxynitrous acid) can directly cause oxidation of biomolecules.
- Decomposition of peroxynitrous acid leads to the formation of hydroxyl radical and nitrogen dioxide (Reaction 12), both of which are highly reactive species capable of damaging biomolecules.
- Peroxynitrite reacts with carbon dioxide (CO₂), resulting in the formation of carbonate radical (CO₃⁻) and nitrogen dioxide via an unstable intermediate nitrosoperoxycarbonate (ONOOCO₂⁻) (Reaction 14).

$$ONOO^{-} + CO_2 \longrightarrow [ONOOCO_2^{-}] \longrightarrow CO_3^{-} + NO_2$$
 (14)

Carbonate radical is a potent oxidant capable of oxidizing biomolecules. Due to the presence of high levels of carbon dioxide in cells and tissues, the above reaction pathway is believed to play a major role in peroxynitrite-induced biological damage.

Glutathione (Chapter 5) is perhaps the most important biomolecule in the detoxification of peroxynitrite in mammalian systems. Both glutathione peroxidase (Chapter 6) and peroxiredoxin (Chapter 11) are able to catalyze the conversion of peroxynitrite to nitrite in vitro. However, the role of these two enzymes in protecting against peroxynitrite toxicity in vivo remains to be established.

2.3.5. Nitric Oxide ('NO) and Related Nitrogen Oxides

Nitric oxide, also referred to as nitric monoxide, has one unpaired electron, and is thus a free radical. The dot in 'NO stands for the unpaired electron. However, nitric oxide is often written as NO with the dot omitted. Nitric oxide is the first gaseous species identified as an endogenously generated signaling molecule. Nitric oxide plays important roles in many physiological processes, such as vascular homeostasis, neurotransmission, and host defense [21]. Nitric oxide is formed in biological systems by two general pathways: nitric oxide synthase-dependent pathway and nitrate-nitrite-nitric oxide pathway [22].

Nitric oxide possesses important physiological functions; however, under certain conditions, such as inflammation it also contributes to pathophysiological processes. Although nitric oxide is reported to exert various adverse effects, most of them are more likely mediated by its oxidation products rather than nitric oxide itself [19]. Indeed, the direct reactions of nitric oxide with most biomolecules, including proteins (except for heme-containing proteins), lipids, and nucleic acids are much limited. Because of its free radical nature, however, nitric oxide reacts readily with other free radical species. As noted earlier, nitric oxide reacts with superoxide at an almost diffusion-limited rate to form the highly damaging species peroxynitrite (Reaction 1). In addition, nitric oxide-mediated redox reactions may lead to the formation of secondary nitrogen oxides, such as nitrogen dioxide. Nitrogen dioxide is a free radical species of high reactivity to various biomolecules. It also exists as an important air pollutant.

2.3.6. Peroxyl Radical (LOO') and Alkoxyl Radical (LO')

Alkoxyl and peroxyl radicals are formed from lipid peroxidation. Hydroxyl radical abstracts a hydrogen atom from a lipid molecule (LH), leading to the formation of a carboncentered radical (L'). In the presence of molecular oxygen, the carbon-centered radical is converted to a peroxyl radical (LOO') (Reactions 15 and 16).

$$OH + LH \longrightarrow L' + H_2O$$
(15)

$$L' + O_2 \longrightarrow LOO'$$
 (16)

Both alkoxyl and peroxyl radicals can abstract hydrogen atoms from adjacent lipid molecules, resulting in the propagation of lipid peroxidation. After abstracting a hydrogen atom (H) from an adjacent lipid molecule, peroxyl radical is reduced to lipid hydroperoxide (LOOH) (Reaction 17). Decomposition of lipid hydroperoxide in the presence of metal ions, such as iron leads to the formation of alkoxyl radical or peroxyl radical depending on the form of the iron ions (Reactions 18 and 19).

$$LOO' + H \rightarrow LOOH$$
 (17)

$$LOOH + Fe^{2+} \longrightarrow LO' + Fe^{3+} + OH^{-}$$
(18)

$$LOOH + Fe^{3+} \longrightarrow LOO' + Fe^{2+} + H^+$$
(19)

Alkoxyl and peroxyl radicals are strong oxidants. In addition to propagating lipid peroxidation in biomembranes or lipoproteins, alkoxyl and peroxyl radicals are also able to oxidize proteins, leading to protein dysfunction and enzyme inactivation. Oxidation of DNA by these free radicals results in oxidative base modifications [23]. Notably, peroxyl radicals react with each other, yielding singlet oxygen [24], a potent oxidant capable of damaging biomolecules (Section 2.3.7).

Both alkoxyl and peroxyl radicals react readily with cellular non-protein antioxidants, including vitamin E, vitamin C, glutathione, and bilirubin (see later chapters for discussion of these antioxidants). These antioxidants act to protect more vital cellular constituents from damage by alkoxyl and peroxyl radicals. At present, no specific enzymes have been found to catalytically metabolize alkoxyl and peroxyl radicals in mammals.

2.3.7. Singlet Oxygen $({}^{1}O_{2})$

Singlet oxygen is an electronically excited form of molecular oxygen. As noted earlier in Section 2.1.2, the term singlet oxygen in biology and medicine usually refers to the ${}^{1}\Delta_{g}$ state, which is not a free radical. As described below, singlet oxygen is formed by different pathways or chemical mechanisms in biological systems.

• Photosensitivity reaction: In a photosensitivity reaction, an endogenous photosensitizer molecule (PS) (e.g., a porphyrin compound) is converted to an excited state (PS*) upon illumination by light. The excitation energy is then transferred to molecular oxygen, converting it to singlet oxygen, while the photosensitizer molecule returns to the ground state (Reactions 20 and 21).

$$PS + light \longrightarrow PS^*$$
(20)

 $PS^* + O_2 \longrightarrow PS + {}^1O_2$

- (21)
- Phagocytic cell respiratory burst: Phagocytic cell activation during inflammation leads to the formation of hypochlorous acid (HOCl), which reacts with hydrogen peroxide, forming singlet oxygen (Reaction 22) [25].

$$HOCl + H_2O_2 \longrightarrow {}^{1}O_2 + H_2O + HCl$$
(22)

• Lipid peroxidation: As stated above in Section 2.3.6, peroxyl radicals (LOO') generated during lipid peroxidation react with each other to yield singlet oxygen along with the formation of alcohol (LOH) and ketone (LO). This reaction proceeds via a linear tetraoxide intermediate (LOOOOL) (Reaction 23).

$$LOO' + LOO' \longrightarrow [LOOOOL] \longrightarrow LOH + LO + {}^{1}O_{2}$$
(23)

Singlet oxygen is a strong oxidant that displays considerable reactivity towards biomolecules, including lipids, proteins, and nucleic acids [26, 27]. The interactions between singlet oxygen and biomolecules may occur via two modes: direct chemical reaction and physical quenching. Physical quenching results in energy transfer and de-excitation of the singlet state without chemical changes in the energy acceptor. Direct chemical reaction is the major mode of reactivity of singlet oxygen in biological systems. Attack of lipid molecules in biomembranes or lipoproteins by singlet oxygen triggers lipid peroxidation. Reaction of singlet oxygen with proteins can result in multiple effects, including oxidation of side-chains, backbone fragmentation, aggregation, unfolding, enzyme inactivation, and enhanced degradation by proteasome pathway. Reaction of singlet oxygen with DNA leads to base modifications and DNA strand breaks.

A number of non-protein antioxidants react with singlet oxygen and thereby protect vital biomolecules from oxidative damage. Notably, dietary carotenoids, such as β -carotene and lycopene (Section 2 of Chapter 22) have been shown to potently quench singlet oxygen. To date, no enzymes have been shown to selectively detoxify singlet oxygen through a catalytic mechanism.

2.3.8. Hypochlorous Acid (HOCI)

Hypochlorous acid is formed from the reaction of hydrogen peroxide and chloride catalyzed by phagocytic myeloperoxidase during inflammation (Reaction 7). Myeloperoxidase also utilizes hydrogen peroxide to oxidize bromide (Br) and thiocyanate (SCN) to generate two other potent oxidants hypobromous acid (HOBr) and hypothiocyanous acid (HOSCN), respectively. Both HOBr and HOSCN are minor products as compared with hypochlorous acid [28].

Hypochlorous acid is a potent oxidizing species that reacts with proteins, lipids, nucleic acids, and carbohydrates. Formation of hypochlorous acid by phagocytic cells during respiratory burst plays an integral part in innate immune defense system. Hypochlorous acid along with other oxidizing species derived from hypochlorous acid, including hydroxyl radical and singlet oxygen contributes to the killing of microorganisms. However, under certain conditions, such as chronic inflammation, uncontrolled formation of hypochlorous acid may attack normal cellular biomolecules, leading to cell and tissue injury [29]. In this context, hypochlorous acid has been implicated in the pathogenesis of various human diseases, including atherosclerosis [30].

Cellular glutathione has been suggested to play a major role in the detoxification of hypochlorous acid. Hypochlorous acid also reacts with the sulfonate amino acid taurine to form an unreactive chloroamine product. Taurine is present at high concentrations in neutrophils, and may act to detoxify hypochlorous acid.

2.3.9. Ozone (O₃)

Ozone is a gas in the stratosphere that protects animals on earth from solar irradiation. Ozone is also an air pollutant near the earth's surface, where it is formed by photochemical reactions. Ozone pollution still represents a significant environmental issue in many urban areas. Recently, it is reported that ozone may be produced by antibodies in biological systems [31]. However, this finding has been questioned by other investigators. Ozone is not a free radical, but has potent oxidizing capacity. It reacts with biomolecules, including lipids, pro-

teins and nucleic acids, resulting in cell and tissue damage. Oxidative injury is a significant mechanism of ozone-induced pulmonary disorders.

2.4. Molecular Targets of ROS/RNS and Cellular Consequences

As discussed above, ROS/RNS are reactive species capable of damaging cellular constituents, including proteins, lipids, and nucleic acids, leading to cell and tissue injury. Reactions of ROS/RNS with proteins can cause enzyme inactivation, structural alterations, and enhanced protein degradation. Attack of lipids by ROS/RNS causes lipid peroxidation, leading to membrane disruption as well as the formation of secondary reactive species, such as aldehydes. Lipid peroxidation-derived reactive aldehydes (e.g., malondialdehyde, 4-hydroxy-2-nonenal, acrolein) are cytotoxic and mutagenic. They also modulate cell signal transduction. Direct DNA damage by ROS/RNS can be manifested as DNA base modifications, strand breakage, and DND-DNA or DNA-protein cross-links. Damage of the above vital biomolecules by ROS/RNS may lead to cell dysfunction, senescence, death, or malignant transformation. Due to the detrimental nature of ROS/RNS, aerobic organisms, including mammals have evolved diverse antioxidant defenses to protect vital biomolecules from ROS/RNS-mediated damage. In addition, a number of compounds derived from dietary sources or synthesized in the laboratories also exhibit antioxidant activities in biological systems. An introduction to the antioxidant system is provided in Chapter 2. The delicate balance between ROS/RNS and antioxidant system is an integral part of physiological homeostasis in mammals. Disruption of this balance may lead to disease pathophysiology.

3. Oxidative Stress and Diseases

3.1. Definitions

3.1.1. Oxidative Stress

The term oxidative stress refers to a condition where the levels of ROS significantly overwhelm the capacity of antioxidant defenses in a biological system. Oxidative stress condition can be caused by either increased ROS formation or decreased activity of antioxidants or both in a biological system. Oxidative stress condition is associated with oxidative damage to biomolecules, including proteins, lipids, and nucleic acids. Moderate oxidative stress may cause cell dysfunction, whereas overt oxidative stress usually causes cell death. It should be noted that increases in ROS levels in a biological system are not always associated with injury. Under certain circumstances, small transient increases in ROS levels can be employed as a signaling mechanism, leading to physiological cellular responses [4, 17]. In this context, the disease conditions in which oxidative stress plays a role usually involve sustained formation of relatively large amounts of ROS via various mechanisms, including activation of inflammatory cells.

Oxidative stress and inflammation are intimately related processes. On the one hand, inflammation leads to the production of ROS/RNS from cellular sources, including NAD(P)H oxidase and inducible nitric oxide synthase. Such formed ROS/RNS mediate the detrimental effects associated with dysregulated inflammation. On the other hand, ROS/RNS cause increased expression of proinflammatory cytokines and adhesion molecules, thus augmenting or propagating the dysregulated inflammatory process. Such a dysregulated inflammatory state is sometimes referred to as inflammatory stress. As oxidative stress and inflammation are closely intertwined, the compound term oxidative/inflammatory stress has been used in the literature to describe settings where both oxidative stress and dysregulated inflammation contribute to disease pathophysiology.

3.1.2. Other Related Terms

Analogous terms have been coined to describe the conditions associated with increased levels of RNS. For example, the term nitrative stress is defined as a condition under which the levels of RNS significantly overwhelm the capacity of RNS-detoxification mechanisms in a biological system. Nitrative stress condition is associated with nitration of biomolecules, leading to cell and tissue injury. Nitration refers to the addition of a nitro (-NO₂) group to a compound. The RNS peroxynitrite readily causes nitrative stress in biological systems. Another term nitrosative stress is defined as a condition induced by nitric oxide or related species, leading to nitrosylation of critical protein cysteine thiols (*S*-nitrosylation) and metallocofactors of proteins. Nitrosylation refers to the addition of a nitros (-NO) group to a thiol group or a redox active metal ion center of a protein. While dysregulation of nitrosylation is associated with pathophysiological conditions, well-controlled nitrosylation plays an important role in cell signal transduction and provides a mechanism of redox-based physiological regulation [32].

3.2. Oxidative Stress in Diseases

3.2.1. Different Roles of Oxidative Stress in Diseases

Increased formation of ROS/RNS and subsequent oxidative damage to biomolecules have been consistently observed under various disease conditions. However, these observations do not necessitate a causal relationship between oxidative stress and disease development. Possible relationships between oxidative stress and diseases, and the effects of controlling ROS/RNS on disease development are summarized below.

- ROS/RNS act as the only initial cause of the disease. Control of the ROS/RNS will prevent or stop the disease development.
- ROS/RNS act as one of the initial causes of the disease. Control of the ROS/RNS will mitigate the disease development.
- ROS/RNS act as a contributing factor in the disease progression. Control of the ROS/RNS will slow the disease progression.
- ROS/RNS are formed as a consequence of the disease and contribute to the progression of the disease pathophysiology. Control of the ROS/RNS will slow the disease progression.
- ROS/RNS are formed as a consequence of the disease and play no role in disease initiation or progression. Control of the ROS/RNS will have no effects on disease process.

3.2.2. Diseases Involving an Oxidative Stress Mechanism

Establishment of a causal or contributing role for ROS/RNS in a particular disease is the basis for developing strategies to prevent and treat that particular disease. Based on findings from extensive investigations in animal models of human diseases as well as studies in human subjects over the last several decades, it has been recognized that ROS/RNS play a causal and/or a contributing role in a wide variety of diseases [33]. Table 1-2 lists some of the disease processes and related conditions in which oxidative stress plays a role. Another line of evidence supporting an important role for elevated ROS/RNS in diseases is the protective effects of diverse antioxidants on disease pathophysiology, which are the main focus of this book.

Disease	Description
Cardiovascular Diseases	Hypertension, Atherosclerosis
	Myocardial ischemia-reperfusion injury
	Heart failure, Cardiotoxicity
Diabetes and Metabolic Syndrome	Diabetes, Diabetic complications
	Obesity, Insulin resistance
Neurological Diseases	Alzheimer's disease, Parkinson's disease
	Stroke, Amyotrophic lateral sclerosis
Pulmonary Diseases	Chronic obstructive pulmonary disease
	Asthma, Hyperoxia-induced lung injury
	Pulmonary toxicity
Hepatic and Gastrointestinal Diseases	Alcoholic fatty liver disease
	Nonalcoholic fatty liver disease
	Isohomia reportusion injury
Panal Disansas	Dishetia nonhronathy
Kenai Diseases	Ischemia-reperfusion injury Nephrotoxicity
Eve Diseases	Cataract
Lyc Discuses	Age-related macular degeneration
Skin Diseases	Ultraviolet light-induced skin injury
	Scleroderma, Contact dermatitis, Psoriasis
Cancer	Chemical carcinogenesis
	Spontaneous cancer development
	Angiogenesis, Cancer metastasis
Aging	Lifespan
	Aging-related organ degeneration
Arthritic Diseases	Rheumatoid arthritis
	Other types of arthritis
Sepsis	Septic shock
	Multiple organ dysfunction
Infections	Viral infections
	Bacterial infections

Table 1-2. Disease processes and related conditions in which reactive oxygen and nitrogen species play a causal or contributing role

4. Conclusion

ROS/RNS are able to cause damage to a variety of cellular constituents, including proteins, lipids, and nucleic acids, leading to tissue injury and disease processes. Under certain circumstances, well-controlled production of ROS/RNS also fulfils important physiological roles, including antimicrobial activity and participation in cell signaling. The biological effects of ROS/RNS, including both beneficial and detrimental effects are dependent on multiple factors. These include the chemical reactivity of the ROS/RNS, the concentrations and duration of exposure, and the types of tissues/cells that the ROS/RNS are produced from or act on. The causal or contributing role of ROS/RNS in various diseases makes it imperative to develop antioxidant-based strategies to mitigate the tissue injury and therefore protect against the disease development. The subsequent Chapter 2 is intended to provide an overview on antioxidants in biology and medicine.

5. References

- [1] Gerschman R, Gilbert DL, Nye SW, Dwyer P, Fenn WO. Oxygen poisoning and xirradiation: a mechanism in common. *Science* 1954;119:623-626.
- [2] McCord JM, Fridovich I. Superoxide dismutase. An enzymic function for erythrocuprein (hemocuprein). J. Biol. Chem. 1969;244:6049-6055.
- [3] Nathan C, Shiloh MU. Reactive oxygen and nitrogen intermediates in the relationship between mammalian hosts and microbial pathogens. *Proc. Natl. Acad. Sci. USA* 2000; 97:8841-8848.
- [4] D'Autreaux B, Toledano MB. ROS as signalling molecules: mechanisms that generate specificity in ROS homeostasis. *Nat. Rev. Mol. Cell Biol.* 2007;8:813-824.
- [5] Halliwell B. Antioxidants in human health and disease. *Annu. Rev. Nutr.* 1996;16:33-50.
- [6] Lambeth JD. NOX enzymes and the biology of reactive oxygen. *Nat. Rev. Immunol.* 2004;4:181-189.
- [7] Li Y, Zhu H, Trush MA. Detection of mitochondria-derived reactive oxygen species production by the chemilumigenic probes lucigenin and luminol. *Biochim. Biophys. Acta* 1999;1428:1-12.
- [8] Ambrosio G, Zweier JL, Duilio C, Kuppusamy P, Santoro G, Elia PP, et al. Evidence that mitochondrial respiration is a source of potentially toxic oxygen free radicals in intact rabbit hearts subjected to ischemia and reflow. J. Biol. Chem. 1993;268:18532-18541.
- [9] Wallace KB. Adriamycin-induced interference with cardiac mitochondrial calcium homeostasis. *Cardiovasc. Toxicol.* 2007;7:101-107.
- [10] Block K, Gorin Y, Abboud HE. Subcellular localization of Nox4 and regulation in diabetes. Proc. Natl. Acad. Sci. USA 2009;106:14385-14390.
- [11] Lin MT, Beal MF. Mitochondrial dysfunction and oxidative stress in neurodegenerative diseases. *Nature* 2006;443:787-795.
- [12] Lu Y, Cederbaum AI. CYP2E1 and oxidative liver injury by alcohol. *Free Radic. Biol. Med.* 2008;44:723-738.

- [13] Xia T, Li N, Nel AE. Potential health impact of nanoparticles. *Annu. Rev. Public Health* 2009;30:137-150.
- [14] Berger MM, Jia XY, Legay V, Aymard M, Tilles JG, Lina B. Nutrition- and virusinduced stress represses the expression of manganese superoxide dismutase in vitro. *Exp. Biol. Med.* (Maywood) 2004;229:843-849.
- [15] Abdalla MY, Britigan BE, Wen F, Icardi M, McCormick ML, LaBrecque DR, et al. Down-regulation of heme oxygenase-1 by hepatitis C virus infection in vivo and by the in vitro expression of hepatitis C core protein. J. Infect. Dis. 2004;190:1109-1118.
- [16] Fridovich I. Superoxide anion radical (O₂⁻), superoxide dismutases, and related matters. *J. Biol. Chem.* 1997;272:18515-18517.
- [17] Rhee SG. Cell signaling. H₂O₂, a necessary evil for cell signaling. Science 2006; 312:1882-1883.
- [18] van der Veen BS, de Winther MP, Heeringa P. Myeloperoxidase: molecular mechanisms of action and their relevance to human health and disease. *Antioxid. Redox Signal* 2009;11:2899-2937.
- [19] Pacher P, Beckman JS, Liaudet L. Nitric oxide and peroxynitrite in health and disease. *Physiol. Rev.* 2007;87:315-424.
- [20] Li S, Matthews J, Sinha A. Atmospheric hydroxyl radical production from electronically excited NO₂ and H₂O. *Science* 2008;319:1657-1660.
- [21] Murad F. Shattuck Lecture. Nitric oxide and cyclic GMP in cell signaling and drug development. *N. Engl. J. Med.* 2006;355:2003-2011.
- [22] Lundberg JO, Weitzberg E, Gladwin MT. The nitrate-nitrite-nitric oxide pathway in physiology and therapeutics. *Nat. Rev. Drug Discov.* 2008;7:156-167.
- [23] Lim P, Wuenschell GE, Holland V, Lee DH, Pfeifer GP, Rodriguez H, et al. Peroxyl radical mediated oxidative DNA base damage: implications for lipid peroxidation induced mutagenesis. *Biochemistry* 2004;43:15339-15348.
- [24] Miyamoto S, Ronsein GE, Prado FM, Uemi M, Correa TC, Toma IN, *et al.* Biological hydroperoxides and singlet molecular oxygen generation. *IUBMB Life* 2007;59:322-331.
- [25] Steinbeck MJ, Khan AU, Karnovsky MJ. Intracellular singlet oxygen generation by phagocytosing neutrophils in response to particles coated with a chemical trap. J. Biol. Chem. 1992;267:13425-13433.
- [26] Davies MJ. Singlet oxygen-mediated damage to proteins and its consequences. *Biochem. Biophys. Res. Commun.* 2003;305:761-770.
- [27] Cadet J, Ravanat JL, Martinez GR, Medeiros MH, Di Mascio P. Singlet oxygen oxidation of isolated and cellular DNA: product formation and mechanistic insights. *Photochem. Photobiol.* 2006;82:1219-1225.
- [28] Pattison DI, Davies MJ. Reactions of myeloperoxidase-derived oxidants with biological substrates: gaining chemical insight into human inflammatory diseases. *Curr. Med. Chem.* 2006;13:3271-3290.
- [29] Klebanoff SJ. Myeloperoxidase: friend and foe. J. Leukoc Biol. 2005;77:598-625.
- [30] Bergt C, Pennathur S, Fu X, Byun J, O'Brien K, McDonald TO, et al. The myeloperoxidase product hypochlorous acid oxidizes HDL in the human artery wall and impairs ABCA1-dependent cholesterol transport. Proc. Natl. Acad. Sci. USA 2004; 101:13032-13037.

- [31] Lerner RA, Eschenmoser A. Ozone in biology. Proc. Natl. Acad. Sci. USA 2003; 100:3013-3015.
- [32] Hess DT, Matsumoto A, Kim SO, Marshall HE, Stamler JS. Protein S-nitrosylation: purview and parameters. *Nat. Rev. Mol. Cell. Biol.* 2005;6:150-166.
- [33] Valko M, Leibfritz D, Moncol J, Cronin MT, Mazur M, Telser J. Free radicals and antioxidants in normal physiological functions and human disease. *Int. J. Biochem. Cell Biol.* 2007;39:44-84.

Chapter 2

Introduction to Antioxidants

Abstract

A wide variety of antioxidants have been evolved to combat reactive oxygen and nitrogen species-mediated damage in mammals. The endogenous antioxidants include proteins and non-protein compounds synthesized by cells. Mammalian cells and tissues also contain antioxidant compounds derived from the diet. In addition, a number of synthesized antioxidant compounds have become available. This chapter defines the term antioxidant and provides an overview of the various classes of antioxidants in biology and medicine. The molecular regulation of endogenous antioxidants and antioxidant-based strategies for disease intervention are also introduced in this chapter.

1. Definition and Mode of Antioxidant Action

The term antioxidant is frequently used in biology and medicine, and has been defined in various ways in the literature. One common way to define this term is that antioxidant is any substance that can prevent, reduce, or repair the reactive oxygen and nitrogen species (ROS/RNS)-induced damage to a target biomolecule [1]. There are three potential modes of action by which antioxidants protect biomolecules from ROS/RNS-induced damage. They are described as follows.

- Scavenging of ROS/RNS: Many antioxidants are able to directly scavenge ROS/RNS. Scavenging of ROS/RNS may occur through enzyme-catalyzed reactions. For example, superoxide dismutase catalyzes the dismutation of superoxide to form hydrogen peroxide and molecular oxygen (Chapter 3). Non-enzymatic antioxidants may scavenge ROS/RNS through direct chemical reactions.
- Inhibition of ROS/RNS formation: ROS/RNS are formed from various cellular sources, such as mitochondria and NAD(P)H oxidases (Section 2.2 of Chapter 1). Antioxidants may act on these cellular sources, inhibiting or preventing the formation of ROS/RNS. For example, mono-O-methylated flavanols inhibit NAD(P)H oxidases in vascular cells, leading to decreased formation of superoxide [2] (Section 2.1.2 of Chapter 23).

Removal or repair of the ROS/RNS-induced Damage: ROS/RNS can cause damage
or modifications to biomolecules. Mammalian cells are equipped with various mechanisms to remove or repair the ROS/RNS-modified biomolecules. For example,
ROS/RNS oxidize methionine residues in proteins, forming methionine sulfoxide.
Such formed methionine sulfoxide can be reduced back to methionine by methionine
sulfoxide reductase [3] (Section 2.1 of Chapter 14). Methionine sulfoxide reductase
is an important repairing system for oxidative protein damage in mammals. Mammalian cells are also equipped with enzymes for repairing oxidative damage of lipids
and nucleic acids.

2. Classification of Antioxidants

Antioxidants are generally classified into two major categories based on their sources, as listed below.

- Endogenous antioxidants
- Exogenous antioxidants

Endogenous antioxidants refer to those made by cells. Endogenous antioxidants can be further classified into two subgroups: (1) protein antioxidants and (2) non-protein antioxidants. Endogenous protein antioxidants include enzymes, such as superoxide dismutase and catalase, and non-enzymes, such as ferritin and metallothionein. Examples of endogenous non-protein antioxidants include the reduced form of glutathione and bilirubin.

Exogenous antioxidants refer to those that are not made by cells. Exogenous antioxidants can be either derived from dietary sources or synthesized in laboratories. Typical examples of exogenous antioxidants derived from the diet include vitamin C, vitamin E, and polyphenols. Antioxidant enzyme mimetics, spin traps, and cerium oxide nanoparticles are examples of synthetic antioxidants. Table 2-1 summarizes the various classes of antioxidants covered in this book.

Source	Name of Antioxidant	Chapter
Endogenous	Superoxide Dismutase	3
Protein Antioxidants	Catalase	4
Made by Cells	Glutathione-synthesizing Enzymes	5
	Glutathione Peroxidase	6
	Glutathione Reductase	7
	Glutathione S-Transferase	8
	Glutaredoxin	9
	Thioredoxin	10
	Peroxiredoxin	11
	Thioredoxin Reductase	12
	Sulfiredoxin	13

Table 2-1. Classification of antioxidants in biology and medicine (continued on next page)

Source	Name of Antioxidant	Chapter
	Methionine Sulfoxide Reductase	14
	Heme Oxygenase	15
	NAD(P)H:Quinone Oxidoreductase	16
	Paraoxonase	17
	Ferritin	18
	Metallothionein	18
Endogenous	Glutathione	5
Non-protein Antioxidants	Bilirubin	19
Made by Cells	Coenzyme Q	19
	Estrogens	19
	α-Lipoic Acid	19
	Melatonin	19
	Pyruvate	19
	Uric Acid	19
Exogenous	Vitamin C	20
Antioxidants Derived	Vitamin E	21
from the Diet	Carotenoids	22
	Phenolic Compounds	23
Exogenous	Antioxidant Enzyme Mimetics	24
Antioxidants Synthesized	Glutathione Precursors	25
in Laboratories	Spin Traps	26
	Nanomaterials	27

3. Antioxidant Gene Regulation

Regulation of antioxidant gene expression is an integral part of cellular defenses against ROS/RNS-induced damage in aerobic organisms, including mammals. Multiple mechanisms are involved in the regulation of mammalian antioxidant gene expression. At the transcriptional level, nuclear factor-erythroid 2 p45-related factor 2 (Nrf2) plays a central role in the regulation of a series of antioxidant genes in mammals. Under certain conditions, other transcription factors may also be involved in antioxidant gene regulation. These include Nrf1, Nrf3, nuclear factor kappa-B (NF- κ B), and AP-1 [4-6]. This section focuses on discussing Nrf2 as a central regulator of mammalian antioxidant gene expression and its role in protection against ROS/RNS-mediated biological damage.

3.1. Regulation of Antioxidant Gene Expression by Nrf2 Signaling

3.1.1. Historical Overview

Early studies during the 1980's showed that administration of phenolic chemicals to experimental animals resulted in simultaneous induction of multiple antioxidants, including glutathione, glutathione S-transferase, and NAD(P)H:quinone oxidoreductase 1, suggesting a possible existence of a central regulator for the above cellular defenses. Studies in the early 1990's demonstrated a crucial role for a cis-acting element, named antioxidant response ele-

ment (ARE) in the regulation of antioxidant gene expression [7]. In 1997, M. Yamomoto and coworkers identified Nrf2 as a critical transcription factor that interacts with the ARE, leading to increased expression of various antioxidant and cytoprotective genes [8]. Subsequent studies from multiple research groups have established that Nrf2 plays a central role in regulating both constitutive and inducible expression of a wide variety of antioxidant genes in mammalian systems [9].



Figure 2-1. Schematic illustration of Keap1-Nrf2-ARE signaling. See text (Section 3.1.2) for detailed description.

3.1.2. Keap1-Nrf2-ARE Signaling

Nrf2 normally resides in the cytosolic compartment through association with a cytosolic actin-binding protein, Keap1 (Kelch-like ECH-associated protein 1). Keap1 plays a central role in controlling Nrf2 activity [10]. Keap1 exists as dimers inside the cells and functions as a substrate linker protein for interaction of Cul3/Rbx1-based E3-ubiquitin ligase complex with Nrf2. This protein-protein interaction leads to continuous ubiquitination of Nrf2 and its proteasomal degradation (Figure 2-1). Hence, the continuous degradation of Nrf2 under basal conditions keeps the Nrf2 level low and thereby the low basal levels of Nrf2-regulated anti-oxidants. When the cells are exposed to moderate amounts of ROS/RNS or chemical inducers, Nrf2 dissociates from Keap1, becomes stabilized, and translocates into the nuclei. Inside the nuclei, Nrf2 interacts with other protein factors, including small Maf (sMaf), and binds to

ARE, resulting in increased transcription of antioxidant genes. The increased production of antioxidants helps detoxify the ROS/RNS and may even render cells more resistant to higher levels of ROS/RNS. Thus, Nrf2-dependent upregulation of antioxidants by moderate levels of ROS/RNS can be viewed as an adaptive response to oxidative stress. Two potential mechanisms have been proposed to explain the dissociation of Nrf2 from Keap1 under stress conditions, which are summarized below [10, 11].

- Oxidation of cysteine residues of Keap1: Keap1 is a cysteine-rich protein, and the cysteine residues may serve as redox sensors for ROS/RNS or chemical inducers. Modifications of these cysteine sulfhydryl groups on Keap1 protein is thought to cause dissociation of Nrf2 from Keap1.
- Phosphorylation of Nrf2: Several protein kinases, including protein kinase C, mitogen-activated protein kinase, and phosphatidylinositol 3-kinase have been suggested to phosphorylate Nrf2, leading to dissociation of Nrf2 from Keap1.

3.1.3. Nrf2-regulated Antioxidants

Numerous in vitro and in vivo studies demonstrate that Nrf2 plays an indispensable role in the regulation of a wide variety of antioxidant gene expression. The antioxidant enzymes known to be regulated by Nrf2 through an ARE-driven mechanism in mammalian cells include ferritin heavy subunit, both catalytic and modifier subunits of γ -glutamylcysteine ligase, glutathione reductase, heme oxygenase-1, peroxiredoxin-1 and -6, multiple isozymes of glutathione S-transferase, NAD(P)H:quinone oxidoreductase 1, NRH:quinone oxidoreductase 2, sulfiredoxin, Cu,Zn superoxide dismutase, thioredoxin, and thioredoxin reductase-1. In addition to the antioxidant genes, Nrf2 signaling is also found to affect the expression of numerous other genes whose products are involved in various cellular processes, such as generation of NADPH and metabolism of xenobiotics [10, 11].

3.2. Nrf2-dependent Induction of Antioxidants and Protection against Oxidative Stress

It has been established that Nrf2 plays a crucial role in mediating the induction of cellular antioxidants by many chemical inducers. Several classes of chemicals are commonly used for induction of cellular antioxidants to protect against disease conditions associated with oxidative stress. These compounds, also known as chemoprotective agents include the following four chemical classes.

- Phenolic compounds
- Dithiolethiones
- Isothiocyanates
- Triterpenoids

The above chemical agents have been shown to play a protective role in a variety of animal models of disease pathophysiology, including oxidative tissue injury, inflammatory disorders, and chemical carcinogenesis [11]. Studies using Nrf2-knockout mouse model demonstrate that the protective effects of the above chemoprotective agents occur primarily via activation of Nrf2 and the subsequent induction of antioxidants as well as other cytoprotective proteins. As coordinated actions of various antioxidants are essential for efficient detoxification of ROS/RNS, the simultaneous upregulation of Nrf2-dependent antioxidants by chemoprotective agents may represent an effective strategy for the intervention of disease conditions involving an oxidative stress component.

4. Antioxidant-based Disease Intervention

The critical involvement of ROS/RNS in a number of disease conditions makes it imperative to develop strategies to control these reactive species-mediated pathophysiological processes. Intervention of ROS/RNS-associated disease conditions could be potentially achieved through various approaches, among which is the antioxidant-based intervention.

4.1. Overall Strategies of Antioxidant-based Disease Intervention

Antioxidant-based intervention of ROS/RNS-associated disease conditions may include two overall strategies, as described below.

- Administration of exogenous antioxidant compounds to scavenge ROS/RNS associated with disease conditions: For disease conditions caused by increased levels of ROS/RNS, scavenging of these reactive species by exogenously administered antioxidant compounds may prevent or diminish the tissue injury caused by the ROS/RNS. In this context, both diet-derived antioxidant compounds (e.g., vitamin C, vitamin E, polyphenols) and synthetic antioxidant chemicals (e.g., antioxidant enzyme mimetics) have been investigated extensively for the intervention of various disease processes in animal models and human subjects (Chapters 20-26).
- Administration of antioxidant enzyme inducers (e.g., chemoprotective agents) to augment endogenous cellular antioxidant defenses: As noted above, the chemical inducibility of endogenous antioxidants makes it possible for protecting against ROS/RNS-mediated injury by administration of antioxidant-inducing agents. Induction of cellular antioxidants and other cytoprotective enzymes by chemoprotective agents has been demonstrated to be a valid approach to the intervention of diseases involving an oxidative stress mechanism in animal models. This approach has also been applied for cancer chemoprotection in humans [12]. An alternative approach to increasing cellular antioxidant defenses is through transgenic overexpression of the individual antioxidant proteins. This is usually done in experimental animals.

4.2. Animal Studies on Antioxidant-based Disease Intervention

Animal studies provide important information on the safety and efficacy as well as mechanisms of actions for drugs that are used for the prevention and treatment of human diseases. This is also true for the development of antioxidant-based strategies for the intervention of disease conditions involving an ROS/RNS-mediated pathophysiological component. Three approaches as outlined below are commonly used to study antioxidant intervention of diseases in experimental animals.

- Administration of exogenous antioxidant compounds
- Transgenic overexpression of antioxidant proteins
- Induction of endogenous antioxidants by chemoprotective agents

Transgenic approach provides the most convincing information regarding the effects of overexpression of a particular antioxidant protein on a disease condition. This also provides the strongest evidence for the involvement of a particular ROS/RNS in disease pathophysiology [13]. Using exogenous antioxidant compounds with selective ROS/RNS-scavenging properties may also yield important insights into the potential role of ROS/RNS in disease pathophysiology. In contrast, the effectiveness of chemical inducers of antioxidants in protecting against a disease condition may result from the simultaneous induction of multiple endogenous antioxidants. Indeed, chemical inducers usually upregulate a series of different antioxidant enzymes via activating Nrf2 signaling [14]. As noted earlier, Nrf2 acts as a central regulator of a wide variety of mammalian antioxidants and other cytoprotective factors. Figure 2-2 summarizes the major approaches to augmenting tissue antioxidant defenses for the intervention of disease conditions with ROS/RNS as etiological or contributing factors.



Figure 2-2. Major experimental approaches to augmenting tissue antioxidant defenses for the intervention of disease conditions that involve an oxidative stress mechanism. See text (Section 4.2) for detailed description.
4.3. Human Studies on Antioxidant-based Disease Intervention

The three approaches used in experimental animal studies, as discussed in Section 4.2 could also be applied to human studies though at present the use of genetic approach (gene therapy) is much limited in humans. In this regard, a more practical antioxidant-based approach in human studies involves the use of exogenous antioxidant compounds or chemoprotective agents capable of inducing endogenous antioxidants. In this context, randomized double-blind, placebo-controlled clinical trials are usually conducted to determine the efficacy of antioxidant-based strategies in the intervention of human diseases that are believed to have an oxidative stress component. The interventional clinical trials can be either preventive or therapeutic.

5. Caveats on Antioxidants and Antioxidant-based Disease Intervention

5.1. Multiple Biological Activities of Antioxidants

Many antioxidants have well-established ability to scavenge ROS/RNS, inhibit their formation, or repair the damage caused by ROS/RNS. However, it is important to note that in addition to their effects on ROS/RNS antioxidants, including both protein and non-protein antioxidants may also exert other biological effects irrelevant to their antioxidant activities. For example, the antioxidant enzyme catalase decomposes hydrogen peroxide to form water and molecular oxygen, a well-known antioxidant activity of catalase. This antioxidant enzyme also exhibits peroxidase and oxidase activities toward a number of substrates, including alcohols [15] (Section 2.1 of Chapter 4). Another example is α -tocopherol, which is known to inhibit lipid peroxidation by scavenging lipid peroxyl radical. α -Tocopherol has been shown to also exert non-antioxidant effects, such as inhibition of protein kinases [16] (Section 2.3 of Chapter 21). Thus, the response of a biological system to a particular antioxidant may result from: (1) the antioxidative activity of the compound only, (2) the non-antioxidative activity of the compound only, or (3) the combined antioxidative and non-antioxidative activities. In antioxidant biology and medicine, another important consideration is that many antioxidants, such as glutathione and glutathione S-transferase are also important factors in the detoxification of electrophilic species, including reactive aldehydes [17]. As described in Section 2.4 of Chapter 1, reactive aldehydes are produced from lipid peroxidation elicited by ROS/RNS, and they can cause damage to biomolecules.

5.2. Genetic Manipulations of Antioxidants

Transgenic overexpression or knockout of particular antioxidant genes is frequently used to study the biological activities of endogenous antioxidant proteins in experimental animals. While these genetic animal models provide the most convincing evidence regarding the biological functions of particular antioxidant proteins, interpretation of the data should be done with caution. This is because deletion of one antioxidant gene may lead to compensatory upregulation of others. Likewise, overexpression of one antioxidant gene may also lead to downregulation of others. Thus, changes in the phenotypes of these genetically manipulated animals likely result from the combined actions of all the antioxidants whose expression is altered due to the genetic manipulation of a single antioxidant gene. Another notion regarding transgenic overexpression of antioxidant genes is that the levels of the overexpressed antioxidant proteins are usually several or even over ten-fold higher than the basal physiological levels. The physiological significance of such dramatically overexpressed antioxidant proteins thus needs to be evaluated with caution. Regardless of the caveats, genetic manipulation of antioxidant genes in experimental animals continues to serve as an important approach to understanding the biology of endogenous antioxidants.

6. Conclusion

Due to the detrimental effects of ROS/RNS, aerobic organisms, including mammals have evolved a number of endogenous antioxidants to counteract ROS/RNS-mediated biological damage. In addition, various exogenous antioxidant compounds have been found in dietary sources or synthesized in laboratories. The complementary effects of diverse antioxidants make it possible to effectively detoxify ROS/RNS and protect against disease conditions involving an oxidative stress mechanism. In this context, coordinated regulation of gene expression of a series of antioxidants primarily through Nrf2 signaling is an integral part of mammalian antioxidant defense mechanisms against oxidative stress. As accumulating evidence suggests a critical role for ROS/RNS in the pathophysiological processes of a wide variety of diseases, it is imperative to develop more effective antioxidant-based strategies for disease intervention. Development of such strategies relies on a profound understanding of the basic biology of antioxidants and their role in health and disease. The subsequent chapters are devoted to the discussion of the various kinds of antioxidants in biology and medicine.

7. References

- [1] Halliwell B. Biochemistry of oxidative stress. *Biochem. Soc. Trans.* 2007;35:1147-1150.
- [2] Steffen Y, Gruber C, Schewe T, Sies H. Mono-O-methylated flavanols and other flavonoids as inhibitors of endothelial NADPH oxidase. *Arch. Biochem. Biophys.* 2008; 469:209-219.
- [3] Kim HY, Gladyshev VN. Methionine sulfoxide reductases: selenoprotein forms and roles in antioxidant protein repair in mammals. *Biochem. J.* 2007;407:321-329.
- [4] Ohtsuji M, Katsuoka F, Kobayashi A, Aburatani H, Hayes JD, Yamamoto M. Nrf1 and Nrf2 play distinct roles in activation of antioxidant response element-dependent genes. *J. Biol. Chem.* 2008;283:33554-33562.
- [5] Sankaranarayanan K, Jaiswal AK. Nrf3 negatively regulates antioxidant-response element-mediated expression and antioxidant induction of NAD(P)H:quinone oxidoreductase1 gene. J. Biol. Chem. 2004;279:50810-50817.
- [6] Yang H, Magilnick N, Lee C, Kalmaz D, Ou X, Chan JY, et al. Nrf1 and Nrf2 regulate

rat glutamate-cysteine ligase catalytic subunit transcription indirectly via NF-kappaB and AP-1. *Mol. Cell. Biol.* 2005;25:5933-5946.

- [7] Rushmore TH, Morton MR, Pickett CB. The antioxidant responsive element: activation by oxidative stress and identification of the DNA consensus sequence required for functional activity. J. Biol. Chem. 1991;266:11632-11639.
- [8] Itoh K, Chiba T, Takahashi S, Ishii T, Igarashi K, Katoh Y, *et al.* An Nrf2/small Maf heterodimer mediates the induction of phase II detoxifying enzyme genes through antioxidant response elements. *Biochem. Biophys. Res. Commun.* 1997;236:313-322.
- [9] Maher J, Yamamoto M. The rise of antioxidant signaling-the evolution and hormetic actions of Nrf2. *Toxicol. Appl. Pharmacol.* 2010;244:4-15.
- [10] Hayes JD, McMahon M. NRF2 and KEAP1 mutations: permanent activation of an adaptive response in cancer. *Trends Biochem. Sci.* 2009;34:176-188.
- [11] Kensler TW, Wakabayashi N. Nrf2: friend or foe for chemoprevention? *Carcinogene-sis* 2010;31:90-99.
- [12] Kensler TW, Egner PA, Wang JB, Zhu YR, Zhang BC, Lu PX, et al. Chemoprevention of hepatocellular carcinoma in aflatoxin endemic areas. *Gastroenterology* 2004;127: S310-318.
- [13] Levonen AL, Vahakangas E, Koponen JK, Yla-Herttuala S. Antioxidant gene therapy for cardiovascular disease: current status and future perspectives. *Circulation* 2008; 117:2142-2150.
- [14] Zhu H, Jia Z, Misra BR, Zhang L, Cao Z, Yamamoto M, *et al.* Nuclear factor E2related factor 2-dependent myocardiac cytoprotection against oxidative and electrophilic stress. *Cardiovasc. Toxicol.* 2008;8:71-85.
- [15] Vetrano AM, Heck DE, Mariano TM, Mishin V, Laskin DL, Laskin JD. Characterization of the oxidase activity in mammalian catalase. J. Biol. Chem. 2005; 280:35372-35381.
- [16] Galli F, Azzi A. Present trends in vitamin E research. *Biofactors* 2010;36:33-42.
- [17] Hayes JD, Flanagan JU, Jowsey IR. Glutathione transferases. Annu. Rev. Pharmacol. Toxicol. 2005;45:51-88.

SECTION II ANTIOXIDANTS SYNTHESIZED BY CELLS

Aerobic organisms, including mammals have evolved a wide variety of antioxidants to combat oxidative stress. These endogenous antioxidants synthesized by cells include both proteins and non-proteins. The protein antioxidants further comprise enzymes and non-enzymes. Superoxide dismutase and catalase, as mentioned in previous chapters are typical examples of antioxidant enzymes synthesized by cells. Examples of non-enzyme protein antioxidants include ferritin and metallothionein. A number of non-protein antioxidants are also synthesized by mammalian cells, which include glutathione, bilirubin, and coenzyme Q. These various endogenous antioxidants, either proteins or non-proteins, have unique antioxidative properties and often act coordinately to detoxify reactive oxygen and nitrogen species. In addition, they also possess other novel biological activities. The seventeen chapters in Section II are devoted to the discussion of each of the major endogenous antioxidants with respect to their basic biology and implications in health and disease.

Chapter 3

Superoxide Dismutase

Abstract

Since its discovery in 1969, superoxide dismutase (SOD) has been extensively studied in biology and medicine. There are three isozymes of SOD in mammals, namely, copper, zinc SOD, manganese SOD, and extracellular SOD. All three isozymes catalyze dismutation of superoxide to hydrogen peroxide and molecular oxygen with a similar reaction rate constant. SOD isozymes play an important role in protecting against various disease processes in animal models, including cardiovascular diseases, diabetes, neurodegeneration, and cancer. This notion from another angle demonstrates a crucial involvement of superoxide in disease pathophysiology. Studies in human subjects also provide evidence indicating protective effects of SODs in certain disease conditions.

1. Overview

1.1. Definition and History

The term superoxide dismutase (SOD) refers to a family of enzymes that catalyze the dismutation of superoxide to hydrogen peroxide and molecular oxygen. There are three iso-zymes of SOD in mammals, as listed below.

- Copper, zinc superoxide dismutase (Cu,ZnSOD or SOD1)
- Manganese superoxide dismutase (MnSOD or SOD2)
- Extracellular superoxide dismutase (ECSOD or SOD3)

Here the term isozyme (also known as isoenzyme or isoform) refers to enzymes that differ in amino acid sequence but catalyze essentially the same biochemical reaction.

As introduced in Chapter 1, the enzymatic activity of Cu,ZnSOD was first identified by J.M. McCord and I. Fridovich in 1969 [1]. But the protein was discovered before. T. Mann and D. Keilin had purified the protein 30 years earlier from bovine blood and liver as copperbinding protein of unknown function. The protein was then named erythrocuprein, hepatocuprein, or cytocuprein. Superoxide, the substrate of SOD was discovered by L. Pauling in the 1930's, and it was not known then if this free radical could be produced biologically. In 1969, P.F. Knowles and coworkers showed that the enzyme xanthine oxidase could produce superoxide. J.M. McCord and I. Fridovich subsequently demonstrated that the copper protein purified by T. Mann and D. Keilin could catalytically eliminate the superoxide initially discovered by L. Pauling. This copper-containing protein is known as Cu,ZnSOD today. In 1970 and 1973, I. Fridovich and coworkers further discovered MnSOD and FeSOD (FeSOD is not present in mammals), respectively. The third SOD in mammals, namely, ECSOD, was discovered by S.L. Marklund and coworkers in 1982 [2].

1.2. Basic Characteristics

In mammals, Cu,ZnSOD is a homodimer with a molecular mass of 32 kDa. Both MnSOD and ECSOD are homotetramers with a molecular mass of 86-88 and 135 kDa, respectively. ECSOD also contains copper and zinc. Cu,ZnSOD is present in cytosol. It is also found in nuclei and mitochondrial intermembrane space. MnSOD exists in mitochondrial matrix. ECSOD is associated with plasma membrane or present in extracellular space. The human genes for Cu,ZnSOD, MnSOD, and ECSOD are localized on chromosomes 21q22, 6q25, and 4q21, respectively. Table 3-1 summarizes the basic characteristics of the three SOD isozymes in mammals.

Isozyme	Molecular	Assembly	Metal	Cellular	Chromosomal
	Mass	of Subunits	Ions	Location	Localization
Cu,ZnSOD	32 kDa	Homodimer	Cu and Zn	Cytosol, nuclei, mitochondrial intermembrane space	21q22
MnSOD	86-88 kDa	Homotetramer	Mn	Mitochondrial matrix	6q25
ECSOD	135 kDa	Homotetramer	Cu and Zn	Plasma mem- brane, extracel- lular space	4q21

Table 3-1. Basic Characteristics of the three isozymes of mammalian superoxide dismutase

2. Biochemistry and Molecular Regulation

2.1. Biochemistry

All three isozymes of SOD catalyze dismutation of superoxide (O_2 ⁻) to form hydrogen peroxide (H_2O_2) and molecular oxygen with a similar reaction rate constant of ~1.6 x 10⁹ M⁻¹s⁻¹ (Reaction 1).

$$2O_2 + 2H^+ \xrightarrow{\text{SOD}} H_2O_2 + O_2 \tag{1}$$

The reaction rate constant of SOD-catalyzed dismutation of superoxide is over three orders of magnitude greater than that of spontaneous dismutation, which is $\sim 5 \times 10^5 \text{ M}^{-1}\text{s}^{-1}$ at pH 7.0. Here the term dismutation refers to a chemical reaction in which the same reactant is both oxidized and reduced. In the dismutation reaction of superoxide, one superoxide is oxidized to molecular oxygen and another is reduced to hydrogen peroxide.

The overall mechanism by which SOD functions has been called "ping-pong" mechanism as it involves the sequential reduction and oxidation of the metal center, with the concomitant oxidation and reduction of superoxide (Reactions 2 and 3, where M stands for the redox metal ion of the SOD).

$$M^{n+}-SOD + O_2^{\cdot} \longrightarrow M^{(n-1)+}-SOD + O_2$$
(2)

$$M^{(n-1)+}-SOD + O_2 + 2H^+ \longrightarrow M^{n+}-SOD + H_2O_2$$
(3)

Cu,ZnSOD also exhibits peroxidase activity. Interaction of hydrogen peroxide with Cu,ZnSOD in the presence of bicarbonate yields carbonate radical [3]. As stated in Section 2.3.4 of Chapter 1, carbonate radical is a potent oxidant capable of inducing damage to bio-molecules in vitro. In addition, Cu,ZnSOD catalyzes tyrosine nitration by peroxynitrite [4]. However, the in vivo significance of these reactions remains to be elucidated.

2.2. Molecular Regulation

A number of transcription factors have been shown to play a role in regulating constitutive or inducible expression of SOD genes. These transcription factors include NF- κ B, AP-1, AP-2, SP1, and CCAAT-enhancer-binding protein (C/EBP) [5]. Recently, it has been shown that Nrf2 regulates Cu,ZnSOD expression via an antioxidant response element-driven mechanism [6]. In addition to the regulation by transcription factors, SOD genes are also subject to epigenetic regulation, such as hypermethylation, leading to epigenetic silencing. The term epigenetic regulation refers to heritable changes at the level of gene expression not related to the underlying DNA sequence.

SOD expression is regulated not only at transcriptional level, but also at posttranscriptional level via changes in mRNA stability, mRNA translation, and post-translational modifications [7]. Alterations in the regulation of SOD expression at these various levels may change tissue SOD activity and thereby modulate the susceptibility to disease pathophysiology involving increased formation of superoxide.

3. Role in Health and Disease

3.1. Animal Studies

3.1.1. Experimental Approaches

Various approaches have been used to study the role of SODs in health and disease in experimental animals. These approaches include the following.

- Gene knockout
- Transgenic overexpression or viral vector-mediated gene delivery
- Administration of exogenous SODs or SOD derivatives
- Pharmacological inhibition or induction of endogenous SODs

Gene knockout and transgenic overexpression models provide the most convincing evidence on the biological activities of SODs. A gene knockout mouse is a mouse that has had one or more specific genes artificially deleted from its genome. For a target gene, if both copies of the gene are deleted (as indicated by the superscript ^{-/-}), the animal is called homozygous. On the other hand, if only one copy of the gene is deleted (as indicated by the superscript ^{+/-}), it is called heterozygous. In contrast to gene knockout, transgenic overexpression animal models are used to achieve selective overexpression of a particular antioxidant protein. Transgenic overexpression animals are created by introducing new DNA sequences into the germline.

Direct administration of antioxidant enzymes or proteins to animals is another way to study the biological activities of the antioxidants. However, limited bioavailability is a major drawback of such an approach. In this regard, antioxidant enzyme mimetics have been commonly used owing to their favorable bioavailability (see Chapter 24 for discussion of antioxidant enzyme mimetics). Here the term bioavailability in pharmacology refers to the fraction of an administered dose of unchanged drug that reaches the systemic circulation.

In addition to genetic approaches, pharmacological inhibition or induction of antioxidant enzymes is also used to study their biological activities. For example, diethyldithiocarbamate is often utilized as an inhibitor of Cu,ZnSOD to study its biological activities in both cell cultures and experimental animals. However, lack of specificity is a major limitation associated with the use of pharmacological inhibitors, including diethyldithiocarbamate. A number of chemical agents, such as polyphenolic compounds have been shown to induce SODs in cell cultures and experimental animals. Again, the limitation is the lack of selectivity as these chemical inducers usually cause upregulation of a battery of antioxidant genes. In addition, they may also exert non-antioxidant effects.

3.1.2. Role in Cardiovascular Diseases

Both gene knockout and transgenic overexpression animal studies demonstrate a critical protective role for all three isozymes of SOD in cardiovascular disorders, including hypertension, atherosclerosis, myocardial ischemia-reperfusion injury, cardiac hypertrophy, heart failure, and drug/xenobiotic-induced cardiovascular complications [8, 9]. With the above animal models of human cardiovascular diseases, knockout of SODs leads to exacerbation, whereas transgenic overexpression of SODs attenuates the pathophysiological processes. It is noteworthy that while homozygous knockout of either Cu,ZnSOD or ECSOD does not affect animal survival, deletion of MnSOD in mice is embryonically lethal or causes early neonatal death due primarily to dilated cardiomyopathy [10].

3.1.3. Role in Diabetes

SODs play an important role in protecting against various complications of diabetes in animal models. Overexpression of Cu,ZnSOD or MnSOD attenuates diabetic renal injury, neuropathy, cardiomyopathy, and retinopathy [11-14]. Conversely, diabetes-induced renal

injury and cataract are exacerbated in Cu,ZnSOD-deficient mice. Cu,ZnSOD deficiency also potentiates diabetic embryopathy in experimental animals, indicating an involvement of superoxide in hyperglycemia-induced adverse effects on fetus. In addition to protecting against diabetic complications, Cu,ZnSOD and MnSOD may also prevent the genesis of diabetes. Pancreatic β -cell-selective overexpression of Cu,ZnSOD in mice protects the β -cells from oxidative stress and enhances the resistance of the animals to the development of alloxan-induced diabetes. Alloxan is a compound capable of causing the pancreatic β -cell destruction and thereby the development of type 1 diabetes in experimental animals. Adenoviral vector-mediated overexpression of MnSOD in the pancreatic islets extends islet graft function in a mouse model of autoimmunity-induced diabetes. These experimental findings suggest a critical role for both Cu,ZnSOD and MnSOD in protecting against the pancreatic β -cell injury and the development of diabetes. In contrast, overexpression of ECSOD in the pancreatic islets has been reported to provide no effects on the development of diabetes in nonobese diabetic mice [15].

3.1.4. Role in Neurological Diseases

SOD overexpression via genetic approaches provides neuroprotection in various animal models of neurological disorders, including ischemic brain injury, Parkinson's disease, and Alzheimer's disease [16, 17]. Intraperitoneal injection of Tat-Cu,ZnSOD fusion protein also protects against ischemic brain injury in mice [18]. Similarly, various SOD mimetics are neuroprotective in animal models of neurological diseases (Chapter 24).

Gene knockout animal models have provided important insights into the protective role of SODs in neurodegeneration. Dilated cardiomyopathy, mitochondrial abnormalities, hepatic lipid accumulation, and early neonatal death have been reported to occur in homozygous MnSOD knockout (MnSOD^{-/-}) mice. Treatment with the SOD mimetic Mn(III) tetrakis (4-benzoic acid) porphyrin (MnTBAP; Chapter 24) rescues the MnSOD^{-/-} mice from the above systemic pathological changes and dramatically prolongs their survival. These animals instead develop pronounced neurodegeneration, which is believed to result from excessive mitochondrial production of superoxide in neuronal cells due to the inability of MnTBAP to cross the blood-brain barrier [19]. These observations are consistent with the notion that mitochondria are a major source of cellular superoxide, and uncontrolled excessive formation of superoxide leads to cell and tissue injury.

In contrast to MnSOD, deletion of Cu,ZnSOD in animals causes milder phenotypes. Homozygous Cu,ZnSOD knockout (Cu,ZnSOD^{-/-}) mice develop normally and show no overt abnormal phenotypes when they reach adulthood except that females show reduced fertility. However, these mice exhibit increased vulnerability to motor neuron loss after axonal injury, suggesting that the neuroprotective effects of Cu,ZnSOD may become evident only when neurons are under stress or exposed to insults [20].

3.1.5. Role in Pulmonary Diseases

A beneficial role for all three isozymes of SOD in pulmonary hyperoxic injury has been demonstrated in experimental animals via both gene knockout and transgenic overexpression [21, 22]. SODs attenuate pulmonary inflammatory responses, oxidative stress, and tissue remodeling following hyperoxia. Overexpression of SODs also results in amelioration of pulmonary injury induced by drugs (e.g., bleomycin), radiation, and environmental chemicals or

dusts (e.g., tobacco smoke, asbestos, air pollutants). Administration of exogenous SOD protein via intratracheal instillation or inhalation protects against oxidative and inflammatory lung injury in experimental animals. Conversely, targeted disruption of SODs aggravates the pathophysiological processes of the pulmonary disorders.

3.1.6. Role in Hepatic and Gastrointestinal Diseases

Multiple animal studies demonstrate a protective role for SODs in hepatic disorders, including liver ischemia-reperfusion injury, graft dysfunction after liver transplantation, nonalcoholic fatty liver disease, and liver injury induced by alcohol and other drugs/xenobiotics. Viral vector-mediated delivery of ECSOD gene to liver tissue suppresses ischemiareperfusion-induced hepatic injury in mice. Viral vector-mediated delivery of either Cu,ZnSOD or MnSOD, but not ECSOD increases animal survival and attenuates graft dysfunction after transplantation of fatty livers in rats. These protective effects may result from the SOD-mediated suppression of NF-KB activation and proinflammatory cytokine production [23]. It is also reported that Cu,ZnSOD deficiency causes ApoB degradation and increases hepatic lipid accumulation in mice [24]. Homozygous MnSOD-deficient mice die within the first 10 days of life with dilated cardiomyopathy and accumulation of lipid in liver [10]. These observations suggest a critical pathophysiological role for superoxide and oxidative stress in hepatic lipid dysregulation and the possible development of nonalcoholic fatty liver disease.

Overexpression of either Cu,ZnSOD or MnSOD in mice inhibits alcohol-induced liver injury, whereas knockout of Cu,ZnSOD aggravates the liver damage [25, 26]. Overexpression of Cu,ZnSOD also protects against carbon tertrachloride-elicited hepatotoxicity in mice. However, mice deficient in Cu,ZnSOD are resistance to acetaminophen-induced hepatotoxicity, suggesting a possible pro-oxidant role for the physiological level of Cu,ZnSOD activity in acetaminophen-mediated liver injury [27].

A protective role for SODs in gastrointestinal disorders has been demonstrated in several animal models. For example, transgenic overexpression of Cu,ZnSOD in mice protects tissue from neutrophil infiltration and lipid peroxidation during intestinal ischemia-reperfusion. Oral coadministration of Cu,ZnSOD and catalase is reported to attenuate stress-induced gastric mucosal lesions in rats. In an animal model of chemically-induced acute colitis, transgenic overexpression of Cu,ZnSOD results in amelioration of colonic inflammation and improves animal survival [28].

3.1.7. Role in Renal Diseases

Either viral vector-mediated or transgenic overexpression of SODs is reported to attenuate renal ischemia-reperfusion injury in animal models. Administration of lecithinized SOD reduces ischemia-induced chronic allograft dysfunction in rats. Targeting a synthetic cationic SOD to renal proximal tubular cells inhibits oxidative stress and nephrotoxicity of cisplatin and increases the survival of cancer-bearing mice [29]. Cisplatin is an anticancer drug that may cause severe renal injury. Viral delivery of Cu,ZnSOD gene reduces cyclosporine Ainduced free radical formation and nephrotoxicity. Cyclosporine A is a commonly used immunosuppressive agent whose clinical use in organ transplantation is hampered by its nephrotoxicity. On the other hand, Cu,ZnSOD deficiency causes salt sensitivity and aggravates hypertension in hydronephrosis in mice [30]. Similarly, mice with heterozygous knockout of MnSOD (MnSOD^{+/-}) develop salt-sensitive hypertension and accelerated renal senescence [31]. Together, these findings from both transgenic overexpression and gene knockout animal models indicate a critical involvement of superoxide in various kidney pathophysiological processes, and also point to the feasibility of using SOD-based modalities for the intervention of renal diseases.

3.1.8. Role in Skin Diseases

Oxidative stress is involved in a variety of skin disorders, including ultraviolet lightinduced injury, skin carcinogenesis, and skin aging [32]. A number of animal studies demonstrate an important role for SODs in protecting against the above skin pathophysiological processes. It is also reported that treatment with Cu,ZnSOD ameliorates skin ischemic injury in experimental animals [33].

3.1.9. Role in Cancer

Oxidative stress is implicated in multistage carcinogenesis, including initiation, promotion, and progression. It is not surprising that SODs are also found to have a beneficial effect in experimental carcinogenesis. Many studies with gene knockout models show that longterm reduction of either Cu,ZnSOD or MnSOD results in increased DNA damage, mutations, and higher incidence of cancer development, including hepatocarcinogenesis [34, 35]. Recently, it has been reported that overexpression of ECSOD attenuates heparanase expression and inhibits breast carcinoma cell growth and invasion [36]. As mentioned above (Section 3.1.8), SOD overexpression also inhibits chemically-induced skin carcinogenesis in experimental animals.

The protective effects of SODs in cancer development have also been investigated via examining their involvement in cancer cell growth in vitro. However, the results from these in vitro studies are inconsistent, with some studies showing SODs as cancer cell survival factors, and others indicating the opposite [36, 37]. Regardless of these in vitro observations, studies using gene knockout and transgenic animal models have provided convincing evidence supporting a critical role for SODs in protecting against carcinogenesis.

3.1.10. Role in Other Diseases and Conditions

SODs may be important protectors in anemia. Deficiency of either Cu,ZnSOD or MnSOD in mice causes oxidative stress in red blood cells and hemolytic anemia [38, 39]. Studies also demonstrate a protective role for SODs in aging and aging-related cognitive decline and muscle atrophy [40]. Moreover, gene transfer of ECSOD to the penis reduces superoxide levels and improves erectile function in aged or diabetic rats [41, 42].

As discussed above, substantial experimental evidence demonstrates an important role for SODs in protecting against various pathophysiological conditions. However, it has also been reported that SODs may exert unfavorable effects under certain experimental conditions. One example is the aforementioned role for Cu,ZnSOD in potentiating acetaminophen-induced hepatotoxicity [27]. As another example, Cu,ZnSOD^{-/-} mice are shown to produce less amounts of caspase-2-dependent cytokines and be less susceptible to lipopolysaccharide-induced endotoxic shock compared with wild-type littermates [43]. These observations point to the complicated involvement of superoxide and SODs in physiology and disease. Nevertheless, animal studies continue providing important insights into the biological activities of

antioxidant enzymes and lay a foundation for investigations on their potential health benefits in human subjects.

3.2. Human Studies and Clinical Perspectives

3.2.1. Study Approaches

Since the discovery of Cu,ZnSOD more than four decades ago, there have been extensive studies on the biological activities of SODs in human health and disease. As outlined below, two major approaches are employed to study the biological activities of SODs as well as other antioxidant enzymes in human subjects.

- Genetic variation studies: Gene mutations and polymorphisms of SODs have been studied in human populations in relation to disease conditions. Here the term polymorphism refers to a genetic variation in the DNA sequence with a measurable frequency of detection above 1%. Any polymorphism in the chromosomal DNA (coding and non-coding) can affect mRNA processing, maturation, and translation as a result of changes in conformation. A polymorphism in an antioxidant gene may result in a decrease, an increase, or no change in the activity/level of the protein encoded by the gene.
- Clinical interventional studies (trials): Recombinant SODs as well as synthetic SOD mimetics have been investigated for their potential use in the intervention of certain human disease conditions.

Another approach of clinical potential in humans is gene therapy. Gene therapy with the use of antioxidant genes is emerging as a promising approach for select disease conditions and is an especially lucrative option for patient groups not suitable for conventional therapies [8]. However, this approach is much limited in human studies due to safety concerns.

3.2.2. Genetic Variation Studies

The discovery in 1993 of the Cu,ZnSOD mutations in familial amyotrophic lateral sclerosis, a fatal, adult-onset neurodegenerative disease primarily affecting motor neurons, has attracted broad attention on the biological role of SODs in human disorders [44]. As illustrated in Figure 3-1, induction of this disease in humans is believed to be due, at least partially, to a toxic gain of function (e.g., an increased pro-oxidant activity) of the mutant Cu,ZnSOD protein, rather than a loss of SOD activity.

Many, though not all, human population studies have demonstrated that gene polymorphisms exist for all three SOD isozymes, and are associated with altered risk of developing various human disease conditions. These include cardiovascular diseases, diabetes, neurodegenerative disorders (e.g., Parkinson's disease, Alzheimer's disease), lung diseases, and certain types of cancer [45-50]. However, it should be borne in mind that the above observational studies do not prove a causal relationship between gene polymorphisms and disease development. In addition to the gene polymorphism studies, there are also reports in human subjects showing a correlation between changes in SOD levels and disease pathophysiology. Again, such studies do not establish a causal role for SODs in the genesis or progression of disease pathophysiology.



Figure 3-1. Schematic illustration of the potential involvement of Cu,ZnSOD-mediated oxidative stress in the pathogenesis of familial amyotrophic lateral sclerosis. It is believed that the mutations in Cu,ZnSOD gene result in production of mutant Cu,ZnSOD with increased pro-oxidant activities. This toxic gain of function causes oxidative stress, leading to motor neuron death. This notion is supported by studies in experimental models showing oxidative neuron degeneration induced by the mutant Cu,ZnSOD protein.

3.2.3. Interventional Clinical Trials

There have been studies in using SODs for the intervention of certain human disease conditions, but the results are inconsistent. An early clinical study reported that recombinant human SOD failed to improve recovery of ventricular function in patients undergoing coronary angioplasty for acute myocardial infarction [51]. A randomized double-blind, placebo-controlled trial showed a beneficial effect of recombinant human SOD on acute and chronic rejection in recipients of cadaveric renal transplants [52]. It has also been reported that in kidney transplantation use of a modified perfusate containing polyethylene glycol-SOD (PEG-SOD), prostaglandin E1, and nitroglycerin improves delayed graft function and survival [53]. In certain clinical studies, topical application of SODs was reported to improve skin or muco-sa lesions associated with various skin disorders, including progressive systemic sclerosis, burns, and pruritis. Cu,ZnSOD has also been used in certain creams for cosmetic purposes with a hope to retard skin aging. The protective effects of SODs when used as exogenously administered agents in human disorders remain to be established through more clinical studies. One major drawback associated with using native SODs for disease intervention is the protein nature of the SODs, which limits their membrane permeability, bioavailability, and

half-life. In this context, small molecular mass SOD mimetics have been developed. These SOD mimetics have been studied for the intervention of ROS-mediated disease processes (Chapter 24).

4. Conclusion and Future Directions

Extensive animal studies demonstrate an important role for all three forms of SOD in protecting against a wide variety of disease conditions. The protective effects apparently result from SOD's ability to scavenge superoxide and attenuate oxidative and inflammatory stress. Consistent with the observations made in animal models, human population studies demonstrate that SOD gene polymorphisms are associated with altered risk of developing various disease conditions. It is established that mutations in Cu,ZnSOD gene causally contribute to the pathogenesis of familial amyotrophic lateral sclerosis. However, randomized controlled trials on using SODs for the intervention of human diseases have been much limited with inconsistent results. Future studies should focus on developing more effective superoxide scavengers, such as small molecular mass SOD mimetics, and testing their effectiveness in disease intervention in well-designed clinical trials. Such studies will further enhance our understanding of the role of superoxide and SODs in human health and disease. They will also contribute to the development of SOD-based modalities to combat human diseases involving a superoxide-mediated oxidative stress mechanism.

5. References

- McCord JM, Fridovich I. Superoxide dismutase. An enzymic function for erythrocuprein (hemocuprein). J. Biol. Chem. 1969;244:6049-6055.
- [2] Marklund SL, Holme E, Hellner L. Superoxide dismutase in extracellular fluids. *Clin. Chim. Acta* 1982;126:41-51.
- [3] Fridovich I. Superoxide anion radical (O₂⁻), superoxide dismutases, and related matters. *J. Biol. Chem.* 1997;272:18515-18517.
- [4] Pacher P, Beckman JS, Liaudet L. Nitric oxide and peroxynitrite in health and disease. *Physiol. Rev.* 2007;87:315-424.
- [5] Miao L, St Clair DK. Regulation of superoxide dismutase genes: implications in disease. *Free Radic. Biol. Med.* 2009;47:344-356.
- [6] Dreger H, Westphal K, Weller A, Baumann G, Stangl V, Meiners S, *et al.* Nrf2dependent upregulation of antioxidative enzymes: a novel pathway for proteasome inhibitor-mediated cardioprotection. *Cardiovasc. Res.* 2009;83:354-361.
- [7] Yamakura F, Kawasaki H. Post-translational modifications of superoxide dismutase. *Biochim. Biophys. Acta* 2010;1804:318-325.
- [8] Levonen AL, Vahakangas E, Koponen JK, Yla-Herttuala S. Antioxidant gene therapy for cardiovascular disease: current status and future perspectives. *Circulation* 2008; 117:2142-2150.
- [9] Yen HC, Oberley TD, Vichitbandha S, Ho YS, St Clair DK. The protective role of manganese superoxide dismutase against adriamycin-induced acute cardiac toxicity in

transgenic mice. J. Clin. Invest 1996;98:1253-1260.

- [10] Li Y, Huang TT, Carlson EJ, Melov S, Ursell PC, Olson JL, et al. Dilated cardiomyopathy and neonatal lethality in mutant mice lacking manganese superoxide dismutase. *Nat. Genet*. 1995;11:376-381.
- [11] Vincent AM, Russell JW, Sullivan KA, Backus C, Hayes JM, McLean LL, et al. SOD2 protects neurons from injury in cell culture and animal models of diabetic neuropathy. *Exp. Neurol.* 2007;208:216-227.
- [12] Shen X, Zheng S, Metreveli NS, Epstein PN. Protection of cardiac mitochondria by overexpression of MnSOD reduces diabetic cardiomyopathy. *Diabetes* 2006;55:798-805.
- [13] Kowluru RA, Kowluru V, Xiong Y, Ho YS. Overexpression of mitochondrial superoxide dismutase in mice protects the retina from diabetes-induced oxidative stress. *Free Radic. Biol. Med.* 2006;41:1191-1196.
- [14] Craven PA, Melhem MF, Phillips SL, DeRubertis FR. Overexpression of Cu²⁺/Zn²⁺ superoxide dismutase protects against early diabetic glomerular injury in transgenic mice. *Diabetes* 2001;50:2114-2125.
- [15] Sandstrom J, Jonsson LM, Edlund H, Holmberg D, Marklund SL. Overexpression of extracellular-SOD in islets of nonobese diabetic mice and development of diabetes. *Free Radic. Biol. Med.* 2002;33:71-75.
- [16] Nakao N, Frodl EM, Widner H, Carlson E, Eggerding FA, Epstein CJ, et al. Overexpressing Cu/Zn superoxide dismutase enhances survival of transplanted neurons in a rat model of Parkinson's disease. *Nat. Med.* 1995;1:226-231.
- [17] Massaad CA, Washington TM, Pautler RG, Klann E. Overexpression of SOD-2 reduces hippocampal superoxide and prevents memory deficits in a mouse model of Alzheimer's disease. *Proc. Natl. Acad. Sci. USA* 2009;106:13576-13581.
- [18] Kim DW, Eum WS, Jang SH, Kim SY, Choi HS, Choi SH, et al. Transduced Tat-SOD fusion protein protects against ischemic brain injury. *Mol. Cells* 2005;19:88-96.
- [19] Melov S, Schneider JA, Day BJ, Hinerfeld D, Coskun P, Mirra SS, et al. A novel neurological phenotype in mice lacking mitochondrial manganese superoxide dismutase. *Nat. Genet.* 1998;18:159-163.
- [20] Reaume AG, Elliott JL, Hoffman EK, Kowall NW, Ferrante RJ, Siwek DF, et al. Motor neurons in Cu/Zn superoxide dismutase-deficient mice develop normally but exhibit enhanced cell death after axonal injury. Nat. Genet. 1996;13:43-47.
- [21] Kinnula VL, Crapo JD. Superoxide dismutases in the lung and human lung diseases. *Am. J. Respir. Crit. Care Med.* 2003;167:1600-1619.
- [22] Gao F, Kinnula VL, Myllarniemi M, Oury TD. Extracellular superoxide dismutase in pulmonary fibrosis. *Antioxid. Redox Signal* 2008;10:343-354.
- [23] Lehmann TG, Wheeler MD, Froh M, Schwabe RF, Bunzendahl H, Samulski RJ, et al. Effects of three superoxide dismutase genes delivered with an adenovirus on graft function after transplantation of fatty livers in the rat. *Transplantation* 2003;76:28-37.
- [24] Uchiyama S, Shimizu T, Shirasawa T. CuZn-SOD deficiency causes ApoB degradation and induces hepatic lipid accumulation by impaired lipoprotein secretion in mice. J. *Biol. Chem.* 2006;281:31713-31719.
- [25] Wheeler MD, Nakagami M, Bradford BU, Uesugi T, Mason RP, Connor HD, *et al.* Overexpression of manganese superoxide dismutase prevents alcohol-induced liver injury in the rat. J. Biol. Chem. 2001;276:36664-36672.

- [26] Kessova IG, Ho YS, Thung S, Cederbaum AI. Alcohol-induced liver injury in mice lacking Cu, Zn-superoxide dismutase. *Hepatology* 2003;38:1136-1145.
- [27] Lei XG, Zhu JH, McClung JP, Aregullin M, Roneker CA. Mice deficient in Cu,Znsuperoxide dismutase are resistant to acetaminophen toxicity. *Biochem. J.* 2006;399: 455-461.
- [28] Kruidenier L, van Meeteren ME, Kuiper I, Jaarsma D, Lamers CB, Zijlstra FJ, et al. Attenuated mild colonic inflammation and improved survival from severe DSS-colitis of transgenic Cu/Zn-SOD mice. Free Radic. Biol. Med. 2003;34:753-765.
- [29] Nishikawa M, Nagatomi H, Nishijima M, Ohira G, Chang BJ, Sato E, *et al.* Targeting superoxide dismutase to renal proximal tubule cells inhibits nephrotoxicity of cisplatin and increases the survival of cancer-bearing mice. *Cancer Lett.* 2001;171:133-138.
- [30] Carlstrom M, Brown RD, Sallstrom J, Larsson E, Zilmer M, Zabihi S, et al. SOD1 deficiency causes salt sensitivity and aggravates hypertension in hydronephrosis. Am. J. Physiol. Regul. Integr. Comp. Physiol. 2009;297:R82-92.
- [31] Rodriguez-Iturbe B, Sepassi L, Quiroz Y, Ni Z, Wallace DC, Vaziri ND. Association of mitochondrial SOD deficiency with salt-sensitive hypertension and accelerated renal senescence. J. Appl. Physiol. 2007;102:255-260.
- [32] Murakami K, Inagaki J, Saito M, Ikeda Y, Tsuda C, Noda Y, et al. Skin atrophy in cytoplasmic SOD-deficient mice and its complete recovery using a vitamin C derivative. Biochem. Biophys. Res. Commun. 2009;382:457-461.
- [33] Galenko-Yaroshevskii VP, Galka VV, Stoyalova ON, Agadzhanova AV, Lapina IV, Gaivoronskaya TV, et al. Antinecrotic and antioxidant effects of superoxide dismutase during skin ischemia. Bull. Exp. Biol. Med. 2006;142:447-449.
- [34] Elchuri S, Oberley TD, Qi W, Eisenstein RS, Jackson Roberts L, Van Remmen H, *et al.* CuZnSOD deficiency leads to persistent and widespread oxidative damage and hepatocarcinogenesis later in life. *Oncogene* 2005;24:367-380.
- [35] Van Remmen H, Ikeno Y, Hamilton M, Pahlavani M, Wolf N, Thorpe SR, *et al.* Lifelong reduction in MnSOD activity results in increased DNA damage and higher incidence of cancer but does not accelerate aging. *Physiol. Genomics* 2003;16:29-37.
- [36] Teoh ML, Fitzgerald MP, Oberley LW, Domann FE. Overexpression of extracellular superoxide dismutase attenuates heparanase expression and inhibits breast carcinoma cell growth and invasion. *Cancer Res.* 2009;69:6355-6363.
- [37] Huang P, Feng L, Oldham EA, Keating MJ, Plunkett W. Superoxide dismutase as a target for the selective killing of cancer cells. *Nature* 2000;407:390-395.
- [38] Friedman JS, Lopez MF, Fleming MD, Rivera A, Martin FM, Welsh ML, et al. SOD2deficiency anemia: protein oxidation and altered protein expression reveal targets of damage, stress response, and antioxidant responsiveness. Blood 2004;104:2565-2573.
- [39] Iuchi Y, Okada F, Takamiya R, Kibe N, Tsunoda S, Nakajima O, et al. Rescue of anaemia and autoimmune responses in SOD1-deficient mice by transgenic expression of human SOD1 in erythrocytes. *Biochem. J.* 2009;422:313-320.
- [40] Muller FL, Song W, Liu Y, Chaudhuri A, Pieke-Dahl S, Strong R, et al. Absence of CuZn superoxide dismutase leads to elevated oxidative stress and acceleration of agedependent skeletal muscle atrophy. *Free Radic. Biol. Med.* 2006;40:1993-2004.
- [41] Bivalacqua TJ, Armstrong JS, Biggerstaff J, Abdel-Mageed AB, Kadowitz PJ, Hellstrom WJ, *et al.* Gene transfer of extracellular SOD to the penis reduces O_2^- and

improves erectile function in aged rats. Am. J. Physiol. Heart Circ. Physiol. 2003;284: H1408-1421.

- [42] Deng W, Bivalacqua TJ, Champion HC, Hellstrom WJ, Murthy SN, Kadowitz PJ. Superoxide dismutase: a target for gene therapeutic approach to reduce oxidative stress in erectile dysfunction. *Methods Mol. Biol.* 2010;610:213-227.
- [43] Meissner F, Molawi K, Zychlinsky A. Superoxide dismutase 1 regulates caspase-1 and endotoxic shock. *Nat. Immunol.* 2008;9:866-872.
- [44] Rosen DR, Siddique T, Patterson D, Figlewicz DA, Sapp P, Hentati A, *et al.* Mutations in Cu/Zn superoxide dismutase gene are associated with familial amyotrophic lateral sclerosis. *Nature* 1993;362:59-62.
- [45] Naganuma T, Nakayama T, Sato N, Fu Z, Soma M, Aoi N, et al. A haplotype-based case-control study examining human extracellular superoxide dismutase gene and essential hypertension. *Hypertens. Res.* 2008;31:1533-1540.
- [46] Alameddine FM, Zafari AM. Genetic polymorphisms and oxidative stress in heart failure. *Congest. Heart Fail.* 2002;8:157-164, 172.
- [47] Flekac M, Skrha J, Hilgertova J, Lacinova Z, Jarolimkova M. Gene polymorphisms of superoxide dismutases and catalase in diabetes mellitus. *BMC Med. Genet.* 2008;9:30.
- [48] Wiener HW, Perry RT, Chen Z, Harrell LE, Go RC. A polymorphism in SOD2 is associated with development of Alzheimer's disease. *Genes Brain Behav.* 2007;6:770-775.
- [49] Arcaroli JJ, Hokanson JE, Abraham E, Geraci M, Murphy JR, Bowler RP, *et al.* Extracellular superoxide dismutase haplotypes are associated with acute lung injury and mortality. *Am. J. Respir. Crit. Care Med.* 2009;179:105-112.
- [50] Nahon P, Sutton A, Rufat P, Ziol M, Akouche H, Laguillier C, *et al.* Myeloperoxidase and superoxide dismutase 2 polymorphisms comodulate the risk of hepatocellular carcinoma and death in alcoholic cirrhosis. *Hepatology* 2009;50:1484-1493.
- [51] Flaherty JT, Pitt B, Gruber JW, Heuser RR, Rothbaum DA, Burwell LR, et al. Recombinant human superoxide dismutase (h-SOD) fails to improve recovery of ventricular function in patients undergoing coronary angioplasty for acute myocardial infarction. *Circulation* 1994;89:1982-1991.
- [52] Land W, Schneeberger H, Schleibner S, Illner WD, Abendroth D, Rutili G, *et al.* The beneficial effect of human recombinant superoxide dismutase on acute and chronic rejection events in recipients of cadaveric renal transplants. *Transplantation* 1994; 57:211-217.
- [53] Guarrera JV, Polyak MM, Arrington B, Boykin J, Brown T, Jean-Jacques MA, *et al.* Pushing the envelope in renal preservation; improved results with novel perfusate modifications for pulsatile machine perfusion of cadaver kidneys. *Transplant. Proc.* 2004;36:1257-1260.

Chapter 4

Catalase

Abstract

Mammalian catalase is a heme-containing enzyme that catalyzes the decomposition of hydrogen peroxide to water and molecular oxygen. It is widely distributed in various tissues and primarily localized in peroxisomes. A number of transcription factors, coactivators, and other signaling molecules are implicated in the regulation of mammalian catalase expression. Studies in animal models of catalase deficiency or overexpression demonstrate a protective role for catalase in disease conditions ranging from cardiovascular injury and diabetes to cancer and aging. Inheritable catalase deficiency and catalase gene polymorphisms have also been identified in humans. Studies on both hereditary catalase deficiency and gene polymorphisms suggest a potential beneficial role for catalase in human health and disease.

1. Overview

1.1. Definition and History

Mammalian catalase belongs to a family of iron-protoporphyrin IX-containing proteins that include a variety of cytochromes, globins and peroxidases. Catalase is one of the best characterized antioxidant enzymes in mammalian tissues. It plays an important role in regulating intracellular hydrogen peroxide levels in mammalian cells. Another major enzyme for detoxifying hydrogen peroxide is glutathione peroxidase (Chapter 6).

Catalase was identified and named by O. Loew in 1900. O. Loew named it catalase due to its catalytic action on hydrogen peroxide [1]. A congenital absence of erythrocyte catalase, named acatalasemia was first identified by S. Takahara about six decade ago among some Japanese patients who had oral gangrene [2]. The disease is also called Takahara's disease. In 1964, R.N. Feinstein and coworkers established catalase-deficient mouse strains, including acatalasemic and hypocatalasemic mice through a large-scale screening of the progeny of irradiated C3H mice [3]. The development of these catalase-deficient mice as well as gene knockout and transgenic overexpression animal models has greatly helped understand the biological activities of mammalian catalase.

1.2. Basic Characteristics

Mammalian catalase is a homotetramer with a molecular mass of ~240 kDa. It is widely distributed throughout the body with high levels in liver, kidney, and red blood cells. In cells, catalase is primarily localized in peroxisomes. It is also found in other intracellular compartments, including mitochondria, nuclei, and endoplasmic reticulum. Human catalase gene is localized on chromosome 11p13.

2. Biochemistry and Molecular Regulation

2.1. Biochemistry

As stated above, mammalian catalase is a homotetramer, and each subunit contains a heme (ferriprotoporphyrin IX) at the active site. The active site is internally located and approachable by a passageway that becomes narrow, thereby explaining the ability of the enzyme to use only substrates of small size [4].

Catalase is best known for its ability to catalyze the decomposition of hydrogen peroxide (H_2O_2) to form water and molecular oxygen (Reaction 1). This is a dismutation reaction because one molecule of hydrogen peroxide is reduced to water and another molecule is oxidized to molecular oxygen.

$$2H_2O_2 \xrightarrow{\text{Catalase}} 2H_2O + O_2$$
 (1)

Catalase catalyzes decomposition of hydrogen peroxide via a two-stage process. In the first step, one hydrogen peroxide molecule oxidizes the heme iron of the resting enzyme to form an oxyferryl species with a π -cationic porphyrin radical, termed compound I (Reaction 2). The second hydrogen peroxide molecule is utilized as a reductant of compound I to regenerate the resting-stage enzyme along with the formation of water and molecular oxygen (Reaction 3, where Por denotes porphyrin).

$$Catalase(Por-Fe^{III}) + H_2O_2 \longrightarrow Compound I(Por^+ - Fe^{IV} = O) + H_2O$$
(2)

Compound I(Por⁺-Fe^{IV}=O) + H₂O₂
$$\longrightarrow$$
 Catalase(Por-Fe^{III}) + H₂O + O₂ (3)

The sum of Reactions 2 and 3 is: $2H_2O_2 \longrightarrow 2H_2O + O_2$

Catalase also possesses peroxidase or oxidase activities toward a number of substrates, including low molecular mass alcohols, the tryptophan precursor indole, and the neurotransmitter precursor β -phenylethylamine [5]. In general, catalase exhibits the peroxidatic activity in the presence of low concentrations of hydrogen peroxide. Reaction 4 illustrates the peroxidatic activity of catalase in converting alcohol (AH₂) to aldehyde (A) in the presence of low levels of hydrogen peroxide.

$$H_2O_2 + AH_2 \xrightarrow{\text{Catalase}} A + 2H_2O$$
 (4)

2.2. Molecular Regulation

Catalase expression is regulated at multiple levels, including transcription, posttranscription, and post-translation. Regulation of mammalian catalase expression has only been recently studied more carefully, and much still remains unknown. It has been reported that the forkhead transcription factor Foxo3a plays an important role in the upregulation of catalase gene expression in mammalian cells [6]. Other regulatory factors involved in the regulation of catalase expression include the transcriptional co-activator PGC-1 α (peroxisome proliferator-activated receptor γ co-activator 1 α), the non-receptor tyrosine kinases c-Abl and Arg, Sirt1, Sirt3, telomerase, and AMP-activated protein kinase α 1 [7-10]. Recently, Nrf2 is also shown to control catalase gene expression though it remains unclear if the promoter of catalase is shown to undergo ubiquitination and proteasomal degradation by a mechanism dependent on c-Abl- and Arg-mediated phosphorylation [12].

3. Role in Health and Disease

3.1. Animal Studies

3.1.1. Experimental Approaches

The experimental approaches described in Chapter 3 are also frequently used to study the biological activities of catalase in animal models. Like superoxide dismutase, much of our current knowledge on the biological activities of mammalian catalase comes from studies in knockout, acatalasemic, and transgenic overexpression animal models, as well as the direct use of catalase or catalase derivatives. A number of catalase derivatives have been developed to selectively deliver the chemically-modified catalase to different tissues or various cell populations within the same tissue [13]. Viral vector-mediated gene delivery is also frequently employed to overexpress catalase in target tissues of experimental animals to study the effects on disease pathophysiology. The chemical inhibitor aminotriazole is a commonly used tool to study the biological activities of catalase in cellular systems.

3.1.2. Role in Cardiovascular Diseases

Studies using both gene knockout and overexpression animal models demonstrate a critical role for catalase in protecting against various cardiovascular disorders with oxidative stress as an important mechanism. These include atherosclerosis, hypertension, myocardial ischemia-reperfusion injury, cardiomyopathy, cardiac hypertrophy, and chronic heart failure [14-16]. Notably, overexpression of catalase targeted to mitochondria attenuates murine cardiac aging, suggesting a crucial role for mitochondria-derived hydrogen peroxide in agingassociated myocardial dysfunction and degeneration [17]. In addition, catalase also protects against cardiotoxicity induced by drugs and xenobiotics, including doxorubicin and ethanol in experimental animals [18].

3.1.3. Role in Diabetes

Overexpression of catalase attenuates oxidative stress, inflammation, and development of

both type 1 and type 2 diabetes in experimental animals. Catalase overexpression also ameliorates diabetic complications, including diabetic cardiomyopathy [19]. In vitro, mitochondrial overexpression of catalase protects the pancreatic β -cells from oxidative stress- and inflammatory cytokine-induced injury [20]. On the other hand, acatalasemic mice have been shown to be more susceptible to chemically-induced pancreatic β -cell injury and diabetes [21]. In contrast to the large body of literature showing a protective role of catalase in diabetes, one study reported that cytoplasmic catalase overexpression in nonobese mice accelerated diabetes development [22].

3.1.4. Role in Neurological Diseases

Homozygous catalase knockout (catalase^{-/-}) mice develop normally and show no gross abnormalities. As compared with wild-type littermates, catalase^{-/-} mice are not more susceptible to hyperoxia-induced lung injury. However, brain mitochondria of the catalase^{-/-} mice are more vulnerable to trauma-induced impairment in oxidative phosphorylation than those of wild-type mice [23]. This suggests a potential role for catalase in neuroprotection. Indeed, transgenic overexpression or viral vector-mediated gene delivery of catalase attenuates the disease severity in various animal models of neurodegenerative disorders, including multiple sclerosis, optic neuropathy, and Friedreich's ataxia [24]. Similarly, administration of the synthetic superoxide dismutase/catalase mimetics (Chapter 24) has also been reported to protect against neurodegeneration in experimental animals [25, 26].

3.1.5. Role in Pulmonary Diseases

As stated above in Section 3.1.4, catalase-deficient mice are not more susceptible to hyperoxic pulmonary injury than wild-type mice [23]. But, viral vector-mediated delivery of catalase gene to lung attenuates hyperoxia-induced lung injury and mortality in mice [27]. Immunotargeting of catalase to the pulmonary endothelium also alleviates oxidative stress and reduces acute lung transplantation injury in rats [28]. However, a study shows that over-expression of catalase in lung tissue of mice fails to attenuate allergic airway disease, in which oxidative stress and inflammation play an important role [29].

3.1.6. Role in Hepatic and Gastrointestinal Diseases

Several studies demonstrate that delivery of catalase gene to liver results in increased hepatic catalase activity and attenuation of liver necrosis and inflammation following ischemiareperfusion in experimental animals. Administration of a catalase derivative, succinylated catalase to mice also increases catalase activity in liver tissue and prevents neutrophilmediated hepatic ischemia-reperfusion injury [30]. Catalase and glutathione peroxidase-1 double-knockout mice are found to be more susceptible to ethanol-induced liver injury than wild-type mice [31], indicating a potential protective role for catalase in hepatotoxicity. Indeed, catalase derivatives are shown to protect against liver failure induced by a number of hepatotoxicants, including carbon tetrachloride and thioacetamide [13]. Delivery of catalase derivatives to liver nonparenchymal cells (e.g., Kupffer cells) is effective in inhibiting liver injury during septic shock in mice. This finding implicates that targeting catalase to Kupffer cells may suppress liver injury due to Kupffer cell activation (i.e., release of reactive oxygen species) during septic shock. Administration of catalase derivatives also attenuates experimental colitis in animal models, suggesting a causal involvement of hydrogen peroxide in the development of colitis. It is also reported that acatalasemic mice exhibit increased susceptibility to oxidant-mediated peritoneal injury and fibrosis. Peritoneal fibrosis is a critical complication of peritoneal dialysis [32].

3.1.7. Role in Renal Diseases

Catalase deficiency in mice sensitizes these animals to renal oxidative stress and fibrosis due to nephronectomy or ureteral obstruction [33, 34]. On the other hand, transgenic overexpression of catalase attenuates interstitial fibrosis and tubular apoptosis in db/db diabetic mice, suggesting a critical involvement of hydrogen peroxide and oxidative stress in diabetic nephropathy [35]. Administration of a catalase derivative (cationized catalase) to mice leads to accumulation of the catalase derivative in kidney, and attenuation of cisplatin-induced nephrotoxicity without compromising its antitumor activity. The survival of the cisplatin-treated tumor-bearing animals is also improved by administration of the catalase derivative [36]. These observations support the oxidative stress mechanism of cisplatin-induced nephrotoxicity and point to the feasibility for using catalase-based modalities to enhance the therapeutic index of cisplatin. Cisplatin is an important anticancer drug whose clinical use is limited by its nephrotoxicity.

3.1.8. Role in Skin Diseases

Multiple studies suggest a protective effect of catalase in ultraviolet (UV) light-induced skin injury. It is known that UV light causes the formation of reactive oxygen species (ROS), including hydrogen peroxide in skin tissue. Viral vector-mediated overexpression of catalase protects normal human reconstructed epidermis from UV light-induced sunburn cell formation, hypertrophy, and oxidative DNA damage. Catalase overexpression also reduces UV light-induced apoptosis in a human xeroderma pigmentosum reconstituted epidermis [37]. It is further reported that catalase overexpression in cultured human keratinocytes inhibits UV light-induced apoptosis via preventing DNA damage caused by ROS [38].

Animal studies demonstrate that skin wound healing is subject to redox regulation. Viral vector-mediated overexpression of catalase in mouse skin is found to retard skin wound healing and tissue remodeling, suggesting a critical role for hydrogen peroxide in promoting skin wound healing in experimental animals [39]. In this regard, hydrogen peroxide at moderate concentrations has been suggested as a potential modality to facilitate the process of skin wound healing. This represents another example that ROS may play a beneficial role in health and disease.

3.1.9. Role in Cancer

Unlike superoxide dismutase (Section 3.1.9 of Chapter 3), the involvement of catalase in experimental carcinogenesis has not been investigated in gene knockout or transgenic overexpression models. Several studies using catalase derivatives have found a protective effect of catalase in tumor metastasis in animal models, including hepatic and pulmonary metastasis as well as peritoneal dissemination of tumor cells [13]. These findings provide evidence for a causal involvement of hydrogen peroxide in tumor metastasis. In vitro, catalase overexpression suppresses tumor cell proliferation and invasiveness. Together, these studies suggest that targeted delivery of catalase to sites of tumor cell metastasis may be a promising approach to inhibiting metastatic tumor growth.

3.1.10. Role in Other Diseases and Conditions

Catalase may also play a beneficial role in other diseases and conditions, including arthritis, eye disorders, and aging. Overexpression of catalase targeted to mitochondria results in extension of murine lifespan and delayed development of cardiac pathology and cataract. Tissue oxidative damage, mitochondrial enzyme inactivation, and mitochondrial deletion are also attenuated in the catalase-overexpressing animals [40]. A recent study also demonstrates that ectopic catalase expression in mitochondria by viral vector-mediated delivery leads to enhanced exercise performance in mice, possibly by increased removing of excessive, pathogenic hydrogen peroxide in muscle [41].

3.2. Human Studies and Clinical Perspectives

The significance of catalase in human health and disease has been investigated largely via studying the phenotypic changes in patients with hereditary catalase deficiency and catalase gene polymorphisms in general populations.

3.2.1. Hereditary Catalase Deficiency

As stated above in Section 1.1, acatalasemia in humans was identified about six decades ago in some Japanese patients. This rare genetic disorder was originally thought to predispose people to oral and facial infections. However, no such predisposition was found in other Japanese people who were later shown to have the same genetic disorder. Hereditary catalase deficiency has been subsequently also observed in people from Switzerland and other countries. Although catalase makes much of the proteins in peroxisomes, patients with hereditary catalase deficiency have no obvious features of peroxisomal diseases. However, these patients may be more susceptible to oxidative stress injury and have an increased risk of developing diabetes [42, 43].

3.2.2. Gene Polymorphisms

There exist various polymorphisms in the catalase gene in human populations. Extensive studies demonstrate an association of gene polymorphisms with a number of disease conditions, including hypertension, diabetes, neurodegeneration, asthma, inflammatory disorders, vitiligo, osteonecrosis, and certain types of cancer [44-52]. Although population gene polymorphism studies do not establish a causal relationship between genetic variations of catalase gene and development of disease conditions, they do provide important clues to the biological significance of this antioxidant enzyme in human health and disease. Such knowledge may serve as a basis for developing strategies to minimize the potential risk associated with genetic variations of catalase gene in human subjects.

4. Conclusion and Future Directions

Studies in various animal models provide a large body of data supporting a critical role for catalase in protecting against a wide range of disease conditions. These studies also provide evidence for a causal involvement of hydrogen peroxide in disease pathophysiology. Although studies in human subjects have not been as convincing as those in experimental animals, there is evidence that catalase may exert beneficial effects in certain human disease conditions, including hypertension, diabetes, and cancer. Future studies should emphasize translational research and focus on developing novel catalase derivatives or mimetics with improved bioavailability and safety profile and testing their efficacy in disease intervention in clinical trials. As superoxide dismutase and catalase act coordinately in the sequential metabolism of superoxide to hydrogen peroxide, and hydrogen peroxide to water (Figure 4-1), more attention needs to be paid to the combined effects of these two antioxidant enzymes in health and disease. Future efforts should also be directed to the continued characterization of novel gene polymorphisms and their impacts on tissue catalase expression as well as individual's susceptibility to disease development.



Figure 4-1. SOD- and catalase-catalyzed sequential metabolism of superoxide and hydrogen peroxide in disease intervention. As illustrated, SOD catalyzes the dismutation of superoxide to hydrogen peroxide. The toxic hydrogen peroxide is further decomposed to water by catalase. Combined effects of both SOD and catalase ensure the detoxification of both superoxide and hydrogen peroxide in biological systems. SOD, superoxide dismutase; O_2 , superoxide; H_2O_2 , hydrogen peroxide.

5. References

- Loew O. A New Enzyme of General Occurrence in Organismis. Science 1900;11:701-702.
- [2] Takahara S. Progressive oral gangrene probably due to lack of catalase in the blood (acatalasaemia); report of nine cases. *Lancet* 1952;2:1101-1104.
- [3] Feinstein RN, Seaholm JE, Howard JB, Russell WL. Acatalasemic Mice. *Proc. Natl. Acad. Sci. USA* 1964;52:661-662.
- [4] Kirkman HN, Gaetani GF. Mammalian catalase: a venerable enzyme with new mysteries. *Trends Biochem. Sci.* 2007;32:44-50.
- [5] Vetrano AM, Heck DE, Mariano TM, Mishin V, Laskin DL, Laskin JD. Characterization of the oxidase activity in mammalian catalase. *J. Biol. Chem.* 2005;

280:35372-35381.

- [6] Olmos Y, Valle I, Borniquel S, Tierrez A, Soria E, Lamas S, *et al.* Mutual dependence of Foxo3a and PGC-1alpha in the induction of oxidative stress genes. *J. Biol. Chem.* 2009;284:14476-14484.
- [7] Cao C, Leng Y, Kufe D. Catalase activity is regulated by c-Abl and Arg in the oxidative stress response. *J. Biol. Chem.* 2003;278:29667-29675.
- [8] Sundaresan NR, Gupta M, Kim G, Rajamohan SB, Isbatan A, Gupta MP. Sirt3 blocks the cardiac hypertrophic response by augmenting Foxo3a-dependent antioxidant defense mechanisms in mice. *J. Clin. Invest.* 2009;119:2758-2771.
- [9] Perez-Rivero G, Ruiz-Torres MP, Diez-Marques ML, Canela A, Lopez-Novoa JM, Rodriguez-Puyol M, *et al.* Telomerase deficiency promotes oxidative stress by reducing catalase activity. *Free Radic. Biol. Med.* 2008;45:1243-1251.
- [10] Colombo SL, Moncada S. AMPKalpha1 regulates the antioxidant status of vascular endothelial cells. *Biochem. J.* 2009;421:163-169.
- [11] Dreger H, Westphal K, Weller A, Baumann G, Stangl V, Meiners S, *et al.* Nrf2dependent upregulation of antioxidative enzymes: a novel pathway for proteasome inhibitor-mediated cardioprotection. *Cardiovasc. Res.* 2009;83:354-361.
- [12] Cao C, Leng Y, Liu X, Yi Y, Li P, Kufe D. Catalase is regulated by ubiquitination and proteosomal degradation: role of the c-Abl and Arg tyrosine kinases. *Biochemistry* 2003;42:10348-10353.
- [13] Nishikawa M, Hashida M, Takakura Y. Catalase delivery for inhibiting ROS-mediated tissue injury and tumor metastasis. *Adv. Drug Deliv. Rev.* 2009;61:319-326.
- [14] Yang H, Shi M, VanRemmen H, Chen X, Vijg J, Richardson A, et al. Reduction of pressor response to vasoconstrictor agents by overexpression of catalase in mice. Am. J. Hypertens. 2003;16:1-5.
- [15] Yang H, Roberts LJ, Shi MJ, Zhou LC, Ballard BR, Richardson A, et al. Retardation of atherosclerosis by overexpression of catalase or both Cu/Zn-superoxide dismutase and catalase in mice lacking apolipoprotein E. Circ. Res. 2004;95:1075-1081.
- [16] Qin F, Lennon-Edwards S, Lancel S, Biolo A, Siwik DA, Pimentel DR, et al. Cardiac-specific overexpression of catalase identifies hydrogen peroxide-dependent and independent phases of myocardial remodeling and prevents the progression to overt heart failure in G(alpha)q-overexpressing transgenic mice. *Circ. Heart Fail.* 2010; 3:306-313.
- [17] Dai DF, Santana LF, Vermulst M, Tomazela DM, Emond MJ, MacCoss MJ, et al. Overexpression of catalase targeted to mitochondria attenuates murine cardiac aging. *Circulation* 2009;119:2789-2797.
- [18] Kang YJ, Chen Y, Epstein PN. Suppression of doxorubicin cardiotoxicity by overexpression of catalase in the heart of transgenic mice. J. Biol. Chem. 1996;271: 12610-12616.
- [19] Ye G, Metreveli NS, Donthi RV, Xia S, Xu M, Carlson EC, et al. Catalase protects cardiomyocyte function in models of type 1 and type 2 diabetes. *Diabetes* 2004;53: 1336-1343.
- [20] Gurgul E, Lortz S, Tiedge M, Jorns A, Lenzen S. Mitochondrial catalase overexpression protects insulin-producing cells against toxicity of reactive oxygen species and proinflammatory cytokines. *Diabetes* 2004;53:2271-2280.

- [21] Kikumoto Y, Sugiyama H, Inoue T, Morinaga H, Takiue K, Kitagawa M, et al. Sensitization to alloxan-induced diabetes and pancreatic cell apoptosis in acatalasemic mice. *Biochim. Biophys. Acta* 2010;1802:240-246.
- [22] Li X, Chen H, Epstein PN. Metallothionein and catalase sensitize to diabetes in nonobese diabetic mice: reactive oxygen species may have a protective role in pancreatic beta-cells. *Diabetes* 2006;55:1592-1604.
- [23] Ho YS, Xiong Y, Ma W, Spector A, Ho DS. Mice lacking catalase develop normally but show differential sensitivity to oxidant tissue injury. J. Biol. Chem. 2004;279: 32804-32812.
- [24] Anderson PR, Kirby K, Orr WC, Hilliker AJ, Phillips JP. Hydrogen peroxide scavenging rescues frataxin deficiency in a Drosophila model of Friedreich's ataxia. *Proc. Natl. Acad. Sci. USA* 2008;105:611-616.
- [25] Liu R, Liu IY, Bi X, Thompson RF, Doctrow SR, Malfroy B, et al. Reversal of agerelated learning deficits and brain oxidative stress in mice with superoxide dismutase/catalase mimetics. Proc. Natl. Acad. Sci. USA 2003;100:8526-8531.
- [26] Peng J, Stevenson FF, Doctrow SR, Andersen JK. Superoxide dismutase/catalase mimetics are neuroprotective against selective paraquat-mediated dopaminergic neuron death in the substantial nigra: implications for Parkinson disease. J. Biol. Chem. 2005;280:29194-29198.
- [27] Danel C, Erzurum SC, Prayssac P, Eissa NT, Crystal RG, Herve P, et al. Gene therapy for oxidant injury-related diseases: adenovirus-mediated transfer of superoxide dismutase and catalase cDNAs protects against hyperoxia but not against ischemiareperfusion lung injury. *Hum. Gene Ther.* 1998;9:1487-1496.
- [28] Kozower BD, Christofidou-Solomidou M, Sweitzer TD, Muro S, Buerk DG, Solomides CC, *et al.* Immunotargeting of catalase to the pulmonary endothelium alleviates oxidative stress and reduces acute lung transplantation injury. *Nat. Biotechnol.* 2003;21:392-398.
- [29] Reynaert NL, Aesif SW, McGovern T, Brown A, Wouters EF, Irvin CG, et al. Catalase overexpression fails to attenuate allergic airways disease in the mouse. J. Immunol. 2007;178:3814-3821.
- [30] Yabe Y, Kobayashi N, Nishihashi T, Takahashi R, Nishikawa M, Takakura Y, *et al.* Prevention of neutrophil-mediated hepatic ischemia/reperfusion injury by superoxide dismutase and catalase derivatives. *J. Pharmacol. Exp. Ther.* 2001;298:894-899.
- [31] Kim SJ, Lee JW, Jung YS, Kwon do Y, Park HK, Ryu CS, et al. Ethanol-induced liver injury and changes in sulfur amino acid metabolomics in glutathione peroxidase and catalase double knockout mice. J. Hepatol. 2009;50:1184-1191.
- [32] Fukuoka N, Sugiyama H, Inoue T, Kikumoto Y, Takiue K, Morinaga H, *et al.* Increased susceptibility to oxidant-mediated tissue injury and peritoneal fibrosis in acatalasemic mice. *Am. J. Nephrol.* 2008;28:661-668.
- [33] Kobayashi M, Sugiyama H, Wang DH, Toda N, Maeshima Y, Yamasaki Y, *et al.* Catalase deficiency renders remnant kidneys more susceptible to oxidant tissue injury and renal fibrosis in mice. *Kidney Int.* 2005;68:1018-1031.
- [34] Sunami R, Sugiyama H, Wang DH, Kobayashi M, Maeshima Y, Yamasaki Y, et al. Acatalasemia sensitizes renal tubular epithelial cells to apoptosis and exacerbates renal fibrosis after unilateral ureteral obstruction. Am. J. Physiol. Renal. Physiol. 2004;286: F1030-1038.

- [35] Brezniceanu ML, Liu F, Wei CC, Chenier I, Godin N, Zhang SL, et al. Attenuation of interstitial fibrosis and tubular apoptosis in db/db transgenic mice overexpressing catalase in renal proximal tubular cells. *Diabetes* 2008;57:451-459.
- [36] Ma SF, Nishikawa M, Hyoudou K, Takahashi R, Ikemura M, Kobayashi Y, et al. Combining cisplatin with cationized catalase decreases nephrotoxicity while improving antitumor activity. *Kidney Int.* 2007;72:1474-1482.
- [37] Rezvani HR, Ged C, Bouadjar B, de Verneuil H, Taieb A. Catalase overexpression reduces UVB-induced apoptosis in a human xeroderma pigmentosum reconstructed epidermis. *Cancer Gene Ther.* 2008;15:241-251.
- [38] Rezvani HR, Mazurier F, Cario-Andre M, Pain C, Ged C, Taieb A, *et al.* Protective effects of catalase overexpression on UVB-induced apoptosis in normal human keratinocytes. *J. Biol. Chem.* 2006;281:17999-18007.
- [39] Roy S, Khanna S, Nallu K, Hunt TK, Sen CK. Dermal wound healing is subject to redox control. *Mol. Ther.* 2006;13:211-220.
- [40] Schriner SE, Linford NJ, Martin GM, Treuting P, Ogburn CE, Emond M, et al. Extension of murine life span by overexpression of catalase targeted to mitochondria. *Science* 2005;308:1909-1911.
- [41] Li D, Lai Y, Yue Y, Rabinovitch PS, Hakim C, Duan D. Ectopic catalase expression in mitochondria by adeno-associated virus enhances exercise performance in mice. *PLoS One* 2009;4:e6673.
- [42] Goth L, Eaton JW. Hereditary catalase deficiencies and increased risk of diabetes. *Lancet* 2000;356:1820-1821.
- [43] Masuoka N, Sugiyama H, Ishibashi N, Wang DH, Masuoka T, Kodama H, et al. Characterization of acatalasemic erythrocytes treated with low and high dose hydrogen peroxide: hemolysis and aggregation. J. Biol. Chem. 2006;281:21728-21734.
- [44] Goth L, Rass P, Pay A. Catalase enzyme mutations and their association with diseases. *Mol. Diagn.* 2004;8:141-149.
- [45] Zhou XF, Cui J, DeStefano AL, Chazaro I, Farrer LA, Manolis AJ, et al. Polymorphisms in the promoter region of catalase gene and essential hypertension. Dis. Markers 2005;21:3-7.
- [46] Chistiakov DA, Zotova EV, Savost'anov KV, Bursa TR, Galeev IV, Strokov IA, et al. The 262T>C promoter polymorphism of the catalase gene is associated with diabetic neuropathy in type 1 diabetic Russian patients. *Diabetes Metab.* 2006;32:63-68.
- [47] Christiansen L, Petersen HC, Bathum L, Frederiksen H, McGue M, Christensen K. The catalase -262C/T promoter polymorphism and aging phenotypes. J. Gerontol. A Biol. Sci. Med. Sci. 2004;59:B886-889.
- [48] Polonikov AV, Ivanov VP, Solodilova MA, Kozhuhov MA, Panfilov VI. Tobacco smoking, fruit and vegetable intake modify association between -21A>T polymorphism of catalase gene and risk of bronchial asthma. J. Asthma 2009;46:217-224.
- [49] Wenten M, Gauderman WJ, Berhane K, Lin PC, Peters J, Gilliland FD. Functional variants in the catalase and myeloperoxidase genes, ambient air pollution, and respiratory-related school absences: an example of epistasis in gene-environment interactions. Am. J. Epidemiol. 2009;170:1494-1501.
- [50] Kim TH, Hong JM, Oh B, Cho YS, Lee JY, Kim HL, *et al.* Genetic association study of polymorphisms in the catalase gene with the risk of osteonecrosis of the femoral head in the Korean population. *Osteoarthritis Cartilage* 2008;16:1060-1066.

- [51] Warchol T, Lianeri M, Wudarski M, Lacki JK, Jagodzinski PP. Catalase -262C>T polymorphism in systemic lupus erythematosus in Poland. *Rheumatol. Int.* 2008;28: 1035-1039.
- [52] Rajaraman P, Hutchinson A, Rothman N, Black PM, Fine HA, Loeffler JS, *et al.* Oxidative response gene polymorphisms and risk of adult brain tumors. *Neuro. Oncol.* 2008;10:709-715.

Chapter 5

Glutathione and its Synthesizing Enzymes

Abstract

The reduced form of glutathione (GSH) is the most abundant intracellular antioxidant thiol in mammalian systems. It is synthesized from three amino acids (i.e., glutamate, cysteine, and glycine) via two successive reactions catalyzed by γ -glutamylcysteine ligase and glutathione synthetase, respectively. In mammalian cells and tissues, GSH participates in diverse reactions, including reacting with reactive oxygen and nitrogen species, electrophiles, and other non-enzymatic antioxidant compounds, as well as protein deglutathionylation. Via participating in these various redox reactions, GSH plays an important part in mammalian physiology. Extensive studies in animal models demonstrate crucial protective effects of GSH in a wide variety of disease processes involving oxidative stress. Consistently, studies in human subjects also suggest potential benefits of GSH in human diseases, including hemolytic anemia, hypertension, diabetes, and chronic obstructive pulmonary disease.

1. Overview

1.1. Definition and History

The term glutathione, if not specified, may refer to both the reduced form (GSH) and the oxidized form (GSSG) of glutathione, as well as the protein-bound glutathione moieties. The oxidized form is also known as glutathione disulfide. This chapter focuses on the reduced form of glutathione, i.e., GSH. In this context, the term glutathione is often used loosely to refer to GSH in the literature.

In 1888, M.J. de Rey Paihade discovered a substance, which was proved to be GSH by F.G. Hopkins in 1929 [1]. In 1935, C.R. Harington and T.H. Mead reported the synthesis of GSH [2]. It is now known that GSH is the most abundant non-protein thiol-containing compound in mammalian cells. Although it was initially believed to mainly act as a reductant, over the past two decades, GSH has been demonstrated to play important roles in various biological processes and disease conditions.

1.2. Basic Characteristics

GSH is a tripeptide (γ -glutamylcysteinylglycine) with a molecular mass of 307 (structure shown in Figure 5-1). In mammalian cells, the cytosolic concentrations of GSH are in the range of 1-10 mM. GSH is also present in high concentrations in mitochondria and nuclei. In contrast, the extracellular levels of GSH are much lower, and GSH levels in plasma are usually in the range of low micromolar concentrations. In mammalian cells, GSH synthesis involves two cytosolic enzymes, namely, γ -glutamylcysteine ligase (GCL) and glutathione synthetase (GSS) (Section 2.1). GCL consists of two subunits: the heavy catalytic subunit designated as GCLC with a molecular mass of ~73 kDa and the light modifier subunit designated as GCLM with a molecular mass of ~31 kDa. GCLC and GCLM are encoded by separate genes that are localized on chromosomes 6p12 and 1p22.1, respectively, in humans. Human GSS gene is localized on chromosome 20q11.2.



Figure 5-1. Structure of γ -glutamylcysteinylglycine (GSH). As indicated, GSH is a tripeptide, and the sulfhydryl group (-SH) is from the cysteine residue.

2. Biochemistry and Molecular Regulation

2.1. Biochemistry

GSH is synthesized from three amino acids via two successive enzymatic reactions in cytoplasm (Figure 5-2). The first step involves combination of cysteine and glutamate to produce the dipeptide γ -glutamylcysteine. The reaction is catalyzed by GCL, also known as γ glutamylcysteine synthetase (GCS). This reaction requires coupled ATP hydrolysis. The next step involves the enzyme GSS, which catalyzes the addition of glycine to the dipeptide to form the tripeptide γ -glutamylcysteinylglycine (GSH). Similar to GCL, GSS also requires coupled ATP hydrolysis.

2.2. Molecular Regulation

 γ -Glutamylcysteine ligase is the key enzyme of GSH biosynthesis. As stated above, this enzyme consists of two subunits, namely, GCLC and GCLM. As the name indicates, GCLM modulates the activity of the enzyme and affects the steady state levels of GSH in mammalian cells and tissues.



Figure 5-2. Schematic illustration of GSH biosynthesis. As depicted, GSH is synthesized from three amino acids through two successive reactions catalyzed by γ -glutamylcysteine ligase (GCL) and glutathione synthetase (GSS), respectively. Two molecules of ATP are consumed for synthesis of each molecule of GSH. It is also worth noting that GCL is the key enzyme of the GSH biosynthesis pathway with cysteine as the rate-limiting substract.

Both subunits are subject to induction by chemical inducers and oxidative stress conditions, and upregulation of either or both of the subunits leads to increased levels of cellular GSH. Transcriptional and post-transcriptional regulation of both subunits has been described [3, 4]. Post-transcriptional regulation includes mRNA stabilization/destabilization and posttranslational modifications.

A number of transcription factors are involved in the regulation of GCLC and GCLM gene expression in mammalian cells. These include AP-1, AP-2, SP1, NF- κ B, c-Myc, Nrf1, and Nrf2 [3]. Among these transcription factors, Nrf2 appears to play the most important role in controlling both basal and inducible expression of GCL under various conditions. Regulation of GCL expression by Nrf2 occurs through an antioxidant response element-driven mechanism (Section 3.1 of Chapter 2).

While much is known about GCL regulation, little attention has been paid to GSS. Recent studies have suggested the involvement of AP-1, NF- κ B, Nrf1, and Nrf2 in the regulation of mammalian GSS gene expression in several cell types [3].

3. Role in Health and Disease

3.1. General Biological Activities

In mammalian cells and tissues, GSH participates in at least four types of biochemical reactions, as listed below. These reactions form the basis for GSH's involvement in physiology and pathophysiology.

- Reaction with reactive oxygen and nitrogen species (ROS/RNS)
- Reaction with electrophiles
- Reaction with other non-enzymatic antioxidants
- Protein deglutathionylation

3.1.1. Reaction of GSH with ROS/RNS

GSH is a major defense against ROS/RNS. GSH can directly react with ROS/RNS, leading to the detoxification of these reactive species. GSH is also used as a cofactor by glutathione peroxidase (Chapter 6) in the detoxification of hydrogen peroxide and other peroxides, as well as peroxynitrite. In addition, reaction of GSH with RNS results in the formation of Snitrosoglutathione. S-Nitrosoglutathione acts as a second messenger to transduce nitric oxide (NO) bioactivity [5].

3.1.2. Reaction of GSH with Electrophiles

Electrophiles (electron-deficient species), such as reactive aldehydes are derived from xenobiotic biotransformation as well as lipid peroxidation. GSH reacts with electrophiles, forming less reactive conjugates. Thus, conjugation with GSH represents an important mechanism of detoxification of electrophilic species. The conjugation reactions with GSH may occur spontaneously, but are markedly accelerated by glutathione S-transferase (Chapter 8). Notably, for certain xenobiotics, GSH conjugation may result in their bioactivation [6]. The term bioactivation (also known as toxification) refers to conversion of less toxic or reactive chemicals to more toxic or reactive metabolites.

3.1.3. Reaction of GSH with Other Non-protein Antioxidants

While GSH is the most abundant non-protein thiol antioxidant in mammalian cells, there are also many other cellular non-protein antioxidants, such as α -tocopherol (Chapter 21) and ascorbate (also known as vitamin C; Chapter 20). α -Tocopherol reduces lipid peroxyl radical to form lipid hydroperoxide, and in the reaction α -tocopherol is oxidized to α -tocopherol radical. The α -tocopherol radical can be reduced back to α -tocopherol by ascorbate, which is oxidized to dehydroascorbate. Dehydroascorbate can then be restored to the reduced form by a GSH-dependent reaction catalyzed by dehydroascorbate reductase (Figure 5-3). GSH deficiency results in decreased tissue levels of vitamin C in animal models, pointing to an in vivo role of GSH in vitamin C metabolism [7].

3.1.4. Protein Deglutathionylation

Reversible protein S-glutathionylation (protein-SSG) is an important post-translational modification involved in redox signaling. Analogous to protein dephosphorylation catalyzed by phosphatases, glutaredoxin isozymes (Section 2.1 of Chapter 9) using GSH as a cofactor

catalyze deglutathionylation of proteins. This participates in regulating diverse intracellular signaling pathways.



Figure 5-3. Role of GSH in the regeneration of other non-protein antioxidants, including α -tocopherol and ascorbate. See text (Section 3.1.3) for detailed description. LOO', lipid peroxyl radical; LOOH, lipid hydroperoxide; DR, dehydroascorbate reductase; GSSG, glutathione disulfide.

3.2. Animal Studies

3.2.1. Experimental Approaches

The biological activities of GSH in mammals have been investigated following modulation of its levels in cells or tissues via various approaches. The levels of cellular or tissue GSH can be elevated by employing the following four approaches.

- Delivery of membrane permeable GSH esters (Chapter 25)
- Increase of the availability of cysteine using *N*-acetylcysteine (Chapter 25)
- Induction of GSH synthesis by chemical inducers (Chapter 2)
- Transgenic overexpression of GCL

Likewise, the levels of cellular or tissue GSH can be decreased via utilizing the following two approaches.

- Buthionine sulfoximine (BSO)-mediated inhibition of GCL activity
- Targeted disruption of GCL gene (i.e., the gene knockout model)

It is noteworthy that BSO is a selective inhibitor of GCL. It has been widely used to decrease cellular or tissue GSH so as to determine the role of GSH in various physiological and pathophysiological processes [8].

Homozygous deletion of GCLC gene (GCLC^{-/-}) in mice is embryonically lethal, indicating an essential role for GCLC and GSH in development. On the other hand, homozygous deletion of GCLM gene (GCLM^{-/-}) in mice has little effects on survival or development. GCLM^{-/-} mice still have the ability to synthesize GSH in limited amounts (~10-20% of normal levels). Similarly, mice with heterozygous deletion of GCLC gene (GCLC^{+/-}) are viable and fertile. GCLC^{+/-} mice exhibit a gene-dose dependent decrease in the GCLC protein and GCL activity, but only about a 20% diminution in GSH levels [9].
3.2.2. Role in Cardiovascular Diseases

The role of GSH and GSH synthesizing enzymes in cardiovascular diseases has been investigated by employing BSO treatment or gene knockout animal models. Multiple studies show that depletion of GSH by BSO causes severe hypertension in normal animals, suggesting a critical role for GSH in controlling oxidative stress and blood pressure regulation [10]. Chronic treatment of rats with BSO decreases plasma GSH levels, resulting in modifications of plasma low-density lipoprotein (LDL). The LDL modifications can be prevented by cotreatment of the animals with GSH ethyl ester. Chronic treatment of mice with BSO also aggravates atherosclerosis in mice lacking apolipoprotein E (ApoE). ApoE-deficient mouse is susceptible to atherosclerosis, and as such is a commonly used model for studying atherogenesis and its protection. On the other hand, it is reported that an increase in macrophage GSH content reduces cell-mediated oxidation of LDL and possibly atherosclerosis in ApoEdeficient mice [11].

Depletion of myocardial GSH by BSO results in impaired post-ischemic contractile function after global ischemia in isolated perfused hearts. Oxidation of myocardial GSH also causes mitochondrial depolarization and cardiac arrhythmias in isolated perfused hearts. Recently, it has been shown that mice lacking GCLM (GCLM^{-/-}) are more susceptible to in vivo myocardial ischemia-reperfusion injury compared with wild-type animals [12]. Chronic depletion of GSH by BSO also leads to oxidative cardiomyopathy and aggravation of heart failure induced by drugs, including doxorubicin and cyclophosphamide.

S-Nitrosoglutathione has also been suggested to exert protective effects in various cardiovascular disorders. A recent study shows that endogenous S-nitrosoglutathione protects against in vivo ischemic myocardial injury in mice due, at least in part, to S-nitrosylationmediated stabilization of hypoxia-inducible factor- 1α (HIF- 1α) and increased angiogenesis [13]. The term angiogenesis refers to growth of new blood vessels from preexisting vessels.

3.2.3. Role in Diabetes

The involvement of GSH in diabetes has been investigated in cultured pancreatic islets and in vivo animal models of diabetes. In vitro studies demonstrate a critical activity for GSH and GCL in protecting against the pancreatic β -cell injury elicited by oxidative stress, inflammatory cytokines, as well as exposure to high concentrations of glucose [14, 15]. Notably, insulin induces GCL and GSH in cell cultures, suggesting a possible interaction between insulin and GSH homeostasis.

There is also evidence supporting a beneficial role for GSH in diabetes in vivo. Chronic depletion of GSH by BSO reduces insulin sensitivity and aggravates diabetes in experimental animals. Conversely, administration of GSH precursors, including GSH ethyl ester and *N*-acetylcysteine ameliorates the pathophysiology of diabetes in animal models [14, 15].

3.2.4. Role in Neurological Diseases

The protective effects of GSH in neurodegenerative disorders have been demonstrated by different approaches in experimental animals. These include the use of BSO to deplete tissue GSH and administration of GSH ethyl ester or *N*-acetylcysteine to augment tissue GSH levels, followed by determination of the changes in the pathophysiology of neurodegeneration. In normal animals, chronic treatment with BSO reduces GSH levels and induces oxidative damage in various tissues or organs, including brain [16]. GSH depletion also sensitizes the

brain to neurotoxin-induced degeneration, which can be protected by co-treatment with either GSH ethyl ester or *N*-acetylcysteine. GSH may also protect against convulsion, aging-associated memory impairment, and noise-induced hearing loss [17]. Another line of evidence supporting an important role for GSH in protecting against neurodegeneration is that GSH decrease usually precedes the neurodegenerative processes in various animal models. Furthermore, deficiency in the transporter for neuronal cysteine uptake in mice has been shown to cause impaired neuronal GSH metabolism, oxidative stress, and aging-dependent neurodegeneration [18].

Consistent with the observations in animal studies, extensive in vitro experiments also demonstrate a critical involvement of GSH in protecting against neuronal cell injury elicited by various neurotoxic insults. It is further shown that GSH is essential for the survival of primary neurons in cultures [19].

3.2.5. Role in Pulmonary Diseases

Reduced levels of GSH are commonly observed in animal models of pulmonary diseases. Administration of BSO to normal animals decreases GSH levels and causes oxidative damage and mitochondrial defects in lung tissue. These deleterious effects can be reversed by cotreatment of the animals with GSH ethyl ester [20], suggesting that GSH is a critical factor for normal physiology of the lung. There is ample evidence showing that GSH also acts as a protective molecule under various pulmonary pathophysiological conditions. For example, inhibition of GSH synthesis by BSO has been reported to potentiate hyperoxia-induced lung injury and mortality in experimental animals. GSH depletion by BSO leads to aggravation of lung injury induced by pulmonary toxins [21]. On the other hand, administration of GSH ethyl ester or other GSH precursors has been shown to attenuate oxidative lung injury, airway inflammation, and allergic asthma in animal models [21, 22].

3.2.6. Role in Hepatic and Gastrointestinal Diseases

Liver is one of the major pools of GSH in mammals. Animal studies along with cell culture experiments have established an important role for GSH in protecting against a number of liver disorders, including drug/xenobiotic-induced hepatotoxicity, alcoholic liver disease, liver ischemia-reperfusion injury, endotoxin-induced liver injury, and nonalcoholic fatty liver disease [23].

Acetaminophen-induced hepatotoxicity remains a major cause of acute liver failure. Studies using GCLC overexpression, GCLC knockdown, and GCLM knockout models demonstrate a crucial involvement of GSH in protecting against this drug-induced liver injury [9]. Although a moderate decrease in hepatic GSH levels caused by BSO doesn't lead to significant liver injury, it markedly sensitizes the animals to hepatotoxicity induced by various liver toxicants, including ethanol. BSO-mediated decrease in hepatic GSH levels also aggravates hepatic ischemia-reperfusion injury as well as endotoxin-induced liver injury, both of which can be ameliorated by co-treatment with GSH ethyl ester or other GSH precursors.

As stated above (Section 3.2.1), global homozygous deletion of GCLC gene is embryonically lethal. This makes it necessary to create models with organ-selective deletion of this gene. In this context, it has been recently shown that hepatocyte-specific GCLC deletion leads to a rapid onset of steatosis with mitochondrial injury and liver failure in mice [24]. This finding suggests that GSH is essential for preserving normal hepatic function and for protecting against the development of nonalcoholic fatty liver disease. In line with this notion, increases of liver GSH by GSH precursors attenuate the progression of nonalcoholic fatty liver disease in animal models [25].

Multiple animal studies demonstrate beneficial effects of GSH in certain gastrointestinal disorders, including experimental colitis and intestinal inflammation induced by bacterial infections [26, 27]. GSH and GSH-related enzymes serve as an important line of defense against oxidative stress and inflammatory injury as well as toxic chemicals in the gut.

3.2.7. Role in Renal Diseases

In contrast to its well-established benefits in many other organs/tissues, the role of GSH in renal physiology and pathophysiology has been a subject of controversies. Early studies showed that GSH ethyl ester aggravated renal ischemia-reperfusion injury, whereas decrease of renal GSH by BSO afforded a protective effect, suggesting a possible detrimental action of GSH in renal ischemia-reperfusion injury [28]. However, administration of *N*-acetylcysteine, another GSH precursor was shown to attenuate renal ischemia-reperfusion injury [29]. It has further been reported that decrease of renal GSH by BSO leads to a concomitant induction of heme oxygenase, which may function to protect against renal ischemia-reperfusion injury [30]. As discussed in Chapter 15, heme oxygenase is an important antioxidant defense against tissue ischemia-reperfusion injury.

The role of GSH in nephrotoxicity induced by drugs and toxicants also varies with the animal models and the nephrotoxicants examined. For example, multiple studies report a potentiation effect of GSH on cisplatin-induced renal injury though other studies show the opposite. GSH mediates bioactivation of certain xenobiotics, such as polychlorinated alkenes, causing nephrotoxicity. In contrast, renal GSH is involved in protecting against kidney injury induced by various other nephrotoxicants, including heavy metals, radiographic contrast materials, and the anticancer drug ifosfamide.

3.2.8. Role in Skin Diseases

Oxidative stress and dysregulated inflammatory responses are frequently involved in skin disorders, including ultraviolet (UV) light-induced skin injury and contact dermatitis. Administration of BSO decreases GSH levels in skin, and sensitizes the animal to both UV light-induced skin injury and contact dermatitis. Conversely, treatment of the animals with GSH ethyl ester or *N*-acetylcysteine attenuates oxidative injury and inflammation in skin tissue [31]. Moreover, GSH is also found to participate in the regulation of skin wound healing in an animal model [32].

3.2.9. Role in Cancer

GSH plays diverse roles in cancer. Elevation of GSH in tumor cells is a major mechanism of anticancer drug resistance. Depletion of tumor cell GSH by BSO has been shown to potentiate the tumor killing activity of various anticancer drugs, such as cisplatin and doxorubicin. However, BSO co-treatment also potentiates the adverse effects of anticancer drugs on normal tissues, which may become a limiting factor for the combined strategy. In vitro studies suggest that high GSH levels may promote tumor cell growth and invasion.

Both ROS and electrophiles are critically involved in chemical carcinogenesis. Depletion of tissue GSH by BSO has been reported to cause DNA damage in normal experimental animals, and potentiate carcinogenesis induced by chemical carcinogens, including aflatoxin B1

[33]. Aflatoxin B1 is a potent human liver carcinogen. Likewise, elevation of tissue GSH levels by GSH precursors or chemical inducers has been demonstrated to be effective in protecting against chemical- as well as UV light-induced carcinogenesis in experimental animals. In addition to inhibiting chemical carcinogenesis, GSH also plays a protective role in spontaneous tumor development. Recently, it is reported that GSH depletion by BSO results in a 5-fold increase in the incidence of colonic tumors in p53-knockout mice, suggesting that GSH deficiency and loss of p53 function may have a synergistic effect on spontaneous colonic tumor development [34].

3.2.10. Role in Other Diseases and Conditions

In addition to the diseases described above, GSH also has a beneficial role in various other disorders and conditions, including HIV infection, scurvy, cataract, and aging in animal models. Moreover, GSH serves as a regulator of immune cell responses, which may participate in controlling viral replication [35]. Indeed, the effectiveness of the GSH precursor *N*acetyl-cysteine in protecting against HIV infection has been demonstrated in experimental animals and suggested in certain clinical trials [36].

3.3. Human Studies and Clinical Perspectives

The well-documented biological activities of GSH in diverse animal models of human diseases have prompted translational and clinical research on GSH over the last two decades. Multiple GSH precursors or analogs have been investigated in clinical trials for the intervention of human diseases. Studies have also identified mutations and gene polymorphisms in GSH synthesizing enzymes and their potential role in disease processes. This section begins with a description of hereditary disorders resulting from gene mutations of GSH synthesizing enzymes. It then summarizes the major findings from gene polymorphism studies and interventional clinical trials on GSH precursors.

3.3.1. GCL and GSS Deficiencies due to Gene Mutations

Mutations have been identified in both GCLC and GSS genes. It has been demonstrated that missense mutations in the heavy subunit gene of GCL cause decreased GCL activity and GSH levels in erythrocytes, and the patients with this rare hereditary GCL deficiency suffer from hemolytic anemia [37]. Glutathione synthetase (GSS) deficiency is also a rare hereditary disorder with decreased GSS activity due to mutations in the corresponding gene. Patients with GSS deficiency also suffer from hemolytic anemia as well as 5-oxoprolinuria [38]. For both GCL and GSS deficiency, hemolytic anemia is believed to result from oxidative stress in red blood cells due to decreased cellular GSH levels. 5-Oxoprolinuria in GSS-deficient patients results from the accumulation of γ -glutamylcysteine and its subsequent conversion to 5-oxoproline. As shown in Figure 5-2, GSS catalyzes the reaction between γ -glutamylcysteine and glycine to form the tripeptide GSH. Hence, GSS deficiency causes accumulation of the substrate γ -glutamylcysteine.

3.3.2. Gene Polymorphisms

A variety of gene polymorphisms exist for both GCLC and GCLM. Many observational

epidemiological studies demonstrate that gene polymorphisms of GCLC or GCLM are associated with a wide variety of disease conditions, including ischemic heart disease, atherosclerosis, hypertension, diabetes, schizophrenia, chronic obstructive pulmonary disease, cystic fibrosis, nonalcoholic liver disease, and certain types of cancer [39-43]. Gene polymorphism studies may thus help identify subpopulations at high risk of developing the above disease processes, and such knowledge may serve as a guide for disease risk assessment and early intervention.

3.3.3. Interventional Clinical Trials

Decreased levels of GSH have been observed in a number of human disease conditions, including those listed above in Section 3.3.2. It is rational that repletion of GSH may thus protect against the disease processes. Multiple clinical trials have suggested a potential benefit of using GSH precursors, especially *N*-acetylcysteine in the intervention of certain human diseases, including hypertension, diabetes, chronic obstructive pulmonary disease, and cystic fibrosis [21, 44-46] (more detailed discussion in Chapter 25).

4. Conclusion and Future Directions

As the most abundant non-protein thiol in mammalian cells, GSH participates in numerous cellular processes that are critical for proper physiological functions of organs/systems. In this context, depletion of GSH in animal models aggravates a series of pathophysiological processes, which can be corrected, at least partially, by administration of GSH or GSH precursors. There are also multiple lines of evidence suggesting a beneficial role for GSH in human health and disease. Although significant progress has been made in the research area of GSH and its clinical applications, there is still much to be done to better understand how GSH synthesis and metabolism may be manipulated to therapeutic advantage. In this regard, future studies should focus on developing novel chemical agents (including both GSH enhancing and depleting agents) that can be used to selectively modulate GSH levels in target tissues to achieve therapeutic goals. It may be pertinent to suggest that therapeutic research involving GSH or its precursors must be carefully designed to take into account of the pharmacokinetic and pharmacodynamic profiles as well as the potential adverse effects of the thiol-containing compounds.

5. References

- [1] Hopkins FG. On glutathione: a reinvestigation. J. Biol. Chem. 1929;84:269-320.
- [2] Harington CR, Mead TH. Synthesis of glutathione. *Biochem. J.* 1935;29:1602-1611.
- [3] Lu SC. Regulation of glutathione synthesis. *Mol. Aspects Med.* 2009;30:42-59.
- [4] Franklin CC, Backos DS, Mohar I, White CC, Forman HJ, Kavanagh TJ. Structure, function, and post-translational regulation of the catalytic and modifier subunits of glutamate cysteine ligase. *Mol. Aspects Med.* 2009;30:86-98.
- [5] Foster MW, Hess DT, Stamler JS. Protein S-nitrosylation in health and disease: a current perspective. *Trends Mol. Med.* 2009;15:391-404.

- [6] Anders MW. Chemical toxicology of reactive intermediates formed by the glutathionedependent bioactivation of halogen-containing compounds. *Chem. Res. Toxicol.* 2008; 21:145-159.
- [7] Meister A. Glutathione-ascorbic acid antioxidant system in animals. J. Biol. Chem. 1994;269:9397-9400.
- [8] Meister A. Glutathione deficiency produced by inhibition of its synthesis, and its reversal; applications in research and therapy. *Pharmacol. Ther.* 1991;51:155-194.
- [9] Botta D, White CC, Vliet-Gregg P, Mohar I, Shi S, McGrath MB, *et al.* Modulating GSH synthesis using glutamate cysteine ligase transgenic and gene-targeted mice. *Drug Metab. Rev.* 2008;40:465-477.
- [10] Vaziri ND, Wang XQ, Oveisi F, Rad B. Induction of oxidative stress by glutathione depletion causes severe hypertension in normal rats. *Hypertension* 2000;36:142-146.
- [11] Rosenblat M, Coleman R, Aviram M. Increased macrophage glutathione content reduces cell-mediated oxidation of LDL and atherosclerosis in apolipoprotein Edeficient mice. *Atherosclerosis* 2002;163:17-28.
- [12] Kobayashi T, Watanabe Y, Saito Y, Fujioka D, Nakamura T, Obata JE, *et al.* Mice lacking the glutamate-cysteine ligase modifier subunit are susceptible to myocardial ischaemia-reperfusion injury. *Cardiovasc. Res.* 2010;85:785-795.
- [13] Lima B, Lam GK, Xie L, Diesen DL, Villamizar N, Nienaber J, et al. Endogenous Snitrosothiols protect against myocardial injury. Proc. Natl. Acad. Sci. USA 2009;106: 6297-6302.
- [14] Robertson RP. Antioxidant drugs for treating beta-cell oxidative stress in type 2 diabetes: glucose-centric versus insulin-centric therapy. *Discov. Med.* 2010;9:132-137.
- [15] Robertson RP, Harmon J, Tran PO, Tanaka Y, Takahashi H. Glucose toxicity in betacells: type 2 diabetes, good radicals gone bad, and the glutathione connection. *Diabetes* 2003;52:581-587.
- [16] Jain A, Martensson J, Stole E, Auld PA, Meister A. Glutathione deficiency leads to mitochondrial damage in brain. Proc. Natl. Acad. Sci. USA 1991;88:1913-1917.
- [17] Yamasoba T, Nuttall AL, Harris C, Raphael Y, Miller JM. Role of glutathione in protection against noise-induced hearing loss. *Brain Res.* 1998;784:82-90.
- [18] Aoyama K, Suh SW, Hamby AM, Liu J, Chan WY, Chen Y, *et al.* Neuronal glutathione deficiency and age-dependent neurodegeneration in the EAAC1 deficient mouse. *Nat. Neurosci.* 2006;9:119-126.
- [19] Diaz-Hernandez JI, Almeida A, Delgado-Esteban M, Fernandez E, Bolanos JP. Knockdown of glutamate-cysteine ligase by small hairpin RNA reveals that both catalytic and modulatory subunits are essential for the survival of primary neurons. J. Biol. Chem. 2005;280:38992-39001.
- [20] Martensson J, Jain A, Frayer W, Meister A. Glutathione metabolism in the lung: inhibition of its synthesis leads to lamellar body and mitochondrial defects. *Proc. Natl. Acad. Sci. USA* 1989;86:5296-5300.
- [21] Biswas SK, Rahman I. Environmental toxicity, redox signaling and lung inflammation: the role of glutathione. *Mol. Aspects Med.* 2009;30:60-76.
- [22] Rahman I. Regulation of glutathione in inflammation and chronic lung diseases. *Mutat. Res.* 2005;579:58-80.
- [23] Yuan L, Kaplowitz N. Glutathione in liver diseases and hepatotoxicity. Mol. Aspects Med. 2009;30:29-41.

- [24] Chen Y, Yang Y, Miller ML, Shen D, Shertzer HG, Stringer KF, *et al.* Hepatocytespecific Gclc deletion leads to rapid onset of steatosis with mitochondrial injury and liver failure. *Hepatology* 2007;45:1118-1128.
- [25] Baumgardner JN, Shankar K, Hennings L, Albano E, Badger TM, Ronis MJ. Nacetylcysteine attenuates progression of liver pathology in a rat model of nonalcoholic steatohepatitis. J. Nutr. 2008;138:1872-1879.
- [26] Oz HS, Chen TS, Nagasawa H. Comparative efficacies of 2 cysteine prodrugs and a glutathione delivery agent in a colitis model. *Transl. Res.* 2007;150:122-129.
- [27] van Ampting MT, Schonewille AJ, Vink C, Brummer RJ, van der Meer R, Bovee-Oudenhoven IM. Intestinal barrier function in response to abundant or depleted mucosal glutathione in Salmonella-infected rats. *BMC Physiol.* 2009;9:6.
- [28] Scaduto RC, Jr., Gattone VH, 2nd, Grotyohann LW, Wertz J, Martin LF. Effect of an altered glutathione content on renal ischemic injury. Am. J. Physiol. 1988;255:F911-921.
- [29] Di Giorno C, Pinheiro HS, Heinke T, Franco MF, Galante NZ, Pacheco-Silva A, *et al.* Beneficial effect of N-acetyl-cysteine on renal injury triggered by ischemia and reperfusion. *Transplant. Proc.* 2006;38:2774-2776.
- [30] Horikawa S, Yoneya R, Nagashima Y, Hagiwara K, Ozasa H. Prior induction of heme oxygenase-1 with glutathione depletor ameliorates the renal ischemia and reperfusion injury in the rat. *FEBS Lett.* 2002;510:221-224.
- [31] Steenvoorden DP, Beijersbergen van Henegouwen G. Glutathione ethylester protects against local and systemic suppression of contact hypersensitivity induced by ultraviolet B radiation in mice. *Radiat. Res.* 1998;150:292-297.
- [32] Adamson B, Schwarz D, Klugston P, Gilmont R, Perry L, Fisher J, *et al.* Delayed repair: the role of glutathione in a rat incisional wound model. *J. Surg. Res.* 1996;62: 159-164.
- [33] Hiruma S, Kimura M, Lehmann K, Gopalan-Kriczky P, Qin GZ, Shinozuka H, et al. Potentiation of aflatoxin B1-induced hepatocarcinogenesis in the rat by pretreatment with buthionine sulfoximine. *Cancer Lett.* 1997;113:103-109.
- [34] Richie JP, Jr., Komninou D, Albino AP. Induction of colon tumorigenesis by glutathione depletion in p53-knock-out mice. *Int. J. Oncol.* 2007;30:1539-1543.
- [35] Yan Z, Garg SK, Kipnis J, Banerjee R. Extracellular redox modulation by regulatory T cells. *Nat. Chem. Biol.* 2009;5:721-723.
- [36] Fraternale A, Paoletti MF, Casabianca A, Nencioni L, Garaci E, Palamara AT, *et al.* GSH and analogs in antiviral therapy. *Mol. Aspects Med.* 2009;30:99-110.
- [37] Ristoff E, Augustson C, Geissler J, de Rijk T, Carlsson K, Luo JL, *et al.* A missense mutation in the heavy subunit of gamma-glutamylcysteine synthetase gene causes hemolytic anemia. *Blood* 2000;95:2193-2196.
- [38] Shi ZZ, Habib GM, Rhead WJ, Gahl WA, He X, Sazer S, *et al.* Mutations in the glutathione synthetase gene cause 5-oxoprolinuria. *Nat. Genet.* 1996;14:361-365.
- [39] Koide S, Kugiyama K, Sugiyama S, Nakamura S, Fukushima H, Honda O, *et al.* Association of polymorphism in glutamate-cysteine ligase catalytic subunit gene with coronary vasomotor dysfunction and myocardial infarction. *J. Am. Coll. Cardiol.* 2003; 41:539-545.
- [40] Bekris LM, Shephard C, Janer M, Graham J, McNeney B, Shin J, *et al.* Glutamate cysteine ligase catalytic subunit promoter polymorphisms and associations with type 1

diabetes age-at-onset and GAD65 autoantibody levels. *Exp. Clin. Endocrinol. Diabetes* 2007;115:221-228.

- [41] Gysin R, Kraftsik R, Sandell J, Bovet P, Chappuis C, Conus P, *et al.* Impaired glutathione synthesis in schizophrenia: convergent genetic and functional evidence. *Proc. Natl. Acad. Sci. USA* 2007;104:16621-16626.
- [42] McKone EF, Shao J, Frangolias DD, Keener CL, Shephard CA, Farin FM, et al. Variants in the glutamate-cysteine-ligase gene are associated with cystic fibrosis lung disease. Am. J. Respir. Crit. Care Med. 2006;174:415-419.
- [43] Oliveira CP, Stefano JT, Cavaleiro AM, Zanella Fortes MA, Vieira SM, Rodrigues Lima VM, et al. Association of polymorphisms of glutamate-cystein ligase and microsomal triglyceride transfer protein genes in non-alcoholic fatty liver disease. J. Gastroenterol. Hepatol. 2010;25:357-361.
- [44] Tirouvanziam R, Conrad CK, Bottiglieri T, Herzenberg LA, Moss RB. High-dose oral N-acetylcysteine, a glutathione prodrug, modulates inflammation in cystic fibrosis. *Proc. Natl. Acad. Sci. USA* 2006;103:4628-4633.
- [45] Atkuri KR, Mantovani JJ, Herzenberg LA. N-Acetylcysteine: a safe antidote for cysteine/glutathione deficiency. *Curr. Opin. Pharmacol.* 2007;7:355-359.
- [46] Martina V, Masha A, Gigliardi VR, Brocato L, Manzato E, Berchio A, et al. Long-term N-acetylcysteine and L-arginine administration reduces endothelial activation and systolic blood pressure in hypertensive patients with type 2 diabetes. Diabetes Care 2008;31:940-944.

Chapter 6

Glutathione Peroxidase

Abstract

Mammalian glutathione peroxidase (GPx) represents a family of six isozymes, namely, GPx1-6. GPxs are critical antioxidant enzymes catalyzing the decomposition of hydrogen peroxide or organic hydroperoxides. Studies in both transgenic overexpression and gene knockout animal models demonstrate an important function for GPx isozymes, especially GPx1-4 in protecting against a wide range of disease processes. Although being not as convincing as that from animal experiments, evidence from human studies also suggests a beneficial role for GPxs in certain disease conditions, particularly cardiovascular disorders.

1. Overview

1.1. Definition and History

Glutathione peroxidase (GPx) is the general name for a family of multiple isozymes that catalyze the reduction of hydrogen peroxide or organic hydroperoxides to water or corresponding alcohols using the reduced form of glutathione (GSH) as the electron donor [1, 2]. In mammalian tissues, there are six GPx isozymes, namely, GPx1, 2, 3, 4, 5, and 6. GPx1, 2, 3, and 4 are selenoproteins. GPx6 is also a selenoprotein in humans and pigs, but not in rats and mice. GPx5 is a non-selenoprotein. Here the term selenoprotein refers to any protein that contains one or more selenocysteine residues. Selenocysteine, also known as the 21st amino acid, has a structure similar to cysteine, but with an atom of selenium taking the place of the usual sulfur.

Twenty-five selenoproteins are identified in humans, among which are many antioxidant enzymes [3]. These selenium-containing antioxidant enzymes include the aforementioned GPxs (GPx1-4, and 6), thioredoxin reductases (TrxR1, 2, and TGR; Chapter 12), and methionine sulfoxide reductase B1 (MsrB1, also known as selenoprotein R; Chapter 14). The other selenoproteins include iodothyronine deiodinases (DIO1, 2, and 3), selenoprotein H (Sel H), Sel I, Sel K, Sel M, Sept15, Sel N, Sel O, Sel P, Sel S, Sel T, Sel V, Sel W, and selenophosphate synthetase 2 (SPS2) [4]. These selenoproteins have diverse biological functions. Some of them, such as Sel P, Sel S and Sel W also possess antioxidant activities. Selenium deficiency and mutations or polymorphisms in selenoprotein genes possess a strong correlation with a variety of human diseases. These include muscle and cardiovascular disorders, immune dysfunction, cancer, neurological disorders, male infertility, and endocrine dysfunction [5]. This chapter focuses on GPx isozymes, the most extensively investigated selenoproteins in biology and medicine.

GPx activity was first reported in erythrocytes by G.C. Mills in 1957 [6]. The enzyme was found to protect hemoglobin from oxidative breakdown. This enzyme was further demonstrated to be a selenoenzyme by L. Flohe and coworkers in 1973 [7]. The enzyme was later named GPx1, and has become the most extensively studied isozyme of the GPx family. Later, several additional mammalian GPx isozymes were isolated and characterized. As stated above, in mammals the GPx family now has six members, and the classification is based on their amino acid sequence, substrate specificity, and subcellular localization.

1.2. Basic Characteristics

GPx1, 2, 3, and 6 are homotetrameric proteins with a subunit molecular mass of ~20-25 kDa, whereas GPx4 is a 20-22 kDa monomeric enzyme. GPx5 is a homodimeric enzyme with a subunit molecular mass of 24 kDa. The nomenclature and cellular and chromosomal localization of the various GPx isozymes are described below.

- GPx1: Also known as classical or cytosolic GPx (cGPx), GPx1 is the first mammalian GPx identified and one of the most abundant and ubiquitously expressed selenoproteins. It is present in cytosol, mitochondria, and nuclei. In humans, GPx1 is localized on chromosome 3p21.3.
- GPx2: Also known as gastrointestinal GPx (GI-GPx), GPx2 is mainly expressed in gastrointestinal tract. It is also found in liver and lung. GPx2 is present in cytosol and nuclei. In humans, GPx2 is localized on chromosome 14q24.1.
- 3) GPx3: Also known as plasma GPx (pGPx), GPx3 is a secreted form of enzyme found in plasma. Kidney is the major source of GPx3 in plasma. This enzyme is also expressed in many other tissues, including lung and heart, and is present in cytosol. In humans, GPx3 is localized on chromosome 5q23.
- 4) GPx4: Also known as phospholipid hydroperoxide GPx (PHGPx), GPx4 enzyme is ubiquitously expressed in a variety of tissues. GPx4 is also a main structural component of the sperm mitochondrial capsule in mature spermatozoa, where it exists as an enzymatically-inactive, oxidatively cross-linked, insoluble protein [8]. Subcellular distribution of GPx4 includes cytosol, nuclei, mitochondria, and membranes. In humans, GPx4 is localized on chromosome 19p13.3.
- 5) GPx5: Also known as epididymal GPx (eGPx), this GPx isozyme is selectively expressed in epididymis. It is secreted into the epididymal lumen. In humans, GPx5 is localized on chromosome 6p22.1.
- 6) GPx6: GPx6 is a recently identified isozyme. The expression of GPx6 is believed to be restricted to the developing embryo and in olfactory epithelium in adults, suggesting a possible role in embryonic development and olfactory function. In humans, GPx6 is localized on chromosome 6p22.1.

2. Biochemistry and Molecular Regulation

2.1. Biochemistry

All of the GPx isozymes are able to catalyze the reduction of hydrogen peroxide (H_2O_2) or organic hydroperoxides (LOOH) to water or corresponding alcohols (LOH) using GSH as the electron donor (Figure 6-1). GPx4 is also able to reduce phospholipid hydroperoxides in membranes, and as such is named phospholipid hydroperoxide GPx (PHGPx). As a main structural component of the sperm mitochondrial capsule in mature spermatozoa, GPx4 is also involved in sperm maturation and male fertility [9]. Due to the low levels of GSH in extracellular fluid, GPx3 may also use extracellular thioredoxin and glutaredoxin as electron donors. In addition to decomposing H_2O_2 and LOOH, GPx isozymes can also reduce peroxynitrite in vitro [10].



Figure 6-1. Glutathione peroxidase (GPx)-catalyzed decomposition of peroxides. As illustrated, GPx catalyzes decomposition of hydrogen peroxide (H_2O_2) or organic hydroperoxides (LOOH) to form water or the corresponding alcohols (LOH). The reactions require GSH as the electron donor. GSSG stands for glutathione disulfide.

2.2. Molecular Regulation

Although the first member of GPx family was identified more than fifty years ago, regulation of gene expression for many of the GPx isozymes is not well-understood. Regulation of GPxs may occur at both transcriptional and post-transcriptional levels [11]. It has been reported that p53 is a critical transcription factor involved in the modulation of GPx1 gene expression. Indeed, a p53-binding site is mapped to the promoter region of GPx1 gene. GPx1 is also subject to post-translational modifications by c-Abl and Arg tyrosine kinases [12]. Multiple transcription factors participate in controlling GPx2 gene expression. These include Nrf2, β -catenin/TCF, and p63 (a member of the p53 tumor suppressor family) [11, 13]. The promoter region of GPx3 has been characterized, and various transcription factors, including AP-1, SP1, and HIF-1 have been implicated in the transcriptional regulation [14]. The transcriptional regulation of GPx4 gene has recently been extensively investigated. A range of transcription factors are shown to participate in regulating GPx4 gene in cellular systems. These factors include C/EBP-epsilon, NF- κ B, SP1, SP3, NF-Y, and Smad [11, 15]. In contrast to the above four isozymes of GPx family, studies on gene regulation of the recently identified members GPx5 and GPx6 are scarce.

3. Role in Health and Disease

3.1. Animal Studies

3.1.1. Experimental Approaches

The biological activities of GPxs have been studied in numerous animal models over the past several decades. Gene knockout or transgenic overexpession studies of the individual GPx isozymes provide the most compelling data on their role in health and disease. Transgenic overexpression of GPx1-5 has been created in animal models. Knockout mouse models for GPx1-5 have also been reported in the literature. Notably, homozygous deletion of GPx4 is embryonically lethal [16]. Primarily based on studies with these gene knockout and transgenic overexpression models, GPx isozymes are found to play important roles in various physiological and pathophysiological processes. In addition to the genetic models, the GPx mimetic ebselen has also been used for disease intervention in both animal models and clinical trials (Chapter 24).

3.1.2. Role in Cardiovascular Diseases

Cardiovascular diseases are among the most extensively studied disease conditions with respect to the role of GPxs in animal models of human diseases. Among the various GPx isozymes, GPx1 has received the most attention. Both gene overexpression and knockout studies have demonstrated a critical protective role for GPx1 in hypertension, myocardial ischemiareperfusion injury, atherosclerosis, and heart failure [17-19]. It has been also found that GPx1 overexpression ameliorates virus-induced myocarditis [20]. In addition to GPx1, there are reports showing beneficial effects of GPx4 in atherosclerosis and myocardial ischemiareperfusion injury in animal models [21, 22].

3.1.3. Role in Diabetes

In vitro studies have conclusively demonstrated a role for GPxs in protecting the pancreatic β -cells from glucose toxicity and oxidative stress. However, the involvement of GPx1 in diabetes in animal models has been a subject of controversies. For example, one study reports that mice with global transgenic overexpression of GPx1 spontaneously develop insulin resistance and obesity [23]. In contrast, another study shows that the pancreatic β -cell-selective overexpression of GPx1 reverses diabetes in db/db mice [24]. Similarly, GPx3 overexpression in adipocytes results in improvement of high glucose-induced insulin resistance and attenuation of proinflammatory gene expression, whereas GPx3 neutralization in adipocytes promotes expression of proinflammatory genes [25]. In addition, reduced plasma levels of GPx3 are found in diabetic mice, further suggesting a protective role for this GPx isozyme in experimental diabetes.

3.1.4. Role in Neurological Diseases

Studies using both transgenic overexpression and gene knockout animal models convincingly show GPx1 as an important cellular factor in protecting against various neurological disorders, including Parkinson's disease, Alzheimer's disease, and cerebral ischemiareperfusion injury [26-28]. Recently, the role of GPx4 in neuroprotection has also been investigated. As stated above in Section 3.1.1, GPx4 is essential for embryonic development as GPx4 knockout (GPx4^{-/-}) mice die in utero by midgestation. Heterozygous (GPx4^{+/-}) mice are born and able to reproduce. However, these mice spontaneously develop neurodegeneration, suggesting a critical involvement of GPx4 in normal physiology of the brain. Brains of GPx4^{+/-} mice are found to have increased lipid peroxidation, which leads to increased amyloidogenesis in experimental Alzheimer's disease [29]. Selective depletion of GPx4 in hippocampus is also shown to result in neurodegeneration, and GPx4 is directly involved in proapoptotic signaling, leading to neuron death [30]. In vitro studies demonstrate that the neuroprotective effects of GPx4 result from its function in removal of lipid hydroperoxides [31]. In this context, GPx4 is associated with cell membrane and acts a major defense against membrane lipid peroxidation. Collectively, these findings support GPx4 acting as an important protective molecule in neurodegenerative diseases.

3.1.5. Role in Pulmonary Diseases

GPx1-deficient mice develop normally and show no increased sensitivity to hyperoxiainduced lung injury [32]. However, as compared with wild-type mice, GPx1-deficient mice are more susceptible to lung injury and mortality induced by lethal amounts of paraquat, a redox-cycling compound [33]. GPx2 in lung tissue is induced by exposure to cigarette smoke, and the induction occurs via an Nrf2-dependent mechanism [34]. The Nrf2-dependent upregulation of GPx2 may be a potential mechanism against cigarette smoke-induced lung injury. Targeted disruption of GPx2 gene in mice sensitizes the animals to allergen-induced airway inflammation, suggesting that GPx2 may also be an important defense against inflammation in the lung [35]. GPx3 is found to be induced in asthmatic lungs via a redox-regulated mechanism; however, the exact role of this isozyme in protecting against oxidative and inflammatory lung injury remains unclear [36].

3.1.6. Role in Hepatic and Gastrointestinal Diseases

GPx1 has been shown to protect liver from oxidative and inflammatory injury induced by diaquat, ethanol, and endotoxin in animal models [37-39]. However, knockout of GPx1 in mice renders the animals more resistant to acetaminophen-induced hepatotoxicity, and GPx1 overexpression sensitizes the animals to this drug-induced liver injury [40, 41]. As noted earlier, GPx1 is an intracellularly expressed protein. In contrast, either overexpression of the plasma GPx3 or intravenous administration of GPx protein is shown to protect against acetaminophen liver toxicity in mice [41]. Under either condition, the GPx protein is mainly present in extracellular fluids. It remains unclear why intracellular overexpression of GPx1 increases acetaminophen-induced liver injury in animal models. It is suggested that this sensitivity may be due to the inability of the GPx1-overexpressing mice to efficiently recover GSH depletion as a result of acetaminophen metabolism. Transgenic overexpression of GPx4 in mice is shown to protect against diaquat-induced lipid peroxidation and apoptosis in liver tissue [42].

Multiple studies demonstrate that GPx isozymes, particularly GPx2 play an important part in protecting against gastrointestinal inflammation. Mice with combined disruption of GPx1 and GPx2 have been shown to spontaneously develop chronic colitis, with symptoms and pathological alterations consistent with inflammatory bowel disease [43]. This finding suggests GPx2 as an integral part of gut homeostasis. As aforementioned, GPx2 is mainly expressed in gastrointestinal tract, and thus it may also serve as an important line of antioxidant defense against peroxides either generated in the gut or introduced via diet.

3.1.7. Role in Renal Diseases

GPx3 is secreted primarily by renal tubular cells. Recently, GPx3 is found to bind to the basal membranes of mouse renal cortex tubular cells [44]. It is further shown that GPx3 overexpression protects against renal ischemia-reperfusion-elicited inflammatory response and kidney injury in mice [45]. Mice overexpressing GPx1 are also more resistant than wild-type littermates to renal ischemia-reperfusion injury [45]. However, GPx1 knockout in mice does not sensitize the animals to streptozotocin-induced diabetic nephropathy [46]. Streptozotocin is a compound that causes cytotoxicity to the pancreatic β -cells, leading to subsequent development of type 1 diabetes in experimental animals.

3.1.8. Role in Skin Diseases

GPx4 exhibits high specific activity in reducing lipid hydroperoxides, and thus may play a critical role in protecting skin from ultraviolet (UV) irradiation-triggered long-term effects like premature skin aging and cancer. Indeed, overexpression of GPx4 in human dermal fibroblasts suppresses UV irradiation-induced lipid peroxidation, inflammation, and activation of interstitial collagenase [47]. Skin cancer cells are shown to have a reduced GPx activity and an elevated peroxidative burden. In vivo knockout of GPx2 predisposes the mice to UV irradiation-induced formation of squamous cell carcinoma [48]. In addition to skin cancer, GPx isozymes also play a critical role in protecting against other types of cancer (see Section 3.1.9 below).

3.1.9. Role in Cancer

ROS, including hydroperoxides are implicated in multistage carcinogenesis. Thus, by decomposition of hydroperoxides, GPxs may function to protect against spontaneous cancer development [49]. In cell cultures, overexpression of GPxs inhibits tumor cell growth and invasiveness, whereas inactivation of GPxs results in increased oxidative DNA damage. Double-knockout of GPx1 and GPx2 causes intestinal inflammation and an increased incidence of intestinal cancer in mice [50]. Deficiency in selenoproteins, including GPxs accelerates prostate carcinogenesis in a transgenic mouse model [51]. Conversely, overexpression of GPx3 in prostate cancer cells is found to suppress the cancer cell growth and metastasis in a xenograft animal model [52]. Together, these observations support a protective role for GPxs in cancer development.

3.1.10. Role in Other Diseases and Conditions

GPxs are implicated in various other physiological and pathophysiological processes, including reproduction, eye disorders, immune regulation, and sepsis. As stated earlier in Section 1.2, GPx4 is not only an antioxidant enzyme, but also a structural protein in spermatozoa. Depletion of GPx4 in spermatocytes causes male infertility in mice [9]. GPx5 is a nonselenoprotein that is selectively expressed in epididymidis, and secreted in significant amounts into the epididymal lumen. This enzyme is suggested to act as an important antioxidant defense in the epididymal lumen to protect spermatozoa from oxidative stress [53]. GPxs may also be an important antioxidant defense of lens in vivo. GPx1 deficiency leads to increased cataract in mice [54]. Recently, it is reported that increased expression of GPx4 in photoreceptors of mice strongly protects retina from oxidative damage [55]. Transgenic mice overexpressing GPx1 and GPx3 are more resistant than wild-type counterparts to endotoxemic conditions, as evidenced by a decreased inflammation and hypotension, and an increased survival [56].

3.2. Human Studies and Clinical Perspectives

The potential involvement of GPxs in human health and disease has been investigated over the past two decades primarily by two approaches: (1) detection of the changes of GPx isozymes in human tissues under various disease conditions and (2) observational epidemiological studies on the association between GPx gene polymorphisms and altered risk of disease development. While these studies in human subjects provide important information on GPx isozymes in human health and disease, it should be borne in mind that neither approach is able to establish a causative role for GPxs in a particular disease condition.

In line with the well-established role for GPx1 in protecting against cardiovascular disorders in animal models, a low activity of red blood cell GPx1 has been demonstrated to be independently associated with an elevated risk of cardiovascular events in humans [57]. Gene polymorphisms have been identified in human GPx1, and individuals possessing one or two ALA6 alleles have been shown to be at a modest increased risk of coronary artery disease [58]. The presence of Pro197Leu substitution of the GPx1 gene may also play a crucial role in determining the genetic susceptibility to coronary atherosclerosis in type 2 diabetes [59]. Recently, multiple studies have shown that the promoter polymorphisms in GPx3 gene are associated with an increased risk of arterial ischemic stroke as well as cerebral venous thrombosis [60, 61], though one study has reported no association of GPx3 gene with cerebral venous thrombosis in a German population [62]. In addition to cardiovascular diseases, several population studies also demonstrate an association between GPx gene polymorphisms and increased risk of developing various types of cancer, including breast cancer, prostate cancer, and colorectal carcinoma [63-66].

4. Conclusion and Future Directions

As a major defense in the detoxification of peroxides, mammalian GPxs play a critical role in protecting against various pathophysiological conditions that involve oxidative stress and inflammation. Both gene knockout and overexpression animal models have contributed greatly to our current understanding of the biological activities of various GPx isozymes in mammals. Such knowledge has facilitated the studies on the role of GPxs in human health and disease. In this context, multiple observational epidemiological studies in human subjects suggest a potential beneficial function for GPxs in certain human disorders, especially cardiovascular diseases. Future studies should focus on investigating the involvement of GPxs in other human disease conditions, and emphasize both basic and translational research on the newly identified GPx isozymes, GPx5 and 6. Such studies will provide novel insights into the complementary actions of various GPx isozymes in disease protection. Future research should also be aimed at the development of pharmacological approaches to augmenting tissue GPx expression and the potential applications of GPx manipulation in the prevention and treatment of common human disorders.

5. References

- [1] Herbette S, Roeckel-Drevet P, Drevet JR. Seleno-independent glutathione peroxidases: more than simple antioxidant scavengers. *FEBS J*. 2007;274:2163-2180.
- [2] Margis R, Dunand C, Teixeira FK, Margis-Pinheiro M. Glutathione peroxidase family: an evolutionary overview. *FEBS J.* 2008;275:3959-3970.
- [3] Kryukov GV, Castellano S, Novoselov SV, Lobanov AV, Zehtab O, Guigo R, *et al.* Characterization of mammalian selenoproteomes. *Science* 2003;300:1439-1443.
- [4] Reeves MA, Hoffmann PR. The human selenoproteome: recent insights into functions and regulation. *Cell Mol. Life Sci.* 2009;66:2457-2478.
- [5] Lu J, Holmgren A. Selenoproteins. J. Biol. Chem. 2009;284:723-727.
- [6] Mills GC. Hemoglobin catabolism. I. Glutathione peroxidase, an erythrocyte enzyme which protects hemoglobin from oxidative breakdown. J. Biol. Chem. 1957;229:189-197.
- [7] Flohe L, Gunzler WA, Schock HH. Glutathione peroxidase: a selenoenzyme. *FEBS Lett.* 1973;32:132-134.
- [8] Ursini F, Heim S, Kiess M, Maiorino M, Roveri A, Wissing J, et al. Dual function of the selenoprotein PHGPx during sperm maturation. *Science* 1999;285:1393-1396.
- [9] Imai H, Hakkaku N, Iwamoto R, Suzuki J, Suzuki T, Tajima Y, *et al.* Depletion of selenoprotein GPx4 in spermatocytes causes male infertility in mice. *J. Biol. Chem.* 2009;284:32522-32532.
- [10] Sies H, Sharov VS, Klotz LO, Briviba K. Glutathione peroxidase protects against peroxynitrite-mediated oxidations: a new function for selenoproteins as peroxynitrite reductase. J. Biol. Chem. 1997;272:27812-27817.
- [11] Stoytcheva ZR, Berry MJ. Transcriptional regulation of mammalian selenoprotein expression. *Biochim. Biophys. Acta* 2009;1790:1429-1440.
- [12] Cao C, Leng Y, Huang W, Liu X, Kufe D. Glutathione peroxidase 1 is regulated by the c-Abl and Arg tyrosine kinases. J. Biol. Chem. 2003;278:39609-39614.
- [13] Banning A, Deubel S, Kluth D, Zhou Z, Brigelius-Flohe R. The GI-GPx gene is a target for Nrf2. *Mol. Cell Biol.* 2005;25:4914-4923.
- [14] Bierl C, Voetsch B, Jin RC, Handy DE, Loscalzo J. Determinants of human plasma glutathione peroxidase (GPx-3) expression. *J. Biol. Chem.* 2004;279:26839-26845.
- [15] Savaskan NE, Ufer C, Kuhn H, Borchert A. Molecular biology of glutathione peroxidase 4: from genomic structure to developmental expression and neural function. *Biol. Chem.* 2007;388:1007-1017.
- [16] Yant LJ, Ran Q, Rao L, Van Remmen H, Shibatani T, Belter JG, *et al.* The selenoprotein GPX4 is essential for mouse development and protects from radiation and oxidative damage insults. *Free Radic. Biol. Med.* 2003;34:496-502.
- [17] Chrissobolis S, Didion SP, Kinzenbaw DA, Schrader LI, Dayal S, Lentz SR, et al. Glutathione peroxidase-1 plays a major role in protecting against angiotensin IIinduced vascular dysfunction. *Hypertension* 2008;51:872-877.
- [18] Torzewski M, Ochsenhirt V, Kleschyov AL, Oelze M, Daiber A, Li H, et al. Deficiency of glutathione peroxidase-1 accelerates the progression of atherosclerosis in apolipoprotein E-deficient mice. Arterioscler. Thromb. Vasc. Biol. 2007;27:850-857.
- [19] Shiomi T, Tsutsui H, Matsusaka H, Murakami K, Hayashidani S, Ikeuchi M, et al.

Overexpression of glutathione peroxidase prevents left ventricular remodeling and failure after myocardial infarction in mice. *Circulation* 2004;109:544-549.

- [20] Beck MA, Esworthy RS, Ho YS, Chu FF. Glutathione peroxidase protects mice from viral-induced myocarditis. *FASEB J.* 1998;12:1143-1149.
- [21] Dabkowski ER, Williamson CL, Hollander JM. Mitochondria-specific transgenic overexpression of phospholipid hydroperoxide glutathione peroxidase (GPx4) attenuates ischemia/reperfusion-associated cardiac dysfunction. *Free Radic. Biol. Med.* 2008;45:855-865.
- [22] Guo Z, Ran Q, Roberts LJ, 2nd, Zhou L, Richardson A, Sharan C, et al. Suppression of atherogenesis by overexpression of glutathione peroxidase-4 in apolipoprotein Edeficient mice. *Free Radic. Biol. Med.* 2008;44:343-352.
- [23] McClung JP, Roneker CA, Mu W, Lisk DJ, Langlais P, Liu F, et al. Development of insulin resistance and obesity in mice overexpressing cellular glutathione peroxidase. *Proc. Natl. Acad. Sci. USA* 2004;101:8852-8857.
- [24] Harmon JS, Bogdani M, Parazzoli SD, Mak SS, Oseid EA, Berghmans M, et al. beta-Cell-specific overexpression of glutathione peroxidase preserves intranuclear MafA and reverses diabetes in db/db mice. *Endocrinology* 2009;150:4855-4862.
- [25] Lee YS, Kim AY, Choi JW, Kim M, Yasue S, Son HJ, et al. Dysregulation of adipose glutathione peroxidase 3 in obesity contributes to local and systemic oxidative stress. *Mol. Endocrinol.* 2008;22:2176-2189.
- [26] Ridet JL, Bensadoun JC, Deglon N, Aebischer P, Zurn AD. Lentivirus-mediated expression of glutathione peroxidase: neuroprotection in murine models of Parkinson's disease. *Neurobiol. Dis.* 2006;21:29-34.
- [27] Weisbrot-Lefkowitz M, Reuhl K, Perry B, Chan PH, Inouye M, Mirochnitchenko O. Overexpression of human glutathione peroxidase protects transgenic mice against focal cerebral ischemia/reperfusion damage. *Brain Res. Mol. Brain Res.* 1998;53:333-338.
- [28] Crack PJ, Taylor JM, Flentjar NJ, de Haan J, Hertzog P, Iannello RC, *et al.* Increased infarct size and exacerbated apoptosis in the glutathione peroxidase-1 (Gpx-1) knockout mouse brain in response to ischemia/reperfusion injury. *J. Neurochem.* 2001; 78:1389-1399.
- [29] Chen L, Na R, Gu M, Richardson A, Ran Q. Lipid peroxidation up-regulates BACE1 expression in vivo: a possible early event of amyloidogenesis in Alzheimer's disease. J. *Neurochem.* 2008;107:197-207.
- [30] Seiler A, Schneider M, Forster H, Roth S, Wirth EK, Culmsee C, *et al.* Glutathione peroxidase 4 senses and translates oxidative stress into 12/15-lipoxygenase dependentand AIF-mediated cell death. *Cell Metab.* 2008;8:237-248.
- [31] Yoo MH, Gu X, Xu XM, Kim JY, Carlson BA, Patterson AD, *et al.* Delineating the role of glutathione peroxidase 4 in protecting cells against lipid hydroperoxide damage and in Alzheimer's disease. *Antioxid. Redox Signal* 2010;12:819-827.
- [32] Ho YS, Magnenat JL, Bronson RT, Cao J, Gargano M, Sugawara M, et al. Mice deficient in cellular glutathione peroxidase develop normally and show no increased sensitivity to hyperoxia. J. Biol. Chem. 1997;272:16644-16651.
- [33] Cheng WH, Ho YS, Valentine BA, Ross DA, Combs GF, Jr., Lei XG. Cellular glutathione peroxidase is the mediator of body selenium to protect against paraquat lethality in transgenic mice. *J. Nutr.* 1998;128:1070-1076.
- [34] Singh A, Rangasamy T, Thimmulappa RK, Lee H, Osburn WO, Brigelius-Flohe R, et

al. Glutathione peroxidase 2, the major cigarette smoke-inducible isoform of GPX in lungs, is regulated by Nrf2. *Am. J. Respir. Cell Mol. Biol.* 2006;35:639-650.

- [35] Dittrich AM, Meyer HA, Krokowski M, Quarcoo D, Ahrens B, Kube SM, et al. Glutathione peroxidase-2 protects from allergen-induced airway inflammation in mice. *Eur. Respir. J.* 2010;35:1148-1154.
- [36] Comhair SA, Bhathena PR, Farver C, Thunnissen FB, Erzurum SC. Extracellular glutathione peroxidase induction in asthmatic lungs: evidence for redox regulation of expression in human airway epithelial cells. *FASEB J.* 2001;15:70-78.
- [37] Fu Y, Cheng WH, Porres JM, Ross DA, Lei XG. Knockout of cellular glutathione peroxidase gene renders mice susceptible to diquat-induced oxidative stress. *Free Radic. Biol. Med.* 1999;27:605-611.
- [38] Kim SJ, Lee JW, Jung YS, Kwon do Y, Park HK, Ryu CS, *et al.* Ethanol-induced liver injury and changes in sulfur amino acid metabolomics in glutathione peroxidase and catalase double knockout mice. *J. Hepatol.* 2009;50:1184-1191.
- [39] Bajt ML, Ho YS, Vonderfecht SL, Jaeschke H. Reactive oxygen as modulator of TNF and fas receptor-mediated apoptosis in vivo: studies with glutathione peroxidasedeficient mice. *Antioxid. Redox Signal* 2002;4:733-740.
- [40] Zhu JH, McClung JP, Zhang X, Aregullin M, Chen C, Gonzalez FJ, et al. Comparative impacts of knockouts of two antioxidant enzymes on acetaminophen-induced hepatotoxicity in mice. *Exp. Biol. Med.* (Maywood) 2009;234:1477-1483.
- [41] Mirochnitchenko O, Weisbrot-Lefkowitz M, Reuhl K, Chen L, Yang C, Inouye M. Acetaminophen toxicity: opposite effects of two forms of glutathione peroxidase. J. Biol. Chem. 1999;274:10349-10355.
- [42] Ran Q, Liang H, Gu M, Qi W, Walter CA, Roberts LJ, 2nd, et al. Transgenic mice overexpressing glutathione peroxidase 4 are protected against oxidative stress-induced apoptosis. J. Biol. Chem. 2004;279:55137-55146.
- [43] Esworthy RS, Aranda R, Martin MG, Doroshow JH, Binder SW, Chu FF. Mice with combined disruption of Gpx1 and Gpx2 genes have colitis. Am. J. Physiol. Gastrointest. Liver Physiol. 2001;281:G848-855.
- [44] Scimeca MS, Lisk DJ, Prolla T, Lei XG. Effects of gpx4 haploid insufficiency on GPx4 activity, selenium concentration, and paraquat-induced protein oxidation in murine tissues. *Exp. Biol. Med.* (Maywood) 2005;230:709-714.
- [45] Ishibashi N, Weisbrot-Lefkowitz M, Reuhl K, Inouye M, Mirochnitchenko O. Modulation of chemokine expression during ischemia/reperfusion in transgenic mice overproducing human glutathione peroxidases. J. Immunol. 1999;163:5666-5677.
- [46] de Haan JB, Stefanovic N, Nikolic-Paterson D, Scurr LL, Croft KD, Mori TA, et al. Kidney expression of glutathione peroxidase-1 is not protective against streptozotocininduced diabetic nephropathy. Am. J. Physiol. Renal. Physiol. 2005;289:F544-551.
- [47] Wenk J, Schuller J, Hinrichs C, Syrovets T, Azoitei N, Podda M, et al. Overexpression of phospholipid-hydroperoxide glutathione peroxidase in human dermal fibroblasts abrogates UVA irradiation-induced expression of interstitial collagenase/matrix metalloproteinase-1 by suppression of phosphatidylcholine hydroperoxide-mediated NFkappaB activation and interleukin-6 release. J. Biol. Chem. 2004;279:45634-45642.
- [48] Walshe J, Serewko-Auret MM, Teakle N, Cameron S, Minto K, Smith L, et al. Inactivation of glutathione peroxidase activity contributes to UV-induced squamous cell carcinoma formation. *Cancer Res.* 2007;67:4751-4758.

- [49] Brigelius-Flohe R, Kipp A. Glutathione peroxidases in different stages of carcinogenesis. *Biochim. Biophys. Acta* 2009;1790:1555-1568.
- [50] Chu FF, Esworthy RS, Chu PG, Longmate JA, Huycke MM, Wilczynski S, *et al.* Bacteria-induced intestinal cancer in mice with disrupted Gpx1 and Gpx2 genes. *Cancer Res.* 2004;64:962-968.
- [51] Diwadkar-Navsariwala V, Prins GS, Swanson SM, Birch LA, Ray VH, Hedayat S, et al. Selenoprotein deficiency accelerates prostate carcinogenesis in a transgenic model. Proc. Natl. Acad. Sci. USA 2006;103:8179-8184.
- [52] Yu YP, Yu G, Tseng G, Cieply K, Nelson J, Defrances M, et al. Glutathione peroxidase 3, deleted or methylated in prostate cancer, suppresses prostate cancer growth and metastasis. *Cancer Res.* 2007;67:8043-8050.
- [53] Chabory E, Damon C, Lenoir A, Kauselmann G, Kern H, Zevnik B, et al. Epididymis seleno-independent glutathione peroxidase 5 maintains sperm DNA integrity in mice. J. Clin. Invest. 2009;119:2074-2085.
- [54] Reddy VN, Giblin FJ, Lin LR, Dang L, Unakar NJ, Musch DC, *et al.* Glutathione peroxidase-1 deficiency leads to increased nuclear light scattering, membrane damage, and cataract formation in gene-knockout mice. *Invest. Ophthalmol. Vis. Sci.* 2001;42: 3247-3255.
- [55] Lu L, Oveson BC, Jo YJ, Lauer TW, Usui S, Komeima K, et al. Increased expression of glutathione peroxidase 4 strongly protects retina from oxidative damage. Antioxid. Redox Signal 2009;11:715-724.
- [56] Mirochnitchenko O, Prokopenko O, Palnitkar U, Kister I, Powell WS, Inouye M. Endotoxemia in transgenic mice overexpressing human glutathione peroxidases. *Circ. Res.* 2000;87:289-295.
- [57] Blankenberg S, Rupprecht HJ, Bickel C, Torzewski M, Hafner G, Tiret L, *et al.* Glutathione peroxidase 1 activity and cardiovascular events in patients with coronary artery disease. *N. Engl. J. Med* .2003;349:1605-1613.
- [58] Winter JP, Gong Y, Grant PJ, Wild CP. Glutathione peroxidase 1 genotype is associated with an increased risk of coronary artery disease. *Coron. Artery Dis.* 2003; 14:149-153.
- [59] Nemoto M, Nishimura R, Sasaki T, Hiki Y, Miyashita Y, Nishioka M, *et al.* Genetic association of glutathione peroxidase-1 with coronary artery calcification in type 2 diabetes: a case control study with multi-slice computed tomography. *Cardiovasc. Diabetol.* 2007;6:23.
- [60] Voetsch B, Jin RC, Bierl C, Benke KS, Kenet G, Simioni P, *et al.* Promoter polymorphisms in the plasma glutathione peroxidase (GPx-3) gene: a novel risk factor for arterial ischemic stroke among young adults and children. *Stroke* 2007;38:41-49.
- [61] Voetsch B, Jin RC, Bierl C, Deus-Silva L, Camargo EC, Annichino-Bizacchi JM, *et al.* Role of promoter polymorphisms in the plasma glutathione peroxidase (GPx-3) gene as a risk factor for cerebral venous thrombosis. *Stroke* 2008;39:303-307.
- [62] Grond-Ginsbach C, Arnold ML, Lichy C, Grau A, Reuner K. No association of the plasma glutathione peroxidase (GPx-3) gene with cerebral venous thrombosis in the German population. *Stroke* 2009;40:e24; author reply e25.
- [63] Hansen RD, Krath BN, Frederiksen K, Tjonneland A, Overvad K, Roswall N, *et al.* GPX1 Pro(198)Leu polymorphism, erythrocyte GPX activity, interaction with alcohol consumption and smoking, and risk of colorectal cancer. *Mutat. Res.* 2009;664:13-19.

- [64] Hu YJ, Diamond AM. Role of glutathione peroxidase 1 in breast cancer: loss of heterozygosity and allelic differences in the response to selenium. *Cancer Res.* 2003; 63:3347-3351.
- [65] Arsova-Sarafinovska Z, Matevska N, Eken A, Petrovski D, Banev S, Dzikova S, et al. Glutathione peroxidase 1 (GPX1) genetic polymorphism, erythrocyte GPX activity, and prostate cancer risk. Int. Urol. Nephrol. 2009;41:63-70.
- [66] Bermano G, Pagmantidis V, Holloway N, Kadri S, Mowat NA, Shiel RS, *et al.* Evidence that a polymorphism within the 3'UTR of glutathione peroxidase 4 is functional and is associated with susceptibility to colorectal cancer. *Genes Nutr.* 2007;2:225-232.

Chapter 7

Glutathione Reductase

Abstract

Glutathione reductase (GR) catalyzes the formation of GSH from glutathione disulfide (GSSG) at the expense of NADPH. As part of the glutathione redox cycle, GR plays an important role in the detoxification of reactive oxygen and nitrogen species and other reactive intermediates. In addition, by maintaining the high GSH/GSSG ratios, GR is also involved in regulating intracellular redox homeostasis. Studies using GR-overexpressing or deficient animal models demonstrate a role for GR in protecting against certain pathophysiological conditions involving oxidative stress. In humans, hereditary GR deficiency has been reported, and the clinical manifestations include shortened lifespan of red blood cells, cataract, and favism. The exact role of GR in common disease conditions in both experimental animals and humans warrants further investigations.

1. Overview

As described in Chapter 6, upon reaction with reactive oxygen and nitrogen species (ROS/RNS), GSH is oxidized to GSSG (also known as glutathione disulfide). In mammalian cells, the ratios of intracellular GSH to GSSG are high usually in the range of 10:1 to 100:1. Maintenance of such high ratios of intracellular GSH to GSSG is critical for normal cellular activities, including redox signaling. Glutathione reductase (GR) reduces GSSG to GSH, and is crucial for maintaining the high ratios of intracellular GSH to GSSG [1].

Mammalian GR is a homodimeric flavoprotein with a molecular mass of 104 kDa. It is widely expressed in various tissues. GR activity is present in both cytosol and mitochondria. The cytosolic and mitochondrial isozymes of GR are biochemically indistinguishable and encoded by a single gene. In humans, GR gene is localized on chromosome 8p21.1.

2. Biochemistry and Molecular Regulation

GR catalyzes the reduction of one molecule of GSSG to form two molecules of GSH using NADPH as the reducing cofactor (Figure 7-1).



Figure 7-1. Glutathione reductase (GR)-catalyzed regeneration of GSH from glutathione disulfide (GSSG). As illustrated, GR catalyzes the reduction of GSSG to GSH using NADPH as the electron donor. NADPH is primarily provided by the pentose phosphate pathway with glucose-6-phosphate dehydrogenase as a key enzyme. Deficiency of glucose-6-phosphate dehydrogenase causes decreased formation of NADPH, compromising GR-catalyzed regeneration of GSH. This may eventually lead to decreased levels of GSH in cells and increased cellular susceptibility to oxidative stress. Deficiency of glucose-6-phosphate dehydrogenase has been identified in humans.

As stated above, mammalian GR is a homodimer. The binding sites for NADPH and GSSG are at the opposite sides of each subunit of the dimer, and FAD is bound to the center of each subunit. NADPH reduces the FAD, which then passes its electrons onto a disulfide bridge in the active site of the enzyme. The two sulfhydryl groups so formed then interact with GSSG and reduce it to two molecules of GSH, and the protein disulfide of the enzyme is reformed [1, 2].

The cloning and sequencing of mammalian GR were reported in 1990 [3]. However, the molecular regulation of mammalian GR expression is poorly understood. Studies suggest a possible role for Nrf2 signaling in the regulation of the constitutive expression of GR in mammalian cells, including cardiomyocytes, macrophages, and bone marrow stromal cells [4-6]. In yeast, GR is regulated by the transcription factor yAP-1 [7]. In both cultured cells and in vivo animal models, GR is induced by various chemical agents, including dithiolethiones. Induction of GR by dithiolethiones occurs via an Nrf2-dependent mechanism. Increased expression of GR is also reported following exposure of cells to oxidative stress conditions.

3. Role in Health and Disease

3.1. Animal Studies

3.1.1. Experimental Approaches

There are multiple studies on the biological activities of GR in animal models. A GR mutant mouse strain (Gr1^{a1Neu}) with decreased tissue GR activities has been reported and used to investigate the effects of GR deficiency on oxidative tissue injury [8, 9]. Transgenic overexpression of GR has been created in *Drosophila melanogaster* [10]. However, to date, transgenic overexpression and gene knockout mouse models of GR have not been reported in the literature. Inhibition of GR activity by pharmacological inhibitors, especially *N*,*N*-bis(2chloroethyl)-*N*-nitrosourea (BCNU) has been frequently employed to study the biological functions of this antioxidant enzyme in both cellular systems and experimental animals. BCNU is an alkylating anticancer agent that may exert many other cellular effects in addition to inhibition of GR. Recently, a selective irreversible inhibitor of GR, namely, 2-acetylamino-3-[4-(2-acetylamino-2 carboxyethylsulfanylthiocarbonylamino)phenylthiocarbamoylsulfanyl] propionic acid (2-AAPA) has been reported [11]. At concentrations that markedly inhibit GR activity, 2-AAPA has no effects on γ -glutamylcysteine ligase, glutathione synthetase, superoxide dismutase, and catalase, but shows minimal inhibition of glutathione S-transferase and glutathione peroxidase [11].

3.1.2. Role in Disease Conditions

By maintaining the high ratios of GSH to GSSG, GR plays an important part in the detoxification of ROS/RNS as well as the maintenance of cellular redox homeostasis. GRdeficient mice (Gr1^{a1Neu}) are shown to be more susceptible to diaquat-induced oxidative stress and renal proximal tubular injury [12]. The GR-deficient mice are also more sensitive to hyperoxia-induced lung injury after the thioredoxin reductase is inhibited pharmacologically [13]. Similarly, inhibition of GR activity in animals or cultured cells by chemical inhibitors or siRNA (small interfering RNA used for knockdown of gene expression) results in increased sensitivity to oxidative injury. On the other hand, transgenic overexpression of GR in Drosophila melanogaster extends survival under hyperoxic conditions [10]. Viral-vector mediated increased expression of GR in macrophages attenuates atherosclerotic lesion formation in low-density lipoprotein (LDL) receptor-deficient mice [14]. In vitro, overexpression of GR protects macrophages from oxidized LDL (oxLDL)-induced cytotoxicity and mitochondrial damage [15]. Although GR is found to protect against oxidative injury in various cellular and animal models, selective expression of GR in type II cells has been shown to provide no benefits on hyperoxic lung injury in mice [16]. It has also been reported that local expression of GR gene in mouse brain is causally associated with anxiety-like behavioral phenotypes [17]; however, the significance of this finding remains unclear.

3.2. Human Studies and Clinical Perspectives

Hereditary GR deficiency has been documented in humans though it is a rare disorder. The first familial deficiency of GR in human red blood cells was reported in 1976 [18]. The GR activity could not be restored by in vivo riboflavin administration or in vitro FAD addition, suggesting a mutation in the gene encoding GR. Recently, a nonsense and a missense mutations of the GR gene have been identified, which cause clinical GR deficiency in patients [19]. The clinical symptoms of GR deficiency include reduced lifespan of red blood cells, cataract, and favism. Favism refers to acute hemolytic anemia caused by ingestion of fava beans, which contain compounds that are capable of inducing oxidative stress in cells, including red blood cells. A recent observational epidemiological study suggested a correlation between GR activity and a series of cardiovascular risk factors, including systolic arterial pressure, total cholesterol levels, and LDL cholesterol levels [20]. GR may also play a role in the pathogenesis of malaria. In this regard, it has been recently reported that GR deficiency and drug-induced GR inhibition may protect against malaria by inducing enhanced ring stage phagocytosis [21]. Thus, in humans GR may play a protective or detrimental role depending on the disease conditions.

4. Conclusion and Future Directions

In mammalian cells, GR is a major enzyme for maintaining the high ratios of GSH to GSSG. In contrast to other GSH-related enzymes, such as glutathione peroxidase (Chapter 6) and glutathione S-transferase (Chapter 8), studies on the biological activities of GR in mammalian models using gene knockout or transgenic overexpression approaches have been scarce. Extensive cell culture studies and several animal experiments demonstrate a role for GR in protecting against oxidative injury. Hereditary GR deficiency due to gene mutations has been reported in humans. However, the role of GR in common human diseases, such as cardiovascular disorders and diabetes warrants further investigations. In this context, continued basic research on GR in animal models of human diseases will provide important insights into the biological significance of this antioxidant enzyme in health and disease, and lay a foundation for translational and clinical research.

5. References

- [1] Meister A. Glutathione metabolism and its selective modification. J. Biol. Chem. 1988;263:17205-17208.
- [2] Karplus PA, Schulz GE. Refined structure of glutathione reductase at 1.54 A resolution. J. Mol. Biol. 1987;195:701-729.
- [3] Tutic M, Lu XA, Schirmer RH, Werner D. Cloning and sequencing of mammalian glutathione reductase cDNA. *Eur. J. Biochem.* 1990;188:523-528.
- [4] Zhu H, Jia Z, Zhang L, Yamamoto M, Misra HP, Trush MA, *et al.* Antioxidants and phase 2 enzymes in macrophages: regulation by Nrf2 signaling and protection against oxidative and electrophilic stress. *Exp. Biol. Med.* (Maywood) 2008;233:463-474.
- [5] Zhu H, Zhang L, Itoh K, Yamamoto M, Ross D, Trush MA, et al. Nrf2 controls bone marrow stromal cell susceptibility to oxidative and electrophilic stress. *Free Radic. Bi*ol. Med. 2006;41:132-143.
- [6] Zhu H, Jia Z, Misra BR, Zhang L, Cao Z, Yamamoto M, et al. Nuclear factor E2related factor 2-dependent myocardiac cytoprotection against oxidative and electrophilic stress. *Cardiovasc. Toxicol.* 2008;8:71-85.
- [7] Grant CM, Collinson LP, Roe JH, Dawes IW. Yeast glutathione reductase is required for protection against oxidative stress and is a target gene for yAP-1 transcriptional regulation. *Mol. Microbiol.* 1996;21:171-179.
- [8] Pretsch W. Glutathione reductase activity deficiency in homozygous Gr1a1Neu mice does not cause haemolytic anaemia. *Genet. Res.* 1999;73:1-5.
- [9] Rogers LK, Tamura T, Rogers BJ, Welty SE, Hansen TN, Smith CV. Analyses of glutathione reductase hypomorphic mice indicate a genetic knockout. *Toxicol. Sci.* 2004;82:367-373.
- [10] Mockett RJ, Sohal RS, Orr WC. Overexpression of glutathione reductase extends survival in transgenic Drosophila melanogaster under hyperoxia but not normoxia. FASEB J. 1999;13:1733-1742.
- [11] Seefeldt T, Zhao Y, Chen W, Raza AS, Carlson L, Herman J, *et al.* Characterization of a novel dithiocarbamate glutathione reductase inhibitor and its use as a tool to mod-

ulate intracellular glutathione. J. Biol. Chem. 2009;284:2729-2737.

- [12] Rogers LK, Bates CM, Welty SE, Smith CV. Diquat induces renal proximal tubule injury in glutathione reductase-deficient mice. *Toxicol. Appl. Pharmacol.* 2006;217: 289-298.
- [13] Tipple TE, Welty SE, Rogers LK, Hansen TN, Choi YE, Kehrer JP, et al. Thioredoxinrelated mechanisms in hyperoxic lung injury in mice. Am. J. Respir. Cell Mol. Biol. 2007;37:405-413.
- [14] Qiao M, Kisgati M, Cholewa JM, Zhu W, Smart EJ, Sulistio MS, et al. Increased expression of glutathione reductase in macrophages decreases atherosclerotic lesion formation in low-density lipoprotein receptor-deficient mice. Arterioscler. *Thromb. Vasc. Biol.* 2007;27:1375-1382.
- [15] Wang Y, Qiao M, Mieyal JJ, Asmis LM, Asmis R. Molecular mechanism of glutathione-mediated protection from oxidized low-density lipoprotein-induced cell injury in human macrophages: role of glutathione reductase and glutaredoxin. *Free Radic. Bi*ol. Med. 2006;41:775-785.
- [16] Heyob KM, Rogers LK, Welty SE. Glutathione reductase targeted to type II cells does not protect mice from hyperoxic lung injury. Am. J. Respir. Cell Mol. Biol. 2008;39: 683-688.
- [17] Hovatta I, Tennant RS, Helton R, Marr RA, Singer O, Redwine JM, *et al.* Glyoxalase 1 and glutathione reductase 1 regulate anxiety in mice. *Nature* 2005;438:662-666.
- [18] Loos H, Roos D, Weening R, Houwerzijl J. Familial deficiency of glutathione reductase in human blood cells. *Blood* 1976;48:53-62.
- [19] Kamerbeek NM, van Zwieten R, de Boer M, Morren G, Vuil H, Bannink N, et al. Molecular basis of glutathione reductase deficiency in human blood cells. Blood 2007;109:3560-3566.
- [20] Apelt N, da Silva AP, Ferreira J, Alho I, Monteiro C, Marinho C, *et al.* ACP1 genotype, glutathione reductase activity, and riboflavin uptake affect cardiovascular risk in the obese. *Metabolism* 2009;58:1415-1423.
- [21] Gallo V, Schwarzer E, Rahlfs S, Schirmer RH, van Zwieten R, Roos D, et al. Inherited glutathione reductase deficiency and Plasmodium falciparum malaria: a case study. *PLoS One* 2009;4:e7303.

Chapter 8

Glutathione S-Transferase

Abstract

Glutathione S-transferase (GST) refers to a superfamily of enzymes that catalyze the conjugation of GSH with a wide variety of xenobiotics. GSTs also play an important role in cell signal transduction and redox regulation. GSTs are major cellular defenses against oxidative and electrophilic stress, and protect against many disease processes in experimental animals. Deficiency of GSTM1 and GSTT1 due to homozygous gene deletions has been identified in human populations. These two null genotypes as well as gene polymorphisms in other members of the GST superfamily are associated with increased risk of developing various diseases, including cardiovascular disorders, diabetes, neurodegeneration, and cancer.

1. Overview

1.1. Definition and History

Glutathione S-transferase (GST), which is now formally called glutathione transferase, is a general name for a superfamily of enzymes that catalyze the conjugation of GSH with a wide variety of xenobiotics. These enzymes play an important role in protecting mammalian cells and tissues from electrophilic and oxidative stress. Some GST isozymes also regulate a number of cellular processes via non-enzymatic reactions [1, 2]. As listed below, this superfamily of enzymes consists of three main families, which are widely distributed in mammalian tissues.

- Cytosolic GSTs
- Mitochondrial GSTs
- Microsomal GSTs, also known as membrane-associated proteins in eicosanoid and glutathione metabolism (MAPEG)

Among the three main families, cytosolic GSTs are the most extensively studied enzymes involved in the detoxification of electrophilic xenobiotics and reactive oxygen species (ROS) in mammalian systems [3].

As stated in Section 1.1 of Chapter 5, M.J. de Rey Paihade in 1888 discovered a substance, which was proved to be GSH by F.G. Hopkins in 1929. In 1959, the role of GSH in the formation of mercapturic acid from bromobenzene in experimental animals was suggested in studies by H.G. Bray and coworkers [4]. The enzymatic catalysis of GSH conjugation reactions with xenobiotics was subsequently reported by J. Booth and coworkers in 1961 [5]. More than a decade later in 1974, GSTs, the enzymes responsible for catalyzing the conjugation reactions were eventually purified and characterized by W.B. Habig and coworkers as the first enzymatic step leading to the formation of mercapturic acid [6]. The GSH-dependent conjugation reactions have been widely used to determine GST activity in biological systems, including cultured cells and animal tissues.

1.2. Basic Characteristics

 Cytosolic GSTs: Mammalian cytosolic GSTs are all dimeric with a molecular mass of ~50 kDa. They are classified into seven classes, namely, alpha (A), mu (M), pi (P), sigma (S), theta (T), omega (O), and zeta (Z). Table 8-1 lists the human and mouse cytosolic GST genes.

Class (human)	Gene (human)	Class (mouse)	Gene (mouse)
Alpha (A)	GSTA1 GSTA2 GSTA3 GSTA4 GSTA5	Alpha (A)	Gsta1 Gsta2 Gsta3 Gsta4 Gsta5
Mu (M)	GSTM1 GSTM2 GSTM3 GSTM4 GSTM5	Mu (M)	Gstm1 Gstm2 Gstm3 Gstm4 Gstm5 Gstm6 Gstm7
Pi (P)	GSTP1	Pi (P)	Gstp1 Gstp2
Sigma (S)	GSTS1	Sigma (S)	Gsts1
Theta (T)	GSTT1 GSTT2	Theta (T)	Gstt1 Gstt2 Gstt3
Omega (O)	GSTO1	Omega (O)	Gsto1 Gsto2
Zeta (Z)	GSTZ1	Zeta (Z)	Gstz1

Table 8-1. Human and mouse cytosolic glutathione S-transferase genes

- Mitochondrial GSTs: Mammalian mitochondrial class kappa GST isozymes are dimeric. Mouse, rat, and human possess only a single kappa GST (GSTK). This enzyme catalyzes GSH conjugation reactions with certain xenobiotics.
- 3) Microsomal GSTs: Most of the mammalian MAPEG enzymes are involved in the production of eicosanoids. Six human MAPEG isozymes have been identified. Some of the isozymes are involved in the detoxification of electrophilic xenobiotics via GSH conjugation.

2. Biochemistry and Molecular Regulation

2.1. Biochemistry

Cytosolic GSTs are among the most extensively studied enzymes involved in the detoxification of xenobiotics and ROS. As illustrated in Figure 8-1, these enzymes catalyze the conjugation reactions of GSH with electrophilic xenobiotics, including reactive aldehydes and quinone compounds to form less reactive conjugates (xenobiotic-GS). Some GST isozymes also exhibit glutathione peroxidase activity, catalyzing reduction of organic hydroperoxides (LOOH) to form alcohols (LOH) [1].



Figure 8-1. Glutathione S-transferase (GST)-catalyzed biochemical reactions. As illustrated, GST-catalyzed conjugation reactions between electrophilic xenobiotics and GSH form less reactive xenobiotic-GSH conjugates (Xenobiotic-GS). GST-catalyzed decomposition of organic hydroperoxides (LOOH) results in the formation of the corresponding alcohols (LOH). In this reaction, GSH is oxidized to glutathione disulfide (GSSG).

2.2. Molecular Regulation

The molecular regulation of GST expression has been a subject of extensive studies during the past decades. GSTs are subject to both transcriptional and post-transcriptional regulation. They are inducible by a wide variety of conditions/factors, including oxidative stress, electrophilic xenobiotics, and chemoprotective agents (Section 3 of Chapter 2). A range of transcription factors have been identified to participate in regulating GST gene expression. Among them are AP-1, NF- κ B, AhR, and Nrf2. The importance of each of these transcription factors in modulating GST gene expression varies with the GST isozymes and the types of cells or tissues examined. Among the various transcription factors, Nrf2 appears to play the most important role in regulating many members of GSTs via an antioxidant response element (ARE)-driven mechanism. Indeed, ARE is identified in the promoter region of the genes for many GST isozymes [1, 7].

3. Role in Health and Disease

3.1. General Biological Activities

The biological activities of mammalian GST isozymes can be summarized into the following three categories. These biological activities are the basis for GST's involvement in health and disease.

- Protection against cytotoxicity of electrophilic xenobiotics and ROS via enzymatic reactions
- Protection against carcinogenesis
- Regulation of cell signaling

3.1.1. Protection against Cytotoxicity of Electrophiles and ROS via Enzymatic Reactions

GSTs play a central role in phase 2 biotransformation, in which they catalyze the GSH conjugation reactions with various substrates containing electrophilic moieties. The resulting more water-soluble conjugates are excreted via the multidrug resistant protein efflux pumps or undergo further metabolism to mercapturic acids. Substrates of GST isozymes include a wide range of endogenous metabolites, environmental xenobiotics, as well as alkylating and free radical-generating anticancer drugs. Some GST isozymes also function as glutathione peroxidases, catalyzing the reduction of organic hydroperoxides, including those derived from lipid peroxidation. In this regard, certain GST isozymes are reported to protect against ROS-induced injury [8]. Thus, GSTs play an important role in the detoxification of both electrophilic species and ROS. It should be noted that although GST-catalyzed glutathione conjugation is generally regarded as a detoxification mechanism, GSH conjugation of certain xenobiotics, such as halogen-containing compounds may lead to their bioactivation, resulting in increased toxicity [9].

3.1.2. Protection against Carcinogenesis

Many carcinogens are electrophilic compounds and also substrates for GSTs. Thus, GSTcatalyzed GSH conjugation of carcinogenic chemicals or their reactive metabolites represents an important detoxification mechanism for chemical carcinogens. As discussed below, GSTs, especially GSTP may also regulate cell signaling, and as such play a protective role in the development of spontaneous tumors.

3.1.3. Regulation of Cell Signaling

In addition to the enzymatic activities, some GST isozymes, particularly GSTP also possess non-enzymatic functions, regulating a number of cellular processes (e.g., kinase cascades) that are involved in the intrinsic ability of mammalian cells to survive genotoxic, metabolic, and oxidative stress [10]. These non-enzymatic reactions may contribute to the protective activities of GST isozymes in disease conditions, especially cancer development. Moreover, recent studies suggest that GSTs may participate in S-glutathionylation [11]. As noted earlier (Section 3.1.4 of Chapter 5), S-glutathionylation is an important mechanism of post-translational regulation involved in cell redox signaling.

3.2. Animal Studies

3.2.1. Experimental Approaches

Animal studies provide much of our current knowledge on the role of GSTs in health and disease. The major experimental approaches in animal models include the following.

- Gene knockout of the individual GSTs
- Transgenic overexpression of the individual GSTs
- Pharmacological induction or inhibition of endogenous GST activity

Both transgenic and gene knockout animal models generate the most convincing evidence on the biological activities of the individual GSTs. On the other hand, although pharmacological approaches also provide important information, the specificity issues associated with the approaches deserve caution when drawing a conclusion on the role of a particular GST in biological systems. In this context, sulfasalazine has been used as a chemical inhibitor of general GST activity, whereas dichloroacetic acid has been reported to deplete GSTZ in experimental animals [12].

3.2.2. Role in Cardiovascular Diseases

Both oxidative and electrophilic stress contribute to the development of cardiovascular disorders, including atherosclerosis, hypertension, myocardial ischemia-reperfusion injury, and xenobiotic/drug-induced cardiotoxicity. By detoxifying oxidative and electrophilic species, GSTs may play a protective role in the above disease processes. In addition, the role of GSTs in redox signaling may also contribute to their beneficial effects in cardiovascular system. In cultured cells, overexpression of GSTs protects cardiovascular cells from oxidative and electrophilic injury. GSTs also regulate vascular smooth muscle cell proliferation and migration, both of which are critical events in atherogenesis [13]. Viral vector-mediated GST overexpression is shown to mitigate transplant atherosclerosis and inflammatory responses in rabbit carotid allografts [14]. Studies using spontaneous hypertension rats also suggest a possible protective role for GSTM1 in vascular oxidative stress [15]. Although the role of GSTs in myocardial ischemia-reperfusion injury has not been reported, GSTs are found to detoxify aldehydes generated during myocardial ischemia-reperfusion in animal models [16]. Aldehydes have been shown to participate in the pathogenesis of myocardial ischemia-reperfusion injury [17].

3.2.3. Role in Neurological Diseases

GSTs are detoxification enzymes that can inhibit oxidative and chemical stress underlying neurodegeneration. In this regard, dopamine-derived quinone metabolites as well as environmental chemicals are implicated in neurodegenerative disorders, especially Parkinson's disease. In vitro studies demonstrate a critical role for GSTs in protecting against dopaminequinone toxicity. In line with the in vitro studies, it has been reported that increased GST activity rescues dopaminergic neuron loss in a Drosophila model of Parkinson's disease [18]. In a mouse model, it has been demonstrated that GSTP is the only GST family member expressed in substantia nigra neurons, and depletion of this GST isozyme sensitizes the animals to experimental parkinsonism induced by 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine (MPTP) [19]. MPTP is known to cause parkinsonism in human subjects, and has been widely used as a model chemical for inducing parkinsonism in experimental animals.

3.2.4. Role in Pulmonary Diseases

Lung is a major target organ of environmental pollutants, including both electrophilic and oxidative species, and GSTs serve as a principal defense against these noxious species. In vitro, GSTs play a critical part in protecting lung cells from injury elicited by oxidants and other reactive species, including tobacco smoke. In experimental animals, inhalation of either GST or GSH ameliorates oxidative stress and airway inflammation induced by ovalbumin, and the combination of GST and GSH affords a synergistic protective effect [20]. Studies with GSTP-null mice suggest that GSTP1 is an endogenous inhibitor of allergic responses in an experimental model of asthma [21]. Together, these in vivo studies support an important role for GSTs in protecting against airway oxidative stress and inflammation.

3.2.5. Role in Hepatic Diseases

Various GST gene knockout models have been used to study the involvement of GST isozymes in liver disorders, including acetaminophen overdose-induced hepatotoxicity [22]. The hepatotoxicity associated with acetaminophen overdose results from liver GSH depletion and the subsequent accumulation of the toxic metabolite, N-acetyl-p-benzoquinoneimine. N-Acetyl-p-benzoquinoneimine is a potent electrophile that causes liver cell injury. It was initially thought that conjugation of N-acetyl-p-benzoquinoneimine with GSH was predominately catalyzed by GSTP in liver cells, and that deficiency of GSTP would aggravate acetaminophen-induced hepatotoxicity. However, GSTP knockout mice are found to be more resistant to acetaminophen-induced liver toxicity than wild-type mice [23]. While the mechanism involved in this unexpected resistance is unclear, GSTP-null mice are shown to be able to rapidly regenerate hepatic GSH, which may be associated with the tolerance to acetaminophen-induced liver toxicity. In contrast to the GSTP-null mice, GSTZ1-deficient mice are very sensitive to acetaminophen toxicity, and show significant hepatotoxicity when exposed to doses that have limited effects in normal mice [24]. Although there is a possibility that GSTZ1 contributes directly to the normal detoxification of acetaminophen, it seems more likely that the relative low level of hepatic GSH in GSTZ1-null mice is the major contributing factor. These observations suggest differential roles for various GST isozymes in a disease condition, and also point to the confounding factors involved in the gene knockout animal models (see Section 5.2 of Chapter 2 for more discussion of caveats on genetic models).

Studies using knockout animal models have shown beneficial effects of GSTA3 and GSTA4 in liver injury induced by other hepatotoxicants, including carbon tetrachloride, ethanol, and aflatoxin B1 [25-27]. Notably, mice deficient in GSTZ (also known as maleylacetoacetate isomerase) exhibit a range of pathological changes in the absence of insults, including hepatitis as well as kidney injury and splenic atrophy. This suggests a role for GSTZ in the detoxification of endogenous toxic metabolites [28].

3.2.6. Role in Cancer

GSTs play multiple roles in cancer, which can be summarized as follows.

- Protection against chemical carcinogenesis
- Protection against spontaneous cancer development
- Induction of GSTs as a mechanism of anticancer drug resistance

Many carcinogens are either direct DNA-reacting electrophiles or metabolized to form electrophilic metabolites. By catalyzing GSH conjugation with the DNA-reacting electrophilic species, GSTs protect cellular DNA as well as other critical targets from damage by chemical carcinogens. Targeted disruption of GST isozymes causes increased susceptibility to chemical carcinogenesis in various animal models. On the other hand, induction of GSTs by pharmacological inducers is an important mechanism for chemoprotection against chemical carcinogenesis [29].

In addition to protecting against chemical carcinogenesis, GSTs also play a beneficial role in spontaneous carcinogenesis. For example, a gene knockout study shows that GSTP deficiency leads to markedly enhanced colon tumorigenesis in APC^{Min} mice [30]. Knockout of GSTP in mice also results in inflammatory responses in colon. As compared with wild-type and heterozygous mice, GSTP homozygous knockout (GSTP^{-/-}) mice spontaneously develop various types of tumors with a significantly higher frequency and with a median tumor free lifespan 20 weeks shorter. In addition, one third of the GSTP^{-/-} mice develop lung adenomas, while less than 10% of GSTP^{+/-} and GSTP^{+/+} mice present such tumors [31]. These observations suggest that GSTP may suppress the development of spontaneous tumors. In addition to tumor development, GSTP deficiency in mice also causes myeloproliferation possibly via a deregulation of JNK and Janus kinase/STAT pathways [32]. As discussed above (Section 3.1.3), GSTP regulates cell signaling and controls cell proliferation via a process independent of the enzymatic activity.

Another scenario related to GSTs in cancer is the anticancer drug resistance due to upregulation of GSTs in cancer cells. Many anticancer drugs, especially the alkylating agents are good substrates for GSTs. Treatment with these anticancer agents can lead to upregulation of GSTs in tumor cells, which in turn renders the tumor cells more resistant to cytotoxicity of these drugs. This has been shown as a major mechanism of anticancer drug resistance [33]. In this context, inhibition of GSTs in tumor cells is desirable as this will enhance the tumorkilling activity of the anticancer drugs.

3.2.7. Role in Other Diseases and Conditions

GSTs may also play a role in several other disease conditions. For example, GSTP-null mice exhibit an increased sensitivity to cyclophosphamide-induced urinary bladder injury [34], a significant side effect associated with the use of cyclophosphamide in patients. As stated in Section 3.2.7 of Chapter 5, GSH may be involved in the bioactivation of the anticancer drug cisplatin, leading to nephrotoxicity. It is believed that GSTs catalyze the reaction between GSH and cisplatin, forming a GSH conjugate. This cisplatin-GSH conjugate is subsequently metabolized to a nephrotoxicant via γ -glutamyl transpeptidase and a cysteine S-conjugate β -lyase in the kidney [35].

GSTZ1 is identical to maleylacetoacetate isomerase and catalyzes a significant step in the

catabolism of phenylalanine and tyrosine. Exposure of GSTZ1-deficient mice to high dietary phenylalanine causes a rapid loss of white blood cells, likely due to the accumulation of maleylacetoacetate or maleylacetone in the circulation [12]. Recently, a protective role for GSTA4 in insulin resistance and diabetes has been suggested. Targeted disruption of adipose GSTA4 leads to increased protein carbonylation, oxidative stress, and mitochondrial dysfunction, which may contribute to insulin resistance and the development of diabetes. In addition, GSTA4 expression is found to be selectively downregulated in adipose tissue of obese insulin resistant mice and in human obesity-linked insulin resistance [36].

3.3. Human Studies and Clinical Perspectives

Gene polymorphisms exist for many GSTs, especially the cytosolic GST isozymes. These polymorphisms may have a significant influence on the structure and function of corresponding GSTs. Two GST gene (GSTM1, GSTT1) deletions (also known as null polymorphisms) have been identified in human populations. The association of these two gene deletions with disease susceptibility, drug effects, and chemical toxicity has been extensively investigated over the past decade. This section begins with a discussion of these two gene deletions, followed by a description of other GST gene polymorphisms and their potential involvement in human health and disease.

3.3.1. GST Deficiencies due to Gene Deletions

Deletion of either GSTM1 or GSTT1 gene results in a null genotype characterized by a general deficiency in the individual GST isozyme activity. The homozygosity for the deficiency occurs at a high frequency in many population groups. Individuals homozygous for either deletion are thought to be at increased risk for tumors as a consequence of decreased capacity to detoxify possible carcinogens. Indeed, extensive studies demonstrate that either null genotype is associated with increased risk of developing various types of cancer [37-39]. Since GSTM1 and GSTT1 are critical enzymes responsible for the detoxification of a range of xenobiotics, including certain anticancer drugs, these enzymes are also involved in anticancer drug resistance. In this context, studies have shown that individuals with the null genotype respond more favorably to certain cancer chemotherapy [40].

Either GSTM1 or GSTT1 deficiency is also associated with elevated risk of developing a number of other diseases. These include hyperlipidemia, atherosclerosis, coronary heart disease, diabetes, neurodegeneration, asthma, chronic obstructive pulmonary disease, and alcoholic liver injury [41-47].

3.3.2. GST Gene Polymorphisms

In addition to the above null genotype for GSTM1 and GSTT1, other gene polymorphisms have also been identified for these two isozymes as well as many other members of the GST superfamily. Numerous observational epidemiological studies in various population groups have suggested an association between GST gene polymorphisms and risk of developing many pathophysiological conditions, including cardiovascular disorders, neurodegenerative diseases, carcinogenesis, and cancer drug resistance [1, 40, 48]. These observations in human subjects are consistent with the established biological activities of GSTs in experimental animals.

4. Conclusion and Future Directions

GSTs play a critical role in the detoxification of oxidative and electrophilic species. GSTs also participate in cell signaling and growth regulation. Via these diverse activities, GSTs provide protection against various disease processes involving oxidative and electrophilic stress as well as growth dysregulation in animal models. Consistent with the findings in experimental animals, extensive population studies on gene polymorphisms also implicate a beneficial role for GSTs in human health and disease. Continued genetic variation studies in human subjects will further enhance our understanding of the association between GSTs and human disease conditions, and facilitate our ability to identify the susceptible subpopulations. Such studies will also contribute to the development of effective strategies for disease intervention in high risk human subpopulations.

5. References

- [1] Hayes JD, Flanagan JU, Jowsey IR. Glutathione transferases. *Annu. Rev. Pharmacol. Toxicol.* 2005;45:51-88.
- [2] Oakley AJ. Glutathione transferases: new functions. *Curr. Opin. Struct. Biol.* 2005;15: 716-723.
- [3] Dourado DF, Fernandes PA, Ramos MJ. Mammalian cytosolic glutathione transferases. *Curr. Protein. Pept. Sci.* 2008;9:325-337.
- [4] Bray HG, Franklin TJ, James SP. The formation of mercapturic acids. 2. The possible role of glutathionase. *Biochem. J.* 1959;71:690-696.
- [5] Booth J, Boyland E, Sims P. An enzyme from rat liver catalysing conjugations with glutathione. *Biochem. J.* 1961;79:516-524.
- [6] Habig WH, Pabst MJ, Jakoby WB. Glutathione S-transferases: the first enzymatic step in mercapturic acid formation. *J. Biol. Chem.* 1974;249:7130-7139.
- [7] Hayes JD, McMahon M. NRF2 and KEAP1 mutations: permanent activation of an adaptive response in cancer. *Trends Biochem. Sci.* 2009;34:176-188.
- [8] Yang Y, Cheng JZ, Singhal SS, Saini M, Pandya U, Awasthi S, *et al.* Role of glutathione S-transferases in protection against lipid peroxidation: overexpression of hGSTA2-2 in K562 cells protects against hydrogen peroxide-induced apoptosis and inhibits JNK and caspase 3 activation. *J. Biol. Chem.* 2001;276:19220-19230.
- [9] Anders MW. Chemical toxicology of reactive intermediates formed by the glutathionedependent bioactivation of halogen-containing compounds. *Chem. Res. Toxicol.* 2008; 21:145-159.
- [10] Saavedra JE, Srinivasan A, Buzard GS, Davies KM, Waterhouse DJ, Inami K, *et al.* PABA/NO as an anticancer lead: analogue synthesis, structure revision, solution chemistry, reactivity toward glutathione, and in vitro activity. *J. Med. Chem.* 2006;49: 1157-1164.
- [11] Townsend DM, Manevich Y, He L, Hutchens S, Pazoles CJ, Tew KD. Novel role for glutathione S-transferase pi: regulator of protein S-Glutathionylation following oxidative and nitrosative stress. J. Biol. Chem. 2009;284:436-445.
- [12] Theodoratos A, Tu WJ, Cappello J, Blackburn AC, Matthaei K, Board PG.
Phenylalanine-induced leucopenia in genetic and dichloroacetic acid generated deficiency of glutathione transferase Zeta. *Biochem. Pharmacol.* 2009;77:1358-1363.

- [13] Yang Y, Parsons KK, Chi L, Malakauskas SM, Le TH. Glutathione S-transferasemicro1 regulates vascular smooth muscle cell proliferation, migration, and oxidative stress. *Hypertension* 2009;54:1360-1368.
- [14] Xu Y, Gong B, Yang Y, Awasthi YC, Boor PJ. Adenovirus-mediated overexpression of glutathione-s-transferase mitigates transplant arteriosclerosis in rabbit carotid allografts. *Transplantation* 2010;89:409-416.
- [15] McBride MW, Brosnan MJ, Mathers J, McLellan LI, Miller WH, Graham D, et al. Reduction of Gstm1 expression in the stroke-prone spontaneously hypertension rat contributes to increased oxidative stress. *Hypertension* 2005;45:786-792.
- [16] Ishikawa T, Esterbauer H, Sies H. Role of cardiac glutathione transferase and of the glutathione S-conjugate export system in biotransformation of 4-hydroxynonenal in the heart. J. Biol. Chem. 1986;261:1576-1581.
- [17] Chen CH, Budas GR, Churchill EN, Disatnik MH, Hurley TD, Mochly-Rosen D. Activation of aldehyde dehydrogenase-2 reduces ischemic damage to the heart. *Science* 2008;321:1493-1495.
- [18] Whitworth AJ, Theodore DA, Greene JC, Benes H, Wes PD, Pallanck LJ. Increased glutathione S-transferase activity rescues dopaminergic neuron loss in a Drosophila model of Parkinson's disease. *Proc. Natl. Acad. Sci. USA* 2005;102:8024-8029.
- [19] Smeyne M, Boyd J, Raviie Shepherd K, Jiao Y, Pond BB, Hatler M, et al. GSTpi expression mediates dopaminergic neuron sensitivity in experimental parkinsonism. *Proc. Natl. Acad. Sci. USA* 2007;104:1977-1982.
- [20] Tripathi P, Nair S, Singh BP, Arora N. Mutated glutathione S-transferase in combination with reduced glutathione shows a synergistic effect in ameliorating oxidative stress and airway inflammation. *Free Radic. Biol. Med* 2010;48:839-844.
- [21] Zhou J, Wolf CR, Henderson CJ, Cai Y, Board PG, Foster PS, *et al.* Glutathione transferase P1: an endogenous inhibitor of allergic responses in a mouse model of asthma. *Am. J. Respir. Crit. Care Med.* 2008;178:1202-1210.
- [22] Board PG. The use of glutathione transferase-knockout mice as pharmacological and toxicological models. *Expert. Opin. Drug. Metab. Toxicol.* 2007;3:421-433.
- [23] Henderson CJ, Wolf CR, Kitteringham N, Powell H, Otto D, Park BK. Increased resistance to acetaminophen hepatotoxicity in mice lacking glutathione S-transferase Pi. Proc. Natl. Acad. Sci. USA 2000;97:12741-12745.
- [24] Blackburn AC, Matthaei KI, Lim C, Taylor MC, Cappello JY, Hayes JD, et al. Deficiency of glutathione transferase zeta causes oxidative stress and activation of antioxidant response pathways. *Mol. Pharmacol.* 2006;69:650-657.
- [25] Dwivedi S, Sharma R, Sharma A, Zimniak P, Ceci JD, Awasthi YC, et al. The course of CCl₄ induced hepatotoxicity is altered in mGSTA4-4 null (-/-) mice. *Toxicology* 2006;218:58-66.
- [26] Sultana R, Raju BS, Sharma V, Reddanna P, Babu PP. Formation of acetaldehyde adducts of glutathione S-transferase A3 in the liver of rats administered alcohol chronically. *Alcohol* 2005;35:57-66.
- [27] Ilic Z, Crawford D, Vakharia D, Egner PA, Sell S. Glutathione-S-transferase A3 knockout mice are sensitive to acute cytotoxic and genotoxic effects of aflatoxin B1. *Toxicol. Appl. Pharmacol.* 2010;242:241-246.

- [28] Lim CE, Matthaei KI, Blackburn AC, Davis RP, Dahlstrom JE, Koina ME, *et al.* Mice deficient in glutathione transferase zeta/maleylacetoacetate isomerase exhibit a range of pathological changes and elevated expression of alpha, mu, and pi class glutathione transferases. *Am. J. Pathol.* 2004;165:679-693.
- [29] Kwak MK, Egner PA, Dolan PM, Ramos-Gomez M, Groopman JD, Itoh K, et al. Role of phase 2 enzyme induction in chemoprotection by dithiolethiones. *Mutat. Res.* 2001; 480-481:305-315.
- [30] Ritchie KJ, Walsh S, Sansom OJ, Henderson CJ, Wolf CR. Markedly enhanced colon tumorigenesis in Apc^{Min} mice lacking glutathione S-transferase Pi. *Proc. Natl. Acad. Sci. USA* 2009;106:20859-20864.
- [31] Gate L, Majumdar RS, Lunk A, Tew KD. Influence of glutathione S-transferase pi and p53 expression on tumor frequency and spectrum in mice. *Int. J. Cancer* 2005;113:29-35.
- [32] Gate L, Majumdar RS, Lunk A, Tew KD. Increased myeloproliferation in glutathione S-transferase pi-deficient mice is associated with a deregulation of JNK and Janus kinase/STAT pathways. J. Biol. Chem. 2004;279:8608-8616.
- [33] McIlwain CC, Townsend DM, Tew KD. Glutathione S-transferase polymorphisms: cancer incidence and therapy. *Oncogene* 2006;25:1639-1648.
- [34] Conklin DJ, Haberzettl P, Lesgards JF, Prough RA, Srivastava S, Bhatnagar A. Increased sensitivity of glutathione S-transferase P-null mice to cyclophosphamideinduced urinary bladder toxicity. J. Pharmacol. Exp. Ther. 2009;331:456-469.
- [35] Townsend DM, Tew KD, He L, King JB, Hanigan MH. Role of glutathione Stransferase Pi in cisplatin-induced nephrotoxicity. *Biomed. Pharmacother.* 2009;63:79-85.
- [36] Curtis JM, Grimsrud PA, Wright WS, Xu X, Foncea RE, Graham DW, *et al.* Downregulation of adipose glutathione S-transferase A4 leads to increased protein carbonylation, oxidative stress, and mitochondrial dysfunction. *Diabetes* 2010;59: 1132-1142.
- [37] Simic T, Savic-Radojevic A, Pljesa-Ercegovac M, Matic M, Mimic-Oka J. Glutathione S-transferases in kidney and urinary bladder tumors. *Nat. Rev. Urol.* 2009;6:281-289.
- [38] Piao JM, Shin MH, Kweon SS, Kim HN, Choi JS, Bae WK, *et al.* Glutathione-Stransferase (GSTM1, GSTT1) and the risk of gastrointestinal cancer in a Korean population. *World J. Gastroenterol.* 2009;15:5716-5721.
- [39] Altinisik J, Balta ZB, Aydin G, Ulutin T, Buyru N. Investigation of glutathione Stransferase M1 and T1 deletions in lung cancer. *Mol. Biol. Rep.* 2010;37:263-267.
- [40] Lo HW, Ali-Osman F. Genetic polymorphism and function of glutathione Stransferases in tumor drug resistance. *Curr. Opin. Pharmacol.* 2007;7:367-374.
- [41] Maciel SS, Pereira Ada C, Silva GJ, Rodrigues MV, Mill JG, Krieger JE. Association between glutathione S-transferase polymorphisms and triglycerides and HDLcholesterol. *Atherosclerosis* 2009;206:204-208.
- [42] Li R, Boerwinkle E, Olshan AF, Chambless LE, Pankow JS, Tyroler HA, *et al.* Glutathione S-transferase genotype as a susceptibility factor in smoking-related coronary heart disease. *Atherosclerosis* 2000;149:451-462.
- [43] Hossaini AM, Zamrroni IM, Kashem RA, Khan ZF. Polymorphism of glutathione Stransferases as genetic risk factors for the development of complications in type 2 diabetes mellitus. J. Crit. Care 2008;23:444-448.

- [44] Bekris LM, Shephard C, Peterson M, Hoehna J, Van Yserloo B, Rutledge E, *et al.* Glutathione-s-transferase M1 and T1 polymorphisms and associations with type 1 diabetes age-at-onset. *Autoimmunity* 2005;38:567-575.
- [45] Perez-Pastene C, Graumann R, Diaz-Grez F, Miranda M, Venegas P, Godoy OT, et al. Association of GST M1 null polymorphism with Parkinson's disease in a Chilean population with a strong Amerindian genetic component. *Neurosci. Lett.* 2007;418: 181-185.
- [46] Kabesch M, Hoefler C, Carr D, Leupold W, Weiland SK, von Mutius E. Glutathione S transferase deficiency and passive smoking increase childhood asthma. *Thorax* 2004; 59:569-573.
- [47] Oniki K, Ueda K, Hori M, Mihara S, Marubayashi T, Nakagawa K. Glutathione-Stransferase (GST) M1 null genotype and combined GSTM1 and GSTT1 null genotypes as a risk factor for alcoholic mild liver dysfunction. *Clin. Pharmacol. Ther.* 2007;81: 634-635.
- [48] Li YJ, Scott WK, Zhang L, Lin PI, Oliveira SA, Skelly T, et al. Revealing the role of glutathione S-transferase omega in age-at-onset of Alzheimer and Parkinson diseases. *Neurobiol. Aging* 2006;27:1087-1093.

Chapter 9

Glutaredoxin

Abstract

Glutaredoxin (Grx) refers to a family of GSH-dependent enzymes that catalyze the reduction of glutathionylated proteins. Grxs also participate in many other cellular processes, including cell signaling, growth regulation, anti-inflammation, and iron homeostasis. Animal studies demonstrate a critical involvement of various Grx isozymes in protecting against cardiovascular disorders, such as myocardial ischemia-reperfusion injury, cardiac hypertrophy, heart failure, and drug-induced cardiotoxicity. Grxs also exert beneficial effects on other disease processes, including neurodegeneration, anemia and iron overload, diabetes, airway inflammation, and cancer in animal models. A mutation that results in Grx5 gene splicing defects has been identified as the cause of anemia and iron overload in a human patient, but the role of Grxs in other human disease conditions remains to be elucidated.

1. Overview

1.1. Definition and History

The term glutaredoxin (Grx) refers to a family of GSH-dependent oxidoreductases. The primary function of Grxs is to catalyze the reduction of glutathionylated proteins, and thereby maintain the cellular redox homeostasis [1]. This reduction reaction is known as deglutathionylation. Protein glutathionylation refers to covalent attachment of glutathione to a protein cysteine residue via a disulfide bridge. Protein glutathionylation increases during oxidative stress. It represents a post-translational modification of proteins. Analogous to protein phosphorylation and dephosphorylation, protein glutathionylation and deglutathionylation play an important role in cell signal transduction. Because of the reversible nature of protein glutathionylation, it is also considered a transient protective mechanism for critical cysteine residues of proteins. This notion means that without glutathionylation the critical cysteine residues of proteins may become irreversibly oxidized or modified during oxidative stress. However, once the cysteine residues first become glutathionylated, they will no longer be irreversibly oxidized or modified. Importantly, the original cysteine residues can be reformed

upon Grx-catalyzed deglutathionylation.

The first Grx was identified in 1970's as a glutathione-dependent reductase of the disulfide formed in ribonucleotide reductase during its catalytic cycle in *Escherichia coli* [2, 3]. It was shown that Grx was able to restore the growth of *E. coli* in a mutant lacking thioredoxin (Trx; Chapter 10). As discussed in Chapter 10, Trxs and Grxs share a number of similar activities. However, it soon became obvious that Grxs, compared with Trxs, are more versatile with respect to the choice of substrates and reaction mechanisms. Moreover, in addition to the early discovered dithiol Grxs that contain the characteristic Cys-Pro-Tyr-Cys active site motif, subsequent studies over the past years have revealed a second group of Grxs. This group, commonly known as monothiol Grxs, lacks the C-terminal active site thiol in its Cys-Gly-Phe-Ser active site motif but contains all structural and functional elements to bind and utilize GSH as the substrate. Monothiol Grxs are further categorized into (1) single domain monothiol Grxs that consist of only one Grx domain and (2) multidomain monothiol Grxs that contain an N-terminal thioredoxin (Trx)-like domain and one to three C-terminal monothiol Grx domains (also known as PICOT homologous domains) [1].

1.2. Basic Characteristics

In addition to the classification described above in Section 1.1, Grx isozymes are usually named by numbers in the order of their discovery in various species. In mammals, there are multiple Grx isozymes, namely, Grx1, 2, 3, and 5 with Grx1 and Grx2 being most extensively investigated. Grx4 is present in lower eukaryotes. Grxs have a small molecular mass of ~10-12 kDa. Both Grx1 and Grx2 are widely distributed in mammalian tissues. Grx1 is mainly located in cytosol. Grx2 is primarily localized in mitochondria, and it can also be present in nuclei. In humans, Grx1 and Grx2 are localized on chromosomes 5q14 and 1q31.2-q31.3, respectively. In mammals, Grx3 is also known as protein kinase C interacting cousin of thioredoxin (PICOT), and is a multidomain monothiol cytosolic Grx. Grx5 is also a monothiol Grx, which is mainly present in mitochondria. In humans, Grxs 3 and 5 are localized on chromosomes 10q26 and 14q32.13, respectively.

2. Biochemistry and Molecular Regulation

2.1. Biochemistry

Grxs contain active cysteine residues and catalyze the reduction (deglutathionylation) of glutathionylated proteins. During the reaction, Grxs are oxidized with the formation of a disulfide bridge, which is then reduced back to the active cysteine groups by utilizing GSH (Figure 9-1). Thus, Grxs are GSH-dependent oxidoreductases. Grxs also catalyze the reduction of disulfide formed in ribonucleotide reductase as well as the reduction of dehydroascorbate to ascorbate. In addition, Grxs are shown to catalyze glutathionylation of proteins [4]. In summary, the biochemical reactions catalyzed by mammalian Grxs include: (1) deglutathionylation, (2) reduction of ribonucleotide reductase, (3) reduction of dehydroascorbate, and (4) glutathionylation.



Figure 9-1. Glutaredoxin (Grx)-catalyzed protein deglutathionylation. As illustrated, protein glutathionylation occurs under oxidative stress and other redox conditions. Grxs catalyze reduction of protein-SSG to protein-SH, and during the reaction Grxs become glutathionylated. The glutathionylated Grxs are reduced back to Grxs by GSH with the concomitant formation of glutathione disulfide (GSSG). GSSG can be reduced back to GSH by glutathione reductase (Chapter 7) utilizing NADPH as the electron donor (not shown).

2.2. Molecular Regulation

Our understanding of the transcriptional regulation of Grx expression in mammalian cells is very limited. The AP-1 binding site has been found in the promoter region of mammalian Grx gene. Mammalian Grxs are inducible by oxidative stress as well as certain chemicals, including 17- β -estradiol. Induction of Grxs by oxidative stress stimuli may occur via an AP-1-dependent mechanism. There is also evidence showing that under certain experimental conditions Grx activity may be enhanced in response to an oxidative stimulus without a concomitant increase in Grx protein expression; however, the exact mechanism of such Grx activation remains to be elucidated [4].

3. Role in Health and Disease

3.1. General Biological Activities

The general biological activities of Grxs can be summarized into the following four categories [1, 5].

- Protection against oxidative injury via mechanisms including maintenance of cellular redox homeostasis
- Regulation of cell growth and apoptosis via participating in cell signaling
- Anti-inflammatory effects via downregulating the expression of proinflammatory cytokines and adhesion molecules
- Regulation of iron homeostasis by acting as iron-sulfur proteins

These biological activities are the molecular basis for the involvement of Grxs in health and disease. Figure 9-2 illustrates the major biological activities of Grx isozymes in mammalian systems.



Figure 9-2. Glutaredoxin (Grx)-mediated biological activities. As illustrated, Grxs protect against oxidative stress and inflammation. In addition, these redox active proteins also participate in cell signaling and regulating iron homeostasis.

3.2. Animal Studies

3.2.1. Experimental Approaches

Both transgenic overexpression and gene knockout animal models have been created to determine the role of Grx isozymes in disease conditions. Homozygous knockout of Grx3 in mice is embryonically lethal, indicating an essential role for Grx3 in embryonic development [6]. In contrast to Grx3, homozygous knockout of other Grxs in mice does not result in embryonic lethality. In addition to the genetically manipulated models, the changes of Grxs in various disease processes have been observed in animal models. However, such observations do not establish any causal relationship between the changes in Grxs and the disease pathophysiology [1, 4].

3.2.2. Role in Cardiovascular Diseases

Cardiovascular disorders are the most extensively studied disease conditions with respect to the role of Grxs in disease pathophysiology. Recently, two studies have reported the effects of Grx1 knockout on myocardial ischemia-reperfusion injury in mice. The first study demonstrated that deficiency of Grx1 did not sensitize the adult mice to myocardial ischemiareperfusion injury, though the embryonic fibroblasts derived from the knockout mice were more susceptible to oxidative stress in vitro [7]. In contrast, the second study showed that Grx1 knockout exacerbated myocardial ischemia-reperfusion injury and enhanced the formation of reactive oxygen species (ROS), indicating a beneficial role for Grx1 in myocardial ischemia-reperfusion injury [8]. The protective role for Grx1 in myocardial ischemiareperfusion injury is further supported by subsequent studies with Grx1-overexpressing mice. Mouse hearts overexpressing Grx1 exhibit significantly improved post-ischemic ventricular recovery and reduced myocardial infarct size compared with wild-type hearts. Grx1 overexpression also attenuates the ROS formation due to myocardial ischemia-reperfusion [8]. More recently, an adenoviral vector-mediated Grx1 gene therapy is shown to increase Grx1 expression in myocardium and protect against myocardial ischemia-reperfusion injury in diabetic mice [9].

Transgenic overexpression of Grx2 is demonstrated to reduce myocardial ischemiareperfusion injury by preventing both apoptosis and necrosis of cardiomyocytes. Isolated hearts from these transgenic mice display significantly improved contractile performance and reduced infarct size [10]. Transgenic overexpression of Grx2 also ameliorates doxorubicininduced cardiotoxicity [11]. Doxorubicin-induced cardiotoxicity occurs via an oxidative stress mechanism.

The role of Grx3 in cardiovascular diseases has recently been extensively investigated using Grx3 transgenic overexpression mice. Grx3 in cardiac cells is upregulated in response to hypertrophic stimuli both in vitro and in vivo. Studies using transgenic mice with cardiacspecific overexpression of Grx3 demonstrate that Grx3 is a potent inhibitor of cardiac hypertrophy induced by pressure overload [12]. Cardiac-specific overexpression of Grx3 also results in increased ventricular function and cardiomyocyte contractility [12]. Further studies show that Grx3 attenuates cardiac hypertrophy by disrupting calcineurin-NFAT signaling, a pathway critically involved in cardiac hypertrophy [13]. Based on these findings, Grx3 has been suggested as an efficient modality for the treatment of cardiac hypertrophy and heart failure [14].

3.2.3. Role in Other Diseases and Conditions

In addition to a critical protective role in cardiovascular disorders, Grxs are also implicated in many other disease processes. Grx1 downregulation is suggested to result in mitochondrial dysfunction in cultured neuronal cells and in animal models of neurodegenerative diseases, such as Parkinson's and motor neuron diseases [15]. Studies using Grx1 knockout mice show that absence of Grx1 increases lens susceptibility to oxidative stress and cataract induced by ultraviolet light [16]. Consistently, the lens epithelial cells derived from Grx1deficient mice exhibit increased sensitivity to oxidative stress in vitro [17].

Both Grxs 3 and 5 are required for iron-sulfur cluster assembly and heme biosynthesis [18, 19]. As such, inactivation or dysfunction of Grx3 or 5 has been implicated in disorders involving iron homeostasis, such as anemia. Other disease conditions, including diabetes, airway inflammation, and cancer may also involve the alterations of Grxs; however, the exact role of Grxs in these disease processes remains to be further elucidated [1, 4].

3.3. Human Studies and Clinical Perspectives

The involvement of Grxs in human health and disease is largely unknown. Although there are reports on the changes of Grx levels in tissues or fluids of human subjects with such disease conditions as lung disorders and cancer, the causal role for Grxs in the disease pathophysiology remains unclear. As stated above, Grx5 is a recently identified Grx isozyme in mammals, and is primarily present in mitochondria. Gene knockout animal models demonstrate an important role for Grx5 in iron homeostasis [20]. A mutation that results in Grx5 gene splicing defects has also been identified as the cause of a sideroblastic-like microcytic anemia in a human patient [21]. The patient had moderate anemia, hepatosplenomegaly, low numbers of ringed sideroblasts, and iron overload [21]. These observations suggest that Grx5 function is highly conserved, and its defect in human subjects may lead to anemia and iron overload. Currently, there have been no reports on gene polymorphisms of Grxs in human populations.

4. Conclusion and Future Directions

Grxs are multifunctional proteins that participate in various cellular processes, including antioxidant defense, cell signaling, cell growth and apoptosis, anti-inflammation, and iron homeostasis. Grxs function to protect against disease processes, especially cardiovascular disorders in experimental animals. Currently, the involvement of Grxs in human health and disease is largely unknown though there has been a single case study demonstrating a role for Grx5 defect in anemia and iron overload in a human subject. Apparently, more basic and clinical research are warranted to further elucidate the role of Grxs in both physiology and disease processes.

5. References

- [1] Lillig CH, Berndt C, Holmgren A. Glutaredoxin systems. *Biochim. Biophys. Acta* 2008; 1780:1304-1317.
- [2] Holmgren A. Hydrogen donor system for Escherichia coli ribonucleoside-diphosphate reductase dependent upon glutathione. *Proc. Natl. Acad. Sci. USA* 1976;73:2275-2279.
- [3] Holmgren A. Glutathione-dependent synthesis of deoxyribonucleotides: purification and characterization of glutaredoxin from Escherichia coli. *J. Biol. Chem.* 1979;254: 3664-3671.
- [4] Mieyal JJ, Gallogly MM, Qanungo S, Sabens EA, Shelton MD. Molecular mechanisms and clinical implications of reversible protein S-glutathionylation. *Antioxid. Redox Signal* 2008;10:1941-1988.
- [5] Rouhier N, Couturier J, Johnson MK, Jacquot JP. Glutaredoxins: roles in iron homeostasis. *Trends Biochem. Sci.* 2010;35:43-52.
- [6] Cha H, Kim JM, Oh JG, Jeong MH, Park CS, Park J, et al. PICOT is a critical regulator of cardiac hypertrophy and cardiomyocyte contractility. J. Mol. Cell Cardiol. 2008;45: 796-803.
- [7] Ho YS, Xiong Y, Ho DS, Gao J, Chua BH, Pai H, *et al.* Targeted disruption of the glutaredoxin 1 gene does not sensitize adult mice to tissue injury induced by ischemia/reperfusion and hyperoxia. *Free Radic. Biol. Med.* 2007;43:1299-1312.
- [8] Malik G, Nagy N, Ho YS, Maulik N, Das DK. Role of glutaredoxin-1 in cardioprotection: an insight with Glrx1 transgenic and knockout animals. J. Mol. Cell. Cardiol. 2008;44:261-269.
- [9] Lekli I, Mukherjee S, Ray D, Gurusamy N, Kim YH, Tosaki A, et al. Functional recovery of diabetic mouse hearts by glutaredoxin-1 gene therapy: role of Akt-FoxOsignaling network. *Gene Ther.* 2010;17:478-485.
- [10] Nagy N, Malik G, Tosaki A, Ho YS, Maulik N, Das DK. Overexpression of glutaredoxin-2 reduces myocardial cell death by preventing both apoptosis and necrosis. J. Mol. Cell Cardiol. 2008;44:252-260.
- [11] Diotte NM, Xiong Y, Gao J, Chua BH, Ho YS. Attenuation of doxorubicin-induced cardiac injury by mitochondrial glutaredoxin 2. *Biochim. Biophys. Acta* 2009;1793:427-438.
- [12] Jeong D, Cha H, Kim E, Kang M, Yang DK, Kim JM, et al. PICOT inhibits cardiac

hypertrophy and enhances ventricular function and cardiomyocyte contractility. *Circ. Res.* 2006;99:307-314.

- [13] Jeong D, Kim JM, Cha H, Oh JG, Park J, Yun SH, et al. PICOT attenuates cardiac hypertrophy by disrupting calcineurin-NFAT signaling. Circ. Res. 2008;102:711-719.
- [14] Samarel AM. PICOT: a multidomain scaffolding inhibitor of hypertrophic signal transduction. *Circ. Res.* 2008;102:625-627.
- [15] Saeed U, Durgadoss L, Valli RK, Joshi DC, Joshi PG, Ravindranath V. Knockdown of cytosolic glutaredoxin 1 leads to loss of mitochondrial membrane potential: implication in neurodegenerative diseases. *PLoS One* 2008;3:e2459.
- [16] Meyer LM, Lofgren S, Ho YS, Lou M, Wegener A, Holz F, et al. Absence of glutaredoxin1 increases lens susceptibility to oxidative stress induced by UVR-B. Exp. Eye Res. 2009;89:833-839.
- [17] Lofgren S, Fernando MR, Xing KY, Wang Y, Kuszynski CA, Ho YS, *et al.* Effect of thioltransferase (glutaredoxin) deletion on cellular sensitivity to oxidative stress and cell proliferation in lens epithelial cells of thioltransferase knockout mouse. *Invest. Ophthalmol. Vis. Sci.* 2008;49:4497-4505.
- [18] Lillig CH, Berndt C, Vergnolle O, Lonn ME, Hudemann C, Bill E, et al. Characterization of human glutaredoxin 2 as iron-sulfur protein: a possible role as redox sensor. *Proc. Natl. Acad. Sci. USA* 2005;102:8168-8173.
- [19] Haunhorst P, Berndt C, Eitner S, Godoy JR, Lillig CH. Characterization of the human monothiol glutaredoxin 3 (PICOT) as iron-sulfur protein. *Biochem. Biophys. Res. Commun.* 2010;394:372-376.
- [20] Wingert RA, Galloway JL, Barut B, Foott H, Fraenkel P, Axe JL, et al. Deficiency of glutaredoxin 5 reveals Fe-S clusters are required for vertebrate haem synthesis. *Nature* 2005;436:1035-1039.
- [21] Camaschella C, Campanella A, De Falco L, Boschetto L, Merlini R, Silvestri L, et al. The human counterpart of zebrafish shiraz shows sideroblastic-like microcytic anemia and iron overload. Blood 2007;110:1353-1358.

Chapter 10

Thioredoxin

Abstract

Thioredoxin (Trx) refers to a family of small redox proteins that undergo NADPHdependent reduction by thioredoxin reductase and in turn reduce the disulfide bridge in target proteins. Trxs participate in a variety of cellular processes, including scavenging reactive oxygen and nitrogen species, anti-inflammation, and regulation of cell signaling. Ample experimental evidence demonstrates an important function for Trxs in protecting against various pathophysiological conditions involving oxidative stress and dysregulated inflammation. These include cardiovascular diseases, diabetes, and neurodegeneration. In contrast, the role of Trxs in cancer development remains a subject of controversies in animal models. Studies in human subjects also suggest a possible involvement of Trxs in certain disease conditions, such as neural tube defects, diabetes, and cancer.

1. Overview

Thioredoxin (Trx or Trn) was first described in 1964 in *Escherichia coli* as an electron donor for ribonucleotide reductase, which is an essential enzyme for DNA synthesis [1]. Trx is conserved from *E. coli* to mammals. The mammalian Trxs are a family of small redox active proteins with a molecular mass of ~12 kDa. There are at least three Trxs in mammals: Trx1, Trx2, and sperm Trx (Sp-Trx, also designed as $p32^{TrxL}$). Trxs 1 and 2 are the most extensively studied Trx isozymes in biology and medicine. They are ubiquitously distributed in mammalian tissues. Trx1 is predominantly located in cytosol and can be excreted into extracellular space or translocate into nuclei, whereas Trx2 is present in mitochondria. In humans, Trx1 and Trx2 are localized on chromosomes 9q13 and 22q13.1, respectively.

2. Biochemistry and Molecular Regulation

2.1. Biochemistry

Trxs are redox proteins that undergo NADPH-dependent reduction by thioredoxin reductase (Chapter 12) and in turn reduce oxidized cysteine groups in proteins [2]. Trxs have a conserved Cys-Gly-Pro-Cys redox catalytic site that is responsible for reducing the protein disulfide bridge. As a result, the two cysteine residues of the Trxs are oxidized to form disulfide (i.e., the oxidized form of Trxs). The oxidized Trxs are then reduced back to Trxs via thioredoxin reductase using NADPH as the electron donor (Figure 10-1).



Figure 10-1. Thioredoxin (Trx)-mediated reduction of protein disulfide bridge. During the reduction reaction, the sulfhydryl groups on Trx are oxidized to form a disulfide bridge. The oxidized Trx is then reduced back to Trx by thioredoxin reductase (TrxR) using NADPH as the electron donor.

Trxs serve as a cofactor for many mammalian enzymes, including ribonucleotide reductase, peroxiredoxins, and glutathione peroxidase-3. In addition, Trx1 can reduce and activate a number of transcription factors (e.g., NF- κ B, AP-1, SP1) that regulate various cellular processes. Trxs also bind to and modify the activities of other cellular proteins, including those involved in apoptosis. Recently, both Trx1 and Trx2 have been shown to regulate protein denitrosylation [3]. Protein denitrosylation, the removal of nitroso (-NO) groups primarily from cysteine residues in proteins, is an important aspect of nitric oxide-based signaling that regulates protein activity and protein-protein interactions [4].

Trxs possess antioxidant activities via various mechanisms, as listed below.

- Trxs may directly scavenge reactive oxygen species (ROS).
- Trxs act as cofactors for peroxiredoxins (Section 2.1 of Chapter 11), which are important enzymes for the detoxification of hydrogen peroxide, organic hydroperoxides, and possibly peroxynitrite.
- Trxs act as cofactors for methionine sulfoxide reductases (Section 2.1 of Chapter 14), enzymes involved in repairing oxidative protein damage and protecting against oxidative injury.
- Trxs may also serve as electron donors for glutathione peroxidase-3 (plasma glutathione peroxidase), an enzyme involved in the detoxification of ROS in extracellular environment (Chapter 6). However, due to the low concentrations of Trxs in plasma, this reaction may not contribute much to the antioxidant activities of Trxs.

2.2. Molecular Regulation

Trx activities are regulated at multiple levels, including transcriptional regulation, subcellular localization (e.g., nuclei, cytosol, mitochondria, extracellular space), interaction with other molecules, and post-translational modifications [2, 5]. Trxs are responsive to a variety of stimuli, such as oxidative stress and inflammatory cytokines. At the transcriptional level, Trx expression is regulated by multiple transcription factors, including AP-1, AP-2, NF- κ B, SP1, and Nrf2. On the other hand, as stated above in Section 2.1, a growing number of transcription factors (e.g., NF- κ B, AP-1, SP1) also require Trx-mediated reduction for efficient DNA binding.

Trx activities are also modulated by interacting proteins. For example, the activities of Trx1 are negatively regulated by thioredoxin interacting protein (Txnip). Txnip is also known as thioredoxin binding protein-2 (TBP-2). In addition to negatively regulating Trx1, Txnip may possess other biological activities, such as acting as a metabolic control protein and a tumor suppressor [6]. Various biochemical and molecular pathways are implicated in post-translational modifications of Trxs, which include oxidation and nitrosylation of the cysteine groups, glutathionylation, and tyrosine nitration. Notably, nitrosylation of certain cysteine residues of Trx1 is found to enhance Trx1 activities, whereas tyrosine nitration leads to Trx1 inhibition [5].

3. Role in Health and Disease

3.1. Animal Studies

3.1.1. Experimental Approaches

The biological activities of Trxs have been a subject of extensive studies over the past few years. Both transgenic overexpression and gene knockout animal models along with other genetic models have been created and used for understanding the functions of Trxs in health and disease. In contrast to Trx1 gene knockout, homozygous deletion of Trx2 gene has been shown to cause embryonic lethality, indicating an essential role for Trx2 in development [7]. In addition to the above genetic approaches, recombinant Trx protein has also been used as therapeutics for various pathophysiological conditions in experimental animals [8]. Primarily based on findings from animal studies, the biological activities of mammalian Trxs can be summarized into the following two categories.

- Protection against disease conditions involving oxidative and inflammatory stress
- Involvement in cancer development and anticancer drug resistance

3.1.2. Role in Cardiovascular Diseases

Trxs play an important protective role in various forms of cardiovascular disorders in animal models, including hypertension, atherosclerosis, myocardial ischemia-reperfusion injury, heart failure, autoimmune myocarditis, and drug-induced cardiotoxicity. It has been reported that transgenic overexpression of Trx2 attenuates angiotensin II-induced vascular dysfunction, oxidative stress, and hypertension in mice [9]. Trx2 overexpression also results in improvement of endothelial function and reduced atherosclerotic lesions in ApoE-deficient mice [10]. Studies with genetic animal models demonstrate a beneficial role for Trxs in myocardial ischemia-reperfusion injury [11]. In addition, endothelial-specific expression of Trx2 is found to promote ischemia-mediated arteriogenesis and angiogenesis, leading to enhanced functional recovery from ischemic injury [12]. In a hypertensive rat model, viral vector-mediated Trx1 gene delivery attenuates the heart failure after a chronic myocardial infarction, and the Trx1 gene delivery also results in increased expression of heme oxygenase-1 [13], a critical antioxidant protein in cardiovascular protection (Section 3.1.2 of Chapter 15). The role of Trx1 in protecting against heart failure is also evidenced by its attenuation of cardiac hyper-trophy and oxidative stress in myocarditis via inhibiting chemokine expression and leukocyte chemotaxis, suggesting a critical role for Trx1 in suppressing inflammation in myocardium [15]. Trx1 is found to be induced by doxorubicin, a cardiotoxic anticancer drug. Transgenic overexpression of Trx1 protects against doxorubicin-induced cardiotoxicity and oxidative stress in mice [16].

3.1.3. Role in Diabetes

Trxs have a protective role in both type 1 and type 2 diabetes in animal models. Transgenic overexpression of Trx1 suppresses streptozotocin-induced pancreatic inflammation, oxidative stress, and diabetes [17]. A lentiviral vector-mediated Trx1 overexpression in islets prolongs graft survival in autoimmune diabetic NOD mice, suggesting a beneficial function for Trx1 in islet transplantation for the treatment of type 1 diabetes in experimental animals [18]. These results also support the notion that oxidative stress and inflammation are key events in type 1 diabetes. Transgenic overexpression of Trxs not only protects against the development of type I diabetes, but also prevents progression of type 2 diabetes [19]. Trxs have been shown to also protect against diabetic complications in animal models, including diabetic cardiomyopathy, nephropathy, and osteopenia [20-22]. In addition, transgenic overexpression of Trx1 inhibits the development of experimental chronic pancreatitis induced by repeated administration of cerulean and lipopolysaccharide in mice [23]. In the same experimental model, Trx1 is observed to attenuate pancreatic fibrosis via suppressing oxidative stress and monocyte chemoattractant protein-1 (MCP)-1-mediated chronic inflammation [23]. These findings suggest that Trx1 not only protects against the pancreatic β -cell injury, but also serves as an important defense against pancreatitis and associated fibrosis.

3.1.4. Role in Neurological Diseases

Overexpression of Trxs in transgenic mice has been shown in multiple studies to protect against ischemic brain injury as well as excitotoxic brain damage [24-26]. Similarly, intravenous administration of recombinant Trx protein is also found to decrease brain damage following transient focal cerebral ischemia in mice [27]. In *Drosophila melanogaster*, overexpression of Trxs suppresses Parkin-associated endothelin receptor-like receptor-induced neurotoxicity and extends longevity, indicating a potential protective effect of Trxs in Parkinson's disease and aging [28]. Homozygous knockout of mitochondrial Trx2 has been shown to cause massive apoptosis, failure of neural tube closure, exencephaly, and early embryonic lethality in mice, suggesting that Trx2 possesses a particularly important function in brain development [7].

3.1.5. Role in Pulmonary Diseases

Numerous studies using transgenic mice as well as recombinant Trx protein have demonstrated a beneficial involvement of Trxs in various pulmonary disorders, including airway inflammation, asthma, emphysema, and drug/toxicant-induced pulmonary toxicity [29-35]. In addition, Trxs also play a role in protecting against pulmonary ischemia-reperfusion injury and influenza virus-induced pneumonia in animal models [36-38]. The above pulmonary protective activities of Trxs are believed to result from both antioxidant and anti-inflammatory functions associated with these redox-active proteins.

3.1.6. Role in Hepatic and Gastrointestinal Diseases

Either transgenic overexpression of Trxs or administration of recombinant Trx protein has been shown to ameliorate hepatic injury induced by ethanol, thioacetamide, and lipopoly-saccharide [39-41]. Trx2 haploinsufficiency $(Trx2^{+/-})$ in mice also results in mitochondrial injury and increased oxidative stress in liver. In addition, $Trx2^{+/-}$ mice are more susceptible to the ROS-generator diaquat-induced liver injury than wild-type littermates [42].

Multiple studies also demonstrate a protective role for Trxs in gastrointestinal disorders in animal models. For example, either transgenic overexpression of Trx1 or administration of recombinant Trx protein attenuates indomethacin-induced gastric mucosal injury in mice [43]. The *Helicobacter felis*-induced gastritis and oxidative stress are also suppressed in mice overexpressing Trx1 [44]. Either transgenic overexpression of Trx1 or administration of recombinant human Trx protein results in amelioration of experimental colitis in mice, and the colonic protective effects are associated with diminished formation of macrophage migration inhibitory factor [45].

3.1.7. Role in Renal Diseases

Trx1 has a beneficial role in certain renal disorders, including ischemia-reperfusion injury and diabetic nephropathy. The endogenous Trx1 protein is rapidly depleted from the cytosol in the cortical proximal tubular cells and detected in the tubular fluid following renal ischemia-reperfusion in mice, pointing to a possible involvement of Trx1 in renal ischemiareperfusion injury. Indeed, transgenic overexpression of human Trx1 in mice leads to increased expression of the Trx in kidney tissue and amelioration of ischemia-reperfusion injury, as assessed by blood creatinine and urea nitrogen levels [46]. Transgenic overexpression of Trx1 in mice also suppresses the development of diabetic nephropathy and oxidative renal tissue injury following induction of type 1 diabetes by streptozotocin [20].

3.1.8. Role in Skin Diseases

The anti-inflammatory and antioxidative activities of Trxs are implicated in dermal protection. A recent study using Trx transgenic mice shows that Trx1 overexpression suppresses contact hypersensitivity response by inhibiting leukocyte recruitment during the elicitation phase of the contact dermatitis [47]. The study also suggests that Trx1 may be a potential target for treating contact dermatitis [47]. In addition to exerting the anti-inflammatory and antioxidative effects, Trxs are also found to reduce the disulfide bonds of allergens and thereby mitigate the allergenicity of commercial wheat preparations in multiple animal models [48]. This finding opens the door to the testing of Trxs in the production of hypoallergenic, moredigestible foods. Trxs may play a role in protecting against skin aging. In cultured normal human fibroblasts, suppression of Trx1 results in premature senescence [49]. Trx1 is also shown to suppress ultraviolet irradiation-induced oxidative stress, proinflammatory responses, and activation of matrix metalloproteinase-1 in human skin cells [50]. Although Trxs may exert a beneficial effect in skin aging, transgenic overexpression of Trx1 in keratinocytes is reported to augment chemically-induced skin carcinogenesis in a two-stage mouse model using topical 7,12-dimethylbenz(a)antracene as an initiator and 12-*O*-tetracecanoylphorbol-13-acetate as a promoting agent [51]. The potential involvement of Trxs in cancer is further discussed in Section 3.1.9 below.

3.1.9. Role in Cancer

While there is ample evidence supporting the protective effects of Trxs in tissue injury induced by oxidative and inflammatory stress, the role of Trxs, especially Trx1 in cancer has been a subject of controversies, and the results are inconsistent among different animal models. Numerous studies have shown an elevated Trx expression in various types of cancer. Trx1 may contribute to many of the hallmarks of cancer, including increased cell proliferation, resistance to cell death, and increased angiogenesis [52]. Some data suggest that Trx expression may also contribute to anticancer drug resistance while others indicate the opposite. As described above in Section 3.1.8, transgenic overexpression of Trx1 in keratinocytes of mice augments chemically-induced skin carcinogenesis [51]. However, Trx1 overexpression in mice is demonstrated to attenuate oxidative stress and prevent benzene-induced hemato-lymphoid toxicity and thymic lymphoma [53]. Currently, there is no evidence showing that transgenic overexpression of Trxs in animals is associated with an increased incidence of cancer. Administration of recombinant Trx protein has not been shown to promote tumor growth or induce anticancer drug resistance in animal models [6]. Thus, the exact role of Trxs in cancer remains to be further elucidated.

3.1.10. Role in Other Diseases and Conditions

Trxs have been implicated as protectors in many other diseases and related conditions, including arthritis, sepsis, eye disorders, male infertility, and aging. Transgenic overexpression of Trx1 is shown to protect against joint destruction and oxidative stress in a murine arthritis model, in which arthritis is induced by administering a mixture of anti-type II collagen monoclonal antibody (mAb) and lipopolysaccharide (LPS) [54]. Administration of recombinant Trx protein also suppresses the mAb/LPS-induced joint swelling in wild-type mice [54]. Either transgenic overexpression of Trx1 or administration of recombinant Trx protein attenuates LPS-induced inflammatory cell activation and septic shock in animal models [55, 56]. These observations are consistent with the ability of Trxs to suppress inflammation and oxidative stress. Oxidative stress and inflammation are important mechanisms underlying eve disorders, such as retinal photic injury. A protective role for Trxs in retinal photooxidative damage has been demonstrated in animal models by using both Trx-transgenic overexpression and administration of exogenous recombinant Trx protein [57, 58]. Transgenic overexpression of Trx1 inhibits germ cell apoptosis induced by experimental cryptorchidism in mice, suggesting that Trx1 intensification may be a useful therapeutic strategy for male infertility associated with heat stress [59]. As discussed above in Section 3.1.8, Trxs may protect against cell senescence in cell cultures. The potential role for Trxs in aging has been investigated using transgenic overexpression models. Trx-overexpressing mice exhibit extended median and maximum lifespans compared with wild-type counterparts. Telomerase activity in spleen tissues of Trx-overexpressing mice is higher than that in wild-type littermates [60]. Tissue oxidative stress is also attenuated in Trx-overexpressing mice. As aforementioned, overexpression of Trxs also extends the longevity of *Drosophila melanogaster* [28] (Section 3.1.4). Taken together, these findings support an antiaging activity of Trxs in experimental animals.

3.2. Human Studies and Clinical Perspectives

The diverse biological activities of Trxs as demonstrated by extensive animal studies make these redox-active proteins also an attractive subject of investigations in the context of human health and disease. The involvement of Trxs in human diseases has been studied primarily via (1) examining the changes of these proteins in accessible tissue samples of patients with various disease conditions and (2) gene polymorphism studies. As discussed in prior chapters, neither of the above approaches establishes a causal relationship between the changes in antioxidant proteins or genes and the altered risk or incidence of disease conditions in the human subjects. But, these studies do provide important information regarding the significance of Trxs in human health and disease, and facilitate identification of risk factors and genetic biomarkers for human diseases.

Animal studies suggest that Trx2 is critical for neural tube closure during embryonic development [7]. This observation has prompted a recent investigation of the association between genetic polymorphisms in Trx2 gene and increased risk for neural tube defects. It has been observed that a novel Ins/Del polymorphism in the proximal promoter region of human Trx2 gene reduces the transcriptional activity of the gene, and is associated with an increased risk of spina bifida [61]. The gene polymorphisms of Trxs are also found to be associated with an increased risk of diabetes [62]. In addition, genetic variations in the thioredoxin interacting protein (Txnip) are demonstrated to be associated with hypertriglyceridemia and increased blood pressure in type 2 diabetes [63]. The involvement of Trxs in human cancer development has been reported in multiple studies. Increased expression of Trx1 is documented in several types of human cancer, including lung, pancreatic, colon, gastric, and breast cancer [52]. Recently, elevated expression of Trx1 has been identified as an independent poor prognosis factor in patients with liver metastasis from colorectal cancer [64]. Although the exact role for Trxs in human cancer remains to be determined, the association of Trx1 expression with aggressive tumor growth, resistance to standard therapy, and decreased patient survival makes it a potential target for cancer drug development [52].

4. Conclusion and Future Directions

Trxs are protective molecules in various disease processes, including cardiovascular injury, diabetes, and neurodegeneration in animal models. However, the exact role of Trxs in cancer remains inconclusive as both suppressive and promoting activities of Trxs with regard to cancer development have been reported in experimental animals. Data on the involvement of Trxs in human diseases are relatively scarce. Limited studies in human subjects suggest a possible protective role for Trxs in spina bifida and diabetes, and a potential detrimental effect on the development of certain types of cancer. Future efforts should be devoted to the identification of novel genetic variations of Trx genes and their influence on tissue Trx expression and susceptibility to disease pathophysiology. As the involvement of Trxs in cancer is controversial, more basic research is warranted to define the exact functions of these redoxactive proteins in multistage carcinogenesis as well as anticancer drug resistance.

5. References

- Laurent TC, Moore EC, Reichard P. Enzymatic synthesis of deoxyribonucleotides. Iv. isolation and characterization of thioredoxin, the hydrogen donor from Escherichia coli B. J. Biol. Chem. 1964;239:3436-3444.
- [2] Powis G, Montfort WR. Properties and biological activities of thioredoxins. *Annu. Rev. Pharmacol. Toxicol.* 2001;41:261-295.
- [3] Benhar M, Forrester MT, Hess DT, Stamler JS. Regulated protein denitrosylation by cytosolic and mitochondrial thioredoxins. *Science* 2008;320:1050-1054.
- [4] Benhar M, Forrester MT, Stamler JS. Protein denitrosylation: enzymatic mechanisms and cellular functions. *Nat. Rev. Mol. Cell Biol.* 2009;10:721-732.
- [5] Ago T, Sadoshima J. Thioredoxin and ventricular remodeling. *J. Mol. Cell Cardiol.* 2006;41:762-773.
- [6] Kaimul AM, Nakamura H, Masutani H, Yodoi J. Thioredoxin and thioredoxin-binding protein-2 in cancer and metabolic syndrome. *Free Radic. Biol. Med.* 2007;43:861-868.
- [7] Nonn L, Williams RR, Erickson RP, Powis G. The absence of mitochondrial thioredoxin 2 causes massive apoptosis, exencephaly, and early embryonic lethality in homozygous mice. *Mol. Cell Biol.* 2003;23:916-922.
- [8] Nakamura H, Hoshino Y, Okuyama H, Matsuo Y, Yodoi J. Thioredoxin 1 delivery as new therapeutics. *Adv. Drug Deliv. Rev.* 2009;61:303-309.
- [9] Widder JD, Fraccarollo D, Galuppo P, Hansen JM, Jones DP, Ertl G, *et al.* Attenuation of angiotensin II-induced vascular dysfunction and hypertension by overexpression of Thioredoxin 2. *Hypertension* 2009;54:338-344.
- [10] Zhang H, Luo Y, Zhang W, He Y, Dai S, Zhang R, *et al.* Endothelial-specific expression of mitochondrial thioredoxin improves endothelial cell function and reduces atherosclerotic lesions. *Am. J. Pathol.* 2007;170:1108-1120.
- [11] Turoczi T, Chang VW, Engelman RM, Maulik N, Ho YS, Das DK. Thioredoxin redox signaling in the ischemic heart: an insight with transgenic mice overexpressing Trx1. J. Mol. Cell Cardiol. 2003;35:695-704.
- [12] Dai S, He Y, Zhang H, Yu L, Wan T, Xu Z, *et al.* Endothelial-specific expression of mitochondrial thioredoxin promotes ischemia-mediated arteriogenesis and angiogenesis. *Arterioscler. Thromb. Vasc. Biol.* 2009;29:495-502.
- [13] Koneru S, Penumathsa SV, Thirunavukkarasu M, Zhan L, Maulik N. Thioredoxin-1 gene delivery induces heme oxygenase-1 mediated myocardial preservation after chronic infarction in hypertensive rats. Am. J. Hypertens. 2009;22:183-190.
- [14] Yamamoto M, Yang G, Hong C, Liu J, Holle E, Yu X, *et al.* Inhibition of endogenous thioredoxin in the heart increases oxidative stress and cardiac hypertrophy. *J. Clin.*

Invest. 2003;112:1395-1406.

- [15] Liu W, Nakamura H, Shioji K, Tanito M, Oka S, Ahsan MK, et al. Thioredoxin-1 ameliorates myosin-induced autoimmune myocarditis by suppressing chemokine expressions and leukocyte chemotaxis in mice. *Circulation* 2004;110:1276-1283.
- [16] Shioji K, Kishimoto C, Nakamura H, Masutani H, Yuan Z, Oka S, *et al.* Overexpression of thioredoxin-1 in transgenic mice attenuates adriamycin-induced cardiotoxicity. *Circulation* 2002;106:1403-1409.
- [17] Hotta M, Tashiro F, Ikegami H, Niwa H, Ogihara T, Yodoi J, et al. Pancreatic beta cell-specific expression of thioredoxin, an antioxidative and antiapoptotic protein, prevents autoimmune and streptozotocin-induced diabetes. J. Exp. Med. 1998;188: 1445-1451.
- [18] Chou FC, Sytwu HK. Overexpression of thioredoxin in islets transduced by a lentiviral vector prolongs graft survival in autoimmune diabetic NOD mice. J. Biomed. Sci. 2009;16:71.
- [19] Yamamoto M, Yamato E, Toyoda S, Tashiro F, Ikegami H, Yodoi J, *et al.* Transgenic expression of antioxidant protein thioredoxin in pancreatic beta cells prevents progression of type 2 diabetes mellitus. *Antioxid. Redox Signal* 2008;10:43-49.
- [20] Hamada Y, Miyata S, Nii-Kono T, Kitazawa R, Kitazawa S, Higo S, et al. Overexpression of thioredoxin1 in transgenic mice suppresses development of diabetic nephropathy. *Nephrol. Dial. Transplant.* 2007;22:1547-1557.
- [21] Hamada Y, Fujii H, Kitazawa R, Yodoi J, Kitazawa S, Fukagawa M. Thioredoxin-1 overexpression in transgenic mice attenuates streptozotocin-induced diabetic osteopenia: a novel role of oxidative stress and therapeutic implications. *Bone* 2009; 44:936-941.
- [22] Samuel SM, Thirunavukkarasu M, Penumathsa SV, Koneru S, Zhan L, Maulik G, et al. Thioredoxin-1 gene therapy enhances angiogenic signaling and reduces ventricular remodeling in infarcted myocardium of diabetic rats. *Circulation* 2010;121:1244-1255.
- [23] Ohashi S, Nishio A, Nakamura H, Asada M, Tamaki H, Kawasaki K, et al. Overexpression of redox-active protein thioredoxin-1 prevents development of chronic pancreatitis in mice. Antioxid. Redox Signal 2006;8:1835-1845.
- [24] Takagi Y, Mitsui A, Nishiyama A, Nozaki K, Sono H, Gon Y, et al. Overexpression of thioredoxin in transgenic mice attenuates focal ischemic brain damage. Proc. Natl. Acad. Sci. USA 1999;96:4131-4136.
- [25] Takagi Y, Hattori I, Nozaki K, Mitsui A, Ishikawa M, Hashimoto N, et al. Excitotoxic hippocampal injury is attenuated in thioredoxin transgenic mice. J. Cereb. Blood Flow Metab. 2000;20:829-833.
- [26] Zhou F, Gomi M, Fujimoto M, Hayase M, Marumo T, Masutani H, et al. Attenuation of neuronal degeneration in thioredoxin-1 overexpressing mice after mild focal ischemia. Brain Res. 2009;1272:62-70.
- [27] Hattori I, Takagi Y, Nakamura H, Nozaki K, Bai J, Kondo N, *et al.* Intravenous administration of thioredoxin decreases brain damage following transient focal cerebral ischemia in mice. *Antioxid. Redox Signal* 2004;6:81-87.
- [28] Umeda-Kameyama Y, Tsuda M, Ohkura C, Matsuo T, Namba Y, Ohuchi Y, et al. Thioredoxin suppresses Parkin-associated endothelin receptor-like receptor-induced neurotoxicity and extends longevity in Drosophila. J. Biol. Chem. 2007;282:11180-11187.

- [29] Imaoka H, Hoshino T, Takei S, Sakazaki Y, Kinoshita T, Okamoto M, et al. Effects of thioredoxin on established airway remodeling in a chronic antigen exposure asthma model. Biochem. Biophys. Res. Commun. 2007;360:525-530.
- [30] Imaoka H, Hoshino T, Okamoto M, Sakazaki Y, Sawada M, Takei S, et al. Endogenous and exogenous thioredoxin 1 prevents goblet cell hyperplasia in a chronic antigen exposure asthma model. *Allergol. Int.* 2009;58:403-410.
- [31] Kinoshita T, Hoshino T, Imaoka H, Ichiki H, Okamoto M, Kawayama T, et al. Thioredoxin prevents the development and progression of elastase-induced emphysema. Biochem. Biophys. Res. Commun. 2007;354:712-719.
- [32] Hoshino T, Nakamura H, Okamoto M, Kato S, Araya S, Nomiyama K, et al. Redoxactive protein thioredoxin prevents proinflammatory cytokine- or bleomycin-induced lung injury. Am. J. Respir. Crit. Care Med. 2003;168:1075-1083.
- [33] Kaimul Ahsan M, Nakamura H, Tanito M, Yamada K, Utsumi H, Yodoi J. Thioredoxin-1 suppresses lung injury and apoptosis induced by diesel exhaust particles (DEP) by scavenging reactive oxygen species and by inhibiting DEP-induced downregulation of Akt. *Free Radic. Biol. Med.* 2005;39:1549-1559.
- [34] Sato A, Hoshino Y, Hara T, Muro S, Nakamura H, Mishima M, *et al.* Thioredoxin-1 ameliorates cigarette smoke-induced lung inflammation and emphysema in mice. *J. Pharmacol. Exp. Ther.* 2008;325:380-388.
- [35] Torii M, Wang L, Ma N, Saito K, Hori T, Sato-Ueshima M, et al. Thioredoxin suppresses airway inflammation independently of systemic Th1/Th2 immune modulation. Eur. J. Immunol. 2010;40:787-796.
- [36] Yagi K, Liu C, Bando T, Yokomise H, Inui K, Hitomi S, *et al.* Inhibition of reperfusion injury by human thioredoxin (adult T-cell leukemia-derived factor) in canine lung transplantation. *J. Thorac. Cardiovasc. Surg.* 1994;108:913-921.
- [37] Okubo K, Kosaka S, Isowa N, Hirata T, Hitomi S, Yodoi J, et al. Amelioration of ischemia-reperfusion injury by human thioredoxin in rabbit lung. J. Thorac. Cardiovasc. Surg. 1997;113:1-9.
- [38] Nakamura H, Tamura S, Watanabe I, Iwasaki T, Yodoi J. Enhanced resistancy of thioredoxin-transgenic mice against influenza virus-induced pneumonia. *Immunol. Lett.* 2002;82:165-170.
- [39] Cohen JI, Roychowdhury S, DiBello PM, Jacobsen DW, Nagy LE. Exogenous thioredoxin prevents ethanol-induced oxidative damage and apoptosis in mouse liver. *Hepatology* 2009;49:1709-1717.
- [40] Okuyama H, Nakamura H, Shimahara Y, Araya S, Kawada N, Yamaoka Y, et al. Overexpression of thioredoxin prevents acute hepatitis caused by thioacetamide or lipopolysaccharide in mice. *Hepatology* 2003;37:1015-1025.
- [41] Okuyama H, Nakamura H, Shimahara Y, Uyama N, Kwon YW, Kawada N, et al. Overexpression of thioredoxin prevents thioacetamide-induced hepatic fibrosis in mice. J. Hepatol. 2005;42:117-123.
- [42] Perez VI, Lew CM, Cortez LA, Webb CR, Rodriguez M, Liu Y, et al. Thioredoxin 2 haploinsufficiency in mice results in impaired mitochondrial function and increased oxidative stress. *Free Radic. Biol. Med.* 2008;44:882-892.
- [43] Tan A, Nakamura H, Kondo N, Tanito M, Kwon YW, Ahsan MK, et al. Thioredoxin-1 attenuates indomethacin-induced gastric mucosal injury in mice. Free Radic. Res. 2007;41:861-869.

- [44] Kawasaki K, Nishio A, Nakamura H, Uchida K, Fukui T, Ohana M, et al. Helicobacter felis-induced gastritis was suppressed in mice overexpressing thioredoxin-1. Lab. Invest. 2005;85:1104-1117.
- [45] Tamaki H, Nakamura H, Nishio A, Nakase H, Ueno S, Uza N, et al. Human thioredoxin-1 ameliorates experimental murine colitis in association with suppressed macrophage inhibitory factor production. *Gastroenterology* 2006;131:1110-1121.
- [46] Kasuno K, Nakamura H, Ono T, Muso E, Yodoi J. Protective roles of thioredoxin, a redox-regulating protein, in renal ischemia/reperfusion injury. *Kidney Int.* 2003;64: 1273-1282.
- [47] Fukunaga A, Horikawa T, Ogura K, Taguchi K, Yu X, Funasaka Y, et al. Thioredoxin suppresses the contact hypersensitivity response by inhibiting leukocyte recruitment during the elicitation phase. Antioxid. Redox Signal 2009;11:1227-1235.
- [48] Buchanan BB, Adamidi C, Lozano RM, Yee BC, Momma M, Kobrehel K, et al. Thioredoxin-linked mitigation of allergic responses to wheat. Proc. Natl. Acad. Sci. USA 1997;94:5372-5377.
- [49] Young JJ, Patel A, Rai P. Suppression of thioredoxin-1 induces premature senescence in normal human fibroblasts. *Biochem. Biophys. Res. Commun.* 2010;392:363-368.
- [50] Buechner N, Schroeder P, Jakob S, Kunze K, Maresch T, Calles C, *et al.* Changes of MMP-1 and collagen type Ialpha1 by UVA, UVB and IRA are differentially regulated by Trx-1. *Exp. Gerontol.* 2008;43:633-637.
- [51] Mustacich D, Wagner A, Williams R, Bair W, Barbercheck L, Stratton SP, et al. Increased skin carcinogenesis in a keratinocyte directed thioredoxin-1 transgenic mouse. *Carcinogenesis* 2004;25:1983-1989.
- [52] Powis G, Kirkpatrick DL. Thioredoxin signaling as a target for cancer therapy. *Curr. Opin. Pharmacol.* 2007;7:392-397.
- [53] Li GX, Hirabayashi Y, Yoon BI, Kawasaki Y, Tsuboi I, Kodama Y, *et al.* Thioredoxin overexpression in mice, model of attenuation of oxidative stress, prevents benzeneinduced hemato-lymphoid toxicity and thymic lymphoma. *Exp. Hematol.* 2006;34: 1687-1697.
- [54] Tsuji G, Koshiba M, Nakamura H, Kosaka H, Hatachi S, Kurimoto C, *et al.* Thioredoxin protects against joint destruction in a murine arthritis model. *Free Radic. Biol. Med.* 2006;40:1721-1731.
- [55] Nakamura H, Herzenberg LA, Bai J, Araya S, Kondo N, Nishinaka Y, et al. Circulating thioredoxin suppresses lipopolysaccharide-induced neutrophil chemotaxis. Proc. Natl. Acad. Sci. USA 2001;98:15143-15148.
- [56] Sano H, Sata T, Nanri H, Ikeda M, Shigematsu A. Thioredoxin is associated with endotoxin tolerance in mice. *Crit. Care Med.* 2002;30:190-194.
- [57] Tanito M, Masutani H, Nakamura H, Ohira A, Yodoi J. Cytoprotective effect of thioredoxin against retinal photic injury in mice. *Invest. Ophthalmol. Vis. Sci.* 2002;43: 1162-1167.
- [58] Tanito M, Masutani H, Nakamura H, Oka S, Ohira A, Yodoi J. Attenuation of retinal photooxidative damage in thioredoxin transgenic mice. *Neurosci. Lett.* 2002;326:142-146.
- [59] Okubo K, Nakamura H, Watanabe J, Kamoto T, Yodoi J, Ogawa O, *et al.* Over expression of thioredoxin-1 in transgenic mice attenuates germ cell apoptosis induced by experimental cryptorchidism. *J. Urol.* 2004;172:2479-2482.

- [60] Mitsui A, Hamuro J, Nakamura H, Kondo N, Hirabayashi Y, Ishizaki-Koizumi S, et al. Overexpression of human thioredoxin in transgenic mice controls oxidative stress and life span. Antioxid. Redox Signal 2002;4:693-696.
- [61] Wen S, Lu W, Zhu H, Yang W, Shaw GM, Lammer EJ, et al. Genetic polymorphisms in the thioredoxin 2 (TXN2) gene and risk for spina bifida. Am. J. Med. Genet. A 2009; 149A:155-160.
- [62] Ikegami H, Ono M, Fujisawa T, Hiromine Y, Kawabata Y, Yamato E. Molecular scanning of the gene for thioredoxin, an antioxidative and antiapoptotic protein, and genetic susceptibility to type 1 diabetes. *Ann. NY Acad. Sci.* 2008;1150:103-105.
- [63] van Greevenbroek MM, Vermeulen VM, Feskens EJ, Evelo CT, Kruijshoop M, Hoebee B, *et al.* Genetic variation in thioredoxin interacting protein (TXNIP) is associated with hypertriglyceridaemia and blood pressure in diabetes mellitus. *Diabet. Med.* 2007;24:498-504.
- [64] Noike T, Miwa S, Soeda J, Kobayashi A, Miyagawa S. Increased expression of thioredoxin-1, vascular endothelial growth factor, and redox factor-1 is associated with poor prognosis in patients with liver metastasis from colorectal cancer. *Hum. Pathol.* 2008;39:201-208.

Chapter 11

Peroxiredoxin

Abstract

Peroxiredoxin (Prx) refers to a family of thiol-dependent peroxidases that reduce hydrogen peroxide and organic hydroperoxides to water and alcohols, respectively. There are six known mammalian isozymes (Prx1-6), classified as typical 2-Cys, atypical 2-Cys, or 1-Cys Prxs based on the mechanism and the number of cysteines involved during catalysis. In addition to their well-established peroxide-scavenging activity, Prxs also participate in the regulation of various cell signaling pathways. Extensive animal studies provide substantial evidence supporting a critical protective role for Prxs in various disease processes involving oxidative and inflammatory stress. In contrast, studies on the involvement of Prxs in human health and disease are much limited.

1. Overview

1.1. Definition and History

Peroxiredoxin (Prx) is a general term that refers to a family of small (22-27 kDa) nonseleno peroxidases currently known to possess six mammalian isozymes, namely, Prx1, 2, 3, 4, 5, and 6. These isozymes are able to reduce hydrogen peroxide, organic hydroperoxides, and possibly peroxynitrite, and thus represent a class of important antioxidants in mammals [1]. As discussed in Chapter 6, glutathione peroxidase represents another family of enzymes that detoxify peroxides and possibly peroxynitrite as well.

The first Prx was discovered by E.R. Stadtman and coworkers in 1988 in *Saccharomyces cerevisae*, where it protected the enzyme glutamine synthase from oxidation in a system that serendipitously included thiols to generate hydrogen peroxide. The enzyme was first named thiol-specific antioxidant (TSA) due to its dependence on thiol compounds [2]. The name was then changed to thioredoxin peroxidase as the protein was found to be oxidized by hydrogen peroxide and reduced back by thioredoxin (see Chapter 10 for discussion of thioredoxin). Subsequent studies showed homologous proteins in other species that shared the peroxidase activity, and the term peroxiredoxin was introduced, which has been widely adopted in the literature.

1.2. Basic Characteristics

The six Prxs expressed in mammalian cells are classified into three subgroups: (1) typical 2-Cys Prxs including Prx1-4, (2) atypical 2-Cys Prx (Prx5), and (3) 1-Cys Prx (Prx6). Prxs are widely distributed in mammalian tissues, and the subcellular localization varies with the isozymes. Prx 1, 2, and 6 are mainly located in cytosol. Prx3 is restricted to mitochondria. Prx4 is present in endoplasmic reticulum and also secreted into extracellular environment. Prx5 is localized intracellularly to cytosol, mitochondria, nuclei and peroxisomes. In humans, Prx1, 2, 3, 4, 5, and 6 are localized on chromosomes 1p34.1, 19p13.2, 10q25-q26, Xp22.11, 11q13, and 1q25.1, respectively. Table 11-1 summarizes the basic characteristics of the six Prx isozymes in mammals.

Isozyme	Subgroup	Cellular Location	Chromosomal Lo- calization
Prx1	Typical 2-Cys	Cytosol	1p34.1
Prx2	Typical 2-Cys	Cytosol	19p13.2
Prx3	Typical 2-Cys	Mitochondria	10q25-q26
Prx4	Typical 2-Cys	Endoplasmic reticulum, extra- cellular space	Xp22.11
Prx5	Atypical 2-Cys	Cytosol, mitochondria, nuclei, peroxisomes	11q13
Prx6	1-Cys	Cytosol	1q25.1

Table 11-1. Basic Characteristics of the six isozymes of mammalian peroxiredoxin

2. Biochemistry and Molecular Regulation

2.1. Biochemistry

Prxs catalyze the reduction of hydrogen peroxide and various organic hydroperoxides to form water and alcohols, respectively through the reactive cysteine (Cys) residues of the enzymes (Figure 11-1). They may also be able to reduce peroxynitrite under certain experimental conditions [3].

The typical 2-Cys Prxs (Prx1, 2, 3, and 4) are homodimers and contain both the N- and C-terminal conserved Cys residues and require both of them for catalytic function. Atypical 2-Cys Prx (Prx5) is a monomer and contains only the N-terminal Cys but requires one additional nonconserved Cys residue for catalytic activity. On the other hand, 1-Cys Prx (Prx6) is a homodimer and contains only the N-terminal Cys and requires no additional Cys for catalytic function.

During reaction with oxidant substrates, the Cys residues of Prxs are oxidized. Thioredoxins provide the electrons for reducing the oxidized Prx1-5, whereas GSH is likely to be employed to reduce the oxidized Prx6. The molecular and biochemical mechanisms underlying the differential use of thioredoxins and GSH by different Prx isozymes remain to be further elucidated.



Figure 11-1. Peroxiredoxin (Prx)-mediated reduction of hydrogen peroxide (H_2O_2) and organic hydroperoxides (LOOH). Shown in the scheme is a 2-Cys Prx-catalyzed reduction of H_2O_2 and LOOH to form water and alcohol (LOH), respectively. During the reaction, one sulfhydryl group of the 2-Cys Prx is oxidized to sulfenic acid intermediate, followed by the formation of a disulfide bridge (i.e., the oxidized form of Prx). The disulfide bridge is reduced by thioredoxin (Trx), and the original Prx is regenerated.

2.2. Molecular Regulation

Prxs are regulated at multiple levels, including both transcriptional regulation and posttranslational modifications. At the transcriptional level, NF- κ B has been shown to suppress the transcriptional expression of Prx6, whereas increased transcription of both Prx1 and Prx6 has been reported to be mediated by Nrf2 via an antioxidant response element-driven mechanism [4-6]. The B cell-specific activator protein, Pax5 is also shown to function as a transactivator of 1-Cys Prx (Prx6) gene expression [7]. In the promoter of human Prx5 gene, the binding sites for both nuclear respiratory factor 1 and nuclear respiratory factor 2 have been predicted. Both of these transcription factors are shown to largely control the basal expression of Prx5 gene [8]. As described in Section 2.2 of Chapter 4, the forkhead transcription factor Foxo3a plays an important role in the regulation of catalase gene expression in mammalian cells. A recent study also demonstrates a critical role for Foxo3a in regulating the gene expression of Prx3 in human cardiac fibroblasts [9].

The post-translational modifications of Prxs have been extensively studied over the past few years. Substantial evidence demonstrates a critical involvement of oxidative modifications of cysteine residues in regulating the activities of various Prx isozymes [10]. As described in Section 2.1 of Chapter 13, hyperoxidation of Prx cysteine residues to sulfinic acid results in irreversible enzyme inactivation. In addition to cysteine thiol modifications, protein phosphorylation and acetylation have also been implicated in the regulation of Prx enzymatic activities [10].

3. Role in Health and Disease

3.1. General Biological Activities

Prxs play an important part in the detoxification of hydrogen peroxide and various organic peroxides, and possibly peroxynitrite. This antioxidant function is the basis for the protective effects of Prxs in various types of diseases involving an oxidative stress mechanism. In addition to their antioxidant function, Prxs are reported to participate in other cellular processes, such as regulation of activities of transcription factors (e.g., NF- κ B, AP-1, p53), growth factor and ROS signaling, cell proliferation, differentiation, and apoptosis [1, 11]. ROS here stands for reactive oxygen species. Regulation of the above cellular processes or signaling pathways may also be an important mechanism underlying the biological activities of Prxs in health and disease.

3.2. Animal Studies

3.2.1. Experimental Approaches

Both transgenic overexpression and gene knockout models have been created to investigate the biological functions of Prxs in experimental animals. Viral vector-based gene delivery and recombinant Prx protein have also been employed to understand the biological activities of Prxs in experimental models. Largely based on studies using these models, Prxs have been demonstrated to be critical protectors in numerous disease conditions, among which are various forms of cardiovascular disorders.

3.2.2. Role in Cardiovascular Diseases

As stated in prior chapters, oxidative stress is a crucial mechanism underlying various types of cardiovascular disorders. Prxs have been found to undergo complex modifications during cardiovascular oxidative stress, suggesting that Prxs may be targets of oxidative stress in cardiovascular system [12, 13]. In this context, multiple transgenic overexpression and gene knockout animal studies demonstrate a critical protective role for different Prx isozymes in cardiovascular disorders, including atherosclerosis, myocardial ischemia-reperfusion injury, and heart failure.

Homozygous deletion of Prx1 (Prx1^{-/-}) in mice causes excessive endothelial activation and inflammation. Deletion of Prx1 also results in early atherosclerosis in ApoE-deficient mice fed normal chow, suggesting an important protective effect of Prx1 in vascular dysfunction and atherogenesis [14]. Platelet-derived growth factor (PDGF) plays a critical role in atherosclerosis and vascular remodeling. It is reported that Prx2 suppresses PDGF signaling and restenosis in a mouse model [15]. The involvement of Prx6 in atherosclerosis remains controversial. Some data suggest a beneficial role for Prx6 in vascular oxidative stress and atherogenesis [16], but others do not show a protective effect [17].

Prxs become oxidatively modified during myocardial ischemia-reperfusion injury. Targeted disruption of Prx6 in mice has been demonstrated to sensitize the hearts to global ischemia-reperfusion injury, as evidenced by reduced recovery of left ventricular function, increased myocardial infarction, and higher amounts of apoptotic cardiomyocytes, as well as increased oxidative stress compared with wild-type mouse hearts [18]. Similarly, overexpression of mitochondrial Prx3 has been found to prevent left ventricular remodeling and failure after myocardial infarction in mice [19]. Overexpression of Prx3 also protects against myocardial mitochondrial oxidative stress and mitochondrial DNA damage [20]. These results point to an important role for Prxs in protecting against myocardial remodeling and heart failure. They also suggest that Prxs may serve as potential therapeutic targets for the intervention of heart failure.

3.2.3. Role in Diabetes

Prxs are expressed in pancreatic tissue, including the β -cells [21]. Prx isozymes are induced in animal models of both type 1 and type 2 diabetes [22, 23]. In cultures, Prxs in the pancreatic β -cells are also inducible by oxidative stress and proinflammatory cytokines [24]. These observations suggest a possible protective role for Prxs in diabetes. Indeed, transgenic overexpression of Prx3 in mice results in decreased mitochondrial ROS production and increased resistance to stress-induced pancreatic β -cell injury [25]. These animals show improved glucose homeostasis, as evidenced by reduced levels of blood glucose and increased glucose clearance. In addition, the Prx3-overexpressing mice are protected from hyperglycemia and glucose intolerance induced by high-fat diet feeding [25].

3.2.4. Role in Neurological Diseases

Prxs are widely distributed in mammalian tissues, including the central nervous system. The basal expression of the six Prxs shows distinct profiles in different brain regions and different cell types. Prx1 and 6 are expressed in glial cells but not in neuron, whereas Prx2, 3, 4, and 5 are expressed in neurons [26]. Multiple studies suggest that Prxs may play an important part in protecting against neurological disorders in animal models, including parkinsonism, cerebral ischemic injury, age-associated hippocampal oxidative damage and cognitive dysfunction, and drug/toxicant-induced neurotoxicity [27-33]. In addition, neuronal overexpression of Prx2 in *Drosophila melanogaster* is shown to attenuate oxidative stress and prolong longevity, suggesting an overall protective function of Prx2 in neurodegeneration and aging [34]. In line with the animal studies, in vitro experiments also show beneficial effects of Prxs on oxidative and inflammatory injury in cultured neuronal cells.

3.2.5. Role in Pulmonary Diseases

A number of animal studies have documented a protective role for Prxs in several pulmonary disorders, including hyperoxia- and paraquat-induced lung injury, asthma, and endotoxin-elicited lung inflammatory injury. Prx6 is probably the most extensively studied Prx isozyme with respect to protection against lung injury [35]. Homozygous knockout of Prx6 sensitizes the mice to hyperoxia-induced lung oxidative stress and inflammation as well as mortality [36]. These knockout mice also show increased pulmonary injury following exposure to paraquat, a redox-cycling compound [37]. On the other hand, either transgenic overexpression or adenovirus-mediated gene transfer of Prx6 results in increased resistance to hyperoxia-induced lung injury and mortality in mice [38, 39]. Prx6 also possesses a phospholipase A2 activity and has an important role in lung surfactant turnover [40]. Recently, it is reported that Prx1 knockout sensitizes the animals to airway inflammation and hyperresponsiveness in a Th2-dominant allergic asthma model [41]. Targeted disruption of Prx3 in mice also leads to increased susceptibility of the animals to lipopolysaccharide-induced lung inflammation, airway wall thickening, and oxidative DNA and protein damage [42]. These animal studies strongly suggest a critical involvement of Prxs in protecting against lung inflammation and oxidative stress.

3.2.6. Role in Hepatic Diseases

The role of Prxs in liver ischemia-reperfusion injury and alcoholic liver injury has been recently studied. Hepatic ischemia-reperfusion is found to alter the cellular localization of Prx6 in mice, and targeted disruption of Prx6 sensitizes the animals to liver ischemia-reperfusion injury, which is associated with increased mitochondrial ROS formation and dys-function [43]. Chronic consumption of alcohol is reported to cause liver injury and oxidative stress as well as decreased expression of Prx6, suggesting a possible protective role for Prx 6 in alcoholic liver injury [44]. However, neither transgenic overexpression nor targeted disruption of Prx6 in mice results in attenuation or exacerbation of alcoholic liver injury [45]. It is further shown that ethanol administration causes significant induction of catalase, glutathione S-transferase, and glutathione peroxidase in both Prx6 transgenic overexpression and knock-out mice. The concomitant upregulation of the above antioxidant defenses by ethanol in both genetic models may explain the failure of genetic alterations of Prx6 to modulate alcoholic liver injury [45].

3.2.7. Role in Skin Diseases

Several recent studies have examined the possible involvement of Prx6 in skin disorders. Overexpression of Prx6 in keratinocytes is shown to protect the skin cells from injury elicited by oxidative stress and ultraviolet (UV) irradiation [46]. In experimental animals, targeted disruption of Prx6 gene leads to increased sensitivity to UV irradiation-induced skin injury and oxidative stress [47]. Knockout animal studies also suggest that Prx6 is required for blood vessel integrity in wounded skin [47].

3.2.8. Role in Cancer

It has been known for decades that ROS are involved in multistage carcinogenesis. Prxs via scavenging peroxides appear to have tumor preventive activities as loss of Prx1 in mice leads to premature death from multiple malignancies, including lymphomas, sarcomas, and carcinomas [48]. Prx1 ablation also increases the susceptibility of mice to Ras-induced breast cancer, suggesting a possible function for Prx1 in regulating oncogenic Ras effector pathways [49]. Indeed, Prx1 is shown to inhibit Ras-induced tumorigenesis via regulating PTEN/AKT

activity [49]. PTEN is a tumor suppressor, and AKT is a protein kinase. In vitro transformation assays demonstrate that the enhanced Ras-mediated transformation by loss of Prx1 can be inhibited by wild-type Prx1 but not by a catalytically inactive mutant of Prx1, in which peroxide catalyzing cysteines are replaced by serines. This suggests that the tumor suppressive properties of Prx1 depend on its antioxidant function [50].

There are studies suggesting that Prxs may also have tumor supportive activities. Increased expression of Prxs is documented in various types of cancer. In a cell culture model, overexpression of Prx6 is shown to promote the in vitro proliferation and invasion of breast cancer cells [51]. In an in vivo study, Prx6-overexpressing breast cancer cells are found to grow much faster and exhibit more pulmonary metastases than control cells [51]. Obviously, the exact role of Prxs in cancer warrants further investigations by using additional animal models of carcinogenesis.

3.2.9. Role in Other Diseases and Conditions

A protective role for Prxs has been suggested for various other pathophysiological conditions, including anemia, sepsis, spermatogenic cell injury, and aging. Prx1 knockout mice are shown to develop hemolytic anemia characterized by an increase in red blood cell ROS, which leads to protein oxidation, hemoglobin instability, Heinz body formation, and decreased red blood cell lifespan [48]. It has also been reported that Prx2 functions as an efficient scavenger of hydrogen peroxide in red blood cells, and may thus act as an important defense against oxidative stress in these cells [52].

Prx2 possesses an important function in regulating proinflammatory responses to lipopolysaccharide (LPS) and protecting against endotoxin-induced lethal shock in experimental animals. LPS is found to induce substantially enhanced inflammatory responses in Prx2deficient macrophages. Mice with targeted disruption of Prx2 show an increased sensitivity to LPS-induced septic shock and lethality, whereas intravenous injection of the adenovirusencoding Prx2 gene into Prx2-deficient mice significantly rescues the animals from LPSinduced lethal shock as compared with the injection of a control viral vector [53]. Recently, a beneficial role of Prx4 in spermatogenic cells has been suggested as Prx4 knockout mice show elevated spermatogenic cell death via oxidative stress [54]. ROS are believed as a major contributor to aging. By scavenging ROS, Prxs may play a protective role in aging process. In cell cultures, Prx2 is demonstrated to function as an enzymatic antioxidant to prevent cellular senescence [55]. Recently, it is reported that Prx5 confers protection against oxidative stress and promotes longevity in *Drosophila melanogaster* [56].

3.3. Human Studies and Clinical Perspectives

The potential involvement of Prxs in human disease conditions have been investigated, but the evidence is far less convincing than that obtained from animal studies. The levels of Prxs are shown to be altered in tissue samples of patients with various diseases, including cardiovascular disorders, neurodegeneration, autoimmunity, and cancer [57-63]. Notably, increased levels of Prxs are reported in most of the above disease conditions. Downregulation of Prx3-6 is observed in patients with dilated cardiomyopathy [58]. Although a causal role for Prxs in human diseases cannot be established solely based on the changes of Prxs in tissue samples, under certain conditions, levels of Prxs may serve as markers for disease prognosis.

For example, elevated Prx1 in lung tissue is shown to be an independent prognostic factor for disease recurrence and reduced survival in stage I non-small cell lung cancer [62]. In contrast, a study in patients with renal cell carcinoma demonstrates that Prx2 is associated with tumors of a lower grade and with a lower frequency of distant metastasis. Patients with tumors expressing Prx2 are further shown to have better prognosis [63]. In view of the controversial findings, the role of Prxs in human cancer development warrants further investigations.

4. Conclusion and Future Directions

Prxs are a family of important antioxidant enzymes that protect against pathophysiological conditions involving oxidative and inflammatory stress in various animal models. These antioxidant proteins also participate in other cellular processes, including regulation of cell signaling, cell proliferation, and apoptosis. Both direct antioxidant and non-antioxidant activities of Prxs form the basis of the beneficial functionality of Prxs in health and disease. In contrast to animal studies, studies in human subjects aiming to establish a causal involvement of Prxs in disease protection are scarce. To date, gene polymorphisms in Prxs and their association with disease risk have not been reported in humans. Future studies should focus on translational and clinical research to better define the role of Prxs in human health and disease. These studies will lay a foundation for developing therapeutic modalities for the intervention of human diseases in which Prxs play a role.

5. References

- [1] Rhee SG, Chae HZ, Kim K. Peroxiredoxins: a historical overview and speculative preview of novel mechanisms and emerging concepts in cell signaling. *Free Radic. Biol. Med.* 2005;38:1543-1552.
- [2] Kim K, Kim IH, Lee KY, Rhee SG, Stadtman ER. The isolation and purification of a specific "protector" protein which inhibits enzyme inactivation by a thiol/Fe(III)/O₂ mixed-function oxidation system. J. Biol. Chem. 1988;263:4704-4711.
- [3] Wood ZA, Schroder E, Robin Harris J, Poole LB. Structure, mechanism and regulation of peroxiredoxins. *Trends Biochem. Sci.* 2003;28:32-40.
- [4] Gallagher BM, Phelan SA. Investigating transcriptional regulation of Prdx6 in mouse liver cells. *Free Radic. Biol. Med.* 2007;42:1270-1277.
- [5] Kim YJ, Ahn JY, Liang P, Ip C, Zhang Y, Park YM. Human prx1 gene is a target of Nrf2 and is up-regulated by hypoxia/reoxygenation: implication to tumor biology. *Cancer Res.* 2007;67:546-554.
- [6] Chowdhury I, Mo Y, Gao L, Kazi A, Fisher AB, Feinstein SI. Oxidant stress stimulates expression of the human peroxiredoxin 6 gene by a transcriptional mechanism involving an antioxidant response element. *Free Radic. Biol. Med.* 2009;46:146-153.
- [7] Lee IS, Choi WH, Kim JY, Jeong JY, Kim MJ, Nam JH, et al. Transcriptional regulation of the murine 1-cys peroxiredoxin gene by the B cell-specific activator protein, Pax5. J. Cell Biochem. 2008;104:465-476.
- [8] Kropotov A, Usmanova N, Serikov V, Zhivotovsky B, Tomilin N. Mitochondrial

targeting of human peroxiredoxin V protein and regulation of PRDX5 gene expression by nuclear transcription factors controlling biogenesis of mitochondria. *FEBS J.* 2007; 274:5804-5814.

- [9] Chiribau CB, Cheng L, Cucoranu IC, Yu YS, Clempus RE, Sorescu D. FOXO3A regulates peroxiredoxin III expression in human cardiac fibroblasts. J. Biol. Chem. 2008;283:8211-8217.
- [10] Aran M, Ferrero DS, Pagano E, Wolosiuk RA. Typical 2-Cys peroxiredoxins: modulation by covalent transformations and noncovalent interactions. *FEBS J.* 2009;276: 2478-2493.
- [11] Flohe L, Budde H, Hofmann B. Peroxiredoxins in antioxidant defense and redox regulation. *Biofactors* 2003;19:3-10.
- [12] Schroder E, Brennan JP, Eaton P. Cardiac peroxiredoxins undergo complex modifications during cardiac oxidant stress. Am. J. Physiol. Heart Circ. Physiol. 2008;295: H425-433.
- [13] Kumar V, Kitaeff N, Hampton MB, Cannell MB, Winterbourn CC. Reversible oxidation of mitochondrial peroxiredoxin 3 in mouse heart subjected to ischemia and reperfusion. *FEBS Lett.* 2009;583:997-1000.
- [14] Kisucka J, Chauhan AK, Patten IS, Yesilaltay A, Neumann C, Van Etten RA, et al. Peroxiredoxin1 prevents excessive endothelial activation and early atherosclerosis. *Circ. Res.* 2008;103:598-605.
- [15] Choi MH, Lee IK, Kim GW, Kim BU, Han YH, Yu DY, et al. Regulation of PDGF signalling and vascular remodelling by peroxiredoxin II. Nature 2005;435:347-353.
- [16] Wang X, Phelan SA, Petros C, Taylor EF, Ledinski G, Jurgens G, et al. Peroxiredoxin 6 deficiency and atherosclerosis susceptibility in mice: significance of genetic background for assessing atherosclerosis. Atherosclerosis 2004;177:61-70.
- [17] Phelan SA, Wang X, Wallbrandt P, Forsman-Semb K, Paigen B. Overexpression of Prdx6 reduces H₂O₂ but does not prevent diet-induced atherosclerosis in the aortic root. *Free Radic. Biol. Med.* 2003;35:1110-1120.
- [18] Nagy N, Malik G, Fisher AB, Das DK. Targeted disruption of peroxiredoxin 6 gene renders the heart vulnerable to ischemia-reperfusion injury. Am. J. Physiol. Heart Circ. Physiol. 2006;291:H2636-2640.
- [19] Matsushima S, Ide T, Yamato M, Matsusaka H, Hattori F, Ikeuchi M, *et al.* Overexpression of mitochondrial peroxiredoxin-3 prevents left ventricular remodeling and failure after myocardial infarction in mice. *Circulation* 2006;113:1779-1786.
- [20] Tsutsui H, Kinugawa S, Matsushima S. Mitochondrial oxidative stress and dysfunction in myocardial remodelling. *Cardiovasc. Res.* 2009;81:449-456.
- [21] Boschero AC, Stoppiglia LF, Collares-Buzato CB, Bosqueiro JR, Delghingaro-Augusto V, Leite A, et al. Expression of a thioredoxin peroxidase in insulin-producing cells. *Diabetes Metab.* 2002;28:3825-28; discussion 23S108-112.
- [22] Xie X, Li S, Liu S, Lu Y, Shen P, Ji J. Proteomic analysis of mouse islets after multiple low-dose streptozotocin injection. *Biochim. Biophys. Acta* 2008;1784:276-284.
- [23] Barati MT, Merchant ML, Kain AB, Jevans AW, McLeish KR, Klein JB. Proteomic analysis defines altered cellular redox pathways and advanced glycation end-product metabolism in glomeruli of db/db diabetic mice. Am. J. Physiol. Renal Physiol. 2007; 293:F1157-1165.
- [24] Bast A, Wolf G, Oberbaumer I, Walther R. Oxidative and nitrosative stress induces

peroxiredoxins in pancreatic beta cells. Diabetologia 2002;45:867-876.

- [25] Chen L, Na R, Gu M, Salmon AB, Liu Y, Liang H, et al. Reduction of mitochondrial H₂O₂ by overexpressing peroxiredoxin 3 improves glucose tolerance in mice. Aging Cell 2008;7:866-878.
- [26] Hattori F, Oikawa S. Peroxiredoxins in the central nervous system. *Subcell Biochem*. 2007;44:357-374.
- [27] Przedborski S. Peroxiredoxin-2 links Cdk5 to neurodegeneration. Nat. Med. 2007;13: 907-909.
- [28] Qu D, Rashidian J, Mount MP, Aleyasin H, Parsanejad M, Lira A, et al. Role of Cdk5mediated phosphorylation of Prx2 in MPTP toxicity and Parkinson's disease. *Neuron* 2007;55:37-52.
- [29] Lee YM, Park SH, Shin DI, Hwang JY, Park B, Park YJ, et al. Oxidative modification of peroxiredoxin is associated with drug-induced apoptotic signaling in experimental models of Parkinson disease. J. Biol. Chem. 2008;283:9986-9998.
- [30] Fang J, Nakamura T, Cho DH, Gu Z, Lipton SA. S-nitrosylation of peroxiredoxin 2 promotes oxidative stress-induced neuronal cell death in Parkinson's disease. *Proc. Natl. Acad Sci. USA* 2007;104:18742-18747.
- [31] Ichimiya S, Davis JG, O'Rourke DM, Katsumata M, Greene MI. Murine thioredoxin peroxidase delays neuronal apoptosis and is expressed in areas of the brain most susceptible to hypoxic and ischemic injury. DNA Cell Biol. 1997;16:311-321.
- [32] Rashidian J, Rousseaux MW, Venderova K, Qu D, Callaghan SM, Phillips M, et al. Essential role of cytoplasmic cdk5 and Prx2 in multiple ischemic injury models, in vivo. J. Neurosci. 2009;29:12497-12505.
- [33] Kim SU, Jin MH, Kim YS, Lee SH, Cho YS, Cho KJ, et al. Peroxiredoxin II preserves cognitive function against age-linked hippocampal oxidative damage. *Neurobiol. Aging* 2010 (in press).
- [34] Lee KS, Iijima-Ando K, Iijima K, Lee WJ, Lee JH, Yu K, et al. JNK/FOXO-mediated neuronal expression of fly homologue of peroxiredoxin II reduces oxidative stress and extends life span. J. Biol. Chem. 2009;284:29454-29461.
- [35] Manevich Y, Fisher AB. Peroxiredoxin 6, a 1-Cys peroxiredoxin, functions in antioxidant defense and lung phospholipid metabolism. *Free Radic. Biol. Med.* 2005; 38:1422-1432.
- [36] Wang Y, Feinstein SI, Manevich Y, Ho YS, Fisher AB. Lung injury and mortality with hyperoxia are increased in peroxiredoxin 6 gene-targeted mice. *Free Radic. Biol. Med.* 2004;37:1736-1743.
- [37] Wang Y, Feinstein SI, Manevich Y, Ho YS, Fisher AB. Peroxiredoxin 6 gene-targeted mice show increased lung injury with paraquat-induced oxidative stress. *Antioxid. Redox Signal* 2006;8:229-237.
- [38] Wang Y, Phelan SA, Manevich Y, Feinstein SI, Fisher AB. Transgenic mice overexpressing peroxiredoxin 6 show increased resistance to lung injury in hyperoxia. *Am. J. Respir. Cell Mol. Biol.* 2006;34:481-486.
- [39] Wang Y, Manevich Y, Feinstein SI, Fisher AB. Adenovirus-mediated transfer of the 1cys peroxiredoxin gene to mouse lung protects against hyperoxic injury. Am. J. Physiol. Lung Cell Mol. Physiol. 2004;286:L1188-1193.
- [40] Fisher AB, Dodia C, Yu K, Manevich Y, Feinstein SI. Lung phospholipid metabolism in transgenic mice overexpressing peroxiredoxin 6. *Biochim. Biophys. Acta* 2006;1761:

785-792.

- [41] Inoue K, Takano H, Koike E, Warabi E, Yanagawa T, Yanagisawa R, et al. Peroxiredoxin I is a negative regulator of Th2-dominant allergic asthma. Int. Immunopharmacol. 2009;9:1281-1288.
- [42] Li L, Shoji W, Takano H, Nishimura N, Aoki Y, Takahashi R, et al. Increased susceptibility of MER5 (peroxiredoxin III) knockout mice to LPS-induced oxidative stress. Biochem. Biophys. Res. Commun. 2007;355:715-721.
- [43] Eismann T, Huber N, Shin T, Kuboki S, Galloway E, Wyder M, *et al.* Peroxiredoxin-6 protects against mitochondrial dysfunction and liver injury during ischemia-reperfusion in mice. *Am. J. Physiol. Gastrointest. Liver Physiol.* 2009;296:G266-274.
- [44] Roede JR, Stewart BJ, Petersen DR. Decreased expression of peroxiredoxin 6 in a mouse model of ethanol consumption. *Free Radic. Biol. Med.* 2008;45:1551-1558.
- [45] Roede JR, Orlicky DJ, Fisher AB, Petersen DR. Overexpression of peroxiredoxin 6 does not prevent ethanol-mediated oxidative stress and may play a role in hepatic lipid accumulation. J. Pharmacol. Exp. Ther. 2009;330:79-88.
- [46] Kumin A, Huber C, Rulicke T, Wolf E, Werner S. Peroxiredoxin 6 is a potent cytoprotective enzyme in the epidermis. *Am. J. Pathol.* 2006;169:1194-1205.
- [47] Kumin A, Schafer M, Epp N, Bugnon P, Born-Berclaz C, Oxenius A, et al. Peroxiredoxin 6 is required for blood vessel integrity in wounded skin. J. Cell Biol. 2007;179:747-760.
- [48] Neumann CA, Krause DS, Carman CV, Das S, Dubey DP, Abraham JL, et al. Essential role for the peroxiredoxin Prdx1 in erythrocyte antioxidant defence and tumour suppression. *Nature* 2003;424:561-565.
- [49] Cao J, Schulte J, Knight A, Leslie NR, Zagozdzon A, Bronson R, et al. Prdx1 inhibits tumorigenesis via regulating PTEN/AKT activity. EMBO J. 2009;28:1505-1517.
- [50] Neumann CA, Fang Q. Are peroxiredoxins tumor suppressors? *Curr. Opin. Pharmacol.* 2007;7:375-380.
- [51] Chang XZ, Li DQ, Hou YF, Wu J, Lu JS, Di GH, *et al.* Identification of the functional role of peroxiredoxin 6 in the progression of breast cancer. *Breast Cancer Res.* 2007; 9:R76.
- [52] Low FM, Hampton MB, Peskin AV, Winterbourn CC. Peroxiredoxin 2 functions as a noncatalytic scavenger of low-level hydrogen peroxide in the erythrocyte. *Blood* 2007; 109:2611-2617.
- [53] Yang CS, Lee DS, Song CH, An SJ, Li S, Kim JM, *et al.* Roles of peroxiredoxin II in the regulation of proinflammatory responses to LPS and protection against endotoxininduced lethal shock. *J. Exp. Med.* 2007;204:583-594.
- [54] Iuchi Y, Okada F, Tsunoda S, Kibe N, Shirasawa N, Ikawa M, et al. Peroxiredoxin 4 knockout results in elevated spermatogenic cell death via oxidative stress. *Biochem. J.* 2009;419:149-158.
- [55] Han YH, Kim HS, Kim JM, Kim SK, Yu DY, Moon EY. Inhibitory role of peroxiredoxin II (Prx II) on cellular senescence. *FEBS Lett.* 2005;579:4897-4902.
- [56] Radyuk SN, Michalak K, Klichko VI, Benes J, Rebrin I, Sohal RS, *et al.* Peroxiredoxin 5 confers protection against oxidative stress and apoptosis and also promotes longevity in Drosophila. *Biochem. J.* 2009;419:437-445.
- [57] Yoshida Y, Yoshikawa A, Kinumi T, Ogawa Y, Saito Y, Ohara K, *et al.* Hydroxyoctadecadienoic acid and oxidatively modified peroxiredoxins in the blood of

Alzheimer's disease patients and their potential as biomarkers. *Neurobiol. Aging* 2009; 30:174-185.

- [58] Brixius K, Schwinger RH, Hoyer F, Napp A, Renner R, Bolck B, et al. Isoformspecific downregulation of peroxiredoxin in human failing myocardium. *Life Sci.* 2007;81:823-831.
- [59] Sorolla MA, Reverter-Branchat G, Tamarit J, Ferrer I, Ros J, Cabiscol E. Proteomic and oxidative stress analysis in human brain samples of Huntington disease. *Free Radic. Biol. Med.* 2008;45:667-678.
- [60] Kim CW, Cho EH, Lee YJ, Kim YH, Hah YS, Kim DR. Disease-specific proteins from rheumatoid arthritis patients. J. Korean. Med. Sci .2006;21:478-484.
- [61] Karasawa R, Ozaki S, Nishioka K, Kato T. Autoantibodies to peroxiredoxin I and IV in patients with systemic autoimmune diseases. *Microbiol. Immunol.* 2005;49:57-65.
- [62] Kim JH, Bogner PN, Ramnath N, Park Y, Yu J, Park YM. Elevated peroxiredoxin 1, but not NF-E2-related factor 2, is an independent prognostic factor for disease recurrence and reduced survival in stage I non-small cell lung cancer. *Clin. Cancer Res.* 2007;13:3875-3882.
- [63] Soini Y, Kallio JP, Hirvikoski P, Helin H, Kellokumpu-Lehtinen P, Kang SW, et al. Oxidative/nitrosative stress and peroxiredoxin 2 are associated with grade and prognosis of human renal carcinoma. APMIS 2006;114:329-337.

Chapter 12

Thioredoxin Reductase

Abstract

Thioredoxin reductase (TrxR) refers to a class of redox-active enzymes that catalyze the reduction of oxidized thioredoxins using NADPH as the electron donor. TrxRs together with thioredoxins and peroxiredoxins form the mammalian thioredoxin system, which is critically involved in antioxidant defenses as well as many other cellular processes. Gene knockout animal studies suggest a critical involvement of TrxRs in the development and function of certain organs, including heart and brain. TrxRs may also play a role in cancer development, either preventing or promoting carcinogenesis depending on the experimental conditions. Limited observational studies in human subjects point to a potential involvement of TrxRs in certain diseases, such as neurodegeneration and cancer.

1. Overview

Thioredoxin reductase (TrxR) refers to a class of enzymes that catalyze the reduction of oxidized thioredoxins (see Chapter 10 for discussion of thioredoxin) with NADPH as the reducing equivalent [1]. TrxRs belong to a pyridine nucleotide disulfide oxidoreductase family. There are three major isozymes of TrxR in mammals: TrxR1, TrxR2, and thioredoxin glutathione reductase (TGR). Similar to thioredoxin-1 and -2, TrxR1 is present in cytosol and nuclei, and TrxR2 is localized in mitochondria. Both TrxR1 and TrxR2 are widely distributed in mammalian tissues.

The third isozyme TGR is so called because it possesses both thioredoxin reductase and glutathione reductase activities. Hence, TGR can be viewed as a dual functioning enzyme that is involved in keeping both thioredoxins and glutathione at reduced state. TGR is also known as TrxR3. TGR is abundant in testis, suggesting a potential role for this protein in male reproduction. It is also expressed in other tissues, including lung, kidney, heart, and brain. Compared with TrxR1 and TrxR2, TGR has received less attention. It should be noted that in mammals the genes for TrxR1, TrxR2, and TGR are denoted as Txnrd1, Txnrd2, and Txnrd3, respectively. In humans, Txnrd1, Txnrd2, and Txnrd3 are localized on chromosomes 12q23-q24.1, 22q11.21, and 3q21.3, respectively.
2. Biochemistry and Molecular Regulation

2.1. Biochemistry

Mammalian TrxRs exist as homodimers. Each subunit is approximately 54-58 kDa in size and contains one FAD binding domain, one NADPH binding domain, and one interface domain that is required for dimerization. TrxRs are selenoproteins that catalyze the reduction of oxidized thioredoxins using NADPH as the electron donor. As stated above, TGR also catalyzes the reduction of glutathione disulfide (GSSG) to GSH, and thus possesses the glutathione reductase activity (Figure 12-1). TGR is found to have an additional protein disulfide isomerase function in sperm cells, and as such may play a role in sperm maturation [2]. In addition to thioredoxins and GSSG, TrxRs also reduce several other substrates, such as glutaredoxin-2 and protein disulfide isomerase.



Figure 12-1. Thioredoxin reductase (TrxR)-mediated reduction of oxidized thioredoxins and glutathione disulfide (GSSG). As illustrated, all three isozymes of mammalian TrxR are able to catalyze the reduction of oxidized thioredoxins using NADPH as the electron donor. In addition, TrxR3 (TGR) also catalyzes the reduction of GSSG to GSH using NADPH as the electron donor.

TrxRs possess antioxidant functions that occur via several potential mechanisms, as described below.

- TrxRs reduce the oxidized form of thioredoxins to the active form. As discussed in Section 2.1 of Chapter 10, thioredoxins play an important role in antioxidant defenses via serving as a cofactor for various antioxidant enzymes.
- TrxRs may also be involved in the regulation of heme oxygenase-1 (HO-1). HO-1 is a critical antioxidant and anti-inflammatory enzyme in mammalian tissues (Section 2.1 of Chapter 15).
- TrxRs may support the antioxidant functions of various small molecule antioxidants, including ascorbate, α-tocopherol, and ubiquinone (also known as coenzyme Q).

Although the antioxidant activities of TrxRs have been well-established in various cellular systems, it should be borne in mind that TrxRs may also participate in other cellular processes (e.g., cell proliferation, DNA repair, angiogenesis) that are not directly related to their antioxidant activities [1]. These non-antioxidant activities may also contribute to the role of these redox-active proteins in health and disease.

2.2. Molecular Regulation

Regulation of TrxR1 expression occurs via various pathways, including transcriptional control and post-translational modifications [1, 3]. At the transcriptional level, Oct-1, SP1, SP3, and Nrf2 are reported to participate in regulating TrxR1 gene expression. TrxR1 is shown to be induced by certain stimuli, such as oxidants and heavy metal ions via an Nrf2-dependent mechanism [4, 5]. Several post-translational modifications of TrxR1 have been suggested, which include glycosylation, phosphorylation, and oxidative modifications [1]. In contrast to TrxR1, little has been reported on the regulation of TrxR2 and TGR expression in mammalian systems.

3. Role in Health and Disease

3.1. Animal Studies

3.1.1. Experimental Approaches

Because of the efficient reduction of oxidized thioredoxins by TrxRs, it is conceivable that many of the biological activities of TrxRs are mediated by thioredoxins. Indeed, the biological functions of thioredoxins and TrxRs overlap. Our current understanding of the biological effects of TrxRs largely results from studies using gene knockout animal models [6]. In this context, homozygous targeted disruption of either Txnrd1 or Txnrd2 in mice causes embryonic lethality, indicating their essential function in embryonic development [7-9]. Mice with heterozygous deficiency of either Txnrd1 or Txnrd2, however, do not exhibit any gross abnormalities.

In addition, various chemical inhibitors have been used to study the biological functions of TrxRs. These pharmacological inhibitors include auranofin, triphenyl phosphine gold chloride, aurothioglucose, and 1-chloro-2,4-dinitrobenzene [10-13]. Although these chemical agents potently inhibit TrxR activity in biological systems, they lack specificity as do many pharmacological inhibitors for other enzymes and pathways.

3.1.2. Role in Cardiovascular Diseases

As stated above, both Txnrd1 and Txnrd2 genes are essential for murine embryonic development. However, homozygous deletion of Txnrd1 or Txnrd2 yields different phenotypes. Homozygous deletion of Txnrd1 in mice is shown to cause early embryonic lethality around embryonic day 9, with severe growth retardation and lack of primitive mesoderm formation. Of note, cardiac development is not affected by a conditional heart-specific deletion of Txnrd1, suggesting that the cytosolic TrxR1 is essential for embryogenesis but dispensable for cardiac development [7]. In contrast, ubiquitous inactivation of Txnrd2 gene results in embryonic death at embryonic day 13, with severe disruption of hematopoiesis and heart development as well as extensive apoptosis in liver [9]. Cardiac tissue-restricted ablation of Txnrd2 leads to fatal dilated cardiomyopathy, a condition reminiscent of that in Keshan disease and Friedreich's ataxia [9]. The cardiomyopathy in Keshan disease (and possibly also in Friedreich's ataxia) is associated with selenium deficiency. The above experimental findings suggest that in contrast to TrxR1, TrxR2 plays a pivotal role in heart development and cardiac function.

3.1.3. Role in Neurological Diseases

The roles of TrxR1 and TrxR2 in brain development and function have been studied using mice with nervous system-specific deletion of Txnrd1 and Txnrd2 [14]. Mice lacking Txnrd1 in nervous system are significantly smaller than control littermates, and display ataxia and tremor. These mice show cerebellar hypoplasia, which appears to be responsible for ataxia and tremor. In contrast, mice with nervous system-selective deletion of Txnrd2 develop normally, do not show any histological deterioration in the brain, and have a similar lifespan as do control littermates [14]. These results indicate that TrxR1, but not TrxR2 plays a critical role in brain development and function in mice. Recently, it has been reported that overexpression of a human chromosome 22q11.2 segment containing Txnrd2, COMT (catechol-*O*methyltransferase), and ARVCF (armadillo repeat gene deleted in velocardiofacial syndrome) in mice affects incentive learning and working memory; however, the exact contribution of Txnrd2 to the phenotype is unclear [15].

3.1.4. Role in Cancer and Other Conditions

The exact role of TrxRs in cancer remains unclear, and the results are inconsistent among different studies [16, 17]. On the one hand, TrxR1 is reported to activate the p53 tumor suppressor and be selectively targeted by carcinogenic electrophilic compounds, implicating a role in protecting against carcinogenesis [18]. On the other hand, TrxR1 is shown to be over-expressed in many types of cancer and involved in the facilitation of cancer cell growth, pointing to a cancer promoting activity [16, 17, 19]. In this context, various anticancer drugs are reported to target TrxR1 in killing cancer cells, and inhibition of TrxRs has been proposed as a novel strategy for cancer intervention [20, 21].

There is also evidence from animal studies in combination with cell culture experiments showing that TrxRs may have a protective role in male infertility, anemia, oxidative lung injury, cellular senescence and aging, and viral infections [1, 12, 22-24].

3.2. Human Studies and Clinical Perspectives

The potential involvement of TrxRs in human health and disease has been investigated largely via observing the changes of these proteins in disease conditions, such as atherosclerosis, neurodegeneration, and cancer [16, 25, 26]. These observational studies, however, do not establish a causative role for TrxRs in the disease processes. Multiple gene polymorphisms have been identified for human Txnrd1, and among them two intronic singlenucleotide polymorphisms are found to be significantly associated with the development of familial amyotrophic lateral sclerosis [27]. Gene polymorphism studies also suggest an association between Txnrd1 variation and risk of developing advanced colorectal adenoma [28], pointing to the possibility of using Txnrd gene variations as biomarkers for certain disease risk assessment in human populations.

4. Conclusion and Future Directions

TrxRs along with thioredoxins and peroxiredoxins form an important line of antioxidant defenses in mammalian cells and tissues. Although studies using gene knockout animal models suggest a critical role for TrxR1 or TrxR2 in the development and function of brain or heart, the involvement of Trxs in the pathophysiological processes of many common diseases remains to be determined. Similarly, human studies on the role of Trxs in health and disease are much limited and inconclusive. Future studies should focus on creating novel genetic animal models (including both transgenic overexpression and gene knockout) and using these models to further elucidate the biological activities of all three TrxR isozymes and their role in disease processes. Future efforts should also be directed to the identification of novel genetic variations of TrxRs in human subjects and the investigations of the impact of the genetic variations on tissue TrxR expression and activity as well as individual's susceptibility to disease development.

5. References

- [1] Arner ES. Focus on mammalian thioredoxin reductases: important selenoproteins with versatile functions. *Biochim. Biophys. Acta* 2009;1790:495-526.
- [2] Su D, Novoselov SV, Sun QA, Moustafa ME, Zhou Y, Oko R, *et al.* Mammalian selenoprotein thioredoxin-glutathione reductase: roles in disulfide bond formation and sperm maturation. *J. Biol. Chem.* 2005;280:26491-26498.
- [3] Stoytcheva ZR, Berry MJ. Transcriptional regulation of mammalian selenoprotein expression. *Biochim. Biophys. Acta* 2009;1790:1429-1440.
- [4] Chen ZH, Saito Y, Yoshida Y, Sekine A, Noguchi N, Niki E. 4-Hydroxynonenal induces adaptive response and enhances PC12 cell tolerance primarily through induction of thioredoxin reductase 1 via activation of Nrf2. *J. Biol. Chem.* 2005;280: 41921-41927.
- [5] Sakurai A, Nishimoto M, Himeno S, Imura N, Tsujimoto M, Kunimoto M, et al. Transcriptional regulation of thioredoxin reductase 1 expression by cadmium in vascular endothelial cells: role of NF-E2-related factor-2. J. Cell Physiol. 2005;203: 529-537.
- [6] Conrad M. Transgenic mouse models for the vital selenoenzymes cytosolic thioredoxin reductase, mitochondrial thioredoxin reductase and glutathione peroxidase 4. *Biochim. Biophys. Acta* 2009;1790:1575-1585.
- [7] Jakupoglu C, Przemeck GK, Schneider M, Moreno SG, Mayr N, Hatzopoulos AK, et al. Cytoplasmic thioredoxin reductase is essential for embryogenesis but dispensable for cardiac development. Mol. Cell Biol. 2005;25:1980-1988.
- [8] Bondareva AA, Capecchi MR, Iverson SV, Li Y, Lopez NI, Lucas O, et al. Effects of thioredoxin reductase-1 deletion on embryogenesis and transcriptome. Free Radic. Biol. Med. 2007;43:911-923.
- [9] Conrad M, Jakupoglu C, Moreno SG, Lippl S, Banjac A, Schneider M, *et al.* Essential role for mitochondrial thioredoxin reductase in hematopoiesis, heart development, and heart function. *Mol. Cell Biol.* 2004;24:9414-9423.

- [10] Omata Y, Folan M, Shaw M, Messer RL, Lockwood PE, Hobbs D, et al. Sublethal concentrations of diverse gold compounds inhibit mammalian cytosolic thioredoxin reductase (TrxR1). *Toxicol. In Vitro* 2006;20:882-890.
- [11] Cox AG, Brown KK, Arner ES, Hampton MB. The thioredoxin reductase inhibitor auranofin triggers apoptosis through a Bax/Bak-dependent process that involves peroxiredoxin 3 oxidation. *Biochem. Pharmacol.* 2008;76:1097-1109.
- [12] Tipple TE, Welty SE, Rogers LK, Hansen TN, Choi YE, Kehrer JP, et al. Thioredoxinrelated mechanisms in hyperoxic lung injury in mice. Am. J. Respir. Cell Mol. Biol. 2007;37:405-413.
- [13] Seyfried J, Wullner U. Inhibition of thioredoxin reductase induces apoptosis in neuronal cell lines: role of glutathione and the MKK4/JNK pathway. *Biochem. Biophys. Res. Commun.* 2007;359:759-764.
- [14] Soerensen J, Jakupoglu C, Beck H, Forster H, Schmidt J, Schmahl W, et al. The role of thioredoxin reductases in brain development. PLoS One 2008;3:e1813.
- [15] Suzuki G, Harper KM, Hiramoto T, Funke B, Lee M, Kang G, *et al.* Over-expression of a human chromosome 22q11.2 segment including TXNRD2, COMT and ARVCF developmentally affects incentive learning and working memory in mice. *Hum. Mol. Genet.* 2009;18:3914-3925.
- [16] Hatfield DL, Yoo MH, Carlson BA, Gladyshev VN. Selenoproteins that function in cancer prevention and promotion. *Biochim. Biophys. Acta* 2009;1790:1541-1545.
- [17] Selenius M, Rundlof AK, Olm E, Fernandes AP, Bjornstedt M. Selenium and the selenoprotein thioredoxin reductase in the prevention, treatment and diagnostics of cancer. *Antioxid. Redox Signal* 2010;12:867-880.
- [18] Moos PJ, Edes K, Cassidy P, Massuda E, Fitzpatrick FA. Electrophilic prostaglandins and lipid aldehydes repress redox-sensitive transcription factors p53 and hypoxiainducible factor by impairing the selenoprotein thioredoxin reductase. J. Biol. Chem. 2003;278:745-750.
- [19] Yoo MH, Xu XM, Carlson BA, Patterson AD, Gladyshev VN, Hatfield DL. Targeting thioredoxin reductase 1 reduction in cancer cells inhibits self-sufficient growth and DNA replication. *PLoS One* 2007;2:e1112.
- [20] Urig S, Becker K. On the potential of thioredoxin reductase inhibitors for cancer therapy. Semin. Cancer Biol. 2006;16:452-465.
- [21] Lu J, Chew EH, Holmgren A. Targeting thioredoxin reductase is a basis for cancer therapy by arsenic trioxide. *Proc. Natl. Acad. Sci. USA* 2007;104:12288-12293.
- [22] Rohrbach S, Gruenler S, Teschner M, Holtz J. The thioredoxin system in aging muscle: key role of mitochondrial thioredoxin reductase in the protective effects of caloric restriction? *Am. J. Physiol. Regul. Integr. Comp. Physiol.* 2006;291:R927-935.
- [23] Volonte D, Galbiati F. Inhibition of thioredoxin reductase 1 by caveolin 1 promotes stress-induced premature senescence. *EMBO Rep.* 2009;10:1334-1340.
- [24] Kalantari P, Narayan V, Natarajan SK, Muralidhar K, Gandhi UH, Vunta H, et al. Thioredoxin reductase-1 negatively regulates HIV-1 transactivating protein Tatdependent transcription in human macrophages. J. Biol. Chem. 2008;283:33183-33190.
- [25] Furman C, Rundlof AK, Larigauderie G, Jaye M, Bricca G, Copin C, et al. Thioredoxin reductase 1 is upregulated in atherosclerotic plaques: specific induction of the promoter in human macrophages by oxidized low-density lipoproteins. *Free Radic. Biol. Med.* 2004;37:71-85.

- [26] Lovell MA, Xie C, Gabbita SP, Markesbery WR. Decreased thioredoxin and increased thioredoxin reductase levels in Alzheimer's disease brain. *Free Radic. Biol. Med.* 2000; 28:418-427.
- [27] Mitchell J, Morris A, de Belleroche J. Thioredoxin reductase 1 haplotypes modify familial amyotrophic lateral sclerosis onset. *Free Radic. Biol. Med.* 2009;46:202-211.
- [28] Peters U, Chatterjee N, Hayes RB, Schoen RE, Wang Y, Chanock SJ, et al. Variation in the selenoenzyme genes and risk of advanced distal colorectal adenoma. Cancer Epidemiol. Biomarkers Prev. 2008;17:1144-1154.

Chapter 13

Sulfiredoxin

Abstract

Sulfiredoxin (Srx or Srx1) is a newly identified redox-active enzyme that catalyzes the ATP-dependent reduction of hyperoxidized peroxiredoxins (Prxs) in the presence of thiol equivalents, such as thioredoxins or glutathione. In addition, Srx catalyzes protein deglutathionylation and participates in regulating cell redox homeostasis and cell proliferation. Multiple in vitro cell experiments and limited in vivo animal studies demonstrate a role for Srx in protecting against oxidative stress. However, the involvement of Srx in common disease processes remains largely unknown.

1. Overview

Sulfiredoxin (Srx or Srx1) is a recently discovered small molecular mass (~13 kDa) antioxidant enzyme that catalyzes the reduction of sulfinic peroxiredoxins (Prxs) in the presence of ATP. Srx was first identified in yeast by B. Biteau and coworkers in 2003 [1]. It is only found in eukaryotic cells. This is thought to be due to the role of Srx in the restoration of hyperoxidized Prxs, whose counterparts in proeukaryotes are not sensitive to oxidative inactivation [2, 3]. Srx is ubiquitously expressed in mammalian tissues [4]. It is primarily present in cytosol and also found to undergo translocation to mitochondria [5]. In humans, Srx is localized on chromosome 20p13.

2. Biochemistry and Molecular Regulation

2.1. Biochemistry

Srx interacts with and binds to 2-Cys Prxs, forming a complex, as evidenced by their coimmunoprecipitation and structural studies [6]. As described in Chapter 11, Prxs are antioxidant enzymes that exert cytoprotective effects in many models of oxidative stress. However, under highly oxidizing conditions Prxs can be inactivated through hyperoxidation of their active site cysteine residue to form sulfinic acid (-SO₂H). This hyperoxidized form of Prxs cannot be reduced by thioredoxins, and is thus considered as an inactivated state. Srx acts by catalyzing the ATP-dependent formation of a sulfinic acid phosphoric ester on Prxs, which is then reduced by thiol equivalents, such as thioredoxins or glutathione, thereby reactivating the hyperoxidized Prxs [2, 3, 7] (Figure 13-1).



Figure 13-1. Hyperoxidation of peroxiredoxin (Prx) to form sulfinic acid, and sulfiredoxin (Srx)mediated reduction of the Prx sulfinic acid to sulfenic acid. During normal reduction of peroxides (LOOH), the cysteine residue of Prx is oxidized to sulfenic acid (-SOH), which then leads to the formation of disulfide bridge. However, under excessive oxidative conditions, the sulfenic acid can be further oxidized to sulfinic acid (-SO₂H), leading to inactivation of the enzyme. Srx catalyzes reduction of sulfinic acid to sulfenic acid. This reaction is dependent on ATP and utilizes thioredoxin (Trx) or GSH (not shown) as the electron donor.

Srx has been shown to protect against oxidative injury primarily via reactivating Prxs. In addition, Srx has been found to catalyze deglutathionylation of proteins, including 2-Cys Prxs [8, 9]. Deglutathionylation is an important mechanism of post-translational modifications of proteins, including signaling molecules, and has been implicated in cell redox homeostasis and growth regulation [10].

2.2. Molecular Regulation

Srx is inducible by stress conditions, including hyperoxia. It has been demonstrated that Nrf2 regulates Srx expression via an antioxidant response element (ARE)-driven mechanism, which appears to be a major signaling event underlying the inducible expression of Srx under oxidative stress [11, 12]. Indeed, in silico analysis of the 5'-promoter-flanking region of Srx has identified multiple AREs that are highly conserved [13]. In addition, Srx expression has been shown to be regulated by the transcription factor AP-1 [12]. The proximal AP-1 binding site of the Srx promoter is found to be embedded within the ARE. Regulation of Srx may also

involve histone acetylation. It is reported that expression of Srx can be induced by enhancing histone acetylation through treatment of neurons with the histone deacetylase inhibitor trichostatin A [14]. Furthermore, inhibition of histone deacetylase by trichostatin A is associated with decreased formation of hyperoxidized Prxs in neurons exposed to oxidative stress [14].

3. Role in Health and Disease

Since its identification in 2003, Srx has been investigated in both cellular systems and animal models. Srx undergoes translocation into mitochondria in response to oxidative stress, and plays a crucial role in reducing hyperoxidized Prx3 within mitochondria [5]. As stated in Section 1.2 of Chapter 11, Prx3 is mainly localized in mitochondria. Overexpression of mitochondria-targeted Srx in a human embryonic kidney cell line also leads to decreased formation of mitochondrial reactive oxygen species (ROS) and increased resistance to apoptosis elicited by rotenone [5]. Rotenone is a pesticide that selectively blocks the complex I of mitochondrial electron transport chain, leading to increased ROS formation. Similarly, in cultured lung epithelial cells, overexpression of Srx protects cells from hydrogen peroxide (H_2O_2)induced toxicity, whereas attenuation of Srx expression by RNA interference potentiates the toxicity of H_2O_2 [13].

Srx is markedly induced in the lungs of mice exposed to hyperoxia. This hyperoxic induction of Srx is shown to be mediated by Nrf2 signaling [11]. Hyperoxia also elicits the accumulation of the sulfinic form of the mitochondrial Prx3 in lung tissue, likely due to mitochondria being the major site of hyperoxia-induced production of ROS. Hyperoxia induces the degradation of Prx3 in the lung tissue of Nrf2-deficient mice but not in that of wildtype animals, suggesting that in the absence of a sufficient amount of Srx, sulfinic Prx3 is converted to a form that is susceptible to proteolysis [11]. Exposure of mice to cigarette smoke is also found to induce Srx expression in the lung tissue via an Nrf2-dependent mechanism, and such an Nrf2-dependent upregulation of Srx has been proposed to protect against cigarette smoke-elicited oxidative stress in lungs [13].

In addition to its involvement in protecting against oxidative stress, Srx is suggested to also participate in controlling cell proliferation and death pathways [15]. However, its role in cancer remains unclear. On the one hand, elevated levels of Srx in epithelial cancer cell lines are shown to promote cell proliferation. On the other hand, increased Srx levels sensitize the cancer cells to cisplatin-induced killing [15]. In a mouse epidermal JB6 cell model, knock-down of Srx is reported to abolish tumor promoter-induced transformation, suggesting that Srx is required for JB6 cell transformation [16]. Furthermore, screening of patient tissues by tissue microarray demonstrates elevated Srx expression in several types of human skin cancer [16]. The exact role of Srx in carcinogenesis awaits further investigations.

4. Conclusion and Future Directions

Srx, a newly identified redox-active enzyme appears to protect against oxidative stress in cellular systems and certain animal models. In addition to its antioxidant function, Srx also plays a role in reversal of protein glutathionylation as well as in cell growth regulation. The

role of Srx in health and disease currently remains largely unknown and thus warrants further investigations both in animal models and in human subjects.

5. References

- [1] Biteau B, Labarre J, Toledano MB. ATP-dependent reduction of cysteine-sulphinic acid by S. cerevisiae sulphiredoxin. *Nature* 2003;425:980-984.
- [2] Findlay VJ, Tapiero H, Townsend DM. Sulfiredoxin: a potential therapeutic agent? *Biomed. Pharmacother*. 2005;59:374-379.
- [3] Rhee SG, Jeong W, Chang TS, Woo HA. Sulfiredoxin, the cysteine sulfinic acid reductase specific to 2-Cys peroxiredoxin: its discovery, mechanism of action, and biological significance. *Kidney Int. Suppl.* 2007:S3-8.
- [4] Chang TS, Jeong W, Woo HA, Lee SM, Park S, Rhee SG. Characterization of mammalian sulfiredoxin and its reactivation of hyperoxidized peroxiredoxin through reduction of cysteine sulfinic acid in the active site to cysteine. *J. Biol. Chem.* 2004; 279:50994-51001.
- [5] Noh YH, Baek JY, Jeong W, Rhee SG, Chang TS. Sulfiredoxin translocation into mitochondria plays a crucial role in reducing hyperoxidized peroxiredoxin III. J. Biol. Chem. 2009;284:8470-8477.
- [6] Jonsson TJ, Johnson LC, Lowther WT. Structure of the sulphiredoxin-peroxiredoxin complex reveals an essential repair embrace. *Nature* 2008;451:98-101.
- [7] Jonsson TJ, Murray MS, Johnson LC, Lowther WT. Reduction of cysteine sulfinic acid in peroxiredoxin by sulfiredoxin proceeds directly through a sulfinic phosphoryl ester intermediate. *J. Biol. Chem.* 2008;283:23846-23851.
- [8] Findlay VJ, Townsend DM, Morris TE, Fraser JP, He L, Tew KD. A novel role for human sulfiredoxin in the reversal of glutathionylation. *Cancer Res.* 2006;66:6800-6806.
- [9] Park JW, Mieyal JJ, Rhee SG, Chock PB. Deglutathionylation of 2-Cys peroxiredoxin is specifically catalyzed by sulfiredoxin. *J. Biol. Chem.* 2009;284:23364-23374.
- [10] Mieyal JJ, Gallogly MM, Qanungo S, Sabens EA, Shelton MD. Molecular mechanisms and clinical implications of reversible protein S-glutathionylation. *Antioxid. Redox Signal* 2008;10:1941-1988.
- [11] Bae SH, Woo HA, Sung SH, Lee HE, Lee SK, Kil IS, et al. Induction of sulfiredoxin via an Nrf2-dependent pathway and hyperoxidation of peroxiredoxin III in the lungs of mice exposed to hyperoxia. Antioxid. Redox Signal 2009;11:937-948.
- [12] Soriano FX, Baxter P, Murray LM, Sporn MB, Gillingwater TH, Hardingham GE. Transcriptional regulation of the AP-1 and Nrf2 target gene sulfiredoxin. *Mol. Cells* 2009;27:279-282.
- [13] Singh A, Ling G, Suhasini AN, Zhang P, Yamamoto M, Navas-Acien A, et al. Nrf2dependent sulfiredoxin-1 expression protects against cigarette smoke-induced oxidative stress in lungs. Free Radic. Biol. Med. 2009;46:376-386.
- [14] Soriano FX, Papadia S, Bell KF, Hardingham GE. Role of histone acetylation in the activity-dependent regulation of sulfiredoxin and sestrin 2. *Epigenetics* 2009;4:152-158.

- [15] Lei K, Townsend DM, Tew KD. Protein cysteine sulfinic acid reductase (sulfiredoxin) as a regulator of cell proliferation and drug response. *Oncogene* 2008;27:4877-4887.
- [16] Wei Q, Jiang H, Matthews CP, Colburn NH. Sulfiredoxin is an AP-1 target gene that is required for transformation and shows elevated expression in human skin malignancies. *Proc. Natl. Acad. Sci. USA* 2008;105:19738-19743.

Chapter 14

Methionine Sulfoxide Reductase

Abstract

Mammalian methionine sulfoxide reductase (Msr) refers to two structurally unrelated classes of thiol-dependent enzymes, namely, MsrA and MsrB that catalyze the reduction of methionine *S*-sulfoxide and methionine *R*-sulfoxide, respectively. Cell experiments and animal studies demonstrate that Msrs play an important part in protecting against oxidative stress and oxidative stress-associated pathophysiological processes. Observational studies with human tissues and in human populations also suggest a possible role for Msrs in certain disease conditions, including neurological diseases and aging, skin disorders, cancer, and obesity.

1. Overview

Oxidative modifications of proteins have been implicated in various degenerative disease conditions and aging. As compared with most other amino acid residues in proteins, methionine is particularly susceptible to oxidation by reactive oxygen species (ROS) and other related species. The products of such oxidation are a diastereomeric mixture of methionine sulfoxides, as listed below.

- Methionine S-sulfoxide
- Methionine *R*-sulfoxide

These two oxidized forms of methionine can be reduced back to methionine by methionine sulfoxide reductase (Msr).

In mammals, Msr refers to two structurally unrelated classes of monomeric enzymes, namely, MsrA and MsrB. MsrA and MsrB catalyze the reduction of methionine *S*-sulfoxide and methionine *R*-sulfoxide, respectively, using thioredoxin as the electron donor [1, 2]. Mammals have a single MsrA gene, and the encoded Msr enzyme can exist in two forms, one in cytosol and the other in mitochondria. In contrast, there are three MsrB isozymes in mammals, namely, MsrB1, MsrB2, and MsrB3. MsrB1 is a selenoprotein and present in cytosol and nuclei. MsrB2 resides in mitochondria. MsrB3 can be localized in mitochondria or en-

doplasmic reticulum. Msr isozymes are ubiquitously expressed in mammalian tissues, but their expression levels vary in different tissues. The highest expression levels of MrsA have been reported in liver, kidney, and brain. MsrB1, like MsrA, is highly expressed in liver and kidney, whereas the highest expression levels of MsrB2 and MsrB3 are found in heart and skeletal muscle.

Among the four Msr isozymes (MsrA, MsrB1, MsrB2, and MsrB3) only MsrB1 is a selenoprotein, and is shown to have the highest catalytic activity due to the selenocysteine in the active site. MsrB1 is also known as selenoprotein R (Sel R; Section 1.1 of Chapter 6). The other Msr isozymes have cysteine residues in the active sites instead of selenocysteine. Msrs are low molecular mass proteins with molecular masses of approximately 25 kDa for MsrA, 12 kDa for MsrB1, and 20 kDa for MsrB2 and MsrB3. In humans, the genes for MrsA, MsrB1, MsrB2, and MsrB3 are localized on chromosomes 8p23.1, 16p13.3, 10p12, and 12q14.3, respectively.

2. Biochemistry and Molecular Regulation

2.1. Biochemistry

As stated above, MsrA and MsrB catalyze the reduction of methionine *S*-sulfoxide and methionine *R*-sulfoxide, respectively, to form the normal methionine using thioredoxin as the electron donor (Figure 14-1).



Figure 14-1. A simplified schematic illustration of methionine sulfoxide reductase (Msr)-catalyzed reduction of protein methionine sulfoxide. As depicted, thioredoxin (Trx) is used as the electron donor for the Msr-catalyzed reaction. Thioredoxin reductase (TrxR) regenerates the Trx from its oxidized form using NADPH as the reducing equivalent. See text (Section 2.1) for detailed description of the biochemical mechanisms underlying Msr-catalyzed reduction of protein methionine sulfoxide.

Mammalian MsrA has three conserved cysteine residues, which participate in the reaction as catalytic and resolving cysteines. A sulfenic acid intermediate at the catalytic cysteine is generated when this cysteine residue attacks the sulfur of methionine *S*-sulfoxide. Then, thioldisulfide exchange involving the two resolving cysteine residues results in the formation of a disulfide bond on the enzyme surface, which is finally reduced by thioredoxin [2].

Mammalian MsrB1 has one conserved cysteine residue in the N-terminal portion and the catalytic selenocysteine in the C-terminal region. During catalysis, the selenocysteine is converted to selenenic acid intermediate upon interacting with the sulfur of methionine *R*-sulfoxide, and the selenenic acid intermediate rearranges to form seleneylsulfide with the help of the resolving cysteine residue. The seleneylsulfide is reduced to selenocysteine by thioredoxin. In contrast, mammalian MsrB2 and MsrB3 have only one conserved cysteine residue and seem to have evolved a different catalytic mechanism for the direct reduction of the sulfenic acid intermediate by thioredoxin [2].

Although thioredoxin is the most important cellular cofactor for mammalian Msr isozymes, other thiol equivalents may also be utilized as reductants under certain in vitro experimental conditions. These include dithiothreitol, thionein, and selenocystamine [1-3]. However, it remains unclear if these thiol compounds also support Msr activity in vivo.

2.2. Molecular Regulation

Some progress has recently been made regarding the regulation of mammalian Msr isozymes. However, the exact transcriptional and post-transcriptional mechanisms remain largely unknown. Studies in non-mammalian cells, including yeast demonstrate an important role for thioredoxin and calcium phospholipid-binding protein in controlling MsrA gene expression [4, 5]. In addition, yeast glutathione peroxidase-3 is shown to directly interact with MsrA and may modulate the MsrA activity in a redox state-dependent manner [6]. In *Caenorhabditis elegans*, MsrA expression is regulated by the DAF-16/FOXO pathway [7]. The transcriptional regulation of MsrA gene in mammalian cells may occur via two different promoters, whose activity is partially regulated by all-trans retinoic acid via retinoic acid receptors [8]. The transcriptional regulation of human MrsB1 has recently been investigated. Human MsrB1 gene promoter is shown to contain multiple SP1 binding sites, and the expression of MsrB1 is tightly controlled by the transcription factor SP1. In addition, the promoter activity of human MsrB1 gene is also subject to epigenetic modifications, such as methylation [9]. In MsrA-null mice, MsrB1 expression is diminished as compared with that in wild-type animals, suggesting that MsrA may have a role in modulating the expression of MsrB1 [10].

3. Role in Health and Disease

3.1. Animal Studies

3.1.1. Experimental Approaches

Similar to other antioxidant enzymes, both gene knockout and transgenic overexpression animal models have been created to understand the biological functions of Msrs. In this regard, gene knockout of MsrA or MsrB1 in mice does not cause embryonic lethality, indicating that neither gene is essential for murine development. However, both MsrA-null mice and MsrB1-null mice develop phenotypes of oxidative stress and aging-related tissue degeneration, suggesting an antioxidant function for Msrs in mammals [11, 12]. In addition to the animal models, cellular systems have also been utilized to study the biological activities of Msrs. In this context, overexpression of Msrs in cultured cells is shown to protect against oxidative stress, whereas decreased Msr expression results in exacerbation of oxidative cell injury [13-15].

3.1.2. Role in Cardiovascular Diseases

A role for MsrA in ischemic cardiac injury has been suggested by the observation that both the cytosolic and mitochondrial Msr activities are reduced during myocardial ischemiareperfusion injury [16]. Overexpression of MsrA in primary cardiomyocytes leads to attenuation of cell injury induced by hypoxia and reoxygenation [17]. Further studies using MsrAnull mice show that MsrA deficiency exaggerates protein oxidation, myocardial apoptosis, impaired cardiac function, and increased mortality after myocardial infarction, supporting a protective role for MsrA in ischemic heart disease in experimental animals [18].

There is also evidence for an involvement of Msrs in protecting against vascular degeneration and atherogenesis. For instance, overexpression of MrsA in vascular smooth muscle cells protects these cells from oxidative damage [19]. It is reported that methionine oxidation impairs reverse cholesterol transport by apolipoprotein A-I, a major component of highdensity lipoprotein (HDL) [20]. HDL-mediated reverse cholesterol transport in the arterial wall is an important mechanism of suppressing atherogenesis. Reversing methionine oxidation with Msr is found to restore HDL's ability to transport cholesterol in vitro [20]. The exact role of Msrs in protecting against atherosclerosis and other vascular disorders warrants further studies in animal models, such as ApoE-null mice.

3.1.3. Role in Aging and Neurological Diseases

Early studies by E.R. Stadtman and coworkers demonstrated that as compared with wildtype mice, MsrA knockout (MsrA^{-/-}) mice exhibit enhanced sensitivity to hyperoxia-induced injury, and a shorter lifespan under both normal and hyperoxic conditions [11]. These MsrA^{-/-} mice also develop an atypical (tip-toe) walking pattern after six months, and have higher tissue levels of oxidized protein under hyperoxia [11]. These results suggest that MsrA may play an important role in aging and neurological disorders. In line with these observations in mice, overexpression of MrsA in Drosophila melanogaster is found to markedly extend the lifespan. The MsrA-overexpressing flies are also more resistant to paraquat-induced oxidative stress, and the onset of senescence-induced decline in the general activity level and reproductive capacity is delayed markedly [21]. Overexpression of MsrA in yeast also prolongs lifespan. Although substantial evidence supports a critical protective role for MsrA in aging process in various models, a recent study reports that lack of MsrA in mice increases sensitivity to oxidative stress but does not diminish lifespan [22]. The effect of MsrB in aging has been recently examined. Overexpression of MsrB in yeast extends lifespan only under calorie restriction conditions. However, overexpression of MsrB has no influence on aging in Drosophila melanogaster [23].

There is evidence showing that MsrA is an important protector in certain neurodegenerative disorders in animal models. As compared with wild-type animals, MsrA knockout mice exhibit lower locomotor activity and altered gait that exacerbate with age. Dopamine regulation and signaling pathways are impaired in these MsrA^{-/-} mice, suggesting a possible involvement of MsrA in experimental parkinsonism [24]. Recently, it is further reported that dopamine D2 receptor function is compromised in the brain of MsrA^{-/-} mice [25]. In *Drosophila melanogaster*, MsrA and S-methyl-L-cysteine (a substrate of MsrA) are shown to prevent parkinsonism caused by ectopic expression of human α -synuclein in the flies' nervous system [26]. In cultured cells, MsrA blocks dopaminergic cell death and protein aggregation induced by rotenone and mutant α -synuclein [27]. Both rotenone and mutant α -synuclein are implicated in the pathophysiology of parkinsonism.

Methionine oxidation has been suggested to play a role in β -amyloid toxicity, an important mechanistic event in Alzheimer's disease. MsrA^{-/-} mice exhibit enhanced neurodegeneration in brain hippocampus as compared with wild-type mice. In addition, a loss of astrocyte integrity, elevated levels of β -amyloid deposition, and tau phosphorylation are dominant in various regions of MsrA^{-/-} hippocampus but not in those of wild-type mice [28]. The cultured brain slices of the hippocampus region of MsrA^{-/-} mice also show increased sensitivity to oxidative stress [28]. These findings indicate that a deficiency of MsrA activity may foster oxidative stress, accumulation of faulty proteins via methionine oxidation, deposition of aggregated proteins, and eventually premature brain cell death.

3.1.4. Role in Other Diseases and Conditions

Msrs have also been implicated in the protection of various other disease conditions involving an oxidative stress mechanism. As stated above in Section 1, MsrB1 is widely distributed throughout different tissues, but the highest levels are found in liver and kidney. MsrB1 knockout mouse model has recently been created, and studies using this model demonstrate that liver and kidney are most affected in terms of susceptibility to oxidative damage of proteins [12]. These observations suggest a role for MsrB1 in protecting against oxidative stress in liver and kidney

Msrs are expressed in eye tissues including lens and retina, and ample evidence supports a beneficial role for Msrs in eye disorders, especially cataract [29, 30]. Deletion of MsrA in mice results in hyperbaric oxygen-induced cataract possibly due to failure in repairing mitochondrial cytochrome c whose methionine residues are oxidized to methionine sulfoxides [30]. Targeted silencing of MsrA in cultured retinal pigmented epithelium or lens cells sensitizes these cells to oxidative injury [31, 32]. MsrB isozymes are also found to be expressed in retinal and lens cells, and protect these cells from oxidative stress [33, 34].

There is also evidence from both in vitro and in vivo models, showing beneficial effects of Msrs in skin disorders, immunity, and infections [35-39].

3.2. Human Studies and Clinical Perspectives

Although being not as convincing as that from animal experiments, evidence from studies in human subjects does exist pointing to a possible role for Msrs in human health and disease. Our current knowledge on the involvement of Msrs in human disease conditions results primarily from determining the changes of Msrs in tissue samples of patients with disease conditions as well as population studies aiming to identify the genetic loci or genetic variations of Msrs in regard to disease risk. Largely based on these studies, a potential involvement of Msrs has been suggested in several disease conditions, including aging and neurological diseases, skin disorders, cancer, and obesity.

3.2.1. Role in Human Aging and Neurological Diseases

Gene polymorphism studies suggest an association between MsrA (-402C/T) polymorphism and longevity in a population of 1534 ethnic Tartars [40]. Studies with brain tissues of Alzheimer's patients demonstrate a significant decrease in MsrA activity compared with control subjects. mRNA analysis further suggests that the decreased MsrA activity is likely due to a post-transcriptional modification [41]. A recent study has identified a novel 4-base pair MsrA deletion associated with schizophrenia. It is shown that the 4-base pair deletion within the MsrA gene is under-transmitted to schizophrenia patients in two different populations, and causes reduced gene expression in individuals homozygous for the deletion [42]. The findings suggest that there may be a link between the MsrA polymorphism and risk of developing schizophrenia. Further studies on the mechanisms by which MsrA is involved in schizophrenia pathophysiology may provide important insights into the biological underpinnings of this neurological disorder.

3.2.2. Role in Human Skin Diseases

Both Msrs A and B are expressed in human skin tissue, including melanocytes. Patients with the depigmentation disorder vitiligo are shown to have low catalase expression and constantly accumulate millimolar concentrations of hydrogen peroxide (H_2O_2) in their skin tissues. Such high concentrations of H_2O_2 can readily oxidize methionine residues in proteins. Indeed, in vivo FT-Raman spectroscopy studies have revealed the presence of methionine sulfoxides in the depigmented skin of patients with active vitiligo [43]. Expression of MsrA and MsrB is also markedly decreased in the epidermis of patients with vitiligo compared with healthy controls. In vitro studies demonstrate that the recombinant human MsrA and MsrB enzymes are readily inactivated by H_2O_2 at concentrations relevant to those seen in the depigmented skin tissues of patients with vitiligo [43]. Together, these observations suggest an important protective role for Msrs in human vitiligo.

Senile hair graying may also involve loss of Msr activity in the hair follicles. An in vivo study with FT-Raman spectroscopy shows that human gray/white scalp hair shafts accumulate H_2O_2 in millimolar concentrations [44]. An almost absent protein expression for catalase and MsrA and MsrB in association with a functional loss of methionine sulfoxide repair is demonstrated in the entire gray hair follicles. Furthermore, oxidation of the methionine residues, including Met 364 in the active site of tyrosinase, the key enzyme in melanogenesis, leads to enzyme inactivation [44]. Collectively, these results point to a causal involvement of Msr inactivation in senile hair graying.

3.2.3. Role in Human Cancer and Obesity

MsrA gene is linked to carcinogenesis in humans. It is reported that chromosome 8p deletions contribute to metastasis of hepatocellular carcinoma. Since human MsrA is localized to chromosome 8p23.1, MsrA is suggested to be a candidate gene. Further studies using oligonucleotide microarrays have identified MsrA gene on chromosome 8p as a possible metastasis suppressor for human hepatitis B virus-positive hepatocellular carcinoma [45]. In a hepatocellular cancer cell line, overexpression of MsrA is shown to inhibit cell colony formation and invasion [45]. The protective role of MsrA as well as other Msr isozymes in carcinogenesis warrants further investigations.

The accumulation of abnormal amounts of intra-abdominal fat (central adiposity) is associated with serious adverse metabolic and cardiovascular outcomes, including type 2 diabetes and atherosclerotic coronary heart disease. A recent genome-wide association scan metaanalysis involving data from 38,580 individuals has identified three loci influencing adiposity and fat accumulation, among which is MsrA. This locus of MsrA is strongly associated with central adiposity [46]. Further mechanistic studies will provide novel insights into the role of MsrA in human physiology and development of obesity.

4. Conclusion and Future Directions

Msr isozymes protect cells from oxidative stress via catalyzing the reduction of methionine sulfoxides to methionine. Studies using genetically manipulated animal models have provided convincing evidence supporting a beneficial role for Msrs in various pathophysiological conditions involving an oxidative and inflammatory stress mechanism. Evidence from studies in human subjects also suggests a potential involvement of Msrs in modulating the risk of developing certain diseases. There is a great need to understand the effects (either protective or detrimental) of various Msr isozymes in pathophysiology of other common diseases in animal models, including diabetes and metabolic syndrome, cancer, as well as disorders affecting lung, liver, and kidney. Future studies should also emphasize translational and clinical research to further explore the biological functions of Msr isozymes in relation to human health and disease.

5. References

- [1] Kim HY, Gladyshev VN. Methionine sulfoxide reductases: selenoprotein forms and roles in antioxidant protein repair in mammals. *Biochem. J.* 2007;407:321-329.
- [2] Lee BC, Dikiy A, Kim HY, Gladyshev VN. Functions and evolution of selenoprotein methionine sulfoxide reductases. *Biochim. Biophys. Acta* 2009;1790:1471-1477.
- [3] Sagher D, Brunell D, Hejtmancik JF, Kantorow M, Brot N, Weissbach H. Thionein can serve as a reducing agent for the methionine sulfoxide reductases. *Proc. Natl. Acad. Sci.* USA 2006;103:8656-8661.
- [4] Hanbauer I, Moskovitz J. The yeast cytosolic thioredoxins are involved in the regulation of methionine sulfoxide reductase A. *Free Radic. Biol. Med.* 2006;40:1391-1396.
- [5] Hanbauer I, Boja ES, Moskovitz J. A homologue of elongation factor 1 gamma regulates methionine sulfoxide reductase A gene expression in Saccharomyces cerevisiae. *Proc. Natl. Acad. Sci. USA* 2003;100:8199-8204.
- [6] Kho CW, Lee PY, Bae KH, Cho S, Lee ZW, Park BC, et al. Glutathione peroxidase 3 of Saccharomyces cerevisiae regulates the activity of methionine sulfoxide reductase in a redox state-dependent way. *Biochem. Biophys. Res. Commun.* 2006;348:25-35.
- [7] Minniti AN, Cataldo R, Trigo C, Vasquez L, Mujica P, Leighton F, et al. Methionine

sulfoxide reductase A expression is regulated by the DAF-16/FOXO pathway in Caenorhabditis elegans. *Aging Cell* 2009;8:690-705.

- [8] Pascual I, Larrayoz IM, Rodriguez IR. Retinoic acid regulates the human methionine sulfoxide reductase A (MSRA) gene via two distinct promoters. *Genomics* 2009;93:62-71.
- [9] De Luca A, Sacchetta P, Nieddu M, Di Ilio C, Favaloro B. Important roles of multiple Sp1 binding sites and epigenetic modifications in the regulation of the methionine sulfoxide reductase B1 (MsrB1) promoter. *BMC Mol. Biol.* 2007;8:39.
- [10] Moskovitz J, Stadtman ER. Selenium-deficient diet enhances protein oxidation and affects methionine sulfoxide reductase (MsrB) protein level in certain mouse tissues. *Proc. Natl. Acad. Sci. USA* 2003;100:7486-7490.
- [11] Moskovitz J, Bar-Noy S, Williams WM, Requena J, Berlett BS, Stadtman ER. Methionine sulfoxide reductase (MsrA) is a regulator of antioxidant defense and lifespan in mammals. *Proc. Natl. Acad. Sci. USA* 2001;98:12920-12925.
- [12] Fomenko DE, Novoselov SV, Natarajan SK, Lee BC, Koc A, Carlson BA, et al. MsrB1 (methionine-R-sulfoxide reductase 1) knock-out mice: roles of MsrB1 in redox regulation and identification of a novel selenoprotein form. *J. Biol. Chem.* 2009;284: 5986-5993.
- [13] Moskovitz J, Flescher E, Berlett BS, Azare J, Poston JM, Stadtman ER. Overexpression of peptide-methionine sulfoxide reductase in Saccharomyces cerevisiae and human T cells provides them with high resistance to oxidative stress. *Proc. Natl. Acad. Sci. USA* 1998;95:14071-14075.
- [14] Cabreiro F, Picot CR, Perichon M, Friguet B, Petropoulos I. Overexpression of methionine sulfoxide reductases A and B2 protects MOLT-4 cells against zinc-induced oxidative stress. *Antioxid. Redox Signal* 2009;11:215-225.
- [15] Picot CR, Petropoulos I, Perichon M, Moreau M, Nizard C, Friguet B. Overexpression of MsrA protects WI-38 SV40 human fibroblasts against H₂O₂-mediated oxidative stress. *Free Radic. Biol. Med.* 2005;39:1332-1341.
- [16] Picot CR, Perichon M, Lundberg KC, Friguet B, Szweda LI, Petropoulos I. Alterations in mitochondrial and cytosolic methionine sulfoxide reductase activity during cardiac ischemia and reperfusion. *Exp. Gerontol.* 2006;41:663-667.
- [17] Prentice HM, Moench IA, Rickaway ZT, Dougherty CJ, Webster KA, Weissbach H. MsrA protects cardiac myocytes against hypoxia/reoxygenation induced cell death. *Biochem. Biophys. Res. Commun.* 2008;366:775-778.
- [18] Erickson JR, Joiner ML, Guan X, Kutschke W, Yang J, Oddis CV, et al. A dynamic pathway for calcium-independent activation of CaMKII by methionine oxidation. *Cell* 2008;133:462-474.
- [19] Haenold R, Wassef R, Brot N, Neugebauer S, Leipold E, Heinemann SH, et al. Protection of vascular smooth muscle cells by over-expressed methionine sulphoxide reductase A: role of intracellular localization and substrate availability. *Free Radic. Res.* 2008;42:978-988.
- [20] Shao B, Cavigiolio G, Brot N, Oda MN, Heinecke JW. Methionine oxidation impairs reverse cholesterol transport by apolipoprotein A-I. *Proc. Natl. Acad. Sci. USA* 2008; 105:12224-12229.
- [21] Ruan H, Tang XD, Chen ML, Joiner ML, Sun G, Brot N, et al. High-quality life extension by the enzyme peptide methionine sulfoxide reductase. *Proc. Natl. Acad. Sci. USA*

2002;99:2748-2753.

- [22] Salmon AB, Perez VI, Bokov A, Jernigan A, Kim G, Zhao H, et al. Lack of methionine sulfoxide reductase A in mice increases sensitivity to oxidative stress but does not diminish life span. FASEB J. 2009;23:3601-3608.
- [23] Shchedrina VA, Vorbruggen G, Lee BC, Kim HY, Kabil H, Harshman LG, et al. Overexpression of methionine-R-sulfoxide reductases has no influence on fruit fly aging. *Mech. Ageing Dev.* 2009;130:429-443.
- [24] Oien DB, Osterhaus GL, Latif SA, Pinkston JW, Fulks J, Johnson M, et al. MsrA knockout mouse exhibits abnormal behavior and brain dopamine levels. *Free Radic. Biol. Med.* 2008;45:193-200.
- [25] Oien DB, Ortiz AN, Rittel AG, Dobrowsky RT, Johnson MA, Levant B, et al. Dopamine D(2) receptor function is compromised in the brain of the methionine sulfoxide reductase A knockout mouse. J. Neurochem. 2010;114:51-61.
- [26] Wassef R, Haenold R, Hansel A, Brot N, Heinemann SH, Hoshi T. Methionine sulfoxide reductase A and a dietary supplement S-methyl-L-cysteine prevent Parkinson's-like symptoms. J. Neurosci. 2007;27:12808-12816.
- [27] Liu F, Hindupur J, Nguyen JL, Ruf KJ, Zhu J, Schieler JL, et al. Methionine sulfoxide reductase A protects dopaminergic cells from Parkinson's disease-related insults. *Free Radic. Biol. Med.* 2008;45:242-255.
- [28] Pal R, Oien DB, Ersen FY, Moskovitz J. Elevated levels of brain-pathologies associated with neurodegenerative diseases in the methionine sulfoxide reductase A knockout mouse. *Exp. Brain Res.* 2007;180:765-774.
- [29] Brennan LA, Kantorow M. Mitochondrial function and redox control in the aging eye: role of MsrA and other repair systems in cataract and macular degenerations. *Exp. Eye Res.* 2009;88:195-203.
- [30] Brennan LA, Lee W, Cowell T, Giblin F, Kantorow M. Deletion of mouse MsrA results in HBO-induced cataract: MsrA repairs mitochondrial cytochrome c. *Mol. Vis.* 2009; 15:985-999.
- [31] Marchetti MA, Lee W, Cowell TL, Wells TM, Weissbach H, Kantorow M. Silencing of the methionine sulfoxide reductase A gene results in loss of mitochondrial membrane potential and increased ROS production in human lens cells. *Exp. Eye Res.* 2006;83: 1281-1286.
- [32] Lee JW, Gordiyenko NV, Marchetti M, Tserentsoodol N, Sagher D, Alam S, et al. Gene structure, localization and role in oxidative stress of methionine sulfoxide reductase A (MSRA) in the monkey retina. *Exp. Eye Res.* 2006;82:816-827.
- [33] Marchetti MA, Pizarro GO, Sagher D, Deamicis C, Brot N, Hejtmancik JF, et al. Methionine sulfoxide reductases B1, B2, and B3 are present in the human lens and confer oxidative stress resistance to lens cells. *Invest. Ophthalmol. Vis. Sci.* 2005;46: 2107-2112.
- [34] Pascual I, Larrayoz IM, Campos MM, Rodriguez IR. Methionine sulfoxide reductase B2 is highly expressed in the retina and protects retinal pigmented epithelium cells from oxidative damage. *Exp. Eye Res.* 2010;90:420-428.
- [35] Picot CR, Moreau M, Juan M, Noblesse E, Nizard C, Petropoulos I, et al. Impairment of methionine sulfoxide reductase during UV irradiation and photoaging. *Exp. Gerontol.* 2007;42:859-863.
- [36] Zhou Z, Li CY, Li K, Wang T, Zhang B, Gao TW. Decreased methionine sulphoxide

reductase A expression renders melanocytes more sensitive to oxidative stress: a possible cause for melanocyte loss in vitiligo. *Br. J. Dermatol.* 2009;161:504-509.

- [37] Agbas A, Moskovitz J. The role of methionine oxidation/reduction in the regulation of immune response. *Curr. Signal Transduct. Ther.* 2009;4:46-50.
- [38] Sasindran SJ, Saikolappan S, Dhandayuthapani S. Methionine sulfoxide reductases and virulence of bacterial pathogens. *Future Microbiol.* 2007;2:619-630.
- [39] Herzberg MC, Nobbs A, Tao L, Kilic A, Beckman E, Khammanivong A, et al. Oral streptococci and cardiovascular disease: searching for the platelet aggregationassociated protein gene and mechanisms of Streptococcus sanguis-induced thrombosis. *J. Periodontol.* 2005;76:2101-2105.
- [40] Pauk VV, Nasibullin TR, Tuktarova IA, Zueva LP, Mustafina OE. Antioxidant gene polymorphism and longevity. Adv. Gerontol. 2008;21:593-595.
- [41] Gabbita SP, Aksenov MY, Lovell MA, Markesbery WR. Decrease in peptide methionine sulfoxide reductase in Alzheimer's disease brain. J. Neurochem. 1999;73:1660-1666.
- [42] Walss-Bass C, Soto-Bernardini MC, Johnson-Pais T, Leach RJ, Ontiveros A, Nicolini H, et al. Methionine sulfoxide reductase: a novel schizophrenia candidate gene. Am. J. Med. Genet. B Neuropsychiatr. Genet. 2009;150B:219-225.
- [43] Schallreuter KU, Rubsam K, Gibbons NC, Maitland DJ, Chavan B, Zothner C, et al. Methionine sulfoxide reductases A and B are deactivated by hydrogen peroxide (H₂O₂) in the epidermis of patients with vitiligo. *J. Invest. Dermatol.* 2008;128:808-815.
- [44] Wood JM, Decker H, Hartmann H, Chavan B, Rokos H, Spencer JD, et al. Senile hair graying: H₂O₂-mediated oxidative stress affects human hair color by blunting methionine sulfoxide repair. *FASEB J.* 2009;23:2065-2075.
- [45] Lei KF, Wang YF, Zhu XQ, Lu PC, Sun BS, Jia HL, et al. Identification of MSRA gene on chromosome 8p as a candidate metastasis suppressor for human hepatitis B viruspositive hepatocellular carcinoma. *BMC Cancer* 2007;7:172.
- [46] Lindgren CM, Heid IM, Randall JC, Lamina C, Steinthorsdottir V, Qi L, et al. Genomewide association scan meta-analysis identifies three Loci influencing adiposity and fat distribution. *PLoS Genet.* 2009;5:e1000508.

Chapter 15

Heme Oxygenase

Abstract

Mammalian heme oxygenase (HO) refers to a family of two major isozymes, namely, HO-1 and HO-2 that catalyze the first and rate-limiting step in the oxidative degradation of heme to eventually produce bilirubin along with the release of carbon monoxide and iron. The antioxidant and anti-inflammatory activities of HO enzymes and their products bilirubin and carbon monoxide are well-recognized. Extensive animal studies have established a critical role for HO enzymes, especially HO-1 in protecting against various disease conditions, ranging from cardiovascular diseases to skin disorders. However, studies also suggest a promoting activity for HO-1 in cancer growth and progression in animal models, which is believed to result from the proproliferative and proangiogenic effects of HO-1. A growing body of evidence from studies in human subjects indicates a beneficial role for HO-1 in various disease processes, which is consistent with findings from animal experiments. On the other hand, studies in cancer patients suggest that HO-1 may play either an inhibitory or a promoting role in human cancer development depending on the types of cancer.

1. Overview

Heme oxygenase (HO) was originally identified in 1968 and 1969 by R. Tenhunen and coworkers in seminal papers where they characterized the enzyme HO and its cellular localization [1, 2]. HO is the rate-limiting enzyme in heme catabolism. In mammals, there are two major isozymes of HO: the inducible form HO-1 and the constitutively expressed form HO-2. Expression of HO-1 occurs at low levels in most tissues under normal conditions with the exception of spleen (the site of red blood cell hemoglobin turnover) and several other unique cell types (e.g., renal inner medullary cells, Kupffer cells in the liver, Purkinje cells in the cerebellum, CD4⁺/CD25⁺ regulatory T cells). HO-2 is constitutively expressed in mammalian tissues under physiological conditions, with the highest levels found in brain and testes. The molecular masses for HO-1 and HO-2 are 32 and 36 kDa, respectively. In mammalian cells, HO-1 and HO-2 are mainly associated with endoplasmic reticulum. HO-1 is also reported to localize to distinct subcellular compartments, including plasma membrane caveolae, mito-chondria, and nuclei. A putative third isozyme of HO, namely, HO-3 is reported to be ex-

pressed in rat tissues. However, the in vivo significance of this putative isozyme in mammalian systems is unclear. In humans, HO-1 and HO-2 are localized on chromosomes 22q13.1 and 16p13.3, respectively.

2. Biochemistry and Molecular Regulation

2.1. Biochemistry

Both HO-1 and HO-2 catalyze the first and rate-limiting step in the oxidative degradation of heme to form the open-chain tetrapyrole biliverdin with concurrent release of ferrous ion (Fe²⁺) and carbon monoxide (CO). This reaction requires molecular oxygen as well as reducing equivalents derived from NADPH cytochrome P450 reductase. The HO-catalyzed reaction displays regiospecificity for the heme molecule such that only the α -isomer of biliverdin is produced. The biliverdin formed is subsequently reduced to bilirubin by biliverdin reductase (Figure 15-1) [3, 4].



Figure 15-1. Heme oxygenase (HO)-catalyzed degradation of heme. As illustrated, HO catalyzes the degradation of heme to form biliverdin along with the release of carbon monoxide (CO) and iron. Biliverdin is further converted to bilirubin by biliverdin reductase (BR). NADPH serves as the electron donor for both HO- and BR-catalyzed reactions.

The biological activities of HO isozymes are largely determined by their biochemical properties, which include the follows.

- Degradation of heme
- Formation of biliverdin
- Formation of bilirubin
- Production of carbon monoxide
- Release of iron

Since heme is a pro-oxidant, its degradation by HO minimizes oxidative stress induced by excess heme. Both biliverdin and bilirubin possess antioxidant activities as well as other cytoprotective functions, such as anti-inflammation and antiproliferation (see Chapter 19 for discussion of bilirubin). Carbon monoxide is also known to exert anti-inflammatory, antiproliferative, and vasodilatory activities [5]. Due to the pro-oxidant activity of iron (e.g., participation in Fenton reaction, leading to the formation of hydroxyl radical, the most reactive species formed in biological systems), the iron released from HO-catalyzed reaction would seem to lead to detrimental effects. However, elevation of HO activity and release of iron are usually associated with concurrent induction of ferritin, an iron-chelating protein (see Chapter 18 for discussion of ferritin). Induction of ferritin thus minimizes the pro-oxidant potential of the released iron. The reaction products biliverdin, bilirubin, and carbon monoxide primarily mediate the antioxidative and anti-inflammatory activities of the HO isozymes under various disease conditions. Recent work has identified HO-1 and its gaseous product, carbon monoxide to also possess potent proangiogenic properties [6, 7]. There is also evidence suggesting that HO-2 may act as an intracellular sensor of oxygen, nitric oxide, and carbon monoxide in mammalian systems [8].

2.2. Molecular Regulation

Regulation of Mammalian HO-1 expression occurs primarily at the transcriptional level though post-transcriptional modifications may also be involved. HO-1 gene expression is highly inducible by diverse stimuli, including heme, reactive oxygen and nitrogen species (ROS/RNS), hyperoxia, hypoxia, electrophiles, heavy metals, ultraviolet light, and inflammatory cytokines [3, 4].

A number of signaling pathways have been implicated in the regulation of HO-1 gene expression in mammalian cells. These include mitogen activated protein kinases (MAPKs), tyrosine kinases, phosphatidylinositol 3-kinase (PI3K) as well as protein kinases A, G, and C. Consistent with the involvement of diverse signaling cascades in the induction of HO-1 gene expression, the promoter region of HO-1 contains a wide variety of regulatory elements for transcription factors, among which are AP-1, NF- κ B, Nrf2, and hypoxia-inducible factor-1 (HIF-1). Evidence also suggests that HO-1 protein localized in nuclei may activate transcription factors (e.g., AP-1) important in oxidative stress [9]. It is thought that nuclear localization of HO-1 protein may serve to upregulate genes that promote cytoprotection against oxidative stress. The exact molecular mechanisms underlying HO-1-mediated transcriptional regulation remain to be elucidated.

In contrast to the structure of the HO-1 gene, which has a simple composition of 5 exons and 4 introns, that of mammalian HO-2 is among the most complex gene structures. Five or more transcripts for HO-2 are present in mammalian tissues, with tissue-dependent patterns of display and abundance. However, studies on transcriptional regulation of HO-2 gene have revealed that the glucocorticoid response element (GRE) is the only functional responsible element in the promoter region of HO-2 gene [8]. Although HO-2 is constitutively expressed, it expression is subject to redox regulation. In addition, there is a direct protein-protein interaction between HO-2 and HO-1 [10]. Furthermore, it has been shown that HO-2 can down-regulate the expression of HO-1, thereby directing the coordinated expression of HO-1 and HO-2 [11].

3. Role in Health and Disease

3.1. Animal Studies

3.1.1. Experimental Approaches

Various experimental approaches have been utilized to investigate the biological functions of both HO-1 and HO-2 enzymes. These include: (1) transgenic overexpression and gene knockout mouse models, (2) viral vector-mediated HO gene transfer, and (3) use of chemical inhibitors (e.g., metalloporphyrins) or chemical inducers (e.g., hemin) to modulate HO activity in experimental animals. Among these various approaches, genetically manipulated mouse models have been extensively employed, and have provided the most convincing evidence on the role of HO-1 and HO-2 enzymes in particular disease conditions. In contrast, although the above chemical inducers or inhibitors of HO exhibit certain selectivity for HO activity, the results should be interpreted with caution due to other potential effects of these chemical agents.

3.1.2. Role in Cardiovascular Diseases

HO-1 is one of the best characterized antioxidants proven to be effective in ameliorating cardiovascular pathophysiology associated with oxidative and inflammatory stress in animal models. The cardiovascular disorders in which HO-1 plays a beneficial role include hypertension, atherosclerosis, myocardial ischemia-reperfusion injury, cardiac transplantation, and heart failure. These well-documented cardiovascular protective effects of HO-1 in experimental animals have been reviewed extensively [12-16].

HO-2 is constitutively expressed in myocardium and vasculature. Multiple studies also suggest a potential protective role for HO-2 in cardiovascular pathophysiology. HO-2 is shown to regulate vascular tone, and deletion of HO-2 causes endothelial cell dysfunction and increased blood pressure in experimental models [17-19]. However, a recent study reports that targeted disruption of HO-2 in mice does not aggravate hypertension induced by either chronic infusion of angiotensin II or inhibition of nitric oxide by N(G)-nitroarginine methyl ester [20]. Hypoxemia is found to induce HO-2 protein expression in myocardium of mice [21]. But, the role of HO-2 in myocardial ischemia-reperfusion injury as well as atherosclerosis has not been documented in the literature. Hence, the function of HO-2 in cardiovascular diseases remains to be further determined.

3.1.3. Role in Diabetes and Metabolic Syndrome

Upregulation of HO-1 by chemical inducers (e.g., hemin, tin chloride, cobalt protoporphyrin) attenuates hyperglycemia and insulin resistance, and protects against diabetic complications in various animal models of diabetes [22-25]. In line with these in vivo observations, overexpression of HO-1 is found to protect cultured pancreatic β -cells from apoptosis caused by prodiabetic stimuli, including inflammatory cytokines [26]. There is also evidence supporting a beneficial role of HO-2 in diabetes and metabolic syndrome in experimental animals. For instance, HO-2-null mice are shown to be obese, display insulin resistance, and have high blood pressure [19]. HO-2-deficient mice are also more susceptible to streptozotocin-induced diabetes, diabetic nephropathy, and oxidative stress. The diabetic nephropathy in these HO-2-deficient mice can be ameliorated by induction of HO-1 activity via treatment with cobalt protoporphyrin, suggesting a complementary function of HO-1 and HO-2 in protecting against experimental diabetes [27].

3.1.4. Role in Neurological Diseases

HO-2 is highly expressed in brain and cerebral vessels, and has been extensively studied in regard to neuroprotection. Substantial evidence supports that HO-2 plays a protective role in various neurological disorders, including focal cerebral ischemic injury, injury due to intracerebral hemorrhage, traumatic brain and spinal cord injury, and neurodegenerative disorders in animal models. In addition, HO-1 is also found to be neuroprotective.

3.1.4.1. CEREBRAL INJURY DUE TO ISCHEMIA AND HEMORRHAGE

Targeted disruption of HO-2 in mice leads to exacerbation of infarct volume after middle cerebral artery occlusion [28]. Infusion of tin protoporphyrin IX, an HO inhibitor, increases infarct volume in wild-type mice but not in HO-2-null mice. It is further shown that the greater infarct volume in HO-2-null mice is not attributable to a larger volume at risk, lower intraischemic blood flow, or poor reflow, but rather to a neuroprotective effect of HO-2 activity [28]. Indeed, overexpression of HO-2 in neurons protects these cells from oxidative injury [29]. Transgenic overexpression of HO-1 in mice has been demonstrated to be neuroprotective in a model of permanent middle cerebral artery occlusion [30]. Stimulation of HO-1 activity by pharmacological agents also suppresses ischemic injury during the acute period of stroke in experimental animals. Consistently, overexpression of HO-1 in neuronal cells is found to protect these cells from toxicity elicited by oxidative insults, including ROS and heme.

Genetic deletion of HO-2 renders cultured neurons more vulnerable to hemin-induced cytotoxicity. Consistently, HO-2 knockout mice exhibit increased brain injury volume and worsened neurological functions following exposure to free heme liberated from hemorrhage [31]. While these in vitro and in vivo findings demonstrate that HO-2 is a crucial neuroprotective enzyme in detoxifying high levels of heme in the brain following intracerebral hemorrhage, other studies suggest a detrimental effect of HO-2 as well as HO-1 in intracerebral hemorrhage [32, 33]. Thus, further work is needed to determine the exact role of HO-2 in intracerebral hemorrhage-elicited brain injury.

3.1.4.2. TRAUMATIC BRAIN AND SPINAL CORD INJURY

Increased formation of ROS/RNS and oxidative stress are implicated in the pathogenesis of traumatic brain injury. The role of HO-2 in attenuating traumatic brain injury has been studied in HO-2-deficient mice. Gene knockout of HO-2 in mice aggravates early pathogenesis after traumatic injury to the immature brain, as evidenced by markedly increased apoptosis in the hippocampal dentate gyrus and a greater cortical lesion volume in the knockouts relative to wild-type counterparts within the first week post-injury [34]. In a controlled cortical impact traumatic brain injury model of adult mice, targeted disruption of HO-2 is found to exacerbate lipid peroxidation-mediated neuron loss and impaired motor recovery. This suggests that the constitutively expressed neuronal HO-2 may protect neurons against traumatic brain injury by reducing lipid peroxidation via the catabolism of free heme [35]. Studies using heterozygous HO-1-deficient (HO-1^{+/-}) mice also demonstrate that HO-1 can stabilize the blood-spinal cord barrier and limit oxidative stress and white matter damage in acutely injured murine spinal cord [36].

3.1.4.3. NEURODEGENERATIVE DISEASES

There is ample evidence showing a beneficial involvement of HO isozymes, especially HO-1 in neurodegenerative disorders, including Alzheimer's disease and Parkinson's disease. Early studies demonstrated that amyloid precursor proteins bind to HO, inhibit the HO activity, and decrease bilirubin formation in animal models [37]. The amyloid precursor protein-HO interactions are also associated with augmented oxidative neurotoxicity, an important event in Alzheimer's disease. Subsequently, exogenous administration of HO-1 protein has been shown to augment the phagocytosis and clearance of $A\beta$ peptides in rat and mouse microglia. Overexpression of HO-1 in neuronal cells also leads to decreased levels of tau protein [38]. These in vivo and in vitro observations suggest that HO-1 acts as a protective molecule in the pathophysiological process of Alzheimer's disease.

Adenovirus-mediated HO-1 gene transfer has been reported to protect dopaminergic neurons from 1-methyl-4-phenylpyridinium (MPP⁺)-induced neurotoxicity, a commonly used experimental model of parkinsonism [39]. The inflammatory stress in substantia nigra induced by MPP⁺ is diminished by HO-1 overexpression. In contrast, inhibition of HO enzymatic activity by zinc protoporphyrin-IX aggravates MPP⁺-elicited parkinsonism in mice. In addition, overexpression of HO-1 in cultured dopaminergic neurons also renders these cells resistant to MPP⁺-induced neurocytotoxicity [39]. Moreover, HO-1 is found to protect mouse brain from glutamatergic excitotoxicity [40], an important mechanism underlying neurodegenerative disorders, including Parkinson's disease. Together, these in vitro and in vivo data support a beneficial effect of HO-1 in experimental parkinsonism.

3.1.5. Role in Pulmonary Diseases

Substantial evidence suggests a critical role for HO-1 in various pulmonary disorders involving oxidative stress and inflammation [41]. These include hyperoxic lung injury, lung ischemia-reperfusion injury, airway inflammation, and drug-induced pulmonary toxicity.

3.1.5.1. HYPEROXIC LUNG INJURY

Overexpression of HO-1 in lung tissue via intratracheal adenovirus-mediated gene transfer markedly attenuates hyperoxia-induced pulmonary injury, as evidenced by decreased lung inflammation and apoptosis as well as improved survival [42]. In line with the in vivo observations, overexpression of HO-1 in lung epithelial cells protects these cells from oxidative injury. However, targeted disruption of HO-1 in mice is reported to render these animals more resistant to hyperoxia-induced lung oxidative stress and mortality as compared with wild-type animals [43]. Deletion of HO-1 gene is also associated with decreased lung reactive iron and iron-associated proteins. It remains unclear why HO-1 overexpression and gene knockout studies yield opposite conclusions on the role of HO-1 in hyperoxic lung injury. Evidence from in vivo models implicates a potential toxic threshold for HO-1 overexpression related to iron accumulation. For instance, hyperoxia sensitivity is observed at high levels of HO-1 overexpression, whereas hyperoxia resistance is seen at moderate levels of transient overexpression of HO-1 in experimental animals [41].

3.1.5.2. LUNG ISCHEMIA-REPERFUSION INJURY

Lung ischemia-reperfusion injury is the inciting event in acute lung failure following transplantation, surgery, and shock. It is reported that HO-1-null mice are much more suscept-

ible to lung ischemia-reperfusion injury as compared with wild-type littermates. HO-1-null mice exhibit lethal ischemic lung injury, but are rescued from death by inhaled carbon monoxide [44]. Intranasal administration of siRNA targeting HO-1 results in selective knockdown of lung HO-1 and exacerbation of ischemia-reperfusion-induced lung cell apoptosis in mice [45]. Together, these findings demonstrate a beneficial effect of HO-1 in lung ischemiareperfusion injury.

3.1.5.3. AIRWAY INFLAMMATION

The increasing recognition of the anti-inflammatory activity of HO-1 has led to extensive studies on this enzyme in protecting against airway inflammation. Induction of HO-1 by hemin markedly attenuates allergen-induced airway inflammation, oxidative stress, and hyper-reactivity in guinea pigs and mice [46, 47]. Conversely, inhibition of HO activity by metalloporphyrins aggravates the above pathophysiological responses in animal models of airway inflammation. Carbon monoxide, one of the products of HO-catalyzed degradation of heme (Scetion 2.1), is also shown to attenuate airway inflammation and hyperreactivity in experimental animals.

3.1.5.4. DRUG-INDUCED PULMONARY TOXICITY

Lung fibrosis represents one of the most serious adverse effects of drugs, such as bleomycin. Adenovirus-mediated HO-1 gene transfer is shown to protect against bleomycininduced pulmonary fibrosis in a mouse model. The protection is associated with decreased epithelial cell apoptosis and increased interferon- γ production, and is independent of Fas-Fas ligand pathway [48]. Paradoxically, treatment of mice with the HO inhibitor, zincdeuteroporphyrin IX-2,4-bisethylene glycol is reported to attenuate bleomycin-induced lung fibrosis and production of transforming growth factor- β [49]. This observation apparently conflicts with that of the aforementioned HO-1 gene transfer study. As stated above in Section 3.1.1, pharmacological inhibitors of HO may also exert other potential actions, which can confound the effects due to HO inhibition.

3.1.6. Role in Hepatic and Gastrointestinal Diseases

HO-1 plays a role in protecting against various liver and gastrointestinal diseases in animal models. Among them are liver ischemia-reperfusion injury, liver injury from systemic inflammatory response syndrome, drug/xenobiotic-induced hepatotoxicity, nonalcoholic fatty liver disease, and colitis.

3.1.6.1. LIVER INJURY FROM ISCHEMIA-REPERFUSION AND SYSTEMIC INFLAMMATORY RESPONSE SYNDROME

Either gene transfer or chemical induction of HO-1 protects liver from ischemiareperfusion injury, as evidenced by prolonged animal survival, attenuated oxidative and inflammatory stress, and decreased apoptosis of hepatocytes [50, 51]. On the other hand, suppression of HO enzyme activity leads to potentiation of liver ischemia-reperfusion injury and oxidative and inflammatory stress in experimental animals. Adenovirus-mediated gene transfer of HO-1 also attenuates hepatocellular injury and liver dysfunction caused by systemic inflammatory response syndrome, supporting a potent anti-inflammatory activity of HO-1 in liver injury [52].

3.1.6.2. HEPATOTOXICITY AND OXIDATIVE STRESS

The antioxidative and anti-inflammatory activities of HO-1 are also implicated in hepatoprotection against drug/xenobiotic-induced toxicity [53-55]. Overexpression of HO-1 in liver either by gene transfer or chemical induction results in attenuation of hepatotoxicity induced by acetaminophen, carbon tetrachloride, alcohol, and heavy metals. It is noted that hepatotoxicity induced by the above insults involves oxidative and inflammatory stress mechanisms. These findings support an important function of HO-1 in protecting hepatocytes from oxidative and inflammatory stress in experimental animals. Recently, hepatocyte-specific deletion of HO-1 is found to cause oxidative stress in the absence of exogenous stressors, indicating a critical role for HO-1 in maintaining the redox homeostasis of hepatocytes under basal physiological conditions [56].

3.1.6.3. NONALCOHOLIC FATTY LIVER DISEASE AND INFLAMMATORY BOWEL DISEASE

There is evidence for a protective effect of HO-1 in experimental steatohepatitis, a severe form of nonalcoholic fatty liver disease. It is further found that the protection by HO-1 results from decreased production of proinflammatory cytokines and modulation of fatty acid turnover [57]. Studies using various models of murine colitis also demonstrate a crucial role for HO-1 in attenuating colonic inflammation and mucosal injury. In this context, the products derived from HO-mediated catabolism of heme, including bilirubin, carbon monoxide, and biliverdin have all been shown to be protective against experimental murine colitis [58, 59]. Upregulation of HO-1 by pharmacological agents also contributes to the inhibition of experimental colitis in animal models. These findings point to the feasibility of developing therapeutic modalities for inflammatory bowel disease via targeting HO-1 in the gut [60-62].

3.1.7. Role in Renal Diseases

While the basal levels of HO-1 in renal tissue are relatively low, constitutive expression of HO-2 is detectable in various cell populations of kidney. There is increasing experimental evidence for an important cytoprotective role of HO isozymes, especially HO-1 in diverse forms of kidney diseases. These include renal ischemia-reperfusion injury and kidney transplantation, nephrotoxicity induced by drugs/xenobiotics, diabetic nephropathy, kidney injury induced by systemic inflammatory response syndrome, and obstructive kidney disease [63, 64]. Notably, increased renal expression of HO-1 is often observed under the above disease conditions, implicating a possible compensatory mechanism of HO-1 induction in protecting against the disease pathophysiology.

3.1.7.1. RENAL ISCHEMIA-REPERFUSION INJURY AND KIDNEY TRANSPLANTATION

Using genetic mouse models and gene transfer of HO-1, extensive studies have established a critical involvement of HO-1 in protecting against ischemia-reperfusion-induced renal cell death, inflammation, and oxidative stress [64]. Expression of HO-1 or administration of carbon monoxide and biliverdin has also been causally linked to prolonged xenograft survival in animal models of kidney transplantation [65-67].

3.1.7.2. DRUG-INDUCED NEPHROTOXICITY AND DIABETIC NEPHROPATHY

Studies in animal models with either overexpression or suppression of HO-1 demonstrate that HO-1 is a critical cytoprotector against nephrotoxicity induced by cisplatin, an effective anticancer drug whose clinical use is limited by nephrotoxicity [68, 69]. There is also evi-

dence suggesting a protective activity of HO-1 in kidney injury induced by cyclosporine A and radiocontrast agents.

Both HO-1 and HO-2 have been implicated in diabetic nephropathy. It is shown that HO-2 deficiency (HO- $2^{-/-}$) in mice aggravates diabetes-mediated renal dysfunction and oxidative stress, as reflected by increased levels of plasma creatinine, acute tubular damage, microvascular pathology, and augmented superoxide production. Upregulation of HO-1 by administration of cobalt protoporphyrin prevents the above pathophysiological changes, whereas inhibition of HO activity by treatment with tin mesoporphyrin accentuates the renal injury in the HO-2-deficient diabetic mice [27]. These observations indicate an important function for both HO-1 and HO-2 in suppressing diabetic nephropathy and associated oxidative stress in experimental animals.

3.1.7.3. Renal Injury from Systemic Inflammatory Response Syndrome and Ureteral Obstruction

On the one hand, deficiency of HO-1 in mice impairs renal hemodynamics and aggravates systemic inflammatory responses to renal ischemia-reperfusion [70]. On the other hand, it is shown that as compared with wild-type counterparts, HO-1-deficient mice are much more susceptible to renal cell apoptosis, kidney dysfunction, and inflammation caused by systemic inflammatory response syndrome [71]. Both scenarios point to HO-1 as a potent antiinflammatory defense in kidney injury.

Targeted disruption of HO-1 in mice is reported to promote epithelial-mesenchymal transition and renal fibrosis following unilateral ureteral obstruction [72]. HO-1 deficiency also results in increased formation of tubular transforming growth factor- β 1 and augments inflammation in the above mouse model of obstructive kidney disease.

3.1.8. Role in Skin Diseases

Skin is constantly exposed to both endogenous and environmental oxidants and proinflammatory stimuli. Oxidative stress and inflammation are critical mechanisms underlying various human skin disorders, including dermatitis. Multiple studies show that overexpression of HO-1 via chemical induction attenuates contact dermatitis elicited by different sensitizers in animal models, whereas inhibition of HO-1 activity potentiates the dermatitis [73-75]. It is further found that HO-1 inhibits T cell-dependent skin inflammation and the function of antigen-presenting cells [76]. Consistent with a protective effect of HO-1 in skin inflammation and dermatitis, this enzyme is often found to be highly induced in the epidermis during dermatitis in experimental animals.

HO-1 is also remarkably elevated in epidermis after skin wounding in animal models. Inhibition of HO activity by tin protoporphyrin-IX results in retardation of cutaneous wound closure [77]. Healing is also delayed in HO-1-deficient mice, in which lack of HO-1 leads to complete suppression of reepithelialization and thereby the formation of extensive skin lesions, accompanied by impaired neovascularization. Transgenic overexpression of HO-1 selectively in keratinocytes improves the neovascularization and hastens the closure of wounds. It is further demonstrated that induction of HO-1 in wounded skin tissue is relatively weak in diabetic (db/db) mice. Angiogenesis and wound healing are also impaired in these diabetic animals. Local delivery of HO-1 transgene using adenoviral vectors promotes wound healing and increases the vascularization in these diabetic mice [77]. Together, these data strongly support a critical function of HO-1 in skin wound healing and vascularization. The promoting activity of HO-1 in angiogenesis may also have implications for tumor growth and progression, as discussed in the next section.

3.1.9. Role in Cancer

Recently, increasing evidence suggests that HO-1 may play a role in cancer development via influencing multiple processes. These include cancer cell growth, angiogenesis, metastasis, resistance to drug therapy, and chemical carcinogenesis [6].

3.1.9.1. CANCER CELL GROWTH

HO-1 overexpression is frequently observed in various types of cancer, and its expression also correlates with the rate of cancer cell proliferation. In cell cultures, overexpression of HO-1 leads to increased cell proliferation and decreased apoptosis, whereas suppression of HO-1 reverses the proproliferative effect. High levels of HO-1 expression resulting either from chemical induction or genetic manipulation are shown to also promote the growth of various types of tumors in experimental animals, including pancreatic cancer, melanoma, angioma, hepatocellular carcinoma, and prostate cancer [78-82]. Conversely, inhibition of HO-1 decreases the tumor growth in the animal models.

3.1.9.2. ANGIOGENESIS AND CANCER METASTASIS

Angiogenesis, formation of new blood vessels from pre-existing ones, plays a critical role in tumor growth and progression. There is substantial in vitro and in vivo evidence showing that HO-1 possesses potent proangiogenic properties, which may be an important mechanism underlying the tumor-promoting activities of this enzyme [6, 7]. The proangiogenic properties of HO-1 may also contribute to tumor progression and metastasis mediated by HO-1 expression, as angiogenesis is essential not only for tumor growth, but also for metastasis.

3.1.9.3. CANCER DRUG RESISTANCE

The ability of HO-1 to protect cells from various stressors, including cytotoxic compounds implicates that this enzyme may also be involved in tumor cell resistance to chemotherapy. Indeed, chemical inhibition or targeted knockdown of HO-1 increases the sensitivity of tumor cells to anticancer drugs, whereas upregulation of HO-1 protects these cells from anticancer drug-induced cytotoxicity. In line with the in vitro observations, studies in animal models of cancer demonstrate that inhibition of HO-1 improves the survival of cancer-bearing animals after treatment with anticancer drugs. Conversely, overexpression of HO-1 renders the tumor cells more resistant to anticancer drugs, and thereby decreases the effectiveness of chemotherapy [6]. As such, suppression of HO-1 has been suggested as a potential approach to sensitizing cancer cells to chemotherapy [6, 83].

3.1.9.4. CHEMICAL CARCINOGENESIS

While a growing body of evidence indicates a role for HO-1 in promoting tumor growth and progression in a variety of animal models, the involvement of HO-1 in chemical carcinogenesis remains unclear. On the one hand, HO-1 acts as a cytoprotective protein in healthy tissues exposed to harmful stimuli, including chemical carcinogens, and may thus protect against chemical carcinogenesis. On the other hand, the antiapoptotic and proproliferative properties of HO-1 may facilitate tumor promotion and progression during multistage chemical carcinogenesis. The exact role of HO-1 in chemical carcinogenesis awaits further investigations using HO-1 gene knockout and overexpression animal models.

3.1.10. Role in Other Diseases and Conditions

As an important defense against oxidative and inflammatory stress, HO enzymes have also been implicated in protecting against various other disease processes. These include anemia [84], arthritis [85], viral and parasitic infections [86-88], sepsis [89], eye disorders [90], sickle cell disease [91], and reproductive disorders [92].

3.2. Human Studies and Clinical Perspectives

Our understanding of the role of HO enzymes, particularly HO-1 in human health and disease, results from studies in three general areas. They are (1) case reports of HO-1 deficiency, (2) observational studies on HO-1 changes in tissue samples of patients with various disease conditions, and (3) HO-1 gene polymorphism studies in human populations in relation to changes in disease risk.

3.2.1. HO-1 Deficiency

The first human case of HO-1 deficiency due to deletion of exons 2 and 3 was reported by A. Yachie and coworkers in 1999 [93]. This patient exhibited marked growth retardation, severe hemolytic anemia, a low level of serum bilirubin, and tissue iron deposition among other clinical abnormalities. In addition, a lymphoblastoid cell line derived from the patient has been shown to be highly vulnerable to oxidative stress injury as compared with normal human cells. Notably, growth retardation, anemia, iron deposition, and vulnerability to oxidative stress injury found in this patient are all characteristics observed in mice lacking HO-1 (HO-1^{-/-}) [84].

3.2.2. HO-1 Activity in Diseased Tissues

HO-1 activity is often induced in tissue samples from patients with various disorders, including cardiovascular diseases, neurodegeneration, pulmonary diseases, and cancer. However, such observations do not allow us to discriminate whether HO-1 upregulation is a cause, a consequence, or an accompanying phenomenon of the disease pathophysiology. Nevertheless, such studies do provide important clues to the possible involvement of HO-1 in human disease conditions, and necessitate further research to assess the causal role of HO-1 in disease processes.

3.2.3. HO-1 Gene Polymorphisms in Non-cancer Diseases

The involvement of HO-1 in human health and disease has been investigated by analysis of clinical data and comparison of HO-1 allele frequency distribution in healthy people versus patients. Human HO-1 gene promoter is highly polymorphic, and several polymorphisms are shown to modulate the expression of HO-1. HO-1 gene polymorphisms are associated with a wide range of disease conditions involving almost all the major human organs or systems. Although some studies report lack of an association between HO-1 gene polymorphisms and risk of developing certain diseases, overwhelming evidence supports the notion that HO-1

may play an important role in protecting against human diseases that involve mechanisms of oxidative stress and inflammation [94-101]. Studies on HO-1 gene polymorphism may help identify human subpopulations susceptible to developing certain diseases, and may thus provide a rational basis for devising effective strategies for disease intervention in the high risk subgroups.

3.2.4. HO-1 Gene Polymorphisms in Cancer

Gene polymorphism studies have also provided important insights into the role of HO-1 in human cancer development. Multiple reports demonstrate that polymorphisms leading to lower HO-1 expression are linked to higher incidence of cancers, including lung adenocarcinoma and oral squamous cell carcinoma associated with use of tobacco products, as well as gastric adenocarcinoma [102-105]. Conversely, polymorphisms leading to increased HO-1 expression are associated with decreased risk of developing oral squamous cell carcinoma as well as a lower tumor recurrence and better relapse-free survival [104]. However, the role of HO-1 gene polymorphisms in carcinogenesis may depend on the types of cancer. For example, polymorphisms leading to increased HO-1 expression are found to be associated with increased risk of malignant melanoma [106]. This suggests that higher levels of HO-1 expression may facilitate the growth and progression of malignant melanoma, which is in line with the observations made in the mouse melanoma model. Polymorphisms leading to increased HO-1 expression are also associated with increased risk of gastrointestinal stromal tumor as well as higher tumor recurrence rates and shorter survival [107]. Hence, the exact function of HO-1 in human cancer development varies with cancer types. Further elucidation of the precise role of HO-1 in various types of human cancer will facilitate the development of effective strategies targeting HO-1 (either upregulation or downregulation of HO-1 depending on the types of cancer) in cancer therapy.

4. Conclusion and Future Directions

HO enzymes, particular HO-1 are among the most extensively studied antioxidative and anti-inflammatory enzymes in mammalian systems. These enzymes protect against various oxidative stress- and inflammation-associated disease processes in animal models. HO-1 may play a dual role (inhibitory or promoting) in cancer development depending on the types of cancer. In line with the experimental findings, studies in human subjects also point to an important involvement of HO-1 in many disease conditions. Future efforts should focus on further elucidation of the precise role of HO enzymes, including both HO-1 and HO-2 in cancer as well as other common human diseases, and addressing how to translate the current discoveries on HO enzymes into new therapeutic and/or preventive modalities to combat human diseases.

5. References

[1] Tenhunen R, Marver HS, Schmid R. The enzymatic conversion of heme to bilirubin by microsomal heme oxygenase. *Proc. Natl. Acad. Sci. USA* 1968;61:748-755.

- [2] Tenhunen R, Marver HS, Schmid R. Microsomal heme oxygenase: characterization of the enzyme. *J. Biol. Chem.* 1969;244:6388-6394.
- [3] Ryter SW, Alam J, Choi AM. Heme oxygenase-1/carbon monoxide: from basic science to therapeutic applications. *Physiol. Rev.* 2006;86:583-650.
- [4] Abraham NG, Kappas A. Pharmacological and clinical aspects of heme oxygenase. *Pharmacol. Rev.* 2008;60:79-127.
- [5] Kim HP, Ryter SW, Choi AM. CO as a cellular signaling molecule. *Annu. Rev. Pharmacol. Toxicol.* 2006;46:411-449.
- [6] Jozkowicz A, Was H, Dulak J. Heme oxygenase-1 in tumors: is it a false friend? *Antioxid. Redox Signal* 2007;9:2099-2117.
- [7] Dulak J, Deshane J, Jozkowicz A, Agarwal A. Heme oxygenase-1 and carbon monoxide in vascular pathobiology: focus on angiogenesis. *Circulation* 2008;117:231-241.
- [8] Maines MD, Gibbs PE. 30 some years of heme oxygenase: from a "molecular wrecking ball" to a "mesmerizing" trigger of cellular events. *Biochem. Biophys. Res. Commun.* 2005;338:568-577.
- [9] Lin Q, Weis S, Yang G, Weng YH, Helston R, Rish K, *et al.* Heme oxygenase-1 protein localizes to the nucleus and activates transcription factors important in oxidative stress. *J. Biol. Chem.* 2007;282:20621-20633.
- [10] Weng YH, Yang G, Weiss S, Dennery PA. Interaction between heme oxygenase-1 and -2 proteins. J. Biol. Chem. 2003;278:50999-51005.
- [11] Ding Y, Zhang YZ, Furuyama K, Ogawa K, Igarashi K, Shibahara S. Down-regulation of heme oxygenase-2 is associated with the increased expression of heme oxygenase-1 in human cell lines. *FEBS J.* 2006;273:5333-5346.
- [12] Chen YH, Yet SF, Perrella MA. Role of heme oxygenase-1 in the regulation of blood pressure and cardiac function. *Exp. Biol. Med.* (Maywood) 2003;228:447-453.
- [13] Hoekstra KA, Godin DV, Cheng KM. Protective role of heme oxygenase in the blood vessel wall during atherogenesis. *Biochem. Cell Biol.* 2004;82:351-359.
- [14] Stocker R, Perrella MA. Heme oxygenase-1: a novel drug target for atherosclerotic diseases? *Circulation* 2006;114:2178-2189.
- [15] Idriss NK, Blann AD, Lip GY. Hemoxygenase-1 in cardiovascular disease. J. Am. Coll. Cardiol. 2008;52:971-978.
- [16] Peterson SJ, Frishman WH, Abraham NG. Targeting heme oxygenase: therapeutic implications for diseases of the cardiovascular system. *Cardiol. Rev.* 2009;17:99-111.
- [17] Bellner L, Martinelli L, Halilovic A, Patil K, Puri N, Dunn MW, et al. Heme oxygenase-2 deletion causes endothelial cell activation marked by oxidative stress, inflammation, and angiogenesis. J. Pharmacol. Exp. Ther. 2009;331:925-932.
- [18] Gagov H, Kadinov B, Hristov K, Boev K, Itzev D, Bolton T, et al. Role of constitutively expressed heme oxygenase-2 in the regulation of guinea pig coronary artery tone. *Pflugers Arch.* 2003;446:412-421.
- [19] Sodhi K, Inoue K, Gotlinger KH, Canestraro M, Vanella L, Kim DH, et al. Epoxyeicosatrienoic acid agonist rescues the metabolic syndrome phenotype of HO-2null mice. J. Pharmacol. Exp. Ther. 2009;331:906-916.
- [20] Stec DE, Vera T, Storm MV, McLemore GR, Jr., Ryan MJ. Blood pressure and renal blow flow responses in heme oxygenase-2 knockout mice. Am. J. Physiol. Regul. Integr. Comp. Physiol. 2009;297:R1822-1828.
- [21] Han F, Takeda K, Ono M, Date F, Ishikawa K, Yokoyama S, et al. Hypoxemia induces expression of heme oxygenase-1 and heme oxygenase-2 proteins in the mouse myocardium. J. Biochem. 2010;147:143-151.
- [22] Ndisang JF, Jadhav A. Heme oxygenase system enhances insulin sensitivity and glucose metabolism in streptozotocin-induced diabetes. Am. J. Physiol. Endocrinol. Metab. 2009;296:E829-841.
- [23] Hu CM, Lin HH, Chiang MT, Chang PF, Chau LY. Systemic expression of heme oxygenase-1 ameliorates type 1 diabetes in NOD mice. *Diabetes* 2007;56:1240-1247.
- [24] Ndisang JF, Lane N, Jadhav A. Upregulation of the heme oxygenase system ameliorates postprandial and fasting hyperglycemia in type 2 diabetes. *Am. J. Physiol. Endocrinol. Metab.* 2009;296:E1029-1041.
- [25] Liu X, Wei J, Peng DH, Layne MD, Yet SF. Absence of heme oxygenase-1 exacerbates myocardial ischemia/reperfusion injury in diabetic mice. *Diabetes* 2005; 54:778-784.
- [26] Tobiasch E, Gunther L, Bach FH. Heme oxygenase-1 protects pancreatic beta cells from apoptosis caused by various stimuli. *J. Investig. Med.* 2001;49:566-571.
- [27] Goodman AI, Chander PN, Rezzani R, Schwartzman ML, Regan RF, Rodella L, et al. Heme oxygenase-2 deficiency contributes to diabetes-mediated increase in superoxide anion and renal dysfunction. J. Am. Soc. Nephrol. 2006;17:1073-1081.
- [28] Goto S, Sampei K, Alkayed NJ, Dore S, Koehler RC. Characterization of a new double-filament model of focal cerebral ischemia in heme oxygenase-2-deficient mice. *Am. J. Physiol. Regul. Integr. Comp. Physiol.* 2003;285:R222-230.
- [29] Dore S, Goto S, Sampei K, Blackshaw S, Hester LD, Ingi T, et al. Heme oxygenase-2 acts to prevent neuronal death in brain cultures and following transient cerebral ischemia. Neuroscience 2000;99:587-592.
- [30] Panahian N, Yoshiura M, Maines MD. Overexpression of heme oxygenase-1 is neuroprotective in a model of permanent middle cerebral artery occlusion in transgenic mice. J. Neurochem. 1999;72:1187-1203.
- [31] Wang J, Zhuang H, Dore S. Heme oxygenase 2 is neuroprotective against intracerebral hemorrhage. *Neurobiol. Dis.* 2006;22:473-476.
- [32] Qu Y, Chen-Roetling J, Benvenisti-Zarom L, Regan RF. Attenuation of oxidative injury after induction of experimental intracerebral hemorrhage in heme oxygenase-2 knockout mice. J. Neurosurg. 2007;106:428-435.
- [33] Wang J, Dore S. Heme oxygenase-1 exacerbates early brain injury after intracerebral haemorrhage. *Brain* 2007;130:1643-1652.
- [34] Yoneyama-Sarnecky T, Olivas AD, Azari S, Ferriero DM, Manvelyan HM, Noble-Haeusslein LJ. Heme oxygenase-2 modulates early pathogenesis after traumatic injury to the immature brain. *Dev. Neurosci.* 2010;32:81-90.
- [35] Chang EF, Wong RJ, Vreman HJ, Igarashi T, Galo E, Sharp FR, et al. Heme oxygenase-2 protects against lipid peroxidation-mediated cell loss and impaired motor recovery after traumatic brain injury. J. Neurosci. 2003;23:3689-3696.
- [36] Lin Y, Vreman HJ, Wong RJ, Tjoa T, Yamauchi T, Noble-Haeusslein LJ. Heme oxygenase-1 stabilizes the blood-spinal cord barrier and limits oxidative stress and white matter damage in the acutely injured murine spinal cord. *J. Cereb. Blood Flow Metab.* 2007;27:1010-1021.
- [37] Takahashi M, Dore S, Ferris CD, Tomita T, Sawa A, Wolosker H, et al. Amyloid

precursor proteins inhibit heme oxygenase activity and augment neurotoxicity in Alzheimer's disease. *Neuron* 2000;28:461-473.

- [38] Takeda A, Perry G, Abraham NG, Dwyer BE, Kutty RK, Laitinen JT, *et al.* Overexpression of heme oxygenase in neuronal cells, the possible interaction with Tau. *J. Biol. Chem.* 2000;275:5395-5399.
- [39] Hung SY, Liou HC, Kang KH, Wu RM, Wen CC, Fu WM. Overexpression of heme oxygenase-1 protects dopaminergic neurons against 1-methyl-4-phenylpyridiniuminduced neurotoxicity. *Mol. Pharmacol.* 2008;74:1564-1575.
- [40] Ahmad AS, Zhuang H, Dore S. Heme oxygenase-1 protects brain from acute excitotoxicity. *Neuroscience* 2006;141:1703-1708.
- [41] Morse D, Lin L, Choi AM, Ryter SW. Heme oxygenase-1, a critical arbitrator of cell death pathways in lung injury and disease. *Free Radic. Biol. Med.* 2009;47:1-12.
- [42] Otterbein LE, Kolls JK, Mantell LL, Cook JL, Alam J, Choi AM. Exogenous administration of heme oxygenase-1 by gene transfer provides protection against hyperoxia-induced lung injury. J. Clin. Invest. 1999;103:1047-1054.
- [43] Dennery PA, Visner G, Weng YH, Nguyen X, Lu F, Zander D, *et al.* Resistance to hyperoxia with heme oxygenase-1 disruption: role of iron. *Free Radic. Biol. Med.* 2003;34:124-133.
- [44] Fujita T, Toda K, Karimova A, Yan SF, Naka Y, Yet SF, et al. Paradoxical rescue from ischemic lung injury by inhaled carbon monoxide driven by derepression of fibrinolysis. Nat. Med. 2001;7:598-604.
- [45] Zhang X, Shan P, Jiang D, Noble PW, Abraham NG, Kappas A, et al. Small interfering RNA targeting heme oxygenase-1 enhances ischemia-reperfusion-induced lung apoptosis. J. Biol. Chem. 2004;279:10677-10684.
- [46] Almolki A, Taille C, Martin GF, Jose PJ, Zedda C, Conti M, et al. Heme oxygenase attenuates allergen-induced airway inflammation and hyperreactivity in guinea pigs. *Am. J. Physiol Lung Cell Mol. Physiol.* 2004;287:L26-34.
- [47] Xia ZW, Zhong WW, Xu LQ, Sun JL, Shen QX, Wang JG, et al. Heme oxygenase-1mediated CD4⁺CD25^{high} regulatory T cells suppress allergic airway inflammation. J. Immunol. 2006;177:5936-5945.
- [48] Tsuburai T, Suzuki M, Nagashima Y, Suzuki S, Inoue S, Hasiba T, *et al.* Adenovirusmediated transfer and overexpression of heme oxygenase 1 cDNA in lung prevents bleomycin-induced pulmonary fibrosis via a Fas-Fas ligand-independent pathway. *Hum. Gene Ther.* 2002;13:1945-1960.
- [49] Atzori L, Chua F, Dunsmore SE, Willis D, Barbarisi M, McAnulty RJ, *et al.* Attenuation of bleomycin induced pulmonary fibrosis in mice using the heme oxygenase inhibitor Zn-deuteroporphyrin IX-2,4-bisethylene glycol. *Thorax* 2004;59: 217-223.
- [50] Amersi F, Buelow R, Kato H, Ke B, Coito AJ, Shen XD, *et al.* Upregulation of heme oxygenase-1 protects genetically fat Zucker rat livers from ischemia/reperfusion injury. *J. Clin. Invest.* 1999;104:1631-1639.
- [51] Coito AJ, Buelow R, Shen XD, Amersi F, Moore C, Volk HD, *et al.* Heme oxygenase-1 gene transfer inhibits inducible nitric oxide synthase expression and protects genetically fat Zucker rat livers from ischemia-reperfusion injury. *Transplantation* 2002;74:96-102.
- [52] McCarter SD, Badhwar A, Scott JR, Akyea TG, Bihari A, Dungey AA, et al. Remote

liver injury is attenuated by adenovirus-mediated gene transfer of heme oxygenase-1 during the systemic inflammatory response syndrome. *Microcirculation* 2004;11:587-595.

- [53] Farombi EO, Surh YJ. Heme oxygenase-1 as a potential therapeutic target for hepatoprotection. J. Biochem. Mol. Biol. 2006;39:479-491.
- [54] Tsui TY, Lau CK, Ma J, Glockzin G, Obed A, Schlitt HJ, et al. Adeno-associated virus-mediated heme oxygenase-1 gene transfer suppresses the progression of micronodular cirrhosis in rats. World J. Gastroenterol. 2006;12:2016-2023.
- [55] Mandal P, Pritchard MT, Nagy LE. Anti-inflammatory pathways and alcoholic liver disease: role of an adiponectin/interleukin-10/heme oxygenase-1 pathway. World. J. Gastroenterol. 2010;16:1330-1336.
- [56] Mamiya T, Katsuoka F, Hirayama A, Nakajima O, Kobayashi A, Maher JM, *et al.* Hepatocyte-specific deletion of heme oxygenase-1 disrupts redox homeostasis in basal and oxidative environments. *Tohoku. J. Exp. Med.* 2008;216:331-339.
- [57] Yu J, Chu ES, Wang R, Wang S, Wu CW, Wong VW, et al. Heme oxygenase-1 protects against steatohepatitis in both cultured hepatocytes and mice. *Gastro*enterology 2010;138:694-704, 704 e691.
- [58] Hegazi RA, Rao KN, Mayle A, Sepulveda AR, Otterbein LE, Plevy SE. Carbon monoxide ameliorates chronic murine colitis through a heme oxygenase 1-dependent pathway. J. Exp. Med. 2005;202:1703-1713.
- [59] Berberat PO, YI AR, Yamashita K, Warny MM, Csizmadia E, Robson SC, et al. Heme oxygenase-1-generated biliverdin ameliorates experimental murine colitis. *Inflamm. Bowel Dis.* 2005;11:350-359.
- [60] Erbil Y, Giris M, Abbasoglu SD, Barbaros U, Yanik BT, Necefli A, et al. Effect of heme oxygenase-1 induction by octreotide on TNBS-induced colitis. J. Gastroenterol. Hepatol. 2007;22:1852-1858.
- [61] Horvath K, Varga C, Berko A, Posa A, Laszlo F, Whittle BJ. The involvement of heme oxygenase-1 activity in the therapeutic actions of 5-aminosalicylic acid in rat colitis. *Eur. J. Pharmacol.* 2008;581:315-323.
- [62] Jun CD, Kim Y, Choi EY, Kim M, Park B, Youn B, *et al.* Gliotoxin reduces the severity of trinitrobenzene sulfonic acid-induced colitis in mice: evidence of the connection between heme oxygenase-1 and the nuclear factor-kappaB pathway in vitro and in vivo. *Inflamm. Bowel Dis.* 2006;12:619-629.
- [63] Abraham NG, Cao J, Sacerdoti D, Li X, Drummond G. Heme oxygenase: the key to renal function regulation. *Am. J. Physiol. Renal. Physiol.* 2009;297:F1137-1152.
- [64] Jarmi T, Agarwal A. Heme oxygenase and renal disease. *Curr. Hypertens. Rep.* 2009; 11:56-62.
- [65] Tullius SG, Nieminen-Kelha M, Bachmann U, Reutzel-Selke A, Jonas S, Pratschke J, *et al.* Induction of heme-oxygenase-1 prevents ischemia/reperfusion injury and improves long-term graft outcome in rat renal allografts. *Transplant. Proc.* 2001;33: 1286-1287.
- [66] Wagner M, Cadetg P, Ruf R, Mazzucchelli L, Ferrari P, Redaelli CA. Heme oxygenase-1 attenuates ischemia/reperfusion-induced apoptosis and improves survival in rat renal allografts. *Kidney Int.* 2003;63:1564-1573.
- [67] Nakao A, Neto JS, Kanno S, Stolz DB, Kimizuka K, Liu F, *et al.* Protection against ischemia/reperfusion injury in cardiac and renal transplantation with carbon monoxide,

biliverdin and both. Am. J. Transplant. 2005;5:282-291.

- [68] Agarwal A, Balla J, Alam J, Croatt AJ, Nath KA. Induction of heme oxygenase in toxic renal injury: a protective role in cisplatin nephrotoxicity in the rat. *Kidney Int.* 1995;48:1298-1307.
- [69] Shiraishi F, Curtis LM, Truong L, Poss K, Visner GA, Madsen K, et al. Heme oxygenase-1 gene ablation or expression modulates cisplatin-induced renal tubular apoptosis. Am. J. Physiol. Renal. Physiol. 2000;278:F726-736.
- [70] Tracz MJ, Juncos JP, Croatt AJ, Ackerman AW, Grande JP, Knutson KL, et al. Deficiency of heme oxygenase-1 impairs renal hemodynamics and exaggerates systemic inflammatory responses to renal ischemia. *Kidney Int.* 2007;72:1073-1080.
- [71] Tracz MJ, Juncos JP, Grande JP, Croatt AJ, Ackerman AW, Rajagopalan G, *et al.* Renal hemodynamic, inflammatory, and apoptotic responses to lipopolysaccharide in HO-1^{-/-} mice. *Am. J. Pathol.* 2007;170:1820-1830.
- [72] Kie JH, Kapturczak MH, Traylor A, Agarwal A, Hill-Kapturczak N. Heme oxygenase-1 deficiency promotes epithelial-mesenchymal transition and renal fibrosis. J. Am. Soc. Nephrol. 2008;19:1681-1691.
- [73] Majewska M, Zajac K, Dulak J, Szczepanik M. Heme oxygenase (HO-1) is involved in the negative regulation of contact sensitivity reaction. *Pharmacol. Rep.* 2008;60:933-940.
- [74] Kirino M, Kirino Y, Takeno M, Nagashima Y, Takahashi K, Kobayashi M, et al. Heme oxygenase 1 attenuates the development of atopic dermatitis-like lesions in mice: implications for human disease. J. Allergy Clin. Immunol. 2008;122:290-297, 297 e291-298.
- [75] Pae HO, Ae Ha Y, Chai KY, Chung HT. Heme oxygenase-1 attenuates contact hypersensitivity induced by 2,4-dinitrofluorobenzene in mice. *Immunopharmacol. Immunotoxicol.* 2008;30:207-216.
- [76] Listopad J, Asadullah K, Sievers C, Ritter T, Meisel C, Sabat R, et al. Heme oxygenase-1 inhibits T cell-dependent skin inflammation and differentiation and function of antigen-presenting cells. Exp. Dermatol. 2007;16:661-670.
- [77] Grochot-Przeczek A, Lach R, Mis J, Skrzypek K, Gozdecka M, Sroczynska P, et al. Heme oxygenase-1 accelerates cutaneous wound healing in mice. PLoS One 2009;4: e5803.
- [78] Sunamura M, Duda DG, Ghattas MH, Lozonschi L, Motoi F, Yamauchi J, et al. Heme oxygenase-1 accelerates tumor angiogenesis of human pancreatic cancer. Angiogenesis 2003;6:15-24.
- [79] Was H, Cichon T, Smolarczyk R, Rudnicka D, Stopa M, Chevalier C, et al. Overexpression of heme oxygenase-1 in murine melanoma: increased proliferation and viability of tumor cells, decreased survival of mice. Am. J. Pathol. 2006;169:2181-2198.
- [80] Marinissen MJ, Tanos T, Bolos M, de Sagarra MR, Coso OA, Cuadrado A. Inhibition of heme oxygenase-1 interferes with the transforming activity of the Kaposi sarcoma herpesvirus-encoded G protein-coupled receptor. *J. Biol. Chem.* 2006;281:11332-11346.
- [81] Alaoui-Jamali MA, Bismar TA, Gupta A, Szarek WA, Su J, Song W, et al. A novel experimental heme oxygenase-1-targeted therapy for hormone-refractory prostate cancer. Cancer Res. 2009;69:8017-8024.

- [82] Sass G, Leukel P, Schmitz V, Raskopf E, Ocker M, Neureiter D, et al. Inhibition of heme oxygenase 1 expression by small interfering RNA decreases orthotopic tumor growth in livers of mice. Int. J. Cancer 2008;123:1269-1277.
- [83] Gleixner KV, Mayerhofer M, Vales A, Gruze A, Hormann G, Cerny-Reiterer S, et al. Targeting of Hsp32 in solid tumors and leukemias: a novel approach to optimize anticancer therapy. Curr. Cancer Drug. Targets 2009;9:675-689.
- [84] Poss KD, Tonegawa S. Heme oxygenase 1 is required for mammalian iron reutilization. *Proc. Natl. Acad. Sci. USA* 1997;94:10919-10924.
- [85] Benallaoua M, Francois M, Batteux F, Thelier N, Shyy JY, Fitting C, et al. Pharmacologic induction of heme oxygenase 1 reduces acute inflammatory arthritis in mice. Arthritis Rheum. 2007;56:2585-2594.
- [86] Zhu Z, Wilson AT, Mathahs MM, Wen F, Brown KE, Luxon BA, et al. Heme oxygenase-1 suppresses hepatitis C virus replication and increases resistance of hepatocytes to oxidant injury. *Hepatology* 2008;48:1430-1439.
- [87] Pamplona A, Ferreira A, Balla J, Jeney V, Balla G, Epiphanio S, *et al.* Heme oxygenase-1 and carbon monoxide suppress the pathogenesis of experimental cerebral malaria. *Nat. Med.* 2007;13:703-710.
- [88] Seixas E, Gozzelino R, Chora A, Ferreira A, Silva G, Larsen R, et al. Heme oxygenase-1 affords protection against noncerebral forms of severe malaria. Proc. Natl. Acad. Sci. USA 2009;106:15837-15842.
- [89] Takamiya R, Hung CC, Hall SR, Fukunaga K, Nagaishi T, Maeno T, et al. Highmobility group box 1 contributes to lethality of endotoxemia in heme oxygenase-1deficient mice. Am. J. Respir. Cell Mol. Biol. 2009;41:129-135.
- [90] Bellner L, Vitto M, Patil KA, Dunn MW, Regan R, Laniado-Schwartzman M. Exacerbated corneal inflammation and neovascularization in the HO-2 null mice is ameliorated by biliverdin. *Exp. Eye Res.* 2008;87:268-278.
- [91] Belcher JD, Vineyard JV, Bruzzone CM, Chen C, Beckman JD, Nguyen J, et al. Heme oxygenase-1 gene delivery by Sleeping Beauty inhibits vascular stasis in a murine model of sickle cell disease. J. Mol. Med. 2010;88:665-675.
- [92] Burnett AL, Johns DG, Kriegsfeld LJ, Klein SL, Calvin DC, Demas GE, et al. Ejaculatory abnormalities in mice with targeted disruption of the gene for heme oxygenase-2. Nat. Med. 1998;4:84-87.
- [93] Yachie A, Niida Y, Wada T, Igarashi N, Kaneda H, Toma T, et al. Oxidative stress causes enhanced endothelial cell injury in human heme oxygenase-1 deficiency. J. Clin. Invest. 1999;103:129-135.
- [94] Exner M, Minar E, Wagner O, Schillinger M. The role of heme oxygenase-1 promoter polymorphisms in human disease. *Free Radic. Biol. Med.* 2004;37:1097-1104.
- [95] Yun L, Xiaoli L, Qi Z, Laiyuan W, Xiangfeng L, Chong S, et al. Association of an intronic variant of the heme oxygenase-1 gene with hypertension in northern Chinese Han population. *Clin. Exp. Hypertens.* 2009;31:534-543.
- [96] Song F, Li X, Zhang M, Yao P, Yang N, Sun X, et al. Association between heme oxygenase-1 gene promoter polymorphisms and type 2 diabetes in a Chinese population. Am. J. Epidemiol. 2009;170:747-756.
- [97] Sheu CC, Zhai R, Wang Z, Gong MN, Tejera P, Chen F, *et al.* Heme oxygenase-1 microsatellite polymorphism and haplotypes are associated with the development of acute respiratory distress syndrome. *Intensive Care Med.* 2009;35:1343-1351.

- [98] Exner M, Bohmig GA, Schillinger M, Regele H, Watschinger B, Horl WH, et al. Donor heme oxygenase-1 genotype is associated with renal allograft function. *Transplantation* 2004;77:538-542.
- [99] Buis CI, van der Steege G, Visser DS, Nolte IM, Hepkema BG, Nijsten M, *et al.* Heme oxygenase-1 genotype of the donor is associated with graft survival after liver transplantation. *Am. J. Transplant.* 2008;8:377-385.
- [100] Rueda B, Oliver J, Robledo G, Lopez-Nevot MA, Balsa A, Pascual-Salcedo D, et al. HO-1 promoter polymorphism associated with rheumatoid arthritis. Arthritis Rheum. 2007;56:3953-3958.
- [101] Infante J, Garcia-Gorostiaga I, Sanchez-Juan P, Sierra M, Martin-Gurpegui JL, Terrazas J, et al. Synergistic effect of two oxidative stress-related genes (heme oxygenase-1 and GSK3beta) on the risk of Parkinson's disease. Eur. J. Neurol. 2010; 17:760-762.
- [102] Kikuchi A, Yamaya M, Suzuki S, Yasuda H, Kubo H, Nakayama K, et al. Association of susceptibility to the development of lung adenocarcinoma with the heme oxygenase-1 gene promoter polymorphism. *Hum. Genet.* 2005;116:354-360.
- [103] Chang KW, Lee TC, Yeh WI, Chung MY, Liu CJ, Chi LY, et al. Polymorphism in heme oxygenase-1 (HO-1) promoter is related to the risk of oral squamous cell carcinoma occurring on male areca chewers. Br. J. Cancer 2004;91:1551-1555.
- [104] Vashist YK, Blessmann M, Trump F, Kalinin V, Kutup A, Schneider C, et al. Microsatellite GTn-repeat polymorphism in the promoter of heme oxygenase-1 gene is an independent predictor of tumor recurrence in male oral squamous cell carcinoma patients. J. Oral. Pathol. Med. 2008;37:480-484.
- [105] Lo SS, Lin SC, Wu CW, Chen JH, Yeh WI, Chung MY, et al. Heme oxygenase-1 gene promoter polymorphism is associated with risk of gastric adenocarcinoma and lymphovascular tumor invasion. Ann. Surg. Oncol. 2007;14:2250-2256.
- [106] Okamoto I, Krogler J, Endler G, Kaufmann S, Mustafa S, Exner M, *et al.* A microsatellite polymorphism in the heme oxygenase-1 gene promoter is associated with risk for melanoma. *Int. J. Cancer* 2006;119:1312-1315.
- [107] Vashist YK, Uzunoglu G, Cataldegirmen G, Kalinin V, Schurr P, Koenig AM, et al. Haeme oxygenase-1 promoter polymorphism is an independent prognostic marker of gastrointestinal stromal tumour. *Histopathology* 2009;54:303-308.

Chapter 16

NAD(P)H:Quinone Oxidoreductase

Abstract

NAD(P)H:quinone oxidoreductase (NQO) refers to a family of flavoproteins that catalyze the two-electron reduction of quinones and derivatives. In mammas, there are two members, namely, NQO1 and NQO2. NQO1 uses NAD(P)H whereas NQO2 employs dihydronicotinamide riboside (NRH) as the electron donors. In addition to the well-recognized function in metabolizing quinones and preventing formation of reactive oxygen species, NQO enzymes also possess other important activities. These include anti-inflammatory effects, direct scavenging of superoxide, stabilization of p53 and other tumor suppressors, and acting as a melatonin-binding site to mediate the biological functions of this neurohormone. Studies in animal models have demonstrated an important role for NQO enzymes in protecting against various pathophysiological processes, including chemical toxicity, carcinogenesis, and cardiovascular injury. Functional gene polymorphisms are identified in both human NQO1 and NQO2 genes. Studies on the association between NQO gene polymorphisms and susceptibility to disease development suggest a beneficial involvement of NQO enzymes in a number of human disorders, especially cancer.

1. Overview

NAD(P)H:quinone oxidoreductase (NQO) refers to a family of flavoproteins that include two members, namely, NQO1 and NQO2 in mammals. NQO1 and NQO2 strand for NAD(P)H:quinone oxidoreductase 1 and NRH:quinone oxidoreductase 2, respectively. NQO2 uses dihydronicotinamide riboside (NRH) rather than NAD(P)H as the electron donor. Hence, using the general term NAD(P)H:quinone oxidoreductase to refer to NQO2 is not accurate. NQO1 was discovered in rat liver and named DT-diaphorase by L. Ernster and F. Navazio in 1958. NQO1 and NQO2 catalyze two-electron reduction of various quinones and derivatives. As described below, these two enzymes also possess other novel biological activities. NQO1 and NQO2 are cytosolic enzymes with a molecular mass of 31 and 25 kDa, respectively. They are widely distributed in mammalian tissues though the highest levels are usually found in liver. In humans, NQO1 and NQO2 are localized on chromosomes 16q22.1 and 6pter-q12, respectively.

2. Biochemistry and Molecular Regulation

2.1. Biochemistry

The biochemical properties of NQO enzymes are summarized as follows.

- Two-electron reduction and detoxification of quinone compounds and derivatives
- Maintenance of α -tocopherol and ubiquinone in their reduced and active forms
- Direct scavenging of superoxide
- Stabilization of the tumor suppressor p53 protein and other tumor suppressors
- As a gatekeeper of the 20S proteasome
- Binding to melatonin

2.1.1. Two-Electron Reduction of Quinones and Derivatives

Both NQO1 and NQO2 catalyze two-electron reduction of quinones to hydroquinones, which then undergo conjugation reactions, leading to their excretion from the body. This twoelectron reduction reaction prevents the unwanted one-electron reduction of quinones by other enzymes, such as cytochrome P450 reductase. The one-electron reduction results in the formation of reactive semiquinone radicals, and the subsequent formation of reactive oxygen species (ROS) via redox cycling (Figure 16-1). Two-electron reduction of quinones by NQO enzymes thus prevents the formation of reactive semiquinone radicals and ROS. In addition to quinones, NQO enzymes also catalyze two-electron reduction of quinoneimines, nitroaromatics, and azo dyes.



Figure 16-1. NAD(P)H:quinone oxidoreductase (NQO) enzyme-catalyzed two-electron reduction of quinones. As illustrated, NQO enzymes catalyze two-electron reduction of quinones to hydroquinones. This prevents quinones from undergoing one-electron reduction catalyzed by enzymes, such as cytochrome P450 reductase (PR). One-electron reduction of quinones leads to the formation of semiquinone radicals, which reduce oxygen to superoxide (O_2^-), leading to the formation of other reactive oxygen species, such as hydrogen peroxide (H_2O_2) and hydroxyl radical (OH).

2.1.2. Maintenance of Endogenous Antioxidants α -Tocopherol and Ubiquinone

NQO1 is thought to help maintain certain endogenous antioxidants in their reduced and active forms. Both ubiquinone (coenzyme Q) and α -tocopherol-quinone, two important lipid-soluble antioxidants are substrates for NQO1 in vitro. NQO1 catalyzes two-electron reduction of ubiquinone and α -tocopherol-quinone to their hydroquinone forms, which then protect against lipid peroxidation of membranes.

2.1.3. Scavenging of Superoxide

It is well-known that both NQO1 and NQO2 prevent ROS formation from quinone redox cycling. In addition, NQO1 also directly scavenges superoxide in an NAD(P)H-dependent manner [1]. This effect may be particularly significant in tissues, such as vasculature, where NQO1 is highly expressed [2].

2.1.4. Stabilization of p53 and Other Tumor Suppressors

The p53 tumor suppressor protein suppresses tumorigenesis by mediating either growth arrest or apoptosis in response to stresses, such as DNA damage. NQO1 is shown to play an important role in stabilizing p53 protein via processes including protein-protein interaction that prevents ubiquitin-independent degradation of p53 by 20S proteasome [3, 4]. Stabilization of p53 protein by NQO1 may contribute to the protective role of NQO1 in carcinogenesis (see Sections 3.1.4 and 3.2.2 for more discussion). In addition to p53, NQO1 also regulates the ubiquitin-independent 20S proteasomal degradation of p73 α and p33, two other tumor suppressors [5, 6].

2.1.5. As a Gatekeeper of the 20S Proteasome

It has been demonstrated that in murine liver tissue majority of NQO1 is associated with the 20S proteasome, and NQO1 may function as a gatekeeper for protein degradation through the 20S proteasome. More recently, NQO1 is found to protect eukaryotic translation initiation factor (eIF) 4GI from proteasomal degradation, and as such may participate in the regulation of translation [7].

2.1.6. Binding to Melatonin

NQO2 has a melatonin-binding site, and as such may mediate some of the biological activities of melatonin, such as the antioxidant effects [8, 9]. However, the exact biological significance of the binding of melatonin to NQO2 remains to be elucidated.

2.2. Molecular Regulation

NQO1 is highly inducible by a wide variety of chemical inducers, including oxidants, electrophiles, and phenolic compounds. It is well-recognized that Keap1/Nrf2/ARE pathway plays a central role in regulating the gene expression of mammalian NQO1 [10, 11]. Here ARE refers to antioxidant response element. Indeed, there are multiple AREs in the promoter region of mammalian NQO1 gene. Mammalian NQO1 gene promoter also contains a functional xenobiotic response element (XRE), and arylhydrocarbon receptor (AhR) via binding

to XRE is crucially involved in the induction of NQO1 gene expression by xenobiotics, such as 2,3,7,8-tetrachlorodibenzo-*p*-dioxin (TCDD). There is also evidence suggesting that Nrf2/ARE and AhR/XRE coordinate in the regulation of NQO1 expression in mammalian tissues [10].

NQO2 expression is also responsive to a variety of xenobiotics, and NQO2 gene expression is coordinately induced with NQO1 in response to xenobiotics and antioxidants. Nucleotide sequence analysis of mammalian NQO2 gene promoter reveals the presence of several cis-elements, including the binding sites for SP1, SP3, XRE, and ARE. It has been demonstrated that SP1 activates whereas SP3 represses the promoter activity of NQO2 gene [12]. A recent study shows that human NQO2 gene expression is also upregulated by antioxidants via an Nrf2-dependent mechanism [13].

3. Role in Health and Disease

3.1. Animal Studies

3.1.1. Experimental Approaches

Gene knockout mouse models have been created for NQO1 and NQO2. Mice deficient in either NQO1 or NQO2 are born and reproduced adeptly as the wild-type mice, indicating that neither gene is essential for murine embryonic development. In addition, NQO1 and NQO2-double-knockout mouse model has been generated, and the mice lacking both NQO1 and NQO2 are born and reproduced normally. Much of our knowledge on the in vivo biological functions of NQO enzymes results from studies using these knockout models.

In addition to the genetic models, several chemicals have been shown to potently inhibit NQO enzyme activity, and as such serve as useful tools for studying the biochemical activities of NQO enzymes in various cellular systems. For example, dicumarol at micromolar concentrations is a commonly used competitive inhibitor of NQO1 in biological systems. However, at concentrations that inhibit NQO1 activity, dicumarol may also interfere with other cellular enzymes, including the enzyme complexes of the mitochondrial electron transport chain [14]. Recently, a more selective and potent NQO1 inhibitor, namely, ES936 has been synthesized and characterized by D. Ross and coworkers [15]. ES936 is a quinone compound and a mechanism-based inhibitor of NOO1 in cellular systems, in which over 90% NQO1 activity can be inhibited by this chemical at 100 nM. More recently, a range of triazoloacridin-6-ones have been synthesized and evaluated for ability to inhibit NQO1 and NQO2, and one of them, NSC66041, is identified as the most potent inhibitor of NQO2 with an IC_{50} of 6 nM [16, 17]. Notably, NQO2 is also shown to be inhibited by novel compounds, including resveratrol [18, 19], the anti-leukemia drug imatinib [20], and the antimalarial quinolines [21, 22]. It is suggested that NQO2 may serve as a unique molecular target of these compounds to mediate their biological activities.

3.1.2. An Overview of the Biological Functions of NQO Enzymes in Animal Models

The biological functions of NQO enzymes are closely associated with their biochemical properties as described above in Section 2.1. Based on studies in cell cultures and animal

models, the best-recognized biological functions of NQO enzymes are (1) protection against quinone compound-induced toxicity and (2) protection against carcinogenesis. In addition, studies primarily in NQO knockout animals have suggested that NQO enzymes may also play an important role in certain other disease conditions, among which are cardiovascular disorders, lung injury, and autoimmunity. The sections below describe in detail these biological activities of NQO enzymes in experimental models.

3.1.3. Role in Chemical Toxicity

The involvement of both NQO1 and NOQ2 in quinone metabolism has been investigated in various in vitro and in vivo studies. Quinones exist as environmental toxic compounds. They are also formed endogenously, such as from metabolism of estrogens. NQO1 knockout (NQO1^{-/-}) mice are more susceptible to menadione-induced tissue injury compared with wild-type littermates [23]. In contrast, NQO2 knockout (NQO2^{-/-}) mice show decreased sensitivity to menadione toxicity, suggesting that NQO2 may metabolically activate menadione to cause tissue injury [24]. NQO1 deficiency in cultured cells also generally aggravates quinone-induced cytotoxicity, whereas overexpression of NQO1 usually protects the cells from quinone toxicity induced by this human leukemogen. Targeted disruption of NQO1 in mice markedly sensitizes the animals to benzene-induced hematotoxicity and bone marrow toxicity, as assessed by peripheral blood counts and bone marrow histopathology [25]. As described below in Section 3.2.2, deficiency of NQO1 in humans also increases the susceptibility to benzene-induced leukemias.

It should be noted that some quinone anticancer drugs, such as mitomycin C can be bioactivated by NQO1 under hypoxic conditions. This is due to the fact that two-electron reduction of these anticancer quinone drugs results in the formation of more reactive metabolites [26]. This unique mechanism of bioactivation may lead to enhanced and selective tumor killing because tumor cells usually express higher levels of NQO1 and have lower oxygen concentrations than do normal cells. On the other hand, NQO1-mediated bioactivation of mitomycin C may also contribute to the adverse effects of this anticancer drug. Indeed, NQO1^{-/-} mice show a complete resistance to mitomycin C-induced bone marrow suppression. The heterozygous (NQO1^{+/-}) mice also exhibit partial resistance to mitomycin C-induced bone marrow toxicity compared with wild-type mice [27]. In cultured cells, NQO2 is also able to bioactivate mitomycin C as well as several other quinones [28]. However, targeted disruption of NQO2 in mice does not aggravate mitomycin C-induced bone marrow suppression and hematotoxicity [27]. Thus, NQO1 and NQO2 may exert differential effects on quinone metabolism and toxicity in mammalian systems.

3.1.4. Role in Carcinogenesis

Metabolism of environmental carcinogens leads to the formation of reactive quinones. Likewise, quinones are also metabolites of endogenous estrogens, which are implicated in estrogen carcinogenesis [29]. Both NQO1^{-/-} and NQO2^{-/-} mice show increased susceptibility to skin carcinogenesis induced by environmental carcinogens, including benzo(a)pyrene and 7,12-dimethylbenz(a)anthracene [30-33]. NQO1^{-/-} mice spontaneously develop myelogenous hyperplasia due to decreased apoptosis [34]. In addition, as compared with wild-type mice, NQO1^{-/-} mice are much more susceptible to benzene-elicited genotoxicity and radiation-induced myeloproliferative disease [35]. Targeted disruption of NQO2 in mice also leads to

myeloid hyperplasia of bone marrow and increased neutrophils, basophils, eosinophils, and platelets in the peripheral blood. It is further shown that decreased apoptosis of bone marrow cells and circulating granulocytes contributes to myeloid hyperplasia and hyperactivity of bone marrow in NQO2^{-/-} mice [24]. NQO2^{-/-} mice are shown to be highly susceptible to radiation-induced B-cell lymphomas, suggesting that NQO2 may function as an endogenous factor in prevention against radiation-induced lymphomas [36].

Both NQO1 and NQO2 are able to reduce estrogen-derived quinone metabolites, thereby protecting cells from these quinone-induced genotoxicity [37, 38]. Estrogen-derived quinones have been implicated in breast carcinogenesis. As discussed below, gene polymorphism studies in humans suggest a protective role for both NQO1 and NQO2 in breast cancer development (Section 3.2). Induction of NQO1 by oltipraz, a chemoprotective agent is shown to attenuate chemically-induced colon carcinogenesis [39, 40]. Targeted disruption of NQO1 in mice also leads to increased oxidative DNA lesions in liver and kidney under basal conditions, indicating that NQO1 may be important in attenuating endogenous oxidative DNA damage in vivo [41].

3.1.5. Role in Cardiovascular Diseases and Related Conditions

NQO1 is a highly expressed antioxidant and anti-inflammatory enzyme in cardiovascular cells and tissues under basal conditions [2, 42, 43]. Its expression in cardiovascular system is also strongly inducible by various stimuli, including oxidants, reactive aldehydes, homocysteine, cigarette smoke, inflammatory cytokines, sheer force, as well as chemoprotective agents. Many of these factors are critically involved in cardiovascular disorders. Overexpression of NQO1 in cultured cardiovascular cells protects them not only from oxidative cytotoxicity, but also from cytokine-induced proinflammatory responses [44, 45]. A recent study reports that activation of NQO1 prevents arterial restenosis by suppressing vascular smooth muscle cell proliferation in an animal model of balloon injury of carotid artery [46]. Adenoviral vector-mediated overexpression of NQO1 in cultured vascular cells attenuates tumor necrosis factor- α -induced endothelial adhesion molecule expression and smooth muscle cell migration [44, 45]. These in vivo and in vitro data suggest that NQO1 functions as an important protector in atherogenesis, and induction of NQO1 in vasculature may thus represent an effective strategy for the intervention of vascular disorders.

NQO1 also plays a role in lipid metabolism and insulin resistance. As compared with wild-type mice, NQO1^{-/-} mice exhibit higher ratios of NAD(P)H to NAD(P)⁺, lower levels of abdominal adipose tissue, and higher levels of hepatic triglycerides. Insulin tolerance test demonstrates that NQO1^{-/-} mice are insulin-resistant [47]. In rodent models of metabolic syndrome, β -lapachone-mediated pharmacological stimulation of NADH oxidation by NQO1 is shown to dramatically ameliorate obesity and related phenotypes, including glucose intolerance, dyslipidemia, and fatty liver [48]. β -Lapachone is a specific and high-affinity substrate of NQO1, and able to potently stimulate NADH oxidation by NQO1 in vitro and in vivo [49, 50]. These observations open the possibility of developing strategies targeting NQO1 for the treatment of metabolic syndrome.

3.1.6. Role in Pulmonary Diseases

As stated above, NQO1 and NQO2 double-knockout (NQO1^{-/-}/NQO2^{-/-}) mice are born and reproduced normally. However, the double-knockout mice show myelogenous hyperpla-

185

sia of bone marrow and development of bronchial-associated lymphoid tissue (BALT) infiltrated with neutrophils and macrophages [51]. The BALT is absent in wild-type and singleknockout mice. The double-knockout mice also develop systemic inflammation, as evidenced by increased serum levels of tumor necrosis factor- α , interleukin-1, and interleukin-6, as well as increased expression of inducible nitric oxide synthase and higher nitric oxide production in lung macrophages. NQO1^{-/-}/NQO2^{-/-} mice upon exposure to hyperoxia demonstrate severe intra-alveolar edema, perivascular inflammation, and massive infiltration of neutrophils as compared with wild-type mice [51]. These findings suggest that NQO1 and NQO2 together function to suppress BALT development, lung inflammation, and hyperoxic lung injury in experimental animals.

Epidemiological studies have identified NQO1 as a host susceptible factor for ozone exposure. This has prompted experimental studies on the role of NQO1 in ozone toxicity in both animal models and cell cultures. Targeted disruption of NQO1 in mice renders them resistant to ozone-induced toxicity, as evidenced by decreased oxidative stress, attenuated airway inflammation, and diminished airway hyperresponsiveness [52]. In cultured primary human airway epithelial cells, inhibition of NQO1 by dicumarol attenuates ozone-induced F₂-isoprostane production and interleukin-8 gene expression [52]. F₂-Isoprostane is a highly reliable and specific biomarker of oxidative stress. These in vivo and in vitro data indicate that NQO1 may play a detrimental role in ozone-induced lung inflammation and oxidative stress. These results on ozone toxicity contradict the observations that NQO1 protects against hyper-oxia-induced lung injury, in which oxidative stress and inflammation play a major role. It remains unknown how NQO1 functions as a susceptible factor in ozone toxicity, and if this is an ozone-specific phenomenon.

3.1.7. Role in Immunity and Autoimmunity

There is ample evidence for a beneficial function of NQO1 in inflammatory responses in both in vitro and in vivo systems. Studies in gene knockout mouse models demonstrate that both NOO1 and NOO2 participate in regulating humoral immunity and autoimmunity. Although targeted disruption of either NQO1 or NQO2 in mice leads to myeloid hyperplasia of bone marrow, these knockout mice show decreased lymphocytes in peripheral blood [53]. NOO1^{-/-} and NOO2^{-/-} mice also exhibit lower germinal center response, altered B cell homing, and impaired primary and secondary immune responses. Both NOO1^{-/-} and NOO2^{-/-} mice display increased susceptibility to autoimmune disease, as revealed by decreased apoptosis of thymocytes and predisposition to collagen-induced arthritis. As compared with wild-type mice, the autoimmunity-induced arthritis in NOO1^{-/-} mice is more severe, has an earlier onset, and lasts longer. There is no significant difference between NQO2^{-/-} and wild-type mice in the onset and severity of the arthritis. However, the arthritis lasts longer in NQO2^{-/-} mice as compared with wild-type mice [53]. These observations implicate that NOO1 and NOO2 are endogenous factors in the regulation of immune response and autoimmunity in animal models. The detailed mechanisms by which NQO1 and NQO2 modulate immune response and autoimmunity await further investigations.

3.1.8. Role in Other Diseases and Conditions

NQO1 is expressed in the central nervous system, and may play a protective role in neurodegenerative disorders, particularly Parkinson's disease. Metabolism of dopamine to reactive quinone metabolites with the concurrent production of ROS is believed to be responsible, at least partially, for dopaminergic neuron degeneration in Parkinson's disease. NQO1 is shown to catalyze two-electron reduction of dopamine-derived quinones, and protect cultured neurons from dopamine toxicity [54]. In contrast, overexpression of NQO2 in a dopaminergic neuronal cell line augments dopamine-mediated ROS formation [55]. Therefore, the exact role of NQO1 and NQO2 in neurodegeneration warrants further studies in animal models. In this regard, NQO1^{-/-} and NQO2^{-/-} mouse models will be important tools for understanding the role of NQO enzymes in experimental parkinsonism as well as other neurodegenerative disorders, such as Alzheimer's disease.

Melatonin is a neurohormone implicated in both biorhythm synchronization and neuroprotection from oxidative stress. Melatonin is also involved in cytoprotection against oxidative stress in various other tissues and disease conditions (see Chapter 19 for discussion of melatonin). The functions of melatonin are believed to be mediated by two G-proteincoupled-receptors (MT1 and MT2) and MT3, which corresponds to NQO2 [8, 9]. Indeed, all tissues from NQO2-null mice are depleted of the melatonin binding site MT3. However, it remains to be elucidated if the specific interaction between melatonin and NQO2 is responsible for the antioxidant activities of this neurohormone.

NQO1 may participate in regulating skin pigmentation process. NQO1 expression is diminished in multiple amelanotic melanoma cell lines. Treatment of both cultured normal human melanocytes and zebrafish with the NQO1 inhibitor ES936 or dicumarol markedly decreases melanogenesis. In contrast, adenoviral vector-mediated overexpression of NQO1 augments melanogenesis, concomitantly with an increased protein level of tyrosinase, the key enzyme involved in the biosynthesis of melanin [56]. These observations indicate that NQO1 may be a positive regulator of the skin pigmentation process via suppression of tyrosinase degradation.

3.2. Human Studies and Clinical Perspectives

3.2.1. Study Approaches and Gene Polymorphisms

The role of NQO enzymes, especially NQO1 in health and disease has been extensively studied in humans. The study approaches in human subjects primarily consist of (1) determination of the changes of NQO enzymes in tissue samples from patients with various disease conditions and (2) clinical and epidemiological studies on the association between NQO gene polymorphisms and disease risk. Both NQO1 and NQO2 genes are polymorphic. Two genetic polymorphisms have been identified in NQO1, which are designated as NQO1*2 and NQO1*3, respectively. The wild-type allele of NQO1 is designated as NQO1*1.

NQO1*2 is a single nucleotide polymorphism, a C to T change at position 609 of the NQO1 cDNA. This nucleotide substitution results in a proline to serine substitution at position 187 of the amino acid sequence of the protein. As such, NQO1*2 is frequently denoted as C609T or Pro187Ser. The mutant NQO1*2 protein is very unstable, and is rapidly ubiquitinated and degraded by the proteasome. Cells with one NQO1*1 allele and one NQO1*2 allele have approximately half of the NQO1 activity of NQO1*1/*1 cells (the wild-type cells), but NQO1*2/*2 cells have no NQO1 activity. Thus, both NQO1 protein and activity are virtually undetectable in tissue samples of individuals carrying the homozygous NQO1*2 genotype (i.e., NQO1*2/*2, also denoted as null genotype). Approximately 10% of the human popula-

tions have the homozygous NQO1*2 genotype, but the frequency varies in different ethical groups [10, 11, 57-59].

A second NQO1 polymorphism has been identified at base 465 of the NQO1 gene also involving a base change from cytosine to thymine, designated as NQO1*3 or C465T. This nucleotide change results in an arginine to tryptophan substitution at position 139 of the amino acid sequence of the protein. As such, this polymorphism is also denoted as Arg139Trp. The NQO1*3 genotype also results in a phenotype with lower NQO1 activity due to the decreased stability of the variant RNA and decreased activity of the resulting protein [58]. However, this polymorphism occurs at a much lower frequency than does NQO1*2. As such, most of the epidemiological studies on the association of NQO1 gene polymorphisms with diseases have dealt with NQO1*2. NQO1*2 polymorphism has been linked to a wide range of disease conditions, including cancer, cardiovascular diseases, diabetes, and neurological disorders. In addition, polymorphisms in NQO2 gene have been shown to affect the expression and activity of NQO2 protein, and be associated with altered risk of developing certain human diseases as well.

3.2.2. Role in Human Cancer

Knowledge on the critical role of NQO enzymes in protecting against experimental carcinogenesis and the p53 stabilizing activity of NQO1 has prompted studies on the involvement of NQO enzymes in human cancer development. Epidemiological studies have shown an increased frequency of NQO1*2 allele in patients with cancer, including leukemias, breast cancer, and colon cancer compared with healthy controls, suggesting a protective role for the normal NQO1 protein in cancer development. An association of the NQO1*2 polymorphism and cancer has been most frequently found in leukemias. Individuals with NQO1 deficiency (NQO1*2/*2) also exhibit an increased risk of developing leukemias following occupational exposure to benzene [60].

It is recently reported that NQO1*2 genotype is a strong prognostic and predicting factor in breast cancer [61]. The homozygous NQO1*2 strongly predicts poor survival in women with breast cancer, which is particularly evident after anthracycline-based adjuvant chemotherapy with epirubicin and in p53-aberrant tumors. Survival after metastasis is reduced among NQO1*2 homozygotes, further implicating NQO1 deficiency in breast cancer progression and treatment resistance. Consistently, response to epirubicin is impaired in NQO1*2homozygous breast carcinoma cells in vitro [61]. In cultured human breast cancer cells, cDNA-mediated expression of the polymorphic variants (NQO1*2 and NQO1*3) of NQO1 is found to poorly reduce estrogen quinones, thereby leading to increased formation of estrogen quinone-DNA adducts [62]. Such DNA adducts are implicated in estrogen-induced breast carcinogenesis. Polymorphisms of NQO2 that lead to decreased NQO2 expression are also associated with increased susceptibility to breast cancer [63]. Wild-type NQO2 is shown to reduce estrogen quinones, leading to decreased DNA adduct formation [38]. Nevertheless, the precise role of NQO-mediated reduction of estrogen quinones in breast cancer development in human patients awaits further evaluation.

Multiple studies suggest that NQO1 plays a significant role in preventing the development of colorectal cancer, and individuals with an NQO1*2/*2 genotype are at an increased risk of developing this disease [64]. This conclusion is in line with results from the experimental studies showing that NQO1 may participate in preventing colorectal carcinogenesis in a rat model [39, 40]. Although extensive studies suggest a significant protective role for NQO enzymes, especially NQO1 in numerous types of cancer, there are also other epidemiological studies reporting no differences or mixed results in occurrence of NQO1*2 in various types of cancer, including renal cancer, lung cancer, gastric cancer, prostate cancer, gliomas, and lymphomas. The involvement of NQO1 and NQO2 gene polymorphisms in human cancer development can be potentially influenced by many factors. These include different ethical backgrounds, stage of cancer, concomitant alterations in other antioxidant genes or disease-modifying genes, dietary and environmental factors, as well as study design. Confirmation in experimental systems, including cell cultures and animal studies will further strengthen the conclusion regarding a positive association between NQO gene polymorphisms and risk of human disease development.

3.2.3. Role in Human Cardiovascular Diseases and Related Conditions

NQO1 is highly expressed and inducible in cardiovascular tissues and cells. As described above in Section 3.1.5, activation of NQO1 attenuates atherogenesis in experimental animals. A recent epidemiological study involving 834 patients with type 2 diabetes reports that patients with NQO1*2 polymorphism exhibit a higher prevalence of atherosclerotic plaques of carotid artery than those without this polymorphism [65]. Another study in patients undergoing standard coronary artery bypass grafting shows a greater inflammatory response, as revealed by increased serum interleukin-6 levels, in the subjects with NQO1*2 genotype as compared with the patients without this polymorphism [66]. Significantly higher levels of C-reactive protein, a biomarker for coronary heart disease, are reported to occur in coronary heart disease patients with lower NQO1 activity [67], suggesting a protective role for NQO1 in cardiovascular inflammation.

Studies in humans have revealed that NQO1 gene is also highly expressed in adipose tissue, and its mRNA expression is reduced during diet-induced weight loss [68]. The expression levels of NOO1 mRNA are found to positively correlate with adiposity, glucose tolerance, and markers of liver dysfunction, suggesting a possible involvement of NQO1 in the metabolic complications of human obesity. However, such studies provide no evidence as to whether NQO1 plays an etiological role in the disease pathophysiology or increased NQO1 mRNA expression is simply an accompanying phenomenon of obesity and its metabolic complications. Alternatively, the increased NQO1 mRNA expression could be an adaptive response to obesity and its metabolic complications, and might serve as a protective mechanism. In this context, NQO1 is inducible by inflammation, a critical event in metabolic syndrome. The involvement of NQO1*2 polymorphism in both type 1 and type 2 diabetes has been investigated in two epidemiological studies in a Chinese population and a Danish population, and no association between NQO1*2 genotype and diabetes (either type 1 or type 2) is demonstrated [69, 70]. The potential involvement of NQO1 in human diabetes and metabolic syndrome deserves further investigations, including studies in other ethical groups and geographical regions.

3.2.4. Role in Human Neurological Diseases

Oxidative stress and dopamine metabolism are implicated in Parkinson's disease. NQO enzymes are proposed to protect against Parkinson's disease via their ability to inhibit oxidative stress and augment the detoxification of dopamine quinones. In this context, NQO1 expression is found in both normal and Parkinsonian substantia nigra [71]. Furthermore, a marked increase in the astroglial and neuronal expression of NOO1 is consistently observed in the Parkinsonian substantia nigra. However, studies to date on the association between NQO1*2 gene polymorphism and risk of developing Parkinson's disease have produced null results [72, 73]. In contrast, polymorphisms in NQO2 gene promoter are shown to be associated with altered susceptibility to Parkinson's disease [55, 72]. The promoter of NOO2 gene is found to be highly polymorphic. Several novel polymorphic alleles are reported, among which are 29-bp insertion (I-29 allele) and 29-bp deletion (D-29 allele). Although the D-29 allele was initially reported to be associated with decreased NOO2 expression, subsequent studies demonstrate the opposite, i.e., the D-29 allele in the promoter directs augmented NQO2 gene expression that results in greater enzymatic activity than wild-type allele [55]. Two studies report a significantly higher frequency of the D-29 allele in patients with Parkinson's disease than in control, suggesting that this NQO2 polymorphism and the resulting elevated NOO2 expression may be associated with increased susceptibility to Parkinson's disease. As described in Section 3.1.8, overexpression of NOO2 in a dopaminergic neuronal cell line leads to increased production of ROS when being exposed to exogenous dopamine, indicating a possible role for NOO2 in augmenting dopamine-mediated oxidative stress and neuronal injury. In contrast to the above two human studies, another population-based casecontrol study with 190 Parkinson's disease patients and 305 unrelated controls matched on age and sex demonstrates no association between the I/D-29 polymorphisms and development of Parkinson's disease [73].

NQO1 expression is found to co-localize with the Alzheimer's disease pathology, and the ratio of frontal to cerebellar NQO1 enzyme activity is significantly increased in patients with Alzheimer's disease as compared with that in control subjects [74], suggesting a possible involvement of NQO1 in this neurodegenerative disorder. A recent study in a Chinese population consisting of 311 Alzheimer's disease patients and 330 controls shows a higher frequency of NQO1*2 allele and a lower frequency of wild-type allele (NQO1*1) in the Alzheimer's patients compared with controls [75]. These observations implicate a possible protective effect of NQO1 in the development of Alzheimer's disease. However, a similar study also in a Chinese population reveals no association between NQO1*2 genotype and late-onset Alzheimer's disease [76].

Multiple studies suggest an involvement of NQO2 in other neurological/psychiatric disorders. For example, a study in a Japanese population of 102 patients with schizophrenia and 234 controls reports that the D-29 polymorphism of NQO2 gene promoter is associated with an increased susceptibility to developing schizophrenia [77]. In the same study, D-29 genotype is found to be associated with a lower NQO2 mRNA in white blood cells. Functional polymorphisms of NQO2 gene are also associated with methamphetamine psychosis [78] as well as the pathogenesis of alcoholism and alcohol withdrawal symptoms [79], indicating a possible role for NQO2 in modulating drug abuse-related mental and behavioral changes. A more recent study using a cohort of 722 community-dwelling older individuals aged 50 years and over demonstrates that a functional NQO2 polymorphism is associated with cognitive decline.

3.2.5. Role in Other Human Diseases

NQO gene polymorphisms are also associated with a number of other human disease conditions. These include asthma [80, 81], acute lung injury [82], inflammatory bowel disease [83], urolithiasis [84], and clozapine-induced agranulocytosis [85]. However, further studies

are needed to better characterize the involvement of NQO gene polymorphisms in modulating the risk of developing the above human diseases.

4. Conclusion and Future Directions

NQO enzymes are multifunctional proteins that play a beneficial role in diverse pathophysiological processes involving oxidative stress and inflammation. This notion has been supported by extensive studies in animal models as well as in human subjects. Future studies should focus on how to translate the enormous amounts of basic and clinical knowledge about NQO enzymes into novel preventive and therapeutic modalities for the intervention of human diseases in which NQO enzymes play a causative role. In this context, further investigations of the role of NQO gene polymorphisms in disease processes in well-designed clinical and epidemiological studies will also contribute to the development of sensitive and selective NQO-based genetic biomarkers of disease susceptibility and progression. Use of these biomarkers will help identify susceptible subjects to whom NQO-based interventional modalities may be selectively applied.

5. References

- Siegel D, Gustafson DL, Dehn DL, Han JY, Boonchoong P, Berliner LJ, *et al.* NAD(P)H:quinone oxidoreductase 1: role as a superoxide scavenger. *Mol. Pharmacol.* 2004;65:1238-1247.
- [2] Zhu H, Jia Z, Mahaney JE, Ross D, Misra HP, Trush MA, *et al.* The highly expressed and inducible endogenous NAD(P)H:quinone oxidoreductase 1 in cardiovascular cells acts as a potential superoxide scavenger. *Cardiovasc. Toxicol.* 2007;7:202-211.
- [3] Asher G, Lotem J, Cohen B, Sachs L, Shaul Y. Regulation of p53 stability and p53dependent apoptosis by NADH quinone oxidoreductase 1. *Proc. Natl. Acad. Sci. USA* 2001;98:1188-1193.
- [4] Anwar A, Dehn D, Siegel D, Kepa JK, Tang LJ, Pietenpol JA, et al. Interaction of human NAD(P)H:quinone oxidoreductase 1 (NQO1) with the tumor suppressor protein p53 in cells and cell-free systems. J. Biol. Chem. 2003;278:10368-10373.
- [5] Asher G, Tsvetkov P, Kahana C, Shaul Y. A mechanism of ubiquitin-independent proteasomal degradation of the tumor suppressors p53 and p73. *Genes Dev.* 2005; 19:316-321.
- [6] Garate M, Wong RP, Campos EI, Wang Y, Li G. NAD(P)H quinone oxidoreductase 1 inhibits the proteasomal degradation of the tumour suppressor p33(ING1b). *EMBO Rep.* 2008;9:576-581.
- [7] Alard A, Fabre B, Anesia R, Marboeuf C, Pierre P, Susini C, et al. NAD(P)H quinoneoxydoreductase 1 protects eukaryotic translation initiation factor 4GI from degradation by the proteasome. *Mol. Cell. Biol.* 2010;30:1097-1105.
- [8] Nosjean O, Ferro M, Coge F, Beauverger P, Henlin JM, Lefoulon F, *et al.* Identification of the melatonin-binding site MT3 as the quinone reductase 2. *J. Biol. Chem.* 2000;275:31311-31317.

- [9] Mailliet F, Ferry G, Vella F, Thiam K, Delagrange P, Boutin JA. Organs from mice deleted for NRH:quinone oxidoreductase 2 are deprived of the melatonin binding site MT3. *FEBS Lett.* 2004;578:116-120.
- [10] Nioi P, Hayes JD. Contribution of NAD(P)H:quinone oxidoreductase 1 to protection against carcinogenesis, and regulation of its gene by the Nrf2 basic-region leucine zipper and the arylhydrocarbon receptor basic helix-loop-helix transcription factors. *Mutat. Res.* 2004;555:149-171.
- [11] Dinkova-Kostova AT, Talalay P. NAD(P)H:quinone acceptor oxidoreductase 1 (NQO1), a multifunctional antioxidant enzyme and exceptionally versatile cytoprotector. Arch. Biochem. Biophys. 2010;501:116-123.
- [12] Wang W, Jaiswal AK. Sp3 repression of polymorphic human NRH:quinone oxidoreductase 2 gene promoter. *Free Radic. Biol. Med.* 2004;37:1231-1243.
- [13] Wang W, Jaiswal AK. Nuclear factor Nrf2 and antioxidant response element regulate NRH:quinone oxidoreductase 2 (NQO2) gene expression and antioxidant induction. *Free Radic. Biol. Med.* 2006;40:1119-1130.
- [14] Preusch PC, Siegel D, Gibson NW, Ross D. A note on the inhibition of DT-diaphorase by dicoumarol. *Free Radic. Biol. Med.* 1991;11:77-80.
- [15] Dehn DL, Siegel D, Swann E, Moody CJ, Ross D. Biochemical, cytotoxic, and genotoxic effects of ES936, a mechanism-based inhibitor of NAD(P)H:quinone oxidoreductase 1, in cellular systems. *Mol. Pharmacol.* 2003;64:714-720.
- [16] Nolan KA, Humphries MP, Barnes J, Doncaster JR, Caraher MC, Tirelli N, et al. Triazoloacridin-6-ones as novel inhibitors of the quinone oxidoreductases NQO1 and NQO2. Bioorg. Med. Chem. 2010;18:696-706.
- [17] Nolan KA, Humphries MP, Bryce RA, Stratford IJ. Imidazoacridin-6-ones as novel inhibitors of the quinone oxidoreductase NQO2. *Bioorg. Med. Chem. Lett.* 2010;20: 2832-2836.
- [18] Buryanovskyy L, Fu Y, Boyd M, Ma Y, Hsieh TC, Wu JM, et al. Crystal structure of quinone reductase 2 in complex with resveratrol. *Biochemistry* 2004;43:11417-11426.
- [19] Wang Z, Hsieh TC, Zhang Z, Ma Y, Wu JM. Identification and purification of resveratrol targeting proteins using immobilized resveratrol affinity chromatography. *Biochem. Biophys. Res. Commun.* 2004;323:743-749.
- [20] Winger JA, Hantschel O, Superti-Furga G, Kuriyan J. The structure of the leukemia drug imatinib bound to human quinone reductase 2 (NQO2). *BMC Struct. Biol.* 2009;9:7.
- [21] Graves PR, Kwiek JJ, Fadden P, Ray R, Hardeman K, Coley AM, et al. Discovery of novel targets of quinoline drugs in the human purine binding proteome. *Mol. Pharmacol.* 2002;62:1364-1372.
- [22] Kwiek JJ, Haystead TA, Rudolph J. Kinetic mechanism of quinone oxidoreductase 2 and its inhibition by the antimalarial quinolines. *Biochemistry* 2004;43:4538-4547.
- [23] Radjendirane V, Joseph P, Lee YH, Kimura S, Klein-Szanto AJ, Gonzalez FJ, et al. Disruption of the DT diaphorase (NQO1) gene in mice leads to increased menadione toxicity. J. Biol. Chem. 1998;273:7382-7389.
- [24] Long DJ, 2nd, Iskander K, Gaikwad A, Arin M, Roop DR, Knox R, et al. Disruption of dihydronicotinamide riboside:quinone oxidoreductase 2 (NQO2) leads to myeloid hyperplasia of bone marrow and decreased sensitivity to menadione toxicity. J. Biol. Chem. 2002;277:46131-46139.

- [25] Bauer AK, Faiola B, Abernethy DJ, Marchan R, Pluta LJ, Wong VA, et al. Genetic susceptibility to benzene-induced toxicity: role of NADPH: quinone oxidoreductase-1. *Cancer Res.* 2003;63:929-935.
- [26] Seow HA, Penketh PG, Belcourt MF, Tomasz M, Rockwell S, Sartorelli AC. Nuclear overexpression of NAD(P)H:quinone oxidoreductase 1 in Chinese hamster ovary cells increases the cytotoxicity of mitomycin C under aerobic and hypoxic conditions. J. Biol. Chem. 2004;279:31606-31612.
- [27] Adikesavan AK, Barrios R, Jaiswal AK. In vivo role of NAD(P)H:quinone oxidoreductase 1 in metabolic activation of mitomycin C and bone marrow cytotoxicity. *Cancer Res.* 2007;67:7966-7971.
- [28] Celli CM, Tran N, Knox R, Jaiswal AK. NRH:quinone oxidoreductase 2 (NQO2) catalyzes metabolic activation of quinones and anti-tumor drugs. *Biochem. Pharmacol.* 2006;72:366-376.
- [29] Yager JD, Davidson NE. Estrogen carcinogenesis in breast cancer. N. Engl. J. Med. 2006;354:270-282.
- [30] Long DJ, 2nd, Waikel RL, Wang XJ, Perlaky L, Roop DR, Jaiswal AK. NAD(P)H:quinone oxidoreductase 1 deficiency increases susceptibility to benzo(a)pyrene-induced mouse skin carcinogenesis. *Cancer Res.* 2000;60:5913-5915.
- [31] Iskander K, Paquet M, Brayton C, Jaiswal AK. Deficiency of NRH:quinone oxidoreductase 2 increases susceptibility to 7,12-dimethylbenz(a)anthracene and benzo(a)pyrene-induced skin carcinogenesis. *Cancer Res.* 2004;64:5925-5928.
- [32] Iskander K, Gaikwad A, Paquet M, Long DJ, 2nd, Brayton C, Barrios R, et al. Lower induction of p53 and decreased apoptosis in NQO1-null mice lead to increased sensitivity to chemical-induced skin carcinogenesis. *Cancer Res.* 2005;65:2054-2058.
- [33] Shen J, Barrios RJ, Jaiswal AK. Inactivation of the quinone oxidoreductases NQO1 and NQO2 strongly elevates the incidence and multiplicity of chemically induced skin tumors. *Cancer Res.* 2010;70:1006-1014.
- [34] Long DJ, 2nd, Gaikwad A, Multani A, Pathak S, Montgomery CA, Gonzalez FJ, et al. Disruption of the NAD(P)H:quinone oxidoreductase 1 (NQO1) gene in mice causes myelogenous hyperplasia. Cancer Res. 2002;62:3030-3036.
- [35] Iskander K, Barrios RJ, Jaiswal AK. Disruption of NAD(P)H:quinone oxidoreductase 1 gene in mice leads to radiation-induced myeloproliferative disease. *Cancer Res.* 2008;68:7915-7922.
- [36] Iskander K, Barrios RJ, Jaiswal AK. NRH:quinone oxidoreductase 2-deficient mice are highly susceptible to radiation-induced B-cell lymphomas. *Clin. Cancer Res.* 2009;15: 1534-1542.
- [37] Gaikwad NW, Rogan EG, Cavalieri EL. Evidence from ESI-MS for NQO1-catalyzed reduction of estrogen ortho-quinones. *Free Radic. Biol. Med.* 2007;43:1289-1298.
- [38] Gaikwad NW, Yang L, Rogan EG, Cavalieri EL. Evidence for NQO2-mediated reduction of the carcinogenic estrogen ortho-quinones. *Free Radic. Biol. Med.* 2009;46: 253-262.
- [39] Begleiter A, Sivananthan K, Curphey TJ, Bird RP. Induction of NAD(P)H quinone: oxidoreductase1 inhibits carcinogen-induced aberrant crypt foci in colons of Sprague-Dawley rats. *Cancer Epidemiol. Biomarkers Prev.* 2003;12:566-572.
- [40] Begleiter A, Sivananthan K, Lefas GM, Maksymiuk AW, Bird RP. Inhibition of colon carcinogenesis by post-initiation induction of NQO1 in Sprague-Dawley rats. *Oncol.*

Rep. 2009;21:1559-1565.

- [41] Zhou GD, Randerath K, Donnelly KC, Jaiswal AK. Effects of NQO1 deficiency on levels of cyclopurines and other oxidative DNA lesions in liver and kidney of young mice. *Int. J. Cancer* 2004;112:877-883.
- [42] Siegel D, Ross D. Immunodetection of NAD(P)H:quinone oxidoreductase 1 (NQO1) in human tissues. *Free Radic. Biol. Med.* 2000;29:246-253.
- [43] Cao Z, Li Y. The chemical inducibility of mouse cardiac antioxidants and phase 2 enzymes in vivo. *Biochem. Biophys. Res. Commun.* 2004;317:1080-1088.
- [44] Chen XL, Varner SE, Rao AS, Grey JY, Thomas S, Cook CK, et al. Laminar flow induction of antioxidant response element-mediated genes in endothelial cells: a novel anti-inflammatory mechanism. J. Biol. Chem. 2003;278:703-711.
- [45] Lee SO, Chang YC, Whang K, Kim CH, Lee IS. Role of NAD(P)H:quinone oxidoreductase 1 on tumor necrosis factor-alpha-induced migration of human vascular smooth muscle cells. *Cardiovasc. Res.* 2007;76:331-339.
- [46] Kim SY, Jeoung NH, Oh CJ, Choi YK, Lee HJ, Kim HJ, et al. Activation of NAD(P)H:quinone oxidoreductase 1 prevents arterial restenosis by suppressing vascular smooth muscle cell proliferation. *Circ. Res.* 2009;104:842-850.
- [47] Gaikwad A, Long DJ, 2nd, Stringer JL, Jaiswal AK. In vivo role of NAD(P)H:quinone oxidoreductase 1 (NQO1) in the regulation of intracellular redox state and accumulation of abdominal adipose tissue. J. Biol. Chem. 2001;276:22559-22564.
- [48] Hwang JH, Kim DW, Jo EJ, Kim YK, Jo YS, Park JH, et al. Pharmacological stimulation of NADH oxidation ameliorates obesity and related phenotypes in mice. *Diabetes* 2009;58:965-974.
- [49] Pink JJ, Planchon SM, Tagliarino C, Varnes ME, Siegel D, Boothman DA. NAD(P)H:Quinone oxidoreductase activity is the principal determinant of betalapachone cytotoxicity. J. Biol. Chem. 2000;275:5416-5424.
- [50] Bey EA, Bentle MS, Reinicke KE, Dong Y, Yang CR, Girard L, et al. An NQO1- and PARP-1-mediated cell death pathway induced in non-small-cell lung cancer cells by beta-lapachone. Proc. Natl. Acad. Sci. USA 2007;104:11832-11837.
- [51] Das A, Kole L, Wang L, Barrios R, Moorthy B, Jaiswal AK. BALT development and augmentation of hyperoxic lung injury in mice deficient in NQO1 and NQO2. *Free Radic. Biol. Med.* 2006;40:1843-1856.
- [52] Voynow JA, Fischer BM, Zheng S, Potts EN, Grover AR, Jaiswal AK, *et al.* NAD(P)H quinone oxidoreductase 1 is essential for ozone-induced oxidative stress in mice and humans. *Am. J. Respir. Cell. Mol. Biol.* 2009;41:107-113.
- [53] Iskander K, Li J, Han S, Zheng B, Jaiswal AK. NQO1 and NQO2 regulation of humoral immunity and autoimmunity. *J. Biol. Chem.* 2006;281:30917-30924.
- [54] Zafar KS, Inayat-Hussain SH, Siegel D, Bao A, Shieh B, Ross D. Overexpression of NQO1 protects human SK-N-MC neuroblastoma cells against dopamine-induced cell death. *Toxicol. Lett.* 2006;166:261-267.
- [55] Wang W, Le WD, Pan T, Stringer JL, Jaiswal AK. Association of NRH:quinone oxidoreductase 2 gene promoter polymorphism with higher gene expression and increased susceptibility to Parkinson's disease. J. Gerontol. A Biol. Sci. Med. Sci. 2008; 63:127-134.
- [56] Choi TY, Sohn KC, Kim JH, Kim SM, Kim CH, Hwang JS, et al. Impact of NAD(P)H:quinone oxidoreductase-1 on pigmentation. J. Invest. Dermatol. 2010;130:

784-792.

- [57] Kelsey KT, Ross D, Traver RD, Christiani DC, Zuo ZF, Spitz MR, *et al.* Ethnic variation in the prevalence of a common NAD(P)H quinone oxidoreductase polymorphism and its implications for anti-cancer chemotherapy. *Br. J. Cancer* 1997; 76:852-854.
- [58] Gaedigk A, Tyndale RF, Jurima-Romet M, Sellers EM, Grant DM, Leeder JS. NAD(P)H:quinone oxidoreductase: polymorphisms and allele frequencies in Caucasian, Chinese and Canadian Native Indian and Inuit populations. *Pharmacogenetics* 1998;8:305-313.
- [59] Traver RD, Siegel D, Beall HD, Phillips RM, Gibson NW, Franklin WA, et al. Characterization of a polymorphism in NAD(P)H: quinone oxidoreductase (DTdiaphorase). Br. J. Cancer 1997;75:69-75.
- [60] Rothman N, Smith MT, Hayes RB, Traver RD, Hoener B, Campleman S, *et al.* Benzene poisoning, a risk factor for hematological malignancy, is associated with the NQO1 609C→T mutation and rapid fractional excretion of chlorzoxazone. *Cancer Res.* 1997;57:2839-2842.
- [61] Fagerholm R, Hofstetter B, Tommiska J, Aaltonen K, Vrtel R, Syrjakoski K, et al. NAD(P)H:quinone oxidoreductase 1 NQO1*2 genotype (P187S) is a strong prognostic and predictive factor in breast cancer. *Nat. Genet.* 2008;40:844-853.
- [62] Singh S, Zahid M, Saeed M, Gaikwad NW, Meza JL, Cavalieri EL, et al. NAD(P)H:quinone oxidoreductase 1 Arg139Trp and Pro187Ser polymorphisms imbalance estrogen metabolism towards DNA adduct formation in human mammary epithelial cells. J. Steroid. Biochem. Mol. Biol. 2009;117:56-66.
- [63] Yu KD, Di GH, Yuan WT, Fan L, Wu J, Hu Z, *et al.* Functional polymorphisms, altered gene expression and genetic association link NRH:quinone oxidoreductase 2 to breast cancer with wild-type p53. *Hum. Mol. Genet.* 2009;18:2502-2517.
- [64] Begleiter A, Hewitt D, Maksymiuk AW, Ross DA, Bird RP. A NAD(P)H:quinone oxidoreductase 1 polymorphism is a risk factor for human colon cancer. *Cancer Epidemiol. Biomarkers Prev.* 2006;15:2422-2426.
- [65] Han SJ, Kang ES, Kim HJ, Kim SH, Chun SW, Ahn CW, et al. The C609T variant of NQO1 is associated with carotid artery plaques in patients with type 2 diabetes. *Mol. Genet. Metab.* 2009;97:85-90.
- [66] Isbir CS, Ergen A, Tekeli A, Zeybek U, Gormus U, Arsan S. The effect of NQO1 polymorphism on the inflammatory response in cardiopulmonary bypass. *Cell Biochem. Funct.* 2008;26:534-538.
- [67] Martin NJ, Collier AC, Bowen LD, Pritsos KL, Goodrich GG, Arger K, *et al.* Polymorphisms in the NQO1, GSTT and GSTM genes are associated with coronary heart disease and biomarkers of oxidative stress. *Mutat. Res.* 2009;674:93-100.
- [68] Palming J, Sjoholm K, Jernas M, Lystig TC, Gummesson A, Romeo S, *et al.* The expression of NAD(P)H:quinone oxidoreductase 1 is high in human adipose tissue, reduced by weight loss, and correlates with adiposity, insulin sensitivity, and markers of liver dysfunction. *J. Clin. Endocrinol. Metab.* 2007;92:2346-2352.
- [69] Wang G, Zhang L, Li Q. Genetic polymorphisms of GSTT1, GSTM1, and NQO1 genes and diabetes mellitus risk in Chinese population. *Biochem. Biophys. Res. Commun.* 2006;341:310-313.
- [70] Kristiansen OP, Larsen ZM, Johannesen J, Nerup J, Mandrup-Poulsen T, Pociot F. No

linkage of P187S polymorphism in NAD(P)H: quinone oxidoreductase (NQO1/DIA4) and type 1 diabetes in the Danish population. DIEGG and DSGD. Danish IDDM Epidemiology and Genetics Group and The Danish Study Group of Diabetes in Childhood. *Hum. Mutat.* 1999;14:67-70.

- [71] van Muiswinkel FL, de Vos RA, Bol JG, Andringa G, Jansen Steur EN, Ross D, *et al.* Expression of NAD(P)H:quinone oxidoreductase in the normal and Parkinsonian substantia nigra. *Neurobiol. Aging* 2004;25:1253-1262.
- [72] Harada S, Fujii C, Hayashi A, Ohkoshi N. An association between idiopathic Parkinson's disease and polymorphisms of phase II detoxification enzymes: glutathione S-transferase M1 and quinone oxidoreductase 1 and 2. *Biochem. Biophys. Res. Commun.* 2001;288:887-892.
- [73] Okada S, Farin FM, Stapleton P, Viernes H, Quigley SD, Powers KM, et al. No associations between Parkinson's disease and polymorphisms of the quinone oxidoreductase (NQO1, NQO2) genes. *Neurosci. Lett.* 2005;375:178-180.
- [74] SantaCruz KS, Yazlovitskaya E, Collins J, Johnson J, DeCarli C. Regional NAD(P)H:quinone oxidoreductase activity in Alzheimer's disease. *Neurobiol. Aging* 2004;25:63-69.
- [75] Bian JT, Zhao HL, Zhang ZX, Bi XH, Zhang JW. Association of NAD(P)H:quinone oxidoreductase 1 polymorphism and Alzheimer's disease in Chinese. J. Mol. Neurosci. 2008;34:235-240.
- [76] Wang B, Jin F, Xie Y, Tang Y, Kan R, Zheng C, et al. Association analysis of NAD(P)H:quinone oxidoreductase gene 609 C/T polymorphism with Alzheimer's disease. *Neurosci. Lett.* 2006;409:179-181.
- [77] Harada S, Tachikawa H, Kawanishi Y. A possible association between an insertion/ deletion polymorphism of the NQO2 gene and schizophrenia. *Psychiatr. Genet.* 2003; 13:205-209.
- [78] Ohgake S, Hashimoto K, Shimizu E, Koizumi H, Okamura N, Koike K, *et al.* Functional polymorphism of the NQO2 gene is associated with methamphetamine psychosis. *Addict. Biol.* 2005;10:145-148.
- [79] Okubo T, Harada S, Higuchi S, Matsushita S. Association analyses between polymorphisms of the phase II detoxification enzymes (GSTM1, NQO1, NQO2) and alcohol withdrawal symptoms. *Alcohol Clin. Exp. Res.* 2003;27:68S-71S.
- [80] Li YF, Tseng PJ, Lin CC, Hung CL, Lin SC, Su WC, et al. NAD(P)H: Quinone oxidoreductase 1, glutathione S-transferase M1, environmental tobacco smoke exposure, and childhood asthma. *Mutat. Res.* 2009;678:53-58.
- [81] Goodrich GG, Goodman PH, Budhecha SK, Pritsos CA. Functional polymorphism of detoxification gene NQO1 predicts intensity of empirical treatment of childhood asthma. *Mutat. Res.* 2009;674:55-61.
- [82] Reddy AJ, Christie JD, Aplenc R, Fuchs B, Lanken PN, Kleeberger SR. Association of human NAD(P)H:quinone oxidoreductase 1 (NQO1) polymorphism with development of acute lung injury. J. Cell Mol. Med. 2009;13:1784-1791.
- [83] Kosaka T, Yoshino J, Inui K, Wakabayashi T, Kobayashi T, Watanabe S, et al. Involvement of NAD(P)H:quinone oxidoreductase 1 and superoxide dismutase polymorphisms in ulcerative colitis. DNA Cell Biol. 2009;28:625-631.
- [84] Schulz WA, Krummeck A, Rosinger I, Schmitz-Drager BJ, Sies H. Predisposition towards urolithiasis associated with the NQO1 null-allele. *Pharmacogenetics*

1998;8:453-454.

[85] Ostrousky O, Meged S, Loewenthal R, Valevski A, Weizman A, Carp H, et al. NQO2 gene is associated with clozapine-induced agranulocytosis. *Tissue Antigens* 2003; 62:483-491.

Chapter 17

Paraoxonase

Abstract

Paraoxonase (PON) refers to a family of three enzymes, namely, PON1, PON2, and PON3. Mammalian PON1 and PON3 are found in circulation bound to high-density lipoprotein, whereas PON2 is an intracellular protein. PON1 was first discovered as an enzyme to hydrolyze paraoxon, an activity that both PON2 and PON3 lack. All three PON enzymes are able to degrade oxidized lipids and protect against oxidative stress. PON enzymes also function to suppress inflammation. Animal studies demonstrate a critical role for PONs, especially PON1 in protecting against diverse disease processes involving oxidative stress and inflammation, such as atherosclerosis, diabetes, and metabolic syndrome. In line with the findings in experimental models, substantial evidence from clinical and epidemiological studies also suggests a potential beneficial role for PONs in a variety of human diseases.

1. Overview

Paraoxonase (PON) refers to a family of enzymes that includes three members in mammals, namely, PON1, PON2, and PON3. PON1 was first identified by W.N. Aldridge in 1953 as an enzyme to hydrolyze paraoxon [1, 2]. PON was thus named for its ability to metabolize paraoxon, the primary metabolite of the organophosphate insecticide parathion. When two closely related enzymes were identified in the mid 1990s, the original PON was referred to as PON1, while the two new enzymes were named PON2 and PON3. In contrast to PON1, neither PON2 nor PON3 exhibits any significant ability to hydrolyze paraoxon or other organophosphate pesticides.

PON1 (43 kDa) is synthesized and secreted by liver, and is primarily associated with high-density lipoprotein (HDL) in plasma. PON1 catalyzes the breakdown of oxidized lipids and inhibits oxidation of HDL as well as low-density lipoprotein (LDL). PON2 is an intracellular protein that is widely distributed in mammalian tissues. Similar to PON1, PON2 with a molecular mass of 44 kDa also acts as an antioxidant protecting against oxidative stress. PON3 (40 kDa) is primarily expressed in liver, and associated with HDL in plasma. Like other members of the paraoxonase family, PON3 has significant antioxidant activity in protect-

ing both HDL and LDL from oxidation. In humans, PON1, PON2, and PON3 are localized next to each other on chromosome 7q21.3-22.1.

2. Biochemistry and Molecular Regulation

2.1. Biochemistry

Since its discovery, mammalian PON1 has been extensively studied with respect to its role in metabolizing organophosphate pesticides, lipid hydroperoxides, and a number of other substrates. In contrast, studies on PON2 and PON3 have been less extensive. Nevertheless, these two new isozymes have been recently found to possess unique biochemical functions. This section focuses on discussing the biochemistry of PON1 along with a description of recent findings on the biochemical properties of PON2 and PON3.

2.1.1. Biochemistry of PON1

The biochemical properties of PON1 are summarized as follows.

- Hydrolysis of organophosphates
- Degradation of oxidized lipids
- Lactonase activity



Figure 17-1. Hydrolysis of paraoxon by paraoxonase 1 (PON1). As illustrated, PON1-catalyzed hydrolysis of paraoxon forms diethyl phosphate and *p*-nitrophenol. See text (Section 2.1.1) for detailed description.

PON1 is an excellent example of a multitasking protein, displaying at least two important biochemical functions: (1) hydrolysis of organophosphates (pesticides and nerve agents) and (2) inhibition of lipoprotein oxidation by catalyzing degradation of oxidized lipids. PON1 effectively hydrolyzes a variety of organophosphate pesticides and nerve agents, including paraoxon, diazoxon, chlorpyrifos-oxon, sarin, and soman [3]. Hydrolysis of these organophosphates by PON1 leads to their detoxification. Figure 17-1 illustrates the chemical reaction of PON1-catalyzed hydrolysis of paraoxon.

Since paraoxon and other organophosphate toxicants are not present in the human body under normal conditions, other molecules would need to be the physiological substrates of PON1. Indeed, extensive studies have demonstrated that PON1 protects LDL and HDL from lipid peroxidation by catalyzing the degradation of oxidized lipids contained in the oxidized lipoproteins [4]. Notably, cysteine 284 is required for the enzyme's ability to hydrolyze oxidized lipids, but not for its activity against organophosphates. This suggests that differential mechanisms of hydrolysis may operate towards organophosphates and oxidized lipids [5]. It has been shown that PON1 can be inactivated by S-glutathionylation [6], a redox signaling mechanism characterized by formation of a mixed disulfide bridge between a protein sulfhydryl group and the oxidized form of glutathione (GSSG) (see Chapter 9 for discussion of protein glutathionylation).

In addition to hydrolyzing organophosphates, PON1 also hydrolyzes aromatic esters, preferably those of acetic acid. Indeed, phenyl acetate is one of the most commonly used substrates for assaying the enzymatic activity of PON1 in serum. More recently, PON1 has been reported to catalyze the hydrolysis of a variety of aromatic and aliphatic lactones as well as the reverse reaction, lactonization, of γ - and δ -hydroxycarboxylic acids [7]. PON1 is suggested to act as a lactonase or lactonizing enzyme. PON1 as well as PON2 and PON3 metabolize very efficiently 5-hydroxy-eicosatetraenoic acid 1,5-lactone and 4-hydroxydocosahexaenoic acid, which are products of both enzymatic and nonenzymatic oxidation of arachidonic acid and docosahexaenoic acid, respectively. These compounds may represent the PONs' endogenous substrates. PON1 may also possess other biochemical activities, such as metabolizing estrogen esters as well as certain drugs.

2.1.2. Biochemistry of PON2 and PON3

As noted above, PON2 and PON3 lack the ability to degrade organophosphates. However, PON2 and PON3 effectively hydrolyze lactones, and as such also act as lactonases. Of particular note is the ability of PONs, especially PON2 to hydrolyze and thereby inactivate acrylhomoserine lactones, which are quorum sensing signals of pathogenic bacteria [8] (see Section 3.1.6 below for discussion of quorum sensing). PON3 has been shown to hydrolyze bulky drug substrates, such as lovastatin and spironolactone in vitro. However, the in vivo significance of PON3 in the metabolism of these important drugs and thereby in modulating their pharmacological effects remains to be further elucidated.

Less is known regarding the detailed activities of PON2 and PON3 in metabolizing oxidized lipids. Similar to PON1, purified PON2 and PON3 are shown to inhibit lipid peroxidation of LDL in vitro. Since PON2 is an intracellular protein and not present in plasma, it is suggested to act as a protector against cellular oxidative stress. Indeed, intracellular overexpression of PON2 attenuates oxidative stress in various types of vascular cells and inhibits cell-mediated oxidation of LDL [9, 10].

2.2. Molecular Regulation

Among the three members of the mammalian PON family, PON1 has been most extensively studied with regard to the regulation of gene expression. Multiple regulatory mechanisms and signaling pathways have been implicated in the modulation of PON1 gene expression in mammalian systems. PON1 expression is inducible by various agents or biomolecules, including dietary phenolic compounds [11-13], aspirin [14], statin drugs [15], alcohol [16], glucose [17], and interleukin-6 [18]. On the other hand, several compounds, such as bile acids, interleukin-1 β , and tumor necrosis factor- α are found to suppress PON1 gene expression [18, 19].

One study in a cultured human liver cell line demonstrates a critical role for SP1 and protein kinase C in the positive regulation of PON1 gene transcription [20]. The transcription factor SP1 as well as sterol regulatory element-binding protein-2 are involved in statininduced activation of PON1 gene expression in human liver and kidney cell lines [15, 21]. SP1 is also implicated in glucose-induced PON1 gene expression in cultured liver cells [17]. On the other hand, induction of PON1 gene expression by dietary polyphenols, such as resveratrol and quercetin occurs via an arylhydrocarbon receptor (AhR)-dependent mechanism [12, 22]. Nucleotide sequence analysis reveals a xenobiotic response element (XRE)-like sequence within the PON1 promoter, and this XRE-like sequence mediates the induction of PON1 gene transcription by phenolic compounds. A more recent study reports a role for peroxisome proliferator-activated receptor-gamma (PPAR γ) pathway in regulating increased expression of PON1 gene by polyphenols in hepatocytes [13].

PON2 is also inducible by polyphenols. In macrophages, induction of PON2 by polyphenols is shown to occur via PPAR γ - and AP-1-dependent mechanisms [23]. PON2 gene expression is also upregulated by unesterified cholesterol and dexamethasone. Induction of PON2 expression by unesterified cholesterol occurs through activation of the phosphatidylinositol 3-kinase (PI3K) pathway [24]. A putative glucocorticoid response element in the promoter region of PON2 is suggested to mediate the activation of PON2 gene expression by glucocorticoid drugs [25]. The in vivo biological relevance of these two regulatory mechanisms of PON2 expression remains to be determined. Currently, the molecular regulation of PON3 gene expression is largely unknown.

3. Role in Health and Disease

3.1. Animal Studies

3.1.1. Experimental Approaches

Both transgenic overexpression and gene knockout mouse models have been created to investigate the biological functions of PON isozymes in health and disease. In addition, recombinant PON proteins have been made available for studying the protective role of PONs in disease conditions in experimental animals. Mice with homozygous knockout of PON1 or PON2 are born and reproduced adeptly, indicating neither of these two PONs is essential for murine embryonic development. To date, PON3-null mouse model has not been reported in the literature. Viral vector-mediated gene transfer and siRNA-mediated gene knockdown

have also been utilized to elucidate the biological functions of PONs in various experimental models.

Selective chemical inhibitors of PON enzymes would be useful tools for studying the biological activities of PONs. In this context, 2-hydroxyquinoline was claimed to be a selective inhibitor of PON1 [26]; however its usefulness as a specific inhibitor of PON1 in biological systems remains to be determined. Recently, a high throughput PON1 assay has been developed for discovery of small molecule modulators of PON1 activity. With this assay, an effective inhibitor of PON1, designated as BAS03551158, has been identified, which is 10-fold more potent than 2-hydroxyquinoline in inhibiting purified PON1 [27]. But, this compound is less potent in inhibiting serum PON1 than 2-hydroxyquinoline. There are also studies showing selective inhibition of PON1 by negatively charged lipids, such as lysophospholipids [28]. However, the value of these PON1 inhibitors in biological systems remains unclear.

3.1.2. An Overview of the Biological Activities of PONs in Animal Models

As stated earlier, PONs hydrolyze oxidized lipids, and protect against lipid peroxidation as well as cellular oxidative stress. In addition, these enzymes possess other novel biological functions, including anti-inflammation and stimulation of macrophage cholesterol efflux [29, 30]. These findings implicate that PONs may potentially affect a variety of disease processes involving oxidative stress and inflammation, such as atherosclerosis, diabetes and metabolic syndrome, hepatic injury, and infectious diseases. Furthermore, PON1 efficiently hydrolyzes various organophosphates, including pesticides and nerve agents [3], implicating that this PON enzyme may also act as an important defense against organophosphate poisoning. The sections below discuss the role of PONs in a number of pathophysiological conditions in animal models.

3.1.3. Role in Cardiovascular Diseases

The best characterized biological activity of PONs in experimental animals is the suppression of atherogenesis. There is substantial evidence for an important role of all three PON enzymes in protecting against experimental atherosclerosis, vascular inflammation, and oxidative stress [31, 32]. In this regard, targeted disruption of PON1 markedly augments high-fat diet-induced vascular inflammation, oxidative stress, and atherosclerosis. Conversely, either transgenic overexpression of PON1 or viral vector-mediated PON1 gene transfer greatly attenuates the above detrimental effects caused by high-fat diet [33-36]. Furthermore, HDL isolated from the PON1-deficient animals is unable to protect LDL from lipid peroxidation, suggesting that PON1 mediates the antioxidant activities of HDL [33]. Recently, PON1 gene transfer is found to also attenuate vascular oxidative stress and improve vasomotor function in ApoE-deficient mice with pre-existing atherosclerosis, pointing to a therapeutic activity of PON1 in atherogenesis [37].

The protective effects of both PON2 and PON3 in experimental atherosclerosis have been recently demonstrated. Despite lower levels of very low-density lipoprotein (VLDL) and LDL cholesterol, mice deficient in PON2 develop significantly larger atherosclerotic lesions compared with their wild-type counterparts [38]. Three potential mechanisms may account for the enhanced atherosclerotic lesions in PON2-deficient mice. They are: (1) enhanced inflammatory properties of LDL, (2) diminished anti-atherogenic capacity of HDL, and (3) a heightened state of oxidative stress coupled with an exacerbated inflammatory response from PON2-deficient macrophages [38]. In line with these findings, adenovirus-mediated expression of

PON2 protects against the development of atherosclerosis in ApoE-deficient mice [39]. Similarly, adenovirus-mediated gene transfer of human PON3 augments HDL antioxidant effects against lipid peroxidation of LDL, and inhibits the progression of atherosclerosis in ApoEdeficient mice [40]. Transgenic overexpression of human PON3 also attenuates atherosclerosis lesion and vascular inflammation in both wild-type mice fed atherogenic diet and in LDL receptor-deficient mice [41]. Notably, a recent study demonstrates that transgenic overexpression of a human PON gene cluster containing PON1, PON2, and PON3 not only represses atherosclerosis, but also promotes atherosclerotic plaque stability in ApoE-deficient mice [42]. In summary, extensive studies in animal models support a critical role for PONs in protecting against atherosclerosis. However, it remains unknown if PONs also have a beneficial role in other forms of cardiovascular disorders.

3.1.4. Role in Diabetes and Metabolic Syndrome

Diabetes is characterized by increased inflammation and oxidative stress, and decreased PON1 activity. Transgenic overexpression of PON1 attenuates diabetes-induced macrophage oxidative stress, represses diabetes development, and decreases mortality in mice treated with streptozotocin. Conversely, mice lacking PON1 are more susceptible to streptozotocininduced diabetes and oxidative stress [43]. Streptozotocin is a commonly used chemical for inducing type 1 diabetes in experimental animals. It has been suggested that pharmacological and nutritional intervention directed toward increasing PON1 activity may be beneficial in arresting diabetes development and its cardiovascular complications [43]. Recently, the potential role of PON2 in diabetes has been studied in cellular systems under hyperglycemic conditions. It is shown that PON2 decreases high glucose-induced macrophage triglyceride accumulation and oxidative stress by inhibiting NAD(P)H oxidase and diacylglycerol acyltransferase-1 [44]. Hypertriglyceridemia is an independent factor for atherosclerosis, and is also associated with the development of diabetes. As such, accumulation of triglycerides by macrophages under diabetic conditions may promote foam cell formation and contribute to diabetic vascular complications. Thus, augmentation of macrophage PON2 expression may attenuate macrophage atherogenicity, foam cell formation, and the resulting vascular complications in diabetes.

There is also evidence supporting a beneficial role for PONs in metabolic syndrome and associated cardiovascular complications. For instance, transgenic overexpression of human PON3 in mice is shown to decrease adiposity and lower circulating leptin levels, implicating a critical role for PON3 in protecting against obesity [41]. In mice with combined leptin and LDL receptor deficiency, a model of metabolic syndrome, adenovirus-mediated overexpression of human PON1 significantly reduces the total atherosclerotic plaque volume, the volume of plaque macrophages, and the amount of plaque-associated oxidized LDL [45].

3.1.5. Role in Liver Diseases

Liver is the major source of PON synthesis, and PON enzymes may play a protective role in liver injury involving oxidative stress and inflammation [46, 47]. Early studies suggest an association between decreased PON1 activity and development of chronic liver disease induced by carbon tetrachloride [48]. Adenovirus-mediated overexpression of either human PON1 or PON3 in mice attenuates carbon tetrachloride-elicited acute hepatic cell death, lipid peroxidation, and inflammation, indicating a causal role for PON1 and PON3 in inhibiting toxic liver injury [49]. The effects of PONs in protecting against other forms of liver disorders, including alcoholic liver injury, acetaminophen overdose-induced hepatotoxicity, and nonalcoholic fatty liver disease deserve investigations with the use of genetically manipulated animal models.

3.1.6. Role in Infectious Diseases

PONs are able to hydrolyze and thereby inactivate acrylhomoserine lactones, which are quorum sensing signals of pathogenic bacteria [8]. Several lines of evidence suggest a role for PONs in innate immunity and in protection against infections by pathogenic microorganisms, including *Pseudomonas aeruginosa* and *Trypanosoma congolense*.

3.1.6.1. PSEUDOMONAS AERUGINOSA INFECTION

P. aeruginosa is an opportunistic bacterium which causes serious infections in immune compromised individuals, such as AIDS patients. Using small signaling molecules called acrylhomoserine lactones, populations of *P. aeruginosa* can coordinate phenotypic changes, including biofilm formation and virulence factor secretion, a process known as quorum sensing. Interference with quorum sensing has been identified as a potential strategy for the intervention of *P. aeruginosa* infection. Human PON enzymes, especially PON2 are able to effectively hydrolyze and inactivate acrylhomoserine lactones, and as such may play an important role in protecting against *P. aeruginosa* infection. Indeed, PON2 is shown to inactivate the quorum sensing molecule *N*-3-oxododecanoyl homoserine lactone in mouse tracheal epithelia [50]. The lysates of tracheal epithelia from PON2, but not PON1 or PON3 knockout mice have impaired *N*-3-oxododecanoyl homoserine lactone inactivation compared with wild-type mice. Using a quorum sensing reporter strain of *P. aeruginosa*, it is further found that quorum sensing is augmented in PON2-deficient tracheal epithelia as compared with those of wild-type, suggesting that PON2 may play a critical role in airway defense against *P. aeruginosa* infection [50].

An in vitro study demonstrates that expression of any of the three PON isozymes in P. aeruginosa leads to inhibition of biofilm formation and decreased resistance to antibiotics gentamicin and ceftazidima [51]. In an in vitro biofilm model, serum PON1 is also shown to prevent P. aeruginosa quorum sensing and biofilm formation by inactivating N-3oxododecanovl homoserine lactone [52]. These in vitro observations suggest a protective function of PON1 in P. aeruginosa infection. However, PON1 knockout mice are found to be paradoxically more resistant to P. aeruginosa infection as compared with wild-type littermates [52]. The PON1 knockout mice show increased levels of PON2 and PON3 mRNA in epithelial tissues, suggesting a possible compensatory mechanism. A recent study demonstrates that transgenic overexpression of human PON1 in Drosophila melanogaster dramatically decreases the lethality caused by *P. aeruginosa* infection. The protection is dependent on the lactonase activity of PON1 and the specific inactivation of the N-3-oxododecanoyl homoserine lactone, which is critical for the virulence of P. aeruginosa in Drosophila melanogaster [53]. Taken together, these novel findings indicate that PON1 can interfere with quorum sensing in vivo, and may play an important role in innate immune response to quorum sensing-dependent pathogens.

3.1.6.2. TRYPANOSOMA CONGOLENSE INFECTION

PONs may also play a role in protecting against parasitic infections. In vitro studies have suggested that a fraction of human HDL, termed trypanosome lysis factor, can suppress try-

panosome infection. The trypanosome factor fraction consists of PON1 and apolipoprotein A-II. Both PON1 transgenic overexpression and gene knockout mice have been used to investigate the involvement of PON1 in protection against trypanosome infection. When challenged with the same dosage of trypanosomes, mice overexpressing PON1 live significantly longer than wild-type mice, and mice deficient in PON1 live significantly shorter. In contrast, mice overexpressing another HDL-associated protein, apolipoprotein A-II have the same survival as wild-type mice following trypanosome infection [54]. The study also shows that HDL particles containing PON1 and those depleted of PON1 do not differ in their ability to lyse the trypanosome parasites in vitro, suggesting that the in vivo protection by PON1 may be indirect. Hence, the exact mechanism underlying the protective effects of PON1 in trypanosome infection remains to be elucidated.

3.1.7. Role in Organophosphate Poisoning

PON1 effectively hydrolyzes organophosphate pesticides and nerve agents, and as such plays an important role in protecting against organophosphate poisoning. Targeted disruption of PON1 in mice markedly sensitizes the animals to organophosphate-induced toxicity [33]. Administration of either rabbit or human PON1 purified from serum to rats and mice also protects them from organophosphate toxicity. Because human PON1 exhibits limited stability, engineered recombinant PON1 variants with higher activity and increased stability have been developed. Recently, it has been reported that injection of such a recombinant human PON1 variant, designated as rHuPON1_{k192}, protects the PON1^{-/-} mice against up to three median lethal doses of the organophosphate diazoxon [55]. rHuPON1_{k192} exhibits greater hydrolysis of diazoxon, chlorpyrifos-oxon, and paraoxon compared with PON1_{R192}, the more efficient one of the two natural human PON1 alloforms (see Section 3.2.2 and Table 17-1). After injection, rHuPON1k192 is found to be nontoxic in mice, persist in serum for at least two days, and protective against diazoxon both prophylactically and therapeutically [55]. These observations point to the potential value for using the engineered recombinant PON1 for: (1) protection of military personnel or civilians from organophosphate chemical warfare agents and (2) protection of agricultural workers and by-standers from accidental poisoning by organophosphate pesticides. Such engineered recombinant human PON1 variants may also be of value for the intervention of chronic cardiovascular disorders like atherosclerosis considering that this enzyme is important in protecting against atherosclerotic vascular diseases in experimental animals (Section 3.1.3).

3.2. Human Studies and Clinical Perspectives

3.2.1. Study Approaches and General Considerations

The potential involvement of PONs in human diseases has been extensively investigated over the last decade. Numerous studies evaluate changes in the levels or activities of PON enzymes in sera of patients with various disease conditions. Many others involve the epidemiological assessment of the association of PON gene polymorphisms with risk or incidence of particular diseases. There are also studies aiming to link gene polymorphisms to changes of PON activities, and further to risk of disease development. These studies in human subjects have provided important information on the involvement of PONs in disease processes; however, they do not establish a casual role for PONs in disease pathophysiology. Nevertheless, the data obtained from both experimental animal models and human studies have yielded valuable insights into the functions of PON enzymes in human health and disease. The sections below begin with a description of common PON gene polymorphisms, followed by a summary of the major clinical and epidemiological findings on the role of PONs in prevalent human diseases.

3.2.2. PON Gene Polymorphisms

Among the three members of human PONs, PON1 gene polymorphisms and their influence on enzyme activity have received the most extensive studies. PON1 gene has two wellstudied polymorphisms in the coding region, which have been associated with various human diseases. Substitution of glutamine (Gln or Q) by arginine (Arg or R) at position 192, and methionine (Met or M) by leucine (Leu or L) at position 55 of the amino acid sequence of the protein independently influence the PON1 activity, and constitute an important molecular basis for interindividual variability. These two coding region polymorphisms of PON1 are designated as Gln192Arg (or Q192R) and Met55Leu (or M55L), respectively.

The Q192R polymorphism exhibits a striking substrate specific difference in the hydrolytic activity of the enzyme. The 192R alloenzyme shows greater activity toward paraoxon hydrolysis than the 192Q alloenzyme, whereas the 192Q alloenzyme provides greater protection against the accumulation of lipid peroxides in LDL than the R alloenzyme in vitro [56, 57]. However, it should be noted that the in vitro effects of the PON alloenzymes on LDL oxidation may not be translated into the corresponding in vivo activities in human subjects (see Section 3.2.3 for further discussion).

The M55L polymorphism also affects PON1 activity/concentration though less than does the Q192R polymorphism. Individuals carrying a leucine at position 55 (L alloenzyme) have a higher PON1 concentration than those with a methionine (M alloenzyme) at this position. There are also five known polymorphisms located within the PON1 promoter region that have been shown to affect PON1 expression. However, their association with human diseases has not been clearly demonstrated [32, 58]. Table 17-1 summarizes the impact of Q192R and M55L polymorphisms on serum PON1 activity/concentration.

Polymorphism	Effects on Serum PON1 Activity/Concentration
Q192R	RR - high serum PON1 activity QQ - low serum PON1 activity
M55L	LL - high serum PON1 concentration MM - low serum PON1 concentration

Table 17-1. PON1 gene polymorphisms and effects on serum PON1 activity/concentration

Note: See text (Section 3.2.2) for detailed description of the Q192R and M55L polymorphisms of PON1 gene and their effects on PON1 activity/concentration.

A number of polymorphisms have been identified in PON2 gene. Among them substitution of cysteine by serine at position 311 of the amino acid sequence of the protein, designated as Cys311Ser (or C311S) has received most attention. The C311S polymorphism has been found to affect PON2 activity (with the C variant showing impaired lactonase activity) [59], and be associated with the development of certain human diseases. Much less is known about the gene polymorphisms of human PON3 and their association with diseases. A study in an Italian population has identified five point mutations in PON3 gene, with three being silent mutations and two being missense mutations. The two missense mutations give rise to amino acid substitutions at position 311 (S to T) and 324 (G to D) [60]. Currently, the effects of the genetic variations on PON3 activity and the linkage between these variations and diseases remain unknown though one study shows an association of PON3 gene variation with PON1 activity [61]. This suggests a potential cross-effect among PON isozymes.

3.2.3. Role in Human Cardiovascular Diseases

A beneficial role for PON enzymes in atherosclerosis is strongly supported by transgenic overexpression and gene knockout studies in experimental animals (Section 3.1.3). There is also ample evidence supporting a protective effect of PONs, especially PON1 in human coronary heart disease. Many epidemiological studies suggest a correlation between low serum PON1 activity and coronary heart disease. Numerous gene polymorphism studies demonstrate an association between functional polymorphisms of PON1, such as Q192R and M55L, and the development of coronary heart disease though some others report conflicting results [32, 62]. In general, serum PON1 activity is a better predictor of risk for coronary heart disease than PON1 genotype. Therefore, it becomes necessary that when performing case-control studies, the observed differences in enzyme activities are matched by genotypes. In this context, a recent study demonstrates that PON1 genotype exhibits a significant dose-dependent association with decreased levels of serum PON activity (QQ192<QR192<RR192) and with increased levels of systemic oxidative stress (QQ192>QR192>RR192) [63]. Hence, among the three genotypes, OO192 is associated with the lowest levels of PON1 activity and the highest levels of systemic oxidative stress, whereas RR192 exhibits the opposite. Compared with individuals possessing either PON1 RR192 or QR192 genotype, individuals with the OO192 genotype have an increased risk of all-cause mortality and major adverse cardiac events. The incidence of major adverse cardiac events is significantly lower in individuals in the highest PON1 activity quartile compared with those in the lowest activity quartile [63]. These observations are consistent with findings in murine models and provide evidence for a cardiovascular protective function of PON1 in humans.

A more recent study shows that among symptomatic male coronary heart disease patients, carriership of the PON1 Q192 or M55 alleles is associated with an increase of the 10year risk of myocardial infarction and mortality from ischemic heart disease. [64]. Both Q192 and M55 isotypes are associated with lower PON1 activity/concentration (Table 17-1). The data thus support the notion that PON1 activity may predict the adverse coronary events, and that PON1 may be a potential target of secondary prevention of coronary heart disease. In line with a protective role of PON1 in cardiovascular diseases as well as other aging-related disorders (see sections below), a recent meta-analysis of studies involving a total of 5962 subjects reveals that individuals carrying PON1 RR192 and QR192 genotypes are favored in reaching extreme ages, implicating that PON1 may be a longevity gene [65].

Gene polymorphisms of PON2 are also associated with an increased risk of cardiovascular diseases. As noted earlier in Section 3.2.2, the C311S polymorphism of PON2 affects PON2 activity, with the C isoform showing impaired lactonase activity [59]. A recent study in an Italian population reports that patients with at least one PON2 C allele represent a category of subjects at a higher risk for the development of acute myocardial infarction with a worse prognosis [66]. The genotype with the C allele of the PON2 gene is also shown to be a risk factor for large vessel disease-associated stroke in a Polish population, further supporting a vasoprotective function of PON2 in humans [67]. These observations, however, contradict an early study showing S allele carriers of the PON2 C311S polymorphism at a greater risk of developing coronary heart disease than C allele carriers [68].

3.2.4. Role in Human Diabetes and Metabolic Syndrome

There are a number of clinical studies showing a significant decrease in serum PON1 activity in diabetic patients as compared with control subjects. The decrease in serum PON1 activity is also associated with the development of diabetic complications. A more recent study demonstrates that lower serum PON1 activity is related to higher plasma C-reactive protein and obesity [69]. These observations indicate a potential beneficial role for PON1 in diabetes and metabolic syndrome, which is also in line with findings in animal models (Section 3.1.4). However, epidemiological studies on the association of gene polymorphisms of PON1 with diabetes have produced inconsistent results. Some studies demonstrate that the PON1 genotypes causing decreased PON1 activity (e.g., PON1-QQ192 and PON1-MM55) are associated with poorer diabetes control and complications [70]. In contrast, other studies report that the PON1-192R allele causing increased PON activity can be an independent cardiovascular risk factor in type 2 diabetic patients [71].

As a further line of supportive evidence for a beneficial effect of PON1 in diabetes, the T allele of PON1 promoter polymorphism (-107C>T) causing reduced expression of PON1 is reported to be associated with lower serum levels of the enzyme and an increased risk for coronary heart disease in type 2 diabetic patients [72]. Moreover, the lower expressor genotype of PON1 (-107TT) is also associated with higher blood glucose concentration in non-diabetic subjects, suggesting that this polymorphism may predispose the individuals to insulin resistance [73]. Thus, a majority of the clinical and epidemiological findings support the view that PON1 may act as a potential protector in diabetes and its complications, as well as in metabolic syndrome.

3.2.5. Role in Human Neurological Diseases

Clinical and epidemiological studies have implicated PONs in various neurological disorders, including Alzheimer's disease, Parkinson's disease, amyotrophic lateral sclerosis, multiple sclerosis, schizophrenia, and anxiety. A decrease in the serum levels of PON1 activity is often observed in patients with Alzheimer's disease compared with control subjects. Some studies report that the R allele of PON1 O192R polymorphism is associated with an increased serum PON1 activity and a decreased risk of developing Alzheimer's disease [74, 75]. Among the Alzheimer's patients, individuals who responded to cholinesterase inhibitors (drugs used for the treatment of Alzheimer's disease) had a significantly higher frequency in the R allele compared to non-responders, suggesting that PON1 may facilitate response to drug therapy [76]. The MM genotype of PON1 L55M polymorphism (causing decreased serum levels of PON1) is also associated with an increased risk of Alzheimer's disease, and correlated with the pathophysiological changes underlying the disease [77]. Taken together, these observations indicate that PON1 may be protective against Alzheimer's disease. However, there are also studies showing no association between PON1 gene polymorphisms and susceptibility to Alzheimer's disease [78]. In addition to PON1, a study in a Chinese population shows that the C allele of the C311S polymorphism of PON2 is associated with Alzhei-
mer's disease [79]. As noted earlier in Section 3.2.2, the C allele displays a decreased lactonase activity of PON2.

A lower PON1 activity is frequently observed in the sera of patients with Parkinson's disease as compared with control subjects. PON1 gene polymorphisms are implicated as risk factors for Parkinson's disease, but case-control studies have generated controversial results. A meta-analysis of the studies relating PON1 M55L and PON1 Q192R polymorphisms to risk of developing Parkinson's disease reveals that the MM55 genotype of PON1 (showing decreased serum levels of PON1) is associated with an increased risk of developing Parkinson's disease compared with the LL55 genotype. However, there is no association between the PON1 Q192R polymorphism and susceptibility to Parkinson's disease [80].

Differences in PON1 polymorphism frequencies and enzyme activity/levels have been investigated in several other neurological disorders. Polymorphisms in PON gene cluster are associated with amyotrophic lateral sclerosis [81]. Decreased serum levels/activity of PON1 are observed in patients with multiple sclerosis [82], schizophrenia [83], or anxiety disorder [84], supporting the notion that oxidative stress-mediated damage may contribute to the pathogenesis of these neurological diseases. Notably, PON1 expression is upregulated in the fetal liver of Down's syndrome patients, which may provide a clue as to why these patients have a lower incidence of atherosclerotic vascular disease than the general population [85].

3.2.6. Role in Human Liver Diseases

The fact that liver is the major synthesis organ for PON1 has prompted clinical investigations into the potential role of this protein in various human liver disorders involving oxidative stress and inflammation. Early studies demonstrated a significant decrease in serum PON1 activity in patients with liver cirrhosis, which has been confirmed by subsequent studies in patients with varying degrees of chronic liver diseases, including chronic hepatitis, chronic alcoholic liver injury, and nonalcoholic fatty liver disease [86-88]. Indeed, serum PON1 activity can reflect liver function, and as such has been proposed as a biomarker for monitoring severity and progression of chronic liver diseases [89]. The association of PON1 gene polymorphisms with chronic hepatitis C virus (HCV) infection has been investigated in one study. It shows that patients with HCV-related chronic hepatitis have a higher frequency of the RR isoform of the Q192R polymorphism than healthy subjects [87]. The plasma concentrations of peroxides are also higher in patients with the hepatitis than control subjects. As noted earlier, the PON1 RR192 genotype is associated with an increased serum activity of the enzyme (Table 17-1). Thus, the significance of above gene polymorphism study remains unclear. Further studies on the association of PON gene polymorphisms with chronic liver diseases will provide insights into the potential impact of PON genotypes on susceptibility to chronic liver diseases. Such studies may lead to identification of PON-associated risk factors for developing chronic liver diseases.

3.2.7. Role in Human Cancer

Oxidative stress and inflammation are critical events underlying carcinogenesis. By suppressing oxidative stress and inflammation, PON enzymes may protect against cancer development. However, to date little is known about the role of PONs in carcinogenesis in animal models. In contrast, there have been many studies in human subjects, suggesting a potential protective effect of PON1 in the development of different types of human cancer. A lower serum PON1 activity has been consistently observed in patients with various types of cancer compared with control subjects. Gene polymorphisms of PON1 that cause decreased serum PON1 activity are associated with increased risk of developing malignancies, including lung cancer [90], breast cancer [91], ovarian epithelial carcinoma [92], prostate cancer [93], and brain cancer [94]. The causative role of PON1 as well as the other two members of PON family in cancer development warrants further investigations in both human subjects and experimental animals.

3.2.8. Role in Other Human Diseases

PON1 has also been implicated in many other human disorders based on observations from clinical and epidemiological studies. These include viral infections (e.g., HIV infection, hepatitis B infection) [95], *Helicobacter pylori* infection [96], idiopathic chronic pancreatitis [97], progressive IgA nephropathy [98], age-related macular degeneration [99], erectile dys-function [100], and organophosphate poisoning [101]. The exact role of PON1 as well as other PON isozymes in the pathophysiology of the above disease conditions awaits further confirmation by additional well-designed studies.

4. Conclusion and Future Directions

PONs are multitasking antioxidant enzymes that have a beneficial role in a variety of oxidative stress- and inflammation-associated pathophysiological processes in experimental animals. Increasing evidence from studies in human subjects also suggests PONs as protective factors in many diseases, including atherosclerotic coronary heart disease, diabetes, neurodegeneration, and cancer. Currently, there is a great need to translate the important discoveries on PON isozymes into effective interventional strategies to prevent and treat human diseases that involve a PON-dependent mechanism. In this regard, future studies should focus on developing effective pharmacological inducers of PONs and evaluating their effectiveness in upregulating PON enzymes and in protecting against oxidative and inflammatory tissue degeneration underlying common human disorders.

5. References

- [1] Aldridge WN. Serum esterases. I. Two types of esterase (A and B) hydrolysing pnitrophenyl acetate, propionate and butyrate, and a method for their determination. *Biochem. J.* 1953;53:110-117.
- [2] Aldridge WN. Serum esterases. II. An enzyme hydrolysing diethyl p-nitrophenyl phosphate (E600) and its identity with the A-esterase of mammalian sera. *Biochem. J.* 1953;53:117-124.
- [3] Chambers JE. PON1 multitasks to protect health. *Proc. Natl. Acad. Sci. USA* 2008;105: 12639-12640.
- [4] Camps J, Marsillach J, Joven J. The paraoxonases: role in human diseases and methodological difficulties in measurement. *Crit. Rev. Clin. Lab. Sci.* 2009;46:83-106.
- [5] Aviram M, Billecke S, Sorenson R, Bisgaier C, Newton R, Rosenblat M, *et al.* Paraoxonase active site required for protection against LDL oxidation involves its free

sulfhydryl group and is different from that required for its arylesterase/paraoxonase activities: selective action of human paraoxonase allozymes Q and R. *Arterioscler*. *Thromb. Vasc. Biol.* 1998;18:1617-1624.

- [6] Rozenberg O, Aviram M. S-Glutathionylation regulates HDL-associated paraoxonase 1 (PON1) activity. *Biochem. Biophys. Res. Commun.* 2006;351:492-498.
- [7] Draganov DI, Teiber JF, Speelman A, Osawa Y, Sunahara R, La Du BN. Human paraoxonases (PON1, PON2, and PON3) are lactonases with overlapping and distinct substrate specificities. *J. Lipid Res.* 2005;46:1239-1247.
- [8] Teiber JF, Horke S, Haines DC, Chowdhary PK, Xiao J, Kramer GL, et al. Dominant role of paraoxonases in inactivation of the Pseudomonas aeruginosa quorum-sensing signal N-(3-oxododecanoyl)-L-homoserine lactone. Infect. Immun. 2008;76:2512-2519.
- [9] Horke S, Witte I, Wilgenbus P, Kruger M, Strand D, Forstermann U. Paraoxonase-2 reduces oxidative stress in vascular cells and decreases endoplasmic reticulum stress-induced caspase activation. *Circulation* 2007;115:2055-2064.
- [10] Ng CJ, Wadleigh DJ, Gangopadhyay A, Hama S, Grijalva VR, Navab M, *et al.* Paraoxonase-2 is a ubiquitously expressed protein with antioxidant properties and is capable of preventing cell-mediated oxidative modification of low density lipoprotein. *J. Biol. Chem.* 2001;276:4444-44449.
- [11] Gong M, Garige M, Varatharajalu R, Marmillot P, Gottipatti C, Leckey LC, *et al.* Quercetin up-regulates paraoxonase 1 gene expression with concomitant protection against LDL oxidation. *Biochem. Biophys. Res. Commun.* 2009;379:1001-1004.
- [12] Gouedard C, Barouki R, Morel Y. Induction of the paraoxonase-1 gene expression by resveratrol. *Arterioscler. Thromb. Vasc. Biol.* 2004;24:2378-2383.
- [13] Khateeb J, Gantman A, Kreitenberg AJ, Aviram M, Fuhrman B. Paraoxonase 1 (PON1) expression in hepatocytes is upregulated by pomegranate polyphenols: a role for PPAR-gamma pathway. *Atherosclerosis* 2010;208:119-125.
- [14] Jaichander P, Selvarajan K, Garelnabi M, Parthasarathy S. Induction of paraoxonase 1 and apolipoprotein A-I gene expression by aspirin. J. Lipid Res. 2008;49:2142-2148.
- [15] Deakin S, Leviev I, Guernier S, James RW. Simvastatin modulates expression of the PON1 gene and increases serum paraoxonase: a role for sterol regulatory elementbinding protein-2. *Arterioscler. Thromb. Vasc. Biol.* 2003;23:2083-2089.
- [16] Rao MN, Marmillot P, Gong M, Palmer DA, Seeff LB, Strader DB, et al. Light, but not heavy alcohol drinking, stimulates paraoxonase by upregulating liver mRNA in rats and humans. *Metabolism* 2003;52:1287-1294.
- [17] Ikeda Y, Suehiro T, Arii K, Kumon Y, Hashimoto K. High glucose induces transactivation of the human paraoxonase 1 gene in hepatocytes. *Metabolism* 2008;57: 1725-1732.
- [18] Kumon Y, Suehiro T, Ikeda Y, Hashimoto K. Human paraoxonase-1 gene expression by HepG2 cells is downregulated by interleukin-1beta and tumor necrosis factor-alpha, but is upregulated by interleukin-6. *Life Sci.* 2003;73:2807-2815.
- [19] Gutierrez A, Ratliff EP, Andres AM, Huang X, McKeehan WL, Davis RA. Bile acids decrease hepatic paraoxonase 1 expression and plasma high-density lipoprotein levels via FXR-mediated signaling of FGFR4. *Arterioscler. Thromb. Vasc. Biol.* 2006;26: 301-306.
- [20] Osaki F, Ikeda Y, Suehiro T, Ota K, Tsuzura S, Arii K, *et al.* Roles of Sp1 and protein kinase C in regulation of human serum paraoxonase 1 (PON1) gene transcription in

HepG2 cells. Atherosclerosis 2004;176:279-287.

- [21] Ota K, Suehiro T, Arii K, Ikeda Y, Kumon Y, Osaki F, *et al.* Effect of pitavastatin on transactivation of human serum paraoxonase 1 gene. *Metabolism* 2005;54:142-150.
- [22] Gouedard C, Barouki R, Morel Y. Dietary polyphenols increase paraoxonase 1 gene expression by an aryl hydrocarbon receptor-dependent mechanism. *Mol. Cell Biol.* 2004;24:5209-5222.
- [23] Shiner M, Fuhrman B, Aviram M. Macrophage paraoxonase 2 (PON2) expression is up-regulated by pomegranate juice phenolic anti-oxidants via PPAR gamma and AP-1 pathway activation. *Atherosclerosis* 2007;195:313-321.
- [24] Shiner M, Fuhrman B, Aviram M. Macrophage paraoxonase 2 (PON2) expression is upregulated by unesterified cholesterol through activation of the phosphatidylinositol 3-kinase (PI3K) pathway. *Biol. Chem.* 2007;388:1353-1358.
- [25] Lim JA, Kim SH. Transcriptional activation of an anti-oxidant mouse Pon2 gene by dexamethasone. *BMB Rep.* 2009;42:421-426.
- [26] Ahmad S, Carter JJ, Scott JE. A homogeneous cell-based assay for measurement of endogenous paraoxonase 1 activity. *Anal. Biochem.* 2010;400:1-9.
- [27] Graves TL, Scott JE. A High Throughput Serum Paraoxonase Assay for Discovery of Small Molecule Modulators of PON1 Activity. *Curr. Chem. Genomics* 2008;2:51-61.
- [28] Nguyen SD, Sok DE. Preferential inhibition of paraoxonase activity of human paraoxonase 1 by negatively charged lipids. *J. Lipid Res.* 2004;45:2211-2220.
- [29] Mackness B, Mackness M. Anti-inflammatory properties of paraoxonase-1 in atherosclerosis. *Adv. Exp. Med. Biol.* 2010;660:143-151.
- [30] Rosenblat M, Coleman R, Reddy ST, Aviram M. Paraoxonase 2 attenuates macrophage triglyceride accumulation via inhibition of diacylglycerol acyltransferase 1. J. Lipid Res. 2009;50:870-879.
- [31] Aviram M, Rosenblat M. Paraoxonases and cardiovascular diseases: pharmacological and nutritional influences. *Curr. Opin. Lipidol.* 2005;16:393-399.
- [32] Seo D, Goldschmidt-Clermont P. The paraoxonase gene family and atherosclerosis. *Curr. Atheroscler. Rep.* 2009;11:182-187.
- [33] Shih DM, Gu L, Xia YR, Navab M, Li WF, Hama S, *et al.* Mice lacking serum paraoxonase are susceptible to organophosphate toxicity and atherosclerosis. *Nature* 1998;394:284-287.
- [34] Tward A, Xia YR, Wang XP, Shi YS, Park C, Castellani LW, *et al.* Decreased atherosclerotic lesion formation in human serum paraoxonase transgenic mice. *Circulation* 2002;106:484-490.
- [35] Miyoshi M, Nakano Y, Sakaguchi T, Ogi H, Oda N, Suenari K, et al. Gene delivery of paraoxonase-1 inhibits neointimal hyperplasia after arterial balloon-injury in rabbits fed a high-fat diet. *Hypertens. Res.* 2007;30:85-91.
- [36] Ng DS, Chu T, Esposito B, Hui P, Connelly PW, Gross PL. Paraoxonase-1 deficiency in mice predisposes to vascular inflammation, oxidative stress, and thrombogenicity in the absence of hyperlipidemia. *Cardiovasc. Pathol.* 2008;17:226-232.
- [37] Guns PJ, Van Assche T, Verreth W, Fransen P, Mackness B, Mackness M, *et al.* Paraoxonase 1 gene transfer lowers vascular oxidative stress and improves vasomotor function in apolipoprotein E-deficient mice with pre-existing atherosclerosis. *Br. J. Pharmacol.* 2008;153:508-516.
- [38] Ng CJ, Bourquard N, Grijalva V, Hama S, Shih DM, Navab M, et al. Paraoxonase-2

deficiency aggravates atherosclerosis in mice despite lower apolipoprotein-Bcontaining lipoproteins: anti-atherogenic role for paraoxonase-2. *J. Biol. Chem.* 2006; 281:29491-29500.

- [39] Ng CJ, Hama SY, Bourquard N, Navab M, Reddy ST. Adenovirus mediated expression of human paraoxonase 2 protects against the development of atherosclerosis in apolipoprotein E-deficient mice. *Mol. Genet. Metab.* 2006;89:368-373.
- [40] Ng CJ, Bourquard N, Hama SY, Shih D, Grijalva VR, Navab M, et al. Adenovirusmediated expression of human paraoxonase 3 protects against the progression of atherosclerosis in apolipoprotein E-deficient mice. Arterioscler. Thromb. Vasc. Biol. 2007;27:1368-1374.
- [41] Shih DM, Xia YR, Wang XP, Wang SS, Bourquard N, Fogelman AM, et al. Decreased obesity and atherosclerosis in human paraoxonase 3 transgenic mice. Circ. Res. 2007; 100:1200-1207.
- [42] She ZG, Zheng W, Wei YS, Chen HZ, Wang AB, Li HL, et al. Human paraoxonase gene cluster transgenic overexpression represses atherogenesis and promotes atherosclerotic plaque stability in ApoE-null mice. Circ. Res. 2009;104:1160-1168.
- [43] Rozenberg O, Shiner M, Aviram M, Hayek T. Paraoxonase 1 (PON1) attenuates diabetes development in mice through its antioxidative properties. *Free Radic. Biol. Med.* 2008;44:1951-1959.
- [44] Meilin E, Aviram M, Hayek T. Paraoxonase 2 (PON2) decreases high glucose-induced macrophage triglycerides (TG) accumulation, via inhibition of NADPH-oxidase and DGAT1 activity: studies in PON2-deficient mice. *Atherosclerosis* 2010;208:390-395.
- [45] Mackness B, Quarck R, Verreth W, Mackness M, Holvoet P. Human paraoxonase-1 overexpression inhibits atherosclerosis in a mouse model of metabolic syndrome. *Arterioscler. Thromb. Vasc. Biol.* 2006;26:1545-1550.
- [46] Marsillach J, Camps J, Ferre N, Beltran R, Rull A, Mackness B, et al. Paraoxonase-1 is related to inflammation, fibrosis and PPAR delta in experimental liver disease. BMC Gastroenterol. 2009;9:3.
- [47] Camps J, Marsillach J, Rull A, Alonso-Villaverde C, Joven J. Interrelationships between paraoxonase-1 and monocyte chemoattractant protein-1 in the regulation of hepatic inflammation. *Adv. Exp. Med. Biol.* 2010;660:5-18.
- [48] Ferre N, Camps J, Cabre M, Paul A, Joven J. Hepatic paraoxonase activity alterations and free radical production in rats with experimental cirrhosis. *Metabolism* 2001; 50:997-1000.
- [49] Peng W, Zhang C, Lv H, Zhu J, Zang Y, Pang X, *et al.* Comparative evaluation of the protective potentials of human paraoxonase 1 and 3 against CCl₄-induced liver injury. *Toxicol. Lett.* 2010;193:159-166.
- [50] Stoltz DA, Ozer EA, Ng CJ, Yu JM, Reddy ST, Lusis AJ, *et al.* Paraoxonase-2 deficiency enhances Pseudomonas aeruginosa quorum sensing in murine tracheal epithelia. *Am. J. Physiol. Lung. Cell. Mol. Physiol.* 2007;292:L852-860.
- [51] Ma F, Wang Y, Zhang Y, Xiong N, Yang B, Chen S. Heterologous expression of human paraoxonases in Pseudomonas aeruginosa inhibits biofilm formation and decreases antibiotic resistance. *Appl. Microbiol. Biotechnol.* 2009;83:135-141.
- [52] Ozer EA, Pezzulo A, Shih DM, Chun C, Furlong C, Lusis AJ, et al. Human and murine paraoxonase 1 are host modulators of Pseudomonas aeruginosa quorum-sensing. FEMS Microbiol. Lett. 2005;253:29-37.

- [53] Stoltz DA, Ozer EA, Taft PJ, Barry M, Liu L, Kiss PJ, et al. Drosophila are protected from Pseudomonas aeruginosa lethality by transgenic expression of paraoxonase-1. J. Clin. Invest. 2008;118:3123-3131.
- [54] Bhasin KK, Yu JM, Tward A, Shih D, Campbell DA, Lusis AJ. Trypanosoma congolense: paraoxonase 1 prolongs survival of infected mice. *Exp. Parasitol.* 2006; 114:240-245.
- [55] Stevens RC, Suzuki SM, Cole TB, Park SS, Richter RJ, Furlong CE. Engineered recombinant human paraoxonase 1 (rHuPON1) purified from Escherichia coli protects against organophosphate poisoning. *Proc. Natl. Acad. Sci. USA* 2008;105:12780-12784.
- [56] Mackness MI, Arrol S, Mackness B, Durrington PN. Alloenzymes of paraoxonase and effectiveness of high-density lipoproteins in protecting low-density lipoprotein against lipid peroxidation. *Lancet* 1997;349:851-852.
- [57] Mackness B, Durrington PN, Mackness MI. Polymorphisms of paraoxonase genes and low-density lipoprotein lipid peroxidation. *Lancet* 1999;353:468-469.
- [58] Goswami B, Tayal D, Gupta N, Mallika V. Paraoxonase: a multifaceted biomolecule. *Clin. Chim. Acta* 2009;410:1-12.
- [59] Stoltz DA, Ozer EA, Recker TJ, Estin M, Yang X, Shih DM, et al. A common mutation in paraoxonase-2 results in impaired lactonase activity. J. Biol. Chem. 2009; 284:35564-35571.
- [60] Campo S, Sardo AM, Campo GM, Avenoso A, Castaldo M, D'Ascola A, et al. Identification of paraoxonase 3 gene (PON3) missense mutations in a population of southern Italy. *Mutat. Res.* 2004;546:75-80.
- [61] Sanghera DK, Manzi S, Minster RL, Shaw P, Kao A, Bontempo F, *et al.* Genetic variation in the paraoxonase-3 (PON3) gene is associated with serum PON1 activity. *Ann. Hum. Genet.* 2008;72:72-81.
- [62] Furlong CE, Suzuki SM, Stevens RC, Marsillach J, Richter RJ, Jarvik GP, et al. Human PON1, a biomarker of risk of disease and exposure. *Chem. Biol. Interact.* 2010; 187:355-361.
- [63] Bhattacharyya T, Nicholls SJ, Topol EJ, Zhang R, Yang X, Schmitt D, *et al.* Relationship of paraoxonase 1 (PON1) gene polymorphisms and functional activity with systemic oxidative stress and cardiovascular risk. *JAMA* 2008;299:1265-1276.
- [64] Regieli JJ, Jukema JW, Doevendans PA, Zwinderman AH, Kastelein JJ, Grobbee DE, *et al.* Paraoxonase variants relate to 10-year risk in coronary artery disease: impact of a high-density lipoprotein-bound antioxidant in secondary prevention. *J. Am. Coll. Cardiol.* 2009;54:1238-1245.
- [65] Lescai F, Marchegiani F, Franceschi C. PON1 is a longevity gene: results of a metaanalysis. *Ageing Res. Rev.* 2009;8:277-284.
- [66] Marchegiani F, Spazzafumo L, Provinciali M, Cardelli M, Olivieri F, Franceschi C, et al. Paraoxonase2 C311S polymorphism and low levels of HDL contribute to a higher mortality risk after acute myocardial infarction in elderly patients. *Mol. Genet. Metab.* 2009;98:314-318.
- [67] Slowik A, Wloch D, Szermer P, Wolkow P, Malecki M, Pera J, *et al.* Paraoxonase 2 gene C311S polymorphism is associated with a risk of large vessel disease stroke in a Polish population. *Cerebrovasc. Dis.* 2007;23:395-400.
- [68] Sanghera DK, Aston CE, Saha N, Kamboh MI. DNA polymorphisms in two

paraoxonase genes (PON1 and PON2) are associated with the risk of coronary heart disease. *Am. J. Hum. Genet.* 1998;62:36-44.

- [69] Dullaart RP, de Vries R, Sluiter WJ, Voorbij HA. High plasma C-reactive protein (CRP) is related to low paraoxonase-I (PON-I) activity independently of high leptin and low adiponectin in type 2 diabetes mellitus. *Clin. Endocrinol.* (Oxf) 2009;70:221-226.
- [70] Flekac M, Skrha J, Zidkova K, Lacinova Z, Hilgertova J. Paraoxonase 1 gene polymorphisms and enzyme activities in diabetes mellitus. *Physiol. Res.* 2008;57:717-726.
- [71] van den Berg SW, Jansen EH, Kruijshoop M, Beekhof PK, Blaak E, van der Kallen CJ, et al. Paraoxonase 1 phenotype distribution and activity differs in subjects with newly diagnosed Type 2 diabetes (the CODAM Study). *Diabet. Med.* 2008;25:186-193.
- [72] James RW, Leviev I, Ruiz J, Passa P, Froguel P, Garin MC. Promoter polymorphism T(-107)C of the paraoxonase PON1 gene is a risk factor for coronary heart disease in type 2 diabetic patients. *Diabetes* 2000;49:1390-1393.
- [73] Leviev I, Kalix B, Brulhart Meynet MC, James RW. The paraoxonase PON1 promoter polymorphism C(-107)T is associated with increased serum glucose concentrations in non-diabetic patients. *Diabetologia* 2001;44:1177-1183.
- [74] Scacchi R, Gambina G, Martini MC, Broggio E, Vilardo T, Corbo RM. Different pattern of association of paraoxonase Gln192→Arg polymorphism with sporadic lateonset Alzheimer's disease and coronary artery disease. *Neurosci. Lett.* 2003;339:17-20.
- [75] He XM, Zhang ZX, Zhang JW, Zhou YT, Tang MN, Wu CB, *et al.* Gln192Arg polymorphism in paraoxonase 1 gene is associated with Alzheimer disease in a Chinese Han ethnic population. *Chin. Med. J.* (Engl) 2006;119:1204-1209.
- [76] Pola R, Flex A, Ciaburri M, Rovella E, Valiani A, Reali G, et al. Responsiveness to cholinesterase inhibitors in Alzheimer's disease: a possible role for the 192 Q/R polymorphism of the PON-1 gene. *Neurosci. Lett.* 2005;382:338-341.
- [77] Leduc V, Poirier J. Polymorphisms at the paraoxonase 1 L55M and Q192R loci affect the pathophysiology of Alzheimer's disease: emphasis on the cholinergic system and beta-amyloid levels. *Neurodegener. Dis.* 2008;5:225-227.
- [78] Cellini E, Tedde A, Bagnoli S, Nacmias B, Piacentini S, Bessi V, et al. Association analysis of the paraoxonase-1 gene with Alzheimer's disease. *Neurosci. Lett.* 2006;408: 199-202.
- [79] Shi J, Zhang S, Tang M, Liu X, Li T, Han H, *et al.* Possible association between Cys311Ser polymorphism of paraoxonase 2 gene and late-onset Alzheimer's disease in Chinese. *Brain Res. Mol. Brain Res.* 2004;120:201-204.
- [80] Zintzaras E, Hadjigeorgiou GM. Association of paraoxonase 1 gene polymorphisms with risk of Parkinson's disease: a meta-analysis. *J. Hum. Genet.* 2004;49:474-481.
- [81] Slowik A, Tomik B, Wolkow PP, Partyka D, Turaj W, Malecki MT, et al. Paraoxonase gene polymorphisms and sporadic ALS. *Neurology* 2006;67:766-770.
- [82] Ferretti G, Bacchetti T, Principi F, Di Ludovico F, Viti B, Angeleri VA, et al. Increased levels of lipid hydroperoxides in plasma of patients with multiple sclerosis: a relationship with paraoxonase activity. *Mult. Scler.* 2005;11:677-682.
- [83] Kucukali CI, Aydin M, Ozkok E, Orhan N, Cakir U, Kilic G, et al. Paraoxonase-1 55/192 genotypes in schizophrenic patients and their relatives in Turkish population. *Psychiatr. Genet.* 2008;18:289-294.

- [84] Sklan EH, Lowenthal A, Korner M, Ritov Y, Landers DM, Rankinen T, et al. Acetylcholinesterase/paraoxonase genotype and expression predict anxiety scores in Health, Risk Factors, Exercise Training, and Genetics study. Proc. Natl. Acad. Sci. USA 2004;101:5512-5517.
- [85] Janel N, Christophe O, Ait Yahya-Graison E, Hamelet J, Paly E, Prieur M, et al. Paraoxonase-1 expression is up-regulated in Down syndrome fetal liver. Biochem. Biophys. Res. Commun. 2006;346:1303-1306.
- [86] Baskol M, Baskol G, Deniz K, Ozbakir O, Yucesoy M. A new marker for lipid peroxidation: serum paraoxonase activity in non-alcoholic steatohepatitis. *Turk. J. Gastroenterol.* 2005;16:119-123.
- [87] Ferre N, Marsillach J, Camps J, Rull A, Coll B, Tous M, et al. Genetic association of paraoxonase-1 polymorphisms and chronic hepatitis C virus infection. *Clin. Chim. Acta* 2005;361:206-210.
- [88] Prakash M, Shetty JK, Tripathy S, Verma M, Vasudev S, Bhandary PV. Serum paraoxonase in alcohol abusers associated with alcoholic liver disease. *Clin. Chim. Acta* 2007;378:232-234.
- [89] Camps J, Marsillach J, Joven J. Measurement of serum paraoxonase-1 activity in the evaluation of liver function. *World J. Gastroenterol.* 2009;15:1929-1933.
- [90] Lee CH, Lee KY, Choe KH, Hong YC, Kim YD, Kang JW, *et al.* Effects of oxidative DNA damage induced by polycyclic aromatic hydrocarbons and genetic polymorphism of the paraoxonase-1 (PON1) gene on lung cancer. *J. Prev. Med. Public Health* 2005; 38:345-350.
- [91] Stevens VL, Rodriguez C, Pavluck AL, Thun MJ, Calle EE. Association of polymorphisms in the paraoxonase 1 gene with breast cancer incidence in the CPS-II Nutrition Cohort. *Cancer Epidemiol. Biomarkers Prev.* 2006;15:1226-1228.
- [92] Arpaci A, Gormus U, Dalan B, Berkman S, Isbir T. Investigation of PON1 192 and PON1 55 polymorphisms in ovarian cancer patients in Turkish population. *In Vivo* 2009;23:421-424.
- [93] Antognelli C, Mearini L, Talesa VN, Giannantoni A, Mearini E. Association of CYP17, GSTP1, and PON1 polymorphisms with the risk of prostate cancer. *Prostate* 2005;63:240-251.
- [94] Searles Nielsen S, Mueller BA, De Roos AJ, Viernes HM, Farin FM, Checkoway H. Risk of brain tumors in children and susceptibility to organophosphorus insecticides: the potential role of paraoxonase (PON1). *Environ. Health Perspect.* 2005;113:909-913.
- [95] Parra S, Marsillach J, Aragones G, Beltran R, Montero M, Coll B, *et al.* Paraoxonase-1 gene haplotypes are associated with metabolic disturbances, atherosclerosis, and immunologic outcome in HIV-infected patients. *J. Infect. Dis.* 2010;201:627-634.
- [96] Aslan M, Nazligul Y, Horoz M, Bolukbas C, Bolukbas FF, Gur M, et al. Serum paraoxonase-1 activity in Helicobacter pylori infected subjects. Atherosclerosis 2008; 196:270-274.
- [97] Verlaan M, Harbers EG, Pap A, Jansen JB, Peters WH, Drenth JP. Paraoxonase 1-192Q allele is a risk factor for idiopathic chronic pancreatitis. *Mol. Diagn.* 2005;9:9-15.
- [98] Kovacs TJ, Harris S, Vas TK, Seres I, Short CD, Wittmann IK, et al. Paraoxonase gene polymorphism and serum activity in progressive IgA nephropathy. J. Nephrol. 2006; 19:732-738.

- [99] Pauer GJ, Sturgill GM, Peachey NS, Hagstrom SA. Protective effect of paraoxonase 1 gene variant Gln192Arg in age-related macular degeneration. *Am. J. Ophthalmol.* 2010;149:513-522.
- [100] Ciftci H, Yeni E, Savas M, Verit A, Celik H. Paraoxonase activity in patients with erectile dysfunction. *Int. J. Impot. Res.* 2007;19:517-520.
- [101] Mohamed Ali S, Chia SE. Interethnic variability of plasma paraoxonase (PON1) activity towards organophosphates and PON1 polymorphisms among Asian populations: a short review. *Ind. Health* 2008;46:309-317.

Chapter 18

Metal Ion Sequestering Proteins

Abstract

Metal ions such as iron and copper are essential elements, yet dysregulation of their homeostasis results in tissue injury due partly to their pro-oxidant properties. Mammals have evolved a number of mechanisms to maintain metal ion homeostasis, among which are ferritin and metallothionein (MT) that possess the ability to sequester metal ions in addition to other biological activities. In mammals, ferritin is a 24-subunit intracellular protein whose principal role within cells is the storage of iron in a nontoxic, but bioavailable form. Studies in experimental models demonstrate that ferritin plays a role in protecting against pathophysiological conditions involving oxidative stress, such as myocardial ischemia-reperfusion injury and neurodegeneration. A number of ferritin genetic disorders have been reported in human subjects with iron-facilitated oxidative stress being a contributing factor to the disease pathogenesis. MT refers to a superfamily of small molecular mass proteins with high cysteine content and the capacity to bind various heavy metals, including copper, zinc, cadmium, and mercury. MTs regulate copper and zinc homeostasis. They are also important defenses against heavy metal toxicity as well as oxidative and inflammatory stress. Extensive studies in animal models demonstrate an important protective function for MTs in a number of disease conditions, such as cardiovascular disorders, diabetes, neurodegeneration, and cancer. Studies in human subjects also support a potential involvement of MTs in certain disease processes, including atherosclerosis and diabetes

1. Introduction

Metal ions, including copper and iron ions are essential elements in mammals. For example, these metal ions form the active centers of a variety of important enzymes in mammalian cells or tissues. However, these metals, particularly copper and iron when exist as free ions can also cause detrimental effects. Three mechanisms are involved in the detrimental effects mediated by copper and iron ions. They are summarized as following.

• Participation in Fenton/Fenton-type reactions to produce the extremely reactive hydroxyl radical (Section 2.3.3 of Chapter 1)

- Catalysis of auto-oxidation of biomolecules (e.g., ascorbate, catecholamines) to form reactive species, such as free radicals and reactive quinones
- Involvement in decomposition of lipid hydroperoxides to form peroxyl and alkoxyl radicals (Section 2.3.6 of Chapter 1)

Due to the adverse effects of free metal ions, mammals have evolved diverse mechanisms for sequestration of these essential but also potentially toxic metal ions. This chapter focuses on discussing two important metal ion-sequestering proteins in mammalian systems: ferritin and metallothionein.

2. Ferritin

2.1. General Aspects, Biochemistry, and Molecular Regulation

2.1.1. General Aspects and Biochemistry

Ferritin was first discovered in 1973 by V. Laufberger, who isolated a new protein from horse spleen that contained up to 23% by dry weight of iron. In mammals, ferritin is a 24-subunit intracellular protein whose principal role within cells is the storage of iron in a non-toxic, but bioavailable form [1]. The assembled ferritin molecule, often referred to as a nano-cage, can store up to 4,500 atoms of iron. Iron in the ferritin nanocage is insoluble and most likely redox inactive. In mammals, there are two ferritin genes that encode for the heavy (H) and light (L) subunits of ferritin, with a molecular mass of 19 and 21 kDa, respectively. Although the two subunits share approximately 55% of their sequences as well as their multihelical three-dimensional structures, they are functionally distinct. The heavy subunit is primarily responsible for the ferritoxidase activity of the ferritin complex, whereas the light subunit facilitates the storage of iron into the ferritin core. The efficient storage of iron requires the coordinated actions of both subunits.

Ferritin is primarily localized in cytosol, existing as a heteropolymers of both heavy and light subunits. It is also present in nuclei and plasma [2, 3]. Nuclear ferritin is found to be the same as cytosolic ferritin, and not the product of a separate gene. Similarly, plasma ferritin is believed to be also derived from cytosolic ferritin though plasma ferritin is composed mainly of the light subunits and is relatively iron-poor. Recently, a third mammalian nuclear gene has been identified to code for the precursor of the mitochondrial ferritin (FtMt). The mature mitochondrial ferritin subunit possessing ferrioxidase activity assembles into homopolymers [4]. In humans, the genes for ferritin heavy subunit, light subunit, and mitochondrial ferritin are localized on chromosomes 11q12, 19q13.33, and 5q21.3, respectively.

2.1.2. Molecular Regulation

The expression of ferritin subunits is under both transcriptional and translational regulation. The translational regulation of ferritin subunits has been well-characterized. The mRNAs of heavy and light subunits contain iron regulatory element (IRE), which binds with high affinity the repressor, named iron response proteins (IRPs including IRP1 and IRP2). The binding activity of IRPs to IRE is modulated by cellular availability of iron. When cellular iron is abundant, IRPs lose their binding activity to IRE, and thereby leading to increased translation of ferritin subunits. On the other hand, under cellular iron deficiency, the binding activity of IRPs to IRE is increased, thus resulting in decreased translation of ferritin subunits [1]. Redox status may also affect the translation of ferritin subunits. Reactive oxygen species (ROS) cause ferritin induction by inactivating IRP1 through reversible oxidation of critical cysteine residues [5]. As discussed below, ROS can also cause ferritin induction via activating transcription of the ferritin subunit genes.

At the transcriptional level, several factors can enhance the expression of ferritin subunits. These factors include inflammatory cytokines, oxidants, and heme. A number of transcription factors have been shown to mediate the transcription of ferritin subunit genes. For example, Nrf2 and Jun D activate transcription of human ferritin heavy subunit gene via an antioxidant response element (ARE)-driven mechanism [6, 7]. This Nrf2/ARE pathway is shown to mediate the induction of ferritin heavy subunit by phenolic compounds, dithiolethiones, and oxidants [7]. Nrf2 may also control the transcriptional expression of ferritin light subunit gene [7]. NF- κ B is demonstrated to mediate the transcriptional activation of ferritin heavy subunit by tumor necrosis factor- α [8]. It is reported that p53 downregulates the promoter activity of ferritin heavy subunit gene [9]. On the other hand, ferritin is shown to interact with p53 and increase p53 protein level and transcriptional activity during oxidative stress [10]. This suggests that ferritin cooperates with p53 to cope with oxidative stress.

2.2. Role in Health and Disease

2.2.1. General Biological Activities

Ferritin has a central role in the regulation of cellular iron homeostasis. Sequestration of iron ions by ferritin is an important mechanism for controlling iron-mediated ROS formation and oxidative damage in mammalian systems (Figure 18-1). As noted earlier, ferritin is mostly cytosolic but also found in mitochondria and nuclei. There is substantial evidence that mitochondrial ferritin may protect this organelle from iron toxicity and oxidative damage [11]. Similarly, nuclear ferritin may bind and protect nuclear DNA from oxidative injury. In addition, nuclear ferritin has been suggested to participate in the transcriptional regulation of β -globin gene [2]. As noted earlier in Section 2.1.1, ferritin also exists in extracellular compartments, such as plasma. Ferritin is plasma has low iron content and a distinct subunit composition. Recently, plasma ferritin is shown to regulate blood vessel formation, a novel role beyond iron storage [12].

Iron can be released from ferritin by several reducing agents, including superoxide in vitro. As described in Section 2.3.1 of Chapter 1, superoxide is also reactive to iron-sulfur clusters in enzymes, leading to release of iron from the enzymes. The released iron further reacts with hydrogen peroxide, generating the highly reactive hydroxyl radical. Indeed, superoxide and iron have been suggested as partners in crime in oxidative tissue injury [13].

2.2.2. Animal Studies

2.2.2.1. EXPERIMENTAL APPROACHES

Animal models of both targeted disruption and transgenic overexpression of ferritin heavy subunit (FTH) have been reported in the literature. Homozygous knockout of the heavy





Figure 18-1. Protection against iron-mediated oxidative stress by ferritin in cytosol, mitochondria, and nuclei. As shown, free iron is redox-active and participates in Fenton reaction, generating hydroxyl radical. In contrast, iron sequestered by ferritin is redox-inert.

In contrast, the heterozygous knockout (FTH^{+/-}) mice are born and reproduced normally [15]. Due to the embryonic lethality of the global homozygous knockout of FTH, conditional FTH knockout mice have recently been characterized with respect to phenotypic changes [16]. Generation and phenotypic characterization of these genetically-manipulated animal models have been key strategies for understanding the biological functions of ferritin in mammals.

2.2.2.2. ROLE IN CARDIOVASCULAR DISEASES AND RELATED CONDITIONS

Evidence indicates a protective role for ferritin in cardiovascular disorders, such as ischemic heart injury. Induction of myocardial ferritin is observed during cardiac ischemia [17]. It has been suggested that cardiac protection by ischemic preconditioning is mediated by ferritin. Cardiac ischemic preconditioning increases de novo synthesis of ferritin in cardiac tissue of mice, and the elevated ferritin may serve as a sink for redox-active iron, thus protecting myocardium from iron-mediated oxidative damage associated with ischemia-reperfusion injury [18]. In cultured cardiomyocytes, downregulation of ferritin heavy subunit by siRNA

increases labile iron pool, oxidative stress, and cell death, supporting a protective role for ferritin in iron-mediated oxidative cardiac injury [19]. As discussed in Chapter 15, heme oxygenase-1 is a critical protector against oxidative pathophysiology underlying various disease conditions, including cardiovascular disorders. Induction of heme oxygenase-1 is often associated with augmented synthesis of ferritin, which has been suggested to be partially responsible for the beneficial effects of heme oxygenase-1. In this regard, adenovirus-mediated overexpression of ferritin heavy subunit is shown to protect cardiovascular cells from apoptosis elicited by various insults, including ROS [20].

A potential role for ferritin in metabolic syndrome has been proposed based on its interaction with adiponectin. Adiponectin is an adipocyte-derived protein that acts to reduce insulin resistance in liver and muscle and also inhibits atherosclerosis. In primary murine skeletal cells, adiponectin is found to upregulate ferritin heavy subunit via an NF-KB-dependent mechanism. The adiponectin-mediated antioxidative stress activity is shown to rely on the induction of ferritin. Hence, upregulation of ferritin may partially explain how adiponectin protects against oxidative stress and insulin resistance in skeletal muscle [21]. In this context, oxidative stress is an important event underlying insulin resistance and other manifestations of metabolic syndrome.

2.2.2.3. ROLE IN NEURODEGENERATIVE DISEASES

As noted earlier, homozygous deletion of ferritin heavy subunit causes embryonic lethality, whereas mice with heterozygous deletion are born and reproduced normally. As compared with wild-type mice, the heterozygous mice have less than half the levels of ferritin heavy subunit and exhibit an increase in the indices of oxidative stress and apoptosis in the brain, implying a beneficial role for ferritin in oxidative neurodegeneration [22]. Transgenic overexpression of ferritin heavy subunit in dopaminergic neurons of the substantia nigra results in increased levels of ferritin in the neurons. This affords protection against neurodegeneration and oxidative stress elicited by two parkinsonism-inducing toxicants, 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine and paraquat, in young mice [23, 24]. Similarly, pharmacological iron chelation also protects against experimental parkinsonism [23, 24]. These observations suggest a causative role for iron-mediated oxidative stress in the pathogenesis of experimental parkinsonism and the feasibility of developing strategies targeting ferritin for the intervention of this neurodegenerative disorder. However, a recent study shows that chronic expression of ferritin heavy subunit in dopaminergic midbrain neurons of older mice results in an agerelated expansion of the labile iron pool and subsequent oxidative neurodegeneration [25], pointing to a complex role of ferritin in iron metabolism and neurodegeneration associated with aging.

Genetic manipulation of ferritin light subunit has been employed to study the molecular involvement of ferritin in neurodegenerative disorders. Insertional mutations in exon 4 of the ferritin light subunit gene are associated with hereditary ferritinopathy (also called neuroferritinopathy), an autosomal dominant neurodegenerative disease characterized by progressive impairment of motor and cognitive functions [1] (Section 2.2.3.1). Transgenic overexpression of one of the mutant human ferritin light subunit cDNAs in mice leads to abnormal iron metabolism likely due to the compromised ability of the mutant ferritin to sequester iron. The transgenic mice exhibit increased lipid peroxidation and oxidative protein damage in the brain tissue [26].

Mitochondrial ferritin, a newly discovered ferritin, is expressed in relatively high levels

in high energy-consuming cells, such as neurons. A recent study demonstrates that overexpression of mitochondrial ferritin inhibits 6-hydroxydopamine-induced dopaminergic neuron damage and oxidative stress [27].

2.2.2.4. ROLE IN HEPATIC DISEASES

As noted above in Section 2.2.2.1, a conditional deletion of ferritin heavy subunit in adult mice has recently been reported. This approach deletes ferritin heavy subunit almost completely in the liver, bone marrow, spleen, and thymus, but less prominently in other tissues. The mice are viable and survive up to two years without visible abnormalities. However, mice fed on a high iron diet prior to ferritin heavy subunit deletion suffer from severe liver damage, as evidenced by markedly increased release of liver enzymes into circulation, hepatic cell death, macrosteatosis, hemorrhage, and infiltration of inflammatory cells [16]. Cell depleted of ferritin heavy subunit show increased sensitivity to iron salt-induced death in cultures, and the cell injury can be prevented by overexpression of a wild-type ferritin heavy subunit but not by a mutant subunit lacking ferrioxidase activity. The cell death is preceded by an increase in cytoplasmic free iron, ROS, and mitochondrial dysfunction [16]. These results indicate that the iron-sequestering function of ferritin plays a major role in preventing iron-mediated cell and tissue damage.

Iron-elicited oxidative stress is involved in tissue ischemia-reperfusion injury. A study demonstrates that adenovirus-mediated overexpression of ferritin heavy subunit gene protects rat livers from ischemia-reperfusion injury and prevents hepatocellular damage upon transplantation into syngeneic recipients [28]. The protective effects of overexpression of ferritin heavy subunit are associated with the inhibition of endothelial cell and hepatocyte apoptosis in vivo. In vitro studies further show that overexpression of the ferritin heavy subunit in endothelial cells suppresses apoptosis induced by a variety of stimuli [28]. These findings demonstrate that ferritin acts as a critical protector in liver degeneration associated with oxidative stress.

2.2.2.5. ROLE IN CANCER

It has been proposed that ferritin by blocking iron-mediated oxidative stress may play a dual role in cancer development. On the one hand, ferritin may prevent cancer development by inhibiting iron-mediated oxidative DNA damage and multistage carcinogenesis. Indeed, overexpression of ferritin heavy subunit decreases intracellular labile iron pool and oxidative stress, and attenuates transforming growth factor-induced hepatic epithelial-mesenchymal transition, a process known to be critical in carcinogenesis [29]. On the other hand, by protecting against oxidative damage and apoptosis, ferritin may promote the survival of the already formed cancer cells. In this regard, serum ferritin levels are often found to be increased in cancer patients, suggesting a possible involvement of ferritin in facilitating cancer development (see Section 2.2.3.2 below). Indeed, serum ferritin has recently been shown to promote angiogenesis in a tumor angiogenesis model [12]. Overexpression of ferritin light subunit is also associated with metastatic phenotype of human melanoma cells. Conversely, downregulation of ferritin light subunit in melanoma cells strongly inhibits their proliferation and invasiveness in both in vitro and in vivo models [30]. Downregulation of ferritin light subunit also sensitizes the melanoma cells to oxidative stress and apoptosis [30]. These data suggest that ferritin may facilitate melanoma progression by promoting cell growth and invasiveness as well as by inhibiting apoptosis in the experimental models. Hence, whether ferritin plays an inhibitory or promoting role in cancer development may depend on the stage of carcinogenesis.

2.2.3. Human Studies and Clinical Perspectives

The involvement of ferritin in human health and disease can be summarized into the following two scenarios.

- Genetic ferritin disorders
- Changes of serum ferritin as a marker or risk factor of disease conditions

2.2.3.1. GENETIC FERRITIN DISORDERS

Genetic variations have been reported for both ferritin heavy and light subunit genes. A single point mutation (A49U) in the IRE motif of H-ferritin mRNA was identified in four of seven members of a Japanese family affected by dominantly inherited iron overload [31]. This mutation is responsible for tissue iron deposition and is a novel cause of hereditary iron overload. A more extensive study of 660 human subjects with abnormalities in serum ferritin and iron status identified only two additional novel mutations in the 5'-untranslated region of ferritin heavy subunit, which however did not seem to contribute to the abnormalities in serum ferritin and iron status [32]. This suggests that ferritin heavy subunit gene is not easily subject to mutations.

In contrast to genetic variations in ferritin heavy subunit, mutations of ferritin light subunit gene are more frequent and often lead to disorders. For example, heterozygous mutations in the 5'-untranslated region of the mRNA cause hereditary hyperferritinemia cataract syndrome (HHCS), whereas the mutations that modify the C-terminal part of the protein sequence cause hereditary ferritinopathies with neurodegeneration. Individuals with HHCS have mutations that alter the structure and functionality of the iron regulatory element (IRE), and reduce its binding affinity to the iron regulatory proteins (IRPs), thereby leading to increased expression of light subunit. Excess of ferritin light subunit in the lens causes aggregation into microcrystals, resulting in bilateral nuclear cataract [1]. Patients with hereditary ferritinopathies (also known as neuroferritinopathies) exhibit movement abnormalities, heterozygous extrapyramidal Parkinson-type symptoms, and spherical inclusions containing ferritin and iron in brain and other organs [1]. Studies in experimental models suggest that these mutations cause dysfunction of ferritin, leading to increased labile iron pool and oxidative stress as well as protein aggregation [26].

2.2.3.2. SERUM FERRITIN IN DISEASE CONDITIONS

Increased levels of serum ferritin, known as hyperferritinemia have been frequently observed in patients with various pathophysiological conditions, including iron overload, acute inflammation and infection, chronic inflammatory disorders, autoimmunity, and malignancies [3]. Under certain circumstances, serum ferritin can serve as a useful marker for the above disorders; however, interpretation of the results should be done with caution due to the multiplicity of conditions affecting serum ferritin levels. On the other hand, certain pathophysiological conditions, such as iron deficiency anemia have been associated with decreased levels of serum ferritin. Indeed, serum ferritin is routinely ordered in the evaluation of anemia in human patients [26]. Extensive studies suggest that changes in serum ferritin may be a risk factor for certain human diseases. For example, higher serum ferritin concentrations in postmenopausal women are associated with an increased risk of ischemic stroke [33]. Several recent studies also demonstrate that hyperferritinemia is associated with insulin resistance and fatty liver [34, 35]. Elevated maternal serum ferritin is also related to a higher risk of gestational diabetes and intra-uterine growth retardation [36]. As mentioned in Section 2.2.2.2, studies in experimental models suggest a protective role for ferritin in ischemic heart disease. However, epidemiological studies on ferritin/iron status and cardiovascular diseases have yielded conflicting results, with two recent larger studies failing to show a linear association between serum ferritin/iron status and risk of ischemic heart disease [37, 38].

Taken together, although serum ferritin can be used as a biomarker for certain pathophysiological conditions, including iron overload and iron-deficient anemia, the value of serum ferritin as a biomarker or risk factor in other prevalent diseases, such as diabetes and cardiovascular disorders remains to be further determined. In addition, it should be noted that the association of hyperferritinemia with increased risk of certain diseases does not establish a causal role for ferritin in disease development. Instead, changes in serum ferritin most likely reflect the body's iron status or responses to pathophysiological conditions. Thus, future studies should focus on identification of the causal involvement (either protective or detrimental) of ferritin in human diseases. In this regard, continued genetic variation studies may lead to discovery of new mutations in ferritin subunit genes as well as in mitochondrial ferritin gene. Characterization of the novel mutations and the associated phenotypic changes will provide important insights into the molecular pathways through which ferritin contributes to human health and disease.

3. Metallothionein

3.1. General Aspects, Biochemistry, and Molecular Regulation

3.1.1. General Aspects and Biochemistry

Metallothionein (MT) refers to a superfamily of small molecular mass proteins (6-7 kDa) with high cysteine content and the capacity to bind various heavy metals, including copper, zinc, cadmium, and mercury. MT was first isolated from horse kidney and characterized by M. Margoshes and B.L. Vallee in 1957. In mammals, there are four types of MT proteins, namely, MT-1, MT-2, MT-3 and MT-4, among which MT-1 and MT-2 have been most extensively investigated. MT-1 and MT-2 are widely distributed in mammalian tissues. MT-3 is found mainly in brain, kidney and reproductive organs, whereas MT-4 is localized primarily in squamous epithelia [39].

In humans, the MT genes are tightly clustered in the q13 region of chromosome 16, consisting of seven functional MT-1 genes (MT-1A, MT-1B, MT-1E, MT-1F, MT-1G, MT-1H, and MT-1X) and a single gene encoding each of the other three MT isozymes, namely, MT-2 (MT-2A gene), MT-3, and MT-4. Mice have a much simpler gene structure with only one functional gene for each of the four isozymes, namely MT-1, MT-2, MT-3, and MT-4, all located on chromosome 8 [39]. It should be noted that majority of MT basic research has been performed in rodents, predominantly mice.

3.1.2. Molecular Regulation

The genes of MT-1 and MT-2 are usually expressed coordinately in mammalian tissues, whereas MT-3 and MT-4 exhibit a much more restricted tissue expression. Regulation of MT-1 and M-2 expression has received much more extensive studies than that of MT-3 and MT-4, and occurs primarily at the transcriptional level. Both MT-1 and MT-2 are highly inducible in response to a wide range of stimuli, including heavy metal ions, glucocorticoids, cytokines, inflammation, electrophiles, and oxidants. A number of transcriptional regulatory pathways have been identified for the upregulation of MT-1 and MT-2 gene expression by various stimuli. For example, metal ion-induced expression is mediated by metal-response elementbinding transcription factor-1 (MTF-1) via a metal response element (MRE)-driven mechanism. Similarly, glucocorticoid-induced expression of MT-1 and MT-2 genes is mediated by a glucocorticoid response element (GRE)-driven mechanism [40]. In addition to MRE and GRE, the promoter region of MT-1 and MT-2 genes also contains the binding sites for a number of other transcription factors, such as SP1, Nrf1, and Nrf2. Although both Nrf1 and Nrf2 are able to bind to the antioxidant response elements (AREs) of the MT-1 and MT-2 genes, only Nrf1 activates the transcription [41]. As discussed in Chapter 2, Nrf2/ARE pathway plays a central role in the upregulation of antioxidant gene expression by metals, oxidants, electrophiles, and inflammatory cytokines. Therefore, the role of Nrf2/ARE in the regulation of MT-1 and MT-2 gene expression under the above conditions merits further investigations. Currently, the molecular regulation of MT-3 and MT-4 gene expression remains largely unknown.

3.2. Role in Health and Disease

3.2.1. General Biological Activities

The biological functions of MT proteins are summarized below.

- Storage of the heavy metals in a nontoxic form
- Regulation of cellular copper and zinc metabolism
- Antioxidant and anti-inflammatory activities

As illustrated in Figure 18-2, the antioxidant activities of MT proteins stem from two main mechanisms. They are: (1) direct ROS-scavenging ability due to the high cysteine content of the MT proteins and (2) chelation of copper ions, thus preventing the formation of hydroxyl radical and other reactive species from copper ion-mediated redox reactions. Since oxidative stress and inflammation are intimately related processes, the antioxidant activities of MT proteins may also explain their anti-inflammatory functions under various pathophysiological conditions.

3.2.2. Animal Studies

3.2.2.1. EXPERIMENTAL APPROACHES

Much of our current knowledge on MTs results from studies in murine models, especially gene knockout and transgenic overexpression mouse lines. Due to the coordinated expression

of MT-1 and MT-2, simultaneous knockout of both genes (MT-1/MT-2-null) in mice has been described, and the mice are born and reproduced normally. MT-3 knockout mouse model has also been reported in the literature, and the mice develop neurological/behavioral disorders [42, 43] (Section 3.2.2.4).



Figure 18-2. Mechanisms underlying metallothionein (MT)-mediated antioxidative activities. As illustrated, MT proteins can directly scavenge reactive oxygen and nitrogen species (ROS/RNS) due to the high content of cysteine residues. MT proteins by binding to copper ion also prevent copper from participating in Fenton-type reaction to generate hydroxyl radical.

3.2.2.2. ROLE IN CARDIOVASCULAR DISEASES

Studies in genetically manipulated animal models have demonstrated a protective role for MT-1/MT-2 in various forms of cardiovascular disorders. These include ischemia-reperfusion injury [44], alcoholic cardiomyopathy [45], doxorubicin-induced cardiotoxicity [46], as well as aging-, high-fat diet-, or sepsis-induced cardiac contractile dysfunction [47-49]. Overex-pression of MT-1/MT-2 attenuates the oxidative damage and inflammatory responses in myo-cardium under the above pathophysiological conditions, pointing to an involvement of the antioxidant activities of MT proteins in disease protection. Recently, it has been shown that MT-1/MT-2 can also enhance angiogenesis and arteriogenesis by upregulating endothelial growth factor expression and modulating smooth muscle cell and macrophage function [50]. These novel activities may result in collateral flow recovery and contribute to the cardioprotective effects of MT proteins in ischemic heart disease.

3.2.2.3. ROLE IN DIABETES AND METABOLIC SYNDROME

Mounting evidence supports a beneficial role of MT proteins in diabetes and metabolic syndrome in experimental models. Overexpression of MT-1/MT-2 in the pancreatic β -cells renders these cells more resistant to oxidative damage [51]. Transgenic overexpression of the MT proteins in mice results in attenuation of streptozotocin-induced diabetes [52]. MT over-expression also protects against diabetic complications, such as diabetic cardiomyopathy in experimental animals [53]. Insulin resistance-induced cardiac dysfunction in mice is alleviated by transgenic MT overexpression in myocardium [54], suggesting a beneficial effect of MT proteins in metabolic syndrome-associated complications. MT proteins may also function to suppress the development of metabolic syndrome. In this regard, an early study demonstrated that targeted disruption of MT-1/MT-2 genes in mice results in moderate obesity and hyperleptinemia [55]. More recently, MT-1/MT-2-null mice have been shown to be more susceptible to high-fat diet-induced obesity as compared with wild-type counterparts [56]. Together, these findings indicate a protective function of MT-1/MT-2 in the genesis of metabolic syndrome.

3.2.2.4. ROLE IN NEUROLOGICAL DISEASES

One of the best characterized functions of MT-1/MT-2 is the protection against various forms of neurological disorders, including cerebral ischemia-reperfusion injury [57, 58], toxin-induced parkinsonism [59, 60], excitotoxicity [61], autoimmune encephalomyelitis [62], and traumatic brain or spinal cord injury [63, 64]. Overall, the beneficial effects of MT-1/MT-2 observed in these experimental models are consistent with the protective functions of the MT proteins in oxidative stress and inflammation. Indeed, MT-1/MT-2 proteins have been proposed as multipurpose neuroprotectants in brain pathogenesis [65]. Recently, studies in MT-3-null and -overexpressing mice also demonstrate an important role for this brain-selective MT isoform in attenuating cerebral ischemic injury and kainic acid-induced seizure [42, 66]. Moreover, MT-3-null mice develop abnormalities in psychological behaviors, suggesting a possible involvement of MT-3 in higher levels of brain function [43].

3.2.2.5. ROLE IN PULMONARY DISEASES

Multiple studies demonstrate a protective role for MT proteins in acute lung injury and inflammation induced by toxicants, such as nickel and ozone [67, 68]. The lung injury and inflammation elicited by lipopolysaccharide (given via either intratracheal instillation or systemic administration) are much more severe in MT-1/MT-2-null mice than that in wild-type animals [69, 70], implicating a potent anti-inflammatory function of MT-1/MT-2. In line with this notion, targeted disruption of MT-1/MT-2 genes in mice also sensitizes the animals to ovalbumin-induced airway inflammation and oxidative stress, as evidenced by increased proinflammatory cytokine production, infiltration of inflammatory cells, and oxidative DNA damage [71]. Ovalbumin-induced airway inflammation is a commonly used experimental model of asthma.

3.2.2.6. ROLE IN HEPATIC AND GASTROINTESTINAL DISEASES

Liver is a major target organ of chemical toxicity. MT proteins play a central part in protecting against hepatic injury induced by heavy metals, such as cadmium [72]. MT proteins also alleviate liver injury caused by other chemicals, including carbon tetrachloride [73], alcohol [74], acetaminophen [75], cisplatin [76], and lipopolysaccharide [77]. There is also evidence indicating an involvement of MT proteins in liver cell regeneration [78]. Moreover, adenovirus-mediated human MT-2A gene transfer is shown to reverse liver fibrosis and promote hepatocyte regeneration in a mouse model of carbon tetrachloride-induced irreversible liver damage [79].

MT proteins possess a beneficial activity in certain gastrointestinal disorders. Increased expression of MT proteins is observed in experimental animals with inflammatory bowel disease or *Helicobacter pylori* infection [80, 81]. A recent study demonstrates that targeted disruption of MT-1/MT-2 genes in mice markedly aggravates *H. pylori*-induced gastric inflammation and ulceration as compared with wild-type mice [82]. However, it remains unknown if selective overexpression or induction of MT proteins in gastrointestinal tissues is also protective against gut inflammatory disorders.

3.2.2.7. ROLE IN RENAL DISEASES

Studies implicate MT-1/MT-2 as important protective molecules in nephrotoxicity induced by toxicants and drugs, including cadmium, arsenicals, and cisplatin. For instance, targeted disruption of MT-1/MT-2 in mice greatly sensitizes the animals to cadmium-induced nephrotoxicity following chronic oral exposure [72]. Long-term exposure to arsenicals leads to renal injury in wild-type mice; however, the renal toxicity is dramatically aggravated in MT-1/MT-2-null mice, as manifested by increased tubular cell vaculolization, inflammatory cell infiltration, and interstitial fibrosis [83].

Cisplatin is a widely used anticancer drug whose clinical application is limited by nephrotoxicity. Targeted disruption of MT-1/MT-2 in mice exacerbates cisplatin-induced renal injury. Pre-induction of renal MT proteins by bismuth nitrate or zinc sulfate protects the wildtype mice from cisplatin-induced nephrotoxicity. However, the same treatment in MT-1/MT-2-null mice does not induce MT proteins and fails to suppress cisplatin-elicited nephrotoxicity [84]. These results indicate that MT proteins are an important endogenous defense against cisplatin-associated renal injury and point to the possibility to attenuate this anticancer druginduced nephrotoxicity by upregulating renal MT proteins.

3.2.2.8. ROLE IN SKIN DISEASES

The proangiogenesis properties of MTs along with their role in cell regeneration implicate that these metal-binding proteins may be involved in skin wound healing. MT-1/MT-2 are upregulated in skin tissue following topical application of zinc and copper. These proteins are also elevated in the wound margins, particularly in regions of high mitotic activity. In addition, MT expression in the wound margins is accompanied by accumulation of zinc, a metal involved in wound healing [85, 86]. Induction of MTs in the wound margins may reflect their role in promoting cell proliferation and reepithelialization. Indeed, either chemical or physical injury-induced epidermal cell proliferation is attenuated in mice with targeted disruption of MT-1/MT-2 [87].

MT-1/MT-2-null mice also exhibit reduced tolerance to ultraviolet B (UVB)-induced skin injury, as revealed by increased numbers of sunburn cells and apoptotic cells compared with wild-type littermates [88]. These experimental data suggest that MT-1/MT-2 have a photoprotective activity in vivo. The in vivo protective effect of MTs on UV irradiation-induced skin injury is also consistent with the observations that MT proteins in skin are highly inducible by UVB, and that induction of MTs in cultured keratinocytes attenuates UVB-elicited cell injury and oxidative stress.

3.2.2.9. ROLE IN CANCER

As described in prior sections, MTs exhibit antioxidant and anti-inflammatory activities, and they also possess antiapoptotic, proliferative, and proangiogenic properties. On the one hand, the antioxidant and anti-inflammatory activities of MTs implicate a protective role for these proteins in carcinogenesis as oxidative stress and inflammation are critical events underlying multistage carcinogenesis. On the other hand, the antiapoptotic, proliferative, and proangiogenic properties of MTs would suggest a potential tumor promoting activity. The protection by MTs against anticancer drug-induced cytotoxicity may also implicate them in anticancer drug resistance. However, direct evidence for a causal tumor promoting activity of MTs in experimental animal models has been largely lacking though multiple studies have shown an increased expression of MT proteins in malignancies [89].

In contrast, extensive studies using genetically manipulated mouse models demonstrate a critical role for MTs in protecting against multistage carcinogenesis induced by various chemical carcinogens as well as γ -irradiation. For example, MT-1/MT-2-null mice are found to be hypersensitive to lead-induced kidney carcinogenesis and cisplatin-induced hepatocellular carcinoma as compared with wild-type counterparts [90, 91]. MT-1/MT-2 deficiency in mice also markedly sensitizes the animals to multistage skin carcinogenesis caused by 7,12-dimethylbezn(a)anthracene and 12-*O*-tetradecanoylphorbol-13-acetate [92]. In addition to chemical carcinogenesis, targeted disruption of MT-1/MT-2 in mice also renders the animals more susceptible to γ -irradiation-induced thymic lymphoma and oxidative DNA damage as compared with wild-type littermates [93].

3.2.2.10. ROLE IN OTHER DISEASES AND CONDITIONS

Studies using mice with genetically manipulated MT genes indicate that these metalbinding antioxidant proteins may be protective in many other pathophysiological conditions. These include chemically-induced retinal neuron damage [94], selective degeneration of central photoreceptors caused by hyperbaric oxygen [95], X-irradiation-induced bone marrow depression [96], and multiple organ damage following lipopolysaccharide-elicited systemic inflammation [70].

3.2.3. Human Studies and Clinical Perspectives

The involvement of MTs in protection against various pathophysiological conditions as revealed in experimental animals suggests that these proteins may also have a beneficial role in human health and disease. Indeed, studies show that levels of MTs are altered in some human disorders, especially malignancies [89]. MT expression is found to be upregulated in certain human malignancies, such as breast and lung cancer, and downregulated in others, including hepatocellular carcinoma. It has been recently reported that MT-1G acts as a potential tumor suppressor gene in human hepatocellular carcinoma [97].

Genetic variation studies in human populations have identified multiple polymorphisms in both promoter and coding regions of the MT genes. Among them several alleles are shown to be associated with increased risk of developing certain disorders, including atherosclerosis [98, 99], diabetes and diabetic complications [100, 101], and heavy metal toxicity [102, 103]. In addition, a recent study in an Italian population demonstrates that a polymorphism in MT-1A gene coding region is associated with longevity [104]. While these studies indicate an association between genetic variations of MT genes and susceptibility to developing certain pathophysiological conditions, they do not establish a causal involvement of MT proteins in the disease processes. As such, it becomes necessary that when performing case-control studies, MT protein levels are detected, and the observed differences in MT protein levels are matched by the genotypes. In addition, the association revealed by epidemiological studies should be further investigated in experimental models, including cell cultures and genetically manipulated animals. Such experimental studies will help establish a causative relationship between genetic variations of MT genes, changes of MT proteins, and susceptibility to disease development.

4. Conclusion and Future Directions

Both ferritin and MTs are critically involved in regulating essential metal ion homeostasis. They also possess antioxidant and anti-inflammatory activities as well as other novel functions, such as regulation of blood vessel formation. These multifunctional proteins play an important part in protecting against various disease processes involving oxidative stress and inflammation in animal models. Although being not as compelling as the findings in animal studies, observations from clinical and epidemiological studies do suggest a possible involvement of these metal-sequestering proteins in human health and disease. Continued basic and clinical research is warranted to further elucidate the causative role (either beneficial or detrimental) for both ferritin and MTs in disease pathophysiology, and develop strategies selectively targeting these proteins to combat human disorders involving oxidative and inflammatory stress.

5. References

- [1] Arosio P, Ingrassia R, Cavadini P. Ferritins: a family of molecules for iron storage, antioxidation and more. *Biochim. Biophys. Acta* 2009;1790:589-599.
- [2] Alkhateeb AA, Connor JR. Nuclear ferritin: A new role for ferritin in cell biology. *Biochim. Biophys. Acta* 2010;1800:793-797.
- [3] Wang W, Knovich MA, Coffman LG, Torti FM, Torti SV. Serum ferritin: past, present and future. *Biochim. Biophy.s Acta* 2010;1800:760-769.
- [4] Arosio P, Levi S. Cytosolic and mitochondrial ferritins in the regulation of cellular iron homeostasis and oxidative damage. *Biochim. Biophys. Acta* 2010;1800:783-792.
- [5] Cairo G, Castrusini E, Minotti G, Bernelli-Zazzera A. Superoxide and hydrogen peroxide-dependent inhibition of iron regulatory protein activity: a protective stratagem against oxidative injury. *FASEB J.* 1996;10:1326-1335.
- [6] Tsuji Y. JunD activates transcription of the human ferritin H gene through an antioxidant response element during oxidative stress. *Oncogene* 2005;24:7567-7578.
- [7] Pietsch EC, Chan JY, Torti FM, Torti SV. Nrf2 mediates the induction of ferritin H in response to xenobiotics and cancer chemopreventive dithiolethiones. J. Biol. Chem. 2003;278:2361-2369.
- [8] Kwak EL, Larochelle DA, Beaumont C, Torti SV, Torti FM. Role for NF-kappa B in the regulation of ferritin H by tumor necrosis factor-alpha. *J. Biol. Chem.* 1995;270:

15285-15293.

- [9] Faniello MC, Di Sanzo M, Quaresima B, Baudi F, Di Caro V, Cuda G, et al. p53mediated downregulation of H ferritin promoter transcriptional efficiency via NF-Y. *Int. J. Biochem. Cell Biol.* 2008;40:2110-2119.
- [10] Lee JH, Jang H, Cho EJ, Youn HD. Ferritin binds and activates p53 under oxidative stress. *Biochem. Biophys. Res. Commun.* 2009;389:399-404.
- [11] Campanella A, Rovelli E, Santambrogio P, Cozzi A, Taroni F, Levi S. Mitochondrial ferritin limits oxidative damage regulating mitochondrial iron availability: hypothesis for a protective role in Friedreich ataxia. *Hum. Mol. Genet.* 2009;18:1-11.
- [12] Coffman LG, Parsonage D, D'Agostino R, Jr., Torti FM, Torti SV. Regulatory effects of ferritin on angiogenesis. *Proc. Natl. Acad. Sci. USA* 2009;106:570-575.
- [13] Liochev SI, Fridovich I. Superoxide and iron: partners in crime. *IUBMB Life* 1999;48:157-161.
- [14] Ferreira C, Bucchini D, Martin ME, Levi S, Arosio P, Grandchamp B, et al. Early embryonic lethality of H ferritin gene deletion in mice. J. Biol. Chem. 2000;275:3021-3024.
- [15] Ferreira C, Santambrogio P, Martin ME, Andrieu V, Feldmann G, Henin D, et al. H ferritin knockout mice: a model of hyperferritinemia in the absence of iron overload. Blood 2001;98:525-532.
- [16] Darshan D, Vanoaica L, Richman L, Beermann F, Kuhn LC. Conditional deletion of ferritin H in mice induces loss of iron storage and liver damage. *Hepatology* 2009;50:852-860.
- [17] Loncar R, Flesche CW, Deussen A. Myocardial ferritin content is closely related to the degree of ischaemia. Acta Physiol. Scand. 2004;180:21-28.
- [18] Chevion M, Leibowitz S, Aye NN, Novogrodsky O, Singer A, Avizemer O, *et al.* Heart protection by ischemic preconditioning: a novel pathway initiated by iron and mediated by ferritin. *J. Mol. Cell Cardiol.* 2008;45:839-845.
- [19] Omiya S, Hikoso S, Imanishi Y, Saito A, Yamaguchi O, Takeda T, *et al.* Downregulation of ferritin heavy chain increases labile iron pool, oxidative stress and cell death in cardiomyocytes. *J. Mol. Cell Cardiol.* 2009;46:59-66.
- [20] Kinderlerer AR, Pombo Gregoire I, Hamdulay SS, Ali F, Steinberg R, Silva G, et al. Heme oxygenase-1 expression enhances vascular endothelial resistance to complement-mediated injury through induction of decay-accelerating factor: a role for increased bilirubin and ferritin. *Blood* 2009;113:1598-1607.
- [21] Ikegami Y, Inukai K, Imai K, Sakamoto Y, Katagiri H, Kurihara S, *et al.* Adiponectin upregulates ferritin heavy chain in skeletal muscle cells. *Diabetes* 2009;58:61-70.
- [22] Thompson K, Menzies S, Muckenthaler M, Torti FM, Wood T, Torti SV, et al. Mouse brains deficient in H-ferritin have normal iron concentration but a protein profile of iron deficiency and increased evidence of oxidative stress. J. Neurosci. Res. 2003;71: 46-63.
- [23] Kaur D, Yantiri F, Rajagopalan S, Kumar J, Mo JQ, Boonplueang R, *et al.* Genetic or pharmacological iron chelation prevents MPTP-induced neurotoxicity in vivo: a novel therapy for Parkinson's disease. *Neuron* 2003;37:899-909.
- [24] McCormack AL, Atienza JG, Johnston LC, Andersen JK, Vu S, Di Monte DA. Role of oxidative stress in paraquat-induced dopaminergic cell degeneration. J. Neurochem. 2005;93:1030-1037.

- [25] Kaur D, Rajagopalan S, Andersen JK. Chronic expression of H-ferritin in dopaminergic midbrain neurons results in an age-related expansion of the labile iron pool and subsequent neurodegeneration: implications for Parkinson's disease. *Brain Res.* 2009; 1297:17-22.
- [26] Barbeito AG, Garringer HJ, Baraibar MA, Gao X, Arredondo M, Nunez MT, et al. Abnormal iron metabolism and oxidative stress in mice expressing a mutant form of the ferritin light polypeptide gene. J. Neurochem. 2009;109:1067-1078.
- [27] Shi ZH, Nie G, Duan XL, Rouault T, Wu WS, Ning B, *et al.* Neuroprotective mechanism of mitochondrial ferritin on 6-hydroxydopamine-induced dopaminergic cell damage: implication for neuroprotection in Parkinson's disease. *Antioxid. Redox Signal* 2010;13:783-796.
- [28] Berberat PO, Katori M, Kaczmarek E, Anselmo D, Lassman C, Ke B, *et al.* Heavy chain ferritin acts as an antiapoptotic gene that protects livers from ischemia reperfusion injury. *FASEB J.* 2003;17:1724-1726.
- [29] Zhang KH, Tian HY, Gao X, Lei WW, Hu Y, Wang DM, et al. Ferritin heavy chainmediated iron homeostasis and subsequent increased reactive oxygen species production are essential for epithelial-mesenchymal transition. Cancer Res. 2009;69: 5340-5348.
- [30] Baldi A, Lombardi D, Russo P, Palescandolo E, De Luca A, Santini D, *et al.* Ferritin contributes to melanoma progression by modulating cell growth and sensitivity to oxidative stress. *Clin. Cancer Res.* 2005;11:3175-3183.
- [31] Kato J, Fujikawa K, Kanda M, Fukuda N, Sasaki K, Takayama T, *et al.* A mutation, in the iron-responsive element of H ferritin mRNA, causing autosomal dominant iron overload. *Am. J. Hum. Genet.* 2001;69:191-197.
- [32] Cremonesi L, Foglieni B, Fermo I, Cozzi A, Paroni R, Ruggeri G, *et al.* Identification of two novel mutations in the 5'-untranslated region of H-ferritin using denaturing high performance liquid chromatography scanning. *Haematologica* 2003;88:1110-1116.
- [33] van der AD, Grobbee DE, Roest M, Marx JJ, Voorbij HA, van der Schouw YT. Serum ferritin is a risk factor for stroke in postmenopausal women. *Stroke* 2005;36:1637-1641.
- [34] Brudevold R, Hole T, Hammerstrom J. Hyperferritinemia is associated with insulin resistance and fatty liver in patients without iron overload. *PLoS One* 2008;3:e3547.
- [35] Kim CH, Kim HK, Bae SJ, Park JY, Lee KU. Association of elevated serum ferritin concentration with insulin resistance and impaired glucose metabolism in Korean men and women. *Metabolism* 2010 (in press).
- [36] Soubasi V, Petridou S, Sarafidis K, Tsantali C, Diamanti E, Buonocore G, *et al.* Association of increased maternal ferritin levels with gestational diabetes and intrauterine growth retardation. *Diabetes Metab.* 2010;36:58-63.
- [37] Galan P, Noisette N, Estaquio C, Czernichow S, Mennen L, Renversez JC, et al. Serum ferritin, cardiovascular risk factors and ischaemic heart diseases: a prospective analysis in the SU.VI.MAX (SUpplementation en VItamines et Mineraux AntioXydants) cohort. Public Health Nutr. 2006;9:70-74.
- [38] Friedrich N, Milman N, Volzke H, Linneberg A, Jorgensen T. Is serum ferritin within the reference range a risk predictor of cardiovascular disease? A population-based, long-term study comprising 2874 subjects. *Br. J. Nutr.* 2009;102:594-600.
- [39] Coyle P, Philcox JC, Carey LC, Rofe AM. Metallothionein: the multipurpose protein.

Cell Mol. Life Sci. 2002;59:627-647.

- [40] Vasak M. Advances in metallothionein structure and functions. J. Trace Elem. Med. Biol. 2005;19:13-17.
- [41] Ohtsuji M, Katsuoka F, Kobayashi A, Aburatani H, Hayes JD, Yamamoto M. Nrf1 and Nrf2 play distinct roles in activation of antioxidant response element-dependent genes. *J. Biol. Chem.* 2008;283:33554-33562.
- [42] Erickson JC, Hollopeter G, Thomas SA, Froelick GJ, Palmiter RD. Disruption of the metallothionein-III gene in mice: analysis of brain zinc, behavior, and neuron vulnerability to metals, aging, and seizures. J. Neurosci. 1997;17:1271-1281.
- [43] Koumura A, Kakefuda K, Honda A, Ito Y, Tsuruma K, Shimazawa M, et al. Metallothionein-3 deficient mice exhibit abnormalities of psychological behaviors. *Neurosci. Lett.* 2009;467:11-14.
- [44] Kang YJ, Li Y, Sun X. Antiapoptotic effect and inhibition of ischemia/reperfusioninduced myocardial injury in metallothionein-overexpressing transgenic mice. Am. J. Pathol. 2003;163:1579-1586.
- [45] Wang L, Zhou Z, Saari JT, Kang YJ. Alcohol-induced myocardial fibrosis in metallothionein-null mice: prevention by zinc supplementation. Am. J. Pathol. 2005; 167:337-344.
- [46] Kang YJ, Chen Y, Yu A, Voss-McCowan M, Epstein PN. Overexpression of metallothionein in the heart of transgenic mice suppresses doxorubicin cardiotoxicity. J. Clin. Invest. 1997;100:1501-1506.
- [47] Fang CX, Doser TA, Yang X, Sreejayan N, Ren J. Metallothionein antagonizes aginginduced cardiac contractile dysfunction: role of PTP1B, insulin receptor tyrosine phosphorylation and Akt. *Aging Cell* 2006;5:177-185.
- [48] Dong F, Li Q, Sreejayan N, Nunn JM, Ren J. Metallothionein prevents high-fat diet induced cardiac contractile dysfunction: role of peroxisome proliferator activated receptor gamma coactivator 1alpha and mitochondrial biogenesis. *Diabetes* 2007;56: 2201-2212.
- [49] Ceylan-Isik AF, Zhao P, Zhang B, Xiao X, Su G, Ren J. Cardiac overexpression of metallothionein rescues cardiac contractile dysfunction and endoplasmic reticulum stress but not autophagy in sepsis. J. Mol. Cell Cardiol. 2010;48:367-378.
- [50] Zbinden S, Wang J, Adenika R, Schmidt M, Tilan JU, Najafi AH, et al. Metallothionein enhances angiogenesis and arteriogenesis by modulating smooth muscle cell and macrophage function. Arterioscler. Thromb. Vasc. Biol. 2010;30:477-482.
- [51] Li X, Chen H, Epstein PN. Metallothionein protects islets from hypoxia and extends islet graft survival by scavenging most kinds of reactive oxygen species. *J. Biol. Chem.* 2004;279:765-771.
- [52] Chen H, Carlson EC, Pellet L, Moritz JT, Epstein PN. Overexpression of metallothionein in pancreatic beta-cells reduces streptozotocin-induced DNA damage and diabetes. *Diabetes* 2001;50:2040-2046.
- [53] Liang Q, Carlson EC, Donthi RV, Kralik PM, Shen X, Epstein PN. Overexpression of metallothionein reduces diabetic cardiomyopathy. *Diabetes* 2002;51:174-181.
- [54] Fang CX, Dong F, Ren BH, Epstein PN, Ren J. Metallothionein alleviates cardiac contractile dysfunction induced by insulin resistance: role of Akt phosphorylation, PTB1B, PPARgamma and c-Jun. *Diabetologia* 2005;48:2412-2421.

- [55] Beattie JH, Wood AM, Newman AM, Bremner I, Choo KH, Michalska AE, et al. Obesity and hyperleptinemia in metallothionein (-I and -II) null mice. Proc. Natl. Acad. Sci. USA 1998;95:358-363.
- [56] Sato M, Kawakami T, Kondoh M, Takiguchi M, Kadota Y, Himeno S, et al. Development of high-fat-diet-induced obesity in female metallothionein-null mice. FASEB J. 2010;24:2375-2384.
- [57] van Lookeren Campagne M, Thibodeaux H, van Bruggen N, Cairns B, Gerlai R, Palmer JT, *et al.* Evidence for a protective role of metallothionein-1 in focal cerebral ischemia. *Proc. Natl. Acad. Sci. USA* 1999;96:12870-12875.
- [58] Trendelenburg G, Prass K, Priller J, Kapinya K, Polley A, Muselmann C, *et al.* Serial analysis of gene expression identifies metallothionein-II as major neuroprotective gene in mouse focal cerebral ischemia. *J. Neurosci.* 2002;22:5879-5888.
- [59] Asanuma M, Miyazaki I, Higashi Y, Tanaka K, Haque ME, Fujita N, et al. Aggravation of 6-hydroxydopamine-induced dopaminergic lesions in metallothionein-I and -II knock-out mouse brain. *Neurosci. Lett.* 2002;327:61-65.
- [60] Rojas P, Klaassen CD. Metallothionein-I and -II knock-out mice are not more sensitive than control mice to 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine neurotoxicity. *Neurosci. Lett.* 1999;273:113-116.
- [61] Carrasco J, Penkowa M, Hadberg H, Molinero A, Hidalgo J. Enhanced seizures and hippocampal neurodegeneration following kainic acid-induced seizures in metallothionein-I + II-deficient mice. *Eur. J. Neurosci.* 2000;12:2311-2322.
- [62] Penkowa M, Espejo C, Martinez-Caceres EM, Poulsen CB, Montalban X, Hidalgo J. Altered inflammatory response and increased neurodegeneration in metallothionein I+II deficient mice during experimental autoimmune encephalomyelitis. J. Neuroimmunol. 2001;119:248-260.
- [63] Penkowa M, Caceres M, Borup R, Nielsen FC, Poulsen CB, Quintana A, et al. Novel roles for metallothionein-I + II (MT-I + II) in defense responses, neurogenesis, and tissue restoration after traumatic brain injury: insights from global gene expression profiling in wild-type and MT-I + II knockout mice. J. Neurosci. Res. 2006;84:1452-1474.
- [64] DeRuisseau LR, Recca DM, Mogle JA, Zoccolillo M, DeRuisseau KC. Metallothionein deficiency leads to soleus muscle contractile dysfunction following acute spinal cord injury in mice. Am. J. Physiol. Regul. Integr. Comp. Physiol. 2009;297:R1795-1802.
- [65] Penkowa M. Metallothioneins are multipurpose neuroprotectants during brain pathology. FEBS J. 2006;273:1857-1870.
- [66] Koumura A, Hamanaka J, Shimazawa M, Honda A, Tsuruma K, Uchida Y, et al. Metallothionein-III knockout mice aggravates the neuronal damage after transient focal cerebral ischemia. *Brain Res.* 2009;1292:148-154.
- [67] Wesselkamper SC, McDowell SA, Medvedovic M, Dalton TP, Deshmukh HS, Sartor MA, et al. The role of metallothionein in the pathogenesis of acute lung injury. Am. J. Respir. Cell Mol. Biol. 2006;34:73-82.
- [68] Inoue K, Takano H, Kaewamatawong T, Shimada A, Suzuki J, Yanagisawa R, et al. Role of metallothionein in lung inflammation induced by ozone exposure in mice. Free Radic. Biol. Med. 2008;45:1714-1722.
- [69] Takano H, Inoue K, Yanagisawa R, Sato M, Shimada A, Morita T, *et al.* Protective role of metallothionein in acute lung injury induced by bacterial endotoxin. *Thorax*

2004;59:1057-1062.

- [70] Inoue K, Takano H, Shimada A, Wada E, Yanagisawa R, Sakurai M, et al. Role of metallothionein in coagulatory disturbance and systemic inflammation induced by lipopolysaccharide in mice. FASEB J. 2006;20:533-535.
- [71] Inoue K, Takano H, Yanagisawa R, Sakurai M, Ichinose T, Sadakane K, et al. Role of metallothionein in antigen-related airway inflammation. Exp. Biol. Med. (Maywood) 2005;230:75-81.
- [72] Klaassen CD, Liu J, Diwan BA. Metallothionein protection of cadmium toxicity. *Toxicol. Appl. Pharmacol.* 2009;238:215-220.
- [73] Davis SR, Samuelson DA, Cousins RJ. Metallothionein expression protects against carbon tetrachloride-induced hepatotoxicity, but overexpression and dietary zinc supplementation provide no further protection in metallothionein transgenic and knockout mice. J. Nutr. 2001;131:215-222.
- [74] Zhou Z, Sun X, James Kang Y. Metallothionein protection against alcoholic liver injury through inhibition of oxidative stress. *Exp. Biol. Med.* (Maywood) 2002;227: 214-222.
- [75] Saito C, Yan HM, Artigues A, Villar MT, Farhood A, Jaeschke H. Mechanism of protection by metallothionein against acetaminophen hepatotoxicity. *Toxicol. Appl. Pharmacol.* 2010;242:182-190.
- [76] Liu J, Liu Y, Habeebu SS, Klaassen CD. Metallothionein (MT)-null mice are sensitive to cisplatin-induced hepatotoxicity. *Toxicol. Appl. Pharmacol.* 1998;149:24-31.
- [77] Kimura T, Itoh N, Takehara M, Oguro I, Ishizaki JI, Nakanishi T, et al. Sensitivity of metallothionein-null mice to LPS/D-galactosamine-induced lethality. Biochem. Biophys. Res. Commun. 2001;280:358-362.
- [78] Cherian MG, Kang YJ. Metallothionein and liver cell regeneration. *Exp. Biol. Med.* (Maywood) 2006;231:138-144.
- [79] Jiang Y, Kang YJ. Metallothionein gene therapy for chemical-induced liver fibrosis in mice. *Mol. Ther.* 2004;10:1130-1139.
- [80] Waeytens A, De Vos M, Laukens D. Evidence for a potential role of metallothioneins in inflammatory bowel diseases. *Mediators Inflamm.* 2009;2009:729172.
- [81] Tran CD, Gobel R, Symonds EL. Metallothionein expression in Helicobacter-infected pregnant mice and their fetuses and pups. *Dig. Dis. Sci.* 2007;52:1527-1532.
- [82] Mita M, Satoh M, Shimada A, Okajima M, Azuma S, Suzuki JS, *et al.* Metallothionein is a crucial protective factor against Helicobacter pylori-induced gastric erosive lesions in a mouse model. *Am. J. Physiol. Gastrointest. Liver. Physiol.* 2008;294:G877-884.
- [83] Liu J, Liu Y, Goyer RA, Achanzar W, Waalkes MP. Metallothionein-I/II null mice are more sensitive than wild-type mice to the hepatotoxic and nephrotoxic effects of chronic oral or injected inorganic arsenicals. *Toxicol. Sci.* 2000;55:460-467.
- [84] Satoh M, Aoki Y, Tohyama C. Protective role of metallothionein in renal toxicity of cisplatinum. *Cancer. Chemother. Pharmacol.* 1997;40:358-362.
- [85] Lansdown AB. Metallothioneins: potential therapeutic aids for wound healing in the skin. *Wound Repair Regen.* 2002;10:130-132.
- [86] Iwata M, Takebayashi T, Ohta H, Alcalde RE, Itano Y, Matsumura T. Zinc accumulation and metallothionein gene expression in the proliferating epidermis during wound healing in mouse skin. *Histochem. Cell Biol.* 1999;112:283-290.
- [87] Hanada K, Sawamura D, Hashimoto I, Kida K, Naganuma A. Epidermal proliferation

of the skin in metallothionein-null mice. J. Invest. Dermatol. 1998;110:259-262.

- [88] Wang WH, Li LF, Zhang BX, Lu XY. Metallothionein-null mice exhibit reduced tolerance to ultraviolet B injury in vivo. *Clin. Exp. Dermatol.* 2004;29:57-61.
- [89] Pedersen MO, Larsen A, Stoltenberg M, Penkowa M. The role of metallothionein in oncogenesis and cancer prognosis. *Prog. Histochem. Cytochem.* 2009;44:29-64.
- [90] Waalkes MP, Liu J, Goyer RA, Diwan BA. Metallothionein-I/II double knockout mice are hypersensitive to lead-induced kidney carcinogenesis: role of inclusion body formation. *Cancer Res.* 2004;64:7766-7772.
- [91] Waalkes MP, Liu J, Kasprzak KS, Diwan BA. Hypersusceptibility to cisplatin carcinogenicity in metallothionein-I/II double knockout mice: production of hepatocellular carcinoma at clinically relevant doses. *Int. J. Cancer* 2006;119:28-32.
- [92] Suzuki JS, Nishimura N, Zhang B, Nakatsuru Y, Kobayashi S, Satoh M, *et al.* Metallothionein deficiency enhances skin carcinogenesis induced by 7,12-dimethylbenz[a]anthracene and 12-O-tetradecanoylphorbol-13-acetate in metallothio-nein-null mice. *Carcinogenesis* 2003;24:1123-1132.
- [93] Shibuya K, Nishimura N, Suzuki JS, Tohyama C, Naganuma A, Satoh M. Role of metallothionein as a protective factor against radiation carcinogenesis. J. Toxicol. Sci. 2008;33:651-655.
- [94] Suemori S, Shimazawa M, Kawase K, Satoh M, Nagase H, Yamamoto T, et al. Metallothionein, an endogenous antioxidant, protects against retinal neuron damage in mice. *Invest. Ophthalmol. Vis. Sci.* 2006;47:3975-3982.
- [95] Nachman-Clewner M, Giblin FJ, Dorey CK, Blanks RH, Dang L, Dougherty CJ, et al. Selective degeneration of central photoreceptors after hyperbaric oxygen in normal and metallothionein-knockout mice. *Invest. Ophthalmol. Vis. Sci.* 2008;49:3207-3215.
- [96] Shibuya K, Suzuki JS, Kito H, Naganuma A, Tohyama C, Satoh M. Protective role of metallothionein in bone marrow injury caused by X-irradiation. J. Toxicol. Sci. 2008;33:479-484.
- [97] Kanda M, Nomoto S, Okamura Y, Nishikawa Y, Sugimoto H, Kanazumi N, et al. Detection of metallothionein 1G as a methylated tumor suppressor gene in human hepatocellular carcinoma using a novel method of double combination array analysis. *Int. J. Oncol.* 2009;35:477-483.
- [98] Giacconi R, Muti E, Malavolta M, Cipriano C, Costarelli L, Bernardini G, et al. The +838 C/G MT2A polymorphism, metals, and the inflammatory/immune response in carotid artery stenosis in elderly people. *Mol. Med.* 2007;13:388-395.
- [99] Giacconi R, Cipriano C, Muti E, Costarelli L, Maurizio C, Saba V, *et al.* Novel 209A/G MT2A polymorphism in old patients with type 2 diabetes and atherosclerosis: relationship with inflammation (IL-6) and zinc. *Biogerontology* 2005;6:407-413.
- [100] Giacconi R, Bonfigli AR, Testa R, Sirolla C, Cipriano C, Marra M, et al. +647 A/C and +1245 MT1A polymorphisms in the susceptibility of diabetes mellitus and cardiovascular complications. *Mol. Genet. Metab.* 2008;94:98-104.
- [101] Yang L, Li H, Yu T, Zhao H, Cherian MG, Cai L, et al. Polymorphisms in metallothionein-1 and -2 genes associated with the risk of type 2 diabetes mellitus and its complications. Am. J. Physiol. Endocrinol. Metab. 2008;294:E987-992.
- [102] Chen HI, Chiu YW, Hsu YK, Li WF, Chen YC, Chuang HY. The association of metallothionein-4 gene polymorphism and renal function in long-term lead-exposed workers. *Biol. Trace Elem. Res.* 2010;137:55-62.

- [103] Miura N. Individual susceptibility to cadmium toxicity and metallothionein gene polymorphisms: with references to current status of occupational cadmium exposure. *Ind. Health.* 2009;47:487-494.
- [104] Cipriano C, Malavolta M, Costarelli L, Giacconi R, Muti E, Gasparini N, *et al.* Polymorphisms in MT1a gene coding region are associated with longevity in Italian Central female population. *Biogerontology* 2006;7:357-365.

Chapter 19

Other Non-Protein Antioxidants Synthesized by Cells

Abstract

Previous chapters have discussed the endogenous protein antioxidants as well as the tripeptide glutathione synthesized in mammalian cells. There are also a variety of non-protein antioxidants produced by mammalian cells, including bilirubin, coenzyme Q, estrogens, α -lipoic acid, melatonin, pyruvate, and uric acid. This chapter describes each of these common non-protein compounds with respect to their antioxidant activities and role in health and disease.

1. Introduction

The most important non-protein antioxidant synthesized by cells is perhaps the tripeptide, glutathione, which is described in chapter 5. In addition to glutathione, a number of other non-protein compounds with antioxidant properties are also synthesized by mammalian cells. These include bilirubin, coenzyme Q, estrogens, α -lipoic acid, melatonin, pyruvate, and uric acid. This chapter discusses the biochemical properties and role in health and disease for each of the above non-protein antioxidants.

2. Bilirubin

2.1. General Aspects and Biochemistry

2.1.1. Bilirubin and Heme Oxygenases

Bilirubin (structure shown in Figure 19-1) is the end product of heme metabolism in mammals. Heme oxygenases catalyze the cleavage of the heme ring to release ferrous ion, carbon monoxide, and biliverdin. The hydrophilic biliverdin is rapidly reduced by biliverdin reductase to hydrophobic bilirubin. As discussed in Chapter 15, heme oxygenases possess

important antioxidative and anti-inflammatory activities. It is now well-recognized that the biological activities of heme oxygenases are primarily mediated by the enzymatic reaction products, including bilirubin, biliverdin, and carbon monoxide. Among these products, bilirubin has received the most extensive studies with respect to the antioxidative and anti-inflammatory functions [1].



Figure 19-1. Structure of bilirubin. See text (Section 2) for description of bilirubin.

2.1.2. Physiological Concentrations and Metabolism

In adult humans, approximately 250-350 mg bilirubin are produced daily primarily via breakdown of hemoglobin. The normal human plasma concentrations of unconjugated bilirubin are between 5 and 15 μ M, and over 99% is bound to albumin. In contrast, tissue concentrations of unconjugated bilirubin are much lower and estimated to be approximately 20-50 nM. Unconjugated, albumin-bound bilirubin is transported to liver, where it dissociates from albumin and enters hepatocytes. Inside hepatocytes, bilirubin undergoes glucuronidation to form water-soluble monoglucuronide and diglucuronide metabolites, which are then excreted into bile and eliminated from the body [1].

2.1.3. Biochemical Properties and Functions

Bilirubin is a potent scavenger of reactive oxygen and nitrogen species (ROS/RNS), including peroxyl radical and peroxynitrite [2, 3]. At physiological tissue concentrations (~20-50 nM), bilirubin is shown to provide cytoprotection against oxidative damage in various biological systems. The potent antioxidant effects of bilirubin at physiologically relevant concentrations are believed to stem from the biliverdin-bilirubin cycle, a concept first proposed by S.H. Snyder and coworkers [4, 5] (Figure 19-2). When bilirubin acts as an antioxidant, it is oxidized to biliverdin, which is rapidly reduced by biliverdin reductase to bilirubin. This cycle ensures the continuous availability of bilirubin for effective detoxification of ROS/RNS. Moreover, bilirubin and glutathione are found to have complementary antioxidant roles, highlighting the coordinated actions of these two important non-protein antioxidants in defense against oxidative stress [6]. In addition to bilirubin, the precursor biliverdin also possesses antioxidant activities in both cellular systems and experimental animals [1]. Due to its water solubility, biliverdin may scavenge ROS/RNS formed mainly in aqueous environment. In contrast, the ROS/RNS-scavenging activities of the lipophilic bilirubin are largely limited to lipid environment, such as lipoproteins and cell membranes.



Figure 19-2. Biliverdin-bilirubin redox cycle in the detoxification of reactive oxygen species (ROS). As illustrated, bilirubin scavenges ROS, and is oxidized to biliverdin. Biliverdin reductase (BR) regenerates bilirubin from biliverdin to ensure the continuous detoxification of ROS. HO, heme oxygenases; CO, carbon monoxide.

2.2. Role in Health and Disease

It is well-recognized that heme oxygenases play an important protective role in various pathophysiological conditions (Section 3 of Chapter 15). In view of the potent antioxidant activity of bilirubin, it is believed that the beneficial effects of heme oxygenases are mediated, at least partially, by bilirubin. This notion is supported by the observations that administration of exogenous bilirubin can replicate the protective effects exerted by induction of heme oxygenases in different animal models of human diseases.

2.2.1. Animal Studies

2.2.1.1. EXPERIMENTAL APPROACHES

Direct administration of bilirubin to animals has been frequently used to study its biological activities in health and disease. It is known that large doses of bilirubin cause adverse effects, such as neurotoxicity. As such, doses of bilirubin are chosen to achieve physiologically relevant plasma concentrations in experimental animals to study the potential protection in disease conditions. Physiologically relevant concentrations of bilirubin have been employed in cell cultures to understand the cytoprotective effects and the underlying mechanisms. In addition to administration of exogenous bilirubin, another important approach to studying the biological activities of bilirubin is the use of Gunn rat model. Gunn rat is deficient in the bilirubin-conjugating enzyme UDP-glucuronosyltransferase 1A1 (UGT1A1), and as such develops hyperbilirubinemia. This rat model has been used to assess the beneficial role of bilirubin in various disease conditions, including cardiovascular disorders. As UGT1A1 is the primary enzyme involved in bilirubin metabolism, another approach to increasing plasma bilirubin in animals is through the use of UGT1A1 inhibitors, such as indinavir [7].

2.2.1.2. ROLE IN CARDIOVASCULAR DISEASES

Administration of bilirubin to animals protects against certain forms of cardiovascular disorders, including myocardial ischemia-reperfusion injury and atherogenesis [8-10]. Studies using animal models of hyperbilirubinemia demonstrate that elevated plasma bilirubin can attenuate angiotensin II- or deoxycorticosterone acetate salt-induced hypertension [11, 12]. In cultured cardiovascular cells, bilirubin at physiologically relevant concentrations inhibits cell proliferation and proinflammatory responses, as well as oxidative stress-induced cytotoxicity [13, 14]. Similar to bilirubin, exogenously administered biliverdin also protects against atherosclerosis in experimental animals and inhibits vascular smooth muscle cell proliferation in cultures [10, 15].

2.2.1.3. ROLE IN DIABETES AND PANCREATIC ISLET TRANSPLANTATION

Exogenously administered bilirubin or biliverdin protects the pancreatic β -cells or islets from oxidative and inflammatory stress. Biliverdin also prevents endothelial sloughing and oxidative stress in a rat model of type 1 diabetes elicited by streptozotocin [16]. Bilirubin is shown to induce tolerance to islet allografts and prolong the survival of the allografts via mechanisms including antioxidant activities and modulation of T regulatory cells [17, 18].

2.2.1.4. ROLE IN NEUROLOGICAL DISEASES

At pathophysiologically high concentrations, bilirubin is known to cause neurotoxicity [19]. In contrast, substantial evidence supports an important role for physiologically relevant concentrations of bilirubin in neuroprotection. In cultured neuronal cells, bilirubin at physiologically relevant concentrations inhibits oxidative stress injury [20]. Administration of bilirubin to SJL/J mice has been demonstrated to also suppress autoimmune encephalomyelitis, a neurological disorder involving an oxidative stress mechanism [21].

2.2.1.5. ROLE IN PULMONARY DISEASES

An early study demonstrated that neonatal Gunn rats are more resistance to hyperoxiainduced lung oxidative stress and tissue injury, suggesting bilirubin as a protective molecule in pulmonary diseases [22]. Subsequent studies have shown that either bilirubin or biliverdin administration alleviates endotoxin-induced acute lung injury [23], and that bilirubin treatment also ameliorates bleomycin-induced pulmonary fibrosis and inflammation in experimental animals [24]. In a murine asthma model, administration of bilirubin inhibits VCAM-1mediated airway inflammation likely through suppression of cellular ROS production [25]. VCAM-1 stands for vascular cell adhesion molecule-1.

2.2.1.6. ROLE IN HEPATIC AND GASTROINTESTINAL DISEASES

Multiple studies demonstrate that exogenously administered bilirubin or biliverdin exhibits beneficial effects in several liver pathophysiological conditions, including hemin-induced hepatic oxidative stress [26], liver ischemia-reperfusion injury [27], and liver transplantation [28]. Bilirubin or biliverdin is also found to protect against intestinal ischemia-reperfusion injury and colitis in animal models [29, 30].

2.2.1.7. ROLE IN RENAL DISEASES

Similar to what observed in liver ischemia-reperfusion injury, administration of bilirubin or biliverdin attenuates renal ischemia-reperfusion injury [31, 32]. A recent study using Gunn

rats shows that hyperbilirubinemia provides marked preservation of renal function and histology in a cisplatin nephrotoxicity model [33]. Notably, exogenous bilirubin does not interfere with the antineoplastic activity of cisplatin in vitro [33], pointing to the feasibility of using bilirubin to enhance the therapeutic index of cisplatin.

2.2.1.8. ROLE IN CANCER

Bilirubin and biliverdin act as potent scavengers of peroxyl radical and other reactive species commonly involved in multistage carcinogenesis. These two pigments are also found to inhibit the mutagenic effects of various classes of mutagens in cultured cells [34]. In addition, bilirubin is shown to suppress tumor cell growth in both in vitro cultures and in vivo cancer xenografts [35, 36].

2.2.1.9. ROLE IN SEPSIS

As noted earlier in Section 2.2.1.5, exogenously administered bilirubin protects against endotoxin-induced acute lung injury. Exogenous bilirubin also reduces mortality and counteracts hypotension elicited by endotoxin through mechanisms involving a decreased induction of nitric oxide synthase 2 (NOS2) secondary to inhibition of NAD(P)H oxidase [37]. Notably, a single bolus injection of bilirubin is shown to improve the overall conditions in a mouse model of endotoxemia [38], indicating a potential value of employing bilirubin-based strategies in the clinical intervention of sepsis.

2.2.2. Human Studies and Clinical Perspectives

2.2.2.1. GENETIC VARIATIONS LEADING TO HYPERBILIRUBINEMIA IN HUMANS

Genetic variations have been demonstrated to cause elevated plasma bilirubin in humans. The most extensively studied genetic disorder of hyperbilirubinemia is Gilbert's syndrome [39]. This common syndrome is caused by decreased hepatic levels of the enzyme UDP-glucuronosyltransferase (mainly UGT1A1) due to gene mutations. As this enzyme is responsible for glucuronidation of bilirubin in the liver, reduced activity of this enzyme leads to an accumulation of unconjugated bilirubin in the circulation. Gilbert's syndrome is found in 5-10% of the population. People with Gilbert's syndrome have mild, chronic unconjugated hyperbilirubinemia in the absence of liver disease or overt hemolysis. Another genetic variation causing hyperbilirubinemia is a gene polymorphism designated as UGT1A1*28 [39]. Individuals with UGT1A1*28 allele show decreased liver UGT1A1 activity and increased plasma bilirubin levels.

2.2.2.2. EPIDEMIOLOGICAL STUDIES

The potent antioxidative and anti-inflammatory functions of bilirubin as reported in experimental models have prompted numerous studies on its potential beneficial effects in human diseases. Meta-analysis of a number of epidemiological studies has demonstrated a correlation between increased plasma bilirubin levels and decreased risk of developing coronary heart disease [39, 40]. In addition, multiple epidemiological studies have also suggested an association between elevated plasma bilirubin levels and reduced risk of developing diabetes and its complications [41, 42]. Moreover, there have been several studies indicating a possible association between hyperbilirubinemia and decreased risk of other diseases, including asthma, inflammatory disorders, and certain types of cancer. These observations in human sub-
jects are in line with the findings in animal models, pointing to a potential protective role for bilirubin in human health and disease.

2.3. Summary

Bilirubin and biliverdin possess potent antioxidative and anti-inflammatory activities. These properties contribute to the beneficial effects of these two pigments in various pathophysiological conditions in animal models. Consistent with the findings in experimental models, epidemiological studies also suggest a protective role for bilirubin in human diseases, including coronary heart disease, diabetes, and certain types of cancer.

3. Coenzyme Q

3.1. General Aspects and Biochemistry

3.1.1. Definition

Coenzyme Q (2,3-dimethoxy-5-methyl-6-multiprenyl-1,4-benzoquinone), also known as ubiquinone is a lipid-soluble compound comprised of a redox-active quinoid nucleus and a hydrophobic side chain containing a number of monounsaturated trans-isoprenoid units. The predominant form of coenzyme Q synthesized by mammalian cells is coenzyme Q_{10} (Co Q_{10}), with 10 isoprenoid units in the side chain (Figure 19-3).



Figure 19-3. Structure of coenzyme Q₁₀. See text (Section 3) for description of coenzyme Q₁₀.

3.1.2. Discovery and Biosynthesis

 CoQ_{10} was first isolated from beef heart mitochondria by F.L. Crane and coworkers in 1957 [43]. It is now known that CoQ_{10} biosynthesis occurs mainly in mitochondria. As noted above, CoQ_{10} is composed of a benzoquinone and a decaprenyl side chain. The quinone ring is derived from tyrosine or phenylalanine, whereas the isoprenoid side chain is produced by addition of isopentenyl diphosphate molecules (derived from the mevalonate pathway) to farnesyl diphosphate in multiple steps catalyzed by decaprenyl diphosphate synthase. Decaprenyl diphosphate and para-hydroxybenzoate (PHB) are condensed in a reaction catalyzed by PHB-polyprenyl transferase (COQ2), and the benzoate ring is then modified by at least six enzymes (COQ3-COQ8), which catalyze methylation, decarboxylation, and hydroxylation reactions to eventually form CoQ_{10} [44].

The molecular regulation of mammalian CoQ_{10} biosynthesis remains largely unknown. An early study demonstrated that peroxisome proliferator-activated receptor is involved in regulating CoQ_{10} biosynthesis [45]. A more recent study shows a critical role for NF- κ B in the transcriptional activation of COQ7 gene and the subsequent induction of CoQ_{10} synthesis [46]. On the other hand, mutations have been identified in some of the genes encoding for enzymes involved in CoQ_{10} biosynthesis and have been linked to CoQ_{10} deficiencies in humans [47] (see Section 3.2.2 below).

3.1.3. Localization and Distribution

 CoQ_{10} is an essential electron carrier of the mitochondrial electron transport chain. It is not only present in mitochondrial membrane, but also in other cellular membranes as well as in lipoproteins. The concentrations of CoQ_{10} in human plasma range from 0.40 to 2.0 μ M [48]. The amounts of CoQ_{10} vary in different tissues, with highest levels found in heart, kidney, and liver.

3.1.4. Exogenous CoQ₁₀

 CoQ_{10} is also found naturally in dietary sources, especially meats, and the daily intake of CoQ_{10} from food is estimated to be 3-5 mg. Dietary supplementation is another exogenous source of CoQ_{10} . Because of its insolubility in water, a variety of formulations have been developed to solubilize CoQ_{10} and enhance its absorption. In addition, CoQ_{10} analogs with improved pharmacokinetic profiles have been developed as promising drugs for disease intervention. Among them, idebenone and MitoQ have been investigated in clinical trials for both safety and efficacy [48]. MitoQ is a mitochondria-targeted antioxidant analogue of idebenone. It possesses a terminal positively charged triphenylphosphonium group, which is responsible for targeting the compound to mitochondria as the inner membrane of mitochondria is negatively charged.

3.1.5. Biochemical Properties and Functions

As mentioned above, CoQ_{10} is an essential electron carrier of the mitochondrial electron transport chain and thus indispensible for mitochondrial ATP production. In addition, CoQ_{10} also possesses other important biochemical activities, including (1) scavenging of ROS/RNS, (2) anti-inflammation, and (3) regulation of mitochondrial uncoupling proteins.

The two-electron reduced form of $CoQ_{10} (CoQ_{10}H_2)$, also known as ubiquinol is a much more potent scavenger of ROS/RNS than the oxidized form of CoQ_{10} . NAD(P)H:quinone oxidoreductase 1 (NQO1) is shown to catalyze the two-electron reduction of CoQ_{10} to $CoQ_{10}H_2$ (Section 2.1.2 of Chapter 16). In particular, $CoQ_{10}H_2$ effectively scavenges peroxyl radical and inhibits lipid peroxidation (Reaction 1). $CoQ_{10}H_2$ is also able to regenerate α tocopherol (α -TOH) from its radical form (α -TO[•]) in cellular membranes or lipoproteins (Reaction 2). The resulting coenzyme Q radical ($CoQ_{10}H^{•}$) is relatively stable, and can be reduced back to $CoQ_{10}H_2$ by reducing agents, such as ascorbate (Section 2.2.2 of Chapter 20) and dihydrolipoic acid (Section 5).

$$CoQ_{10}H_2 + LOO' \longrightarrow CoQ_{10}H' + LOOH$$
 (1)

$$CoQ_{10}H_2 + \alpha - TO^{-} \longrightarrow CoQ_{10}H^{-} + \alpha - TOH$$
 (2)

 CoQ_{10} exerts multiple anti-inflammatory effects by suppressing the expression of proinflammatory cytokines and adhesion molecules, and by modulating the activities of inflammatory cells [49]. The antioxidant activities of CoQ_{10} also contribute to its anti-inflammatory effects as oxidative stress and inflammation are intimately related processes. In addition to the well-recognized antioxidative and anti-inflammatory properties, CoQ_{10} is also involved in the regulation of uncoupling proteins. Mitochondrial uncoupling proteins play an important role in coordinating mitochondrial oxidative phosphorylation and heat production, as well as formation of ROS. There is evidence showing CoQ_{10} is an obligatory cofactor for uncoupling protein function [50]. The enhancement of uncoupling protein function by CoQ_{10} may lead to decreased formation of ROS from mitochondrial electron transport chain as uncoupling proteins act to decrease mitochondrial ROS production.

3.2. Role in Health and Disease

3.2.1. Animal Studies

The beneficial effects of both CoQ_{10} and its analogs (e.g., idebenone, MitoQ) have been extensively studied in various experimental systems, including cell cultures and animal models. At physiologically relevant concentrations, CoQ_{10} and analogs afford cytoprotection against ROS/RNS-mediated damage. They also exhibit marked anti-inflammatory activities in cell cultures. In experimental animals, administration of CoQ_{10} and its analogs results in increased levels of these antioxidant compounds in plasma and tissues, including the central nervous system. Numerous studies demonstrate that CoQ_{10} and its analogs protect against various pathophysiological processes involving oxidative and inflammatory stress mechanisms. These include cardiovascular disorders (e.g., hypertension, atherosclerosis, myocardial ischemia-reperfusion injury, heart failure, cardiotoxicity), diabetes and metabolic syndrome, neurodegenerative disorders (e.g., Parkinson's disease, Alzheimer's disease, Huntington's disease), sepsis, skin disorders, male infertility, and cancer. The beneficial effects of CoQ_{10} and its analogs on the above pathophysiological conditions are believed to result from their antioxidative and anti-inflammatory activities as well as their critical involvement in mitochondrial energetics.

3.2.2. Human Studies and Clinical Perspectives

Current information on the role of CoQ_{10} in human health and disease results primarily from studies on CoQ_{10} deficiencies as well as clinical trials on CoQ_{10} supplementation. In addition, the CoQ_{10} analogs idebenone and MitoQ have also been investigated in clinical trials for the intervention of human diseases.

3.2.2.1. HUMAN COQ₁₀ DEFICIENCIES

Deficiencies of CoQ_{10} are classified into primary and secondary deficiencies, and have been associated with four major clinical phenotypes [51]: (1) encephalomyopathy characterized by a triad of recurrent myoglobinuria, brain involvement, and ragged-red fibers, (2) infantile multisystemic disease typically with prominent nephropathy and encephalopathy, (3) cerebellar ataxia with marked cerebellar atrophy, and (4) pure myopathy with lipid storage myopathy and mitochondrial respiratory chain dysfunction.

Primary CoQ₁₀ deficiencies due to mutations in CoQ₁₀ biosynthetic genes have been

identified in patients with the infantile multisystemic and cerebellar ataxic phenotypes. On the other hand, secondary CoQ_{10} deficiencies due to mutations in genes not directly related to ubiquinone biosynthesis have been recognized in patients with cerebellar ataxia and pure myopathy. In addition, CoQ_{10} deficiencies have been reported in patients with other disorders, including Leigh syndrome encephalopathy, growth retardation, deafness, and cardiofaciocutaneous syndrome. In many patients with CoQ_{10} deficiencies, the causative genetic defects are unknown; therefore, it is likely that mutations in additional genes will be identified as causes of CoQ_{10} deficiencies are believed to be attributed to the compromised functions of CoQ_{10} in energy production as well as in protecting against oxidative and inflammatory stress. Numerous studies demonstrate that CoQ_{10} supplementation improves the clinical conditions in patients with CoQ_{10} deficiencies, further supporting a causal role of CoQ_{10} deficiencies in disease pathophysiology [48].

3.2.2.2. CLINICAL STUDIES

In line with the findings in experimental models, supplementation of CoQ_{10} has been shown to provide protection against cardiovascular disorders, diabetes, neurodegenerative diseases, and certain types of cancer in numerous clinical trials [48, 52]. In addition, clinical studies also demonstrate excellent safety profiles for oral CoQ_{10} supplementation. Although a large majority of clinical trials have provided compelling evidence for a beneficial role of CoQ_{10} supplementation in the intervention of common human diseases, inconsistent results are also reported. It is worthy of mentioning that most of the clinical trials so far on CoQ_{10} are relatively small-scaled with potential design limitations, which may explain the inconsistency in conclusions. In this context, several large-scale double-blind, placebo-controlled trials are currently underway to further determine the efficacy of CoQ_{10} supplementation in the intervention of common human diseases [48].

3.3. Summary

In addition to its essential role in mitochondrial ATP production, CoQ_{10} is also a potent antioxidant and anti-inflammatory compound. A growing body of evidence from studies in experimental animals supports CoQ_{10} as an important protector in various disease conditions involving oxidative stress and inflammation. Similarly, the beneficial effects of CoQ_{10} supplementation in human diseases have been suggested by numerous clinical studies. Ongoing and future large-scale double-blind, placebo-controlled trials will further define the exact role of this multifunctional compound in human health and disease.

4. Estrogens

4.1. General Aspects and Biochemistry

The estrogen hormones include estradiol, estrone, and estriol. Physiological levels of estrogens are within the nanomolar range. The beneficial effects of estrogens in human health and disease have been increasingly recognized. However, the underlying molecular mechanisms remain to be elucidated. At non-physiologically relevant micromolar concentrations, estrogens are able to scavenge ROS/RNS and inhibit lipid peroxidation in cell-free systems. Estrogens at micromolar concentrations also induce cellular antioxidant proteins, including superoxide dismutase, glutathione peroxidase, and thioredoxin [53-55]. Induction of antioxidant enzymes by estrogens and estrogen derivatives is shown to be mediated by pathways including estrogen receptor activation as well as Nrf2 signaling [56, 57]. Consistent with these in vitro findings, administration of pharmacological doses of estrogens has been demonstrated to upregulate various tissue antioxidants in experimental animals. In humans, increased estrogen levels also lead to enhanced superoxide dismutase (ECSOD and MnSOD) expression in circulating monocytes [53]. In addition, estrogens are shown to suppress inflammatory responses in cultured cells and animal models [58]. These studies suggest potential antioxidative and anti-inflammatory activities of estrogens at pharmacological doses in vitro and in vivo. However, currently there is no clear evidence showing that estrogens at physiologically relevant concentrations possess any significant in vivo antioxidative and antiinflammatory functions.

4.2. Role in Health and Disease

Consistent with their antioxidative and anti-inflammatory activities, estrogens given to animals at pharmacological doses are found to protect against pathophysiological conditions involving oxidative stress and inflammation. These include cardiovascular diseases, neurological disorders, and others.

4.2.1. Role in Cardiovascular Diseases

Administration of estradiol to animals attenuates hypertension and vascular oxidative stress, and improves endothelial function in spontaneous hypertensive rats [59]. Long-term treatment of ovarectomized ApoE^{-/-} mice with oral estradiol (6 μ g daily for 12 weeks) inhibits the development atherosclerosis [60]. In addition, the estradiol treatment also leads to decreased vascular oxidative stress and induction of vascular Cu,ZnSOD and MnSOD in these mice, suggesting that the antioxidative actions in the vessel wall may contribute to the atheroprotective effects of estrogens. Pharmacological doses of estrogens have also been shown to protect against myocardial ischemia-reperfusion injury in animal models, which is, at least partially, dependent on the antioxidative and anti-inflammatory activities of estrogens [61]. These findings in animal models are consistent with the cardiovascular protective effects of estrogens in humans. However, it should be borne in mind that the beneficial actions of estrogens in human health and disease are likely attributed to the multiple molecular and biochemical activities of these hormonal compounds, rather than their specific antioxidative and anti-inflammatory properties only.

4.2.2. Role in Neurological Diseases and Other Disease Conditions

Extensive studies indicate the neuroprotective effects of estrogens and estrogen derivatives at pharmacological doses in animal models of human neurological disorders, especially cerebral ischemic injury [62, 63]. The neuroprotective actions of estrogens are believed to largely result from their antioxidant activities. Indeed, estrogens are shown to not only inhibit ROS formation from NAD(P)H oxidase, but also induce antioxidant enzymes in brain tissue of experimental animals [63, 64].

Estrogens also exert beneficial effects in other disease conditions. For instance, treatment with estrogens upregulates pancreatic antioxidants and attenuates experimental diabetes and diabetic complications in animal models [65, 66]. Administration of pharmacological doses of estrogens is also found to protect against oxidative multiorgan damage in experimental animals with chronic renal failure [67], attenuate liver ischemia-reperfusion injury and oxidative stress [68], and ameliorate toxicant-induced hepatic fibrosis [69, 70].

4.3. Summary

Estrogens at micromolar concentrations possess antioxidant activities, as evidenced by scavenging ROS/RNS and inducing endogenous cellular antioxidant enzymes. Administration of pharmacological doses of estrogens has been shown to attenuate oxidative and inflammatory pathophysiological conditions in various animal models. These experimental findings are in line with the established beneficial effects of estrogens in humans.

5. α-Lipoic Acid

5.1. General Aspects and Biochemistry

 α -Lipoic Acid (LA; 1,2-dithiolane-3-pentanoic acid) is a naturally occurring sulfurcontaining compound synthesized enzymatically in mitochondria from octanoic acid [71]. LA is a necessary cofactor for mitochondrial α -ketoacid dehydrogenase, and thus plays a critical role in energy metabolism of mammalian cells. In addition to its cellular synthesis, LA is also absorbed intact from dietary sources and distributes to various tissues. Micromolar concentrations of LA in plasma can be achieved following administration of LA in both experimental animals and humans.

LA and its reduced form dihydrolipoic acid (structures shown in Figure 19-4) at micromolar concentrations have been shown to scavenge various ROS/RNS, chelate redox-active metal ions, and restore cellular glutathione levels. LA at micromolar concentrations has also been found to induce endogenous cellular antioxidant enzymes and suppress inflammatory responses [72-74].

5.2. Role in Health and Disease

LA exhibits cytoprotective activities against cell injury elicited by ROS/RNS in vitro. Administration of LA to experimental animals provides protection against many pathophysiological processes involving oxidative stress and dysregulated inflammation. These include cardiovascular disorders, diabetes and metabolic syndrome, neurodegeneration, sepsis, and some others [73].



Figure 19-4. Structures of α -lipoic acid (top) and dihydrolipoic acid (bottom). See text (Section 5) for description of α -lipoic acid.

In humans, LA is used to prevent diabetic polyneuropathies [75]. Clinical trials also demonstrate an efficacy of LA supplementation in attenuating vascular dysfunction and high blood pressure in patients with metabolic syndrome [76, 77].

5.3. Summary

LA is a naturally occurring sulfur compound synthesized in mitochondria, and is also derived from dietary sources. LA at micromolar concentrations scavenges ROS/RNS, chelates redox-active metal ions, and upregulates cellular antioxidant enzymes. LA is also able to suppress inflammatory responses. Administration of LA to animals affords protection against a wide range of disease processes involving oxidative stress and inflammation. Studies in human subjects also show effectiveness of LA supplementation in the intervention of such common diseases as diabetes and metabolic syndrome.

6. Melatonin

6.1. General Aspects and Biochemistry

Melatonin (*N*-acetyl-5-methoxytryptamine) is synthesized in pineal gland of mammals, and the best known function of this molecule is to regulate circadian rhythms. It is also produced by many extrapineal tissues, including retina, ciliary body, lens, Harderian gland, brain, thymus, airway epithelium, bone marrow, gut, ovary, testes, placenta, lymphocytes, and skin. Although the physiological concentrations of melatonin are extremely low in plasma (usually less than 1 nM), the intracellular levels of melatonin in certain tissues, such as skin are estimated to be in micromolar range or even higher [78].

The highest intracellular concentrations of melatonin are found in mitochondria. In this context, melatonin is shown to protect mitochondria from oxidative damage by preventing cardiolipin oxidation. Melatonin is a potent scavenger of ROS/RNS in vitro. Melatonin at micromolar concentrations also induces certain endogenous cellular antioxidant enzymes and inhibits proinflammatory responses. In addition to the antioxidative and anti-inflammatory actions, melatonin participates in other cellular processes, including cell signal transduction, growth regulation, and angiogenesis [79-81].

6.2. Role in Health and Disease

As noted above, melatonin exhibits potent antioxidative and anti-inflammatory activities. At micromolar concentrations, melatonin has been shown to protect cells or tissues from ROS/RNS-mediated damage. Administration of melatonin to experimental animals has been demonstrated to provide protection against various disease conditions, including cardiovascular diseases, diabetes, neurodegeneration, cancer, skin disorders, sepsis, and many other inflammatory conditions [78, 79, 81]. The above beneficial effects of melatonin have also been suggested by extensive clinical studies [78, 79, 81]. The clinical studies also demonstrate excellent safety profiles for oral melatonin supplementation over a wide range of doses. The protective effects of melatonin on oxidative stress and inflammation together with its excellent safety profiles and other beneficial activities make this neurohormone an attractive compound for continued clinical research.

6.3. Summary

A growing body of evidence shows that melatonin is capable of attenuating oxidative and inflammatory stress. Administration of pharmacological doses of melatonin has been shown to be protective in a variety of animal models of human diseases involving oxidative stress and inflammation. Consistently, numerous clinical studies also suggest a beneficial role for melatonin supplementation in the intervention of many common human disease conditions.

7. Pyruvate

7.1. General Aspects and Biochemistry

Pyruvic acid (CH₃COCOOH) is a three-carbon α -keto acid that is present in cells and extracellular fluids as its conjugate anion, pyruvate. Pyruvic acid is the final product of glycolysis and the starting substrate for tricarboxylic acid cycle.

Pyruvic acid reacts nonenzymatically with hydrogen peroxide and protects cells from hydrogen peroxide-induced injury. Pyruvic acid also scavenges other ROS/RNS, such as hydroxyl radical and peroxynitrite. Due to the unstable nature of pyruvic acid in solution, a relatively stable derivative, namely, ethyl pyruvate has been developed and studied for its protective effects on ROS/RNS-induced damage. Ethyl pyruvate also effectively scavenges ROS/RNS. In addition, ethyl pyruvate potently suppresses inflammatory responses via, at least partially, inhibiting NF-KB-mediated expression of proinflammatory cytokines [82, 83].

7.2. Role in Health and Disease

In line with its antioxidative and anti-inflammatory activities, treatment with ethyl pyruvate (or pyruvate in certain studies) has been shown to improve survival and ameliorate organ dysfunction in a variety of preclinical animal models of acute illnesses. These include severe sepsis, hemorrhagic shock, stroke, spinal cord ischemia-reperfusion injury, myocardial ischemia-reperfusion injury, ethanol-induced liver injury, liver ischemia-reperfusion injury, acute pancreatitis, bacterial peritonitis, acute burn injury, and whole-body radiation-induced injury [82, 83]. On the other hand, more prolonged treatment with ethyl pyruvate has been found to ameliorate inflammatory bowel disease and slow the growth rate of malignant tumors in animal models [84-86]. Prompted by these promising preclinical findings, ethyl pyruvate was investigated in a phase 2 clinical trial for the intervention of certain acute conditions in high-risk patients undergoing cardiac surgery with cardiopulmonary bypass [87]. Although ethyl pyruvate was found to be safe, it failed to improve the clinical outcome.

7.3. Summary

Pyruvate reacts nonenzymatically with hydrogen peroxide. The more stable pyruvate derivative, ethyl pyruvate also scavenges ROS/RNS and possesses potent anti-inflammatory activities. Ethyl pyruvate has been shown to be protective against disease pathophysiology in various animal models of human diseases; however, its effectiveness in disease intervention in clinical trials is unclear and warrants further investigations.

8. Uric Acid

8.1. General Aspects and Biochemistry

Uric acid is the end product of purine metabolism in humans. The final two reactions in the metabolism pathway that lead to the formation of uric acid are the conversion of hypoxanthine to xanthine, and xanthine to uric acid. These two reactions are catalyzed by xanthine oxidoreductase. As illustrated in Figure 19-5, in most animal species, uric acid is metabolized to allantoin by urate oxidase. Humans lack urate oxidase, and as such the plasma levels of uric acid in humans are much higher (200-400 μ M) than those in other mammals.

Uric acid is also present intracellularly and in other body fluids, but usually at lower levels than in plasma. At physiological pH, almost all uric acid is ionized to urate. Uric acid at physiologically relevant concentrations is able to scavenge various ROS/RNS, such as peroxyl radical, hydroxyl radical, singlet oxygen, peroxynitrite, and ozone [88-90]. Uric acid also chelates redox-active metal ions, including copper and iron. These findings suggest that uric acid may be a physiologically relevant antioxidant compound in vivo.



Figure 19-5. Formation of uric acid from purine metabolism. As illustrated, xanthine oxidoreductase (XO) catalyzes the formation of xanthine from hypoxanthine. XO further converts xanthine to uric acid. Uric acid is then metabolized by urate oxidase (UO) to allantoin. Humans lack UO, and as such plasma levels of urate in humans are high and usually in the range of 300 to 400 μ M. Hyperuricemia, a condition with abnormally elevated levels of plasma urate, is a major risk factor for the development of gout in humans. (courtesy of Jason Z. Li)

8.2. Role in Health and Disease

8.2.1. Animal Studies

In experimental animals, uric acid is shown to protect against allergic encephalomyelitis and multiple sclerosis likely through its scavenging of peroxynitrite, a reactive species implicated in the pathogenesis of experimental encephalomyelitis and multiple sclerosis [91, 92]. Administration of uric acid at the onset of spinal cord injury in a mouse model attenuates the pathological changes, including oxidative damage and inflammatory cell infiltration in the spinal cord [93]. Uric acid also provides protection in other animal models of human diseases involving oxidative stress and inflammation, such as experimental parkinsonism [94], brain injury induced by acute ischemia [95, 96], and liver injury resulting from hemorrhagic shock [97] or obesity [98]. These findings are consistent with the antioxidant activities of urate.

8.2.2. Human Studies and Clinical Perspectives

In human healthy volunteers, administration of uric acid is shown to increase the serum free radical-scavenging capacity [99] and reduce exercise-induced oxidative stress [100]. Multiple epidemiological studies suggest that high plasma urate concentrations are associated with decreased risk of developing Parkinson's disease [101, 102] as well as other neurological disorders, including Huntington's disease and mild cognitive impairment [103, 104]. Furthermore, a recent clinical study demonstrates that decreased levels of plasma uric acid in acute ischemic stroke are correlated with increased stroke severity, unfavorable disease evolution, and poor long-term clinical outcome [105]. A pilot interventional study reported that in acute stroke patients treated with recombinant tissue plasminogen activator (rt-PA), coadministration of uric acid could reduce lipid peroxidation and prevent a fall of uric acid in blood that occurred early after stroke onset [106]. These clinical and epidemiological observations are in line with the findings in animal studies, pointing to a potential protective role of uric acid in human neurological disorders. However, there have been studies reporting inconsistent or contradictory results on the effects of uric acid in ischemic stroke in human subjects [107, 108], justifying the need for large-scale double-blind, placebo-controlled trials to further determine the efficacy of uric acid in stroke intervention.

On the one hand, there is substantial evidence supporting the antioxidant activities of uric acid in both in vitro systems and animal models, as well as in certain human disorders. On the other hand, epidemiological studies suggest that high levels of plasma uric acid are strongly associated with, and in many cases predict development of hypertension, diabetes, metabolic syndrome, kidney diseases, and cardiovascular events [109-111]. Hence, the effects of uric acid in human diseases vary with pathological conditions, and the potential value of uric acid in disease intervention warrants further clinical studies.

8.3. Summary

At physiologically relevant concentrations, uric acid scavenges ROS/RNS and protects against oxidative cell injury. Administration of uric acid to animals attenuates the pathophysiological processes of various diseases, especially neurological disorders. Epidemiological studies also suggest an inverse association between plasma uric acid levels and risk of developing Parkinson's disease. However, some epidemiological studies also demonstrate a positive correlation between plasma uric acid levels and risk of developing many other diseases, including cardiovascular disorders and diabetes. The causal involvement of uric acid in human diseases and the underlying mechanisms deserve further exploration so as to better define the beneficial and detrimental effects of uric acid in health and disease.

9. Conclusion and Future Directions

Bilirubin, CoQ₁₀, estrogens, LA, melatonin, pyruvate, and uric acid are endogenous nonprotein compounds synthesized by mammalian cells. A growing body of experimental evidence supports that these compounds at physiologically or pharmacologically relevant concentrations possess antioxidative and anti-inflammatory activities. Some of them, such as estrogens, LA, and melatonin are also able to upregulate endogenous cellular antioxidant enzymes. Both animal studies and clinical trials have provided substantial evidence for the beneficial effects of these antioxidant compounds in various disease conditions though other studies have reported inconsistent results. It should be noted that the beneficial effects of these compounds may also result from their non-antioxidant functions. Future studies should focus on characterization of the dose-response relationship when using these antioxidant compounds in disease intervention. It should be kept in mind that although being synthesized by cells, these antioxidant compounds may cause adverse effects when they are supplemented at pharmacological doses.

10. References

- [1] Kirkby KA, Adin CA. Products of heme oxygenase and their potential therapeutic applications. *Am. J. Physiol. Renal. Physiol.* 2006;290:F563-571.
- [2] Stocker R, Yamamoto Y, McDonagh AF, Glazer AN, Ames BN. Bilirubin is an antioxidant of possible physiological importance. *Science* 1987;235:1043-1046.
- [3] Mancuso C, Pani G, Calabrese V. Bilirubin: an endogenous scavenger of nitric oxide and reactive nitrogen species. *Redox Rep.* 2006;11:207-213.
- [4] Baranano DE, Rao M, Ferris CD, Snyder SH. Biliverdin reductase: a major physiologic cytoprotectant. *Proc. Natl. Acad. Sci. USA* 2002;99:16093-16098.
- [5] Sedlak TW, Snyder SH. Bilirubin benefits: cellular protection by a biliverdin reductase antioxidant cycle. *Pediatrics* 2004;113:1776-1782.
- [6] Sedlak TW, Saleh M, Higginson DS, Paul BD, Juluri KR, Snyder SH. Bilirubin and glutathione have complementary antioxidant and cytoprotective roles. *Proc. Natl. Acad. Sci. USA* 2009;106:5171-5176.
- [7] Vera T, Granger JP, Stec DE. Inhibition of bilirubin metabolism induces moderate hyperbilirubinemia and attenuates ANG II-dependent hypertension in mice. *Am. J. Physiol. Regul. Integr. Comp. Physiol.* 2009;297:R738-743.
- [8] Clark JE, Foresti R, Sarathchandra P, Kaur H, Green CJ, Motterlini R. Heme oxygenase-1-derived bilirubin ameliorates postischemic myocardial dysfunction. *Am. J. Physiol. Heart Circ. Physiol.* 2000;278:H643-651.
- [9] Ollinger R, Bilban M, Erat A, Froio A, McDaid J, Tyagi S, et al. Bilirubin: a natural inhibitor of vascular smooth muscle cell proliferation. *Circulation* 2005;112:1030-1039.
- [10] Ollinger R, Yamashita K, Bilban M, Erat A, Kogler P, Thomas M, *et al.* Bilirubin and biliverdin treatment of atherosclerotic diseases. *Cell Cycle* 2007;6:39-43.
- [11] Pflueger A, Croatt AJ, Peterson TE, Smith LA, d'Uscio LV, Katusic ZS, *et al.* The hyperbilirubinemic Gunn rat is resistant to the pressor effects of angiotensin II. *Am. J. Physiol. Renal. Physiol.* 2005;288:F552-558.
- [12] Nath KA, d'Uscio LV, Juncos JP, Croatt AJ, Manriquez MC, Pittock ST, et al. An analysis of the DOCA-salt model of hypertension in HO-1^{-/-} mice and the Gunn rat. Am. J. Physiol. Heart Circ. Physiol. 2007;293:H333-342.
- [13] Mazzone GL, Rigato I, Ostrow JD, Bossi F, Bortoluzzi A, Sukowati CH, et al. Bilirubin inhibits the TNFalpha-related induction of three endothelial adhesion molecules. Biochem. Biophys. Res. Commun. 2009;386:338-344.
- [14] Kim DS, Chae SW, Kim HR, Chae HJ. CO and bilirubin inhibit doxorubicin-induced cardiac cell death. *Immunopharmacol. Immunotoxicol.* 2009;31:64-70.
- [15] Nakao A, Murase N, Ho C, Toyokawa H, Billiar TR, Kanno S. Biliverdin administra-

tion prevents the formation of intimal hyperplasia induced by vascular injury. *Circulation* 2005;112:587-591.

- [16] Rodella L, Lamon BD, Rezzani R, Sangras B, Goodman AI, Falck JR, et al. Carbon monoxide and biliverdin prevent endothelial cell sloughing in rats with type I diabetes. *Free Radic. Biol. Med.* 2006;40:2198-2205.
- [17] Wang H, Lee SS, Dell'Agnello C, Tchipashvili V, d'Avila JC, Czismadia E, *et al.* Bilirubin can induce tolerance to islet allografts. *Endocrinology* 2006;147:762-768.
- [18] Lee SS, Gao W, Mazzola S, Thomas MN, Csizmadia E, Otterbein LE, et al. Heme oxygenase-1, carbon monoxide, and bilirubin induce tolerance in recipients toward islet allografts by modulating T regulatory cells. FASEB J. 2007;21:3450-3457.
- [19] Ostrow JD, Pascolo L, Brites D, Tiribelli C. Molecular basis of bilirubin-induced neurotoxicity. *Trends Mol. Med.* 2004;10:65-70.
- [20] Dore S, Takahashi M, Ferris CD, Zakhary R, Hester LD, Guastella D, et al. Bilirubin, formed by activation of heme oxygenase-2, protects neurons against oxidative stress injury. Proc. Natl. Acad. Sci. USA 1999;96:2445-2450.
- [21] Liu Y, Li P, Lu J, Xiong W, Oger J, Tetzlaff W, et al. Bilirubin possesses powerful immunomodulatory activity and suppresses experimental autoimmune encephalomyelitis. J. Immunol. 2008;181:1887-1897.
- [22] Dennery PA, McDonagh AF, Spitz DR, Rodgers PA, Stevenson DK. Hyperbilirubinemia results in reduced oxidative injury in neonatal Gunn rats exposed to hyperoxia. *Free Radic. Biol. Med.* 1995;19:395-404.
- [23] Sarady-Andrews JK, Liu F, Gallo D, Nakao A, Overhaus M, Ollinger R, et al. Biliverdin administration protects against endotoxin-induced acute lung injury in rats. Am. J. Physiol. Lung Cell Mol. Physiol. 2005;289:L1131-1137.
- [24] Wang HD, Yamaya M, Okinaga S, Jia YX, Kamanaka M, Takahashi H, et al. Bilirubin ameliorates bleomycin-induced pulmonary fibrosis in rats. Am. J. Respir. Crit. Care Med. 2002;165:406-411.
- [25] Keshavan P, Deem TL, Schwemberger SJ, Babcock GF, Cook-Mills JM, Zucker SD. Unconjugated bilirubin inhibits VCAM-1-mediated transendothelial leukocyte migration. J. Immunol. 2005;174:3709-3718.
- [26] Erario MA, Gonzales S, Noriega GO, Tomaro ML. Bilirubin and ferritin as protectors against hemin-induced oxidative stress in rat liver. *Cell Mol. Biol.* (Noisy-le-grand) 2002;48:877-884.
- [27] Fondevila C, Shen XD, Tsuchiyashi S, Yamashita K, Csizmadia E, Lassman C, *et al.* Biliverdin therapy protects rat livers from ischemia and reperfusion injury. *Hepatology* 2004;40:1333-1341.
- [28] Kato Y, Shimazu M, Kondo M, Uchida K, Kumamoto Y, Wakabayashi G, et al. Bilirubin rinse: A simple protectant against the rat liver graft injury mimicking heme oxygenase-1 preconditioning. *Hepatology* 2003;38:364-373.
- [29] Hammerman C, Goldschmidt D, Caplan MS, Kaplan M, Bromiker R, Eidelman AI, *et al.* Protective effect of bilirubin in ischemia-reperfusion injury in the rat intestine. *J. Pediatr. Gastroenterol. Nutr.* 2002;35:344-349.
- [30] Berberat PO, YI AR, Yamashita K, Warny MM, Csizmadia E, Robson SC, et al. Heme oxygenase-1-generated biliverdin ameliorates experimental murine colitis. *Inflamm. Bowel. Dis.* 2005;11:350-359.
- [31] Kirkby K, Baylis C, Agarwal A, Croker B, Archer L, Adin C. Intravenous bilirubin provides incomplete protection against renal ischemia-reperfusion injury in vivo. Am. J. Physiol. Renal Physiol. 2007;292:F888-894.
- [32] Li Volti G, Rodella LF, Di Giacomo C, Rezzani R, Bianchi R, Borsani E, *et al.* Role of carbon monoxide and biliverdin in renal ischemia/reperfusion injury. *Nephron. Exp.*

Nephrol. 2006;104:e135-139.

- [33] Barabas K, Milner R, Farese J, Baylis C, Croker B, Archer L, et al. Hyperbilirubinemia's protective effect against cisplatin nephrotoxicity in the Gunn rat. Anticancer Drugs 2008;19:495-502.
- [34] Bulmer AC, Ried K, Blanchfield JT, Wagner KH. The anti-mutagenic properties of bile pigments. *Mutat. Res.* 2008;658:28-41.
- [35] Rao P, Suzuki R, Mizobuchi S, Yamaguchi T, Sasaguri S. Bilirubin exhibits a novel anti-cancer effect on human adenocarcinoma. *Biochem. Biophys. Res. Commun.* 2006; 342:1279-1283.
- [36] Ollinger R, Kogler P, Troppmair J, Hermann M, Wurm M, Drasche A, *et al.* Bilirubin inhibits tumor cell growth via activation of ERK. *Cell Cycle* 2007;6:3078-3085.
- [37] Lanone S, Bloc S, Foresti R, Almolki A, Taille C, Callebert J, et al. Bilirubin decreases nos2 expression via inhibition of NAD(P)H oxidase: implications for protection against endotoxic shock in rats. FASEB J. 2005;19:1890-1892.
- [38] Kadl A, Pontiller J, Exner M, Leitinger N. Single bolus injection of bilirubin improves the clinical outcome in a mouse model of endotoxemia. *Shock* 2007;28:582-588.
- [39] Schwertner HA, Vitek L. Gilbert syndrome, UGT1A1*28 allele, and cardiovascular disease risk: possible protective effects and therapeutic applications of bilirubin. *Atherosclerosis* 2008;198:1-11.
- [40] Breimer LH, Mikhailidis DP. Bilirubin and peripheral arterial disease: 15 years later. *Expert Opin. Ther. Targets* 2009;13:139-140.
- [41] Ohnaka K, Kono S, Inoguchi T, Yin G, Morita M, Adachi M, *et al.* Inverse associations of serum bilirubin with high sensitivity C-reactive protein, glycated hemoglobin, and prevalence of type 2 diabetes in middle-aged and elderly Japanese men and women. *Diabetes Res. Clin. Pract.* 2010;88:103-110.
- [42] Han SS, Na KY, Chae DW, Kim YS, Kim S, Chin HJ. High serum bilirubin is associated with the reduced risk of diabetes mellitus and diabetic nephropathy. *Tohoku J. Exp. Med.* 2010;221:133-140.
- [43] Crane FL, Hatefi Y, Lester RL, Widmer C. Isolation of a quinone from beef heart mitochondria. *Biochim. Biophys. Acta* 1957;25:220-221.
- [44] Bentinger M, Tekle M, Dallner G. Coenzyme Q: biosynthesis and functions. *Biochem. Biophys. Res. Commun.* 2010;396:74-79.
- [45] Turunen M, Peters JM, Gonzalez FJ, Schedin S, Dallner G. Influence of peroxisome proliferator-activated receptor alpha on ubiquinone biosynthesis. J. Mol. Biol. 2000; 297:607-614.
- [46] Brea-Calvo G, Siendones E, Sanchez-Alcazar JA, de Cabo R, Navas P. Cell survival from chemotherapy depends on NF-kappaB transcriptional up-regulation of coenzyme Q biosynthesis. *PLoS One* 2009;4:e5301.
- [47] DiMauro S, Quinzii CM, Hirano M. Mutations in coenzyme Q10 biosynthetic genes. J. Clin. Invest. 2007;117:587-589.
- [48] Villalba JM, Parrado C, Santos-Gonzalez M, Alcain FJ. Therapeutic use of coenzyme Q(10) and coenzyme Q(10)-related compounds and formulations. *Expert Opin. Investig. Drugs* 2010;19:535-554.
- [49] Schmelzer C, Lindner I, Rimbach G, Niklowitz P, Menke T, Doring F. Functions of coenzyme Q10 in inflammation and gene expression. *Biofactors* 2008;32:179-183.
- [50] Echtay KS, Winkler E, Klingenberg M. Coenzyme Q is an obligatory cofactor for uncoupling protein function. *Nature* 2000;408:609-613.
- [51] Quinzii CM, Lopez LC, Naini A, DiMauro S, Hirano M. Human CoQ10 deficiencies. *Biofactors* 2008;32:113-118.
- [52] Littarru GP, Tiano L. Clinical aspects of coenzyme Q10: an update. Nutrition 2010;

26:250-254.

- [53] Strehlow K, Rotter S, Wassmann S, Adam O, Grohe C, Laufs K, *et al.* Modulation of antioxidant enzyme expression and function by estrogen. *Circ. Res.* 2003;93:170-177.
- [54] Chiueh C, Lee S, Andoh T, Murphy D. Induction of antioxidative and antiapoptotic thioredoxin supports neuroprotective hypothesis of estrogen. *Endocrine* 2003;21:27-31.
- [55] Baltgalvis KA, Greising SM, Warren GL, Lowe DA. Estrogen regulates estrogen receptors and antioxidant gene expression in mouse skeletal muscle. *PLoS One* 2010;5:e10164.
- [56] Lee JM, Anderson PC, Padgitt JK, Hanson JM, Waters CM, Johnson JA. Nrf2, not the estrogen receptor, mediates catechol estrogen-induced activation of the antioxidant responsive element. *Biochim. Biophys. Acta* 2003;1629:92-101.
- [57] Rangwala SM, Li X, Lindsley L, Wang X, Shaughnessy S, Daniels TG, et al. Estrogenrelated receptor alpha is essential for the expression of antioxidant protection genes and mitochondrial function. *Biochem. Biophys. Res. Commun.* 2007;357:231-236.
- [58] Xing D, Nozell S, Chen YF, Hage F, Oparil S. Estrogen and mechanisms of vascular protection. *Arterioscler. Thromb. Vasc. Biol.* 2009;29:289-295.
- [59] Dantas AP, Tostes RC, Fortes ZB, Costa SG, Nigro D, Carvalho MH. In vivo evidence for antioxidant potential of estrogen in microvessels of female spontaneously hypertensive rats. *Hypertension* 2002;39:405-411.
- [60] Wing LY, Chen YC, Shih YY, Cheng JC, Lin YJ, Jiang MJ. Effects of oral estrogen on aortic ROS-generating and -scavenging enzymes and atherosclerosis in apoE-deficient mice. *Exp. Biol. Med.* (Maywood) 2009;234:1037-1046.
- [61] Booth EA, Lucchesi BR. Estrogen-mediated protection in myocardial ischemiareperfusion injury. *Cardiovasc. Toxicol.* 2008;8:101-113.
- [62] Zhang Y, Milatovic D, Aschner M, Feustel PJ, Kimelberg HK. Neuroprotection by tamoxifen in focal cerebral ischemia is not mediated by an agonist action at estrogen receptors but is associated with antioxidant activity. *Exp. Neurol.* 2007;204:819-827.
- [63] Zhang QG, Raz L, Wang R, Han D, De Sevilla L, Yang F, et al. Estrogen attenuates ischemic oxidative damage via an estrogen receptor alpha-mediated inhibition of NADPH oxidase activation. J. Neurosci. 2009;29:13823-13836.
- [64] Ozgonul M, Oge A, Sezer ED, Bayraktar F, Sozmen EY. The effects of estrogen and raloxifene treatment on antioxidant enzymes in brain and liver of ovarectomized female rats. *Endocr. Res.* 2003;29:183-189.
- [65] Hamden K, Carreau S, Boujbiha MA, Lajmi S, Aloulou D, Kchaou D, et al. Hyperglycaemia, stress oxidant, liver dysfunction and histological changes in diabetic male rat pancreas and liver: protective effect of 17beta-estradiol. Steroids 2008;73: 495-501.
- [66] Ceylan-Isik AF, Erdogan-Tulmac OB, Ari N, Ozansoy G, Ren J. Effect of 17betaoestradiol replacement on vascular responsiveness in ovariectomized diabetic rats. *Clin. Exp. Pharmacol. Physiol.* 2009;36:e65-71.
- [67] Kasimay O, Sener G, Cakir B, Yuksel M, Cetinel S, Contuk G, *et al.* Estrogen protects against oxidative multiorgan damage in rats with chronic renal failure. *Ren. Fail.* 2009;31:711-725.
- [68] Shen SQ, Zhang Y, Xiong CL. The protective effects of 17beta-estradiol on hepatic ischemia-reperfusion injury in rat model, associated with regulation of heat-shock protein expression. *J. Surg. Res.* 2007;140:67-76.
- [69] Lu G, Shimizu I, Cui X, Itonaga M, Tamaki K, Fukuno H, *et al.* Antioxidant and antiapoptotic activities of idoxifene and estradiol in hepatic fibrosis in rats. *Life Sci.* 2004;74:897-907.
- [70] Xu JW, Gong J, Chang XM, Luo JY, Dong L, Hao ZM, et al. Estrogen reduces CCl₄-

induced liver fibrosis in rats. World. J. Gastroenterol. 2002;8:883-887.

- [71] Smith AR, Shenvi SV, Widlansky M, Suh JH, Hagen TM. Lipoic acid as a potential therapy for chronic diseases associated with oxidative stress. *Curr. Med. Chem.* 2004; 11:1135-1146.
- [72] Petersen Shay K, Moreau RF, Smith EJ, Hagen TM. Is alpha-lipoic acid a scavenger of reactive oxygen species in vivo? Evidence for its initiation of stress signaling pathways that promote endogenous antioxidant capacity. *IUBMB Life* 2008;60:362-367.
- [73] Shay KP, Moreau RF, Smith EJ, Smith AR, Hagen TM. Alpha-lipoic acid as a dietary supplement: molecular mechanisms and therapeutic potential. *Biochim. Biophys. Acta* 2009;1790:1149-1160.
- [74] Cao Z, Tsang M, Zhao H, Li Y. Induction of endogenous antioxidants and phase 2 enzymes by alpha-lipoic acid in rat cardiac H9C2 cells: protection against oxidative injury. *Biochem. Biophys. Res. Commun.* 2003;310:979-985.
- [75] Singh U, Jialal I. Alpha-lipoic acid supplementation and diabetes. *Nutr. Rev.* 2008;66: 646-657.
- [76] Sola S, Mir MQ, Cheema FA, Khan-Merchant N, Menon RG, Parthasarathy S, et al. Irbesartan and lipoic acid improve endothelial function and reduce markers of inflammation in the metabolic syndrome: results of the Irbesartan and Lipoic Acid in Endothelial Dysfunction (ISLAND) study. *Circulation* 2005;111:343-348.
- [77] Pershadsingh HA. Alpha-lipoic acid: physiologic mechanisms and indications for the treatment of metabolic syndrome. *Expert Opin. Investig. Drugs* 2007;16:291-302.
- [78] Tan DX, Manchester LC, Terron MP, Flores LJ, Reiter RJ. One molecule, many derivatives: a never-ending interaction of melatonin with reactive oxygen and nitrogen species? *J. Pineal Res.* 2007;42:28-42.
- [79] Reiter RJ, Tan DX, Fuentes-Broto L. Melatonin: a multitasking molecule. *Prog. Brain Res.* 2010;181:127-151.
- [80] Park SY, Jang WJ, Yi EY, Jang JY, Jung Y, Jeong JW, *et al.* Melatonin suppresses tumor angiogenesis by inhibiting HIF-1alpha stabilization under hypoxia. *J. Pineal. Res.* 2010;48:178-184.
- [81] Bonnefont-Rousselot D, Collin F. Melatonin: action as antioxidant and potential applications in human disease and aging. *Toxicology* 2010;278:55-67.
- [82] Kao KK, Fink MP. The biochemical basis for the anti-inflammatory and cytoprotective actions of ethyl pyruvate and related compounds. *Biochem. Pharmacol.* 2010;80:151-159.
- [83] Fink MP. Ethyl pyruvate: a novel anti-inflammatory agent. J. Intern. Med. 2007;261: 349-362.
- [84] Dave SH, Tilstra JS, Matsuoka K, Li F, DeMarco RA, Beer-Stolz D, *et al.* Ethyl pyruvate decreases HMGB1 release and ameliorates murine colitis. *J. Leukoc. Biol.* 2009;86:633-643.
- [85] Liang X, Chavez AR, Schapiro NE, Loughran P, Thorne SH, Amoscato AA, *et al.* Ethyl pyruvate administration inhibits hepatic tumor growth. *J. Leukoc. Biol.* 2009; 86:599-607.
- [86] Muller AJ, DuHadaway JB, Jaller D, Curtis P, Metz R, Prendergast GC. Immunotherapeutic suppression of indoleamine 2,3-dioxygenase and tumor growth with ethyl pyruvate. *Cancer Res.* 2010;70:1845-1853.
- [87] Bennett-Guerrero E, Swaminathan M, Grigore AM, Roach GW, Aberle LG, Johnston JM, et al. A phase II multicenter double-blind placebo-controlled study of ethyl pyruvate in high-risk patients undergoing cardiac surgery with cardiopulmonary bypass. J. Cardiothorac. Vasc. Anesth. 2009;23:324-329.
- [88] Ames BN, Cathcart R, Schwiers E, Hochstein P. Uric acid provides an antioxidant

defense in humans against oxidant- and radical-caused aging and cancer: a hypothesis. *Proc. Natl. Acad. Sci. USA* 1981;78:6858-6862.

- [89] Glantzounis GK, Tsimoyiannis EC, Kappas AM, Galaris DA. Uric acid and oxidative stress. *Curr. Pharm. Des.* 2005;11:4145-4151.
- [90] Sautin YY, Johnson RJ. Uric acid: the oxidant-antioxidant paradox. *Nucleosides Nucleotides Nucleic Acids* 2008;27:608-619.
- [91] Hooper DC, Spitsin S, Kean RB, Champion JM, Dickson GM, Chaudhry I, *et al.* Uric acid, a natural scavenger of peroxynitrite, in experimental allergic encephalomyelitis and multiple sclerosis. *Proc. Natl. Acad. Sci. USA* 1998;95:675-680.
- [92] Scott GS, Spitsin SV, Kean RB, Mikheeva T, Koprowski H, Hooper DC. Therapeutic intervention in experimental allergic encephalomyelitis by administration of uric acid precursors. *Proc. Natl. Acad. Sci. USA* 2002;99:16303-16308.
- [93] Scott GS, Cuzzocrea S, Genovese T, Koprowski H, Hooper DC. Uric acid protects against secondary damage after spinal cord injury. *Proc. Natl. Acad. Sci. USA* 2005; 102:3483-3488.
- [94] Duan W, Ladenheim B, Cutler RG, Kruman, II, Cadet JL, Mattson MP. Dietary folate deficiency and elevated homocysteine levels endanger dopaminergic neurons in models of Parkinson's disease. J. Neurochem. 2002;80:101-110.
- [95] Yu ZF, Bruce-Keller AJ, Goodman Y, Mattson MP. Uric acid protects neurons against excitotoxic and metabolic insults in cell culture, and against focal ischemic brain injury in vivo. J. Neurosci. Res. 1998;53:613-625.
- [96] Romanos E, Planas AM, Amaro S, Chamorro A. Uric acid reduces brain damage and improves the benefits of rt-PA in a rat model of thromboembolic stroke. *J. Cereb. Blood Flow Metab.* 2007;27:14-20.
- [97] Tsukada K, Hasegawa T, Tsutsumi S, Katoh H, Kuwano H, Miyazaki T, *et al.* Effect of uric acid on liver injury during hemorrhagic shock. *Surgery* 2000;127:439-446.
- [98] Garcia-Ruiz I, Rodriguez-Juan C, Diaz-Sanjuan T, del Hoyo P, Colina F, Munoz-Yague T, et al. Uric acid and anti-TNF antibody improve mitochondrial dysfunction in ob/ob mice. *Hepatology* 2006;44:581-591.
- [99] Waring WS, Webb DJ, Maxwell SR. Systemic uric acid administration increases serum antioxidant capacity in healthy volunteers. *J. Cardiovasc. Pharmacol.* 2001;38:365-371.
- [100] Waring WS, Convery A, Mishra V, Shenkin A, Webb DJ, Maxwell SR. Uric acid reduces exercise-induced oxidative stress in healthy adults. *Clin. Sci.* (Lond) 2003;105: 425-430.
- [101] Ascherio A, LeWitt PA, Xu K, Eberly S, Watts A, Matson WR, et al. Urate as a predictor of the rate of clinical decline in Parkinson disease. Arch. Neurol. 2009;66: 1460-1468.
- [102] Winquist A, Steenland K, Shankar A. Higher serum uric acid associated with decreased Parkinson's disease prevalence in a large community-based survey. *Mov. Disord.* 2010; 25:932-936.
- [103] Auinger P, Kieburtz K, McDermott MP. The relationship between uric acid levels and Huntington's disease progression. *Mov. Disord.* 2010;25:224-228.
- [104] Irizarry MC, Raman R, Schwarzschild MA, Becerra LM, Thomas RG, Peterson RC, et al. Plasma urate and progression of mild cognitive impairment. *Neurodegener. Dis.* 2009;6:23-28.
- [105] Brouns R, Wauters A, Van De Vijver G, De Surgeloose D, Sheorajpanday R, De Deyn PP. Decrease in uric acid in acute ischemic stroke correlates with stroke severity, evolution and outcome. *Clin. Chem. Lab. Med.* 2010;48:383-390.
- [106] Amaro S, Soy D, Obach V, Cervera A, Planas AM, Chamorro A. A pilot study of dual

treatment with recombinant tissue plasminogen activator and uric acid in acute ischemic stroke. *Stroke* 2007;38:2173-2175.

- [107] Chen JH, Chuang SY, Chen HJ, Yeh WT, Pan WH. Serum uric acid level as an independent risk factor for all-cause, cardiovascular, and ischemic stroke mortality: a Chinese cohort study. *Arthritis Rheum.* 2009;61:225-232.
- [108] Seet RC, Kasiman K, Gruber J, Tang SY, Wong MC, Chang HM, et al. Is uric acid protective or deleterious in acute ischemic stroke? A prospective cohort study. *Atherosclerosis* 2010;209:215-219.
- [109] Lippi G, Montagnana M, Franchini M, Favaloro EJ, Targher G. The paradoxical relationship between serum uric acid and cardiovascular disease. *Clin. Chim. Acta* 2008;392:1-7.
- [110] Cain L, Shankar A, Ducatman AM, Steenland K. The relationship between serum uric acid and chronic kidney disease among Appalachian adults. *Nephrol. Dial. Transplant.* 2010;25:3593-3599.
- [111] Wiik BP, Larstorp AC, Hoieggen A, Kjeldsen SE, Olsen MH, Ibsen H, *et al.* Serum uric acid is associated with new-onset diabetes in hypertensive patients with left ventricular hypertrophy: The LIFE Study. *Am. J. Hypertens.* 2010;23:845-851.

SECTION III ANTIOXIDANTS DERIVED FROM THE DIET

The human diet contains a number of compounds that have antioxidant properties. Among them are vitamin C, vitamin E, carotenoids, and phenolic compounds. These compounds are not made by human cells though vitamin C is synthesized in many mammalian species. It is important to note that these diet-derived non-protein compounds also possess other biological activities in addition to their antioxidant effects. Thus, the role of these compounds in health and disease may results from either the antioxidant or the non-antioxidant activities, or both. The biochemical properties and role in health and disease for each of the above compounds are discussed in the subsequent Chapters 20-23.

Chapter 20

Vitamin C

Abstract

Vitamin C is a water-soluble antioxidant synthesized in many animal species, but not in humans. Humans obtain vitamin C from dietary sources as well as via vitamin supplementation. Vitamin C possesses important biological functions, including serving as a cofactor for many enzymes, acting as an antioxidant, and participating in regulating cell growth, apoptosis, and signaling. Extensive research in animal models has conclusively established diverse beneficial functions of this antioxidant vitamin in a wide variety of disease conditions involving oxidative stress and inflammation. In line with the findings in experimental animals, observational epidemiological studies demonstrate an association between high dietary vitamin C intake and decreased risk of human diseases, including cardiovascular disorders and cancer. However, large-scale interventional trials in nonselected populations have failed to conclude a consistent beneficial role for vitamin C supplementation in cardiovascular diseases and cancer. In contrast, small-scale clinical trials in selected subpopulations of patients with significant oxidative stress have provided evidence supporting a potential beneficial effect of vitamin C in the intervention of certain disease conditions.

1. Overview

1.1. Definition and Dietary Sources

Vitamin C, also known as ascorbic acid or ascorbate, is a water-soluble molecule synthesized endogenously in animals except humans, monkeys, guinea pigs, and several other animal species [1]. Humans lost this capability as a result of a series of inactivating mutations of the gene encoding gulonolactone oxidase (GULO), a key enzyme in vitamin C biosynthesis. Humans normally acquire vitamin C from various dietary sources through a substratesaturable transport mechanism. Dietary sources of vitamin C are mainly vegetables and fruits, including brussels sprouts, broccoli, bell peppers, lettuce, tomatoes, citrus fruits, strawberries, papayas, and mangoes. Because vitamin C is vulnerable to heat, raw fruits and vegetables usually have a higher nutrient density than their cooked counterparts. Another source is vitamin C supplement. Vitamin C is also frequently added to juice.

1.2. Pharmacokinetics

Oral vitamin C intake produces plasma concentrations that are tightly controlled; once its oral intake exceeds 200 mg daily, it is difficult to further increase its plasma concentration. The maximal plasma concentration attainable by oral intake of vitamin C has been estimated to be approximately 200 μ M though the physiological plasma concentrations in healthy humans range from 60 to 100 μ M. In contrast, intravenous injection of large doses of vitamin C produces millimolar concentrations of plasma vitamin C [2].

Under physiological conditions, intracellular levels of vitamin C can be in the millimolar range. This is due to selective intracellular accumulation via vitamin C transport system present in the plasma membrane [3].

1.3. Deficiency

Vitamin C deficiency in diet causes the dread disease scurvy. Other manifestations of vitamin C deficiency may include anemia, poor wound healing, rough skin, blotchy bruises, and frequent infections.

2. Biochemical Properties and Functions

The biochemical properties and functions of vitamin C can be summarized into the following three categories.

- As a cofactor for various enzymes
- As an antioxidant or a potential pro-oxidant
- Other novel activities

2.1. As a Cofactor for Various Enzymes

Vitamin C serves as a cofactor for at least eight enzymes in mammals. The most notable ones are proline hydroxylase and lysine hydroxylase, which are involved in collagen synthesis. The other enzymes for which vitamin C acts as a cofactor are involved in carnitine synthesis, catecholamine synthesis, peptide amidation, and tyrosine metabolism [4]. Due to its critical role in collagen synthesis, deficiency of vitamin C compromises the integrity of blood vessels, leading to scorbutic gums and pinpoint hemorrhage.

2.2. As an Antioxidant or a Potential Pro-oxidant

2.2.1. Redox Chemistry

The redox chemical properties of vitamin C are responsible for its potential antioxidant or pro-oxidant activities. Ascorbic acid (AscH₂) has two ionizable hydroxyl groups. At physio-

logical pH, ascorbic acid exists predominantly as a monoanion, i.e., ascorbate monoanion (AscH⁻). AscH⁻ acts as a reducing agent and is converted to ascorbate radical (Asc⁻, also known as semidehydroascorbate) after donation of one electron. After losing another electron, Asc⁻ is converted to dehydroascorbate (DHA). Asc⁻ can be reduced back to AscH⁻. DHA can also be reduced by either one electron to Asc⁻ or by two electrons to AscH⁻ (Figure 20-1). Due to its high instability, DHA, if not rapidly reduced, undergoes hydrolytic ring opening to 2,3-diketogulonic acid.



Figure 20-1. Vitamin C structure and related redox reactions. See text (Section 2.2.1) for detailed description.

2.2.2. Antioxidant Functions

In many in vitro systems, vitamin C has been found to scavenge various reactive oxygen and nitrogen species (ROS/RNS), and protect cells from oxidative damage. Vitamin C is able to regenerate α -tocopherol and coenzyme Q from α -tocopherol radical and coenzyme Q radical, respectively, and thereby plays a role in maintaining the antioxidant activities of α tocopherol and coenzyme Q. It has recently been demonstrated that vitamin C is also able to reduce 1-Cys peroxiredoxin [5]. As discussed in Section 2.1 of Chapter 11, peroxiredoxin is critical for the detoxification of hydrogen peroxide, organic hydroperoxides, and possibly peroxynitrite. Early studies also showed that vitamin C and glutathione cooperate as an efficient antioxidant system in mammals [6]. In addition to the above activities involved in detoxifying ROS/RNS, vitamin C is found to inhibit NAD(P)H oxidase, thus decreasing the formation of ROS from this important cellular source [7].

2.2.3. Potential Pro-oxidant Activities

Under certain conditions, such as in the presence of redox-active metal ions, vitamin C may behave as a pro-oxidant, giving rise to ROS, damaging DNA, and causing protein glycation. Vitamin C is shown to induce decomposition of lipid hydroperoxides to genotoxins even in the absence of redox-active metal ions [8]. In a humanized mouse model, vitamin C is found to mediate chemical aging of lens crystallins via the Maillard reaction [9]. In experimental animals, administration of pharmacological doses of vitamin C causes formation of vitamin C radical and ROS in extracellular fluid and decreases growth of aggressive tumor xenografts [10] (see Section 3.1.10 below for more discussion of the role of vitamin C in cancer). Hence, the pro-oxidant activities of vitamin C may exert either detrimental or beneficial effects depending on the physiological and pathophysiological conditions.

2.3. Other Novel Activities

Vitamin C reduces ferric ion to ferrous ion and thus facilitates iron absorption in the duodenum. Recent studies suggest an important role for vitamin C in nucleic acid and histone demethylation, as well as proteoglycan deglycanation. Due to its crucial involvement in hydroxylation reactions, vitamin C participates in downregulating hypoxia-inducible transcription factor-1 α (HIF-1 α). In this regard, proline hydroxylation targets HIF-1 α for ubiquitinmediated degradation [11]. Moreover, vitamin C is shown to regulate angiogenesis [12, 13]. These novel mechanisms also contribute to the biological functions of vitamin C in health and disease.

3. Role in Health and Disease

3.1. Animal Studies

3.1.1. Experimental Approaches

A number of approaches have been employed to study the biological activities of vitamin C in experimental animals. These include direct administration of vitamin C either orally or parenterally. As noted earlier in Section 1.2, intravenous injection of large doses of vitamin C produces millimolar plasma concentrations, causing pharmacological effects such as antitumor activity. Due to limited oral bioavailability of vitamin C, lipophilic vitamin C derivatives (e.g., ascorbyl stearate) with improved bioavailability have been developed as potential protectors in animal models of human diseases.

As aforementioned, humans are unable to synthesize vitamin C endogenously due to inactivating mutations of the gene encoding gulonolactone oxidase (GULO). Targeted disruption of GULO gene in mice leads to creation of mutant mice of vitamin C deficiency [14]. This vitamin C deficient GULO knockout mouse model has been used to study the biological functions of vitamin C under experimental conditions. For example, vitamin C deficiency in GULO knockout mice is shown to cause aortic wall damage and augment the instability of atherosclerotic plaque likely due to the compromised synthesis of collagen and enhanced oxidative stress [14, 15].

3.1.2. Role in Cardiovascular Diseases

One of the most extensively investigated biological activities of vitamin C in experimental animals is cardiovascular protection. Numerous studies have conclusively established that vitamin C protects against many forms of cardiovascular disorders in animal models, including hypertension, atherosclerosis, and ischemic heart disease [16-18]. Mechanistic studies have suggested that the beneficial effects of vitamin C in cardiovascular pathophysiology may result from the following mechanisms.

- Protection against ROS/RNS-mediated oxidative modifications of low-density lipoprotein (LDL) [16]
- Maintenance of vascular nitric oxide bioavailability and amelioration of endothelial dysfunction [17]

• Stabilization of atherosclerotic lesion due to production of collagen and inhibition of oxidative stress and inflammation [17].

3.1.3. Role in Diabetes and Metabolic Syndrome

There is evidence supporting the protective effects of vitamin C in diabetes and metabolic syndrome in experimental animals. For example, vitamin C supplementation is found to decrease insulin glycation and improve glucose homeostasis in obese hyperglycemic mice [19]. Administration of vitamin C alone or in combination with vitamin E also attenuates oxidative stress and pancreatic β -cell injury, as well as protects against diabetic complications in various animal models [20-22].

3.1.4. Role in Neurological Diseases

The highest concentrations of vitamin C are found in brain and in neuroendocrine tissues (e.g., adrenal gland). There is increasing experimental evidence supporting that vitamin C is a vital antioxidant in brain. For instance, vitamin C deficiency in GULO knockout mice results in elevated oxidative stress in brain tissue as well as sensorimotor deficits [23]. Neurological disorders, such as ischemic stroke, Alzheimer's disease, Parkinson's disease, and Huntington's disease usually involve oxidative stress. Vitamin C as an important antioxidant in brain has thus been implicated in the protection against these diseases in animal models [24]. Indeed, an early study showed that when monkeys were given 1 g/day of vitamin C parenterally for six days before middle cerebral artery occlusion, brain infarct size was decreased by ~50% in the vitamin C-treated group compared with the control group [25]. Administration of vitamin C also results in amelioration of experimental parkinsonism [26], β -amyloid peptide-induced oxidative damage in rat brain [27], as well as Huntington's behavioral phenotype in mice [28].

The neuroprotection by vitamin C is believed to result mainly from its antioxidant function. In addition, vitamin C is proposed to act as a neuromodulator of glutamatergic, dopaminergic, cholinergic, and GABAergic transmission and related behaviors. Such neurotransmission modulating activities may also contribute to the beneficial effects of vitamin C in neurological disorders [24].

3.1.5. Role in Pulmonary Diseases

Pulmonary protection is also a well-recognized biological function of vitamin C in experimental animals. Early work demonstrated that cigarette smoke depletes vitamin C levels in lung tissue, and either oral or intravenous administration of vitamin C prevents cigarette smoke-induced lung inflammation and oxidative stress [29]. A recent study shows that highdose vitamin supplementation attenuates airway inflammation by modulating Th1/Th2 balance in a murine model of asthma [30]. Vitamin C may also play a role in pulmonary immune response to infections. In this regard, vitamin C deficiency in GULO knockout mice is found to be associated with an increased susceptibility to influenza virus-induced lung inflammation and tissue injury [31].

3.1.6. Role in Hepatic and Gastrointestinal Diseases

Administration of vitamin C or its derivatives to experimental animals ameliorates hepatic ischemia-reperfusion injury and improves survival of liver grafts following transplantation [32-34]. Vitamin C supplementation also protects experimental animals from hepatotoxicity and oxidative stress induced by acetaminophen, ethanol, or carbon tetrachloride. Recently, vitamin C is shown to attenuate liver injury due to metabolic syndrome in a murine model [35]. Although the majority of studies suggest a protective function for vitamin C in various liver diseases, conflicting results have also been reported in the literature. For example, a recent study demonstrates that vitamin C deficiency attenuates liver fibrosis via upregulating peroxisome proliferator-activated receptor- γ expression in senescence marker protein 30 knockout mice [36]. However, the relevance of this finding to the human health remains to be further investigated. In addition to liver diseases, there is evidence showing that vitamin C treatment also protects against gastrointestinal disorders, including intestinal ischemiareperfusion injury and aspirin-induced gastric mucosal damage [37].

3.1.7. Role in Renal Diseases

A number of studies independently demonstrate a beneficial role for vitamin C in renal ischemia-reperfusion injury and kidney transplantation [38, 39]. Such protective effects of vitamin C are associated with decreased oxidative stress and inflammation in renal tissue. Vitamin C alone or in combination with vitamin E also attenuates nephrotoxicity induced by cisplatin, cyclosporine A, or radiocontrast agents. Moreover, vitamin C administration in combination with vitamin E or desferrioxamine ameliorates diabetic nephropathy in animal models [40, 41]. Desferrioxamine is an iron-chelating agent.

3.1.8. Role in Skin Diseases

Ultraviolet (UV) irradiation is found to decrease vitamin C levels in skin tissue, suggesting a potential involvement of this antioxidant vitamin in suppressing UV light-mediated skin damage. Indeed, topical application of vitamin C protects against UV light-induced skin inflammation and oxidative DNA damage. Vitamin C is also effective in protecting against acne vulgaris as well as minocycline pigment formation [42, 43]. Notably, co-application of vitamin C and other antioxidant compounds, such as vitamin E and fullerene (fullerene is a nanomaterial, which is discussed in Chapter 27) is highly effective in ameliorating oxidative and inflammatory skin injury induced by UV irradiation in animal models [44]. This suggests that vitamin C may synergize with other antioxidants in counteracting UV irradiation-induced oxidative stress.

3.1.9. Role in Sepsis

Oxidative stress and dysregulated inflammation are key events leading to multiple organ damage in sepsis. A number of studies in animal models demonstrate that administration of vitamin C protects against both lipopolysaccharide and bacteria-induced sepsis and the subsequent multiple organ damage [45]. Several mechanisms have been proposed to account for the beneficial effects of vitamin C in sepsis. These are listed below.

- Direct scavenging of ROS/RNS formed during sepsis
- Inhibition of activation of inducible nitric oxide synthase (iNOS) and the subsequent formation of nitric oxide during sepsis [46]
- Inhibition of activation of NAD(P)H oxidase and the subsequent formation of ROS as well as attenuation of activation of phosphatase type 2A during sepsis [47]

• Modulation of the functionality of inflammatory cells, resulting in decreased inflammatory responses [48, 49]

3.1.10. Role in Cancer

Early studies by E. Cameron and L. Pauling (a two-time Nobel Laureate) reported that high pharmacological doses of vitamin C could prolong survival of patients with terminal cancer [50, 51] (also see Section 3.2.3). Pharmacological millimolar concentrations of vitamin C are found to selectively kill cancer cells via an ROS-dependent mechanism [10, 52]. High doses of vitamin C given via intravenous injection also exhibit antitumor activities in many animal models of human cancer [10, 13, 53], supporting the early findings by E. Cameron and L. Pauling. The anticancer activities of pharmacological doses of vitamin C in animal models are also in line with the results from mechanistic studies, which are outlined below.

- Vitamin C-mediated inhibition of HIF-1α, a transcription factor involved in tumor cell growth under hypoxic conditions [11]
- Vitamin C-mediated inhibition of angiogenesis [12, 13]
- Vitamin C-mediated stabilization of p53 tumor suppressor [54]
- Vitamin C-mediated apoptosis [55, 56]
- Vitamin C-mediated inhibition of NF-kB, a transcription factor involved in inflammation and oncogenesis [57]

It should be noted that the above molecular and cellular mechanisms underlying vitamin C-mediated antitumor activities are not mutually exclusive. In fact, many of them are closely interrelated. For example, inhibition of HIF-1 α may also lead to suppression of angiogenesis. Similarly, stabilization of p53 may promote this tumor suppressor-dependent cell cycle arrest and apoptosis.

3.2. Human Studies and Clinical Perspectives

3.2.1. Study Approaches

The role of vitamin C in human disease conditions has been investigated by applying various study approaches, including observational epidemiological studies, randomized controlled clinical trials, and case reports. Although treatment of scurvy has been so far the only fully established therapeutic use of vitamin C with a proven mechanism of action in humans, there are increasing numbers of studies suggesting a potential beneficial effect for vitamin C supplementation in the intervention of many common human diseases, such as cardiovascular disorders and cancer.

3.2.2. Role in Human Cardiovascular Diseases

It has long been recognized that high dietary intake of vegetables and fruits is associated with decreased risk of developing cardiovascular diseases. This association is particularly attributable to antioxidant compounds, such as vitamins C and E, present in these foods. Many observational epidemiological studies and randomized clinical trials have examined the relationship between antioxidant vitamin supplementation and incidence of cardiovascular diseases [58]. However, the results and conclusions of these studies are not consistent. While observational epidemiological studies suggest an association between high vitamin C intake and decreased incidence of cardiovascular diseases, large-scale randomized, placebocontrolled trials in non-selected populations largely fail to conclude a beneficial effect of vitamin C supplementation in cardiovascular diseases [58]. The findings from these studies have been systematically reviewed and analyzed, and the potential causes of their discrepancy discussed [16, 59]. Among the potential causes are doses, duration of intervention, and confounding effects. In addition, these studies often did not consider the specific physiological conditions (e.g., vitamin C status) of the subjects. Because vitamin C may have opposing effects (antioxidant versus pro-oxidant activities) under different physiological conditions, cancellation of positive and negative outcomes within a pooled large sample population may result in lack of treatment effects. In this context, multiple relatively small-scale clinical trials in selected subpopulations of patients with significant oxidative stress have yielded promising results, supporting a potential therapeutic role for vitamin C alone or in combination with vitamin E in certain forms of cardiovascular diseases, including hypertension and ischemic heart disease [60-62].

3.2.3. Role in Human Cancer

Similar to the findings related to cardiovascular diseases, observational epidemiological studies have also suggested that consumption of diets rich in vitamin C is associated with decreased risk of developing cancer. However, large-scale randomized, placebo-controlled interventional trials have mostly concluded that dietary antioxidant vitamin supplementation does not reduce the incidence of cancer or cancer-related mortality [63]. This section focuses on discussion of the findings related to the utilization of high intravenous doses of vitamin C in therapeutic intervention of human malignancies.

3.2.3.1. A HISTORICAL OVERVIEW ON HIGH-DOSE VITAMIN C THERAPY

As noted earlier in Section 3.1.10, studies conducted by E. Cameron and L. Pauling in the 1970s suggested that large doses (10 g/day via intravenous infusion for about 10 days and orally thereafter) of vitamin C were effective in increasing survival and improving quality of life of terminal cancer patients [50, 51]. Similar findings were also reported by clinical trials in Japan [64]. However, subsequent studies on large doses of vitamin C given orally (10 g/day) at the Mayo Clinic did not reach the same conclusion [65, 66], and the routes of administration (intravenous infusion versus oral intake) may explain the discrepant results. Indeed, as discussed above in Section 1.2, oral vitamin C intake produces plasma concentrations that are tightly controlled, and the maximal plasma concentration attainable by oral intake of vitamin C has been estimated to be approximately 200 μ M. In contrast, intravenous administration of pharmacological doses of vitamin C have been shown to selectively kill cancer cells in vitro [52].

3.2.3.2. RECENT CLINICAL STUDIES ON HIGH-DOSE VITAMIN C THERAPY

Recently, S.J. Padayatty and coworkers reported three well-documented cases of advanced malignancies in which patients had unexpectedly long survival times after receiving high doses of vitamin C via intravenous administration [67]. The high intravenous dosing regimens for the 3 patients were (1) 65 g twice per week for 10 months, (2) 30 g twice per week for 3 months, followed by 30 g once every 1-2 months for 4 years, interspersed with periods of 1-2 months during which the patient had more frequent infusion, and (3) 15 g twice per week for about 7 months followed by 15 g once every 2-3 months for about one year. Although the findings from this study supported a potential therapeutic role for high intravenous doses of vitamin C in cancer, other possibilities could not be excluded to account for the unexpectedly prolonged survival of the patients. In this context, a recent phase 1 clinical trial reported that high-dose intravenous vitamin C (0.4-1.5 g/kg body weight, three times weekly) was well-tolerated but failed to demonstrate anticancer activity when administered to patients with previously treated advanced malignancies [68]. Thus, large-scale randomized, placebo-controlled trials are needed to establish a possible therapeutic value for high intravenous dos-es of vitamin C in human cancer.

3.2.4. Role in Other Human Diseases

Multiple clinical trials have also suggested potential benefits of vitamin C supplementation in the intervention of some other human diseases, including sepsis, gut disorders, and metabolic syndrome.

3.2.4.1. SEPSIS

Parenteral administration of vitamin C may decrease morbidity and mortality in critically ill patients who are septic or at risk of becoming septic. In a randomized double-blind, placebo-controlled trial with 216 critically ill patients, intravenous infusion of vitamin C and vitamin E was associated with a decreased mortality [69]. Another randomized trial with 595 critically ill surgical patients also found that a combination of vitamin C and vitamin E administration begun within 24 hours of traumatic injury or major surgery decreased relative risk of pulmonary edema and multiple organ failure [70]. These two clinical trials were not designed to distinguish between the effects of vitamin C and vitamin E. A third randomized trial on vitamin C alone reported decreased morbidity (as evidenced by significant reductions in edema formation, fluid resuscitation volume, and respiratory dysfunction) for severely burned patient who received high-dose vitamin C therapy intravenously [71].

3.2.4.2. GUT DISORDERS

Dietary vitamin C supplementation may exert beneficial effects in certain gastrointestinal disorders. A randomized controlled trial with 244 Japanese subjects having atrophic gastritis reported that taking vitamin C supplement for 5 years was associated with decreased oxidative stress markers in serum [72]. Multiple clinical trials demonstrate that addition of vitamin C to the standard antibiotic treatment regimens for *Helicobacter pylori* infections may increase *H. pylori* eradication rate [73] or reduce the dosage of the antibiotic clarithromycin without compromising the high eradication efficacy for clarithromycin-susceptible *H. pylori* infection [74].

3.2.4.3. METABOLIC SYNDROME, ANEMIA, ASTHMA, AND SKIN DISORDERS

There have also been clinical trials examining the potential therapeutic value of vitamin C supplementation alone or in combination with other antioxidants in metabolic syndrome, anemia, asthma, and skin disorders. A small-scale randomized, placebo-controlled trial reported that vitamin C infusion was associated with attenuated oxidative stress-mediated ar-

terial dysfunction in patients with metabolic syndrome [75]. Meta-analysis of clinical studies suggests a beneficial role of vitamin C for anemia management in hemodialysis patients [76]. Systematic review and meta-analysis reveal that relatively low dietary intake of vitamin C and vitamin A is associated with significantly increased odds of asthma and wheeze [77]. The therapeutic activities of topical application of vitamin C in skin disorders, such as acne vulgaris and UV light-induced dermatitis have also been demonstrated by randomized controlled clinical trials [42, 78].

4. Conclusion and Future Directions

The antioxidant effects of vitamin C have been well-documented in diverse experimental systems. This antioxidant vitamin exerts beneficial effects in various oxidative stress-associated disease processes in animal models. While multiple large-scale randomized clinical trials in non-selected populations fail to show a benefit of vitamin C supplementation in the intervention of cardiovascular diseases and cancer, many small-scale clinical trials in selected subpopulations of patients with overt oxidative stress have yielded promising results, suggesting a potential value for using vitamin C in disease intervention. Future studies should put more emphasis on developing reliable biomarkers to assess the status of antioxidants and oxidative stress in selected patient subpopulations, and test the efficacy of vitamin C supplementation in patient groups with marked oxidative stress and/or antioxidant deficiency in well-designed clinical trials. There is also a need to further characterize the pharmacokinetics and pharmacodynamics of vitamin C not only in normal individuals, but also in selected patient groups under oxidative stress and/or with antioxidant deficiency. Such studies will provide important insights into the pharmacological basis of vitamin C-based modalities in both preventive and therapeutic intervention of human diseases.

5. References

- [1] Bruno EJ, Jr., Ziegenfuss TN, Landis J. Vitamin C: research update. *Curr. Sports Med. Rep.* 2006;5:177-181.
- [2] Padayatty SJ, Sun H, Wang Y, Riordan HD, Hewitt SM, Katz A, et al. Vitamin C pharmacokinetics: implications for oral and intravenous use. Ann. Intern. Med. 2004; 140:533-537.
- [3] Wilson JX. Regulation of vitamin C transport. Annu. Rev. Nutr. 2005;25:105-125.
- [4] Mandl J, Szarka A, Banhegyi G. Vitamin C: update on physiology and pharmacology. *Br. J. Pharmacol.* 2009;157:1097-1110.
- [5] Monteiro G, Horta BB, Pimenta DC, Augusto O, Netto LE. Reduction of 1-Cys peroxiredoxins by ascorbate changes the thiol-specific antioxidant paradigm, revealing another function of vitamin C. *Proc. Natl. Acad. Sci. USA* 2007;104:4886-4891.
- [6] Meister A. Glutathione-ascorbic acid antioxidant system in animals. J. Biol. Chem. 1994;269:9397-9400.
- [7] Wu F, Schuster DP, Tyml K, Wilson JX. Ascorbate inhibits NADPH oxidase subunit p47phox expression in microvascular endothelial cells. *Free Radic. Biol. Med.* 2007;

42:124-131.

- [8] Lee SH, Oe T, Blair IA. Vitamin C-induced decomposition of lipid hydroperoxides to endogenous genotoxins. *Science* 2001;292:2083-2086.
- [9] Fan X, Reneker LW, Obrenovich ME, Strauch C, Cheng R, Jarvis SM, et al. Vitamin C mediates chemical aging of lens crystallins by the Maillard reaction in a humanized mouse model. Proc. Natl. Acad. Sci. USA 2006;103:16912-16917.
- [10] Chen Q, Espey MG, Sun AY, Pooput C, Kirk KL, Krishna MC, et al. Pharmacologic doses of ascorbate act as a prooxidant and decrease growth of aggressive tumor xenografts in mice. Proc. Natl. Acad. Sci. USA 2008;105:11105-11109.
- [11] Knowles HJ, Raval RR, Harris AL, Ratcliffe PJ. Effect of ascorbate on the activity of hypoxia-inducible factor in cancer cells. *Cancer Res.* 2003;63:1764-1768.
- [12] Mikirova NA, Ichim TE, Riordan NH. Anti-angiogenic effect of high doses of ascorbic acid. J. Transl. Med. 2008;6:50.
- [13] Yeom CH, Lee G, Park JH, Yu J, Park S, Yi SY, *et al.* High dose concentration administration of ascorbic acid inhibits tumor growth in BALB/C mice implanted with sarcoma 180 cancer cells via the restriction of angiogenesis. *J. Transl. Med.* 2009;7:70.
- [14] Maeda N, Hagihara H, Nakata Y, Hiller S, Wilder J, Reddick R. Aortic wall damage in mice unable to synthesize ascorbic acid. *Proc. Natl. Acad. Sci. USA* 2000;97:841-846.
- [15] Nakata Y, Maeda N. Vulnerable atherosclerotic plaque morphology in apolipoprotein E-deficient mice unable to make ascorbic Acid. *Circulation* 2002;105:1485-1490.
- [16] Li Y, Schellhorn HE. New developments and novel therapeutic perspectives for vitamin C. J. Nutr. 2007;137:2171-2184.
- [17] Aguirre R, May JM. Inflammation in the vascular bed: importance of vitamin C. *Pharmacol. Ther* .2008;119:96-103.
- [18] Buttros JB, Bergamaschi CT, Ribeiro DA, Fracalossi AC, Campos RR. Cardioprotective actions of ascorbic acid during isoproterenol-induced acute myocardial infarction in rats. *Pharmacology* 2009;84:29-37.
- [19] Abdel-Wahab YH, O'Harte FP, Mooney MH, Barnett CR, Flatt PR. Vitamin C supplementation decreases insulin glycation and improves glucose homeostasis in obese hyperglycemic (ob/ob) mice. *Metabolism* 2002;51:514-517.
- [20] Hamden K, Boujbiha MA, Masmoudi H, Ayadi FM, Jamoussi K, Elfeki A. Combined vitamins (C and E) and insulin improve oxidative stress and pancreatic and hepatic injury in alloxan diabetic rats. *Biomed. Pharmacother*. 2009;63:95-99.
- [21] Al-Shamsi M, Amin A, Adeghate E. Effect of vitamin C on liver and kidney functions in normal and diabetic rats. Ann. NY Acad. Sci. 2006;1084:371-390.
- [22] Hasanein P, Shahidi S. Effects of combined treatment with vitamins C and E on passive avoidance learning and memory in diabetic rats. *Neurobiol. Learn. Mem.* 2010; 93:472-478.
- [23] Harrison FE, Yu SS, Van Den Bossche KL, Li L, May JM, McDonald MP. Elevated oxidative stress and sensorimotor deficits but normal cognition in mice that cannot synthesize ascorbic acid. *J. Neurochem.* 2008;106:1198-1208.
- [24] Harrison FE, May JM. Vitamin C function in the brain: vital role of the ascorbate transporter SVCT2. *Free Radic. Biol. Med.* 2009;46:719-730.
- [25] Ranjan A, Theodore D, Haran RP, Chandy MJ. Ascorbic acid and focal cerebral ischaemia in a primate model. Acta Neurochir. (Wien) 1993;123:87-91.
- [26] Desole MS, Esposito G, Fresu L, Migheli R, Enrico P, Miele M, et al. Correlation

between 1-methyl-4-phenylpyridinium ion (MPP⁺) levels, ascorbic acid oxidation and glutathione levels in the striatal synaptosomes of the 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine (MPTP)-treated rat. *Neurosci. Lett.* 1993;161:121-123.

- [27] Rosales-Corral S, Tan DX, Reiter RJ, Valdivia-Velazquez M, Martinez-Barboza G, Acosta-Martinez JP, *et al.* Orally administered melatonin reduces oxidative stress and proinflammatory cytokines induced by amyloid-beta peptide in rat brain: a comparative, in vivo study versus vitamin C and E. *J. Pineal. Res.* 2003;35:80-84.
- [28] Rebec GV, Barton SJ, Marseilles AM, Collins K. Ascorbate treatment attenuates the Huntington behavioral phenotype in mice. *Neuroreport* 2003;14:1263-1265.
- [29] Lehr HA, Frei B, Arfors KE. Vitamin C prevents cigarette smoke-induced leukocyte aggregation and adhesion to endothelium in vivo. *Proc. Natl. Acad. Sci. USA* 1994; 91: 7688-7692.
- [30] Chang HH, Chen CS, Lin JY. High dose vitamin C supplementation increases the Th1/Th2 cytokine secretion ratio, but decreases eosinophilic infiltration in bronchoalveolar lavage fluid of ovalbumin-sensitized and challenged mice. J. Agric. Food Chem. 2009;57:10471-10476.
- [31] Li W, Maeda N, Beck MA. Vitamin C deficiency increases the lung pathology of influenza virus-infected gulo^{-/-} mice. J. Nutr. 2006;136:2611-2616.
- [32] Ichiki A, Miyazaki T, Nodera M, Suzuki H, Yanagisawa H. Ascorbate inhibits apoptosis of Kupffer cells during warm ischemia/reperfusion injury. *Hepatogastroenterology* 2008;55:338-344.
- [33] Matsukawa H, Yagi T, Matsuda H, Kawahara H, Yamamoto I, Matsuoka J, et al. Ascorbic acid 2-glucoside prevents sinusoidal endothelial cell apoptosis in supercooled preserved grafts in rat liver transplantation. *Transplant. Proc.* 2000;32:313-317.
- [34] De Tata V, Brizzi S, Saviozzi M, Lazzarotti A, Fierabracci V, Malvaldi G, *et al.* Protective role of dehydroascorbate in rat liver ischemia-reperfusion injury. *J. Surg. Res.* 2005;123:215-221.
- [35] Massip L, Garand C, Paquet ER, Cogger VC, O'Reilly JN, Tworek L, et al. Vitamin C restores healthy aging in a mouse model for Werner syndrome. FASEB J. 2010;24:158-172.
- [36] Park JK, Ki MR, Lee HR, Hong IH, Ji AR, Ishigami A, et al. Vitamin C deficiency attenuates liver fibrosis by way of up-regulated peroxisome proliferator-activated receptor-gamma expression in senescence marker protein 30 knockout mice. *Hepatology* 2010;51:1766-1777.
- [37] Konturek PC, Kania J, Hahn EG, Konturek JW. Ascorbic acid attenuates aspirininduced gastric damage: role of inducible nitric oxide synthase. J. Physiol. Pharmacol. 2006;57 Suppl 5:125-136.
- [38] Korkmaz A, Kolankaya D. The protective effects of ascorbic acid against renal ischemia-reperfusion injury in male rats. *Ren. Fail.* 2009;31:36-43.
- [39] Lee JI, Son HY, Kim MC. Attenuation of ischemia-reperfusion injury by ascorbic acid in the canine renal transplantation. *J. Vet. Sci.* 2006;7:375-379.
- [40] Farvid MS, Jalali M, Siassi F, Hosseini M. Comparison of the effects of vitamins and/or mineral supplementation on glomerular and tubular dysfunction in type 2 diabetes. *Diabetes Care* 2005;28:2458-2464.
- [41] Iino K, Iwase M, Sonoki K, Yoshinari M, Iida M. Combination treatment of vitamin C and desferrioxamine suppresses glomerular superoxide and prostaglandin E production

in diabetic rats. Diabetes Obes. Metab. 2005;7:106-109.

- [42] Klock J, Ikeno H, Ohmori K, Nishikawa T, Vollhardt J, Schehlmann V. Sodium ascorbyl phosphate shows in vitro and in vivo efficacy in the prevention and treatment of acne vulgaris. *Int. J. Cosmet. Sci.* 2005;27:171-176.
- [43] Bowles WH. Protection against minocycline pigment formation by ascorbic acid (vitamin C). J. Esthet. Dent. 1998;10:182-186.
- [44] Ito S, Itoga K, Yamato M, Akamatsu H, Okano T. The co-application effects of fullerene and ascorbic acid on UV-B irradiated mouse skin. *Toxicology* 2010;267:27-38.
- [45] Wilson JX. Mechanism of action of vitamin C in sepsis: ascorbate modulates redox signaling in endothelium. *Biofactors* 2009;35:5-13.
- [46] Wu F, Wilson JX, Tyml K. Ascorbate inhibits iNOS expression and preserves vasoconstrictor responsiveness in skeletal muscle of septic mice. Am. J. Physiol. Regul. Integr. Comp. Physiol. 2003;285:R50-56.
- [47] Han M, Pendem S, Teh SL, Sukumaran DK, Wu F, Wilson JX. Ascorbate protects endothelial barrier function during septic insult: role of protein phosphatase type 2A. *Free Radic. Biol. Med.* 2010;48:128-135.
- [48] Victor VV, Guayerbas N, Puerto M, Medina S, De la Fuente M. Ascorbic acid modulates in vitro the function of macrophages from mice with endotoxic shock. *Immunopharmacology* 2000;46:89-101.
- [49] Carcamo JM, Pedraza A, Borquez-Ojeda O, Golde DW. Vitamin C suppresses TNF alpha-induced NF kappa B activation by inhibiting I kappa B alpha phosphorylation. *Biochemistry* 2002;41:12995-13002.
- [50] Cameron E, Pauling L. Supplemental ascorbate in the supportive treatment of cancer: prolongation of survival times in terminal human cancer. *Proc. Natl. Acad. Sci. USA* 1976;73:3685-3689.
- [51] Cameron E, Pauling L. Supplemental ascorbate in the supportive treatment of cancer: reevaluation of prolongation of survival times in terminal human cancer. *Proc. Natl. Acad. Sci. USA 1978*;75:4538-4542.
- [52] Chen Q, Espey MG, Krishna MC, Mitchell JB, Corpe CP, Buettner GR, *et al.* Pharmacologic ascorbic acid concentrations selectively kill cancer cells: action as a pro-drug to deliver hydrogen peroxide to tissues. *Proc. Natl. Acad. Sci. USA* 2005;102: 13604-13609.
- [53] Verrax J, Calderon PB. Pharmacologic concentrations of ascorbate are achieved by parenteral administration and exhibit antitumoral effects. *Free Radic. Biol. Med.* 2009; 47:32-40.
- [54] Reddy VG, Khanna N, Singh N. Vitamin C augments chemotherapeutic response of cervical carcinoma HeLa cells by stabilizing P53. *Biochem. Biophys. Res. Commun.* 2001;282:409-415.
- [55] Naidu KA, Tang JL, Prockop LD, Nicosia SV, Coppola D. Antiproliferative and apoptotic effect of ascorbyl stearate in human glioblastoma multiforme cells: modulation of insulin-like growth factor-I receptor (IGF-IR) expression. J. Neurooncol. 2001;54:15-22.
- [56] Takemura Y, Satoh M, Satoh K, Hamada H, Sekido Y, Kubota S. High dose of ascorbic acid induces cell death in mesothelioma cells. *Biochem. Biophys. Res. Commun.* 2010;394:249-253.

- [57] Han SS, Kim K, Hahm ER, Lee SJ, Surh YJ, Park HK, et al. L-ascorbic acid represses constitutive activation of NF-kappaB and COX-2 expression in human acute myeloid leukemia, HL-60. J. Cell Biochem. 2004;93:257-270.
- [58] Honarbakhsh S, Schachter M. Vitamins and cardiovascular disease. *Br. J. Nutr.* 2009; 101:1113-1131.
- [59] Lykkesfeldt J, Poulsen HE. Is vitamin C supplementation beneficial? Lessons learned from randomised controlled trials. *Br. J. Nutr.* 2010;103:1251-1259.
- [60] McRae MP. Is vitamin C an effective antihypertensive supplement? A review and analysis of the literature. J. Chiropr. Med. 2006;5:60-64.
- [61] Shinke T, Shite J, Takaoka H, Hata K, Inoue N, Yoshikawa R, *et al.* Vitamin C restores the contractile response to dobutamine and improves myocardial efficiency in patients with heart failure after anterior myocardial infarction. *Am. Heart. J.* 2007;154:645 e641-648.
- [62] Jaxa-Chamiec T, Bednarz B, Herbaczynska-Cedro K, Maciejewski P, Ceremuzynski L. Effects of vitamins C and E on the outcome after acute myocardial infarction in diabetics: a retrospective, hypothesis-generating analysis from the MIVIT study. *Cardiology* 2009;112:219-223.
- [63] Myung SK, Kim Y, Ju W, Choi HJ, Bae WK. Effects of antioxidant supplements on cancer prevention: meta-analysis of randomized controlled trials. *Ann. Oncol.* 2010; 21:166-179.
- [64] Murata A, Morishige F, Yamaguchi H. Prolongation of survival times of terminal cancer patients by administration of large doses of ascorbate. *Int. J. Vitam. Nutr. Res. Suppl.* 1982;23:103-113.
- [65] Creagan ET, Moertel CG, O'Fallon JR, Schutt AJ, O'Connell MJ, Rubin J, *et al.* Failure of high-dose vitamin C (ascorbic acid) therapy to benefit patients with advanced cancer: a controlled trial. *N. Engl. J. Med.* 1979;301:687-690.
- [66] Moertel CG, Fleming TR, Creagan ET, Rubin J, O'Connell MJ, Ames MM. High-dose vitamin C versus placebo in the treatment of patients with advanced cancer who have had no prior chemotherapy: a randomized double-blind comparison. *N. Engl. J. Med.* 1985;312:137-141.
- [67] Padayatty SJ, Riordan HD, Hewitt SM, Katz A, Hoffer LJ, Levine M. Intravenously administered vitamin C as cancer therapy: three cases. *CMAJ* 2006;174:937-942.
- [68] Hoffer LJ, Levine M, Assouline S, Melnychuk D, Padayatty SJ, Rosadiuk K, et al. Phase I clinical trial of i.v. ascorbic acid in advanced malignancy. Ann. Oncol. 2008; 19:1969-1974.
- [69] Crimi E, Liguori A, Condorelli M, Cioffi M, Astuto M, Bontempo P, et al. The beneficial effects of antioxidant supplementation in enteral feeding in critically ill patients: a prospective, randomized, double-blind, placebo-controlled trial. Anesth. Analg. 2004;99:857-863.
- [70] Nathens AB, Neff MJ, Jurkovich GJ, Klotz P, Farver K, Ruzinski JT, *et al.* Randomized, prospective trial of antioxidant supplementation in critically ill surgical patients. *Ann. Surg.* 2002;236:814-822.
- [71] Tanaka H, Matsuda T, Miyagantani Y, Yukioka T, Matsuda H, Shimazaki S. Reduction of resuscitation fluid volumes in severely burned patients using ascorbic acid administration: a randomized, prospective study. *Arch. Surg.* 2000;135:326-331.
- [72] Sasazuki S, Hayashi T, Nakachi K, Sasaki S, Tsubono Y, Okubo S, et al. Protective

effect of vitamin C on oxidative stress: a randomized controlled trial. *Int. J. Vitam. Nutr. Res.* 2008;78:121-128.

- [73] Zojaji H, Talaie R, Mirsattari D, Haghazali M, Molaei M, Mohsenian N, et al. The efficacy of Helicobacter pylori eradication regimen with and without vitamin C supplementation. *Dig. Liver Dis.* 2009;41:644-647.
- [74] Chuang CH, Sheu BS, Kao AW, Cheng HC, Huang AH, Yang HB, et al. Adjuvant effect of vitamin C on omeprazole-amoxicillin-clarithromycin triple therapy for Helicobacter pylori eradication. *Hepatogastroenterology* 2007;54:320-324.
- [75] Cangemi R, Angelico F, Loffredo L, Del Ben M, Pignatelli P, Martini A, et al. Oxidative stress-mediated arterial dysfunction in patients with metabolic syndrome: effect of ascorbic acid. Free Radic. Biol. Med. 2007;43:853-859.
- [76] Deved V, Poyah P, James MT, Tonelli M, Manns BJ, Walsh M, et al. Ascorbic acid for anemia management in hemodialysis patients: a systematic review and meta-analysis. *Am. J. Kidney Dis.* 2009;54:1089-1097.
- [77] Allen S, Britton JR, Leonardi-Bee JA. Association between antioxidant vitamins and asthma outcome measures: systematic review and meta-analysis. *Thorax* 2009;64:610-619.
- [78] Woolery-Lloyd H, Baumann L, Ikeno H. Sodium L-ascorbyl-2-phosphate 5% lotion for the treatment of acne vulgaris: a randomized, double-blind, controlled trial. J. Cosmet. Dermatol. 2010;9:22-27.
Chapter 21

Vitamin E

Abstract

Vitamin E is a nutritional term, and natural dietary vitamin E comprises eight different forms: the α -, β -, γ -, and δ -tocopherols and the α -, β -, γ -, and δ -tocotrienols. Among the eight forms of vitamin E, α -tocopherol has received the most extensive investigations, accounting for over 90% of the total literature on vitamin E research. Recently, the non- α -tocopherol forms of vitamin E have also received attention due to their significant biological activities. The antioxidative and anti-inflammatory effects of vitamin E have been demonstrated in various in vitro and in vivo experimental models. In addition, both tocopherols and tocotrienols also possess other novel biological functions, including modulation of cell signaling, antiproliferation, induction of apoptosis, and antiangiogenesis. These diverse biological activities obviously contribute to the well-established beneficial effects of vitamin E in a variety of disease conditions in animal models. However, large-scale randomized controlled clinical trials on vitamin E (predominantly α tocopherol) in non-selected populations have mostly failed to show a benefit in the intervention of human diseases, including cardiovascular disorders and cancer. In contrast, small-scale clinical trials in selected patient populations provide evidence for a possible beneficial role of vitamin E supplementation in disease intervention.

1. Overview

1.1. Definition and Dietary Sources

Vitamin E was discovered by H.M. Evans and K.S. Bishop in 1922 as an essential dietary factor for reproduction in rats [1]. Vitamin E is a nutritional term, and the major dietary sources of vitamin E include vegetable oils, seeds, and nuts. Natural dietary vitamin E comprises of eight different forms: α -, β -, γ -, and δ -tocopherols and α -, β -, γ -, and δ -tocotrienols. Tocotrienols have unsaturated side chain, whereas tocopherols contain a phytyl tail with three chiral centers that naturally occur in the *RRR* configuration. Among the eight forms of dietary vitamin E, α -tocopherol is the predominant form in human tissues due to efficient retention by liver and distribution to peripheral tissues (see Section 1.4 below). α -Tocopherol is also the most studied vitamin E isoform in animals and humans, presently accounting for over

90% of the total literature on vitamin E research. Other forms of vitamin E, such as γ -tocopherol and tocotrienols have also recently received attention due to their significant biological activities in mammalian systems [2].

1.2. Blood Levels and Tissue Distribution

The normal average plasma concentration of α -tocopherol ranges from 22 to 28 μ M, which is about 10 and 100 times higher than that of γ -tocopherol (2.5 μ M) and δ -tocopherol (0.3 μ M), respectively. In tissues, the highest concentrations of α -tocopherol are found in adipose tissue and the adrenal glands. In contrast to α -tocopherol, the uptake and distribution of tocotrienols in the body are not very efficient with the plasma concentration usually below 1 μ M [3].

1.3. Additional Nomenclatures

The natural form of α -tocopherol is designated as *RRR*- α -tocopherol (formerly called d- α -tocopherol; structure shown in Figure 21-1). As mentioned above, each tocopherol contains a phytyl tail with three chiral centers, thus giving eight stereoisomers. In contrast to the natural *RRR*- α -tocopherol, the synthetic α -tocopherol, designated as *all-rac*- α -tocopherol (formerly called dl- α -tocopherol) contains ~12.5% of the *RRR* form together with seven other stereoisomers that are less biologically active. As such, the natural form of α -tocopherol is believed to be more biologically active than the synthetic form. Recently, a novel form of α -tocopherol, namely, α -tocopheryl phosphate has been reported to be present in tissues and possess biological activities, including modulation of cell signaling [4, 5].



Figure 21-1. Structure of *RRR*- α -tocopherol. See text (Sections 1.1. and 1.3) for description of α -tocopherol and other forms of vitamin E.

1.4. Deficiency

As noted above, α -tocopherol is efficiently retained by liver and distributed to peripheral tissues via circulation. This is due to the presence of a liver protein that selectively recognizes α -tocopherol absorbed from the diet and is involved in the secretion of α -tocopherol into plasma for distribution to peripheral tissues. This protein is named α -tocopherol transfer pro-

tein (α -TTP) [6]. A defect in the gene for α -TTP in humans causes a disease known as ataxia with vitamin E deficiency [7] (see Section 3.2.2 below). In addition to the above genetic defect, insufficient dietary intake may also lead to vitamin E deficiency.

2. Biochemical Properties and Functions

The biological activities of α -tocopherol as well as other forms of vitamin E can be summarized into three general categories, as listed below.

- Antioxidant and pro-oxidant activities
- Anti-inflammatory activities
- Other novel effects

2.1. Antioxidant and Pro-oxidant Effects

Due to high lipophilicity, α -tocopherol and other forms of vitamin E are believed to be the major antioxidants for protecting against lipid peroxidation [8]. In various in vitro systems, both tocopherols and tocotrienols are able to scavenge peroxyl radicals. Indeed, the reaction rate between α -tocopherol and peroxyl radicals is much greater than that between peroxyl radicals and polyunsaturated fatty acid side chains of the membrane lipids. α -Tocopherol also reacts with others reactive oxygen and nitrogen species (ROS/RNS), such as singlet oxygen and peroxynitrite in certain in vitro systems. In addition, α -tocopherol has been shown to decrease ROS formation from NAD(P)H oxidase [9].

Pro-oxidant activities of α -tocopherol have been reported in vitro. α -Tocopherol is able to reduce redox-active metal ions, such as copper and iron. The α -tocopherol radical formed during reaction of α -tocopherol with ROS/RNS is also reactive and can cause hydrogen abstraction from polyunsaturated fatty acid side chains, leading to lipid peroxidation.

2.2. Anti-inflammatory Effects

Various forms of vitamin E have been demonstrated to suppress inflammatory responses in both in vitro and in vivo models. The anti-inflammatory effects of vitamin E may result from the inhibition of NF- κ B, a transcription factor critically involved in the expression of proinflammatory cytokines [10]. The long-chain carboxychromanol metabolites of various forms of vitamin E also potently inhibit cyclooxygenases, important enzymes in the process of inflammation [11].

2.3. Other Novel Effects

 α -Tocopherol and other forms of vitamin E possess many novel activities irrelevant to the antioxidant effects. These include modulation of enzyme activities (e.g., inhibition of pro-

tein kinase C), cell signal transduction, and gene expression. These non-antioxidant effects are believed to be the molecular basis of certain vitamin E-mediated biological actions, including antiproliferation, induction of apoptosis, and antiangiogenesis [2, 3]. In addition, to-cotrienols also decrease cholesterol synthesis by inhibiting 3-hydroxy-3-methylglutaryl coenzyme A reductase, the key enzyme in cholesterol biosynthesis [2]. Moreover, vitamin E isoforms are able to induce the gene expression of enzymes involved in xenobiotic biotransformation [12]. These non-antioxidant functions along with the aforementioned antioxidative and anti-inflammatory properties may explain the diverse roles of vitamin E in health and disease.

3. Role in Health and Disease

3.1. Animal Studies

3.1.1. Experimental Approaches

The biological activities of vitamin E have been extensively investigated in experimental animals via administration of natural vitamin E isoforms, including tocopherols and tocotrienols. Since natural vitamin isoforms are relatively unstable and not soluble in water, several vitamin E derivatives with increased stability and water solubility have been synthesized and studied in biological systems. These include α -tocopherol acetate, α -tocopherol succinate, and α -tocopheryl phosphate [13]. As noted earlier in Section 1.3, α -tocopheryl phosphate may also be formed endogenously. These relatively stable vitamin E derivatives are used in food supplements and cosmetics. In addition, α -tocopherol has been modified by coupling it to a lipophilic triphenylphosphonium cation through an alkyl linker to increase its selective accumulation in mitochondria. This mitochondria-targeted vitamin E derivative (MitoVitE) has been used to protect against pathophysiological conditions involving oxidative mitochondrial damage [13].

As mentioned above (Section 1.4), a defect in the gene for α -tocopherol transfer protein (α -TTP) in humans causes a disease known as ataxia with vitamin E deficiency. Targeted disruption of α -TTP gene in mice has been reported, and the knockout mice develop α -tocopherol deficiency and delayed-onset ataxia, which can be prevented by α -tocopherol supplementation [14]. This α -TTP knockout mouse model has been used to investigate the biological functions of α -tocopherol in disease conditions, including cardiovascular disorders.

3.1.2. Role in Cardiovascular Diseases

As a major form of vitamin E in the body, α -tocopherol has been extensively studied with respect to its benefit in cardiovascular diseases in animal models. There is substantial evidence showing an important role for α -tocopherol in protecting against cardiovascular pathophysiology, including atherosclerosis, hypertension, and myocardial ischemia-reperfusion injury [15-17]. Recently, other forms of vitamin E, such as α -tocopheryl phosphate and tocotrienols are also found to provide significant protection against cardiovascular diseases in animal models [18, 19]. The cardioprotective functions of vitamin E in experimental animals have been related to the following major mechanisms.

- ROS/RNS scavenging activity
- Inhibition of ROS formation from NAD(P)H oxidase
- Anti-inflammatory activity
- Augmentation of endothelial nitric oxide bioavailability
- Inhibition of cholesterol synthesis
- Modulation of cell signaling

3.1.3. Role in Diabetes and Metabolic Syndrome

Oxidative stress and inflammation are critical events underlying diabetes and metabolic syndrome. A large body of evidence suggests a role for natural vitamin E and derivatives in suppressing the development of diabetes in animal models [20, 21]. Various vitamin E isoforms are shown to also attenuate oxidative stress and inflammation in diabetic complications [22]. Supplementation of vitamin E isoforms ameliorates oxidative stress and inflammatory response as well as insulin resistance in animal models of metabolic syndrome [23]. A recent study further demonstrates that tocotrienols improve insulin sensitivity through activating peroxisome proliferator-activated receptors in experimental animals [24].

3.1.4. Role in Neurological Diseases

Depletion of brain α -tocopherol due to targeted disruption of α -TTP gene in mice causes lipid peroxidation of brain tissue along with neurodegeneration and ataxia [14]. These α tocopherol-deficient mice also exhibit augmented β -amyloid accumulation and aggregation likely due to increased lipid peroxidation in brain tissue [25]. These findings indicate that α tocopherol acts as a neuroprotective molecule. In addition to α -tocopherol, other forms of vitamin E (e.g., γ -tocopherol, tocotrienols) have also been shown to exert potent neuroprotective effects in animal models of neurological disorders, including stroke, Parkinson's disease, and diabetic neuropathy [2, 26, 27].

3.1.5. Role in Pulmonary Diseases

A beneficial role for vitamin E in animal models of pulmonary diseases has been demonstrated by multiple studies. Supplementation of both vitamin C and α -tocopherol attenuates acute lung inflammatory response elicited by cigarette smoke in mice [28]. Administration of α -tocopherol alone also protects experimental animals from lung oxidative stress and inflammation elicited by pulmonary toxicants and allergens [29-31]. The role of α -tocopherol in lung diseases has also been examined in α -TTP knockout mice. Severe deficiency of α tocopherol in α -TTP knockout mice leads to exacerbation of acute hyperoxic lung injury [32]. In contrast, as compared with wild-type mice the α -TTP knockout mice with severe α tocopherol deficiency are shown to have blunted airway allergic inflammatory response in an ovalbumin-induced asthma model [33]. The mechanisms by which severe α -tocopherol deficiency differentially modulates airway inflammation in different animal models warrant further investigations.

Recently, the role of γ -tocopherol in pulmonary protection has received more attention. This vitamin E isoform is shown to reduce ozone-induced exacerbation of allergic rhinosinusitis in rats [34]. Administration of γ -tocopherol also prevents airway inflammation in ovalbumin-elicited rhinitis and asthma [35]. Similarly, ozone-mediated enhancement of lower airway allergic inflammation can be prevented by γ -tocopherol treatment in experimental animals [36]. These observations indicate that supplementation with γ -tocopherol may be a novel complementary therapy for allergic airway diseases.

3.1.6. Role in Hepatic and Gastrointestinal Diseases

 α -Tocopherol protects against certain liver disease conditions in experimental animals. These include hepatic ischemia-reperfusion injury, liver transplantation, nonalcoholic fatty liver disease, and toxicant-induced hepatotoxicity [37-40]. The antioxidative and anti-inflammatory activities of α -tocopherol apparently contribute to the above protective effects on liver pathophysiology.

Administration of α -tocopherol or its derivatives has been demonstrated to ameliorate experimental colitis and intestinal oxidative stress in murine models [41]. Dietary supplementation of α -tocopherol also suppresses gastric mucosal damage elicited by *Helicobacter pylori* infection through, at least partially, inhibition of inflammatory cell infiltration [42]. As significant amounts of vitamin E isoforms, including tocotrienols occur in the gut following dietary intake, vitamin E may serve as an important defense against gut inflammation and oxidative stress.

3.1.7. Role in Renal Diseases

Multiple studies demonstrate that administration of α -tocopherol or tocotrienols results in amelioration of the pathophysiological processes underlying various renal diseases, including ischemia-reperfusion injury [43], chronic renal allograft nephropathy [44], diabetic nephropathy [45], and drug/xenobiotic-induced nephrotoxicity [46, 47]. The protective effects are associated with attenuation of oxidative stress and inflammation in renal tissue under these pathophysiological conditions.

3.1.8. Role in Skin Diseases

Similar to vitamin C (Chapter 20), α -tocopherol is present in mammalian skin and may serve as an important line of dermal antioxidant defenses. Topical application of vitamin E isoforms, including tocopherols and tocotrienols exerts dermal protection against certain skin disease conditions, such as ultraviolet (UV) irradiation-induced skin injury, contact dermatitis, and chemically-induced oxidative stress in animal models [48-50]. Notably, both vitamin C and vitamin E have been used in cosmetics [51]. Multiple mechanisms have been proposed to account for the dermal protective effects of vitamin E, which are outlined below.

- Direct ROS/RNS-scavenging effects
- Suppression of cytokine production and inflammatory cell infiltration
- Stabilization of both keratinocytes and mast cells [49, 52]

In addition, administration of an α -tocopherol-derived analog alone or in combination with celecoxib (a selective cyclooxygenas-2 inhibitor) has been shown to reduce the multiplicity of UV irradiation-induced skin cancer in mice [53].

3.1.9. Role in Cancer

Both natural vitamin E and vitamin E derivatives have been extensively investigated with

respect to their anticancer activities. A large body of evidence suggests a critical role for vitamin E in cancer protection in animal models [54, 55]. Early studies have focused on the anticancer activities of α -tocopherol and the potential clinical applications. Recent studies also demonstrate the importance of γ -tocopherol as well as tocotrienols in cancer protection. In fact, under certain conditions these non- α -tocopherol forms of vitamin E have more pronounced anticancer activities than α -tocopherol [2]. The anticancer effects of vitamin E in animal models most likely result from mechanisms involving multiple targets [54-56], as listed below.

- Inhibition of cell cycle
- Induction of apoptosis
- Suppression of angiogenesis
- Regulation of immunity and augmentation of tumor immunosurveillance

3.1.10. Role in Other Diseases and Conditions

Studies in experimental models also suggest a protective role for vitamin E supplementation in some other disease conditions. These include endotoxemia [57], arthritis [58], erectile dysfunction [59], cataract [60], and HIV infection [61]. Notably, a recent study shows that severe deficiency of α -tocopherol in α -TTP knockout mice confers resistance to malarial infection. This increased resistance may result from augmented oxidative damage to the parasites due to diminished antioxidant action of α -tocopherol [62].

3.2. Human Studies and Clinical Perspectives

3.2.1. Study Approaches

The potential beneficial effects of vitamin E in human disease conditions have been investigated by three major approaches: (1) observational epidemiological studies to assess the association between dietary vitamin E intake and risk of disease development, (2) randomized controlled clinical trials to determine the benefits of vitamin E supplementation in preventive or therapeutic intervention of diseases, and (3) clinical case reports on genetic variations or disorders that may potentially affect the metabolism and biological activities of vitamin E. In this context, as described below, a genetic disorder known as ataxia with vitamin E deficiency has been identified in humans. Studies on this genetic disorder and the underlying pathophysiology have provided valuable insights into the biological significance of vitamin E in human health and disease.

3.2.2. Ataxia with Vitamin E Deficiency

Ataxia with vitamin E deficiency (AVED), an autosomal recessive neurodegenerative disease, is caused by a defect in the gene for α -TTP, a protein critically involved in selectively transporting α -tocopherol from liver to plasma and its subsequent distribution to peripheral tissues. The neurological symptoms of AVED include ataxia, dysarthria, hyporeflexia, and decreased vibration sense, sometimes associated with cardiomyopathy and retinitis pigmentosa. Vitamin E supplementation to patients with this disease improves symptoms and prevents disease progression [7]. Pathophysiologically, patients with AVED develop neuropathy characterized as a dying back of the large caliber axons in the sensory nerves. The dying back of the sensory neurons is likely caused by nerve deficiency of α -tocopherol and the subsequent apoptosis [8].

3.2.3. Role in Human Cardiovascular Diseases

Observational epidemiological studies suggest an inverse relationship between consumption of vitamin E and risk of developing cardiovascular diseases. However, randomized controlled interventional trials have been controversial, with some positive findings, many null findings, and some suggestion of harm in certain populations [63-65]. Possible reasons for the controversies include the doses and forms (natural versus synthetic) of the vitamin E used, the time-window of intervention, lack of reliable biomarkers to assess both the oxidative stress and antioxidant status in the human subjects, as well as the general populations included in the trials. It has been suggested that vitamin E may be beneficial to those individuals who are antioxidant-deficient or exposed to increased levels of oxidative stress, such as smokers, diabetics, and elderly patients, emphasizing the importance of subpopulation targeting. In this regard, recent smaller interventional studies with carefully chosen subpopulations, such as those under high levels of oxidative stress, have yielded largely positive results [63]. For example, a randomized controlled clinical trial reported that vitamin E supplementation (400 IU daily) significantly reduced cardiovascular events in a subgroup of middle-aged individuals with both type 2 diabetes and the haptoglobin 2-2 genotype [66]. Nevertheless, additional large-scale randomized controlled trials in selected populations are needed to better understand the effectiveness of vitamin E supplementation in the intervention of human cardiovascular diseases.

It should be noted that almost all of the major clinical trials on vitamin E have only involved the use of α -tocopherol. As discussed above, other isoforms of vitamin E also possess important biological activities, and their beneficial effects have been demonstrated in various animal models of human diseases, including cardiovascular diseases, cancer, and neurological disorders [2]. Thus, future clinical trials should also include the non- α -tocopherol forms of vitamin E for disease intervention.

3.2.4. Role in Human Cancer

Similar to what was found with cardiovascular diseases, observational epidemiological studies also show an inverse association between consumption of vitamin E and risk of cancer development. However, most large-scale randomized controlled trials have failed to show a beneficial role for α -tocopherol in cancer intervention [67-69]. The same reasons as described for the failure of the clinical trials on vitamin E and cardiovascular diseases (Section 3.2.3) could also account for the failure of clinical trials on α -tocopherol and cancer. Of note is the possibility that α -tocopherol may not be the most effective form of vitamin E with respect to cancer intervention. Indeed, as described above, the non- α -tocopherol forms of vitamin E exhibit marked anticancer activities in a variety of animal models of human cancer (Section 3.1.9). Thus, future studies should also focus on the non- α -tocopherol isoforms of vitamin E in human cancer intervention.

Although the majority of the large-scale randomized controlled trials have failed to show a benefit for α -tocopherol in cancer intervention, results from multiple small-scale studies have provided supporting evidence for a possible role of both α -tocopherol and γ -tocopherol in primary prevention of certain specific types of cancer, including gastric noncardia adenocarcinoma, pancreatic cancer, ovarian cancer, and breast cancer [5]. The efficacy of vitamin E supplementation in human cancer intervention awaits further investigations in large-scale randomized controlled trials with better study design, including (1) choosing of the right vitamin E isoforms and the appropriate dosage and duration of intervention, (2) selection of the appropriate patient populations with marked antioxidant deficiency or oxidative stress, and (3) development of reliable biomarkers for assessing both the antioxidant and oxidative stress status in the human subjects.

3.2.5. Role in Other Human Diseases

3.2.5.1. NONALCOHOLIC STEATOHEPATITIS

A recent randomized controlled trial in a total of 247 adults with nonalcoholic steatohepatitis reported that as compared with placebo, intake of vitamin E (*RRR*- α -tocopherol) at a dose of 800 IU daily for 96 weeks was associated with a significantly higher rate of improvement in disease condition as assessed by histological features of nonalcoholic steatohepatitis, inflammation, and serum enzyme levels [70]. The trial concluded that vitamin E is superior to placebo for the treatment of nonalcoholic steatohepatitis in adults without diabetes. This study also examined the effect of pioglitazone, an antidiabetic drug, at a dose of 30 mg daily for 96 weeks, and found no benefit of pioglitazone over placebo for the primary outcome (i.e., the histological features of nonalcoholic steatohepatitis). However, significant benefits of pioglitazone were observed for some of the secondary outcomes (e.g., improvement in insulin resistance) [70].

The above randomized controlled trial did not examine the effectiveness of combined therapy of vitamin E and pioglitazone. In this context, a previous pilot study reported that a combination therapy of vitamin E (400 IU/day) and pioglitazone (30 mg/day) produced marked improvement in nonalcoholic steatohepatitis histology although vitamin E (400 IU/day) alone had no significant effects [71]. The lack of effectiveness of vitamin E therapy in this pilot study could be due to the lower dose of vitamin E (400 IU/day in the pilot study versus 800 IU/day in the randomized controlled trial), shorter duration of treatment, as well as limitations in sample size and study design. The efficacy of vitamin E therapy in nonalcoholic steatohepatitis observed in the randomized controlled trial described above is also in line with the finding in experimental animals that administration of α -tocopherol protects against nonalcoholic fatty liver disease (Section 3.1.6).

3.2.5.2. Additional Diseases and Conditions

In addition to nonalcoholic steatohepatitis, small-scale clinical trials also suggest a possible beneficial role for vitamin E (α -tocopherol or γ -tocopherol) in several other pathophysiological conditions. These include severe sepsis [72], active ulcerative colitis [73], oxidative stress in normal and asthmatic subjects [74], digital ulcers in systemic sclerosis [75], UV irradiation-induced skin injury [76], and HIV infection [77]. Due to the relatively small numbers of patients involved as well as other design limitations associated with these small-scale clinical trials or pilot studies, the potential beneficial effects of vitamin E therapy in the above disease processes need to be further evaluated in well-designed, large-scale randomized controlled trials.



Figure 21-2. Vitamin E isoforms as multitasking molecules. As illustrated, vitamin E isoforms have antioxidative and anti-inflammatory activities. They also participate in modulating cell signal transduction, leading to growth inhibition, induction of apoptosis, and inhibition of angiogenesis. These mechanisms contribute to the beneficial effects of vitamin E in disease conditions involving oxidative stress, inflammation, and growth dysregulation.

4. Conclusion and Future Directions

Vitamin E isoforms are multitasking molecules that possess antioxidative and antiinflammatory activities, and participate in regulating cell signaling, cell proliferation, and cell death (Figure 21-2). The disease-protective effects of vitamin E, especially α -tocopherol have been well-documented by extensive studies in animal models of human diseases involving oxidative stress and inflammation. The benefits of vitamin E (primarily α -tocopherol) have also been suggested by many observational epidemiological studies and small-scale interventional trials in selected patient groups. However large-scale interventional trials in nonselected populations have largely concluded the ineffectiveness of vitamin E (primarily α tocopherol) supplementation in disease intervention. Possible reasons for the controversies include the doses and forms of the vitamin E used, the time-window of intervention, the lack of reliable biomarkers to assess both the oxidative stress and antioxidant status in the human subjects, as well as the general populations included in the trials. Indeed, nearly all major clinical trials on vitamin E have only investigated α -tocopherol. As discussed earlier, other forms of vitamin E, such as γ - and δ -tocopherol as well as tocotrienols also possess significant disease-protective activities. Hence, more research efforts should be devoted to the investigations of these vitamin E isoforms with regard to their biological activities, pharmacokinetics, and efficacy in disease intervention.

5. References

- [1] Evans HM, Bishop KS. On the existence of a hitherto unrecognized dietary factor essential for reproduction. *Science* 1922;56:650-651.
- [2] Sen CK, Khanna S, Roy S. Tocotrienols in health and disease: the other half of the natural vitamin E family. *Mol. Aspects Med.* 2007;28:692-728.
- [3] Zingg JM. Vitamin E: an overview of major research directions. *Mol. Aspects Med.* 2007;28:400-422.
- [4] Gianello R, Libinaki R, Azzi A, Gavin PD, Negis Y, Zingg JM, et al. Alpha-tocopheryl phosphate: a novel, natural form of vitamin E. *Free Radic. Biol. Med.* 2005;39:970-976.
- [5] Galli F, Azzi A. Present trends in vitamin E research. Biofactors 2010;36:33-42.
- [6] Manor D, Morley S. The alpha-tocopherol transfer protein. *Vitam. Horm.* 2007;76:45-65.
- [7] Di Donato I, Bianchi S, Federico A. Ataxia with vitamin E deficiency: update of molecular diagnosis. *Neurol. Sci.* 2010;31:511-515.
- [8] Traber MG, Atkinson J. Vitamin E, antioxidant and nothing more. *Free Radic. Biol. Med.* 2007;43:4-15.
- [9] Cachia O, Benna JE, Pedruzzi E, Descomps B, Gougerot-Pocidalo MA, Leger CL. alpha-tocopherol inhibits the respiratory burst in human monocytes. Attenuation of p47(phox) membrane translocation and phosphorylation. J. Biol. Chem. 1998;273: 32801-32805.
- [10] Glauert HP. Vitamin E and NF-kappaB activation: a review. *Vitam. Horm.* 2007; 76:135-153.
- [11] Jiang Q, Yin X, Lill MA, Danielson ML, Freiser H, Huang J. Long-chain carboxychromanols, metabolites of vitamin E, are potent inhibitors of cyclooxygenases. *Proc. Natl. Acad. Sci. USA* 2008;105:20464-20469.
- [12] Mustacich DJ, Gohil K, Bruno RS, Yan M, Leonard SW, Ho E, *et al.* Alpha-tocopherol modulates genes involved in hepatic xenobiotic pathways in mice. *J. Nutr. Biochem.* 2009;20:469-476.
- [13] Zingg JM. Molecular and cellular activities of vitamin E analogues. *Mini Rev. Med. Chem.* 2007;7:543-558.
- [14] Yokota T, Igarashi K, Uchihara T, Jishage K, Tomita H, Inaba A, et al. Delayed-onset ataxia in mice lacking alpha -tocopherol transfer protein: model for neuronal degeneration caused by chronic oxidative stress. Proc. Natl. Acad. Sci. USA 2001;98: 15185-15190.
- [15] Munteanu A, Zingg JM. Cellular, molecular and clinical aspects of vitamin E on atherosclerosis prevention. *Mol. Aspects Med.* 2007;28:538-590.
- [16] Newaz MA, Nawal NN, Rohaizan CH, Muslim N, Gapor A. alpha-Tocopherol increased nitric oxide synthase activity in blood vessels of spontaneously hypertensive rats. Am. J. Hypertens. 1999;12:839-844.
- [17] Saremi A, Arora R. Vitamin E and cardiovascular disease. Am. J. Ther. 2010;17:e56-65.
- [18] Negis Y, Aytan N, Ozer N, Ogru E, Libinaki R, Gianello R, et al. The effect of tocopheryl phosphates on atherosclerosis progression in rabbits fed with a high

cholesterol diet. Arch. Biochem. Biophys. 2006;450:63-66.

- [19] Mukherjee S, Lekli I, Das M, Azzi A, Das DK. Cardioprotection with alpha-tocopheryl phosphate: amelioration of myocardial ischemia reperfusion injury is linked with its ability to generate a survival signal through Akt activation. *Biochim. Biophys. Acta* 2008;1782:498-503.
- [20] Beales PE, Williams AJ, Albertini MC, Pozzilli P. Vitamin E delays diabetes onset in the non-obese diabetic mouse. *Horm. Metab. Res.* 1994;26:450-452.
- [21] Papaccio G, Baccari GC, Frascatore S, Sellitti S, Pisanti FA. The vitamin-E derivative U-83836-E in the low-dose streptozocin-treated mouse: effects on diabetes development. *Diabetes Res. Clin. Pract.* 1995;30:163-171.
- [22] Pazdro R, Burgess JR. The role of vitamin E and oxidative stress in diabetes complications. *Mech. Ageing Dev.* 2010;131:276-286.
- [23] Shen X, Tang Q, Wu J, Feng Y, Huang J, Cai W. Effect of vitamin E supplementation on oxidative stress in a rat model of diet-induced obesity. *Int. J. Vitam. Nutr. Res.* 2009;79:255-263.
- [24] Fang F, Kang Z, Wong C. Vitamin E tocotrienols improve insulin sensitivity through activating peroxisome proliferator-activated receptors. *Mol. Nutr. Food Res.* 2010;54: 345-352.
- [25] Nishida Y, Ito S, Ohtsuki S, Yamamoto N, Takahashi T, Iwata N, *et al.* Depletion of vitamin E increases amyloid beta accumulation by decreasing its clearances from brain and blood in a mouse model of Alzheimer disease. *J. Biol. Chem.* 2009;284:33400-33408.
- [26] Kuhad A, Chopra K. Tocotrienol attenuates oxidative-nitrosative stress and inflammatory cascade in experimental model of diabetic neuropathy. *Neuropharmacology* 2009;57:456-462.
- [27] Itoh N, Masuo Y, Yoshida Y, Cynshi O, Jishage K, Niki E. gamma-Tocopherol attenuates MPTP-induced dopamine loss more efficiently than alpha-tocopherol in mouse brain. *Neurosci. Lett.* 2006;403:136-140.
- [28] Silva Bezerra F, Valenca SS, Lanzetti M, Pimenta WA, Castro P, Goncalves Koatz VL, *et al.* Alpha-tocopherol and ascorbic acid supplementation reduced acute lung inflammatory response by cigarette smoke in mouse. *Nutrition* 2006;22:1192-1201.
- [29] Deger Y, Yur F, Ertekin A, Mert N, Dede S, Mert H. Protective effect of alphatocopherol on oxidative stress in experimental pulmonary fibrosis in rats. *Cell Biochem. Funct.* 2007;25:633-637.
- [30] Okamoto N, Murata T, Tamai H, Tanaka H, Nagai H. Effects of alpha tocopherol and probucol supplements on allergen-induced airway inflammation and hyperresponsiveness in a mouse model of allergic asthma. *Int. Arch. Allergy Immunol.* 2006;141:172-180.
- [31] Mabalirajan U, Aich J, Leishangthem GD, Sharma SK, Dinda AK, Ghosh B. Effects of vitamin E on mitochondrial dysfunction and asthma features in an experimental allergic murine model. J. Appl. Physiol. 2009;107:1285-1292.
- [32] Yamaoka S, Kim HS, Ogihara T, Oue S, Takitani K, Yoshida Y, *et al.* Severe Vitamin E deficiency exacerbates acute hyperoxic lung injury associated with increased oxidative stress and inflammation. *Free Radic. Res.* 2008;42:602-612.
- [33] Lim Y, Vasu VT, Valacchi G, Leonard S, Aung HH, Schock BC, *et al.* Severe vitamin E deficiency modulates airway allergic inflammatory responses in the murine asthma

model. Free Radic. Res. 2008;42:387-396.

- [34] Wagner JG, Harkema JR, Jiang Q, Illek B, Ames BN, Peden DB. Gamma-tocopherol attenuates ozone-induced exacerbation of allergic rhinosinusitis in rats. *Toxicol. Pathol.* 2009;37:481-491.
- [35] Wagner JG, Jiang Q, Harkema JR, Ames BN, Illek B, Roubey RA, et al. Gammatocopherol prevents airway eosinophilia and mucous cell hyperplasia in experimentally induced allergic rhinitis and asthma. *Clin. Exp. Allergy* 2008;38:501-511.
- [36] Wagner JG, Jiang Q, Harkema JR, Illek B, Patel DD, Ames BN, *et al.* Ozone enhancement of lower airway allergic inflammation is prevented by gamma-tocopherol. *Free Radic. Biol. Med.* 2007;43:1176-1188.
- [37] Evans ZP, Ellett JD, Fariss MW, Schnellmann RG, Schmidt MG, Chavin K. Vitamin E succinate reduces ischemia/reperfusion injury in steatotic livers. *Transplant. Proc.* 2008;40:3327-3329.
- [38] Soden JS, Devereaux MW, Haas JE, Gumpricht E, Dahl R, Gralla J, et al. Subcutaneous vitamin E ameliorates liver injury in an in vivo model of steatocholestasis. *Hepatology* 2007;46:485-495.
- [39] Iida C, Fujii K, Koga E, Washino Y, Kitamura Y, Ichi I, et al. Effect of alphatocopherol on carbon tetrachloride intoxication in the rat liver. Arch. Toxicol. 2009;83: 477-483.
- [40] Sener G, Sehirli AO, Ayanoglu-Dulger G. Protective effects of melatonin, vitamin E and N-acetylcysteine against acetaminophen toxicity in mice: a comparative study. J. *Pineal Res.* 2003;35:61-68.
- [41] Isozaki Y, Yoshida N, Kuroda M, Takagi T, Handa O, Kokura S, *et al.* Effect of a novel water-soluble vitamin E derivative as a cure for TNBS-induced colitis in rats. *Int. J. Mol. Med.* 2006;17:497-502.
- [42] Sugimoto N, Yoshida N, Nakamura Y, Ichikawa H, Naito Y, Okanoue T, *et al.* Influence of vitamin E on gastric mucosal injury induced by Helicobacter pylori infection. *Biofactors* 2006;28:9-19.
- [43] Salehipour M, Monabbati A, Salahi H, Nikeghbalian S, Bahador A, Marvasti VE, et al. Protective effect of parenteral vitamin E on ischemia-reperfusion injury of rabbit kidney. Urology 2010;75:858-861.
- [44] Gottmann U, Oltersdorf J, Schaub M, Knoll T, Back WE, van der Woude FJ, et al. Oxidative stress in chronic renal allograft nephropathy in rats: effects of long-term treatment with carvedilol, BM 91.0228, or alpha-tocopherol. J. Cardiovasc. Pharmacol. 2003;42:442-450.
- [45] Kuhad A, Chopra K. Attenuation of diabetic nephropathy by tocotrienol: involvement of NFkB signaling pathway. *Life Sci.* 2009;84:296-301.
- [46] Akpolat T, Akpolat I, Ozturk H, Sarikaya S, Cosar AM, Bedir A, et al. Effect of vitamin E and pentoxifylline on glycerol-induced acute renal failure. Nephron 2000;84:243-247.
- [47] Zal F, Mostafavi-Pour Z, Vessal M. Comparison of the effects of vitamin E and/or quercetin in attenuating chronic cyclosporine A-induced nephrotoxicity in male rats. *Clin. Exp. Pharmacol. Physiol.* 2007;34:720-724.
- [48] McVean M, Liebler DC. Inhibition of UVB induced DNA photodamage in mouse epidermis by topically applied alpha-tocopherol. *Carcinogenesis* 1997;18:1617-1622.
- [49] Kuriyama K, Shimizu T, Horiguchi T, Watabe M, Abe Y. Vitamin E ointment at high

dose levels suppresses contact dermatitis in rats by stabilizing keratinocytes. *Inflamm. Res.* 2002;51:483-489.

- [50] Rahman S, Bhatia K, Khan AQ, Kaur M, Ahmad F, Rashid H, *et al.* Topically applied vitamin E prevents massive cutaneous inflammatory and oxidative stress responses induced by double application of 12-O-tetradecanoylphorbol-13-acetate (TPA) in mice. *Chem. Biol. Interact.* 2008;172:195-205.
- [51] Manela-Azulay M, Bagatin E. Cosmeceuticals vitamins. *Clin. Dermatol.* 2009;27:469-474.
- [52] Zingg JM. Vitamin E and mast cells. *Vitam. Horm.* 2007;76:393-418.
- [53] Riedel SB, Fischer SM, Sanders BG, Kline K. Vitamin E analog, alpha-tocopherol ether-linked acetic acid analog, alone and in combination with celecoxib, reduces multiplicity of ultraviolet-induced skin cancers in mice. *Anticancer Drugs* 2008;19: 175-181.
- [54] Constantinou C, Papas A, Constantinou AI. Vitamin E and cancer: An insight into the anticancer activities of vitamin E isomers and analogs. *Int. J. Cancer* 2008;123:739-752.
- [55] Ju J, Picinich SC, Yang Z, Zhao Y, Suh N, Kong AN, *et al.* Cancer-preventive activities of tocopherols and tocotrienols. *Carcinogenesis* 2010;31:533-542.
- [56] Nesaretnam K. Multitargeted therapy of cancer by tocotrienols. *Cancer Lett.* 2008;269: 388-395.
- [57] Ochiai J, Takano H, Ichikawa H, Naito Y, Yoshida N, Yanagisawa R, *et al.* A novel water-soluble vitamin E derivative, 2-(alpha-D-glucopyranosyl)methyl-2,5,7,8-tetramethylchroman-6-ol, protects against acute lung injury and mortality in endotoxemic rats. *Shock* 2002;18:580-584.
- [58] Choi EJ, Bae SC, Yu R, Youn J, Sung MK. Dietary vitamin E and quercetin modulate inflammatory responses of collagen-induced arthritis in mice. J. Med. Food 2009;12: 770-775.
- [59] Ushiyama M, Kuramochi T, Yagi S, Katayama S. Antioxidant treatment with alphatocopherol improves erectile function in hypertensive rats. *Hypertens. Res.* 2008;31: 1007-1013.
- [60] Karslioglu I, Ertekin MV, Kocer I, Taysi S, Sezen O, Gepdiremen A, *et al.* Protective role of intramuscularly administered vitamin E on the levels of lipid peroxidation and the activities of antioxidant enzymes in the lens of rats made cataractous with gamma-irradiation. *Eur. J. Ophthalmol.* 2004;14:478-485.
- [61] Heredia A, Davis C, Amoroso A, Taylor G, Le N, Bamba D, *et al.* In vitro suppression of latent HIV-1 activation by vitamin E: potential clinical implications. *AIDS* 2005; 19:836-837.
- [62] Herbas MS, Ueta YY, Ichikawa C, Chiba M, Ishibashi K, Shichiri M, et al. Alphatocopherol transfer protein disruption confers resistance to malarial infection in mice. *Malar. J.* 2010;9:101.
- [63] Willcox BJ, Curb JD, Rodriguez BL. Antioxidants in cardiovascular health and disease: key lessons from epidemiologic studies. Am. J. Cardiol. 2008;101:75D-86D.
- [64] Honarbakhsh S, Schachter M. Vitamins and cardiovascular disease. *Br. J. Nutr.* 2009; 101:1113-1131.
- [65] Cordero Z, Drogan D, Weikert C, Boeing H. Vitamin E and risk of cardiovascular diseases: a review of epidemiologic and clinical trial studies. *Crit. Rev. Food Sci. Nutr.*

2010;50:420-440.

- [66] Milman U, Blum S, Shapira C, Aronson D, Miller-Lotan R, Anbinder Y, et al. Vitamin E supplementation reduces cardiovascular events in a subgroup of middle-aged individuals with both type 2 diabetes mellitus and the haptoglobin 2-2 genotype: a prospective double-blinded clinical trial. Arterioscler. Thromb. Vasc. Biol. 2008;28: 341-347.
- [67] Lippman SM, Klein EA, Goodman PJ, Lucia MS, Thompson IM, Ford LG, *et al.* Effect of selenium and vitamin E on risk of prostate cancer and other cancers: the Selenium and Vitamin E Cancer Prevention Trial (SELECT). *JAMA* 2009;301:39-51.
- [68] Gaziano JM, Glynn RJ, Christen WG, Kurth T, Belanger C, MacFadyen J, et al. Vitamins E and C in the prevention of prostate and total cancer in men: the Physicians' Health Study II randomized controlled trial. JAMA 2009;301:52-62.
- [69] Lin J, Cook NR, Albert C, Zaharris E, Gaziano JM, Van Denburgh M, et al. Vitamins C and E and beta carotene supplementation and cancer risk: a randomized controlled trial. J. Natl. Cancer Inst. 2009;101:14-23.
- [70] Sanyal AJ, Chalasani N, Kowdley KV, McCullough A, Diehl AM, Bass NM, et al. Pioglitazone, vitamin E, or placebo for nonalcoholic steatohepatitis. N. Engl. J. Med. 2010;362:1675-1685.
- [71] Sanyal AJ, Mofrad PS, Contos MJ, Sargeant C, Luketic VA, Sterling RK, *et al.* A pilot study of vitamin E versus vitamin E and pioglitazone for the treatment of nonalcoholic steatohepatitis. *Clin. Gastroenterol. Hepatol.* 2004;2:1107-1115.
- [72] Weber SU, Lehmann LE, Schewe JC, Thiele JT, Schroder S, Book M, *et al.* Low serum alpha-tocopherol and selenium are associated with accelerated apoptosis in severe sepsis. *Biofactors* 2008;33:107-119.
- [73] Mirbagheri SA, Nezami BG, Assa S, Hajimahmoodi M. Rectal administration of dalpha tocopherol for active ulcerative colitis: a preliminary report. World J. Gastroenterol. 2008;14:5990-5995.
- [74] Wiser J, Alexis NE, Jiang Q, Wu W, Robinette C, Roubey R, et al. In vivo gammatocopherol supplementation decreases systemic oxidative stress and cytokine responses of human monocytes in normal and asthmatic subjects. *Free Radic. Biol. Med.* 2008; 45:40-49.
- [75] Fiori G, Galluccio F, Braschi F, Amanzi L, Miniati I, Conforti ML, et al. Vitamin E gel reduces time of healing of digital ulcers in systemic sclerosis. *Clin. Exp. Rheumatol.* 2009;27:51-54.
- [76] Lin JY, Selim MA, Shea CR, Grichnik JM, Omar MM, Monteiro-Riviere NA, et al. UV photoprotection by combination topical antioxidants vitamin C and vitamin E. J. Am. Acad. Dermatol. 2003;48:866-874.
- [77] de Souza Junior O, Treitinger A, Baggio GL, Michelon C, Verdi JC, Cunha J, et al. alpha-Tocopherol as an antiretroviral therapy supplement for HIV-1-infected patients for increased lymphocyte viability. *Clin. Chem. Lab Med.* 2005;43:376-382.

Chapter 22

Carotenoids

Abstract

Carotenoids represent a large family of compounds of over six hundred fat-soluble plant pigments that provide much of the color that we see in nature. The major carotenoids in normal human diets include β -carotene, α -carotene, lycopene, lutein, zeaxanthin, β -cryptoxanthin, and astaxanthin. Some carotenoids, such as β -carotene have pro-vitamin A activity. Carotenoids are effective scavengers of singlet oxygen and peroxyl radicals. They also possess anti-inflammatory activities and other novel functions in mammalian systems. The health benefits of carotenoids are evidenced by their protective effects in a wide variety of disease processes that involve oxidative stress and inflammation in animal models. The beneficial role of carotenoids in various human disease conditions, including cardiovascular disorders and cancer has been suggested by extensive observational epidemiological studies and many small-scale interventional trials. However, large-scale randomized controlled trials on β -carotene supplementation have failed to show a beneficial effect in cardiovascular diseases and cancer. The exact value of carotenoid supplementation in human disease intervention awaits further investigations in welldesigned clinical trials.

1. Overview

1.1. Definition and Dietary Sources

Carotenoids are a family of lipophilic pigmented compounds that are synthesized by plants. In plants, carotenoids contribute to the photosynthetic machinery and protect plant tissues from photodamage [1]. Carotenoids are responsible for the yellow, orange, and red colors of fruits and vegetables, which constitute the major sources of human dietary carotenoids. Over six hundred carotenoids have so far been identified in nature. The major ones in human diet include β -carotene, α -carotene, lycopene, lutein, zeaxanthin, β -cryptoxanthin, and astaxanthin. As shown in Figure 22-1, notable major dietary sources for the above carotenoids include carrot (β -carotene and α -carotene), tomato and watermelon (lycopene), kale and spinach (lutein and zeaxanthin), papaya and peach (β -cryptoxanthin), and salmon (astaxanthin).



Figure 22-1. Major carotenoids in human diet and their dietary sources. See text (Sections 1.1 and 1.2) for description of carotenoids. (courtesy of Jason Z. Li)

1.2. Classification

Based on their chemical composition, carotenoids are classified into two groups: (1) carotenes which are composed of only carbon and hydrogen atoms and (2) xanthophylls which also carry at least one oxygen atom. β -Carotene, α -carotene, and lycopene are predominant members of the carotene group, whereas lutein, zeaxanthin, β -cryptoxanthin, and astaxanthin belong to the xanthophylls group.

1.3. Blood Levels and Tissue Distribution

The tissue and blood levels of carotenoids in humans vary with diet. Carotenoids in human plasma are estimated to be at low micromolar or submicromolar concentrations. The plasma concentrations of carotenoids are considered useful biomarkers of dietary intake of fruits and vegetables. Lutein and zeaxanthin are also present in the retina.

Among the carotenoids, β -carotene, lycopene, lutein, zeaxanthin, and astaxanthin have been extensively studied with respect to their biological activities in both experimental animals and human subjects.

2. Biochemical Properties and Functions

The biochemical properties and functions of carotenoids can be summarized into four general categories, as listed below.

- As a precursor of vitamin A
- Antioxidant activities
- Anti-inflammatory activities
- Other novel activities, such as modulation of cell signaling

2.1. As a Precursor of Vitamin A

 β -Carotene is best known as a precursor of vitamin A. Carotenoids that can be converted to vitamin A also include α -carotene and β -cryptoxanthin [2]. These carotenoids may thus prevent vitamin A deficiency. Vitamin A is essential for growth and development, as well as visual function. In contrast, lycopene, lutein, zeaxanthin, and astaxanthin lack pro-vitamin A activity.

2.2. Antioxidant Activities

Carotenoids are potent quenchers of singlet oxygen [2]. They also scavenge other reactive species, such as peroxyl radicals, nitrogen oxides, and peroxynitrite. In addition, carotenoids, especially lycopene are found to induce cellular antioxidant enzymes via Nrf2dependent signaling mechanism [3]. Due to the extremely low levels of carotenoids in tissues and blood, the in vivo antioxidant effects of these compounds under physiological conditions remain to be established. Carotenoids at physiologically or pharmacologically relevant concentrations have been shown to exert non-antioxidant activities, as described below.

2.3. Anti-inflammatory Activities

Carotenoids suppress inflammatory responses in both in vitro and in vivo systems. The anti-inflammatory activities may result from multiple mechanisms, including detoxification of reactive oxygen and nitrogen species (ROS/RNS) and suppression of proinflammatory cyto-kine formation. In this regard, carotenoids, such as lycopene and astaxanthin are shown to potently inhibit NF- κ B signaling [4, 5]. NF- κ B is a critical transcription factor involved in the regulation of proinflammatory cytokine gene expression. Carotenoids may also inhibit other pathways involved in inflammation, such as cyclooxygenases [6].

2.4. Other Novel Activities

Besides the antioxidant and anti-inflammatory activities, carotenoids have been shown to possess other biological functions, such as inhibition of growth factor signaling and tumor cell growth, modulation of androgen metabolism, induction of apoptosis, immunomodulation, and antiangiogenesis [1, 2, 7, 8]. These novel molecular events also contribute to the beneficial activities of carotenoids in health and disease.

3. Role in Health and Disease

3.1. Animal Studies

The beneficial effects of carotenoids in health and disease have been extensively investigated in experimental animals. The following sections summarize recent findings on the role of carotenoids in protecting against diverse disease processes in animal models.

3.1.1. Role in Cardiovascular Diseases

Substantial evidence from animal studies supports a beneficial role of dietary supplementation of carotenoids in cardiovascular pathophysiology involving oxidative stress and inflammation. These include atherosclerosis [9, 10], hypertension [11], myocardial ischemiareperfusion injury [12], cardiac hypertrophy and heart failure [13], as well as drug-induced cardiotoxicity [14]. The protective carotenoids include β -carotene, lycopene, lutein, zeaxanthin, astaxanthin, and crocetin. It is important to note that different carotenoids may exhibit different cardiovascular protective effects. In this context, astaxanthin has recently received great attention because of its marked cardiovascular protection in animal models [15, 16]. Several mechanisms have been suggested to account for the cardiovascular protective activities of carotenoids, which are outlined below.

- Detoxification of ROS/RNS
- Decreased ROS formation from cellular sources, such as NAD(P)H oxidase
- Increased vascular nitric oxide bioavailability
- Anti-inflammation
- Inhibition of low-density lipoprotein (LDL) oxidation

3.1.2. Role in Diabetes and Metabolic Syndrome

A number of studies have reported a beneficial role for various carotenoids, including lycopene, lutein, zeaxanthin, and astaxanthin in amelioration of experimental diabetes and diabetic complications. Treatment with carotenoids is associated with decreased oxidative stress and inflammation in the pancreatic islets, which apparently contribute to the improvement of diabetic conditions, including insulin resistance, hyperglycemia, as well as development and progression of diabetic complications. Notably, the retina-associated carotenoids lutein and zeaxanthin exhibit a marked protection on diabetes-associated eye disorders, such as cataract and retinopathy [17, 18].

Carotenoids, especially the recently identified seaweed carotenoid fucoxanthin have been shown to protect against obesity and other metabolic abnormalities in animal models of metabolic syndrome [19]. The antiobesity effect of fucoxanthin is mediated, at least partially, by its altering lipid-regulating enzymes and mitochondrial uncoupling proteins of visceral adipose tissue in mice [20]. A beneficial effect of astaxanthin in controlling obesity, hypertension, insulin resistance, and blood glucose levels in animal models has also been reported in the literature [21, 22].

3.1.3. Role in Neurological Diseases

Oxidative stress and inflammation are critical mechanisms underlying various neurological disorders, including cerebral ischemia-reperfusion injury, Parkinson's disease, and Alzheimer's disease. The neuroprotective effects of retinoic acid in the above disorders have been well-recognized in animal models. There is also evidence suggesting a neuroprotective role for carotenoids in experimental animals. For instance, administration of lycopene or dietary intake of tomato powder rich in lycopene is shown to protect experimental animals from cerebral ischemia-reperfusion injury as well as toxin-induced parkinsonism [23-25]. Other carotenoids, including astaxanthin and crocetin have also been reported to exhibit neuroprotection in cerebral ischemia-reperfusion injury and toxin-induced parkinsonism in animal models [26-28].

3.1.4. Role in Pulmonary Diseases

Multiple studies have demonstrated a beneficial role for lycopene in airway inflammatory disorders. Administration of lycopene ameliorates ovalbumin-induced airway inflammation in a murine model of asthma [29]. It is further shown that lycopene supplementation suppresses Th2 responses and lung eosinophil infiltration [30]. In cultured airway epithelial cells, lycopene decreases proinflammatory responses induced by rhinovirus infection and lipopolysaccharide [31]. Oral intake of lycopene or tomato extracts rich in lycopene also prevents cigarette smoke-induced emphysema and alleviates bleomycin-elicited lung fibrosis and inflammation [32, 33]. A recent study in β -carotene 15,15'-monooxygenase-1 knockout mice demonstrates that genetically altered β -carotene supplementation, suggesting a critical involvement of β -carotene in controlling lung inflammation [34].

3.1.5. Role in Hepatic and Gastrointestinal Diseases

The hepatoprotective effects of carotenoids have been demonstrated by numerous studies in animal models of human liver disorders, including ischemia-reperfusion injury [35, 36], nonalcoholic fatty liver disease [37, 38], viral hepatitis [39], and drug/xenobiotic-induced hepatotoxicity [40]. Carotenoids also exhibit a beneficial role in gastrointestinal disorders, such as drug-induced peptic ulcer [41], *Helicobacter pylori* infection [42], inflammatory bowel disease [43], and intestinal ischemia-reperfusion injury [44]. The benefits of carotenoids in liver and gastrointestinal diseases most likely result from their antioxidative and anti-inflammatory activities. In addition, dietary administration of carotenoids, such as lycopene and astaxanthin also results in marked induction of multiple antioxidant enzymes in target tissues, which may further contribute to the protective effects of carotenoids.

It should be mentioned that high doses of carotenoids (e.g., lycopene) are able to induce phase 1 biotransformation enzymes, such as P4502E1, which may lead to bioactivation of alcohol, causing exacerbation of alcoholic liver injury [45]. In addition, the apoptosis-promoting activity of lycopene is found to worsen dextran sulfate sodium-induced acute colitis, an experimental model of human ulcerative colitis [46].

3.1.6. Role in Renal Diseases

Administration of β -carotene is shown to suppress renal ischemia-reperfusion injury in experimental animals, as evidenced by amelioration of renal dysfunction and oxidative stress [47, 48]. In db/db mice, a rodent model of type 2 diabetes, treatment with astaxanthin prevents the progression of diabetic nephropathy [49]. In vitro, astaxanthin is found to protect against oxidative stress, inflammation, and apoptosis in high glucose-exposed proximal tubular epithelial cells [50]. It also protects renal mesangial cells from high glucose-induced oxidative stress and proinflammatory cytokine expression [51]. Notably, astaxanthin accumulates in the mitochondria of mesangial cells and suppresses high glucose-induced mitochondrial ROS formation [51]. These biochemical and cellular mechanisms may contribute to the beneficial effects of astaxanthin in ameliorating diabetic nephropathy observed in experimental animals.

The protective effects of dietary carotenoids, such as lycopene and astaxanthin on drug/xenobiotic-induced nephrotoxicity and oxidative stress have been documented by many studies. The drugs/xenobiotics whose nephrotoxicity can be ameliorated by carotenoids include cisplatin [52], cyclosporine A [53], gentamicin [54], and mercury chloride [55].

3.1.7. Role in Eye Diseases

The macular region of the retina is yellow in color due to the presence of two diet-derived carotenoids lutein and zeaxanthin, which are thus also called macular pigments [56]. These two carotenoids absorb blue light and possess potent antioxidative and anti-inflammatory activities. Lutein and zeaxanthin have been demonstrated to protect against age-related macular degeneration in animal models [56]. They also exert beneficial effects in other eye disorders, including diabetic retinopathy [17], cataract [18], and retinal ischemia-reperfusion injury [57]. In addition to the macular pigments, other carotenoids, such as astaxanthin and lycopene also protect against eye disorders, including age-related macular degeneration and cataract in experimental animals [58, 59]. Notably, the protective effects of carotenoids in the above eye disorders are strongly associated with their ability to suppress inflammation and oxidative stress, as well as modulation of cell signaling. Consistent with the above in vivo findings, carotenoids are shown to protect retinal or lens cells from oxidative and inflammatory stress in vitro [58-61].

3.1.8. Role in Skin Diseases

Carotenoids as light-harvesting pigments play an important part in protecting plants from excessive light and photo-oxidative stress. Carotenoids are also present in mammalian skin tissue. As noted earlier in Section 2.2, these pigments are effective scavengers of singlet oxygen, peroxyl radicals, and other reactive species that participate in photodamage. The role of carotenoids in skin photoprotection has been investigated in animal models and human subjects [2, 62]. There is ample evidence supporting that carotenoids when given via dietary intake can increase their levels in skin tissue and afford protection against ultraviolet (UV) irradiation-induced skin damage, as evidenced by amelioration of erythema, inflammation, and oxidative stress. Carotenoids are shown to also protect skin from environmental pollutant-induced oxidative damage and inflammation [63].

Carotenoids may have a role in skin carcinogenesis. Dietary intake of lutein and zeaxanthin protects against UV irradiation-induced epidermal hyperproliferation and skin carcinogenesis [64, 65]. Dietary administration of β -carotene also inhibits chemically-induced skin carcinogenesis [66]. In contrast, β -carotene is shown to enhance photocarcinogenesis in animal models under certain conditions [67]. Hence, β -carotene may possess both anti- and procarcinogenic activities depending on the experimental conditions.

3.1.9. Role in Cancer

Carotenoids suppress spontaneous development of cancer in various animal models, and protect against multistage carcinogenesis induced by chemical carcinogens [1, 68-70]. The anticancer activities of carotenoids have been observed with almost all of the major types of human cancer in animal models, including lung, liver, breast, prostate, and brain cancer. Carotenoids not only decrease cancer growth, but also suppress cancer invasion and metastasis. Several potential mechanisms have been proposed to account for the anticancer activities of carotenoids in experimental animals [1, 7, 68, 69, 71]. These are listed below.

- Antiproliferation
- Induction of apoptosis
- Antiangiogenesis
- Induction of phase 2 enzymes involved in the detoxification of carcinogens
- Antioxidative stress
- Anti-inflammation
- Induction of gap junctional communication

Antiproliferation, induction of apoptosis, and antiangiogenesis are obviously important mechanisms involved in the anticancer activities of carotenoids. Induction of phase 2 enzymes results in enhanced detoxification of electrophilic carcinogens and diminished formation of carcinogen-DNA adducts, thus suppressing the initiation step of multistage carcinogenesis. As oxidative stress and inflammation are important events in cancer development, suppression of these events by carotenoids may also contribute to inhibition of carcinogenesis. Gap junctional communication allows small molecular signaling between cells through channels that are formed by gap junction proteins, such as connexin 43. Virtually all human tumors are deficient in gap junctional communication, and restoration of gap junctional communication allows suppresses the phenotypes of neoplasia.

Carotenoids by inducing connexin 43 have been found to augment gap junctional communication, which may play an important role in their in vivo anticancer activities [71].

3.2. Human Studies and Clinical Perspectives

3.2.1. Role in Human Cardiovascular Diseases

There have been a number of observational epidemiological studies showing an inverse relationship between dietary intakes or blood levels of carotenoids and risk of developing cardiovascular diseases, including atherosclerosis, hypertension, and coronary heart disease [1, 72, 73]. These observations are in line with the experimental findings that dietary carotenoids afford significant cardiovascular protection in animal models of human cardiovascular diseases (Section 3.1.1). The results from observational epidemiological studies are also consistent with the well-recognized ability of carotenoids to suppress oxidative stress and inflammation in experimental systems. In this context, oxidative stress and dysregulated inflammation have been increasingly recognized as important mechanisms underlying human cardiovascular diseases.

Although the potential cardiovascular benefits of carotenoids have been suggested by extensive observational epidemiological studies, large-scale randomized controlled trials have mostly failed to show an efficacy of β -carotene supplementation in the intervention of human cardiovascular diseases [1, 72, 73]. The lack of benefits in these trials could be due to the relatively short duration of intervention, use of non-optimal dosages of β -carotene, and the general populations involved in the studies. Furthermore, β -carotene may not be the most effective carotenoid in cardiovascular protection. In this context, other carotenoids, such as lycopene and astaxanthin have been shown to exhibit marked cardiovascular protective effects in animal models (Section 3.1.1).

Notably, recent randomized controlled trials in selected subgroup populations suggest a possible beneficial effect of carotenoids in hyperlipidemia, hypertension, and vascular inflammation [74-77]. Large-scale randomized controlled trials in selected patient groups are needed to further investigate the potential role of carotenoid supplementation in the intervention of cardiovascular diseases.

3.2.2. Role in Human Cancer

As described above, animal studies provide substantial evidence supporting that carotenoids are protective against various types of cancer (Section 3.1.9). Results from these animal studies are in line with the findings from observational epidemiological studies demonstrating that increased consumption of diet rich in carotenoids is associated with decreased risk of many types of cancer [1, 2, 78]. In addition, blood levels of β -carotene have been inversely associated with risk of developing lung cancer. However, most large-scale randomized controlled trials have failed to show any beneficial effects of β -carotene supplementation in cancer intervention. Two interventional trials even reported an increased risk for lung cancer when high doses of β -carotene were supplemented to a population who were smokers or exposed to asbestos [2]. Several possible reasons for the unexpected results have been proposed, among which is that β -carotene at high doses may act as a pro-oxidant, thereby promoting carcinogenesis [2]. The failure of β -carotene supplementation for cancer intervention has prompted investigations of other dietary carotenoids, especially lycopene as potential protective agents against human cancer. Multiple observational epidemiological studies have found an inverse relationship between dietary intakes or circulation levels of lycopene and risk of prostate cancer [79, 80]. Several relatively small-scale randomized controlled trials also suggest a beneficial effect of lycopene supplementation in slowing the progression of prostate cancer. However, largescale well-designed interventional trials are warranted to ascertain the efficacy of lycopene in the intervention of prostate cancer as well as other types of human cancer [79, 80]. In addition, well-designed randomized controlled trials are also needed to assess the potential value of other dietary carotenoids in human cancer intervention, including both primary and secondary prevention.

3.2.3. Role in Other Human Diseases

Cardiovascular diseases and cancer have been the major focus of studies related to the health benefits of carotenoids. Carotenoids have also been studied with regard to their potential effects in many other human diseases. The sections below provide a brief summary of the recent findings from both observational epidemiological studies and interventional trials on the benefits of carotenoids in these diseases.

3.2.3.1. EYE DISEASES

The macula lutea, also known as yellow spot is part of the retina and the area of maximal visual acuity. The carotenoids lutein and zeaxanthin are the pigments responsible for the color of this tissue. As aforementioned, these two macular pigments are protective against age-related macular degeneration and cataract in animal models (Section 3.1.7). Consistently, both observational epidemiological studies and small interventional trials have suggested a beneficial role for lutein and zeaxanthin in patients with age-associated macular degeneration or cataract [81-83]. Supplementation of a mix of β -carotene, vitamins C and E, and zinc was also found to attenuate the development of advanced age-related macular degeneration in a randomized controlled trial [84].

3.2.3.2. Skin Diseases

Carotenoids accumulate in skin tissue following dietary intake. Multiple interventional trials have shown a protective effect of dietary supplementation of carotenoids, such as β -carotene, lycopene, lutein, and zeaxanthin in UV irradiation-induced skin injury [62, 85]. This is consistent with the findings in animal studies that carotenoids are protective against skin photodamage by UV light (Section 3.1.8). It has been suggested that dietary carotenoids may contribute to life-long protection against UV irradiation in humans. In addition to dietary intake, topical application of carotenoids has also been shown to afford protection against skin oxidative stress and aging in human subjects [62].

3.2.3.3. ADDITIONAL DISEASES AND CONDITIONS

Observational epidemiological studies demonstrate that plasma levels of lycopene are significantly lower in asthmatic patients than in healthy control subjects, suggesting a potential involvement of this carotenoid in asthma [86, 87]. An early randomized controlled interventional trial reported a beneficial effect of lycopene supplementation on exercise-induced

asthma and oxidative stress [88]. However, this finding was not replicated in a subsequent interventional trial [89].

As discussed above in Section 3.1, extensive animal studies have demonstrated astaxanthin as an effective protective agent in various disease conditions via its potent antioxidative and anti-inflammatory activities. In line with this, a recent randomized controlled trial reports that supplementation of astaxanthin decreases oxidative stress and inflammation in healthy human subjects [90]. Randomized controlled interventional trials also show a beneficial role for astaxanthin supplementation in improving the symptoms in patients with *Helicobacter pylori* infection [91], and in ameliorating oxidative stress, inflammation, and vascular dysfunction in renal transplant patients [92]. These observations indicate the feasibility of using astaxanthin in the intervention of human diseases involving oxidative stress and dysregulated inflammation.

There is substantial evidence supporting that inflammation and oxidative stress contribute to the development of human diabetes and metabolic syndrome. An inverse association between blood levels of carotenoids and risk of diabetes or metabolic syndrome has been demonstrated by observational epidemiological studies in various populations [93-97]. However, randomized controlled clinical trials on β -carotene supplementation so far have not produced positive results with regard to the intervention of diabetes and metabolic syndrome in human subjects [98-100].

4. Conclusion and Future Directions

Carotenoids are an important group of chemicals found in human diet that have antioxidative and anti-inflammatory activities. In addition, they also possess other novel biological functions, such as antiproliferation and modulation of cell signaling. Carotenoids are protective against various disease processes in animal models. Consistently, many observational epidemiological studies and multiple small-scale interventional trials in selected patient groups suggest a beneficial role for carotenoids in human diseases, including cardiovascular disorders and cancer. However, most large-scale randomized controlled trials predominantly on β -carotene supplementation in non-selected populations have failed to show a benefit in cardiovascular diseases and cancer. Large-scale randomized controlled trials with selected subgroup targeting are warranted to ascertain the value of carotenoid supplementation in the intervention of human diseases. Future efforts should also be focused on investigating the involvement of carotenoids other than β -carotene in human health and disease. It should be kept in mind that human diet contains many other bioactive compounds, such as polyphenols (Chapter 23), which may interact with carotenoids to influence the development of human diseases. In this context, future research including both basic and clinical research should be directed to study the combined effects of carotenoids with other diet-derived phytochemicals.

5. References

[1] Krinsky NI, Johnson EJ. Carotenoid actions and their relation to health and disease. *Mol. Aspects Med.* 2005;26:459-516.

- [2] Stahl W, Sies H. Bioactivity and protective effects of natural carotenoids. *Biochim. Biophys. Acta* 2005;1740:101-107.
- [3] Ben-Dor A, Steiner M, Gheber L, Danilenko M, Dubi N, Linnewiel K, et al. Carotenoids activate the antioxidant response element transcription system. Mol. Cancer Ther. 2005;4:177-186.
- [4] Kim GY, Kim JH, Ahn SC, Lee HJ, Moon DO, Lee CM, et al. Lycopene suppresses the lipopolysaccharide-induced phenotypic and functional maturation of murine dendritic cells through inhibition of mitogen-activated protein kinases and nuclear factor-kappaB. *Immunology* 2004;113:203-211.
- [5] Suzuki Y, Ohgami K, Shiratori K, Jin XH, Ilieva I, Koyama Y, *et al.* Suppressive effects of astaxanthin against rat endotoxin-induced uveitis by inhibiting the NF-kappaB signaling pathway. *Exp. Eye Res.* 2006;82:275-281.
- [6] Nagendraprabhu P, Sudhandiran G. Astaxanthin inhibits tumor invasion by decreasing extracellular matrix production and induces apoptosis in experimental rat colon carcinogenesis by modulating the expressions of ERK-2, NFkB and COX-2. *Invest. New Drugs* 2010 (in press).
- [7] Chew BP, Brown CM, Park JS, Mixter PF. Dietary lutein inhibits mouse mammary tumor growth by regulating angiogenesis and apoptosis. *Anticancer Res.* 2003;23:3333-3339.
- [8] Rao AV, Rao LG. Carotenoids and human health. *Pharmacol. Res.* 2007;55:207-216.
- [9] Dwyer JH, Navab M, Dwyer KM, Hassan K, Sun P, Shircore A, et al. Oxygenated carotenoid lutein and progression of early atherosclerosis: the Los Angeles atherosclerosis study. *Circulation* 2001;103:2922-2927.
- [10] Harari A, Harats D, Marko D, Cohen H, Barshack I, Kamari Y, et al. A 9-cis betacarotene-enriched diet inhibits atherogenesis and fatty liver formation in LDL receptor knockout mice. J. Nutr. 2008;138:1923-1930.
- [11] Hussein G, Nakamura M, Zhao Q, Iguchi T, Goto H, Sankawa U, et al. Antihypertensive and neuroprotective effects of astaxanthin in experimental animals. *Biol. Pharm. Bull.* 2005;28:47-52.
- [12] Bansal P, Gupta SK, Ojha SK, Nandave M, Mittal R, Kumari S, *et al.* Cardioprotective effect of lycopene in the experimental model of myocardial ischemia-reperfusion injury. *Mol. Cell. Biochem.* 2006;289:1-9.
- [13] Cai J, Yi FF, Bian ZY, Shen DF, Yang L, Yan L, et al. Crocetin protects against cardiac hypertrophy by blocking MEK-ERK1/2 signalling pathway. J. Cell Mol. Med. 2009;13:909-925.
- [14] Karimi G, Ramezani M, Abdi A. Protective effects of lycopene and tomato extract against doxorubicin-induced cardiotoxicity. *Phytother. Res.* 2005;19:912-914.
- [15] Pashkow FJ, Watumull DG, Campbell CL. Astaxanthin: a novel potential treatment for oxidative stress and inflammation in cardiovascular disease. Am. J. Cardiol. 2008; 101:58D-68D.
- [16] Fassett RG, Coombes JS. Astaxanthin, oxidative stress, inflammation and cardiovascular disease. *Future Cardiol*. 2009;5:333-342.
- [17] Kowluru RA, Menon B, Gierhart DL. Beneficial effect of zeaxanthin on retinal metabolic abnormalities in diabetic rats. *Invest. Ophthalmol. Vis. Sci.* 2008;49:1645-1651.
- [18] Arnal E, Miranda M, Almansa I, Muriach M, Barcia JM, Romero FJ, et al. Lutein

prevents cataract development and progression in diabetic rats. *Graefe's Arch. Clin. Exp. Ophthalmol.* 2009;247:115-120.

- [19] Maeda H, Tsukui T, Sashima T, Hosokawa M, Miyashita K. Seaweed carotenoid, fucoxanthin, as a multi-functional nutrient. *Asia Pac. J. Clin. Nutr.* 2008;17 Suppl 1:196-199.
- [20] Woo MN, Jeon SM, Shin YC, Lee MK, Kang MA, Choi MS. Anti-obese property of fucoxanthin is partly mediated by altering lipid-regulating enzymes and uncoupling proteins of visceral adipose tissue in mice. *Mol. Nutr. Food Res.* 2009;53:1603-1611.
- [21] Hussein G, Nakagawa T, Goto H, Shimada Y, Matsumoto K, Sankawa U, *et al.* Astaxanthin ameliorates features of metabolic syndrome in SHR/NDmcr-cp. *Life Sci.* 2007;80:522-529.
- [22] Ikeuchi M, Koyama T, Takahashi J, Yazawa K. Effects of astaxanthin in obese mice fed a high-fat diet. *Biosci. Biotechnol. Biochem.* 2007;71:893-899.
- [23] Hsiao G, Fong TH, Tzu NH, Lin KH, Chou DS, Sheu JR. A potent antioxidant, lycopene, affords neuroprotection against microglia activation and focal cerebral ischemia in rats. *In Vivo* 2004;18:351-356.
- [24] Suganuma H, Hirano T, Arimoto Y, Inakuma T. Effect of tomato intake on striatal monoamine level in a mouse model of experimental Parkinson's disease. J. Nutr. Sci. Vitaminol. (Tokyo) 2002;48:251-254.
- [25] di Matteo V, Pierucci M, Di Giovanni G, Dragani LK, Murzilli S, Poggi A, et al. Intake of tomato-enriched diet protects from 6-hydroxydopamine-induced degeneration of rat nigral dopaminergic neurons. J. Neural. Transm. Suppl. 2009:333-341.
- [26] Shen H, Kuo CC, Chou J, Delvolve A, Jackson SN, Post J, et al. Astaxanthin reduces ischemic brain injury in adult rats. FASEB J. 2009;23:1958-1968.
- [27] Ochiai T, Shimeno H, Mishima K, Iwasaki K, Fujiwara M, Tanaka H, et al. Protective effects of carotenoids from saffron on neuronal injury in vitro and in vivo. Biochim. Biophys. Acta 2007;1770:578-584.
- [28] Ahmad AS, Ansari MA, Ahmad M, Saleem S, Yousuf S, Hoda MN, et al. Neuroprotection by crocetin in a hemi-parkinsonian rat model. *Pharmacol. Biochem. Behav.* 2005;81:805-813.
- [29] Lee CM, Chang JH, Moon DO, Choi YH, Choi IW, Park YM, et al. Lycopene suppresses ovalbumin-induced airway inflammation in a murine model of asthma. Biochem. Biophys. Res. Commun. 2008;374:248-252.
- [30] Hazlewood LC, Wood LG, Hansbro PM, Foster PS. Dietary lycopene supplementation suppresses Th2 responses and lung eosinophilia in a mouse model of allergic asthma. J. Nutr. Biochem. 2011;22:95-100.
- [31] Saedisomeolia A, Wood LG, Garg ML, Gibson PG, Wark PA. Lycopene enrichment of cultured airway epithelial cells decreases the inflammation induced by rhinovirus infection and lipopolysaccharide. J. Nutr. Biochem. 2009;20:577-585.
- [32] Kasagi S, Seyama K, Mori H, Souma S, Sato T, Akiyoshi T, et al. Tomato juice prevents senescence-accelerated mouse P1 strain from developing emphysema induced by chronic exposure to tobacco smoke. Am. J. Physiol. Lung Cell Mol. Physiol. 2006; 290:L396-404.
- [33] Zhou C, Han W, Zhang P, Cai M, Wei D, Zhang C. Lycopene from tomatoes partially alleviates the bleomycin-induced experimental pulmonary fibrosis in rats. *Nutr. Res.* 2008;28:122-130.

- [34] van Helden YG, Heil SG, van Schooten FJ, Kramer E, Hessel S, Amengual J, *et al.* Knockout of the Bcmo1 gene results in an inflammatory response in female lung, which is suppressed by dietary beta-carotene. *Cell Mol. Life Sci.* 2010;67:2039-2056.
- [35] Curek GD, Cort A, Yucel G, Demir N, Ozturk S, Elpek GO, et al. Effect of astaxanthin on hepatocellular injury following ischemia/reperfusion. *Toxicology* 2010;267:147-153.
- [36] Bignotto L, Rocha J, Sepodes B, Eduardo-Figueira M, Pinto R, Chaud M, et al. Antiinflammatory effect of lycopene on carrageenan-induced paw oedema and hepatic ischaemia-reperfusion in the rat. Br. J. Nutr. 2009;102:126-133.
- [37] Chamberlain SM, Hall JD, Patel J, Lee JR, Marcus DM, Sridhar S, *et al.* Protective effects of the carotenoid zeaxanthin in experimental nonalcoholic steatohepatitis. *Dig. Dis. Sci.* 2009;54:1460-1464.
- [38] Wang Y, Ausman LM, Greenberg AS, Russell RM, Wang XD. Dietary lycopene and tomato extract supplementations inhibit nonalcoholic steatohepatitis-promoted hepatocarcinogenesis in rats. *Int. J. Cancer* 2010;126:1788-1796.
- [39] Seren S, Mutchnick M, Hutchinson D, Harmanci O, Bayraktar Y, Mutchnick S, et al. Potential role of lycopene in the treatment of hepatitis C and prevention of hepatocellular carcinoma. Nutr. Cancer 2008;60:729-735.
- [40] Baranowitz SA, Maderson PF. Acetaminophen toxicity is substantially reduced by beta-carotene in mice. *Int. J. Vitam. Nutr. Res.* 1995;65:175-180.
- [41] Kim JH, Kim YS, Song GG, Park JJ, Chang HI. Protective effect of astaxanthin on naproxen-induced gastric antral ulceration in rats. *Eur. J. Pharmacol.* 2005;514:53-59.
- [42] Bennedsen M, Wang X, Willen R, Wadstrom T, Andersen LP. Treatment of H. pylori infected mice with antioxidant astaxanthin reduces gastric inflammation, bacterial load and modulates cytokine release by splenocytes. *Immunol Lett.* 1999;70:185-189.
- [43] Reifen R, Nur T, Matas Z, Halpern Z. Lycopene supplementation attenuates the inflammatory status of colitis in a rat model. *Int. J. Vitam. Nutr. Res.* 2001;71:347-351.
- [44] Ogura W, Itagaki S, Kurokawa T, Noda T, Hirano T, Mizuno S, *et al.* Protective effect of lutein on ischemia-reperfusion injury in rat small intestine. *Biol. Pharm. Bull.* 2006;29:1764-1766.
- [45] Veeramachaneni S, Ausman LM, Choi SW, Russell RM, Wang XD. High dose lycopene supplementation increases hepatic cytochrome P4502E1 protein and inflammation in alcohol-fed rats. J. Nutr. 2008;138:1329-1335.
- [46] Joo YE, Karrasch T, Muhlbauer M, Allard B, Narula A, Herfarth HH, et al. Tomato lycopene extract prevents lipopolysaccharide-induced NF-kappaB signaling but worsens dextran sulfate sodium-induced colitis in NF-kappaBEGFP mice. PLoS One 2009;4:e4562.
- [47] Hosseini F, Naseri MK, Badavi M, Ghaffari MA, Shahbazian H, Rashidi I. Protective effect of beta carotene pretreatment on renal ischemia/reperfusion injury in rat. *Pak J. Biol. Sci.* 2009;12:1140-1145.
- [48] Hosseini F, Naseri MK, Badavi M, Ghaffari MA, Shahbazian H, Rashidi I. Effect of beta carotene on lipid peroxidation and antioxidant status following renal ischemia/ reperfusion injury in rat. Scand. J. Clin. Lab. Invest. 2010;70:259-263.
- [49] Naito Y, Uchiyama K, Aoi W, Hasegawa G, Nakamura N, Yoshida N, *et al.* Prevention of diabetic nephropathy by treatment with astaxanthin in diabetic db/db mice. *Biofactors* 2004;20:49-59.

- [50] Kim YJ, Kim YA, Yokozawa T. Protection against oxidative stress, inflammation, and apoptosis of high-glucose-exposed proximal tubular epithelial cells by astaxanthin. *J. Agric. Food Chem.* 2009;57:8793-8797.
- [51] Manabe E, Handa O, Naito Y, Mizushima K, Akagiri S, Adachi S, et al. Astaxanthin protects mesangial cells from hyperglycemia-induced oxidative signaling. J. Cell Biochem. 2008;103:1925-1937.
- [52] Atessahin A, Yilmaz S, Karahan I, Ceribasi AO, Karaoglu A. Effects of lycopene against cisplatin-induced nephrotoxicity and oxidative stress in rats. *Toxicology* 2005; 212:116-123.
- [53] Atessahin A, Ceribasi AO, Yilmaz S. Lycopene, a carotenoid, attenuates cyclosporineinduced renal dysfunction and oxidative stress in rats. *Basic Clin. Pharmacol. Toxicol.* 2007;100:372-376.
- [54] Karahan I, Atessahin A, Yilmaz S, Ceribasi AO, Sakin F. Protective effect of lycopene on gentamicin-induced oxidative stress and nephrotoxicity in rats. *Toxicology* 2005; 215:198-204.
- [55] Augusti PR, Conterato GM, Somacal S, Sobieski R, Spohr PR, Torres JV, *et al.* Effect of astaxanthin on kidney function impairment and oxidative stress induced by mercuric chloride in rats. *Food Chem. Toxicol.* 2008;46:212-219.
- [56] Carpentier S, Knaus M, Suh M. Associations between lutein, zeaxanthin, and agerelated macular degeneration: an overview. *Crit. Rev. Food. Sci. Nutr.* 2009;49:313-326.
- [57] Li SY, Fu ZJ, Ma H, Jang WC, So KF, Wong D, et al. Effect of lutein on retinal neurons and oxidative stress in a model of acute retinal ischemia/reperfusion. *Invest. Ophthalmol. Vis. Sci.* 2009;50:836-843.
- [58] Nakajima Y, Inokuchi Y, Shimazawa M, Otsubo K, Ishibashi T, Hara H. Astaxanthin, a dietary carotenoid, protects retinal cells against oxidative stress in-vitro and in mice in-vivo. J. Pharm. Pharmacol. 2008;60:1365-1374.
- [59] Gupta SK, Trivedi D, Srivastava S, Joshi S, Halder N, Verma SD. Lycopene attenuates oxidative stress induced experimental cataract development: an in vitro and in vivo study. *Nutrition* 2003;19:794-799.
- [60] Chan CM, Fang JY, Lin HH, Yang CY, Hung CF. Lycopene inhibits PDGF-BBinduced retinal pigment epithelial cell migration by suppression of PI3K/Akt and MAPK pathways. *Biochem. Biophys. Res. Commun.* 2009;388:172-176.
- [61] Nakajima Y, Shimazawa M, Otsubo K, Ishibashi T, Hara H. Zeaxanthin, a retinal carotenoid, protects retinal cells against oxidative stress. *Curr. Eye Res.* 2009;34:311-318.
- [62] Roberts RL, Green J, Lewis B. Lutein and zeaxanthin in eye and skin health. *Clin. Dermatol.* 2009;27:195-201.
- [63] Valacchi G, Pecorelli A, Mencarelli M, Maioli E, Davis PA. Beta-carotene prevents ozone-induced proinflammatory markers in murine skin. *Toxicol. Ind. Health.* 2009;25: 241-247.
- [64] Gonzalez S, Astner S, An W, Goukassian D, Pathak MA. Dietary lutein/zeaxanthin decreases ultraviolet B-induced epidermal hyperproliferation and acute inflammation in hairless mice. J. Invest. Dermatol. 2003;121:399-405.
- [65] Astner S, Wu A, Chen J, Philips N, Rius-Diaz F, Parrado C, *et al.* Dietary lutein/ zeaxanthin partially reduces photoaging and photocarcinogenesis in chronically UVB-

irradiated Skh-1 hairless mice. Skin. Pharmacol. Physiol. 2007;20:283-291.

- [66] Wolf G. The effect of beta-carotene on lung and skin carcinogenesis. *Carcinogenesis* 2002;23:1263-1265.
- [67] Black HS. Pro-oxidant and anti-oxidant mechanism(s) of BHT and beta-carotene in photocarcinogenesis. *Front Biosci.* 2002;7:d1044-1055.
- [68] Erdman JW, Jr., Ford NA, Lindshield BL. Are the health attributes of lycopene related to its antioxidant function? *Arch. Biochem. Biophys.* 2009;483:229-235.
- [69] Tapiero H, Townsend DM, Tew KD. The role of carotenoids in the prevention of human pathologies. *Biomed. Pharmacother*. 2004;58:100-110.
- [70] Nishino H, Murakoshi M, Tokuda H, Satomi Y. Cancer prevention by carotenoids. *Arch. Biochem. Biophys.* 2009;483:165-168.
- [71] Bertram JS, Vine AL. Cancer prevention by retinoids and carotenoids: independent action on a common target. *Biochim. Biophys. Acta* 2005;1740:170-178.
- [72] Sesso HD. Carotenoids and cardiovascular disease: what research gaps remain? *Curr. Opin. Lipidol.* 2006;17:11-16.
- [73] Voutilainen S, Nurmi T, Mursu J, Rissanen TH. Carotenoids and cardiovascular health. Am. J. Clin. Nutr. 2006;83:1265-1271.
- [74] Yoshida H, Yanai H, Ito K, Tomono Y, Koikeda T, Tsukahara H, et al. Administration of natural astaxanthin increases serum HDL-cholesterol and adiponectin in subjects with mild hyperlipidemia. Atherosclerosis 2010;209:520-523.
- [75] Riccioni G, Mancini B, Di Ilio E, Bucciarelli T, D'Orazio N. Protective effect of lycopene in cardiovascular disease. *Eur. Rev. Med. Pharmacol. Sci.* 2008;12:183-190.
- [76] Engelhard YN, Gazer B, Paran E. Natural antioxidants from tomato extract reduce blood pressure in patients with grade-1 hypertension: a double-blind, placebocontrolled pilot study. *Am. Heart J.* 2006;151:100.
- [77] Paran E, Novack V, Engelhard YN, Hazan-Halevy I. The effects of natural antioxidants from tomato extract in treated but uncontrolled hypertensive patients. *Cardiovasc. Drugs Ther.* 2009;23:145-151.
- [78] Musa-Veloso K, Card JW, Wong AW, Cooper DA. Influence of observational study design on the interpretation of cancer risk reduction by carotenoids. *Nutr. Rev.* 2009; 67:527-545.
- [79] Dahan K, Fennal M, Kumar NB. Lycopene in the prevention of prostate cancer. J. Soc. Integr. Oncol. 2008;6:29-36.
- [80] Haseen F, Cantwell MM, O'Sullivan JM, Murray LJ. Is there a benefit from lycopene supplementation in men with prostate cancer? A systematic review. *Prostate Cancer Prostatic Dis.* 2009;12:325-332.
- [81] Rhone M, Basu A. Phytochemicals and age-related eye diseases. *Nutr. Rev.* 2008;66: 465-472.
- [82] Christen WG, Liu S, Glynn RJ, Gaziano JM, Buring JE. Dietary carotenoids, vitamins C and E, and risk of cataract in women: a prospective study. *Arch. Ophthalmol.* 2008; 126:102-109.
- [83] Johnson EJ. Age-related macular degeneration and antioxidant vitamins: recent findings. Curr. Opin. Clin. Nutr. Metab. Care 2010;13:28-33.
- [84] Age-Related Eye Disease Study Research Group. A randomized, placebo-controlled, clinical trial of high-dose supplementation with vitamins C and E, beta carotene, and zinc for age-related macular degeneration and vision loss: AREDS report no. 8. *Arch.*

Ophthalmol. 2001;119:1417-1436.

- [85] Stahl W, Sies H. Carotenoids and flavonoids contribute to nutritional protection against skin damage from sunlight. *Mol. Biotechnol.* 2007;37:26-30.
- [86] Riccioni G, Bucciarelli T, Mancini B, Di Ilio C, D'Orazio N. Plasma lycopene concentration is low in subjects with stable asthma. *Allergy* 2006;61:1371-1372.
- [87] Riccioni G, Bucciarelli T, Mancini B, Di Ilio C, Della Vecchia R, D'Orazio N. Plasma lycopene and antioxidant vitamins in asthma: the PLAVA study. J. Asthma 2007; 44:429-432.
- [88] Neuman I, Nahum H, Ben-Amotz A. Reduction of exercise-induced asthma oxidative stress by lycopene, a natural antioxidant. *Allergy* 2000;55:1184-1189.
- [89] Falk B, Gorev R, Zigel L, Ben-Amotz A, Neuman I. Effect of lycopene supplementation on lung function after exercise in young athletes who complain of exerciseinduced bronchoconstriction symptoms. *Ann. Allergy Asthma Immunol.* 2005; 94:480-485.
- [90] Park JS, Chyun JH, Kim YK, Line LL, Chew BP. Astaxanthin decreased oxidative stress and inflammation and enhanced immune response in humans. *Nutr. Metab.* (Lond) 2010;7:18.
- [91] Kupcinskas L, Lafolie P, Lignell A, Kiudelis G, Jonaitis L, Adamonis K, et al. Efficacy of the natural antioxidant astaxanthin in the treatment of functional dyspepsia in patients with or without Helicobacter pylori infection: a prospective, randomized, double blind, and placebo-controlled study. *Phytomedicine* 2008;15:391-399.
- [92] Fassett RG, Healy H, Driver R, Robertson IK, Geraghty DP, Sharman JE, *et al.* Astaxanthin vs placebo on arterial stiffness, oxidative stress and inflammation in renal transplant patients (Xanthin): a randomised controlled trial. *BMC Nephrol.* 2008;9:17.
- [93] Coyne T, Ibiebele TI, Baade PD, Dobson A, McClintock C, Dunn S, *et al.* Diabetes mellitus and serum carotenoids: findings of a population-based study in Queensland, Australia. *Am. J. Clin. Nutr.* 2005;82:685-693.
- [94] Hozawa A, Jacobs DR, Jr., Steffes MW, Gross MD, Steffen LM, Lee DH. Associations of serum carotenoid concentrations with the development of diabetes and with insulin concentration: interaction with smoking: the Coronary Artery Risk Development in Young Adults (CARDIA) Study. Am. J. Epidemiol. 2006;163:929-937.
- [95] Arnlov J, Zethelius B, Riserus U, Basu S, Berne C, Vessby B, *et al.* Serum and dietary beta-carotene and alpha-tocopherol and incidence of type 2 diabetes mellitus in a community-based study of Swedish men: report from the Uppsala Longitudinal Study of Adult Men (ULSAM) study. *Diabetologia* 2009;52:97-105.
- [96] Coyne T, Ibiebele TI, Baade PD, McClintock CS, Shaw JE. Metabolic syndrome and serum carotenoids: findings of a cross-sectional study in Queensland, Australia. Br. J. Nutr. 2009;102:1668-1677.
- [97] Sugiura M, Nakamura M, Ogawa K, Ikoma Y, Matsumoto H, Ando F, et al. Associations of serum carotenoid concentrations with the metabolic syndrome: interaction with smoking. Br. J. Nutr. 2008;100:1297-1306.
- [98] Kataja-Tuomola M, Sundell JR, Mannisto S, Virtanen MJ, Kontto J, Albanes D, *et al.* Effect of alpha-tocopherol and beta-carotene supplementation on the incidence of type 2 diabetes. *Diabetologia* 2008;51:47-53.
- [99] Song Y, Cook NR, Albert CM, Van Denburgh M, Manson JE. Effects of vitamins C and E and beta-carotene on the risk of type 2 diabetes in women at high risk of

cardiovascular disease: a randomized controlled trial. Am. J. Clin. Nutr. 2009;90:429-437.

[100] Kataja-Tuomola MK, Kontto JP, Mannisto S, Albanes D, Virtamo JR. Effect of alphatocopherol and beta-carotene supplementation on macrovascular complications and total mortality from diabetes: results of the ATBC Study. Ann. Med. 2010;42:178-186.

Chapter 23

Phenolic Compounds

Abstract

Phenolic compounds are an important bioactive component of human diet rich in fruits and vegetables. Dietary phenolic compounds are usually classified into five groups, namely, flavonoids (e.g., genistein), stilbenes (e.g., resveratrol), phenolic acids (e.g., caffeic acid), lignans (e.g., secoisolariciresinol), and others (e.g., curcumin). Dietary phenolic compounds possess diverse important biochemical properties and functions, including antioxidative and anti-inflammatory activities as well as other novel effects on cellular processes. These important biochemical properties and functions are translated into their critical role in protecting against disease pathophysiology in many animal models of human diseases. Both observational epidemiological studies and interventional clinical trials also suggest potential benefits of dietary phenolic compounds in various human disease processes.

1. Overview

1.1. Definition and Dietary Sources

Phenolic compounds, also known as phenols or phenolics, are a class of chemicals consisting of one or more hydroxyl groups (-OH) bonded directly to one or more six-membered aromatic rings. The simplest of the class is phenol (C_6H_5OH), which is a monophenol. Monophenols contain one such aromatic hydroxyl group, whereas compounds containing two such hydroxyl groups are called biphenols. Polyphenols are those that contain more than two such hydroxyl groups. However, the polyphenol literature often includes discussion of biphenols and monophenols because they often possess similar biological activities as do polyphenols. Notably, biphenols are also frequently considered as polyphenols in the literature. But, strictly this is not correct.

Phenolic compounds, particularly polyphenols are widely distributed in plants. Fruits and vegetables as well as wine and tea constitute the major dietary sources of phenolic compounds, with an average consumption of approximately one gram per day for a normal human adult [1, 2].
1.2. Classification

Dietary phenolic compounds are generally classified into five major groups, which are listed below.

- Flavonoids (e.g., genistein)
- Stilbenes (e.g., resveratrol)
- Phenolic acids (e.g., caffeic acid)
- Lignans (e.g., secoisolariciresinol)
- Others (e.g., curcumin)

Flavonoids, the most commonly studied dietary phenols, share a common structure consisting of two aromatic rings that are bound together by three carbon atoms that usually form an oxygenated heterocycle (Figure 23-1). Flavonoids are further classified into the following six subclasses.

- Flavonols (e.g., quercetin)
- Flavones (e.g., luteolin)
- Isoflavones (e.g., genistein)
- Flavanones (e.g., taxifolin)
- Anthocyanidins (e.g., delphinidin)
- Flavanols (e.g., epigallocatechin-3-gallate)



Figure 23-1. The basic chemical structure of flavonoids. This shared structure of flavonoids consists of two aromatic rings A and B that are bound together by the third ring C. See text (Section 1.2) for description of flavonoids.

The structures of some commonly studied phenolic compounds in biology and medicine and their major dietary sources are shown in Figure 23-2.

1.3. Pharmacokinetic Properties

Following dietary intake, the levels of phenolic compounds in tissues and blood are usually in the submicromolar range. Administration of pharmacological doses of pure phenolic compounds may produce low micromolar plasma concentrations in animals and human subjects. Due to the hydroxyl groups, phenolic compounds undergo rapid conjugation reactions catalyzed by phase 2 biotransformation enzymes in vivo. It has been shown that gastrointestinal tissues may be exposed to high levels of phenolic compounds following their oral supplementation or dietary intake [3].



Figure 23-2. Structures of some commonly studied phenolic compounds in biology and medicine and their major dietary sources. Quercetin is mainly found in citrus fruits and apples. Soy beans and tea are major sources of genistein and epigallocatechin-3-gallate, respectively. Resveratrol is present in grape seeds and skin. Caffeic acid is a major phenolic compound in coffee. Curcumin is the principal curcuminoid of the popular Indian spice turmeric.

2. Biochemical Properties and Functions

Dietary phenolic compounds have been shown to possess various biochemical properties and functions. These are summarized into four general categories, as listed below.

- Antioxidant and pro-oxidant activities
- Anti-inflammatory activities
- Effects on nitric oxide bioavailability and endothelial function
- Other novel functions

2.1. Antioxidant and Pro-oxidant Activities

The antioxidant activities of phenolic compounds may result from the following mechanisms: (1) scavenging of reactive oxygen and nitrogen species (ROS/RNS) and other reactive species, (2) inhibition of ROS formation from cellular sources, and (3) induction of endogenous cellular antioxidant enzymes. Notably, these mechanisms may operate coordinately to account for the efficient antioxidative activities observed with certain phenolic compounds. In addition to the antioxidant functions, phenolic compounds may also exert pro-oxidant properties under certain conditions.

2.1.1. Scavenging of ROS/RNS and Other Reactive Species

In many in vitro systems, plant phenolic compounds have been shown to possess antioxidant activities. At high micromolar concentrations, they are able to scavenge various ROS/RNS as well as other reactive species, such as reactive carbonyls [4]. In addition, many phenolic compounds chelate redox active metal ions, thus preventing these metal ions from mediating the formation of more reactive forms of ROS, such as hydroxyl radical.

Due to the extreme low concentrations reached in blood from regular dietary intake, phenolic compounds are generally considered to exert insignificant ROS/RNS-scavenging effects in vivo. However, supplementation of pharmacological doses of phenolic compounds may result in markedly elevated levels of these compounds in vivo that are high enough to directly scavenge ROS/RNS and other reactive species.

2.1.2. Inhibition of ROS Formation

In cultured cells, dietary phenolic compounds at micromolar concentrations have been shown to decrease superoxide formation from NAD(P)H oxidase, an important cellular source of ROS involved in disease pathophysiology. The decreased ROS formation may result from their suppression of NAD(P)H oxidase expression, disruption of signaling pathways leading to NAD(P)H oxidase activation, or direct inhibition of the enzymatic activity of the oxidase. Both parent phenolic compounds and their metabolites may be directly involved in the inhibition of ROS formation from NAD(P)H oxidase. Indeed, the metabolites of flavonoid compounds have recently been found to potently inhibit NAD(P)H oxidase in cells, leading to decreased formation of superoxide [5]. In addition to NAD(P)H oxidase, other ROSproducing enzymes, such as xanthine oxidoreductase may also be inhibited by dietary phenolic compounds.

2.1.3. Induction of Endogenous Antioxidant Enzymes

At micromolar concentrations, many phenolic compounds have been found to upregulate endogenous cellular antioxidant enzymes in cell cultures. Administration of pharmacological doses of certain phenolic compounds has also been reported to induce tissue antioxidant enzymes in experimental animals. The transcription factor Nrf2 appears to be the key regulator involved in the upregulation of antioxidant enzymes by phenolic compounds both in vitro and in vivo [6, 7].

2.1.4. Potential Pro-oxidant Activities

Under certain in vitro conditions, many phenolic compounds at high concentrations exhi-

bit pro-oxidant activities, particularly in the presence of redox-active metals, including copper and iron ions. Due to the presence of redox metal ions in culture media, phenolic compounds undergo auto-oxidation to form ROS in cell cultures [8]. There is also evidence that certain phenolic compounds, such as epigallocatechin-3-gallate at pharmacological doses may increase ROS levels in vivo [9]. Such pro-oxidant effects have been proposed as an underlying mechanism of phenolic compound-mediated anticancer activities in animal models (see Section 3.1.8 below).

2.2. Anti-inflammatory Activities

Phenolic compounds are effective inhibitors of inflammatory responses in both in vitro and in vivo systems [10, 11]. They may affect various cascades of inflammation, including (1) inhibition of NF- κ B, a critical transcription factor regulating expression of proinflammatory cytokines, (2) inhibition of cyclooxygenases, lipoxygenases, and phospholipase A2, enzymes involved in the formation of inflammatory mediators, such as prostaglandins and leukotrienes, and (3) inhibition of expression of adhesion molecules and other proinflammatory factors. In addition, the antioxidant activities of phenolic compounds may contribute to their anti-inflammatory functions as oxidative stress and inflammation are two intimately related processes.

2.3. Effects on Nitric Oxide Bioavailability and Endothelial Function

As decreased nitric oxide bioavailability and compromised endothelial function are crucial events underlying cardiovascular diseases, the beneficial role of phenolic compounds in these events has recently received attention [12, 13]. Phenolic compounds, such as flavanols and resveratrol at low micromolar concentrations increase nitric oxide levels in cultured endothelial cells. These compounds and/or their metabolites inhibit NAD(P)H oxidase, thus reducing superoxide formation from this enzyme, a major source of vascular superoxide. As discussed in Section 2.3.1 of Chapter 1, superoxide reacts with nitric oxide at a near diffusion-limited rate to form peroxynitrite. Phenolic compounds also increase the expression of endothelial nitric oxide synthase (eNOS), leading to increased formation of nitric oxide. Through decreasing superoxide formation and induction of eNOS, phenolic compounds may significantly enhance endothelial nitric oxide bioavailability.

2.4. Other Novel Functions

At micromolar concentrations or pharmacological doses, phenolic compounds exhibit a variety of other biological activities. These effects include: (1) modulation of many cellular signaling molecules, leading to decreased cell proliferation, migration, and angiogenesis, (2) induction of apoptosis in tumor cells, (3) modulation of estrogen or androgen receptors, and (4) activation of sirtuins.

Activation of sirtuins by phenolic compounds, especially resveratrol has recently received great attention. Sirtuins are a highly conserved family of NAD⁺-dependent enzymes that regulate lifespan in lower organisms. To date, seven mammalian homologues (Sirt1-7) have been identified, and they are linked to many activities, including cellular stress resistance, genomic stability, tumorigenesis, and energy metabolism [14]. Activation of Sirt1 by pharmacological doses of resveratrol leads to protection against metabolic disease as well as improvement of survival in mice on a high-calorie diet [15] (see Section 3.1.2 below for more discussion). A number of other plant phenolic compounds, including butein, quercetin, fise-tin, isoliquiritigenin and piceatannol have been shown to activate Sirt1 [16, 17].

In summary, as illustrated in Figure 23-3, phenolic compounds affect numerous cellular processes and possess diverse biochemical properties and functions. Increasing evidence suggests that these diverse biochemical properties and functions contribute to the beneficial role of dietary phenolic compounds in health and disease.



Figure 23-3. Dietary phenolic compounds as multitasking molecules. As illustrated, dietary phenolic compounds possess antioxidative and anti-inflammatory activities. In addition, they also participate in modulating cell signal transduction and growth regulation. Moreover, phenolic compounds are able to cause activation of Sirt1 as well as modulate the receptors of sex hormones. These diverse mechanisms contribute to the disease-protective effects of dietary phenolic compounds.

3. Role in Health and Disease

3.1. Animal Studies

The beneficial effects of phenolic compounds in disease pathophysiology have been extensively investigated in a wide variety of animal models over the last two decades. A large body of experimental evidence supports an important protective role for numerous phenolic compounds in animal models of human diseases involving nearly all organs and systems. It is impossible to include this vast amount of information in a single chapter. Hence, the sections below are intended to summarize the key findings from these studies.

3.1.1. Role in Cardiovascular Diseases

Cardiovascular diseases are among the best studied disorders with regard to the protection by dietary phenolic compounds. The most notable polyphenol with established cardiovascular benefits in animal models is probably resveratrol [18]. The cardiovascular protective effects of numerous other phenolic compounds have also been demonstrated in animal models [13, 19-21]. In addition to the individual pure phenolic compounds, fruit and plant extracts rich in polyphenols have been reported to be also cardiovascular protective. The disease entities in which dietary phenolic compounds exhibit a beneficial role include hypertension, atherosclerosis, myocardial ischemia-reperfusion injury, cardiac hypertrophy and heart failure, cardiac arrhythmia, and drug/xenobiotic-induced cardiotoxicity.

The cardiovascular protective effects of phenolic compounds are usually manifested as prevention of disease development and/or retardation of disease progression. Certain phenolic compounds, such as curcumin have been demonstrated to even reverse cardiac hypertrophy in a murine model induced by aortic banding and phenylephrine infusion. Although overwhelming evidence supports beneficial activities of phenolic compounds in cardiovascular pathophysiology, there are also studies suggesting deleterious effects of certain phenolic compounds under specific experimental conditions. For example, it was recently reported that late chronic catechin treatment could worsen endothelial dysfunction in low-density lipoprotein receptor-deficient mice with established atherosclerosis although the same catechin treatment reversed the age-related vascular dysfunctions in wild-type mice [22]. The significance of this observation to human health remains unclear.

The biochemical properties and functions of phenolic compounds described in Section 2 are apparently the molecular and cellular basis of cardiovascular protection by these compounds. Specifically, the following major mechanisms may particularly account for the beneficial effects of phenolic compounds in cardiovascular pathophysiology.

- Inhibition of oxidative stress and inflammation in cardiovascular tissues
- Augmentation of nitric oxide bioavailability in vasculature
- Antiplatelet activities
- Stimulation of Sirt1 signaling to augment cardiovascular function
- Stimulation of high-density lipoprotein-mediated cholesterol efflux from macrophages, preventing foam cell formation

Obviously, the antioxidative and anti-inflammatory activities are important in phenolic compound-mediated cardiovascular protection as oxidative stress and inflammation are key events of cardiovascular pathophysiology. Augmentation of nitric oxide bioavailability will improve vasodilation and endothelial function. In addition, vascular nitric oxide is able to suppress inflammation as well as thrombogenesis. Phenolic compounds are shown to also have direct antiplatelet activities via interference with signaling involved in platelet activation and aggregation. The antiplatelet activities may contribute to the in vivo inhibition of thrombogenesis by phenolic compounds [20]. As Sirt1 plays an important role in energy metabolism and cardiovascular function, stimulation of this signaling pathway by phenolic compounds, especially resveratrol has been shown to improve cardiac function of experimental animals with cardiomyopathy [23]. Recently, phenolic compounds from coffee are demonstrated to stimulate high-density lipoprotein-mediated cholesterol efflux from macrophages in

vitro and in vivo, providing a novel mechanism for the antiatherosclerotic effects of dietary phenolic compounds [24, 25].

In addition to the above listed mechanisms, phenolic compounds may target other cellular pathways or enzymes to exert cardiovascular protection. For example, inhibition of matrix metalloproteinases by phenolic compounds may attenuate cardiovascular remodeling, a critical event involved in hypertension, atherosclerosis, cardiac hypertrophy, and heart failure. Recently, the dietary polyphenol curcumin is found to protect against cardiac hypertrophy and heart failure via its inhibition of p300 histone acetyltransferase activity in cardiac tissue [26, 27]. Histone acetylation plays a critical role in the pathophysiology of cardiac hypertrophy and heart failure.

3.1.2. Role in Diabetes and Metabolic Syndrome

3.1.2.1. DIABETES

The antidiabetic effects of phenolic compounds are well-recognized in animal models of both type 1 and type 2 diabetes. Phenolic compounds have been found to prevent the development of diabetes as well as retard the progression of diabetic complications. Multiple mechanisms have been proposed to explain the protective effects of phenolic compounds in diabetes and diabetic complications in addition to their antioxidative and anti-inflammatory activities. These additional mechanisms, as listed below, involve multiple targeting sites, including gut, pancreas, liver, and peripheral tissues (e.g., muscle, adipose) [28, 29].

- Inhibition of carbohydrate digestion and glucose absorption
- Protection against pancreatic β-cell dysfunction, and stimulation of insulin secretion from pancreatic β-cells
- Suppression of glucose release from liver storage
- Improvement of glucose uptake in the insulin-sensitive tissues, such as muscle and adipose

A variety of phenolic compounds are able to inhibit the activity of α -amylase and α glucosidase, enzymes involved in the digestion of dietary carbohydrates to glucose in the gut. Intestinal absorption of glucose is mediated by active transport via the sodium-dependent glucose transporter SGLT1 and by facilitated sodium-independent transport via the glucose transporter GLUT2. Numerous phenolic compounds have been demonstrated to inhibit these glucose transporting systems, thereby diminishing the absorption of glucose in the gut. Both inhibition of dietary carbohydrate digestion and suppression of glucose absorption in the gut may greatly contribute to the amelioration of postprandial hyperglycemia in diabetes.

Pancreatic β -cells are an important targeting site for the antidiabetic activities of many phenolic compounds, especially the soy-derived isoflavones. Phenolic compounds have been found to protect pancreatic β -cells from glucotoxicity as well as oxidative and inflammatory stress. They also stimulate insulin release from pancreatic β -cells possibly via modulation of Ca²⁺ signaling and AMP-activated protein kinase pathway.

Liver plays a major role in the regulation of blood glucose levels in tight cooperation with peripheral tissues. A large number of dietary phenolic compounds, including tea catechins and soy isoflavones have been demonstrated to reduce hepatic glucose output by suppressing gluconeogenetic enzyme expression and increasing the activity of glucokinase to improve glycogenesis and glucose utilization. Activation of AMP-activated protein kinase pathway has been implicated in the above effects of phenolic compounds on hepatic glucose regulation [30].

There is also evidence suggesting that the antidiabetic activities of phenolic compounds may result from their stimulation of glucose uptake in insulin-sensitive peripheral tissues, such as muscle and adipose tissue. Phenolic compounds have been found to both increase the sensitivity of insulin receptor and augment the activity of glucose transporters in cultured skeletal cells and adipocytes.

3.1.2.2. METABOLIC SYNDROME

Substantial evidence supports a beneficial role for dietary phenolic compounds in metabolic syndrome in animal models. In particular, various phenolic compounds, including tea catechins, soy isoflavones, and resveratrol have been demonstrated to exhibit antiobesity effects in animal models of high fat diet-induced obesity [31-33]. The benefits of phenolic compounds in obesity and other manifestations of metabolic syndrome may involve multiple mechanisms related to lipid metabolism as well as the direct effects on adipocytes. In addition, the effects of phenolic compounds on carbohydrate metabolism as described above (Section 3.1.2.1) may also contribute to their antiobesity activity. Obviously, the antioxidative and anti-inflammatory effects of phenolic compounds are beneficial in metabolic syndrome as oxidative stress and inflammation are key events involved in obesity, insulin resistance, and other abnormalities associated with metabolic syndrome.

Several signaling pathways have been suggested to be the molecular targets of dietary phenolic compounds in metabolic syndrome. These include AMP-activated protein kinase and Sirt1 signaling. Activation of AMP-activated protein kinase has been demonstrated as an important protective mechanism in diabetes and metabolic syndrome [30]. Recently, activation of Sirt1 signaling by phenolic compounds, especially resveratrol has been found to improve the efficiency of energy metabolism and insulin resistance, as well as suppress obesity and prevent early mortality associated with obesity in animal models [34]. There is also evidence suggesting a close interaction between AMP-activated protein kinase and Sirt1signaling in ameliorating obesity and other metabolic abnormalities caused by high-fat diet in experimental animals [35].

3.1.3. Role in Neurological Diseases

Extensive studies in animal models have demonstrated a protective role for a variety of different dietary phenolic compounds in many neurological disorders, including stroke, Parkinson's disease, and Alzheimer's disease. [36-40]. The neuroprotection has been observed with both pure phenolic compounds and fruit or vegetable extracts rich in polyphenols. Furthermore, dietary supplementation of phenolic compounds attenuates aging-related motor and cognitive deficits in experimental animals.

The potential mechanisms underlying the neuroprotective effects of phenolic compounds may include their antioxidative and anti-inflammatory activities. An important aspect related to the neuroprotection of phenolic compounds is their ability to penetrate the blood-brain barrier and reach the brain tissue. In addition, certain phenolic compounds have been found to induce antioxidant enzymes in brain tissue [41]. As redox-active metal ions (e.g., iron, copper) are particularly involved in neurodegenerative disorders, such as Parkinson's disease, chelation of these redox-active metal ions by phenolic compounds may also contribute to their neuroprotection [42].

Sirt1 signaling has been demonstrated to promote survival and stress tolerance in neurons of the central nervous system. Recently, a role for the activation of Sirt1 by dietary phenolic compounds in neuroprotection has been suggested in the literature [43]. For example, resveratrol pretreatment is found to increase the activity of Sirt1 in rat brain tissue, and protect against cerebral ischemic damage. The protective effect of resveratrol on cerebral ischemic injury is abolished by the Sirt1-specific inhibitor, sirtinol, suggesting that the neuroprotection occurs via activation of Sirt1 signaling pathway. Moreover, activation of Sirt1 by resveratrol leads to inhibition of mitochondrial uncoupling protein 2, which may also contribute to resveratrol-mediated neuroprotection [44]. Inhibition of uncoupling protein 2 improves mitochondrial ATP production.

3.1.4. Role in Pulmonary Diseases

A number of dietary phenolic compounds, including resveratrol, curcumin, and tea catechins have been shown to be protective against various pulmonary diseases in animal models, such as asthma, chronic obstructive pulmonary disease (COPD), and drug-induced pulmonary fibrosis.

Ovalbumin-induced murine allergic airway inflammation is a commonly used experimental model of asthma. In this murine model, a variety of phenolic compounds are reported to attenuate airway inflammation and oxidative stress. There is also evidence that phenolic compounds, such as quercetin protect experimental animals from ovalbumin-induced airway inflammation via regulating Th1/Th2 balance [45].

Treatment with phenolic compounds ameliorates cigarette smoke-induced COPD-like airway inflammation, oxidative stress, and lung tissue injury in experimental animals. Cigarette smoke is a major cause of COPD in humans. As asthma and COPD are characterized by inflammation and oxidative stress, the protective effects of phenolic compounds in these pulmonary diseases are thus most likely attributed to their antioxidative and anti-inflammatory functions. Phenolic compounds, such as resveratrol, curcumin, and tea polyphenols induce antioxidant enzymes in lung tissue, which may also contribute to the pulmonary protection [46, 47]. More recently, Sirt1 signaling pathway has been implicated as a protective mechanism in cigarette smoke-induced lung injury [48]. Thus, activation of Sirt1 by phenolic compounds, such as resveratrol may represent a novel mechanism for protection against the pathogenesis of COPD.

In addition to asthma and COPD, viral infection-elicited lung inflammation is attenuated by dietary administration of curcumin in mice. Curcumin, resveratrol, and epigallocatechin-3gallate also protect experimental animals from bleomycin-induced pulmonary fibrosis, inflammation, and oxidative stress. It is further found that the beneficial effects of epigallocatechin-3-gallate in bleomycin-induced pulmonary injury are mediated, at least partially, by augmentation of antioxidant enzymes via Nrf2 signaling [47].

3.1.5. Role in Hepatic and Gastrointestinal Diseases

3.1.5.1. HEPATIC DISEASES

Many phenolic compounds are protective in diverse hepatic diseases, including ischemiareperfusion injury, nonalcoholic fatty liver disease, alcoholic fatty liver disease, endotoxinelicited liver injury, drug/xenobiotic-induced hepatotoxicity, and viral hepatitis. While phenolic compound-mediated hepatoprotective effects are well-documented, the exact underlying mechanisms are often difficult to be delineated. This is due to the fact that phenolic compounds possess multiple biological activities, and often time these various activities all contribute to hepatoprotection. Nevertheless, the hepatoprotective effects of phenolic compounds are usually associated with attenuation of inflammation and oxidative stress in hepatic tissue. This suggests that the antioxidative and anti-inflammatory activities of phenolic compounds may primarily contribute to hepatoprotection. In addition, other novel pathways may also be involved. For example, the protective effects of resveratrol on both alcoholic and nonalcoholic fatty liver diseases in mice are shown to be mediated partially by activation of Sirt1 and AMP-activated protein kinase signaling pathways [49, 50].

3.1.5.2. GASTROINTESTINAL DISEASES

Although blood or tissue levels of phenolic compounds are usually in the range of low or submicromolar concentrations following dietary supplementation, the concentrations of these compounds in the gastrointestinal tract have been found to be much higher [3]. Thus, phenolic compounds may exert significant beneficial effects in gastrointestinal disorders that involve inflammatory and oxidative stress [51, 52]. Indeed, dietary supplementation of pure phenolic compounds, including resveratrol, curcumin, tea catechins, and soy isoflavones has been found to protect against inflammatory bowel disease, *Helicobacter pylori* infection, and intestinal ischemia-reperfusion injury in experimental animals. Protection against intestinal inflammatory and oxidative disorders has also been observed with dietary supplementation of plant extracts rich in polyphenols.

3.1.6. Role in Renal Diseases

Oxidative stress and inflammation play an important role in various kidney diseases, such as ischemia-reperfusion injury, diabetic nephropathy, and drug/xenobiotic-induced nephrotoxicity. A variety of phenolic compounds, including tea polyphenols, soy isoflavones, resveratrol, and curcumin have been shown to exert beneficial effects on the above pathophysiological processes. Dietary supplementation of plant or fruit extracts rich in polyphenols also results in protection against renal diseases, including diabetic nephropathy and drug-induced nephrotoxicity. In addition to the antioxidative and anti-inflammatory activities, other novel actions of phenolic compounds may participate in the renal protection. For instance, resveratrol and flavonoids (e.g., quercetin, daidzein) are shown to promote mitochondrial biogenesis in renal tubular cells via Sirt1-dependent signaling [53]. The improved energy metabolism via activating Sirt1 may contribute to phenolic compound-mediated renal protection against ischemia-reperfusion injury. More recently, Sirt1 activation is found to protect mouse renal medulla from oxidative injury [54]. How exactly Sirt1 activation leads to protection against oxidative stress remains to be elucidated. Possible mechanisms include upregulation of cellular antioxidants and improvement of mitochondrial function [50]. Targeting Sirt1 signaling may thus represent a novel approach to protecting against renal diseases involving oxidative stress and dysregulated energy metabolism.

3.1.7. Role in Skin Diseases

A wealth of data from animal studies has revealed beneficial effects of various phenolic compounds (e.g., green tea polyphenols) in skin disorders, especially ultraviolet (UV) irradia-

tion-induced skin injury [55]. The photoprotective effects of dietary phenolic compounds can be achieved by either oral administration or topical application. Green tea polyphenols also provide protection against UV irradiation-induced oxidative DNA damage, immunosuppression, and skin carcinogenesis in animal models [56, 57]. Notably, green tea polyphenols promote rapid repair of UV-induced DNA damage likely through stimulation of a nucleotide excision repair mechanism [58]. In addition, chemically-induced skin inflammation is attenuated by oral administration of green tea polyphenols in mice [59]. Moreover, green tea polyphenols have been demonstrated to promote keratinocyte differentiation and skin wound healing [60], protect hair follicles from γ -irradiation-induced apoptosis in vivo [61], and reduce psoriasiform lesions in the flaky skin mouse model [62]. The flaky skin mouse is a commonly used animal model of human psoriasis.

3.1.8. Role in Cancer

In addition to cardiovascular diseases, cancer is another most extensively studied disease with regard to phenolic compound-mediated protection in experimental animals [63-65]. Almost all of the classes of phenols listed in Section 1.2 are demonstrated to exhibit anticancer activities. Notable examples include resveratrol, curcumin, tea catechins, and soy flavonoids. The anticancer activities of phenolic compounds have been shown in animal models of both spontaneous cancer development and chemical carcinogenesis. It is known that cancer development involves multiple etiological factors as well as the complex interactions between genetic abnormalities and dysregulated cell signaling. As such, protective effects of phenolic compounds in cancer are unlikely due to their action on any specific pathway of cancer development. As described earlier in Section 2, phenolic compounds possess multiple biological activities, which may act together to contribute to the protection against cancer development. Largely based on studies in animal models and cell cultures, several mechanisms have been proposed to explain the beneficial effects of dietary phenolic compounds in cancer. These are listed below.

- Antioxidative effects
- Anti-inflammatory activities
- Antiproliferation
- Induction of apoptosis
- Antiangiogenesis
- Modulation of xenobiotic biotransformation enzymes

Dysregulated inflammation and oxidative stress are critical events in cancer development. Thus, suppression of oxidative stress and inflammation by phenolic compounds may contribute to cancer chemoprotection. Phenolic compounds, such as resveratrol, curcumin, and tea polyphenols are able to block cell cycle and induce apoptosis in cancer cells via modulating cancer cell signaling pathways. The pro-oxidant activities of phenolic compounds, such as epigallocatechin-3-gallate may also be involved in induction of apoptosis in cancer cells [9]. These effects along with antiangiogenesis may be primarily responsible for the suppression of cancer growth and metastasis. In this context, phenolic compounds, such as curcumin and tea polyphenols have been shown to potently block cancer invasion and metastasis in animal models [66, 67].

Phenolic compounds have long been known for their ability to protect against chemical carcinogenesis. Among the various mechanisms involved, induction of phase 2 enzymes has been demonstrated to play an important part in the detoxification of electrophilic carcinogens, thereby inhibiting the formation of carcinogen-DNA adducts, a critical event leading to initiation of the multistage carcinogenesis. In addition, some phenolic compounds, such as resveratorl can also inhibit P450 enzymes, resulting in decreased bioactivation of carcinogens to form DNA-damaging species [68]. Hence, modulation of carcinogen-metabolizing enzymes

3.1.9. Role in Other Diseases and Conditions

chemical carcinogenesis.

Substantial evidence suggests that dietary polyphenolic compounds also exert beneficial effects in various other pathophysiological conditions involving oxidative stress and inflammation in animal models. These include sepsis [69], arthritis [70, 71], HIV infection [72], and aging [73].

by phenolic compounds may be a particularly relevant mechanism for protection against

3.2. Human Studies and Clinical Perspectives

It has long been recognized that diet plays an important part in the maintenance of optimal health in humans. In this context, consumption of diet rich in fruits and vegetables has been associated with decreased risk of developing prevalent diseases, including cardiovascular disorders and cancer. As discussed in previous chapters, various bioactive components from diet rich in fruits and vegetables, including vitamin C, vitamin E, and carotenoids have been suggested to play a beneficial role in human diseases. Similarly, the health benefits of dietary phenolic compounds have also been extensively assessed over the last two decades in human subjects through both observational epidemiological studies and interventional clinical trials. Although cardiovascular diseases and cancer have been the focus of human studies, the potential beneficial effects of dietary phenolic compounds in other human diseases have also recently received attention. The sections below summarize the major findings from these human studies.

3.2.1. Role in Human Cardiovascular Diseases

A potential beneficial effect of dietary phenolic compounds in cardiovascular diseases was first suggested by an observational epidemiological study, the Zutphen Elderly Study, showing that flavonoid intake in regularly consumed foods was inversely associated with mortality from coronary heart disease in elderly men [74]. Since the Zutphen Elderly Study, there have been numerous observational epidemiological studies assessing the relationship between dietary intake of various classes of phenolic compounds and risk of developing cardiovascular diseases. Despite of null and inconsistent findings from some studies, the large majority of observational studies overwhelmingly suggest an inverse association between dietary intake of phenolic compounds and risk of developing different forms of cardiovascular diseases. In line with these findings, a number of small-scale interventional trials also suggest a potential benefit of supplementation of dietary phenolic compounds, such as cocoa polyphenols in the intervention of cardiovascular pathophysiology. The beneficial actions of polyphenols include: (1) improvement of hyperlipidemia and hypertension [75-77], (2) aug-

mentation of vascular nitric oxide formation and amelioration of endothelial dysfunction [21, 78], and (3) inhibition of platelet activation and aggregation [79]. These observations in human studies are also consistent with the findings from animal experiments, and point to a potential value of using dietary phenolic compounds in the intervention of human cardiovascular diseases.

3.2.2. Role in Human Cancer

The discovery that the wine polyphenol resveratrol protected against carcinogenesis in experimental animals in 1997 [80] sparked the recent surge of interest in the potential benefits of phenolic compounds in human cancer. The well-documented cancer chemoprotective effects of a variety of dietary phenolic compounds in animal models of human cancer have also prompted the studies on these compounds in protecting against cancer in human subjects. An increasing number of observational epidemiological studies have demonstrated an inverse association between dietary consumption of phenolic compounds, such as green tea polyphenols and soy flavonoids, and risk of cancer development.

In line with the findings in observational epidemiological studies, multiple small-scale interventional trials show a possible protective role for supplementation of dietary phenolic compounds in certain types of human cancer, such as prostate cancer. For example, in a recent randomized controlled trial, supplementation of tea catechins (600 mg per day) for one year was found to prevent the development of invasive prostate cancer in men with highgrade intra-epithelial neoplasia, suggesting that tea catechins may be effective for treating premalignant lesions of prostate [81]. A more recent interventional trial in prostate cancer patients reported that brief treatment with tea polyphenols (mainly epigallocatechin-3-gallate) resulted in a significant reduction of serum markers, including prostate-specific antigen, hepatocyte growth factor, and endothelial growth factor [82]. In addition to prostate cancer intervention, supplementation of green tea extract in a randomized controlled trial was found to improve clinical response rate in patients with high-risk oral premalignant lesions in a dosedependent manner, indicating a potential value for using green tea polyphenols in oral cancer prevention [83]. In another pilot trial, supplementation of green tea extract in patients who had colorectal adenomas removed by polypectomy was shown to reduce the development of metachronous colorectal adenoma compared with a group of patients who did not take the green tea extract [84].

It should be noted that there are also many small-scale clinical trials that have failed to show an effectiveness of supplementation of phenolic compounds, including green tea polyphenols in human cancer intervention. Thus, at the present time the exact value of phenolic compounds in human cancer intervention remains inconclusive. Hence, well-designed largescale interventional trials are needed to firmly establish the potential efficacy of various classes of dietary phenolic compounds in the intervention (both preventive and therapeutic) of common types of human cancer.

3.2.3. Role in Other Human Diseases and Conditions

Over the last two decades, the benefits of phenolic compounds in other human disorders have also been investigated through both observational epidemiological studies and small-scale interventional trials. Although many such studies have produced inconsistent results, some do point to a potential value of supplementation of dietary phenolic compounds in the intervention of certain disease conditions. These include metabolic syndrome [32, 33, 85, 86],

329

kidney transplantation [87, 88], bone loss in menopausal women [89-91], and UV irradiationinduced skin damage [92, 93].

4. Conclusion and Future Directions

Phenolic compounds represent an important component of human diet. In addition to having antioxidative and anti-inflammatory activities, these compounds also possess many other novel functions, such as modulation of cell signaling. The diseases-protective effects of dietary phenolic compounds have been well-recognized in various animal models of human diseases associated with oxidative stress and inflammation. Observational epidemiological studies and interventional clinical trials in selected patient groups also suggest potential health benefits of supplementation of dietary phenolic compounds in certain human diseases. However, the exact value of dietary phenolic compounds in human disease intervention remains to be established via large-scale well-designed interventional trials. Future studies should also emphasize the dose-response relationship of dietary phenolic compounds in disease intervention. As multiple classes of phenolic compounds as well as other antioxidant compounds coexist in regular human diet rich in fruits and vegetables, both basic and clinical studies should be directed to determine the interactions between the different groups of dietary antioxidant compounds with respect to their combined biological activities and role in disease intervention. Although dietary phenolic compounds have been shown to be well-tolerated in human subjects, potential adverse effects of these compounds, especially upon long-term supplementation at high doses should not be ignored. In this context, there are an increasing number of case reports of liver injury in human subjects associated with intake of green tea dietary supplements.

5. References

- [1] Stevenson DE, Hurst RD. Polyphenolic phytochemicals-just antioxidants or much more? *Cell Mol. Life Sci.* 2007;64:2900-2916.
- [2] Scalbert A, Williamson G. Dietary intake and bioavailability of polyphenols. J. Nutr. 2000;130:2073S-2085S.
- [3] Halliwell B, Rafter J, Jenner A. Health promotion by flavonoids, tocopherols, tocotrienols, and other phenols: direct or indirect effects? Antioxidant or not? *Am. J. Clin. Nutr.* 2005;81:268S-276S.
- [4] Sang S, Shao X, Bai N, Lo CY, Yang CS, Ho CT. Tea polyphenol (-)-epigallocatechin-3-gallate: a new trapping agent of reactive dicarbonyl species. *Chem. Res. Toxicol.* 2007;20:1862-1870.
- [5] Steffen Y, Gruber C, Schewe T, Sies H. Mono-O-methylated flavanols and other flavonoids as inhibitors of endothelial NADPH oxidase. *Arch. Biochem. Biophys.* 2008; 469:209-219.
- [6] Patel R, Maru G. Polymeric black tea polyphenols induce phase II enzymes via Nrf2 in mouse liver and lungs. *Free Radic. Biol. Med.* 2008;44:1897-1911.
- [7] Andreadi CK, Howells LM, Atherfold PA, Manson MM. Involvement of Nrf2, p38, B-

Raf, and nuclear factor-kappaB, but not phosphatidylinositol 3-kinase, in induction of hemeoxygenase-1 by dietary polyphenols. *Mol. Pharmacol.* 2006;69:1033-1040.

- [8] Halliwell B. Are polyphenols antioxidants or pro-oxidants? What do we learn from cell culture and in vivo studies? *Arch. Biochem. Biophys.* 2008;476:107-112.
- [9] Lambert JD, Elias RJ. The antioxidant and pro-oxidant activities of green tea polyphenols: a role in cancer prevention. *Arch. Biochem. Biophys.* 2010;501:65-72.
- [10] Rahman I, Biswas SK, Kirkham PA. Regulation of inflammation and redox signaling by dietary polyphenols. *Biochem. Pharmacol.* 2006;72:1439-1452.
- [11] Biesalski HK. Polyphenols and inflammation: basic interactions. *Curr. Opin. Clin. Nutr. Metab. Care* 2007;10:724-728.
- [12] Mann GE, Rowlands DJ, Li FY, de Winter P, Siow RC. Activation of endothelial nitric oxide synthase by dietary isoflavones: role of NO in Nrf2-mediated antioxidant gene expression. *Cardiovasc. Res.* 2007;75:261-274.
- [13] Stoclet JC, Chataigneau T, Ndiaye M, Oak MH, El Bedoui J, Chataigneau M, *et al.* Vascular protection by dietary polyphenols. *Eur. J. Pharmacol.* 2004;500:299-313.
- [14] Finkel T, Deng CX, Mostoslavsky R. Recent progress in the biology and physiology of sirtuins. *Nature* 2009;460:587-591.
- [15] Baur JA, Pearson KJ, Price NL, Jamieson HA, Lerin C, Kalra A, *et al.* Resveratrol improves health and survival of mice on a high-calorie diet. *Nature* 2006;444:337-342.
- [16] Howitz KT, Bitterman KJ, Cohen HY, Lamming DW, Lavu S, Wood JG, et al. Small molecule activators of sirtuins extend Saccharomyces cerevisiae lifespan. *Nature* 2003;425:191-196.
- [17] Chung S, Yao H, Caito S, Hwang JW, Arunachalam G, Rahman I. Regulation of SIRT1 in cellular functions: Role of polyphenols. Arch. Biochem. Biophys. 2010;501: 70-79.
- [18] Das DK, Mukherjee S, Ray D. Resveratrol and red wine, healthy heart and longevity. *Heart Fail. Rev.* 2010;15:467-477.
- [19] Manach C, Mazur A, Scalbert A. Polyphenols and prevention of cardiovascular diseases. Curr. Opin. Lipidol. 2005;16:77-84.
- [20] Stangl V, Dreger H, Stangl K, Lorenz M. Molecular targets of tea polyphenols in the cardiovascular system. *Cardiovasc. Res.* 2007;73:348-358.
- [21] Grassi D, Desideri G, Croce G, Tiberti S, Aggio A, Ferri C. Flavonoids, vascular function and cardiovascular protection. *Curr. Pharm. Des.* 2009;15:1072-1084.
- [22] Gendron ME, Theoret JF, Mamarbachi AM, Drouin A, Nguyen A, Bolduc V, et al. Late chronic catechin antioxidant treatment is deleterious to the endothelial function in aging mice with established atherosclerosis. Am. J. Physiol. Heart Circ. Physiol. 2010; 298:H2062-2070.
- [23] Sulaiman M, Matta MJ, Sunderesan NR, Gupta MP, Periasamy M, Gupta M. Resveratrol, an activator of SIRT1, upregulates sarcoplasmic calcium ATPase and improves cardiac function in diabetic cardiomyopathy. Am. J. Physiol. Heart Circ. Physiol. 2010;298:H833-843.
- [24] Uto-Kondo H, Ayaori M, Ogura M, Nakaya K, Ito M, Suzuki A, et al. Coffee consumption enhances high-density lipoprotein-mediated cholesterol efflux in macrophages. Circ. Res. 2010;106:779-787.
- [25] Burke MF, Khera AV, Rader DJ. Polyphenols and cholesterol efflux: is coffee the next red wine? *Circ. Res.* 2010;106:627-629.

- [26] Li HL, Liu C, de Couto G, Ouzounian M, Sun M, Wang AB, et al. Curcumin prevents and reverses murine cardiac hypertrophy. J. Clin. Invest. 2008;118:879-893.
- [27] Morimoto T, Sunagawa Y, Kawamura T, Takaya T, Wada H, Nagasawa A, *et al.* The dietary compound curcumin inhibits p300 histone acetyltransferase activity and prevents heart failure in rats. *J. Clin. Invest.* 2008;118:868-878.
- [28] Hanhineva K, Torronen R, Bondia-Pons I, Pekkinen J, Kolehmainen M, Mykkanen H, et al. Impact of dietary polyphenols on carbohydrate metabolism. Int. J. Mol. Sci. 2010; 11:1365-1402.
- [29] Dembinska-Kiec A, Mykkanen O, Kiec-Wilk B, Mykkanen H. Antioxidant phytochemicals against type 2 diabetes. Br. J. Nutr. 2008;99 E Suppl 1:ES109-117.
- [30] Hwang JT, Kwon DY, Yoon SH. AMP-activated protein kinase: a potential target for the diseases prevention by natural occurring polyphenols. *N. Biotechnol.* 2009;26:17-22.
- [31] Hsu CL, Yen GC. Phenolic compounds: evidence for inhibitory effects against obesity and their underlying molecular signaling mechanisms. *Mol. Nutr. Food Res.* 2008;52: 53-61.
- [32] Thielecke F, Boschmann M. The potential role of green tea catechins in the prevention of the metabolic syndrome: a review. *Phytochemistry* 2009;70:11-24.
- [33] Grove KA, Lambert JD. Laboratory, epidemiological, and human intervention studies show that tea (Camellia sinensis) may be useful in the prevention of obesity. *J. Nutr.* 2010;140:446-453.
- [34] Chaudhary N, Pfluger PT. Metabolic benefits from Sirt1 and Sirt1 activators. *Curr. Opin. Clin. Nutr. Metab. Care* 2009;12:431-437.
- [35] Feige JN, Lagouge M, Canto C, Strehle A, Houten SM, Milne JC, et al. Specific SIRT1 activation mimics low energy levels and protects against diet-induced metabolic disorders by enhancing fat oxidation. Cell Metab. 2008;8:347-358.
- [36] Mandel S, Youdim MB. Catechin polyphenols: neurodegeneration and neuroprotection in neurodegenerative diseases. *Free Radic. Biol. Med.* 2004;37:304-317.
- [37] Joseph J, Cole G, Head E, Ingram D. Nutrition, brain aging, and neurodegeneration. J. Neurosci. 2009;29:12795-12801.
- [38] Arab L, Liebeskind DS. Tea, flavonoids and stroke in man and mouse. *Arch. Biochem. Biophys.* 2010;501:31-36.
- [39] Sun AY, Wang Q, Simonyi A, Sun GY. Resveratrol as a therapeutic agent for neurodegenerative diseases. *Mol. Neurobiol.* 2010;41:375-383.
- [40] Darvesh AS, Carroll RT, Bishayee A, Geldenhuys WJ, Van der Schyf CJ. Oxidative stress and Alzheimer's disease: dietary polyphenols as potential therapeutic agents. *Expert. Rev. Neurother.* 2010;10:729-745.
- [41] Shah ZA, Li RC, Ahmad AS, Kensler TW, Yamamoto M, Biswal S, *et al.* The flavanol (-)-epicatechin prevents stroke damage through the Nrf2/HO1 pathway. *J. Cereb. Blood Flow Metab.* 2010;30:1951-1961.
- [42] Weinreb O, Amit T, Mandel S, Youdim MB. Neuroprotective molecular mechanisms of (-)-epigallocatechin-3-gallate: a reflective outcome of its antioxidant, iron chelating and neuritogenic properties. *Genes Nutr.* 2010 (in press).
- [43] Pallas M, Casadesus G, Smith MA, Coto-Montes A, Pelegri C, Vilaplana J, et al. Resveratrol and neurodegenerative diseases: activation of SIRT1 as the potential pathway towards neuroprotection. *Curr. Neurovasc. Res.* 2009;6:70-81.

- [44] Della-Morte D, Dave KR, DeFazio RA, Bao YC, Raval AP, Perez-Pinzon MA. Resveratrol pretreatment protects rat brain from cerebral ischemic damage via a sirtuin 1-uncoupling protein 2 pathway. *Neuroscience* 2009;159:993-1002.
- [45] Park HJ, Lee CM, Jung ID, Lee JS, Jeong YI, Chang JH, et al. Quercetin regulates Th1/Th2 balance in a murine model of asthma. Int. Immunopharmacol. 2009;9:261-267.
- [46] Kode A, Rajendrasozhan S, Caito S, Yang SR, Megson IL, Rahman I. Resveratrol induces glutathione synthesis by activation of Nrf2 and protects against cigarette smoke-mediated oxidative stress in human lung epithelial cells. Am. J. Physiol. Lung Cell Mol. Physiol. 2008;294:L478-488.
- [47] Sriram N, Kalayarasan S, Sudhandiran G. Epigallocatechin-3-gallate augments antioxidant activities and inhibits inflammation during bleomycin-induced experimen-tal pulmonary fibrosis through Nrf2-Keap1 signaling. *Pulm. Pharmacol. Ther.* 2009;22: 221-236.
- [48] Hwang JW, Chung S, Sundar IK, Yao H, Arunachalam G, McBurney MW, et al. Cigarette smoke-induced autophagy is regulated by SIRT1-PARP-1-dependent mechanism: implication in pathogenesis of COPD. Arch. Biochem. Biophys. 2010;500:203-209.
- [49] Ajmo JM, Liang X, Rogers CQ, Pennock B, You M. Resveratrol alleviates alcoholic fatty liver in mice. *Am. J. Physiol. Gastrointest. Liver Physiol.* 2008;295:G833-842.
- [50] Pfluger PT, Herranz D, Velasco-Miguel S, Serrano M, Tschop MH. Sirt1 protects against high-fat diet-induced metabolic damage. *Proc. Natl. Acad. Sci. USA* 2008;105: 9793-9798.
- [51] Dryden GW, Song M, McClain C. Polyphenols and gastrointestinal diseases. *Curr. Opin. Gastroenterol.* 2006;22:165-170.
- [52] Romier B, Schneider YJ, Larondelle Y, During A. Dietary polyphenols can modulate the intestinal inflammatory response. *Nutr. Rev.* 2009;67:363-378.
- [53] Rasbach KA, Schnellmann RG. Isoflavones promote mitochondrial biogenesis. J. *Pharmacol. Exp. Ther.* 2008;325:536-543.
- [54] He W, Wang Y, Zhang MZ, You L, Davis LS, Fan H, *et al.* Sirt1 activation protects the mouse renal medulla from oxidative injury. *J. Clin. Invest.* 2010;120:1056-1068.
- [55] Nichols JA, Katiyar SK. Skin photoprotection by natural polyphenols: antiinflammatory, antioxidant and DNA repair mechanisms. Arch. Dermatol. Res. 2010; 302:71-83.
- [56] Hsu S. Green tea and the skin. J. Am. Acad. Dermatol. 2005;52:1049-1059.
- [57] Yusuf N, Irby C, Katiyar SK, Elmets CA. Photoprotective effects of green tea polyphenols. *Photodermatol. Photoimmunol. Photomed.* 2007;23:48-56.
- [58] Katiyar SK, Vaid M, van Steeg H, Meeran SM. Green tea polyphenols prevent UVinduced immunosuppression by rapid repair of DNA damage and enhancement of nucleotide excision repair genes. *Cancer Prev. Res* (Phila Pa) 2010;3:179-189.
- [59] Katiyar SK, Agarwal R, Ekker S, Wood GS, Mukhtar H. Protection against 12-Otetradecanoylphorbol-13-acetate-caused inflammation in SENCAR mouse ear skin by polyphenolic fraction isolated from green tea. *Carcinogenesis* 1993;14:361-365.
- [60] Qin Y, Wang HW, Karuppanapandian T, Kim W. Chitosan green tea polyphenol complex as a released control compound for wound healing. *Chin. J. Traumatol.* 2010; 13:91-95.

- [61] Kim SH, Kim SR, Lee HJ, Oh H, Ryu SY, Lee YS, et al. Apoptosis in growing hair follicles following gamma-irradiation and application for the evaluation of radioprotective agents. In Vivo 2003;17:211-214.
- [62] Hsu S, Dickinson D, Borke J, Walsh DS, Wood J, Qin H, et al. Green tea polyphenol induces caspase 14 in epidermal keratinocytes via MAPK pathways and reduces psoriasiform lesions in the flaky skin mouse model. Exp. Dermatol. 2007;16:678-684.
- [63] Yang CS, Wang X, Lu G, Picinich SC. Cancer prevention by tea: animal studies, molecular mechanisms and human relevance. *Nat. Rev. Cancer* 2009;9:429-439.
- [64] Guo W, Kong E, Meydani M. Dietary polyphenols, inflammation, and cancer. *Nutr. Cancer* 2009;61:807-810.
- [65] Korkina LG, De Luca C, Kostyuk VA, Pastore S. Plant polyphenols and tumors: from mechanisms to therapies, prevention, and protection against toxicity of anti-cancer treatments. *Curr. Med. Chem.* 2009;16:3943-3965.
- [66] Bachmeier B, Nerlich AG, Iancu CM, Cilli M, Schleicher E, Vene R, *et al.* The chemopreventive polyphenol Curcumin prevents hematogenous breast cancer metastases in immunodeficient mice. *Cell Physiol. Biochem.* 2007;19:137-152.
- [67] Bracke ME, Vanhoecke BW, Derycke L, Bolca S, Possemiers S, Heyerick A, *et al.* Plant polyphenolics as anti-invasive cancer agents. *Anticancer Agents Med. Chem.* 2008;8:171-185.
- [68] Chen ZH, Hurh YJ, Na HK, Kim JH, Chun YJ, Kim DH, et al. Resveratrol inhibits TCDD-induced expression of CYP1A1 and CYP1B1 and catechol estrogen-mediated oxidative DNA damage in cultured human mammary epithelial cells. Carcinogenesis 2004;25:2005-2013.
- [69] Shapiro H, Lev S, Cohen J, Singer P. Polyphenols in the prevention and treatment of sepsis syndromes: rationale and pre-clinical evidence. *Nutrition* 2009;25:981-997.
- [70] Singh R, Akhtar N, Haqqi TM. Green tea polyphenol epigallocatechi3-gallate: inflammation and arthritis. *Life Sci.* 2010;86:907-918.
- [71] Aggarwal BB, Harikumar KB. Potential therapeutic effects of curcumin, the antiinflammatory agent, against neurodegenerative, cardiovascular, pulmonary, metabolic, autoimmune and neoplastic diseases. *Int. J. Biochem. Cell. Biol.* 2009;41:40-59.
- [72] Hauber I, Hohenberg H, Holstermann B, Hunstein W, Hauber J. The main green tea polyphenol epigallocatechin-3-gallate counteracts semen-mediated enhancement of HIV infection. *Proc. Natl. Acad. Sci. USA* 2009;106:9033-9038.
- [73] Queen BL, Tollefsbol TO. Polyphenols and aging. Curr. Aging Sci. 2010;3:34-42.
- [74] Hertog MG, Feskens EJ, Hollman PC, Katan MB, Kromhout D. Dietary antioxidant flavonoids and risk of coronary heart disease: the Zutphen Elderly Study. *Lancet* 1993;342:1007-1011.
- [75] Jia L, Liu X, Bai YY, Li SH, Sun K, He C, *et al.* Short-term effect of cocoa product consumption on lipid profile: a meta-analysis of randomized controlled trials. *Am. J. Clin. Nutr.* 2010;92:218-225.
- [76] Desch S, Schmidt J, Kobler D, Sonnabend M, Eitel I, Sareban M, et al. Effect of cocoa products on blood pressure: systematic review and meta-analysis. Am. J. Hypertens. 2010;23:97-103.
- [77] Egert S, Bosy-Westphal A, Seiberl J, Kurbitz C, Settler U, Plachta-Danielzik S, et al. Quercetin reduces systolic blood pressure and plasma oxidised low-density lipoprotein concentrations in overweight subjects with a high-cardiovascular disease risk pheno-

type: a double-blinded, placebo-controlled cross-over study. Br. J. Nutr. 2009; 102: 1065-1074.

- [78] Schroeter H, Heiss C, Balzer J, Kleinbongard P, Keen CL, Hollenberg NK, et al. (-)-Epicatechin mediates beneficial effects of flavanol-rich cocoa on vascular function in humans. Proc. Natl. Acad. Sci. USA 2006;103:1024-1029.
- [79] Ostertag LM, O'Kennedy N, Kroon PA, Duthie GG, de Roos B. Impact of dietary polyphenols on human platelet function--a critical review of controlled dietary intervention studies. *Mol. Nutr. Food Res.* 2010;54:60-81.
- [80] Jang M, Cai L, Udeani GO, Slowing KV, Thomas CF, Beecher CW, et al. Cancer chemopreventive activity of resveratrol, a natural product derived from grapes. Science 1997;275:218-220.
- [81] Bettuzzi S, Brausi M, Rizzi F, Castagnetti G, Peracchia G, Corti A. Chemoprevention of human prostate cancer by oral administration of green tea catechins in volunteers with high-grade prostate intraepithelial neoplasia: a preliminary report from a one-year proof-of-principle study. *Cancer Res.* 2006;66:1234-1240.
- [82] McLarty J, Bigelow RL, Smith M, Elmajian D, Ankem M, Cardelli JA. Tea polyphenols decrease serum levels of prostate-specific antigen, hepatocyte growth factor, and vascular endothelial growth factor in prostate cancer patients and inhibit production of hepatocyte growth factor and vascular endothelial growth factor in vitro. *Cancer Prev. Res.* (Phila Pa) 2009;2:673-682.
- [83] Tsao AS, Liu D, Martin J, Tang XM, Lee JJ, El-Naggar AK, et al. Phase II randomized, placebo-controlled trial of green tea extract in patients with high-risk oral premalignant lesions. *Cancer Prev. Res.* (Phila Pa) 2009;2:931-941.
- [84] Shimizu M, Fukutomi Y, Ninomiya M, Nagura K, Kato T, Araki H, et al. Green tea extracts for the prevention of metachronous colorectal adenomas: a pilot study. Cancer Epidemiol. Biomarkers. Prev. 2008;17:3020-3025.
- [85] Grassi D, Desideri G, Necozione S, Lippi C, Casale R, Properzi G, *et al.* Blood pressure is reduced and insulin sensitivity increased in glucose-intolerant, hypertensive subjects after 15 days of consuming high-polyphenol dark chocolate. *J. Nutr.* 2008; 138:1671-1676.
- [86] Orgaard A, Jensen L. The effects of soy isoflavones on obesity. *Exp. Biol. Med.* (Maywood) 2008;233:1066-1080.
- [87] Shoskes DA, Thomas M, Pobgee R, Fromkin B, Copley JB. Phase I study of oral bioflavonoids in cadaveric renal transplant recipients: effects on delayed graft function and calcineurin inhibitor toxicities. *Transplant. Proc.* 2003;35:841-842.
- [88] Shoskes D, Lapierre C, Cruz-Correa M, Muruve N, Rosario R, Fromkin B, et al. Beneficial effects of the bioflavonoids curcumin and quercetin on early function in cadaveric renal transplantation: a randomized placebo controlled trial. *Transplantation* 2005;80:1556-1559.
- [89] Ma DF, Qin LQ, Wang PY, Katoh R. Soy isoflavone intake increases bone mineral density in the spine of menopausal women: meta-analysis of randomized controlled trials. *Clin. Nutr.* 2008;27:57-64.
- [90] Ma DF, Qin LQ, Wang PY, Katoh R. Soy isoflavone intake inhibits bone resorption and stimulates bone formation in menopausal women: meta-analysis of randomized controlled trials. *Eur. J. Clin. Nutr.* 2008;62:155-161.
- [91] Taku K, Melby MK, Takebayashi J, Mizuno S, Ishimi Y, Omori T, et al. Effect of soy

isoflavone extract supplements on bone mineral density in menopausal women: metaanalysis of randomized controlled trials. *Asia Pac. J. Clin. Nutr.* 2010;19:33-42.

- [92] Katiyar SK, Matsui MS, Elmets CA, Mukhtar H. Polyphenolic antioxidant (-)epigallocatechin-3-gallate from green tea reduces UVB-induced inflammatory responses and infiltration of leukocytes in human skin. *Photochem. Photobiol.* 1999; 69:148-153.
- [93] Malhomme de la Roche H, Seagrove S, Mehta A, Divekar P, Campbell S, Curnow A. Using natural dietary sources of antioxidants to protect against ultraviolet and visible radiation-induced DNA damage: An investigation of human green tea ingestion. J. Photochem. Photobiol. B 2010;101:169-173.

SECTION IV SYNTHETIC ANTIOXIDANTS

A number of synthetic antioxidant compounds have been developed over the last several decades. Many of them are mimetics of endogenous antioxidant enzymes, including superoxide dismutase, catalase, and glutathione peroxidase. Some of these antioxidant enzyme mimetics possess certain specificities in terms of scavenging reactive oxygen and nitrogen species (ROS/RNS). The synthetic antioxidants also include glutathione precursors (e.g., *N*acetylcysteine), spin traps (e.g., α -phenyl-*tert*-butylnitrone), and nanomaterials (e.g., cerium oxide nanoparticles). Due to the non-protein nature, synthetic antioxidant compounds are relatively stable and usually able to penetrate the cells, and as such some of them can be administered orally. Synthetic antioxidants are widely used in both cell cultures and experimental animals to study the protective effects on ROS/RNS-mediated damage or to help identify the ROS/RNS involved in particular pathophysiological processes. In addition, certain synthetic antioxidant compounds have been studied in clinical trials for the intervention of diseases involving an oxidative stress mechanism. The subsequent Chapters 24-27 describe the chemical and biochemical properties of the above four classes of synthetic antioxidants with an emphasis on their protective role in disease pathophysiology.

Chapter 24

Antioxidant Enzyme Mimetics

Abstract

Antioxidant enzyme mimetics are synthetic small molecular mass compounds that catalytically metabolize reactive oxygen and nitrogen species, including superoxide, hydrogen peroxide, and peroxynitrite. The commonly studied antioxidant enzyme mimetics in biology and medicine include mimetics of superoxide dismutase, catalase, and glutathione peroxidase, as well as the catalytic scavengers of peroxynitrite. The protective effects of the various classes of antioxidant enzyme mimetics in disease pathophysiology have been demonstrated in a variety of animal models of human diseases, including cardiovascular disease, diabetes and metabolic syndrome, neurological disorders, and cancer. In contrast to the extensive experimental animal studies, there have been very limited clinical investigations on the safety and efficacy of antioxidant enzyme mimetics in human subjects.

1. Overview

The recognition of the critical involvement of reactive oxygen and nitrogen species (ROS/RNS) in disease pathophysiology has sparked the surge of interest in the potential use of antioxidant enzymes for disease intervention. Due to the protein nature and the associated instability, limited bioavailability, and antigenicity, systematic administration of native anti-oxidant enzymes for disease intervention has been much restricted. To circumvent these limitations, a number of synthetic antioxidant compounds that mimic the catalytic activities of endogenous antioxidant enzymes have been developed over the last two decades. These include mimetics of superoxide dismutase (SOD), catalase, and glutathione peroxidase (GPx). Some of the mimetics possess certain specificities in terms of catalytically scavenging or decomposing ROS/RNS. Compared with the native enzymes, antioxidant enzyme mimetics are relatively stable, able to penetrate the cells, and can be administered orally. These compounds are widely used as protective agents against oxidative stress in vitro and in vivo. They are also commonly employed as tools to study the involvement of ROS/RNS in disease pathophysiology. This chapter focuses on discussion of the commonly used mimetics of SOD, catalase, and GPx. The peroxynitrite scavengers are also covered in this chapter.

2. Chemistry and Biochemistry

2.1. Superoxide Dismutase Mimetics

Superoxide is crucially involved in various disease processes, and SOD enzymes play an important role in protecting against superoxide-mediated cell and tissue injury (Chapter 3). However, as stated above, use of the native SODs for suppressing oxidative tissue injury has many drawbacks, including limited bioavailability, antigenicity, and high expense. As such, several low molecular mass SOD mimetics have been developed to overcome the above limitations. Commonly used SOD mimetics are compounds that contain manganese (Mn) as the redox-active center, and include the following [1, 2].

- Mn(III) metalloporphyrins
- Mn(III) salen complexes
- Mn(II) pentaazamacrocyclic ligand-based complexes

In addition to these Mn-containing mimetics, several other structurally distinct chemicals have also been found to possess SOD-like activity. These include the nitroxide compound 4-hydroxy-2,2,6,6-tetramethylpiperidine-*N*-oxyl (Tempol) and fullerene derivatives [3, 4]. Cerium oxide nanoparticles may also have SOD-like activity. Fullerenes and cerium oxide nanoparticles are covered in Chapter 27.

2.1.1. Mn(III) Metalloporphyrins

The Mn-based metalloporphyrin complexes scavenge superoxide, hydrogen peroxide, peroxynitrite, and lipid peroxyl radicals. Although being referred to as SOD mimetics in the literature, these compounds apparently do not selectively scavenge superoxide. The commonly investigated Mn(III) metalloporphyrins include Mn(III) tetrakis (4-benzoic acid) porphyrin (abbreviation: MnTBAP), Mn(III) tetrakis (1-methyl-4-pyridyl) porphyrin (abbreviation: MnTMPyP), and the AEOL series, such as AEOL10113. The chemical name for AEOL10113 is Mn(III) tetrakis (*N*-ethylpyridinium-2-yl) porphyrin (abbreviation: MnTE-2-PyP⁵⁺). Due to its positive charge, MnTE-2-PyP⁵⁺ can accumulate in mitochondria [5]. The chemical structures for MnTMPyP, MnTBAP, and MnTE-2-PyP⁵⁺ are shown in Figure 24-1.

2.1.2. Mn(III) Salen Complexes

The Mn(III) salen complexes, also known as EUK series, scavenge superoxide and hydrogen peroxide, and thus have SOD and catalase activities. The members of this class include EUK-8, EUK-134, EUK-189, EUK-207, EUK-418, and EUK-423 [1].

2.1.3. Mn(II) Pentaazamacrocyclic Ligand-based Complexes

Mn(II) pentaazamacrocyclic ligand-based SOD mimetics, such as M40401 and M40403 are selective for superoxide, which is different from the other two classes of SOD mimetics described above. This selectivity for superoxide is believed to be due to the fact that the Mn atom at the center of the pentaazamacrocyclic ligand-based compounds is held by five coordination points and is only available for one-electron reduction [6]. In this regard, scavenging of hydrogen peroxide and peroxynitrite requires transfer of two electrons.



Figure 24-1. Structures of some commonly used superoxide dismutase (SOD) mimetics. See text (Section 2.1) for detailed description of various types of SOD mimetics. MnTBAP, Mn(III) tetrakis (4-benzoic acid) porphyrin; MnTMPyP, Mn(III) tetrakis (1-methyl-4-pyridyl) porphyrin; MnTE-2-PyP⁵⁺, Mn(III) tetrakis (*N*-ethylpyridinium-2-yl) porphyrin. (courtesy of Jason Z. Li)

2.2. Catalase Mimetics

Catalase is a heme-containing protein that catalyzes the dismutation of hydrogen peroxide to form water and molecular oxygen. Catalase mimetics are compounds with redox-active metal centers that often contain either manganese (Mn) or iron (Fe). As discussed above, the SOD mimetics Mn(III) metalloporphyrins and Mn(III) salen complexes also have catalase activity [7]. As such, these compounds are frequently designated as SOD/catalase mimetics.

2.3. GPx Mimetics

Another class of endogenous catalytic hydrogen peroxide scavengers is the seleniumcontaining GPx (Chapter 6). One of the best studied GPx mimetics is the selenoorganic compound 2-phenyl-1,2-benzisoselenazol-3(2H)-one, also known as ebselen (structure shown in Figure 24-2). Ebselen is one of the first selenium-based GPx mimetics developed. It catalytically scavenges hydrogen peroxide and other organic hydroperoxides in the presence of reducing cofactors, including glutathione, *N*-acetylcysteine, and dihydrolipoic acid. Ebselen also scavenges peroxynitrite and possibly other ROS/RNS. It is shown to be a substrate for thioredoxin reductase [8]. In addition to scavenging ROS/RNS, ebselen may also inhibit ROS-producing enzymes, including 5-lipoxygenase and NAD(P)H oxidase, and possess anti-inflammatory activity [7].



Figure 24-2. Structure of the glutathione peroxidase (GPx) mimetic ebselen. See text (Section 2.3) for detailed description of ebselen. The chemical name of ebselen is 2-phenyl-1,2-benzisoselenazol-3(2H)-one.

A number of diselenide and ditelluride containing compounds have been reported to catalytically scavenge peroxides with higher GPx-like activity than ebselen. For example, the diselenide compound 2,2'-diseleno-bis-beta-cyclodextrin (2-SeCD) can efficiently scavenge a variety of peroxides, including hydrogen peroxide using glutathione as a cofactor [7].

2.4. Peroxynitrite Scavengers

As noted above, peroxynitrite can be scavenged by manganese-based metalloporphyrin compounds with SOD and catalase activities as well as by GPx mimetics (e.g., ebselen). In contrast to manganese-based metalloporpyrins (e.g., MnTBAP, MnTMPyP), iron (Fe)-based metalloporphyrins display much higher rate of peroxynitrite decomposition and are relatively selective for peroxynitrite. Examples of commonly used iron-based peroxynitrite scavengers include 5,10,15,20-tetrakis (2,4,6-trimethyl-3,3-disulfonatophenyl) porphyrinato iron (III) (abbreviation: FeTMPS), 5,10,15,20-tetrakis (4-sulfonatophenyl) porphyrinato iron (III) (abbreviation: FeTPPS), and 5,10,15,20-tetrakis (*N*-methyl-4'-pyridyl) porphyrinato iron (III) (abbreviation: FeTMPY) [9].

3. Role in Disease Intervention

3.1. Animal Studies

The critical involvement of superoxide and other ROS/RNS in disease pathophysiology has prompted the use of catalytic antioxidant enzyme mimetics for disease intervention in experimental animals. As described above, many of the catalytic antioxidant enzyme mimetics lack selectivity for ROS/RNS. In addition, the catalytic scavenging ability of these compounds toward ROS/RNS as initially demonstrated in cell-free systems may not be the same in more complex biological systems. Moreover, these mimetics have been found to exert other biological activities, such as inhibition or induction of enzymes involved in ROS and xenobiotic metabolism [10, 11]. As such, interpretation of the protective effects of these antioxidant enzyme mimetics in disease pathophysiology should be done with caution. In this context, simultaneous assessment of the effects of the mimetics on ROS/RNS levels and oxidative stress markers in disease processes will help identify the mechanisms of action for these compounds in vivo.

Substantial evidence from studies in animal models shows beneficial effects of various antioxidant enzyme mimetics in disease conditions involving an oxidative stress mechanism. The sections below summarize the major experimental findings in this research area.

3.1.1. Role in Cardiovascular Diseases

A number of antioxidant enzyme mimetics are protective against hypertension, endothelial dysfunction, atherogenesis, myocardial ischemia-reperfusion injury, and heart failure. Notably, treatment with superoxide-selective Mn(II) pentaazamacrocyclic ligand-based SOD mimetics (e.g., M40401, M40403) is shown to improve endothelial function in atherosclerotic mice [12], ameliorate myocardial ischemia-reperfusion injury [13], and attenuate the development of heart failure [14]. Recently, the GPx mimetic ebselen is found to reduce atherosclerotic lesions in diabetic ApoE-deficient mice [15]. The above protective effects are correlated with decreased oxidative stress and inflammation in the target tissues, which support a critical role for superoxide as well as other ROS/RNS in the pathophysiology of cardiovascular diseases. These observations are also in line with the findings that genetic manipulations of endogenous antioxidant enzymes (e.g., SOD, catalase, GPx) significantly modulate the pathophysiology of the above cardiovascular diseases in animal models. Together, these studies support the notion that protection against cardiovascular diseases can be achieved by using catalytic antioxidant enzyme mimetics.

3.1.2. Role in Diabetes and Metabolic Syndrome

As discussed in previous chapters, genetic overexpression of endogenous antioxidant enzymes has been shown to protect against diabetes and diabetic complications in experimental animals. There is also substantial evidence for the beneficial effects of antioxidant mimetics, including Tempol, MnTMPyP, the superoxide-selective mimetics (e.g., M40401, M40403), and ebselen in diabetes and diabetic complications [16, 17]. Consistent with these in vivo findings, treatment with antioxidant mimetics protects the pancreatic β -cells from oxidative stress in vitro.

Oxidative stress and inflammation have been causally linked not only to diabetes, but also to obesity and obesity-associated complications. Treatment with antioxidant enzyme mimetics, such as Tempol reduces oxidative stress and improves insulin resistance in obese Zucker rats, indicating a potential value of these mimetics in the management of metabolic syndrome [18]. Obesity-associated vascular and renal dysfunction are also ameliorated by treatment with Tempol and other antioxidant enzyme mimetics [19]. Notably, supplementation of Tempol in the food of mice has been shown to cause significant weight loss and prevent obesity without toxicity [20]. More recently, Tempol is demonstrated to inhibit adipogenesis in vitro, which may underlie, at least in part, some of its in vivo effects on body weight and obesity [21]. These observations are consistent with the notion that oxidative stress is an important pathophysiological component of adiposity.

3.1.3. Role in Neurological Diseases

The increasing recognition of the critical role of oxidative stress and inflammation in various neurological disorders has prompted studies on the neuroprotective effects of catalytic antioxidant enzyme mimetics in experimental animals. Administration of antioxidant enzyme mimetics, including mimetics of SOD, catalase, and GPx, as well as peroxynitrite scavengers and fullerene derivatives has been demonstrated to protect against many neurological disorders [22]. These include cerebral ischemic injury, Parkinson's diseases, Alzheimer's disease, and neuroexcitotoxicity. Notably, treatment with antioxidant enzyme mimetics has also been found to prevent or reverse aging-related learning deficits and brain oxidative stress, and extent the lifespan in experimental animals [23-25]. These observations from a different angle support the oxidative stress mechanism of neurodegeneration and aging, and point to the feasibility of using antioxidant enzyme mimetics in the intervention of neurological disorders that involve an oxidative stress component.

3.1.4. Role in Pulmonary Diseases

Lung is exposed to higher oxygen tension than most other organs under both physiological and pathophysiological conditions. Indeed, oxidative stress has long been recognized as a causative factor in various pulmonary diseases. The protective effects of antioxidant enzyme mimetics have been demonstrated in many animal models of human lung diseases, including hyperoxic lung injury, pulmonary ischemia-reperfusion injury, asthma, chronic obstructive lung disease, acute respiratory distress syndrome, as well as radiation- and drug-induced lung injury [6, 26]. A notable benefit of these mimetics in the above lung pathophysiological processes is marked attenuation of inflammatory responses and oxidative stress. In addition, there is also evidence supporting the ability of antioxidant enzyme mimetics to inhibit lung fibrosis in experimental animals [27].

3.1.5. Role in Hepatic and Gastrointestinal Diseases

Antioxidant enzyme mimetics are protective against many liver disorders in animal models. For example, treatment with metalloporphyrin SOD mimetics (e.g., MnTMPyP, MnTBAP) or the diselenide GPx mimetic 2-SeCD attenuates hepatic ischemia-reperfusion injury, as evidenced by decreased release of liver enzymes and reduced inflammation and oxidative stress in liver tissue [28, 29]. Administration of MnTBAP also prevents Fasinduced acute liver failure and oxidative stress in mice [30]. Both alcoholic and nonalcoholic fatty liver diseases are ameliorated by treatment with antioxidant enzyme mimetics, such as Tempol and ebselen [31, 32]. Some antioxidant enzyme mimetics, including MnTBAP, ebselen, EUK-8, and EUK-134 have been shown to also protect against endotoxin-elicited liver injury and drug/xenobiotic-induced hepatotoxicity [33-35]. In addition to liver diseases, treatment with Tempol or M40403 suppresses chemically-induced colitis in animal models [36, 37]. Tempol also exerts a beneficial effect on intestinal ischemia-reperfusion injury via inhibiting superoxide formation and inflammatory responses in the intestine [38]. Together, these findings point to the potential value of catalytic antioxidant enzyme mimetics in the intervention of liver and gastrointestinal diseases associated with oxidative stress and dysregulated inflammation.

3.1.6. Role in Renal Diseases

The protective effects of antioxidant enzyme mimetics in kidney diseases have been investigated in experimental animals over the last decade. The renal disease conditions in which antioxidant enzyme mimetics have a protective role include ischemia-reperfusion injury [39], diabetic nephropathy [40], and drug/xenobiotic-induced nephrotoxicity [41, 42]. Treatment with antioxidant enzyme mimetics, such as Tempol also attenuates kidney inflammatory responses to acute sodium overload and protects against autoimmunity-induced glomerular injury [43, 44]. More recently, the SOD/catalase-mimetic MnTBAP has been shown to prevent the development of insulin resistance and glucose intolerance due to uremia in a mouse model of surgically induced chronic renal failure [45]. Treatment with MnTBAP also blocks the development of insulin resistance in normal mice after urea infusion [45]. These studies suggest broad applications of antioxidant enzyme mimetics in the management of diverse pathophysiological conditions involving kidney.

3.1.7. Role in Cancer

An early study by T.W. Kensler and coworkers in 1983 showed an inhibitory effect on tumor promotion by a copper-coordination complex with SOD mimetic activity [46]. This study provided the first evidence that a low molecular mass antioxidant enzyme mimetic may have an impact on multistage carcinogenesis in experimental animals. Subsequent studies have demonstrated that several other antioxidant enzyme mimetics, including MnTBAP, MnTE-2-PyP⁵⁺, and ebselen are also able to suppress chemical carcinogenesis in animal models [47-49].

Antioxidant enzyme mimetics are also protective against carcinogenesis in animal models unrelated to chemical carcinogens. For example, treatment with antioxidant enzyme mimetics Tempol and EUK-189 prolongs the latency of thymic lymphoma and the animal survival, and reduces oxidative stress in Atm-deficient mice. [50-52]. The Atm-deficient mouse is a murine model of human ataxia telangiectasia, a hereditary disorder characterized by a high incidence of lymphoid malignancies, premature aging, and other abnormalities. Recently, treatment with MnTE-2-PyP⁵⁺ is shown to inhibit angiogenesis and tumor growth via suppressing oxidative stress in a mouse model of breast cancer [53]. These findings point to the possibility of utilizing antioxidant enzyme mimetics for the intervention of human cancer.

The potential effects of antioxidant enzyme mimetics on cancer therapy, including chemotherapy and radiotherapy have been investigated in experimental animals. In a mouse model of prostate cancer, co-treatment with MnTE-2-PyP⁵⁺ does not significantly affect radiation-induced inhibition of tumor growth, but protects against radiation-induced decreases in red blood cell counts, hemoglobin, and hematocrit [54]. This suggests a role for MnTE-2-PyP⁵⁺ in reducing adverse effects of radiation therapy. However, in a murine model of lymphosarcoma, pretreatment with Tempol significantly reduces cyclophosphamide-induced tumor-killing activity. It is further shown that pretreatment with Tempol upregulates aldehyde dehydrogenase in liver, an enzyme known to detoxify the active metabolites of cyclophosphamide [11]. Such active metabolites are believed to be responsible for the antitumor activity of cyclophosphamide. Hence, the concurrent use of antioxidant enzyme mimetics may differentially affect the outcomes of cancer therapy depending on the types of the therapy as well as other experimental conditions.

3.1.8. Role in Other Diseases and Conditions

Catalytic antioxidant enzyme mimetics may also exert beneficial effects in other diseases and conditions in experimental animals. These include endotoxin-induced sepsis and multiple organ dysfunction [55], skin disorders (e.g., ultraviolet-induced injury) [56], arthritis [57], and total body irradiation-induced lethality [58].

3.2. Human Studies and Clinical Perspectives

In contrast to the substantial experimental evidence from animal studies, the clinical evidence for the effectiveness of antioxidant enzyme mimetics in disease intervention in human subjects is much limited. The GPx mimetic ebselen (Section 2.3 and Figure 24-2) is probably the only antioxidant enzyme mimetic whose potential therapeutic effects have been evaluated in multiple clinical studies. Several small-scale clinical trials in 1990s demonstrated a potential benefit of ebselen treatment in patients with acute ischemic stroke and patients with delayed neurological deficits after aneurysmal subarachnoid hemorrhage [59-61]. However, there are also trials reporting no beneficial effects of ebselen supplementation in patients with critical illness, including cerebrovascular events [62]. Thus, the neuroprotective effects of ebselen in humans remain to be established.

Topical application of antioxidant enzyme mimetics for disease intervention has been investigated in clinical trials. For example, a phase 1 trial reported that topical application of Tempol to the scalp before whole brain radiation was safe and might protect against radiationinduced alopecia in patients with metastatic cancer to the brain [63]. There is also evidence indicating that topical application of EUK-134 may suppress ultraviolet irradiation-induced oxidant formation on the skin surface in human subjects [64].

4. Conclusion and Future Directions

Antioxidant enzyme mimetics are able to catalytically metabolize ROS/RNS. While most of the available antioxidant enzyme mimetics lack selectivity for a particular ROS/RNS, some, such as Mn(II) pentaazamacrocyclic ligand-based SOD mimetics do show high specificity for superoxide. Antioxidant enzyme mimetics protect cells from oxidative stress. These compounds also exhibit beneficial effects in a variety of disease conditions involving oxidative stress and dysregulated inflammation in animal models. However, the potential usefulness of antioxidant enzyme mimetics in the intervention of human diseases currently remains largely unknown due to very limited clinical trials. Future studies should focus on the continued evaluation of both safety and efficacy of the existing antioxidant enzyme mimetics in preclinical models as well as developing more effective new compounds for preclinical assessment. These preclinical studies will lay a foundation for subsequent well-designed clinical trials to determine the potential effectiveness of antioxidant enzyme mimetics in human disease intervention, including both prevention and treatment.

5. References

- [1] Muscoli C, Cuzzocrea S, Riley DP, Zweier JL, Thiemermann C, Wang ZQ, *et al.* On the selectivity of superoxide dismutase mimetics and its importance in pharmacological studies. *Br. J. Pharmacol.* 2003;140:445-460.
- [2] Batinic-Haberle I, Reboucas JS, Spasojevic I. Superoxide dismutase mimics: chemistry, pharmacology, and therapeutic potential. *Antioxid. Redox Signal* 2010;13: 877-918.
- [3] Simonsen U, Christensen FH, Buus NH. The effect of tempol on endotheliumdependent vasodilatation and blood pressure. *Pharmacol. Ther.* 2009;122:109-124.
- [4] Ali SS, Hardt JI, Quick KL, Kim-Han JS, Erlanger BF, Huang TT, *et al.* A biologically effective fullerene (C60) derivative with superoxide dismutase mimetic properties. *Free Radic. Biol. Med.* 2004;37:1191-1202.
- [5] Spasojevic I, Chen Y, Noel TJ, Yu Y, Cole MP, Zhang L, et al. Mn porphyrin-based superoxide dismutase (SOD) mimic, MnIIITE-2-PyP⁵⁺, targets mouse heart mitochondria. *Free Radic. Biol. Med.* 2007;42:1193-1200.
- [6] Day BJ. Catalytic antioxidants: a radical approach to new therapeutics. *Drug Discov. Today* 2004;9:557-566.
- [7] Day BJ. Catalase and glutathione peroxidase mimics. *Biochem. Pharmacol.* 2009;77: 285-296.
- [8] Zhao R, Masayasu H, Holmgren A. Ebselen: a substrate for human thioredoxin reductase strongly stimulating its hydroperoxide reductase activity and a superfast thioredoxin oxidant. *Proc. Natl. Acad. Sci*. USA 2002;99:8579-8584.
- [9] Rabkin SW, Klassen SS. Metalloporphyrins as a therapeutic drug class against peroxynitrite in cardiovascular diseases involving ischemic reperfusion injury. *Eur. J. Pharmacol.* 2008;586:1-8.
- [10] Cotgreave IA, Duddy SK, Kass GE, Thompson D, Moldeus P. Studies on the antiinflammatory activity of ebselen. Ebselen interferes with granulocyte oxidative burst by dual inhibition of NADPH oxidase and protein kinase C? *Biochem. Pharmacol.* 1989;38:649-656.
- [11] Kaledin VI, Popova NA, Nikolin VP, Vasilieva ED, Kirilyuk IA, Grigor'ev IA. Tempol reduces the therapeutic effect of cyclophosphamide on an experimental tumour model. *Free Radic. Res.* 2009;43:685-690.
- [12] Jiang F, Guo Y, Salvemini D, Dusting GJ. Superoxide dismutase mimetic M40403 improves endothelial function in apolipoprotein(E)-deficient mice. *Br. J. Pharmacol.* 2003;139:1127-1134.
- [13] Masini E, Cuzzocrea S, Mazzon E, Marzocca C, Mannaioni PF, Salvemini D. Protective effects of M40403, a selective superoxide dismutase mimetic, in myocardial ischaemia and reperfusion injury in vivo. *Br. J. Pharmacol.* 2002;136:905-917.
- [14] Lu Z, Xu X, Hu X, Lee S, Traverse JH, Zhu G, et al. Oxidative stress regulates left ventricular PDE5 expression in the failing heart. *Circulation* 2010;121:1474-1483.
- [15] Chew P, Yuen DY, Koh P, Stefanovic N, Febbraio MA, Kola I, et al. Site-specific antiatherogenic effect of the antioxidant ebselen in the diabetic apolipoprotein Edeficient mouse. Arterioscler. Thromb. Vasc. Biol. 2009;29:823-830.
- [16] Nassar T, Kadery B, Lotan C, Da'as N, Kleinman Y, Haj-Yehia A. Effects of the

superoxide dismutase-mimetic compound tempol on endothelial dysfunction in streptozotocin-induced diabetic rats. *Eur. J. Pharmacol.* 2002;436:111-118.

- [17] Vassort G, Turan B. Protective role of antioxidants in diabetes-induced cardiac dysfunction. *Cardiovasc. Toxicol.* 2010;10:73-86.
- [18] Xiang L, Dearman J, Abram SR, Carter C, Hester RL. Insulin resistance and impaired functional vasodilation in obese Zucker rats. Am. J. Physiol. Heart Circ. Physiol. 2008; 294:H1658-1666.
- [19] Ebenezer PJ, Mariappan N, Elks CM, Haque M, Francis J. Diet-induced renal changes in Zucker rats are ameliorated by the superoxide dismutase mimetic TEMPOL. *Obesity* (Silver Spring) 2009;17:1994-2002.
- [20] Mitchell JB, Xavier S, DeLuca AM, Sowers AL, Cook JA, Krishna MC, et al. A low molecular weight antioxidant decreases weight and lowers tumor incidence. Free Radic. Biol. Med. 2003;34:93-102.
- [21] Samuni Y, Cook JA, Choudhuri R, Degraff W, Sowers AL, Krishna MC, et al. Inhibition of adipogenesis by Tempol in 3T3-L1 cells. Free Radic. Biol. Med. 2010;49: 667-673.
- [22] Pong K. Oxidative stress in neurodegenerative diseases: therapeutic implications for superoxide dismutase mimetics. *Expert. Opin. Biol. Ther.* 2003;3:127-139.
- [23] Liu R, Liu IY, Bi X, Thompson RF, Doctrow SR, Malfroy B, et al. Reversal of agerelated learning deficits and brain oxidative stress in mice with superoxide dismutase/catalase mimetics. Proc. Natl. Acad. Sci. USA 2003;100:8526-8531.
- [24] Quick KL, Ali SS, Arch R, Xiong C, Wozniak D, Dugan LL. A carboxyfullerene SOD mimetic improves cognition and extends the lifespan of mice. *Neurobiol. Aging* 2008; 29:117-128.
- [25] Clausen A, Doctrow S, Baudry M. Prevention of cognitive deficits and brain oxidative stress with superoxide dismutase/catalase mimetics in aged mice. *Neurobiol. Aging* 2010;31:425-433.
- [26] Rabbani ZN, Batinic-Haberle I, Anscher MS, Huang J, Day BJ, Alexander E, et al. Long-term administration of a small molecular weight catalytic metalloporphyrin antioxidant, AEOL 10150, protects lungs from radiation-induced injury. Int. J. Radiat. Oncol. Biol. Phys. 2007;67:573-580.
- [27] Day BJ. Antioxidants as potential therapeutics for lung fibrosis. *Antioxid. Redox Signal* 2008;10:355-370.
- [28] Moon KH, Hood BL, Mukhopadhyay P, Rajesh M, Abdelmegeed MA, Kwon YI, et al. Oxidative inactivation of key mitochondrial proteins leads to dysfunction and injury in hepatic ischemia reperfusion. *Gastroenterology* 2008;135:1344-1357.
- [29] Llacuna L, Mari M, Lluis JM, Garcia-Ruiz C, Fernandez-Checa JC, Morales A. Reactive oxygen species mediate liver injury through parenchymal nuclear factorkappaB inactivation in prolonged ischemia/reperfusion. *Am. J. Pathol.* 2009;174:1776-1785.
- [30] Malassagne B, Ferret PJ, Hammoud R, Tulliez M, Bedda S, Trebeden H, et al. The superoxide dismutase mimetic MnTBAP prevents Fas-induced acute liver failure in the mouse. *Gastroenterology* 2001;121:1451-1459.
- [31] Kono H, Arteel GE, Rusyn I, Sies H, Thurman RG. Ebselen prevents early alcoholinduced liver injury in rats. *Free Radic. Biol. Med.* 2001;30:403-411.
- [32] Wei Y, Clark SE, Thyfault JP, Uptergrove GM, Li W, Whaley-Connell AT, et al.

Oxidative stress-mediated mitochondrial dysfunction contributes to angiotensin IIinduced nonalcoholic fatty liver disease in transgenic Ren2 rats. *Am. J. Pathol.* 2009; 174:1329-1337.

- [33] Bianca RV, Wayman NS, McDonald MC, Pinto A, Shape MA, Chatterjee PK, et al. Superoxide dismutase mimetic with catalase activity, EUK-134, attenuates the multiple organ injury and dysfunction caused by endotoxin in the rat. Med. Sci. Monit. 2002;8:BR1-7.
- [34] Koyanagi T, Nakamuta M, Enjoji M, Iwamoto H, Motomura K, Sakai H, *et al.* The selenoorganic compound ebselen suppresses liver injury induced by Propionibacterium acnes and lipopolysaccharide in rats. *Int. J. Mol. Med.* 2001;7:321-327.
- [35] Ferret PJ, Hammoud R, Tulliez M, Tran A, Trebeden H, Jaffray P, *et al.* Detoxification of reactive oxygen species by a nonpeptidyl mimic of superoxide dismutase cures acetaminophen-induced acute liver failure in the mouse. *Hepatology* 2001;33:1173-1180.
- [36] Karmeli F, Eliakim R, Okon E, Samuni A, Rachmilewitz D. A stable nitroxide radical effectively decreases mucosal damage in experimental colitis. *Gut* 1995;37:386-393.
- [37] Cuzzocrea S, Mazzon E, Dugo L, Caputi AP, Riley DP, Salvemini D. Protective effects of M40403, a superoxide dismutase mimetic, in a rodent model of colitis. *Eur. J. Pharmacol.* 2001;432:79-89.
- [38] Abdallah DM, El-Abhar HS, Abdel-Aziz DH. TEMPOL, a membrane-permeable radical scavenger, attenuates gastric mucosal damage induced by ischemia/reperfusion: a key role for superoxide anion. *Eur. J. Pharmacol.* 2009;603:93-97.
- [39] Liang HL, Hilton G, Mortensen J, Regner K, Johnson CP, Nilakantan V. MnTMPyP, a cell-permeant SOD mimetic, reduces oxidative stress and apoptosis following renal ischemia-reperfusion. Am. J. Physiol. Renal. Physiol. 2009;296:F266-276.
- [40] Peixoto EB, Pessoa BS, Biswas SK, Lopes de Faria JB. Antioxidant SOD mimetic prevents NADPH oxidase-induced oxidative stress and renal damage in the early stage of experimental diabetes and hypertension. *Am. J. Nephrol.* 2009;29:309-318.
- [41] Naghibi B, Ghafghazi T, Hajhashemi V, Talebi A, Taheri D. The effect of 2,3dihydroxybenzoic acid and tempol in prevention of vancomycin-induced nephrotoxicity in rats. *Toxicology* 2007;232:192-199.
- [42] Samai M, Sharpe MA, Gard PR, Chatterjee PK. Comparison of the effects of the superoxide dismutase mimetics EUK-134 and tempol on paraquat-induced nephrotoxicity. *Free Radic. Biol. Med* .2007;43:528-534.
- [43] Roson MI, Della Penna SL, Cao G, Gorzalczany S, Pandolfo M, Toblli JE, *et al.* Different protective actions of losartan and tempol on the renal inflammatory response to acute sodium overload. *J. Cell Physiol.* 2010;224:41-48.
- [44] Duann P, Datta PK, Pan C, Blumberg JB, Sharma M, Lianos EA. Superoxide dismutase mimetic preserves the glomerular capillary permeability barrier to protein. J. *Pharmacol. Exp. Ther.* 2006;316:1249-1254.
- [45] D'Apolito M, Du X, Zong H, Catucci A, Maiuri L, Trivisano T, et al. Urea-induced ROS generation causes insulin resistance in mice with chronic renal failure. J. Clin. Invest. 2010;120:203-213.
- [46] Kensler TW, Bush DM, Kozumbo WJ. Inhibition of tumor promotion by a biomimetic superoxide dismutase. *Science* 1983;221:75-77.
- [47] Park KK, Park JH, Jung YJ, Chung WY. Inhibitory effects of chlorophyllin, hemin and

tetrakis(4-benzoic acid)porphyrin on oxidative DNA damage and mouse skin inflammation induced by 12-O-tetradecanoylphorbol-13-acetate as a possible anti-tumor promoting mechanism. *Mutat. Res.* 2003;542:89-97.

- [48] Yang CF, Liu J, Wasser S, Shen HM, Tan CE, Ong CN. Inhibition of ebselen on aflatoxin B(1)-induced hepatocarcinogenesis in Fischer 344 rats. *Carcinogenesis* 2000;21:2237-2243.
- [49] Zhao Y, Chaiswing L, Oberley TD, Batinic-Haberle I, St Clair W, Epstein CJ, et al. A mechanism-based antioxidant approach for the reduction of skin carcinogenesis. *Cancer Res.* 2005;65:1401-1405.
- [50] Schubert R, Erker L, Barlow C, Yakushiji H, Larson D, Russo A, *et al.* Cancer chemoprevention by the antioxidant tempol in Atm-deficient mice. *Hum. Mol. Genet.* 2004;13:1793-1802.
- [51] Reliene R, Fleming SM, Chesselet MF, Schiestl RH. Effects of antioxidants on cancer prevention and neuromotor performance in Atm deficient mice. *Food Chem. Toxicol.* 2008;46:1371-1377.
- [52] Reliene R, Schiestl RH. Antioxidants suppress lymphoma and increase longevity in Atm-deficient mice. J. Nutr. 2007;137:2298-2328.
- [53] Rabbani ZN, Spasojevic I, Zhang X, Moeller BJ, Haberle S, Vasquez-Vivar J, et al. Antiangiogenic action of redox-modulating Mn(III) meso-tetrakis(N-ethylpyridinium-2-yl)porphyrin, MnTE-2-PyP⁵⁺, via suppression of oxidative stress in a mouse model of breast tumor. *Free Radic. Biol. Med.* 2009;47:992-1004.
- [54] Makinde AY, Luo-Owen X, Rizvi A, Crapo JD, Pearlstein RD, Slater JM, et al. Effect of a metalloporphyrin antioxidant (MnTE-2-PyP) on the response of a mouse prostate cancer model to radiation. Anticancer Res. 2009;29:107-118.
- [55] Esposito E, Cuzzocrea S. Role of nitroso radicals as drug targets in circulatory shock. *Br. J. Pharmacol.* 2009;157:494-508.
- [56] Fujimori T, Yasui H, Hiromura M, Sakurai H. Suppressive effect of orally administered copper(II)-aspirinate (Cu₂(asp)₄) complex on the generation of reactive oxygen species in the skin of animals subjected to UVA exposure. *Exp. Dermatol.* 2007; 16:746-752.
- [57] Salvemini D, Mazzon E, Dugo L, Serraino I, De Sarro A, Caputi AP, et al. Amelioration of joint disease in a rat model of collagen-induced arthritis by M40403, a superoxide dismutase mimetic. Arthritis Rheum. 2001;44:2909-2921.
- [58] Thompson JS, Chu Y, Glass J, Tapp AA, Brown SA. The manganese superoxide dismutase mimetic, M40403, protects adult mice from lethal total body irradiation. *Free Radic. Res.* 2010;44:529-540.
- [59] Yamaguchi T, Sano K, Takakura K, Saito I, Shinohara Y, Asano T, et al. Ebselen in acute ischemic stroke: a placebo-controlled, double-blind clinical trial. Ebselen Study Group. Stroke 1998;29:12-17.
- [60] Ogawa A, Yoshimoto T, Kikuchi H, Sano K, Saito I, Yamaguchi T, *et al.* Ebselen in acute middle cerebral artery occlusion: a placebo-controlled, double-blind clinical trial. *Cerebrovasc. Dis.* 1999;9:112-118.
- [61] Saito I, Asano T, Sano K, Takakura K, Abe H, Yoshimoto T, *et al.* Neuroprotective effect of an antioxidant, ebselen, in patients with delayed neurological deficits after aneurysmal subarachnoid hemorrhage. *Neurosurgery* 1998;42:269-277.
- [62] Avenell A, Noble DW, Barr J, Engelhardt T. Selenium supplementation for critically ill

adults. Cochrane Database Syst. Rev. 2004:CD003703.

- [63] Metz JM, Smith D, Mick R, Lustig R, Mitchell J, Cherakuri M, *et al.* A phase I study of topical Tempol for the prevention of alopecia induced by whole brain radiotherapy. *Clin. Cancer Res*. 2004;10:6411-6417.
- [64] Declercq L, Sente I, Hellemans L, Corstjens H, Maes D. Use of the synthetic superoxide dismutase/catalase mimetic EUK-134 to compensate for seasonal antioxidant deficiency by reducing pre-existing lipid peroxides at the human skin surface. *Int. J. Cosmet. Sci.* 2004;26:255-263.
Chapter 25

Glutathione Precursors

Abstract

The reduced form of glutathione (GSH) is an important biomolecule in mammalian health and disease. Glutathione precursors include the compounds that directly give rise to GSH and those that provide substrates for de novo GSH biosynthesis. Thus, GSH precursors represent an important approach to increasing cellular and tissue GSH levels. GSH precursors may possess other novel biological activities in addition to augmenting cellular and tissue GSH levels. The protective effects of GSH precursors, especially *N*-acetylcysteine in disease pathophysiology have been well-recognized in a variety of animal models of human diseases. A number of clinical trials also suggest a potential beneficial role for *N*-acetylcysteine in the intervention of many human diseases, including cardiovascular disorders, diabetes, chronic obstructive pulmonary disease, and contrast medium-induced nephropathy.

1. Overview

The reduced form of glutathione (GSH) is a critical antioxidant in mammals, and increased GSH levels have been shown to provide protection against oxidative and electrophilic injury under various disease conditions (see Chapter 5 for discussion of GSH). However, direct administration of GSH to increase its tissue levels has proven inefficient due to the membrane impermeability and instability of GSH. As such, many membrane permeable and relatively stable compounds, known as GSH precursors or prodrugs have been developed, which can be administered to elevate GSH levels in vivo via various mechanisms. The term GSH precursors refers to synthetic compounds that can increase tissue GSH levels via either directly giving rise to GSH or providing substrates for de novo GSH biosynthesis. The GSH precursors acting via the former mechanism include GSH esters, whereas those acting through the latter mechanism consist of N-acetylcysteine and a number of other cysteine analogs. N-Acetylcysteine is an old drug approved by the United States Food and Drug Administration (FDA), which has been used as the specific antidote for acetaminophen overdose. It has also been extensively studied as a potential protective agent for the management of a variety of other pathophysiological conditions. This chapter discusses the basic chemical and biochemical properties of GSH precursors and their involvement in health and disease with an emphasis on N-acetylcysteine.

2. Chemistry and Biochemistry

2.1. Glutathione Esters

Since GSH is not significantly taken up by cells, an effective approach to increasing cellular GSH levels is to supply a membrane permeable GSH-containing compound that is well transported into cells and converted to GSH. In this regard, membrane permeable GSH esters have been developed to provide an effective way of delivering GSH into cells. There are two types of GSH esters: GSH monoethyl ester and GSH diethyl ester. When administered to experimental animals, GSH esters can be transported into many tissues and hydrolyzed by tissue esterases to release GSH, resulting in increased tissue levels of GSH [1].

2.2. N-Acetylcysteine

Another way of augmenting tissue GSH levels is to increase the availability of the ratelimiting substrate cysteine using the non-toxic precursor *N*-acetylcysteine (structure shown in Figure 25-1). Orally delivered *N*-acetylcysteine is readily absorbed and converted by esterases to cysteine in tissues (mainly liver), where it is used for synthesis of GSH. In addition to supplying cysteine for GSH synthesis, *N*-acetylcysteine also directly scavenges ROS/RNS and binds redox-active metal ions. *N*-Acetylcysteine has been shown to possess anti-inflammatory activities through mechanisms including inhibition of NF- κ B activation and suppression of the expression of proinflammatory cytokines and adhesion molecules [2, 3]. Moreover, it has been reported that *N*-acetylcysteine inhibits platelet aggregation via increasing platelet nitric oxide [4]. These novel effects may also contribute to *N*-acetylcysteine-mediated protection against disease pathophysiology.



Figure 25-1. Structure of N-acetylcysteine. See text (Section 2.2) for description of N-acetylcysteine.

2.3. Other Glutathione Precursors

In addition to GSH esters and *N*-acetylcysteine, there are many other GSH precursors that can be utilized to increase tissue GSH levels. These include various analogs of cysteine, such as *S*-carboxymethylcysteine, *N*-isobutyrylcysteine, *N*-acystelyn, erdosteine, and fudosteine. In general, these cysteine-derived analogs have improved stability and bioavailability compared

with *N*-acetylcysteine. These compounds have also been investigated in animal studies and clinical trials for the intervention of certain diseases, such as chronic obstructive pulmonary disease [5].

3. Role in Disease Intervention

3.1. Animal Studies

3.1.1. Role in Cardiovascular Diseases

The protective effects of GSH precursors have been demonstrated in animal models of human cardiovascular diseases, including hypertension, atherosclerosis, myocardial ischemiareperfusion injury, heart failure, and drug-induced cardiotoxicity. Notably, administration of either GSH monoethyl ester or *N*-acetylcysteine attenuates atherosclerotic plaque formation and reduces oxidative stress in vascular wall in experimental animals fed atherosclerotic diet [6, 7]. *N*-Acetylcysteine also prevents accelerated atherosclerosis, vascular inflammation, and oxidative stress in uremic ApoE-deficient mice [8]. These findings indicate GSH as an important protective molecule and point to the feasibility for using thiol compounds in the intervention of vascular degeneration.

3.1.2. Role in Diabetes and Metabolic Syndrome

GSH is a critical molecule in protecting the pancreatic β -cells from oxidative stress and inflammation, and in suppressing the development of diabetes in animal models (Chapter 5). There is substantial evidence supporting the beneficial effects of *N*-acetylcysteine and GSH esters in diabetic complications, including cardiomyopathy, neuropathy, vascular dysfunction, and cataract. It is of note that *N*-acetylcysteine protects against hyperglycemia-mediated myocardial hypertrophy likely through attenuation of PKC- β 2 overexpression [9]. PKC- β 2 overexpression is an important mechanism underlying diabetic cardiac hypertrophy. It has been reported that treatment with *N*-acetylcysteine restores myocardial manganese superoxide dismutase activity in diabetic animals, which may partially contribute to the improvement of diabetic cardiomyopathy [10].

N-Acetylcysteine is also beneficial in treating metabolic syndrome. For example, oral administration of *N*-acetylcysteine ameliorates arterial hypertension, insulin resistance, and oxidative stress in chronically glucose-fed rats [11]. Dietary supplementation of *N*-acetylcysteine also alleviates sucrose-induced oxidative stress and insulin resistance in rats. Notably, the beneficial effects of *N*-acetylcysteine on sucrose-induced oxidative stress and insulin resistance can be duplicated by consumption of a cysteine-rich diet [12]. More recently, *N*-acetylcysteine is found to prevent high-sucrose diet-induced obesity and metabolic disturbances in experimental animals [13].

3.1.3. Role in Neurological Diseases

The neuroprotective effects of GSH esters have been well-documented in various animal models of human neurological disorders. For example, infusion of GSH monoethyl ester into the third ventricle during middle cerebral artery occlusion and reperfusion is found to markedly reduce the infarct size in experimental animals [14]. Administration of GSH monoethyl

ester prevents mitochondrial GSH depletion during focal cerebral ischemia, which may be an important mechanism underlying its protection against stroke in animal models [15]. Intraperitoneal injection of GSH monoethyl ester improves functional recovery, enhances neuron survival, and stabilizes spinal cord blood flow after spinal cord injury in rats [16]. There is also evidence showing beneficial actions of both GSH esters and *N*-acetylcysteine in other neurodegenerative diseases, such as Parkinson's disease [17]. In this regard, GSH depletion in dopaminergic neurons is an early pathophysiological event of Parkinson's disease. Thus, repletion of GSH via GSH esters and *N*-acetylcysteine may represent an effective approach to the management of Parkinson's disease. In addition to Parkinson's disease, brain GSH deficiencies have also been observed in aged animals. Supplementation of GSH monoethyl ester is shown to restore the aging-related GSH deficiencies, oxidative stress, and neurochemical alterations [18-20].

3.1.4. Role in Pulmonary Diseases

The best studied pulmonary disease with respect to protection by GSH precursors is chronic obstructive pulmonary disease (COPD). There is substantial evidence showing a beneficial role for GSH precursors, including *N*-acetylcysteine and *S*-carboxymethylcysteine in COPD in experimental animals [5, 21, 22]. The protective effects of these GSH precursors in COPD have been associated with decreased pulmonary inflammation and oxidative stress. In line with this, GSH precursors, including *N*-acetylcysteine, *S*-carboxymethylcysteine, and GSH monoethyl ester have also been found to attenuate airway inflammation and oxidative stress is shown to alleviate hyperoxia-induced lung injury [25, 26]. Moreover, oral supplementation of *N*-acetylcysteine protects against bleomycin-elicited lung oxidative stress, inflammation, and fibrosis in experimental animals [27, 28]. In cultured bronchial epithelial cells, *N*-acetylcysteine treatment inhibits bleomycin-induced secretion of proinflammatory cytokines, which is consistent with the in vivo observations [29].

3.1.5. Role in Hepatic and Gastrointestinal Diseases

GSH precursors are protective in liver diseases, including ischemia-reperfusion injury, nonalcoholic fatty liver disease, and drug/xenobiotic-induced liver injury. Intravenous infusion of *N*-acetylcysteine or GSH monoethyl ester attenuates liver ischemia-reperfusion injury, as evidenced by decreased release of liver enzymes and reduced oxidative stress [30, 31]. In addition, GSH monoethyl ester treatment protects liver mitochondria from oxidative damage during ischemia-reperfusion. The beneficial effects of both *N*-acetylcysteine and GSH esters in nonalcoholic fatty liver disease have been well-documented in various animal models, including the nutritional models of either feeding a diet deficient in both methionine and choline or feeding a high-fat diet [32, 33].

N-Acetylcysteine is best known as the specific antidote for acetaminophen overdose. This is primarily due to the increased GSH levels in liver, leading to enhanced detoxification of the reactive metabolite of acetaminophen, *N*-acetyl-*p*-benzoquinoneimine (Figure 25-2). *N*-Acetylcysteine or GSH esters also ameliorate liver injury induced by other drugs/xenobiotics, such as cyclophosphamide, carmustine, ethanol, and carbon tetrachloride.

GSH precursors, such as *N*-acetylcysteine and GSH monoethyl ester exert beneficial effects in certain gastrointestinal disorders. These include *Helicobacter pylori* infection, inflammatory bowel disease, and drug-induced peptic ulcer disease. Notably, treatment with *N*-

acetylcysteine plus mesalamine, a first line drug for ulcerative colitis, caused a significantly greater reduction in colonic injury than either agent alone in a rodent model of colitis [34]. This suggests that *N*-acetylcysteine may have added benefits when combined with established therapy in the management of diseases involving oxidative stress and inflammation.



Figure 25-2. *N*-Acetylcysteine as the specific antidote for acetaminophen overdose. As illustrated, GSH-mediated conjugation reaction is a major detoxification mechanism for the toxic metabolite of acetaminophen, *N*-acetyl-*p*-benzoquinoneimine in liver cells. The cytochrome P450 2E1 enzyme is primarily responsible for the bioactivation of acetaminophen to produce the highly reactive *N*-acetyl-*p*-benzoquinoneimine is produced, which overwhelms the normal GSH-dependent detoxification capacity of the liver cells, causing liver injury. By releasing cysteine, *N*-acetylcysteine increases liver GSH levels, leading to enhanced detoxification of *N*-acetyl-*p*-benzoquinoneimine.

3.1.6. Role in Renal Diseases

The most extensively studied renal disorder regarding *N*-acetylcysteine-mediated protection in experimental animals (as well as in human clinical trials) is probably contrast mediuminduced nephropathy [35, 36]. While the exact mechanisms remain to be elucidated, the radical scavenging and anti-inflammatory activities of *N*-acetylcysteine seem to play an important part in the protection against contrast medium-induced nephropathy. Multiple studies also demonstrate a protective effect of *N*-acetylcysteine in renal ischemia-reperfusion injury [37, 38]. Treatment with *N*-acetylcysteine is found to attenuate oxidative stress and inflammation, and improve the graft function after kidney transplantation in experimental animals. *N*acetylcysteine and GSH esters also ameliorate diabetic nephropathy and drug/xenobioticinduced nephrotoxicity [39, 40].

3.1.7. Role in Cancer

Administration of *N*-acetylcysteine protects experimental animals from carcinogenesis induced by a variety of chemical carcinogens [41, 42]. Mechanistically, *N*-acetylcysteine treatment inhibits oxidative DNA damage as well as the formation of carcinogen-DNA adducts during chemical carcinogenesis. The anticancer activities of *N*-acetylcysteine have also been demonstrated in genetic models of spontaneous carcinogenesis. For example, long-term dietary supplementation of *N*-acetylcysteine is shown to reduce the incidence and multiplicity of lymphoma in Atm-deficient mice [43]. In the same animal model, *N*-acetylcysteine treatment inhibits oxidative DNA damage and reduces the frequency of DNA deletions, which may contribute to its cancer chemoprotection [44]. Supplementation of *N*-acetylcysteine also prevents chronic colitis-associated colorectal carcinoma in wild-type mice and hepatocarcinogenesis in transaldolase-deficient mice [45, 46]. Furthermore, it has been reported that *N*-acetylcysteine synergizes with the anticancer drug doxorubicin in the suppression of tumorigenesis and metastasis in murine models [47]. The major mechanisms that have been proposed to account for the anticancer activities of *N*-acetylcysteine in experimental animals are summarized below [48, 49].

- Direct reactions between N-acetylcysteine and carcinogens
- Increased levels of GSH, leading to enhanced detoxification of carcinogens
- Free radical-scavenging activities
- Anti-inflammatory effects
- Inhibition of angiogenesis

3.1.8. Role in Other Diseases and Conditions

N-Acetylcysteine, GSH esters, and other cysteine analogs have been shown to be protective in several other disease processes involving oxidative stress and inflammation in experimental animals. These include sepsis [50], ultraviolet irradiation-induced skin injury [51], wound healing [52], arthritis [53], viral (e.g., HIV, H1N1) infections [54-56], cataract [57, 58], and noise-induced hearing loss [59, 60].

3.2. Human Studies and Clinical Perspectives

3.2.1. N-Acetylcysteine as the Specific Antidote for Acetaminophen Overdose in Humans

Acetaminophen overdose is responsible for more than 40% of all cases of acute liver failure each year in the United States. *N*-Acetylcysteine has been used as the specific antidote for acetaminophen overdose [61]. In acetaminophen overdose, hepatic GSH is depleted by the toxic metabolite of acetaminophen, *N*-acetyl-*p*-benzoquinoneimine, and if left untreated, *N*acetyl-*p*-benzoquinoneimine may accumulate, causing liver cell death (also see Figure 25-2). Timely administration of *N*-acetylcysteine prevents liver injury by replenishing hepatic GSH [62]. Notably, even if administered following the development of acetaminophen-induced fulminate hepatic failure, *N*-acetylcysteine still improves survival rates. Recently, it has been suggested that *N*-acetylcysteine be administered to all patients with acute liver failure regardless of its etiology [61].

3.2.2. Role in Human Cardiovascular Diseases and Diabetes

The potential beneficial effects N-acetylcysteine in human cardiovascular diseases and diabetes have been investigated in multiple small-scale clinical trials. In a randomized trial, N-acetylcysteine as an additive to blood cardioplegia in patients undergoing on-pump coronary artery bypass graft surgery was found to reduce coronary endothelial activation and myocardial oxidative stress [63]. A more recent randomized controlled trial demonstrated that high-dose intravenous N-acetylcysteine resulted in attenuation of oxidative stress, but did not provide an additional clinical benefit to placebo with respect to myocardial reperfusion injury in unselected patients with ST-segment elevation myocardial infarction undergoing primary percutaneous coronary intervention [64]. A meta-analysis reveals that N-acetylcysteine may reduce post-cardiothoracic surgery complications [65]. In a clinical trial with hypertensive patients having type 2 diabetes, long-term administration of N-acetylcysteine and L-arginine was found to reduce endothelial activation and systolic blood pressure [66]. However, this trial did not investigate the effects of N-acetylcysteine alone. A subsequent study by the same group demonstrated that oral supplementation of N-acetylcysteine alone was able to attenuate oxidative stress and endothelial activation after a high-glucose content meal in patients with type 2 diabetes [67]. While these findings suggest the potential protective effects of Nacetylcysteine in human cardiovascular pathophysiology and diabetes, large-scale randomized controlled trials are needed to further ascertain the clinical benefits of N-acetylcysteine-based intervention.

3.2.3. Role in Human Pulmonary Diseases

Clinical trials in patients with chronic obstructive pulmonary disease (COPD) collectively showed that orally administered *N*-acetylcysteine significantly improved responses to steroids, increased general wellbeing, and decreased exacerbation rates and emergency room visits [68]. In addition to *N*-acetylcysteine, *S*-carboxymethylcysteine has been recently recognized as a potentially effective and safe agent for COPD, as evidenced by its ability to reduce the incidence of exacerbation and improve quality of life. In a four-week, dose-escalating phase 1 trial, high-dose oral *N*-acetylcysteine was found to modulate lung inflammation in cystic fibrosis [69]. A subsequent phase 2 study reported that high-dose oral *N*-acetylcysteine didn't alter clinical or inflammatory parameters in patients with cystic fibrosis though it had a potential to increase extracellular GSH levels in the airways [70]. The therapeutic value of *N*-acetylcysteine and other GSH precursors in cystic fibrosis and COPD warrants further evaluation in large-scale well-designed clinical trials.

3.2.4. Role in Contrast Medium-induced Nephropathy

Approximately ten million procedures using radiological contrast agents are performed in the United States annually. The contrast medium-induced nephropathy remains a major clinical problem. The involvement of oxidative and inflammatory stress in contrast mediuminduced nephropathy has prompted studies on using *N*-acetylcysteine as a potential protective agent in human subjects. In 2000, a randomized placebo-controlled trial showed that *N*acetylcysteine was effective as prophylaxis against contrast medium-induced nephropathy [71]. Many subsequent clinical trials have attempted to replicate this report, but the results have been mixed and controversial. The inconsistent outcomes may be a result of the heterogeneity of the patient populations, different inclusion criteria, stage of kidney disease, and the differences in doses and routes of administration of *N*-acetylcysteine [72, 73]. In this context, recent trials in selected patient populations reported beneficial effects of high-dose *N*-acetylcysteine alone or in combination with sodium bicarbonate in contrast medium-induced nephropathy [74-76]. Because of the dichotomous findings from clinical trials, prophylaxis of contrast medium-induced nephropathy with *N*-acetylcysteine is currently not considered standard care. Nevertheless, use of *N*-acetylcysteine has increased due to its demonstrated safety and potential effectiveness in the prevention of contrast medium-induced nephropathy in human subjects.

3.2.5. Role in Other Human Diseases

As an antioxidant and GSH precursor, *N*-acetylcysteine has potential uses in many other disease conditions involving oxidative stress and inflammation. Multiple pilot clinical trials suggest that *N*-acetylcysteine may be effective in the management of nonalcoholic steatohepatitis, kidney transplantation, ulcerative colitis, *Helicobacter pylori* infection, HIV infection, adenomatous colonic polyps, cigarette smoke-induced carcinogenesis, and addictive behaviors [68, 77-80]. Large-scale well-designed clinical trials are needed to further confirm the effectiveness of *N*-acetylcysteine in the intervention of these human diseases.

4. Conclusion and Future Directions

Glutathione precursors, including GSH esters, N-acetylcysteine, and other cysteine analogs can be administered to increase cellular and tissue levels of GSH. In addition to increasing GSH levels, GSH precursors, especially N-acetylcysteine also exert other novel biological effects, such as anti-inflammation, antiplatelet aggregation, and antiangiogenesis. The protective effects of GSH precursors, particularly N-acetylcysteine have been extensively demonstrated in animal models of human diseases, including cardiovascular diseases, diabetes and metabolic syndrome, neurological disorders, pulmonary diseases, liver and gastrointestinal disorders, kidney diseases, and cancer. In line with the findings in animal studies, multiple clinical trials have suggested beneficial effects *N*-acetylcysteine in various disease conditions. However, large-scale well-designed clinical trials are warranted to further establish the efficacy of N-acetylcysteine in human disease intervention. As N-acetylcysteine synergizes with other drugs in disease treatment, future clinical studies should put more emphasis on evaluating the added benefits of N-acetylcysteine in established treatment regimens for common human disease conditions. Future efforts should also focus on the development of novel GSH precursors as well as the preclinical assessment of their efficacy and safety in animal models of human diseases.

5. References

- [1] Anderson ME. Glutathione: an overview of biosynthesis and modulation. *Chem. Biol. Interact.* 1998;111-112:1-14.
- [2] Majano PL, Medina J, Zubia I, Sunyer L, Lara-Pezzi E, Maldonado-Rodriguez A, *et al.* N-Acetyl-cysteine modulates inducible nitric oxide synthase gene expression in human

hepatocytes. J. Hepatol. 2004;40:632-637.

- [3] Chen G, Shi J, Hu Z, Hang C. Inhibitory effect on cerebral inflammatory response following traumatic brain injury in rats: a potential neuroprotective mechanism of N-acetylcysteine. *Mediators Inflamm.* 2008;2008:716458.
- [4] Anfossi G, Russo I, Massucco P, Mattiello L, Cavalot F, Trovati M. N-acetyl-Lcysteine exerts direct anti-aggregating effect on human platelets. *Eur. J. Clin. Invest.* 2001;31:452-461.
- [5] Biswas SK, Rahman I. Environmental toxicity, redox signaling and lung inflammation: the role of glutathione. *Mol. Aspects Med.* 2009;30:60-76.
- [6] Iqbal MP, Mehboobali N, Pervez S. Plaque formation reduction with glutathione monoester in mice fed on atherogenic diet. J. Coll. Physicians. Surg. Pak. 2006;16: 571-575.
- [7] Shimada K, Murayama T, Yokode M, Kita T, Uzui H, Ueda T, *et al.* N-acetylcysteine reduces the severity of atherosclerosis in apolipoprotein E-deficient mice by reducing superoxide production. *Circ. J.* 2009;73:1337-1341.
- [8] Ivanovski O, Szumilak D, Nguyen-Khoa T, Ruellan N, Phan O, Lacour B, et al. The antioxidant N-acetylcysteine prevents accelerated atherosclerosis in uremic apolipoprotein E knockout mice. *Kidney Int.* 2005;67:2288-2294.
- [9] Xia Z, Kuo KH, Nagareddy PR, Wang F, Guo Z, Guo T, *et al.* N-acetylcysteine attenuates PKCbeta2 overexpression and myocardial hypertrophy in streptozotocin-induced diabetic rats. *Cardiovasc. Res.* 2007;73:770-782.
- [10] Xia Z, Guo Z, Nagareddy PR, Yuen V, Yeung E, McNeill JH. Antioxidant N-acetylcysteine restores myocardial Mn-SOD activity and attenuates myocardial dysfunction in diabetic rats. *Eur. J. Pharmacol.* 2006;544:118-125.
- [11] El Midaoui A, Ismael MA, Lu H, Fantus IG, de Champlain J, Couture R. Comparative effects of N-acetyl-L-cysteine and ramipril on arterial hypertension, insulin resistance, and oxidative stress in chronically glucose-fed rats. *Can. J. Physiol. Pharmacol.* 2008; 86:752-760.
- [12] Blouet C, Mariotti F, Azzout-Marniche D, Mathe V, Mikogami T, Tome D, et al. Dietary cysteine alleviates sucrose-induced oxidative stress and insulin resistance. Free Radic. Biol. Med. 2007;42:1089-1097.
- [13] Novelli EL, Santos PP, Assalin HB, Souza G, Rocha K, Ebaid GX, *et al.* N-acetylcysteine in high-sucrose diet-induced obesity: energy expenditure and metabolic shifting for cardiac health. *Pharmacol. Res.* 2009;59:74-79.
- [14] Anderson MF, Nilsson M, Eriksson PS, Sims NR. Glutathione monoethyl ester provides neuroprotection in a rat model of stroke. *Neurosci. Lett.* 2004;354:163-165.
- [15] Anderson MF, Nilsson M, Sims NR. Glutathione monoethylester prevents mitochondrial glutathione depletion during focal cerebral ischemia. *Neurochem. Int.* 2004;44: 153-159.
- [16] Guizar-Sahagun G, Ibarra A, Espitia A, Martinez A, Madrazo I, Franco-Bourland RE. Glutathione monoethyl ester improves functional recovery, enhances neuron survival, and stabilizes spinal cord blood flow after spinal cord injury in rats. *Neuroscience* 2005;130:639-649.
- [17] Park SW, Kim SH, Park KH, Kim SD, Kim JY, Baek SY, et al. Preventive effect of antioxidants in MPTP-induced mouse model of Parkinson's disease. *Neurosci. Lett.* 2004;363:243-246.

- [18] Chen TS, Richie JP, Nagasawa HT, Lang CA. Glutathione monoethyl ester protects against glutathione deficiencies due to aging and acetaminophen in mice. *Mech. Ageing. Dev.* 2000;120:127-139.
- [19] Murali G, Panneerselvam C. Age-associated oxidative macromolecular damages in rat brain regions: role of glutathione monoester. J. Gerontol. A Biol. Sci. Med. Sci. 2007; 62:824-830.
- [20] Murali G, Dhivya S, Rasappan P. Role of Glutathione monoester on age-related neurochemical alterations in rat brain. *Int. J. Dev. Neurosci.* 2009;27:643-648.
- [21] Hooper C, Calvert J. The role for S-carboxymethylcysteine (carbocisteine) in the management of chronic obstructive pulmonary disease. Int. J. Chron. Obstruct. Pulmon. Dis. 2008;3:659-669.
- [22] Sadowska AM, Manuel YKB, De Backer WA. Antioxidant and anti-inflammatory efficacy of NAC in the treatment of COPD: discordant in vitro and in vivo dose-effects: a review. *Pulm. Pharmacol. Ther.* 2007;20:9-22.
- [23] Kloek J, Van Ark I, De Clerck F, Bloksma N, Nijkamp FP, Folkerts G. Modulation of airway hyperresponsiveness by thiols in a murine in vivo model of allergic asthma. *Inflamm. Res.* 2003;52:126-131.
- [24] Kloek J, Mortaz E, van Ark I, Lilly CM, Nijkamp FP, Folkerts G. Glutathione prevents the early asthmatic reaction and airway hyperresponsiveness in guinea pigs. J. Physiol. Pharmacol .2010;61:67-72.
- [25] Langley SC, Kelly FJ. N-acetylcysteine ameliorates hyperoxic lung injury in the preterm guinea pig. *Biochem. Pharmacol.* 1993;45:841-846.
- [26] Singhal RK, Jain A. Glutathione ethyl ester supplementation prevents mortality in newborn rats exposed to hyperoxia. *Biol. Neonate* 2000;77:261-266.
- [27] Mata M, Ruiz A, Cerda M, Martinez-Losa M, Cortijo J, Santangelo F, *et al.* Oral N-acetylcysteine reduces bleomycin-induced lung damage and mucin Muc5ac expression in rats. *Eur. Respir. J.* 2003;22:900-905.
- [28] Yildirim Z, Kotuk M, Iraz M, Kuku I, Ulu R, Armutcu F, et al. Attenuation of bleomycin-induced lung fibrosis by oral sulfhydryl containing antioxidants in rats: erdosteine and N-acetylcysteine. *Pulm. Pharmacol. Ther.* 2005;18:367-373.
- [29] Gon Y, Hashimoto S, Nakayama T, Matsumoto K, Koura T, Takeshita I, *et al.* N-acetyl-L-cysteine inhibits bleomycin-induced interleukin-8 secretion by bronchial epithelial cells. *Respirology* 2000;5:309-313.
- [30] Grattagliano I, Vendemiale G, Lauterburg BH. Reperfusion injury of the liver: role of mitochondria and protection by glutathione ester. J. Surg. Res. 1999;86:2-8.
- [31] Fusai G, Glantzounis GK, Hafez T, Yang W, Quaglia A, Sheth H, *et al.* N-Acetylcysteine ameliorates the late phase of liver ischaemia/reperfusion injury in the rabbit with hepatic steatosis. *Clin. Sci.* (Lond) 2005;109:465-473.
- [32] Thong-Ngam D, Samuhasaneeto S, Kulaputana O, Klaikeaw N. N-acetylcysteine attenuates oxidative stress and liver pathology in rats with non-alcoholic steatohepatitis. *World J. Gastroenterol.* 2007;13:5127-5132.
- [33] Caballero F, Fernandez A, Matias N, Martinez L, Fucho R, Elena M, et al. Specific contribution of methionine and choline in nutritional nonalcoholic steatohepatitis: impact on mitochondrial S-adenosyl-L-methionine and glutathione. J. Biol. Chem. 2010;285:18528-18536.
- [34] Siddiqui A, Ancha H, Tedesco D, Lightfoot S, Stewart CA, Harty RF. Antioxidant

therapy with N-acetylcysteine plus mesalamine accelerates mucosal healing in a rodent model of colitis. *Dig. Dis. Sci.* 2006;51:698-705.

- [35] Sochman J, Peregrin JH, Burgelova M, Kopkan L, Kramer HJ, Cervenka L. Nacetylcysteine attenuates iodine contrast agent-induced nephropathy in 5/6-nephrectomized rats. *Kidney Blood Press. Res.* 2010;33:149-156.
- [36] Cronin RE. Contrast-induced nephropathy: pathogenesis and prevention. *Pediatr. Nephrol.* 2010;25:191-204.
- [37] Lin A, Sekhon C, Sekhon B, Smith A, Chavin K, Orak J, et al. Attenuation of ischemia-reperfusion injury in a canine model of autologous renal transplantation. *Transplantation* 2004;78:654-659.
- [38] Nitescu N, Grimberg E, Ricksten SE, Guron G. Effects of N-acetyl-L-cysteine on renal haemodynamics and function in early ischaemia-reperfusion injury in rats. *Clin. Exp. Pharmacol. Physiol.* 2006;33:53-57.
- [39] Babu E, Gopalakrishnan VK, Sriganth IN, Gopalakrishnan R, Sakthisekaran D. Cisplatin induced nephrotoxicity and the modulating effect of glutathione ester. *Mol. Cell Biochem.* 1995;144:7-11.
- [40] Duru M, Nacar A, Yonden Z, Kuvandik G, Helvaci MR, Koc A, et al. Protective effects of N-acetylcysteine on cyclosporine-A-induced nephrotoxicity. *Ren. Fail.* 2008; 30:453-459.
- [41] Nishikawa-Ogawa M, Wanibuchi H, Morimura K, Kinoshita A, Nishikawa T, Hayashi S, et al. N-acetylcysteine and S-methylcysteine inhibit MeIQx rat hepatocarcinogenesis in the post-initiation stage. Carcinogenesis 2006;27:982-988.
- [42] Balansky R, Ganchev G, Iltcheva M, Steele VE, De Flora S. Prenatal N-acetylcysteine prevents cigarette smoke-induced lung cancer in neonatal mice. *Carcinogenesis* 2009; 30:1398-1401.
- [43] Reliene R, Schiestl RH. Antioxidant N-acetyl cysteine reduces incidence and multiplicity of lymphoma in Atm deficient mice. *DNA Repair* (Amst) 2006;5:852-859.
- [44] Reliene R, Fischer E, Schiestl RH. Effect of N-acetyl cysteine on oxidative DNA damage and the frequency of DNA deletions in atm-deficient mice. *Cancer Res.* 2004; 64:5148-5153.
- [45] Gommeaux J, Cano C, Garcia S, Gironella M, Pietri S, Culcasi M, et al. Colitis and colitis-associated cancer are exacerbated in mice deficient for tumor protein 53-induced nuclear protein 1. Mol. Cell Biol. 2007;27:2215-2228.
- [46] Hanczko R, Fernandez DR, Doherty E, Qian Y, Vas G, Niland B, et al. Prevention of hepatocarcinogenesis and increased susceptibility to acetaminophen-induced liver failure in transaldolase-deficient mice by N-acetylcysteine. J. Clin. Invest. 2009;119: 1546-1557.
- [47] De Flora S, D'Agostini F, Masiello L, Giunciuglio D, Albini A. Synergism between Nacetylcysteine and doxorubicin in the prevention of tumorigenicity and metastasis in murine models. *Int. J. Cancer* 1996;67:842-848.
- [48] Agarwal A, Munoz-Najar U, Klueh U, Shih SC, Claffey KP. N-acetyl-cysteine promotes angiostatin production and vascular collapse in an orthotopic model of breast cancer. *Am. J. Pathol.* 2004;164:1683-1696.
- [49] Ma Q, Cavallin LE, Yan B, Zhu S, Duran EM, Wang H, et al. Antitumorigenesis of antioxidants in a transgenic Rac1 model of Kaposi's sarcoma. Proc. Natl. Acad. Sci. USA 2009;106:8683-8688.

- [50] Hsu BG, Lee RP, Yang FL, Harn HJ, Chen HI. Post-treatment with N-acetylcysteine ameliorates endotoxin shock-induced organ damage in conscious rats. *Life Sci.* 2006; 79:2010-2016.
- [51] Goodson AG, Cotter MA, Cassidy P, Wade M, Florell SR, Liu T, et al. Use of oral N-acetylcysteine for protection of melanocytic nevi against UV-induced oxidative stress: towards a novel paradigm for melanoma chemoprevention. *Clin. Cancer Res.* 2009; 15:7434-7440.
- [52] Aktunc E, Ozacmak VH, Ozacmak HS, Barut F, Buyukates M, Kandemir O, *et al.* N-acetyl cysteine promotes angiogenesis and clearance of free oxygen radicals, thus improving wound healing in an alloxan-induced diabetic mouse model of incisional wound. *Clin. Exp. Dermatol.* 2010;35:902-909.
- [53] Nakagawa S, Arai Y, Mazda O, Kishida T, Takahashi KA, Sakao K, et al. Nacetylcysteine prevents nitric oxide-induced chondrocyte apoptosis and cartilage degeneration in an experimental model of osteoarthritis. J. Orthop. Res. 2010;28:156-163.
- [54] Fraternale A, Paoletti MF, Casabianca A, Orlandi C, Schiavano GF, Chiarantini L, et al. Inhibition of murine AIDS by pro-glutathione (GSH) molecules. Antiviral Res. 2008;77:120-127.
- [55] Lai KY, Ng WY, Osburga Chan PK, Wong KF, Cheng F. High-dose N-acetylcysteine therapy for novel H1N1 influenza pneumonia. *Ann. Intern. Med.* 2010;152:687-688.
- [56] Fraternale A, Paoletti MF, Casabianca A, Nencioni L, Garaci E, Palamara AT, *et al.* GSH and analogs in antiviral therapy. *Mol. Aspects Med.* 2009;30:99-110.
- [57] Kobayashi S, Kasuya M, Shimizu K, Ishii Y, Takehana M, Sakai K, *et al.* Glutathione isopropyl ester (YM737) inhibits the progression of X-ray-induced cataract in rats. *Curr. Eye Res.* 1993;12:115-122.
- [58] Aydin B, Yagci R, Yilmaz FM, Erdurmus M, Karadag R, Keskin U, *et al.* Prevention of selenite-induced cataractogenesis by N-acetylcysteine in rats. *Curr. Eye Res.* 2009; 34:196-201.
- [59] Ohinata Y, Yamasoba T, Schacht J, Miller JM. Glutathione limits noise-induced hearing loss. *Hear Res.* 2000;146:28-34.
- [60] Wu HP, Hsu CJ, Cheng TJ, Guo YL. N-acetylcysteine attenuates noise-induced permanent hearing loss in diabetic rats. *Hear Res.* 2010;267:71-77.
- [61] Stravitz RT, Kramer DJ. Management of acute liver failure. *Nat. Rev. Gastroenterol. Hepatol.* 2009;6:542-553.
- [62] Heard KJ. Acetylcysteine for acetaminophen poisoning. N. Engl. J. Med. 2008;359: 285-292.
- [63] Rodrigues AJ, Evora PR, Bassetto S, Alves L, Jr., Scorzoni Filho A, Origuela EA, *et al.* Blood cardioplegia with N-acetylcysteine may reduce coronary endothelial activation and myocardial oxidative stress. *Heart Surg. Forum* 2009;12:E44-48.
- [64] Thiele H, Hildebrand L, Schirdewahn C, Eitel I, Adams V, Fuernau G, et al. Impact of high-dose N-acetylcysteine versus placebo on contrast-induced nephropathy and myocardial reperfusion injury in unselected patients with ST-segment elevation myocardial infarction undergoing primary percutaneous coronary intervention. The LIPSIA-N-ACC (Prospective, Single-Blind, Placebo-Controlled, Randomized Leipzig Immediate PercutaneouS Coronary Intervention Acute Myocardial Infarction N-ACC) Trial. J. Am. Coll. Cardiol. 2010;55:2201-2209.

- [65] Baker WL, Anglade MW, Baker EL, White CM, Kluger J, Coleman CI. Use of Nacetylcysteine to reduce post-cardiothoracic surgery complications: a meta-analysis. *Eur. J. Cardiothorac. Surg.* 2009;35:521-527.
- [66] Martina V, Masha A, Gigliardi VR, Brocato L, Manzato E, Berchio A, *et al.* Long-term N-acetylcysteine and L-arginine administration reduces endothelial activation and systolic blood pressure in hypertensive patients with type 2 *diabetes*. *Diabetes Care* 2008;31:940-944.
- [67] Masha A, Brocato L, Dinatale S, Mascia C, Biasi F, Martina V. N-acetylcysteine is able to reduce the oxidation status and the endothelial activation after a high-glucose content meal in patients with Type 2 diabetes mellitus. J. Endocrinol. Invest. 2009; 32:352-356.
- [68] Atkuri KR, Mantovani JJ, Herzenberg LA. N-Acetylcysteine: a safe antidote for cysteine/glutathione deficiency. *Curr. Opin. Pharmacol.* 2007;7:355-359.
- [69] Tirouvanziam R, Conrad CK, Bottiglieri T, Herzenberg LA, Moss RB. High-dose oral N-acetylcysteine, a glutathione prodrug, modulates inflammation in cystic fibrosis. *Proc. Natl. Acad. Sci. USA* 2006;103:4628-4633.
- [70] Dauletbaev N, Fischer P, Aulbach B, Gross J, Kusche W, Thyroff-Friesinger U, et al. A phase II study on safety and efficacy of high-dose N-acetylcysteine in patients with cystic fibrosis. Eur. J. Med. Res. 2009;14:352-358.
- [71] Tepel M, van der Giet M, Schwarzfeld C, Laufer U, Liermann D, Zidek W. Prevention of radiographic-contrast-agent-induced reductions in renal function by acetylcysteine. *N. Engl. J. Med.* 2000;343:180-184.
- [72] Stenstrom DA, Muldoon LL, Armijo-Medina H, Watnick S, Doolittle ND, Kaufman JA, *et al.* N-acetylcysteine use to prevent contrast medium-induced nephropathy: premature phase III trials. *J. Vasc. Interv. Radiol.* 2008;19:309-318.
- [73] Liu R, Nair D, Ix J, Moore DH, Bent S. N-acetylcysteine for the prevention of contrastinduced nephropathy: a systematic review and meta-analysis. J. Gen. Intern. Med. 2005;20:193-200.
- [74] Marenzi G, Assanelli E, Marana I, Lauri G, Campodonico J, Grazi M, *et al.* N-acetylcysteine and contrast-induced nephropathy in primary angioplasty. *N. Engl. J. Med.* 2006;354:2773-2782.
- [75] Jo SH, Koo BK, Park JS, Kang HJ, Kim YJ, Kim HL, et al. N-acetylcysteine versus AScorbic acid for preventing contrast-Induced nephropathy in patients with renal insufficiency undergoing coronary angiography NASPI study: a prospective randomized controlled trial. Am. Heart. J. 2009;157:576-583.
- [76] Brown JR, Block CA, Malenka DJ, O'Connor GT, Schoolwerth AC, Thompson CA. Sodium bicarbonate plus N-acetylcysteine prophylaxis: a meta-analysis. JACC Cardiovasc. Interv. 2009;2:1116-1124.
- [77] Guijarro LG, Mate J, Gisbert JP, Perez-Calle JL, Marin-Jimenez I, Arriaza E, et al. N-acetyl-L-cysteine combined with mesalamine in the treatment of ulcerative colitis: randomized, placebo-controlled pilot study. World J. Gastroenterol. 2008;14:2851-2857.
- [78] Dodd S, Dean O, Copolov DL, Malhi GS, Berk M. N-acetylcysteine for antioxidant therapy: pharmacology and clinical utility. *Expert. Opin. Biol. Ther.* 2008;8:1955-1962.
- [79] Millea PJ. N-acetylcysteine: multiple clinical applications. *Am. Fam. Physician* 2009; 80:265-269.

[80] de Oliveira CP, Stefano JT, de Siqueira ER, Silva LS, de Campos Mazo DF, Lima VM, *et al.* Combination of N-acetylcysteine and metformin improves histological steatosis and fibrosis in patients with non-alcoholic steatohepatitis. *Hepatol. Res.* 2008;38:159-165.

Chapter 26

Spin Traps

Abstract

Spin traps are chemicals that react with unstable free radical species to form relatively stable free radical spin adducts that can be detected by electron paramagnetic resonance (EPR) spectroscopy. Nitrone compounds are the most commonly used spin traps with *a*-phenyl-*tert*-butylnitrone (PBN) as a notable example. Spin traps possess antioxidant activities due to their ability to scavenge free radicals. In addition, they also exert other novel biological activities, such as induction of cellular antioxidant enzymes, antiinflammation, and antiangiogenesis. Animal studies have demonstrated a protective role for spin traps, especially PBN and its derivatives in the pathophysiology of a number of diseases. However, pooled analysis of clinical trials on the PBN derivative 2,4disulfophenyl-*N-tert*-butylnitrone (also known as NXY-059) has concluded that this nitrone compound is ineffective for the treatment of acute ischemic stroke in human subjects though its efficacy in treating this neurological disorder is well-established in preclinical studies.

1. Overview

Spin traps are used to react with the short-lived free radicals to form relatively long-lived radical adducts that can be detected by electron paramagnetic resonance (EPR) techniques. EPR spin trapping has been considered the most specific way of detecting free radical species formed in biological systems. Some of the commonly used spin traps have been found to also exert antioxidant effects in both in vitro and in vivo systems. Notable examples include the nitrone compounds 5,5-dimethyl-1-pyrroline *N*-oxide (DMPO), α -phenyl-*tert*-butylnitrone (PBN), and PBN derivatives. These compounds have been demonstrated to protect against various pathophysiological conditions involving oxidative stress. As discussed below, in addition to directly scavenging free radicals, some spin traps are shown to upregulate cellular antioxidant enzymes and exert other novel biological effects. This chapter begins with a brief discussion of the chemical and biochemical properties of the antioxidant spin traps followed by a summary of the major findings with respect to the role of these compounds in disease intervention in both animal models and human clinical trials.

2. Chemistry and Biochemistry

2.1. Nitrone Spin Traps

Nitrones have been the most widely used spin traps for detecting free radicals, including oxygen radicals and carbon-centered radical species in biological systems. There are two classes of nitrone spin traps: the linear and the pyrroline-based cyclic nitrones. The linear nitrone spin traps include PBN (structure shown in Figure 26-1) and α -(4-pyridyl-1-oxide)-*N*-*tert*-butylnitrone (POBN). The pyrroline-based cyclic nitrone spin traps consist of DMPO, 5-diethoxyphosphoryl-5-methyl-1-pyrroline *N*-oxide (DEPMPO), and 5-ethoxycarbonyl-5-methyl-1-pyrroline *N*-oxide (EMPO) [1]. These spin traps differ with respect to their affinity to free radicals and the stability of the radical adducts formed. In addition, the toxicity also varies with different spin traps. In this context, PBN and several PBN derivatives have been shown to be well-tolerated.



Figure 26-1. Structure of α -phenyl-*tert*-butylnitrone (PBN). See text (Section 2.1) for description of PBN.

The nitrone chemical structure in its simplest form can be written as X-CH=NO-Y. The nitrone trapping of a free radical intermediate is represented in the following simple reaction in which the free radical intermediate (R[•]) is trapped by the nitrone to form a nitroxyl free radical spin adduct (X-CHR-NO[•]-Y) (Reaction 1) [2].

$$X-CH=NO-Y+R^{\cdot} \longrightarrow X-CHR-NO^{\cdot}-Y$$
(1)

The nitroxyl free radical spin adduct is usually much more stable than the original free radical intermediate (R⁻), therefore making it possible in principle to detect and characterize the original free radical intermediate using EPR spectroscopy. Likewise, the ability of nitrones to trap free radical intermediates to form relatively less reactive spin adducts makes it rational to use these compounds to protect against oxidative stress injury in biological systems. Indeed, the nitrone spin trap PBN has been well-recognized for its protective effects in oxidative stress-associated pathophysiology. Several PBN derivatives, including 2,4-disulfophenyl-*N-tert*-butylnitrone (also known as NXY-059; see Sections 3.1.3, 3.1.7, and 3.2 for more discussion of NXY-059) have been developed over the last two decades, which are shown to also possess potent antioxidant activities in biological systems. In addition to their well-recognized free radical-scavenging properties, nitrone spin traps have been demonstrated to possess other novel activities, such as induction of endogenous antioxidant enzymes, anti-

inflammation, and antiangiogenesis [3-5]. These novel effects may also contribute to nitrone spin trap-mediated disease protection.

2.2. Other Spin Traps

In addition to nitrones, nitroso compounds have also been used as spin traps, but the application of nitroso compounds as spin traps and potential antioxidant agents is limited by their instability and toxicity [1]. Iron-dithiocarbamate complexes are employed as spin traps for detecting and characterizing nitric oxide in biological systems [6]. The disease protective effects of the nitric oxide spin traps in animal models have recently been reported. Section 3 below summarizes the key findings regarding the beneficial effects of spin traps in disease pathophysiology with an emphasis on PBN and PBN derivatives.

3. Role in Disease Intervention

3.1. Animal Studies

3.1.1. Role in Cardiovascular Diseases

Spin traps, such as DMPO, PBN, and iron-dithiocarbamate complexes have been frequently utilized to study free radical formation in cardiovascular diseases. Consequently, these spin traps have been shown to protect against cardiovascular pathophysiology involving oxidative stress and inflammation. For example, DMPO, PBN, and PBN derivatives are able to mitigate myocardial ischemia-reperfusion injury in animal models [7-9]. It is further found that DMPO protects against myocardial ischemia-reperfusion injury through scavenging oxygen free radicals and preservation of mitochondrial electron transport chain [10]. PBN has been demonstrated to be beneficial in cardiac allograft transplantation via abrogation of inflammatory cytokine gene expression during alloimmune activation in rats [11]. Similarly, the nitric oxide spin trap iron-diethyldithiocarbamate complex is reported to attenuate inflammatory responses, prolong graft survival, and decrease histological rejection scores in a rat model of acute cardiac allograft rejection [12]. The protective effects of nitrone spin traps have also been observed in other cardiovascular disorders, including post-ischemic cardiac arrhythmias and hypertension in experimental animals [13, 14].

3.1.2. Role in Diabetes

There have been multiple studies on the potential effects of PBN in chemically-induced type 1 diabetes in animal models. An early study demonstrated that administration of PBN protects against streptozotocin-induced diabetes in mice, as evidenced by inhibition of hyper-glycemia, hemoglobin glycation, nitric oxide formation, and pancreatic β -cell destruction [15]. These initial observations have been confirmed in a subsequent study, which further shows a protective role for PBN in alloxan-induced diabetes. PBN treatment attenuates both alloxan- and streptozotocin-induced free radical formation and activation of NF- κ B in pancreatic tissue [16]. However, treatment with PBN fails to prevent spontaneous type 1 diabetes associated with autoimmunity in rats though lipid peroxidation in pancreatic tissue is inhi-

bited by the spin trap [17]. To date, the effects of nitrone spin traps on type 2 diabetes have not been reported in the literature.

3.1.3. Role in Neurological Diseases

Over the last two decades the neuroprotective effects of nitrone spin traps, especially PBN and its derivatives have been extensively investigated in various animal models of neurological disorders. The most studied neurological disease regarding the protection by nitrone spin traps in experimental animals is probably acute ischemic stroke [2, 18]. The beneficial effects of PBN and its derivative 2,4-disulfophenyl-*N-tert*-butylnitrone (NXY-059) in acute ischemic stroke have been demonstrated in both rodents and primates, which led to the initiation of clinical trials on NXY-059 for the treatment of acute ischemic stroke in humans [2, 18] (see Section 3.2 below). Although the exact mechanisms underlying nitrone spin trapmediated stroke protection remain to be elucidated, attenuation of free radical damage and inflammation, and improvement of energy metabolism may play an important part. As oxidative stress and inflammation are common mechanisms of neurodegenerative disorders, including Parkinson's disease. Chemically-induced parkinsonism in mice is ameliorated by treatment with PBN [19]. PBN treatment also protects against toxin-induced seizure, excitotoxicity, and neuroinflammatory disorders in experimental animals [20-22].

3.1.4. Role in Pulmonary Diseases

The herbicide paraquat causes lung injury through an oxidative stress mechanism. Indeed, formation of oxygen free radicals has been identified by EPR spectroscopy with PBNspin trapping in experimental animals following exposure to paraquat. Paraquat-induced acute broncoconstriction and edema in guinea pig lungs are markedly attenuated by PBN treatment [23]. Recently, it is reported that administration of PBN to neonatal rats protects these animals from hyperoxia-induced lung injury. It has been further observed that the hyperoxiainduced superoxide formation from NAD(P)H oxidase and gene expression of proinflammatory cytokines in lung tissue are inhibited by PBN treatment. This suggests that suppression of inflammation may play a part in PBN-mediated pulmonary protection against hyperoxic injury [24]. In line with this notion, PBN is shown to inhibit hyperoxia-induced formation of nitric oxide, an important event in hyperoxia-elicited inflammatory injury [25].

3.1.5. Role in Hepatic and Gastrointestinal Diseases

The protective effects of nitrone spin traps on chemically-induced liver disorders have been demonstrated in rat models, including carbon tetrachloride-induced hepatotoxicity and copper-induced fulminant hepatitis with jaundice [26, 27]. Recently, a new PBN derivative *N*-{[4-(lactobionamido)methyl]benzylidene}-1,1-dimethyl-2-(octylsulfanyl)ethylamine *N*-oxide is shown to suppress copper-induced fulminant hepatitis and oxidative stress at an effective dose 5-10 orders of magnitudes lower than that of PBN [28].

Treatment with PBN improves dextran sulfate sodium-induced colitis in mice, as evidenced by attenuated colonic inflammation, oxidative stress, and tissue necrosis. Notably, activation of NF-kB and NF-kB-regulated proinflammatory cytokine gene expression is diminished by PBN treatment in this murine model of colitis [29]. More recently, PBN treatment has been reported to reduce visceral pain associated with zymosan-induced colitis and colonic oxidative stress in rats, further supporting a beneficial effect of PBN in experimental colitis [30]. In another murine model of colitis elicited by trinitrobenzene sulfonic acid, combination of PBN with mesalamine, a first line drug for treating ulcerative colitis, failed to improve the disease condition beyond that produced by mesalamine alone. However, the study did not report the effects of PBN alone on trinitrobenzene sulfonic acid-induced colitis [31].

3.1.6. Role in Renal Diseases

The nitrone spin trap PBN exerts beneficial effects in renal diseases, especially ischemiareperfusion injury in animal models. Early studies reported that administration of PBN reduces ischemia-reperfusion-induced acute renal failure in rats, as revealed by improvement in creatinine and urea clearance [32]. Subsequently, studies have demonstrated that PBN treatment causes suprainduction of heme oxygenase-1 in renal tissue, which may mediate, at least partially, PBN's protection against ischemia-reperfusion injury [3]. In this context, transgenic overexpression of HO-1 is highly protective against renal ischemia-reperfusion injury (see Chapter 15 for discussion of heme oxygenase).

3.1.7. Role in Cancer

The anticancer activities of PBN were first observed by D. Nakae and coworkers in a rat model of hepatocarcinogenesis induced by a diet deficient in choline [33, 34]. In this model, chronic administration of PBN is found to inhibit the development of hepatocellular carcinoma. PBN treatment also inhibits the formation of oxidative DNA damage as well as the activity of cyclooxygenase-2 in the target tissue [33]. In addition, PBN is shown to selectively induce apoptosis in preneoplastic lesions during the early phase of hepatocarcinogenesis induced by the choline-deficient diet [35]. Notably, the metabolite of PBN, namely, 4-hydroxyl-PBN, also possesses anticancer activities [35].

Recently, PBN and its derivatives have been demonstrated to exert anticancer effects in other animal models of cancer. For example, administration of PBN leads to marked induction of tumor regression, inhibition of angiogenesis, and prolongation of survival in a C6 rat glioma model [5]. The PBN derivative 2,4-disulfophenyl-*N-tert*-butylnitrone (also known as OKN007 or NXY-059) also decreases tumor volumes and increases survival in the C6 rat glioma model [36]. More recently, PBN is shown to be protective in a murine model of colorectal carcinoma [37].

3.1.8. Role in Other Diseases and Conditions

There are a number of other diseases and conditions in which nitrone spin traps have been shown to exert beneficial effects in experimental animals. These include sepsis [38], ionizing radiation injury [39], light-induced retinal degeneration [40], acute acoustical trauma [41], and aging [42].

3.2. Human Studies and Clinical Perspectives

The enormous amount of data obtained from preclinical studies on nitrone spin trapmediated disease protection has spurred the interest in utilizing these compounds for the intervention of human diseases. Indeed, as noted earlier in Section 3.1.3, the PBN derivative NXY-059 was developed as a potential neuroprotectant for human stroke. Supported by the available preclinical data on its efficacy in treating stroke as well as the safety profiles in human subjects, a randomized double-blind, placebo-controlled trial (SAINT I) was initiated in 2003, and the SAINT I trial involving 1722 patients with acute ischemic stroke reported in 2006 that administration of NXY-059 within 6 hours after the onset of acute ischemic stroke significantly improved the primary outcome (reduced disability at 90 days) [43]. However, a subsequent larger trial (SAINT II) involving 3306 patients with acute ischemic stroke reported in 2007 that NXY-059 was ineffective for the treatment of acute ischemic stroke given within 6 hours after the onset of the symptoms [44]. Furthermore, a pooled analysis of the SAINT I and II trials concluded that NXY-059 lacks efficacy in the treatment of acute ischemic stroke ischemic stroke [45]. As such, NXY-509 was not further developed as a drug for human ischemic stroke.

4. Conclusion and Future Directions

In addition to their free radical-scavenging activities, spin traps, especially the nitrone spin traps also possess other important biological properties, including upregulation of endogenous antioxidant enzymes, downregulation of inflammatory responses, and inhibition of angiogenesis (Figure 26-2).



Figure 26-2. Biological activities of spin traps in disease intervention. As illustrated, in addition to their free radical-scavenging ability, spin traps also possess other novel biological activities, including induction of endogenous antioxidant enzymes, anti-inflammation, and antiangiogenesis. These diverse mechanistic events contribute to the biological role of spin traps in disease intervention.

The beneficial effects of spin traps, such as PBN and the PBN derivative NXY-059 have been demonstrated in many animal models of human diseases, including both rodent and primate models of acute ischemic stroke. However, pooled analysis of the SAINT trials concluded that NXY-059 lacks effectiveness in treating human acute ischemic stroke. There have been many discussions and critiques of the SAINT trials as well as the preclinical studies on NXY-059, which provide suggestions for future studies on neuroprotective agents. These include (1) better characterization of drug candidate-target interaction and its relationship to pharmacodynamic treatment endpoints, (2) combination therapies with different compounds targeting different pathways of disease pathophysiology to achieve a better chance of success than single medications, and (3) testing candidate drugs in more carefully selected patient groups and better designed trials. In view of the demonstrated safety of NXY-059 in human subjects, this nitrone spin trap should be further investigated for its potential benefits in the intervention of other human diseases in well-designed clinical trials.

5. References

- [1] Villamena FA, Zweier JL. Detection of reactive oxygen and nitrogen species by EPR spin trapping. *Antioxid. Redox Signal* 2004;6:619-629.
- [2] Floyd RA, Kopke RD, Choi CH, Foster SB, Doblas S, Towner RA. Nitrones as therapeutics. *Free Radic. Biol. Med.* 2008;45:1361-1374.
- [3] Maines MD, Raju VS, Panahian N. Spin trap (N-t-butyl-alpha-phenylnitrone)-mediated suprainduction of heme oxygenase-1 in kidney ischemia/reperfusion model: role of the oxygenase in protection against oxidative injury. *J. Pharmacol. Exp. Ther.* 1999;291: 911-919.
- [4] Sang H, Wallis GL, Stewart CA, Kotake Y. Expression of cytokines and activation of transcription factors in lipopolysaccharide-administered rats and their inhibition by phenyl N-tert-butylnitrone (PBN). Arch. Biochem. Biophys. 1999;363:341-348.
- [5] Doblas S, Saunders D, Kshirsagar P, Pye Q, Oblander J, Gordon B, *et al.* Phenyl-tertbutylnitrone induces tumor regression and decreases angiogenesis in a C6 rat glioma model. *Free Radic. Biol. Med.* 2008;44:63-72.
- [6] Hogg N. Detection of nitric oxide by electron paramagnetic resonance spectroscopy. *Free Radic. Biol. Med.* 2010;49:122-129.
- [7] Tosaki A, Blasig IE, Pali T, Ebert B. Heart protection and radical trapping by DMPO during reperfusion in isolated working rat hearts. *Free Radic. Biol. Med.* 1990;8:363-372.
- [8] Bolli R, Jeroudi MO, Patel BS, DuBose CM, Lai EK, Roberts R, *et al.* Direct evidence that oxygen-derived free radicals contribute to postischemic myocardial dysfunction in the intact dog. *Proc. Natl. Acad. Sci. USA* 1989;86:4695-4699.
- [9] Tanguy S, Durand G, Reboul C, Polidori A, Pucci B, Dauzat M, *et al.* Protection against reactive oxygen species injuries in rat isolated perfused hearts: effect of LPBNAH, a new amphiphilic spin-trap derived from PBN. *Cardiovasc. Drugs Ther.* 2006;20:147-149.
- [10] Zuo L, Chen YR, Reyes LA, Lee HL, Chen CL, Villamena FA, et al. The radical trap 5,5-dimethyl-1-pyrroline N-oxide exerts dose-dependent protection against myocardial ischemia-reperfusion injury through preservation of mitochondrial electron transport. J. Pharmacol. Exp. Ther. 2009;329:515-523.
- [11] Pieper GM, Nilakantan V, Zhou X, Khanna AK, Johnson CP, Roza AM, et al.

Treatment with alpha-phenyl-N-tert-butylnitrone, a free radical-trapping agent, abrogates inflammatory cytokine gene expression during alloimmune activation in rat cardiac allografts. *J. Pharmacol. Exp. Ther.* 2005;312:774-779.

- [12] Pieper GM, Nilakantan V, Hilton G, Halligan NL, Felix CC, Kampalath B, et al. Mechanisms of the protective action of diethyldithiocarbamate-iron complex on acute cardiac allograft rejection. Am. J. Physiol. Heart Circ. Physiol. 2003;284:H1542-1551.
- [13] Hearse DJ, Tosaki A. Free radicals and reperfusion-induced arrhythmias: protection by spin trap agent PBN in the rat heart. *Circ. Res.* 1987;60:375-383.
- [14] Saito K, Kobayashi C, Ikeda M. Effect of radical scavenger N-tert-butyl-alphaphenylnitrone on stroke in a rat model using a telemetric system. J. Pharm. Pharm. Sci. 2008;11:25-31.
- [15] Tabatabaie T, Kotake Y, Wallis G, Jacob JM, Floyd RA. Spin trapping agent phenyl Ntert-butylnitrone protects against the onset of drug-induced insulin-dependent diabetes mellitus. *FEBS Lett.* 1997;407:148-152.
- [16] Ho E, Chen G, Bray TM. Alpha-phenyl-tert-butylnitrone (PBN) inhibits NFkappaB activation offering protection against chemically induced diabetes. *Free Radic. Biol. Med.* 2000;28:604-614.
- [17] Iovino G, Kubow S, Marliss EB. Effect of alpha-phenyl-N-tert-butylnitrone on diabetes and lipid peroxidation in BB rats. *Can. J. Physiol. Pharmacol.* 1999;77:166-174.
- [18] Green AR, Ashwood T, Odergren T, Jackson DM. Nitrones as neuroprotective agents in cerebral ischemia, with particular reference to NXY-059. *Pharmacol. Ther.* 2003; 100:195-214.
- [19] Fredriksson A, Eriksson P, Archer T. MPTP-induced deficits in motor activity: neuroprotective effects of the spintrapping agent, alpha-phenyl-tert-butyl-nitrone (PBN). J. Neural. Transm. 1997;104:579-592.
- [20] Zivin M, Milatovic D, Dettbarn WD. Nitrone spin trapping compound N-tert-butylalpha-phenylnitrone prevents seizures induced by anticholinesterases. *Brain Res.* 1999; 850:63-72.
- [21] Milatovic D, Zivin M, Hustedt E, Dettbarn WD. Spin trapping agent phenyl-N-tertbutylnitrone prevents diisopropylphosphorofluoridate-induced excitotoxicity in skeletal muscle of the rat. *Neurosci. Lett.* 2000;278:25-28.
- [22] Fan LW, Chen RF, Mitchell HJ, Lin RC, Simpson KL, Rhodes PG, et al. alpha-Phenyln-tert-butyl-nitrone attenuates lipopolysaccharide-induced brain injury and improves neurological reflexes and early sensorimotor behavioral performance in juvenile rats. J. Neurosci. Res. 2008;86:3536-3547.
- [23] Sata T, Kubota E, Misra HP, Mojarad M, Pakbaz H, Said SI. Paraquat-induced lung injury: prevention by N-tert-butyl-alpha-phenylnitrone, a free-radical spin-trapping agent. *Am. J. Physiol.* 1992;262:L147-152.
- [24] Chang YS, Kim YJ, Yoo HS, Sung DK, Kim SY, Kang S, et al. Alpha-phenyl-N-tertbutylnitrone attenuates hyperoxia-induced lung injury by down-modulating inflammation in neonatal rats. Exp. Lung Res. 2009;35:234-249.
- [25] Saito K, Yoshioka H. Protective effect of spin trap agent, N-tert-butyl-alphaphenylnitrone on hyperoxia-induced oxidative stress and its potential as a nitric oxide donor. *Free Radic. Res.* 2002;36:143-149.
- [26] Towner RA, Janzen EG, Zhang YK, Yamashiro S. MRI study of the inhibitory effect of new spin traps on in vivo CCl₄-induced hepatotoxicity in rats. *Free Radic. Biol.*

Med. 1993;14:677-681.

- [27] Yamashita T, Ohshima H, Asanuma T, Inukai N, Miyoshi I, Kasai N, *et al.* The effects of alpha-phenyl-tert-butyl nitrone (PBN) on copper-induced rat fulminant hepatitis with jaundice. *Free Radic. Biol. Med.* 1996;21:755-761.
- [28] Asanuma T, Yasui H, Inanami O, Waki K, Takahashi M, Iizuka D, et al. A new amphiphilic derivative, N-{[4-(lactobionamido)methyl]benzylidene}-1,1-dimethyl-2-(octylsulfanyl)ethylamine N-oxide, has a protective effect against copper-induced fulminant hepatitis in Long-Evans Cinnamon rats at an extremely low concentration compared with its original form alpha-phenyl-N-(tert-butyl) nitrone. *Chem. Biodivers.* 2007;4:2253-2267.
- [29] Naito Y, Takagi T, Ishikawa T, Handa O, Matsumoto N, Yagi N, et al. alpha-Phenyl-N-tert-butylnitrone provides protection from dextran sulfate sodium-induced colitis in mice. Antioxid. Redox Signal 2002;4:195-206.
- [30] Wang J, Cochran V, Abdi S, Chung JM, Chung K, Kim HK. Phenyl N-t-butylnitrone, a reactive oxygen species scavenger, reduces zymosan-induced visceral pain in rats. *Neurosci. Lett.* 2008;439:216-219.
- [31] Ancha HR, Kurella RR, McKimmey CC, Lightfoot S, Harty RF. Luminal antioxidants enhance the effects of mesalamine in the treatment of chemically induced colitis in rats. *Exp. Biol. Med.* (Maywood) 2008;233:1301-1308.
- [32] Pedraza-Chaverri J, Tapia E, Bobadilla N. Ischemia-reperfusion induced acute renal failure in the rat is ameliorated by the spin-trapping agent alpha-phenyl-N-tert-butyl nitrone (PBN). *Ren. Fail.* 1992;14:467-471.
- [33] Nakae D, Kotake Y, Kishida H, Hensley KL, Denda A, Kobayashi Y, *et al.* Inhibition by phenyl N-tert-butyl nitrone of early phase carcinogenesis in the livers of rats fed a choline-deficient, L-amino acid-defined diet. *Cancer Res.* 1998;58:4548-4551.
- [34] Nakae D, Uematsu F, Kishida H, Kusuoka O, Katsuda S, Yoshida M, *et al.* Inhibition of the development of hepatocellular carcinomas by phenyl N-tert-butyl nitrone in rats fed with a choline-deficient, L-amino acid-defined diet. *Cancer Lett.* 2004;206:1-13.
- [35] Nakae D, Kishida H, Enami T, Konishi Y, Hensley KL, Floyd RA, *et al.* Effects of phenyl N-tert-butyl nitrone and its derivatives on the early phase of hepatocarcinogenesis in rats fed a choline-deficient, L-amino acid-defined diet. *Cancer Sci.* 2003; 94:26-31.
- [36] Garteiser P, Doblas S, Watanabe Y, Saunders D, Hoyle J, Lerner M, et al. Multiparametric assessment of the anti-glioma properties of OKN007 by magnetic resonance imaging. J. Magn. Reson. Imaging. 2010;31:796-806.
- [37] Floyd RA, Towner RA, Wu D, Abbott A, Cranford R, Branch D, et al. Anti-cancer activity of nitrones in the Apc^{Min/+} model of colorectal cancer. *Free Radic. Res.* 2010; 44:108-117.
- [38] Kotake Y, Sang H, Tabatabaie T, Wallis GL, Moore DR, Stewart CA. Interleukin-10 overexpression mediates phenyl-N-tert-butyl nitrone protection from endotoxemia. *Shock* 2002;17:210-216.
- [39] Lee JH, Kim IS, Park JW. The use of N-t-butyl hydroxylamine for radioprotection in cultured cells and mice. *Carcinogenesis* 2004;25:1435-1442.
- [40] Tomita H, Kotake Y, Anderson RE. Mechanism of protection from light-induced retinal degeneration by the synthetic antioxidant phenyl-N-tert-butylnitrone. *Invest. Ophthalmol. Vis. Sci.* 2005;46:427-434.

- [41] Choi CH, Chen K, Vasquez-Weldon A, Jackson RL, Floyd RA, Kopke RD. Effectiveness of 4-hydroxy phenyl N-tert-butylnitrone (4-OHPBN) alone and in combination with other antioxidant drugs in the treatment of acute acoustic trauma in chinchilla. *Free Radic. Biol. Med.* 2008;44:1772-1784.
- [42] Floyd RA. Nitrones as therapeutics in age-related diseases. Aging Cell 2006;5:51-57.
- [43] Lees KR, Zivin JA, Ashwood T, Davalos A, Davis SM, Diener HC, et al. NXY-059 for acute ischemic stroke. N. Engl. J. Med. 2006;354:588-600.
- [44] Shuaib A, Lees KR, Lyden P, Grotta J, Davalos A, Davis SM, *et al.* NXY-059 for the treatment of acute ischemic stroke. *N. Engl. J. Med.* 2007;357:562-571.
- [45] Diener HC, Lees KR, Lyden P, Grotta J, Davalos A, Davis SM, et al. NXY-059 for the treatment of acute stroke: pooled analysis of the SAINT I and II Trials. Stroke 2008; 39:1751-1758.

Chapter 27

Nanomaterials

Abstract

Nanomaterials have received increasing attention in biology and medicine due to their unique interactions with biological systems and their potential applications in both diagnostics and therapeutics. Several kinds of nanomaterials, including cerium oxide nanoparticles, platinum nanoparticles, and fullerene derivatives have been shown to possess antioxidative and anti-inflammatory activities. Extensive studies have demonstrated a protective role for these antioxidant nanomaterials in a variety of animal models of human diseases, including cardiovascular diseases, neurological disorders, aging, and cancer. On the other hand, nanomaterials, like any existing drugs, also have the potential to cause adverse effects in biological systems. Future studies are warranted to further determine the efficacy and safety of the antioxidant nanomaterials in animal models as well as in clinical trials.

1. Overview

Nanomedicine, a combinatorial approach using nanotechnology and medicine, has become an increasingly important field of research in both diagnostics and therapeutics [1]. The field of nanomedicine involves the design and development of novel nanomaterials, such as metal nanoparticles, liposomal nanoparticles, fullerenes, carbon nanotubes, dendrimers, and nanoshells. Once engineered, these novel nanomaterials are investigated for their potential value in the diagnosis and treatment of diseases. Drug delivery, diagnostic imaging, and antioxidant activities are among the major areas of focus in nanomedicine.

Regarding the antioxidant properties of nanomaterials in biological systems, two scenarios may exit: (1) nanomaterials directly scavenge reactive oxygen and nitrogen species (ROS/RNS) via the unique redox chemistry of the surface and (2) nanomaterials serve as carriers for known antioxidant compounds or enzymes. Many antioxidants (either non-protein antioxidant compounds or antioxidant enzymes) exhibit low bioavailability due to limited solubility, instability, and first-pass effect. These limitations can be potentially circumvented by encapsulating the antioxidants to form antioxidant-loaded polymeric materials (nanocapsules or nanospheres). Certain antioxidant-loaded nanomaterials have been reported to protect against oxidative damage in cultured cells and animal models [2-4]. Hence, the second scenario represents a form of drug delivery. This chapter focuses on the antioxidant activities related to direct scavenging of ROS/RNS by the redox-active nanomaterials, which mainly include cerium oxide nanoparticles, platinum nanoparticles, and fullerenes.

2. Chemistry and Biochemistry

2.1. Cerium Oxide Nanoparticles

Cerium is a rare earth element of the lanthanide series. The oxide form (CeO_2) is used as a support material in automotive catalysis. Cerium oxide nanoparticles, also known as nanoceria have been shown to scavenge superoxide possibly via the redox properties of Ce^{3+}/Ce^{4+} (Reactions 1 and 2) [5, 6].

$$O_2 \cdot + Ce^{4+} \longrightarrow O_2 + Ce^{3+}$$
(1)

$$O_2 \stackrel{\cdot}{\cdot} + Ce^{3^+} + 2H^+ \longrightarrow H_2O_2 + Ce^{4^+}$$
(2)

Because of the above redox reactions, cerium oxide nanoparticles have been suggested to possess superoxide dismutase (SOD) mimetic activity, i.e., the autocatalytic dismutation of superoxide. It is further reported that the reaction rate constant for cerium oxide nanoparticle-catalyzed dismutation of superoxide may exceed that determined for the enzyme SOD [5]. More recently, cerium oxide nanoparticles are found to exhibit redox state-dependent catalase mimetic activity [7]. There is also evidence for the ability of cerium oxide to catalytically reduce nitrogen dioxide (NO_2) to nitric oxide (NO), and NO to N₂ [8, 9]. In contrast to the ROS/RNS-scavenging activities, formation of hydroxyl radicals from a Fenton-type reaction between cerium and hydrogen peroxide has been observed in vitro [10]. It remains to be further elucidated if the above redox reactions also occur in vivo. Figure 27-1 illustrates the redox chemistry of cerium oxide nanoparticles in relation to metabolism of ROS/RNS.



Figure 27-1. Metabolism of reactive oxygen and nitrogen species (ROS/RNS) catalyzed by cerium oxide nanoparticles. See text (Section 2.1) for description of the redox chemistry of cerium oxide nanoparticles in relation to catalytic metabolism of ROS/RNS.

2.2. Platinum Nanoparticles

Synthetic platinum nanoparticles have recently been shown to scavenge both superoxide and hydrogen peroxide in cell-free systems [11, 12]. The hydrogen peroxide-scavenging activity of platinum nanoparticles is much higher than that of EUK-8, a superoxide dismutase/ catalase mimetic (Section 2.1.2 of Chapter 24). In addition, platinum nanoparticles scavenge peroxyl radicals and inhibit peroxyl radical-induced lipid peroxidation [13]. More recently, it is reported that platinum nanoparticles may have an activity similar to mitochondrial NADH:ubiquinone oxidoreductase [14]. This suggests a unique opportunity for using the nanoparticles to protect against disease conditions, such as Parkinson's disease, in which increased ROS formation and decreased mitochondrial complex I activity are critical pathophysiological events.

2.3. Fullerenes

The fullerene revolution began with the discovery by a team of scientists in 1985 of the buckyball (C_{60}) composed of 60 carbon atoms arranged in a hollow soccer ball shape [15] (structure shown in Figure 27-2). Because of this discovery, three of the team members were awarded the 1996 Nobel Prize in Chemistry. C_{60} is also known as buckminsterfullerene (or fullerene for short), named after Richard Buckminster Fuller, the architect who created the dome in 1967 with the same shape as that of the carbon cluster. Now, it is known that fullerenes represent a family of carbon allotrope molecules in the form of a hollow sphere, ellipsoid, tube, or plane with various numbers of carbon atoms [16, 17]. Fullerenes, especially C_{60} have received great attention for their potential usefulness in biomedicine. A number of derivatives of C_{60} fullerene, especially the polyhydroxylated C_{60} fullerenes (also known as fullerenols) have been developed to improve their biocompatibility (e.g., water solubility) and activities in biological systems. These functionalized C_{60} fullerenes have been shown to exert a number of biological activities, including enzyme inhibition, antimicrobial activity, and scavenging of free radicals [16, 17]. C₆₀ fullerene derivatives may possess SOD activity, in that the fullerenes catalytically convert superoxide to hydrogen peroxide and molecular oxygen [18-20]. In addition, C_{60} fullerene derivatives are found to scavenge peroxyl radicals, singlet oxygen, and hydroxyl radicals [21, 22]. The free radical scavenging properties of fullerene derivatives may result from the capacity of the carbon-cluster to absorb electrons as well as the redox behaviors of the surface structures, including the hydroxyl groups introduced in the derivatives [18-20, 23].



Figure 27-2. Structure of C₆₀ fullerene. See text (Section 2.3) for description of fullerenes.

2.4. Other Nanomaterials

The free radical-scavenging properties of several other metal nanomaterials have also been reported in the literature. For example, both Nano-Se (also known as nano red element selenium) and nickel oxide nanoparticles are shown to scavenge carbon-centered free radicals in vitro [24, 25]. Nano-Se may also scavenge superoxide and singlet oxygen [25]. Gold-poly(amidoamine) dendrimer nanocomposites are found to potently scavenge hydroxyl radicals generated from the reaction between hydrogen peroxide and iron [26]. The redox chemistry underlying the potential free radical scavenging activities of these metal nanomaterials warrants investigations.

3. Role in Disease Intervention

3.1. Animal Studies

The increasing recognition of ROS-scavenging activities of cerium oxide nanoparticles, nickel oxide nanoparticles, and fullerenes in both cell-free and cell culture systems has prompted studies on using these nanomaterials for the intervention of diseases involving an oxidative stress mechanism. In addition, cerium oxide nanoparticles and fullerenes have been found to also possess anti-inflammatory activities, as evidenced by their ability to reduce the expression of inflammatory cytokines, adhesion molecules, and inducible nitric oxide synthase [27-29]. As oxidative stress and inflammation are intimately related, the ROS-scavenging ability of cerium oxide nanoparticles and fullerenes may primarily contribute to their anti-inflammatory activities. The sections below summarize the recent findings regarding the protective effects of nanomaterials on oxidative and inflammatory stress-mediated disease pathophysiology in various animal models of human diseases.

3.1.1. Role in Cardiovascular Diseases

Multiple in vitro experiments have demonstrated an important role for cerium oxide nanoparticles and C₆₀ fullerene derivatives in suppressing oxidative stress injury and proinflammatory responses in cultured cardiovascular cells [30, 31]. The in vivo cardioprotection of cerium oxide nanoparticles was first studied by J. Niu and coworkers in a transgenic murine model of ischemic cardiomyopathy [32]. They demonstrated that treatment with cerium oxide nanoparticles attenuates the progressive cardiac dysfunction and remodeling as a result of inhibition of myocardial oxidative stress, endoplasmic reticulum stress, and inflammatory processes. The autoregenerative antioxidant properties of cerium oxide nanoparticles are suggested to account for the above beneficial effects. In addition to cerium oxide nanoparticles, fullerene derivatives may also have a protective effect on oxidative cardiomyopathy. As described in Chapter 3, mice with homozygous deletion of MnSOD die in utero or within a few days of birth due to catastrophic oxidative damage to mitochondria and subsequent development of dilated cardiomyopathy. Notably, treatment with a tris-malonic derivative of C_{60} fullerene dramatically increases the lifespan of the MnSOD^{-/-} mice [18]. This observation coupled with the evidence that the C₆₀ fullerene derivative is localized to mitochondria suggests that this fullerene derivative may function as MnSOD to protect against disease pathophysiology associated with mitochondrial oxidative stress [18]. Recently, a polyhydroxylated fullerene derivative $C_{60}OH_{24}$ is shown to also protect against doxorubicin-induced acute and chronic cardiotoxicity as well as myocardial oxidative stress in rats [33, 34].

3.1.2. Role in Neurological Diseases

Neuroprotection is an effect that may result in salvage, recovery, or regeneration of the nervous system under disease conditions. A promising neuroprotective approach involves mitigating oxidative and inflammatory stress, which are believed to be key pathophysiological events underlying many types of neurological diseases, including ischemia, trauma, and degenerative disorders. Hence, recent efforts have been focused on determining the potential neuroprotective effects of nanomaterials that have antioxidant and anti-inflammatory properties. A number of in vitro studies demonstrate that cerium oxide nanoparticles and fullerene derivatives protect cultured neurons from oxidative stress injury elicited by exogenous ROS. excitotoxic compounds, and β -amyloid peptides [35-39]. In line with the in vitro findings, in vivo studies show that administration of a carboxyfullerene or a water-soluble hexasulfonated C_{60} fullerene derivative protects against cerebral ischemia-reperfusion injury and oxidative stress in experimental animals [40-42]. Treatment with carboxyfullerene derivatives also results in prolongation of survival and amelioration of oxidative functional degeneration in a murine model of familial amyotrophic lateral sclerosis [38]. In addition, carboxyfullerene derivatives suppress iron-induced oxidative stress in nigrostriatal dopaminergic system in vivo [43]. Concomitant local injection of a carboxyfullerene derivative also attenuates oxidative degeneration of substantia nigra induced by intranigral infusion of 1-methyl-4phenylpyridinium (MPP⁺), the active metabolite of the parkinsonism-inducing agent 1methyl-4-phenyl-1,2,3,6-tetrahydropyridine (MPTP) [44]. In contrast, concomitant systemic administration of the carboxyfullerene and MPTP leads to potentiation of MPTP-induced dopaminergic degeneration. An increase in the striatal MPP⁺ level is observed in the carboxyfullerene and MPTP-cotreated mice, which may explain the above potentiation effect [44]. These findings implicate that fullerene-based nanomaterials may exert both beneficial and detrimental effects on disease pathophysiology depending on the experimental conditions.

3.1.3. Role in Pulmonary Diseases

Lung can be readily exposed to nanomaterials via inhalation, and as such the potential harmful effects of nanomaterials on respiratory system have been the focus of research in nanotoxicology [45]. On the other hand, the potent antioxidative and anti-inflammatory activities of certain nanomaterials may also have implications for protection against pulmonary diseases involving a mechanism of oxidative and inflammatory stress. Early studies demonstrated that administration of water-soluble fullerene derivatives leads to attenuation of lung ischemia-reperfusion injury and associated oxidative stress [46, 47]. More recently, fullerene derivatives are shown to inhibit IgE-dependent activation of human mast cells and peripheral basophils possibly as a result of the fullerene-mediated free radical scavenging [48]. This suggests that antioxidant fullerenes may be useful for the intervention of airway inflammatory disorders, such as asthma. In addition to fullerenes, the antioxidant cerium oxide and platinum nanoparticles also possess pulmonary protective activities. For example, intraperitoneal administration of cerium oxide nanoparticles to mice prevents the onset of radiation-induced pneumonia and prolongs the animal survival [49]. Intranasal administration of platinum na-

noparticles prior to exposure to cigarette smoke attenuates pulmonary inflammation, oxidative stress, and histological damage in mice [50]. Notably, nasal administration of platinum nanoparticles does not elicit pulmonary inflammation and toxicity. These observations suggest a potential value for using nanoparticles to prevent inflammatory pulmonary injury caused by environmental toxicants.

3.1.4. Role in Hepatic and Gastrointestinal Diseases

Pretreatment with C_{60} fullerene and its derivatives protects against carbon tetrachlorideor doxorubicin-induced hepatotoxicity and oxidative stress in experimental animals [51, 52]. Administration of a water-soluble fullerene derivative also inhibits intestinal ischemiareperfusion injury via its free-radical scavenging activities [53]. Recently, it is shown that cerium oxide nanoparticles protect gastrointestinal epithelium from radiation-induced damage by reduction of ROS and upregulation of MnSOD in mice [54]. In cultured normal human colon cells, cerium oxide nanoparticles also inhibit radiation-induced ROS formation and cell death. Similar to the in vivo findings, treatment with cerium oxide nanoparticles increases the expression of MnSOD in cultured colon cells [54]. As MnSOD is a critical antioxidant defense, the above observations indicate a novel mechanism by which cerium oxide nanoparticles suppress oxidative stress. The molecular mechanism underlying cerium oxide nanoparticle-mediated induction of MnSOD awaits further studies.

3.1.5. Role in Eye Diseases

Retinal degeneration is a common cause of blindness. Photoreceptor cells of the retina are incessantly bombarded with photons of light, which, along with the cells' high rate of oxygen metabolism, continuously expose them to elevated levels of ROS. Indeed, accumulating evidence supports a critical involvement of oxidative stress in retinal degeneration. The potent autocatalytic properties of cerium oxide nanoparticles in detoxifying ROS have led to the use of the nanoparticles for treating oxidative retinal degeneration in experimental models [55-57]. Treatment of cultured retinal neurons with cerium oxide nanoparticles reduces intracellular accumulation of ROS. In vivo injection of cerium oxide nanoparticles also prevents loss of vision due to damaging light-induced degeneration of photoreceptor cells in mice. Notably, even when injected after light damage, the cerium oxide nanoparticles are still partially effective, suggesting that as long as the cells have not died, they may be prevented from entering or preceding down a death pathway [55]. These observations indicate that cerium oxide nanoparticles may be an effective modality for preventing and/or treating eye disorders associated with oxidative retinal degeneration.

3.1.6. Role in Aging

Exposure to cerium oxide nanoparticles prolongs the longevity of *Drosophila melanogaster*, suggesting a possible beneficial effect of nanomaterials in aging [58]. Chronic administration of a carboxyfullerene having SOD mimetic activity to mice not only reduces agingassociated oxidative damage and mitochondrial free radical production, but also significantly extends lifespan [59]. The fullerene-treated mice also exhibit improved performance on the Morris water maze learning and memory task, suggesting that the antioxidant fullerene derivative also rescues aging-related cognitive impairment [59]. Recently, the antioxidant platinum nanoparticles are found to possess significant antiaging activities in a nematode *Caenorhabdi*- *tis elegans* model [60, 61]. Notably, treatment with platinum nanoparticles also recovers the shortened lifespan of the mev-1 (kn1) mutant. The shortened lifespan of the mutant is due to excessive oxidative stress. Collectively, these findings point to the feasibility of using anti-oxidant nanomaterials to combat aging.

3.1.7. Role in Cancer

Using nanomaterials as a novel strategy for anticancer drug delivery has recently received much attention [62]. This strategy significantly enhances the effectiveness of the well-established anticancer drugs in animal models. On the other hand, there is evidence suggesting that certain nanomaterials having antioxidative and anti-inflammatory properties may directly play a role in protecting against cancer development in experimental models. In vitro studies demonstrate that treatment with platinum nanoparticles suppresses phorbol-12-myristate-13-acetate (PMA)-induced ROS formation and promotion of cell transformation in a two-stage cell transformation model [63]. PMA is a classical tumor promoting agent capable of eliciting inflammation and oxidative stress. In cell cultures, platinum nanoparticles also exhibit selective inhibition of carcinoma cells over normal cells [64]. More recently, polyhydroxylated C_{82} fullerene derivatives are shown to potently inhibit angiogenesis and tumor growth in a mouse model of breast cancer [65]. In contrast to water soluble polyhydroxylated fullerenes, nanocrystalline C_{60} fullerene is reported to augment tumor growth in a mouse model of breast cancer [65]. In contrast to water soluble polyhydroxylated fullerenes, nanocrystalline C₆₀ fullerene is reported to augment tumor growth in a mouse model of be further investigated.

3.1.8. Role in Other Diseases and Conditions

Antioxidant nanomaterials, especially the polyhydroxylated fullerene derivatives also have a protective role in various other disease processes in experimental animals. These include ultraviolet irradiation-induced oxidative skin injury [67], loss of hair [68], inflammatory arthritis [69], ionizing irradiation-induced immune dysfunction and multi-organ damage [70], and infections [17].

3.2. Human Studies and Clinical Perspectives

As discussed above, nanomaterials, including cerium oxide nanoparticles, platinum nanoparticles, and fullerenes possess antioxidative and anti-inflammatory activities, and their administration to experimental animals affords protection against a variety of disease conditions that involve oxidative stress and inflammation. However, clinical studies on using these nanomaterials for the intervention of human diseases have not yet been reported in the literature. One likely reason is that nanomedicine is still in its infancy, and much remains to be learned from studies using experimental models. Nanomaterials as potential drugs need to be further investigated in animal models with respect to not only their pharmacological effects, but also their toxicological profiles. In this regard, nanomaterials, including the antioxidant nanomaterials discussed in this Chapter have the potential to cause adverse effects [45]. In fact, nanotoxicology has been established recently to specifically address the detrimental effects of nanomaterials in biological systems. Although still in its early stage, nanotoxicology has already yielded enormous amounts of information regarding the adverse effects of various kinds of nanomaterials. Knowledge on both efficacy and safety of antioxidant nanomaterials in experimental models will lay a basis for subsequent clinical studies to determine their value in the prevention and/or treatment of human diseases.

4. Conclusion and Future Directions

It has been increasingly recognized that nanomaterials, such as nanoceria, platinum nanoparticles, and fullerenes possess antioxidant activities via scavenging various ROS/RNS and other related reactive species. Moreover, nanomaterials, including cerium oxide nanoparticles and fullerene derivatives, exhibit SOD mimetic activity, suggesting a potential for these nanomaterials to catalytically detoxify superoxide in biological systems. Indeed, accumulating evidence points to the effectiveness of the above nanomaterials in protecting against disease pathophysiology involving an oxidative stress mechanism in animal models. It should be borne in mind that nanomaterials may possess other biological activities irrelevant to free radical scavenging, and these non-antioxidant activities may also play a role in disease intervention. Future studies should focus on continued characterization of both efficacy and safety of the already existing antioxidant nanomaterials in preclinical studies. Such studies will lay a foundation for subsequent clinical trials to evaluate the potential benefits of these nanomaterials in human disease intervention. Future efforts should also be devoted to the development of novel nanomaterials or derivatives of the existing nanomaterials with improved antioxidant properties and reduced adverse effects.

5. References

- [1] Murday JS, Siegel RW, Stein J, Wright JF. Translational nanomedicine: status assessment and opportunities. *Nanomedicine* 2009;5:251-273.
- [2] Reddy MK, Labhasetwar V. Nanoparticle-mediated delivery of superoxide dismutase to the brain: an effective strategy to reduce ischemia-reperfusion injury. *FASEB J.* 2009;23:1384-1395.
- [3] Carroll RT, Bhatia D, Geldenhuys W, Bhatia R, Miladore N, Bishayee A, *et al.* Braintargeted delivery of Tempol-loaded nanoparticles for neurological disorders. *J. Drug Target* 2010 (in press).
- [4] Yen FL, Wu TH, Tzeng CW, Lin LT, Lin CC. Curcumin nanoparticles improve the physicochemical properties of curcumin and effectively enhance its antioxidant and antihepatoma activities. *J. Agric. Food Chem.* 2010;58:7376-7382.
- [5] Korsvik C, Patil S, Seal S, Self WT. Superoxide dismutase mimetic properties exhibited by vacancy engineered ceria nanoparticles. *Chem. Commun.* (Camb) 2007: 1056-1058.
- [6] Heckert EG, Karakoti AS, Seal S, Self WT. The role of cerium redox state in the SOD mimetic activity of nanoceria. *Biomaterials* 2008;29:2705-2709.
- [7] Pirmohamed T, Dowding JM, Singh S, Wasserman B, Heckert E, Karakoti AS, et al. Nanoceria exhibit redox state-dependent catalase mimetic activity. *Chem. Commun.* (Camb) 2010;46:2736-2738.
- [8] Nolan M, Parker SC, Watson GW. CeO₂ catalysed conversion of CO, NO₂ and NO

from first principles energetics. Phys. Chem. Chem. Phys. 2006;8:216-218.

- [9] Nolan M, Parker SC, Watson GW. Reduction of NO₂ on ceria surfaces. J. Phys. Chem. B 2006;110:2256-2262.
- [10] Heckert EG, Seal S, Self WT. Fenton-like reaction catalyzed by the rare earth inner transition metal cerium. *Environ. Sci. Technol.* 2008;42:5014-5019.
- [11] Kajita M, Hikosaka K, Iitsuka M, Kanayama A, Toshima N, Miyamoto Y. Platinum nanoparticle is a useful scavenger of superoxide anion and hydrogen peroxide. *Free Radic. Res.* 2007;41:615-626.
- [12] Hamasaki T, Kashiwagi T, Imada T, Nakamichi N, Aramaki S, Toh K, *et al.* Kinetic analysis of superoxide anion radical-scavenging and hydroxyl radical-scavenging activities of platinum nanoparticles. *Langmuir.* 2008;24:7354-7364.
- [13] Watanabe A, Kajita M, Kim J, Kanayama A, Takahashi K, Mashino T, et al. In vitro free radical scavenging activity of platinum nanoparticles. *Nanotechnology* 2009;20: 455105.
- [14] Hikosaka K, Kim J, Kajita M, Kanayama A, Miyamoto Y. Platinum nanoparticles have an activity similar to mitochondrial NADH:ubiquinone oxidoreductase. *Colloids Surf. B Biointerfaces* 2008;66:195-200.
- [15] Kroto H, Heath J, BO'Brien S, Curl R, Smaller R. C₆₀: buckminsterfullerene. *Nature* 1985;318:162-163.
- [16] Partha R, Conyers JL. Biomedical applications of functionalized fullerene-based nanomaterials. *Int. J. Nanomedicine* 2009;4:261-275.
- [17] Chawla P, Chawla V, Maheshwari R, Saraf SA, Saraf SK. Fullerenes: from carbon to nanomedicine. *Mini. Rev. Med. Chem.* 2010;10:662-677.
- [18] Ali SS, Hardt JI, Quick KL, Kim-Han JS, Erlanger BF, Huang TT, *et al.* A biologically effective fullerene (C₆₀) derivative with superoxide dismutase mimetic properties. *Free Radic. Biol. Med.* 2004;37:1191-1202.
- [19] Liu GF, Filipovic M, Ivanovic-Burmazovic I, Beuerle F, Witte P, Hirsch A. High catalytic activity of dendritic C₆₀ monoadducts in metal-free superoxide dismutation. *Angew Chem. Int. Ed. Engl.* 2008;47:3991-3994.
- [20] Osuna S, Swart M, Sola M. On the mechanism of action of fullerene derivatives in superoxide dismutation. *Chemistry* 2010;16:3207-3214.
- [21] Horie M, Fukuhara A, Saito Y, Yoshida Y, Sato H, Ohi H, et al. Antioxidant action of sugar-pendant C₆₀ fullerenes. *Bioorg. Med. Chem. Lett.* 2009;19:5902-5904.
- [22] Yin JJ, Lao F, Fu PP, Wamer WG, Zhao Y, Wang PC, et al. The scavenging of reactive oxygen species and the potential for cell protection by functionalized fullerene materials. *Biomaterials* 2009;30:611-621.
- [23] Krusic PJ, Wasserman E, Keizer PN, Morton JR, Preston KF. Radical reactions of C₆₀. Science 1991;254:1183-1185.
- [24] Saikia JP, Paul S, Konwar BK, Samdarshi SK. Nickel oxide nanoparticles: a novel antioxidant. *Colloids Surf. B Biointerfaces* 2010;78:146-148.
- [25] Huang B, Zhang J, Hou J, Chen C. Free radical scavenging efficiency of Nano-Se in vitro. *Free Radic. Biol. Med.* 2003;35:805-813.
- [26] Esumi K, Houdatsu H, Yoshimura T. Antioxidant action by gold-PAMAM dendrimer nanocomposites. *Langmuir* 2004;20:2536-2538.
- [27] Hirst SM, Karakoti AS, Tyler RD, Sriranganathan N, Seal S, Reilly CM. Antiinflammatory properties of cerium oxide nanoparticles. *Small* 2009;5:2848-2856.

- [28] Tzeng SF, Lee JL, Kuo JS, Yang CS, Murugan P, Ai Tai L, et al. Effects of malonate C₆₀ derivatives on activated microglia. *Brain Res.* 2002;940:61-68.
- [29] Dellinger A, Zhou Z, Lenk R, MacFarland D, Kepley CL. Fullerene nanomaterials inhibit phorbol myristate acetate-induced inflammation. *Exp. Dermatol.* 2009;18:1079-1081.
- [30] Younce CW, Wang K, Kolattukudy PE. Hyperglycaemia-induced cardiomyocyte death is mediated via MCP-1 production and induction of a novel zinc-finger protein MCPIP. *Cardiovasc. Res.* 2010;87:665-674.
- [31] Maeda R, Noiri E, Isobe H, Homma T, Tanaka T, Negishi K, et al. A water-soluble fullerene vesicle alleviates angiotensin II-induced oxidative stress in human umbilical venous endothelial cells. *Hypertens. Res.* 2008;31:141-151.
- [32] Niu J, Azfer A, Rogers LM, Wang X, Kolattukudy PE. Cardioprotective effects of cerium oxide nanoparticles in a transgenic murine model of cardiomyopathy. *Cardio-vasc. Res.* 2007;73:549-559.
- [33] Injac R, Perse M, Boskovic M, Djordjevic-Milic V, Djordjevic A, Hvala A, et al. Cardioprotective effects of fullerenol C₆₀(OH)₂₄ on a single dose doxorubicin-induced cardiotoxicity in rats with malignant neoplasm. *Technol. Cancer Res. Treat.* 2008;7:15-25.
- [34] Injac R, Perse M, Cerne M, Potocnik N, Radic N, Govedarica B, *et al.* Protective effects of fullerenol $C_{60}(OH)_{24}$ against doxorubicin-induced cardiotoxicity and hepatotoxicity in rats with colorectal cancer. *Biomaterials* 2009;30:1184-1196.
- [35] Schubert D, Dargusch R, Raitano J, Chan SW. Cerium and yttrium oxide nanoparticles are neuroprotective. *Biochem. Biophys. Res. Commun.* 2006;342:86-91.
- [36] Das M, Patil S, Bhargava N, Kang JF, Riedel LM, Seal S, et al. Auto-catalytic ceria nanoparticles offer neuroprotection to adult rat spinal cord neurons. *Biomaterials* 2007;28:1918-1925.
- [37] Dugan LL, Gabrielsen JK, Yu SP, Lin TS, Choi DW. Buckminsterfullerenol free radical scavengers reduce excitotoxic and apoptotic death of cultured cortical neurons. *Neurobiol. Dis.* 1996;3:129-135.
- [38] Dugan LL, Turetsky DM, Du C, Lobner D, Wheeler M, Almli CR, *et al.* Carboxyfullerenes as neuroprotective agents. *Proc. Natl. Acad. Sci. USA* 1997;94:9434-9439.
- [39] Lee CM, Huang ST, Huang SH, Lin HW, Tsai HP, Wu JY, *et al.* C₆₀ Fullerenepentoxifylline dyad nanoparticles enhance autophagy to avoid cytotoxic effects caused by the beta-amyloid peptide. *Nanomedicine* 2010 (in press).
- [40] Lin AM, Fang SF, Lin SZ, Chou CK, Luh TY, Ho LT. Local carboxyfullerene protects cortical infarction in rat brain. *Neurosci. Res.* 2002;43:317-321.
- [41] Huang SS, Tsai SK, Chih CL, Chiang LY, Hsieh HM, Teng CM, et al. Neuroprotective effect of hexasulfobutylated C₆₀ on rats subjected to focal cerebral ischemia. Free Radic. Biol. Med. 2001;30:643-649.
- [42] Yang DY, Wang MF, Chen IL, Chan YC, Lee MS, Cheng FC. Systemic administration of a water-soluble hexasulfonated C₆₀ (FC₄S) reduces cerebral ischemia-induced infarct volume in gerbils. *Neurosci. Lett.* 2001;311:121-124.
- [43] Lin AM, Chyi BY, Wang SD, Yu HH, Kanakamma PP, Luh TY, et al. Carboxyfullerene prevents iron-induced oxidative stress in rat brain. J. Neurochem. 1999;72:1634-1640.
- [44] Lin AM, Yang CH, Ueng YF, Luh TY, Liu TY, Lay YP, et al. Differential effects of

carboxyfullerene on MPP⁺/MPTP-induced neurotoxicity. *Neurochem. Int.* 2004;44:99-105.

- [45] Xia T, Li N, Nel AE. Potential health impact of nanoparticles. *Annu. Rev. Public Health* 2009;30:137-150.
- [46] Lai YL, Murugan P, Hwang KC. Fullerene derivative attenuates ischemia-reperfusioninduced lung injury. *Life Sci.* 2003;72:1271-1278.
- [47] Chen YW, Hwang KC, Yen CC, Lai YL. Fullerene derivatives protect against oxidative stress in RAW 264.7 cells and ischemia-reperfused lungs. Am. J. Physiol. Regul. Integr. Comp. Physiol. 2004;287:R21-26.
- [48] Ryan JJ, Bateman HR, Stover A, Gomez G, Norton SK, Zhao W, *et al.* Fullerene nanomaterials inhibit the allergic response. *J. Immunol.* 2007;179:665-672.
- [49] Colon J, Herrera L, Smith J, Patil S, Komanski C, Kupelian P, et al. Protection from radiation-induced pneumonitis using cerium oxide nanoparticles. *Nanomedicine* 2009; 5:225-231.
- [50] Onizawa S, Aoshiba K, Kajita M, Miyamoto Y, Nagai A. Platinum nanoparticle antioxidants inhibit pulmonary inflammation in mice exposed to cigarette smoke. *Pulm. Pharmacol. Ther.* 2009;22:340-349.
- [51] Gharbi N, Pressac M, Hadchouel M, Szwarc H, Wilson SR, Moussa F. [60]fullerene is a powerful antioxidant in vivo with no acute or subacute toxicity. *Nano Lett.* 2005; 5: 2578-2585.
- [52] Injac R, Perse M, Obermajer N, Djordjevic-Milic V, Prijatelj M, Djordjevic A, *et al.* Potential hepatoprotective effects of fullerenol $C_{60}(OH)_{24}$ in doxorubicin-induced hepatotoxicity in rats with mammary carcinomas. *Biomaterials* 2008;29:3451-3460.
- [53] Lai HS, Chen WJ, Chiang LY. Free radical scavenging activity of fullerenol on the ischemia-reperfusion intestine in dogs. *World J. Surg.* 2000;24:450-454.
- [54] Colon J, Hsieh N, Ferguson A, Kupelian P, Seal S, Jenkins DW, *et al.* Cerium oxide nanoparticles protect gastrointestinal epithelium from radiation-induced damage by reduction of reactive oxygen species and upregulation of superoxide dismutase 2. *Nanomedicine* 2010;6:698-705.
- [55] Chen J, Patil S, Seal S, McGinnis JF. Rare earth nanoparticles prevent retinal degeneration induced by intracellular peroxides. *Nat. Nanotechnol.* 2006;1:142-150.
- [56] Silva GA. Nanomedicine: seeing the benefits of ceria. Nat. Nanotechnol. 2006;1:92-94.
- [57] Chen J, Patil S, Seal S, McGinnis JF. Nanoceria particles prevent ROI-induced blindness. *Adv. Exp. Med. Biol.* 2008;613:53-59.
- [58] Rzigalinski BA, Meehan K, Davis RM, Xu Y, Miles WC, Cohen CA. Radical nanomedicine. *Nanomedicine* (Lond) 2006;1:399-412.
- [59] Quick KL, Ali SS, Arch R, Xiong C, Wozniak D, Dugan LL. A carboxyfullerene SOD mimetic improves cognition and extends the lifespan of mice. *Neurobiol. Aging* 2008; 29:117-128.
- [60] Kim J, Takahashi M, Shimizu T, Shirasawa T, Kajita M, Kanayama A, et al. Effects of a potent antioxidant, platinum nanoparticle, on the lifespan of Caenorhabditis elegans. *Mech. Ageing Dev.* 2008;129:322-331.
- [61] Kim J, Shirasawa T, Miyamoto Y. The effect of TAT conjugated platinum nanoparticles on lifespan in a nematode Caenorhabditis elegans model. *Biomaterials* 2010; 31:5849-5854.
- [62] Chithrani DB. Nanoparticles for improved therapeutics and imaging in cancer therapy.
Recent Pat. Nanotechnol. 2010 (in press).

- [63] Nishikawa R, Teruya K, Katakura Y, Osada K, Hamasaki T, Kashiwagi T, et al. Electrolyzed Reduced Water Supplemented with Platinum Nanoparticles Suppresses Promotion of Two-stage Cell Transformation. Cytotechnology 2005;47:97-105.
- [64] Saitoh Y, Yoshimura Y, Nakano K, Miwa N. Platinum nanocolloid-supplemented hydrogendissolved water inhibits growth of human tongue carcinoma cells preferentially over normal cells. *Exp. Oncol.* 2009;31:156-162.
- [65] Meng H, Xing G, Sun B, Zhao F, Lei H, Li W, *et al.* Potent angiogenesis inhibition by the particulate form of fullerene derivatives. *ACS Nano.* 2010;4:2773-2783.
- [66] Zogovic NS, Nikolic NS, Vranjes-Djuric SD, Harhaji LM, Vucicevic LM, Janjetovic KD, *et al.* Opposite effects of nanocrystalline fullerene (C_{60}) on tumour cell growth in vitro and in vivo and a possible role of immunosupression in the cancer-promoting activity of C_{60} . *Biomaterials* 2009;30:6940-6946.
- [67] Ito S, Itoga K, Yamato M, Akamatsu H, Okano T. The co-application effects of fullerene and ascorbic acid on UV-B irradiated mouse skin. *Toxicology* 2010;267:27-38.
- [68] Zhou Z, Lenk R, Dellinger A, MacFarland D, Kumar K, Wilson SR, et al. Fullerene nanomaterials potentiate hair growth. *Nanomedicine* 2009;5:202-207.
- [69] Yudoh K, Karasawa R, Masuko K, Kato T. Water-soluble fullerene (C₆₀) inhibits the development of arthritis in the rat model of arthritis. *Int. J. Nanomedicine* 2009;4:217-225.
- [70] Cai X, Hao J, Zhang X, Yu B, Ren J, Luo C, *et al.* The polyhydroxylated fullerene derivative C₆₀(OH)₂₄ protects mice from ionizing-radiation-induced immune and mitochondrial dysfunction. *Toxicol. Appl. Pharmacol.* 2010;243:27-34.

Index

1

1,2-dithiolane-3-pentanoic acid, 249
12-O-tetracecanoylphorbol-13-acetate, 116
17-β-estradiol, 105
1-chloro-2,4-dinitrobenzene, 137
1-Cys Prx, 123, 124, 125
1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine, 96, 221, 234, 276, 381
¹O₂, 4, 12

2

- 2,3,7,8-tetrachlorodibenzo-p-dioxin, 182
- 2,3-dimethoxy-5-methyl-6-multiprenyl-1,4benzoquinone, 244
- 2,4-disulfophenyl-N-tert-butylnitrone, 367, 368, 370, 371
- 20S proteasome, 180, 181
- 2-AAPA, 87
- 2-Cys Prxs, 143, 144
- 2-hydroxyquinoline, 201
- 2-phenyl-1,2-benzisoselenazol-3(2H)-one, 342 2-SeCD, 344

3

3-hydroxy-3-methylglutaryl coenzyme A reductase, 284

4-hydroxy-2,2,6,6-tetramethylpiperidine-N-oxyl, 340

4

4-hydroxy-2-nonenal, 14 4-hydroxy-docosahexaenoic acid, 199 4-hydroxyl-PBN, 371

5

- 5,5-dimethyl-1-pyrroline N-oxide, 367, 373
- 5-diethoxyphosphoryl-5-methyl-1-pyrroline N-oxide, 368
- 5-ethoxycarbonyl-5-methyl-1-pyrroline N-oxide, 368 5-hydroxy-eicosatetraenoic acid 1,5-lactone, 199
- 5-lipoxygenase, 342
- 5-oxoproline, 67
- 5-oxoprolinuria, 67, 70

6

7

Α

6-hydroxydopamine, 222, 232, 234, 308

7,12-dimethylbenz(a)anthracene, 183, 192

absorption, 245, 268, 322

abstraction, 283

abuse, 189 acatalasemia, 47, 52

acatalasemic mice, 50, 51, 55

acetaminophen, 38, 39, 44, 77, 82, 96, 100, 166, 203, 227, 235, 270, 293, 349, 353, 356, 357, 358, 362, 363, 364

adulthood, 37

acetaminophen overdose, 96, 203, 353, 356, 357, 358 acetic acid, 199, 294 acetylation, 126, 145, 322 acid, 4, 6, 9, 10, 12, 13, 14, 18, 33, 37, 55, 69, 73, 74, 82, 87, 92, 95, 99, 100, 125, 126, 133, 144, 146, 147, 149, 151, 156, 166, 171, 174, 186, 187, 199, 205, 206, 227, 234, 239, 245, 249, 250, 251, 252, 253, 254, 259, 260, 261, 265, 266, 268, 274, 275, 276, 277, 278, 279, 283, 292, 294, 301, 315, 316, 317, 340, 341, 342, 349, 350, 365, 371, 375, 388 acne, 270, 274, 277, 279 acne vulgaris, 270, 274, 277, 279 aconitase, 8 acoustical trauma, 371 acrolein, 14 acrylhomoserine lactones, 199, 203 active centers, 217 active site, 48, 86, 104, 144, 146, 150, 154, 209 active transport, 322 acute burn injury, 252 acute inflammation, 223, 310 acute ischemic stroke, 254, 260, 261, 346, 350, 367, 370, 372, 376 acute liver failure, 65, 344, 348, 349, 358, 364 acute lung injury, 45, 189, 195, 227, 234, 242, 243, 256, 294 acute myeloid leukemia, 278 acute pancreatitis, 252 acute renal failure, 293, 371, 375 acute respiratory distress syndrome, 176, 344 addictive behaviors, 360 adenocarcinoma, 170, 177, 257, 289 adenoma, 138, 141, 328 adenomatous colonic polyps, 360 adenovirus, 43, 55, 128, 129, 164, 174, 201, 202, 221, 222, 228 adhesion, 15, 105, 184, 242, 246, 255, 276, 319, 354, 380 adhesion molecule, 15, 105, 184, 242, 246, 255, 319, 354, 380 adipocyte, 221 adipogenesis, 344, 348 adiponectin, 174, 214, 221, 311 adipose, 81, 98, 101, 184, 188, 193, 194, 282, 301, 308, 322, 323 adipose tissue, 98, 184, 188, 193, 194, 282, 301, 308, 323 adiposity, 155, 158, 188, 194, 202, 344 adrenal gland, 269, 282 adrenal glands, 282 adult T-cell, 120

AEOL series, 340 AEOL10113, 340 aerobic organisms, 1, 3, 4, 14, 23, 29 aflatoxin, 30, 66, 70, 96, 100, 350 aflatoxin B1, 66, 70, 96, 100 age-related macular degeneration, 209, 216, 302, 305, 310, 311 aggregation, 13, 56, 153, 158, 223, 276, 285, 321, 328, 354, 360 aging process, 129, 152 agonist, 171, 258 agranulocytosis, 189, 196 AhR, 93, 181, 200 AIDS, 203, 294, 364 air pollutants, 7, 38 airway epithelial cells, 82, 185, 301, 308 airway epithelium, 250 airway hyperresponsiveness, 185, 362 airway inflammation, 65, 77, 82, 96, 100, 103, 107, 115, 120, 128, 164, 165, 173, 185, 227, 235, 242, 269, 285, 292, 301, 308, 324, 356 airway wall thickening, 128 airways, 55, 359 AKT, 128, 133 ALA6 alleles, 79 albumin, 240 alcohol, 12, 17, 38, 43, 48, 83, 100, 125, 128, 166, 195, 200, 210, 215, 227, 302, 309, 348 alcohol abuse, 215 alcohol consumption, 83 alcohol withdrawal, 189, 195 alcoholic cardiomyopathy, 226 alcoholic cirrhosis, 45 alcoholic fatty liver disease, 71, 324 alcoholic liver disease, 65, 174, 215 alcoholic liver injury, 6, 98, 128, 203, 208, 235, 302 alcoholism, 189 alcohols, 28, 48, 73, 75, 93, 123, 124 aldehyde, 48, 100, 345 aldehvde dehvdrogenase, 100, 345 aldehydes, 14, 28, 62, 93, 95, 140, 184 aliphatic lactones, 199 alkenes. 66 alkoxyl radical, 4, 11 alkylating agents, 97 allantoin, 252, 253 allele, 169, 186, 187, 189, 194, 195, 206, 207, 215, 243, 257 allergenicity, 115 allergen-induced airway inflammation, 77, 82, 165, 173, 292 allergens, 115, 285

allergic airway diseases, 286 allergic asthma, 65, 128, 133, 292, 308, 362 allergic inflammation, 286, 293 allergic rhinitis, 293 allografts, 95, 100, 174, 242, 374 alloxan, 37, 55, 275, 364, 369 all-rac- α -tocopherol, 282 all-trans retinoic acid, 151 alopecia, 346, 351 α -(4-pyridyl-1-oxide)-N-tert-butylnitrone, 368 α -amvlase. 322 α-carotene, 297, 299 α -glucosidase, 322 a-keto acid, 251 α -ketoacid dehydrogenase, 249 α-lipoic acid, 23, 239, 249, 250 α -phenyl-tert-butylnitrone, 367, 368 α -synuclein, 153 α-tocopherol, 28, 62, 63, 136, 180, 181, 245, 281, 282, 283, 284, 285, 286, 287, 288, 289, 290 α-tocopherol acetate, 284 α -tocopherol radical, 62, 267, 283 α -tocopherol succinate, 284 α -tocopherol transfer protein, 283, 284 α -tocopherol-quinone, 181 α -tocopheryl phosphate, 282, 284 α-TTP, 283-285, 287 ALS. 214 Alzheimer's disease, 16, 37, 40, 76, 77, 153, 154, 164, 189, 207, 208, 246, 269, 301, 344, ambient air, 56 amino acids, 59, 60, 61 aminotriazole, 49 AMP-activated protein kinase, 49, 322, 323, 325, 331 AMP-activated protein kinase $\alpha 1$, 49 amylase, 322 amyloid beta, 292 amyloid precursor proteins, 164 amyloidogenesis, 77, 81 amyotrophic lateral sclerosis, 40, 41, 42, 45, 138, 141, 207, 208, 381 androgen, 300, 319 anemia, 39, 44, 59, 67, 70, 87, 103, 107, 108, 109, 129, 138, 169, 223, 224, 266, 273, 279 angiogenesis, 64, 114, 116, 136, 168, 171, 175, 222, 226, 231, 233, 251, 259, 268, 271, 275, 287, 290, 307, 319, 345, 358, 364, 371, 372, 373, 383, 388 angiography, 365 angioma, 168 angioplasty, 41, 45, 365 angiotensin II, 80, 113, 118, 162, 242, 255, 349, 386

anthracycline, 187 antiangiogenesis, 281, 284, 300, 303, 326, 360, 367, 369, 372 antibiotic, 212, 273 antibiotic resistance, 212 antibody, 116, 260 anti-cancer, 194, 257, 333 anticancer activity, 273 anticancer drug, 5, 38, 51, 66, 94, 97, 98, 113, 114, 116, 118, 138, 166, 168, 183, 228, 229, 358, 383 anticancer drugs, 66, 94, 97, 98, 138, 168, 183, 383 antigen, 120, 167, 175, 235, 328, 334 antigenicity, 339, 340 antigen-presenting cells, 167, 175 antimalarial guinolines, 182, 191 antimicrobial activity, 3, 17, 379 antioxidant, 1, 6, 14, 17, 21, 23, 25, 26, 27, 28, 29, 30, 31, 35, 36, 40, 44, 47, 49, 52, 53, 54, 59, 61, 62, 66, 69, 73, 77, 78, 80, 82, 87, 88, 93, 100, 108, 112, 114, 115, 117, 119, 123, 125, 126, 128, 129, 130, 131, 132, 133, 135, 136, 139, 143, 144, 145, 151, 155, 156, 159, 161, 181, 184, 186, 188, 191, 193, 197, 201, 202, 209, 210, 213, 219, 225, 226, 229, 230, 233, 236, 239, 240, 241, 242, 245, 246, 247, 248, 249, 250, 251, 252, 253, 254, 255, 258, 259, 260, 263, 265, 266, 267, 269, 270, 271, 272, 274, 278, 279, 283, 286, 287, 288, 289, 290, 291, 294, 299, 300, 302, 307, 308, 309, 311, 312, 318, 319, 323, 324, 329, 330, 331, 332, 333, 335, 337, 339, 342, 343, 344, 345, 346, 347, 348, 350, 353, 360, 361, 365, 367, 368, 369, 372, 375, 376, 377, 380, 381, 382, 383, 384, 385, 387 antioxidant enzyme mimetics, 26, 36, 337, 339, 342, 343, 344, 345, 346 antioxidant gene regulation, 23 antioxidant response element, 24, 49, 94, 144, 191, 219, 230, 307 antioxidative activity, 28 antiplatelet, 321, 360 antitumor, 51, 56, 268, 271, 345 anti-type II collagen monoclonal antibody, 116 antiviral therapy, 70, 364 anxiety, 87, 89, 207, 208, 215 anxiety disorder, 208 AP-1, 23, 30, 35, 61, 75, 93, 105, 112, 113, 126, 144, 146, 147, 161, 200, 211 AP-2, 35, 61, 113 ApoB, 38, 43 ApoE, 64, 114, 126, 152, 201, 202, 212, 248, 343, 355 ApoE-deficient mice, 64, 114, 127, 201, 202, 343, 355

apolipoprotein A-I, 152, 156, 204, 210

apolipoprotein E, 54, 64, 69, 80, 81, 211, 212, 275, 347.361 apoptosis, 51, 55, 56, 77, 81, 82, 99, 105, 107, 108, 112, 114, 116, 118, 120, 121, 126, 130, 132, 133, 137, 140, 145, 152, 162, 163, 164, 165, 167, 168, 172, 173, 174, 175, 181, 183, 185, 190, 192, 221, 222, 265, 271, 276, 281, 284, 287, 288, 290, 295, 300, 302, 303, 307, 310, 319, 326, 349, 364, 371 apples, 317 arachidonic acid, 199 ARE, 94, 181, 219, 225 Arg, 54, 75, 80, 205, 214 Arg139Trp, 187, 194 arginine, 71, 187, 205, 359, 365 aromatic esters, 199 aromatic hydrocarbons, 215 aromatic rings, 315, 316 arrhythmia, 321 arsenic, 140 arsenicals, 228, 235 arterial hypertension, 355, 361 arteriogenesis, 114, 118, 226, 233 arteriosclerosis, 100 artery, 18, 79, 83, 163, 171, 172, 184, 188, 194, 213, 214, 236, 269, 350, 355, 359 arthritis, 16, 52, 116, 121, 134, 169, 176, 177, 185, 287, 294, 327, 333, 346, 350, 358, 383, 388 ARVCF, 138, 140 aryl hydrocarbon receptor, 211 arylhydrocarbon receptor, 181, 191, 200 asbestos, 6, 38, 304 Asc⁻⁻, 267 AscH⁻, 267 ascorbate, 62, 63, 104, 136, 218, 245, 265, 267, 274, 275, 277, 278 ascorbate monoanion, 267 ascorbate radical, 267 ascorbic acid, 69, 265, 267, 274, 275, 276, 277, 278, 279, 292, 388 ascorbyl stearate, 268, 277 aspirin, 200, 210, 270, 276 assessment, 68, 138, 204, 343, 346, 360, 375, 384 astaxanthin, 297, 299, 300, 301, 302, 304, 306, 307, 308, 309, 310, 311, 312 asthma, 52, 56, 65, 96, 98, 100, 102, 115, 120, 127, 133, 189, 195, 227, 242, 243, 269, 273, 279, 285, 292, 293, 301, 305, 308, 312, 324, 332, 344, 356, 362, 381 astrocyte, 153 ataxia, 50, 55, 138, 231, 246, 247, 283, 284, 285, 287, 291, 345 ataxia telangiectasia, 345 ataxia with vitamin E deficiency, 283, 287

atherogenesis, 64, 81, 95, 127, 152, 171, 184, 188, 201, 212, 242, 307, 343 atherosclerosis, 13, 36, 49, 54, 64, 68, 69, 76, 79, 80, 95, 98, 113, 126, 131, 138, 152, 162, 197, 201, 202, 204, 206, 211, 212, 215, 217, 221, 229, 236, 242, 246, 248, 258, 268, 284, 291, 300, 304, 307, 321, 322, 330, 355, 361 atherosclerotic diet, 355 atherosclerotic lesion, 87, 89, 114, 118, 201, 211, 269, 343 atherosclerotic plaque, 140, 188, 202, 212, 268, 275, 355 atherosclerotic vascular disease, 204, 208 athletes, 312 Atm-deficient mice, 345, 350, 358 atopic dermatitis, 175 ATP, 60, 61, 143, 144, 146, 245, 247, 324 ATP hydrolysis, 60 atrophy, 39, 44, 96, 246 atypical 2-Cys Prx, 124 auranofin, 137, 140 aurothioglucose, 137 autoimmune diseases, 134 autoimmune encephalomyelitis, 227, 234 autoimmune myocarditis, 113, 119 autoimmunity, 37, 129, 183, 185, 193, 223, 345, 369 auto-oxidation, 218, 319 autosomal dominant, 221, 232 autosomal recessive, 287 AVED, 287, 288 axons, 288 azo dyes, 180 AB peptides, 164

B

bacteria, 5, 7, 199, 203, 270 bacterial infection, 66 bacterial peritonitis, 252 bacterium, 203 BALT, 185, 193 BAS03551158, 201 base modifications, 12, 13, 14 base pair, 154 basophils, 184, 381 B-cell lymphomas, 184, 192 **BCNU**, 86 beef, 244, 257 behavioral change, 189 behavioral disorders, 226 behaviors, 227, 233, 269, 360, 379 bell peppers, 265

beneficial effect, 39, 41, 45, 53, 66, 76, 95, 96, 103, 116, 127, 153, 164, 165, 221, 227, 241, 242, 243, 244, 246, 247, 249, 251, 255, 265, 267, 268, 269, 270, 272, 273, 274, 278, 281, 287, 288, 289, 290, 297, 300, 301, 302, 304, 305, 320, 321, 324, 325, 326, 327, 334, 343, 344, 346, 355, 356, 359, 360, 369, 370, 371, 372, 380, 382 benzene, 116, 121, 183, 187, 192 benzo(a)pyrene, 183, 192 benzoguinone, 244 β -amyloid peptide, 269, 281 β-amyloid toxicity, 153 β-carotene, 13, 297, 298, 299, 300, 301, 302, 303, 304, 305, 306 β-carotene 15,15'-monooxygenase-1, 301 β-catenin/TCF, 75 β-cryptoxanthin, 297, 298, 299 β-globin gene, 219 β -lapachone, 184 β -phenylethylamine, 48 bicarbonate, 35, 360, 365 bile, 200, 240, 257 bile acids, 200 bilirubin, 12, 22, 31, 159, 160, 161, 164, 166, 169, 170, 231, 239, 240, 241, 242, 243, 244, 255, 256, 257 biliverdin, 160, 161, 166, 174, 175, 176, 239, 240, 241, 242, 243, 244, 255, 256 biliverdin reductase, 160, 239, 240, 255 bioactivation, 62, 66, 69, 94, 97, 99, 183, 302, 327, 357 bioavailability, 36, 41, 53, 268, 285, 300, 317, 319, 321, 329, 339, 340, 354, 377 biocompatibility, 379 biofilm, 203, 212 bioflavonoids, 334 biological activity, 201 biological processes, 59 biological systems, 3, 4, 6, 7, 8, 9, 11, 12, 13, 14, 15, 53, 95, 137, 161, 182, 201, 240, 284, 343, 367, 368, 369, 377, 379, 383, 384 biomarkers, 117, 134, 138, 190, 194, 274, 288, 289, 290, 299 biorhythm, 186 biorhythm synchronization, 186 biosynthesis, 60, 61, 107, 186, 244, 245, 247, 257, 265, 284, 353, 360 biotransformation, 100, 302 biphenols, 315 bismuth nitrate, 228 black tea, 329 bleomycin, 37, 120, 165, 173, 242, 256, 301, 308, 324, 332, 356, 362

blindness, 382, 387

- blood cardioplegia, 359 blood flow, 163, 356, 361
- blood pressure, 64, 71, 117, 122, 162, 171, 250, 311,
- 333, 347, 359, 365
- blood vessels, 64, 168, 266, 291
- blood-brain barrier, 37, 323
- blood-spinal cord barrier, 163, 172
- body fluid, 252
- body weight, 273, 344
- bone, 88, 183, 184, 185, 191, 192, 222, 229, 236, 250, 329, 334, 335
- bone marrow, 86, 88, 183, 184, 185, 191, 192, 222, 229, 236, 250
- bone marrow depression, 229
- bone marrow toxicity, 183
- bone resorption, 334
- bowel, 16, 77, 166, 189, 228, 235, 252, 302, 325, 356
- brain, 37, 43, 50, 55, 57, 64, 69, 77, 81, 87, 114, 119, 127, 132, 134, 135, 138, 139, 140, 141, 150, 153, 154, 157, 158, 159, 163, 164, 172, 173, 209, 215, 221, 223, 224, 227, 233, 234, 246, 249, 250, 253, 258, 260, 269, 275, 276, 285, 292, 303, 308, 323, 324, 331, 332, 344, 346, 348, 351, 356, 361, 362, 374, 384, 386 brain cancer, 209, 303 brain damage, 114, 119, 260
- brain tumor, 57, 215
- breast cancer, 79, 84, 117, 128, 129, 133, 184, 187, 192, 194, 209, 215, 289, 333, 345, 363, 383
- breast carcinogenesis, 184, 187
- breast carcinoma, 39, 44, 187 broccoli. 265
- bromide, 13
- bronchial asthma, 56
- bronchial epithelial cells, 356, 362
- bronchial-associated lymphoid tissue, 185
- bronchoconstriction, 312
- broncoconstriction, 370
- brussels sprouts, 265
- BSO, 63, 64, 65, 66
- buckminsterfullerene, 379, 385
- buckyball (C₆₀), 379
- burn, 252
- butein, 320
- buthionine sulfoximine, 70
- bypass graft, 188, 359

С

C isoform, 206 C/EBP-epsilon, 75 C465T. 187 C609T, 186, 194 C60OH24, 381 Ca^{2+} , 322 c-Abl, 49, 54, 75, 80 cadaver, 45 cadmium, 139, 217, 224, 227, 228, 235, 237 Caenorhabditis elegans, 151, 156, 383, 387 caffeic acid, 315, 316 calcification, 83 calcineurin-NFAT signaling, 107, 109 calcium, 17, 151, 156, 330 calcium phospholipid-binding protein, 151 caloric restriction, 140 calorie, 152, 320, 330 calorie restriction, 152 cancer, 16, 26, 30, 33, 38, 39, 40, 44, 47, 52, 53, 66, 68, 74, 78, 79, 83, 84, 91, 95, 97, 98, 99, 101, 103, 107, 111, 113, 116, 117, 118, 121, 122, 128, 129, 133, 134, 135, 138, 140, 145, 149, 154, 155, 159, 168, 169, 170, 175, 179, 184, 187, 188, 192, 193, 194, 208, 209, 215, 217, 222, 229, 230, 236, 243, 244, 246, 247, 251, 257, 260, 265, 267, 271, 272, 273, 274, 275, 277, 278, 281, 286, 287, 288, 294, 295, 297, 303, 304, 305, 306, 311, 326, 327, 328, 330, 333, 334, 339, 345, 346, 350, 358, 360, 363, 371, 375, 377, 383, 386, 387, 388 cancer cells, 44, 78, 97, 129, 138, 140, 145, 168, 187, 193, 222, 271, 272, 275, 277, 326 cancer chemoprotection, 26, 326, 358 cancer chemotherapy, 98, 194 cancer drug resistance, 98 cancer intervention, 138, 288, 304, 305, 328 cancer progression, 187 capillary, 349 capsule, 74, 75 carbohydrate, 322, 323, 331 carbohydrate metabolism, 323, 331 carbohydrates, 13, 322 carbon, 10, 11, 38, 50, 96, 159, 160, 161, 165, 166, 171, 173, 174, 176, 202, 227, 235, 239, 241, 251, 256, 270, 293, 298, 316, 356, 368, 370, 377, 379, 380, 382, 385 carbon atoms, 316, 379 carbon dioxide, 10, 11 carbon monoxide, 159, 160, 161, 165, 166, 171, 173, 174, 176, 239, 241, 256 carbon nanotubes, 377 carbon tetrachloride, 50, 96, 166, 202, 227, 235, 270, 293, 356, 370, 382 carbonate radical, 10, 35 carbon-centered radical, 11, 368 carboxyfullerene, 348, 381, 382, 386, 387

carcinogen, 67, 192, 303, 327, 358 carcinogen-DNA adducts, 303, 327, 358 carcinogenesis, 16, 25, 39, 51, 66, 78, 83, 94, 97, 98, 116, 118, 121, 128, 129, 135, 138, 145, 154, 168, 170, 179, 181, 183, 184, 187, 191, 192, 208, 222, 229, 236, 243, 303, 304, 307, 311, 326, 327, 328, 345, 350, 358, 360, 375 carcinogenicity, 236 carcinogens, 66, 94, 97, 98, 168, 183, 229, 303, 327, 345, 358 carcinoma, 30, 39, 44, 45, 78, 79, 82, 130, 134, 154, 158, 168, 170, 177, 187, 209, 229, 236, 277, 309, 358, 371, 383, 388 carcinomas, 128, 375, 387 cardiac aging, 49, 54 cardiac arrhythmia, 64, 321, 369 cardiac arrhythmias, 64, 369 cardiac contractile dysfunction, 226, 233 cardiac fibroblasts, 126, 131 cardiac hypertrophy, 36, 49, 103, 107, 108, 109, 114, 118, 300, 307, 321, 322, 331, 355 cardiac ischemia, 156, 220 cardiac surgery, 252, 259 cardiac transplantation, 162 cardiofaciocutaneous syndrome, 247 cardiolipin, 251 cardiomyocyte, 54, 86, 107, 108, 109, 127, 152, 220, 231, 386 cardiomyopathy, 36, 37, 38, 43, 49, 50, 64, 114, 129, 138, 226, 227, 233, 287, 321, 330, 355, 380, 386 cardioplegia, 359, 364 cardiopulmonary bypass, 194, 252, 259 cardiotoxicity, 5, 49, 54, 95, 103, 107, 113, 119, 226, 233, 246, 300, 307, 321, 355, 381, 386 cardiovascular cells, 95, 184, 190, 221, 242, 380 cardiovascular disease, 30, 33, 36, 40, 42, 64, 79, 107, 111, 158, 159, 162, 169, 171, 187, 206, 211, 224, 232, 248, 251, 257, 261, 265, 271, 272, 274, 278, 284, 288, 291, 294, 297, 304, 306, 307, 311, 313, 319, 326, 327, 330, 333, 339, 343, 347, 355, 359, 360, 369, 377 cardiovascular diseases, 33, 36, 40, 64, 79, 107, 111, 159, 162, 169, 187, 206, 211, 224, 248, 251, 265, 271, 272, 274, 284, 288, 294, 297, 304, 306, 319, 326, 327, 330, 343, 347, 355, 359, 360, 369, 377 cardiovascular function, 321 cardiovascular risk, 87, 89, 207, 213, 232 cardiovascular system, 95, 126, 171, 184, 330 carnitine, 266 carotene, 13, 295, 297, 299, 300, 301, 302, 303, 304, 305, 306, 307, 309, 310, 311, 312, 313 carotenes, 298

carotenoids, 13, 263, 297, 298, 299, 300, 301, 302, 303, 304, 305, 306, 307, 308, 311, 312, 327 carotid artery, 184, 188, 194, 236 carrot, 297 cartilage, 364 case study, 89, 108 caspase-2, 39 catabolism, 80, 98, 159, 163, 166 catalase, 6, 22, 28, 30, 31, 38, 45, 47, 48, 49, 50, 51, 52, 53, 54, 55, 56, 82, 87, 125, 128, 154, 337, 339, 340, 341, 342, 343, 344, 345, 348, 349, 351, 378, 379, 384 catalase mimetics, 50, 55, 348 catalysis, 3, 92, 123, 151, 378 catalytic activity, 124, 150, 385 cataract, 37, 52, 67, 78, 83, 85, 87, 107, 153, 157. 223, 287, 301, 302, 305, 308, 310, 311, 355, 358, 364 catechin, 321, 330 catecholamines, 218 cation, 284 cationized catalase, 51, 56 causal relationship, 15, 40, 52, 106, 117 caveolae, 159 CCAAT-enhancer-binding protein, 35 CD4⁺/CD25⁺ regulatory T cells, 159 cDNA, 88, 173, 186, 187 CdTe quantum dots, 6 ceftazidima, 203 celecoxib, 286, 294 cell culture, 36, 43, 64, 65, 78, 88, 116, 129, 138, 168, 182, 185, 188, 230, 241, 246, 260, 318, 319, 326, 330, 337, 380, 383 cell cycle, 271, 287, 326 cell cycle arrest, 271 cell death, 14, 43, 81, 108, 116, 129, 132, 133, 153, 156, 166, 173, 193, 202, 221, 222, 231, 255, 277, 290, 358, 382 cell dysfunction, 14, 162, 322 cell line, 140, 145, 155, 169, 171, 186, 189, 200 cell lines, 140, 145, 171, 186, 200 cell membranes, 240 cell migration, 184, 310 cell proliferation, 51, 95, 97, 100, 109, 116, 126, 130, 136, 143, 145, 147, 168, 184, 193, 228, 242, 255, 290, 319 cell signaling, 17, 18, 94, 97, 99, 103, 105, 106, 108, 111, 123, 130, 281, 282, 285, 290, 299, 302, 306, 326, 329 central adiposity, 155 central nervous system, 127, 132, 185, 246, 324 CeO₂, 378, 384 cerebellar ataxia, 246, 247

cerebellar atrophy, 246 cerebellar hypoplasia, 138 cerebellum, 159 cerebral ischemia-reperfusion injury, 76, 227, 301, 381 cerebrovascular events, 346 cerium, 6, 22, 337, 340, 377, 378, 380, 381, 382, 383, 384, 385, 386, 387 cerium oxide nanoparticles, 6, 22, 337, 340, 377, 378, 380, 381, 382, 383, 384, 385, 386, 387 cerulean, 114 cGPx, 74 chelates, 250, 252 chemical carcinogenesis, 25, 66, 97, 168, 229, 326, 327, 345, 358 chemical properties, 6, 7, 266 chemical reactions, 21 chemical reactivity, 7, 17 chemical stress, 95 chemical structures, 340 chemokine, 82, 114, 119 chemoprevention, 30, 350, 364 chemoprotection, 97, 101 chemoprotective agents, 25, 26, 27, 28, 93, 184 chemotaxis, 114, 119, 121 chemotherapy, 98, 168, 187, 194, 257, 278, 345 childhood, 102, 195 chiral center, 281, 282 chloroamine, 13 chlorpyrifos-oxon, 199, 204 cholesterol, 18, 87, 101, 152, 156, 200, 201, 211, 284, 285, 292, 311, 321, 330 choline-deficient diet, 371 cholinergic, 214, 269 cholinesterase, 207, 214 cholinesterase inhibitors, 207, 214 chondrocyte, 364 chromatography, 191, 232 chromosome, 48, 60, 74, 85, 138, 140, 143, 154, 158, 198, 224 chronic diseases, 259 chronic hepatitis, 208, 215 chronic liver disease, 202, 208 chronic obstructive pulmonary disease, 59, 68, 98, 324, 353, 355, 356, 359, 362 chronic renal failure, 249, 258, 345, 349 cigarette smoke, 77, 82, 120, 145, 146, 184, 269, 276, 285, 292, 301, 324, 332, 360, 363, 382, 387 ciliary body, 250 circadian rhythm, 250 circadian rhythms, 250 circulation, 36, 98, 197, 222, 243, 282, 305 cirrhosis, 45, 174, 208, 212

community, 189, 260, 312

cis-acting element, 23 cisplatin, 38, 44, 51, 56, 66, 97, 101, 145, 166, 175, 227, 228, 229, 235, 236, 243, 257, 270, 302, 310 citrus fruits, 265, 317 clarithromycin, 273, 279 cleavage, 9, 239 clinical symptoms, 87 clinical trials, 28, 42, 53, 67, 68, 76, 245, 246, 247, 252, 255, 265, 271, 272, 273, 274, 281, 287, 288, 289, 290, 297, 306, 315, 327, 328, 329, 337, 346, 353, 355, 357, 359, 360, 367, 370, 373, 377, 384 cloning, 86 clozapine, 189, 196 clusters, 8, 109, 219 c-Myc, 61 CO, 171, 255 CO₂, 10 CO3^{-,} 10 cobalt, 162, 167 cobalt protoporphyrin, 162, 167 cocoa, 327, 333, 334 cocoa polyphenols, 327 coding, 40, 205, 229, 237 coenzyme, 31, 136, 181, 239, 244, 245, 257, 267, 284 coenzyme O, 31, 136, 181, 239, 244, 245, 257, 267 coenzyme Q radical, 245, 267 coenzyme Q₁₀, 244, 257 coffee, 317, 321, 330 cognition, 275, 348, 387 cognitive decline, 39, 189 cognitive deficit, 323, 348 cognitive deficits, 323, 348 cognitive dysfunction, 127 cognitive function, 132, 221 cognitive impairment, 254, 260, 382 co-immunoprecipitation, 143 colitis, 38, 44, 50, 66, 70, 77, 82, 115, 121, 165, 166, 174, 195, 242, 256, 259, 286, 289, 293, 295, 302, 309, 344, 349, 357, 358, 360, 363, 365, 370, 375 collagen, 116, 121, 185, 266, 268, 269, 294, 350 collagen synthesis, 266 collagenase, 78, 82 collagen-induced arthritis, 185, 294, 350 collateral, 226 colon, 70, 97, 101, 117, 184, 187, 192, 194, 307, 382 colon cancer, 187, 194 colon carcinogenesis, 184, 192, 307 color, 158, 297, 302, 305 colorectal adenoma, 138, 141, 328, 334 colorectal cancer, 83, 84, 117, 122, 187, 375, 386 combination therapy, 289 combined effect, 53, 306

complement, 231 complex interactions, 326 complications, 16, 36, 50, 101, 114, 162, 188, 202, 207, 227, 229, 236, 243, 249, 269, 285, 301, 313, 322, 343, 355, 359, 365 composition, 161, 219, 298 compound I, 48 compounds, 5, 7, 21, 25, 26, 27, 28, 29, 36, 68, 69, 87, 93, 94, 99, 123, 138, 140, 151, 168, 180, 181, 182, 183, 199, 200, 219, 239, 246, 248, 254, 257, 259, 263, 270, 271, 297, 299, 306, 315, 316, 317, 318, 319, 320, 321, 322, 323, 324, 325, 326, 327, 328, 329, 331, 337, 339, 340, 341, 342, 343, 346, 353, 355, 367, 368, 369, 371, 373, 377, 381 computed tomography, 83 COMT, 138, 140 configuration, 281 conjugation, 62, 91, 92, 93, 94, 96, 97, 180, 317, 357 connexin 43, 303 consensus, 30 consumption, 83, 128, 133, 272, 288, 304, 315, 327, 328, 330, 333, 355 contact dermatitis, 66, 115, 167, 286, 294 contrast medium-induced nephropathy, 353, 357, 359, 365 controlled trials, 42, 247, 254, 272, 273, 278, 288, 289, 297, 304, 305, 306, 333, 334, 335, 359 controversies, 66, 76, 111, 116, 288, 290 convulsion, 65 coordination, 340, 345 COPD, 324, 332, 356, 359, 362 copper, 3, 6, 33, 34, 217, 224, 225, 226, 228, 252, 283, 319, 323, 345, 350, 370, 375 CoQ₁₀, 244, 245, 246, 247, 257 CoQ10 deficiencies, 245, 246, 257 CoQ10H, 245 CoQ₁₀H₂, 245 COQ3, 244 COQ7, 245 COO8, 244 coronary angioplasty, 41, 45 coronary artery bypass graft, 188, 359 coronary artery disease, 79, 83, 213, 214 coronary heart disease, 98, 101, 155, 188, 194, 206, 207, 209, 214, 243, 244, 304, 327, 333 correlation, 40, 74, 87, 206, 243, 254 cortex, 78 cortical neurons, 386 cosmetics, 284, 286 C-reactive protein, 188, 207, 214, 257 creatinine, 115, 167, 371 crocetin, 300, 301, 308

cross-sectional study, 312 CRP, 214 cryptorchidism, 116, 121 Cu,ZnSOD, 33, 34, 35, 36, 37, 38, 39, 40, 41, 42, 248 Cul3/Rbx1-based E3-ubiquitin ligase complex, 24 culture, 43, 65, 88, 129, 138, 260, 319, 330, 380 culture media, 319 curcumin, 315, 316, 321, 322, 324, 325, 326, 331, 333, 334, 384 cycling, 5, 7, 77, 128, 180, 181 cyclooxygenase, 371 cyclooxygenase-2, 371 cyclooxygenases, 283, 300, 319 cyclophosphamide, 64, 97, 101, 345, 347, 356 cyclosporine, 38, 167, 270, 293, 302, 310, 363 cyclosporine A, 38, 167, 270, 293, 302 Cys311Ser, 205, 214 cysteine, 15, 25, 60, 61, 63, 65, 67, 68, 69, 70, 97, 103, 104, 111, 112, 113, 124, 126, 144, 146, 147, 150, 151, 199, 205, 217, 219, 224, 225, 226, 353, 354, 358, 360, 361, 362, 363, 364, 365 cysteine 284, 199 cysteine S-conjugate β -lyase, 97 cysteine-rich protein, 25 cystic fibrosis, 68, 71, 359, 365 cytochrome, 6, 153, 157, 160, 180, 309 cytochrome P450 reductase, 180 cytochromes, 47 cytocuprein, 33 cytokines, 15, 39, 54, 64, 105, 113, 127, 161, 162, 166, 184, 219, 225, 246, 252, 276, 283, 319, 354, 356, 370, 373, 380 cytoplasm, 60 cytoprotectant, 255 cytoprotection, 30, 88, 161, 186, 240, 246 cytosine, 187 cytosol, 34, 74, 85, 104, 111, 113, 115, 124, 135, 143, 149, 218, 220 cytosolic ferritin, 218 cvtosolic GPx, 74 cytosolic GSTs, 91, 92 cytotoxicity, 78, 87, 94, 97, 163, 168, 183, 184, 192, 193, 229, 242

D

D-29 allele, 189 DAF-16/FOXO, 151, 156 daidzein, 325 deafness, 247 decaprenyl diphosphate, 244 decaprenyl diphosphate synthase, 244 decarboxylation, 244 decomposition, 47, 48, 73, 75, 78, 93, 218, 267, 275, 342 defects, 65, 69, 103, 107, 111, 117, 247 defense mechanisms, 29, 54 deficiencies, 56, 245, 246, 257, 356, 362 deficiency, 6, 37, 38, 43, 44, 47, 51, 52, 54, 55, 62, 65, 67, 69, 71, 74, 78, 83, 85, 86, 87, 88, 89, 96, 97, 98, 100, 102, 106, 131, 137, 138, 152, 153, 167, 169, 172, 175, 176, 183, 187, 192, 193, 202, 211, 212, 219, 223, 229, 231, 234, 236, 260, 266, 268, 269, 270, 274, 276, 283, 284, 285, 287, 288, 289, 291, 292, 299, 351, 365 degeneration, 16, 41, 49, 65, 119, 152, 186, 209, 222, 229, 231, 236, 291, 302, 305, 308, 311, 355, 364, 381, 382 deglutathionylation, 62, 63, 103, 104, 144 degradation, 13, 14, 24, 38, 43, 49, 54, 145, 159, 160, 161, 165, 181, 186, 190, 199, 268 dehydroascorbate, 62, 63, 104, 267, 276 dehydroascorbate reductase, 62, 63 delphinidin, 316 delta state, 4 dendrimers, 377 dendritic cell, 307 deoxycorticosterone acetate, 242 dephosphorylation, 62, 103 depigmentation, 154 DEPMPO, 368 depolarization, 64 deposition, 153, 169, 223 depression, 229 derivatives, 36, 49, 50, 51, 53, 55, 179, 180, 248, 259, 268, 269, 284, 285, 286, 340, 344, 367, 368, 369, 370, 371, 375, 377, 379, 380, 381, 382, 383, 384, 385, 386, 387, 388 dermatitis, 16, 66, 115, 167, 175, 274, 286, 294 desferrioxamine, 270, 276 destruction, 37, 116, 121, 369 detoxification, 11, 13, 15, 26, 28, 53, 62, 79, 85, 87, 91, 93, 94, 95, 96, 98, 99, 112, 126, 180, 188, 195, 199, 240, 241, 267, 300, 303, 327, 356, 357, 358 dexamethasone, 200, 211 dextran sulfate sodium, 302, 309, 370, 375 DHA, 267 diabetes, 17, 33, 36, 40, 43, 45, 47, 50, 52, 53, 54, 55, 56, 59, 64, 68, 69, 71, 76, 78, 79, 81, 83, 88, 91, 98, 101, 102, 103, 107, 111, 114, 115, 117, 119, 122, 127, 155, 162, 167, 172, 176, 187, 188, 194, 195, 197, 201, 202, 207, 209, 212, 214, 217,

224, 227, 229, 232, 233, 236, 242, 243, 244, 246,

disturbances, 215, 355

disulfide bridge, 86, 103, 104, 111, 112, 125, 144,

247, 249, 250, 251, 254, 256, 257, 259, 261, 269, 276, 285, 288, 289, 292, 295, 301, 302, 306, 312, 313, 322, 323, 331, 339, 343, 348, 349, 353, 355, 359, 360, 365, 369, 374 diabetic cardiomyopathy, 43, 50, 114, 227, 233, 330, 355 diabetic complications, 37, 50, 114, 162, 207, 227, 229, 249, 269, 285, 301, 322, 343, 355 diabetic embryopathy, 37 diabetic nephropathy, 51, 78, 82, 115, 119, 162, 166, 167, 257, 270, 286, 293, 302, 309, 325, 345, 357 diabetic neuropathy, 43, 56, 285, 292 diabetic patients, 207, 214 diabetic polyneuropathies, 250 diabetic renal injury, 36 diabetic retinopathy, 302 diacylglycerol, 202, 211 diacylglycerol acyltransferase-1, 202 diagnosis, 291, 377 diagnostic imaging, 377 dialysis, 51 diazoxon, 199, 204 dichloroacetic acid, 95, 100 dicumarol, 182, 185, 186 diet, 21, 22, 26, 77, 127, 131, 156, 188, 201, 202, 211, 222, 226, 227, 233, 234, 263, 266, 282, 292, 297, 298, 299, 302, 304, 306, 307, 308, 315, 320, 323, 327, 329, 330, 331, 332, 355, 356, 361, 371, 375 dietary intake, 271, 274, 283, 286, 299, 301, 303, 304, 305, 316, 318, 327 dietary supplementation, 300, 305, 323, 325, 358 diethyl phosphate, 198 diethyldithiocarbamate, 36, 369, 374 diffusion, 8, 10, 11, 319 digestion, 322 diglucuronide, 240 dihydrolipoic acid, 245, 249, 250, 342 dihydronicotinamide riboside, 179 dilated cardiomyopathy, 36, 37, 38, 129, 138, 380 dimerization, 136 DIO1, 73 DIO2, 73 DIO3, 73 dioxin, 182 disability, 372 disease progression, 15, 260, 287, 321 diselenide, 342, 344 dismutation, 3, 5, 8, 21, 33, 34, 35, 48, 53, 341, 378, 385 disorder, 50, 52, 67, 87, 154, 189, 203, 208, 221, 242, 243, 287, 345, 357, 367 dissociation, 25

199 ditelluride, 342 dithiol Grxs, 104 dithiolethiones, 86, 101, 219, 230 dithiothreitol, 151 DMPO, 369, 373 DNA, 10, 12, 13, 14, 18, 30, 35, 36, 39, 40, 44, 51, 66, 78, 83, 97, 111, 113, 127, 128, 132, 136, 140, 181, 184, 187, 193, 194, 195, 213, 215, 219, 222, 227, 229, 233, 267, 270, 293, 303, 326, 327, 332, 333, 335, 350, 358, 363, 371 DNA adduct, 187, 194 DNA damage, 14, 39, 44, 51, 66, 78, 127, 181, 184, 215, 222, 227, 229, 233, 270, 326, 332, 333, 335, 350, 358, 363, 371 DNA lesions, 184, 193 DNA repair, 136, 332 DNA strand breaks, 13 DNA synthesis, 111 docosahexaenoic acid, 199 dogs, 387 donors, 75, 112, 179 dopamine, 95, 153, 157, 185, 188, 193, 292 dopamine metabolism, 188 dopamine toxicity, 186 dopamine-derived quinone, 95, 186 dopaminergic, 55, 96, 100, 153, 157, 164, 173, 186, 189, 221, 222, 231, 232, 234, 260, 269, 308, 356, 381 dopaminergic midbrain neurons, 221 dopaminergic neuron, 55, 96, 100, 164, 173, 186, 189, 221, 222, 260, 308, 356 dosage, 204, 273, 289 dose-response relationship, 255, 329 dosing, 272 Down's syndrome, 208, 215 doxorubicin, 5, 49, 54, 64, 66, 107, 108, 114, 226, 233, 255, 307, 358, 363, 381, 382, 386, 387 Drosophila, 55, 86, 87, 88, 96, 100, 114, 117, 119, 127, 129, 133, 152, 153, 203, 213, 382 Drosophila melanogaster, 86, 87, 88, 114, 117, 127, 129, 152, 153, 203, 382 drug abuse, 189 drug delivery, 378, 383 drug resistance, 66, 97, 98, 101, 113, 116, 118, 229 drug targets, 350 drug therapy, 168, 207 drugs, 6, 26, 37, 38, 49, 64, 66, 69, 94, 97, 98, 165, 166, 168, 183, 191, 192, 199, 200, 207, 228, 245, 302, 356, 360, 373, 376, 377, 383 DT-diaphorase, 179, 191, 194

duodenum, 268 dusts, 6, 38 dyes, 180 dysarthria, 287 dyslipidemia, 184 dyspepsia, 312 $d-\alpha$ -tocopherol, 282

Е

ebselen, 76, 342, 343, 344, 345, 346, 347, 349, 350 ECSOD, 33, 34, 36, 37, 38, 39, 248 edema, 185, 273, 370 eGPx, 74 eicosanoids, 93 electron, 4, 5, 6, 7, 9, 11, 62, 73, 75, 86, 105, 111, 112, 135, 136, 144, 145, 149, 150, 160, 179, 180, 181, 182, 183, 186, 245, 246, 267, 340, 367, 369, 373 electron paramagnetic resonance, 367, 373 electron transport chain, 5, 7, 145, 182, 245, 246, 369 electrons, 4, 86, 124, 267, 340, 379 electrophiles, 59, 62, 66, 97, 161, 181, 225 electrophilic carcinogens, 303, 327 elongation, 155 embryogenesis, 137, 139 embryonic development, 74, 76, 106, 117, 137 embryonic fibroblasts, 106 embryonic kidney cell, 145 embryonic lethality, 106, 113, 114, 118, 137, 152, 220, 221, 231 emphysema, 115, 120, 301, 308 encephalomyelitis, 227, 234, 242, 253, 260 encephalomyopathy, 246 encephalopathy, 246, 247 encoding, 87, 129, 224, 245, 265, 268 endocrine, 74 endocrine dysfunction, 74 endoplasmic reticulum, 48, 124, 150, 159, 210, 233, 380 endothelial activation, 71, 126, 131, 359, 364, 365 endothelial cells, 54, 139, 193, 222, 274, 319, 386 endothelial dysfunction, 268, 321, 328, 343, 348 endothelial growth factor, 122, 226, 328, 334 endothelial nitric oxide, 285, 319, 330 endothelial sloughing, 242 endothelin receptor, 114, 119 endothelium, 50, 55, 276, 277, 347 endotoxemia, 176, 243, 257, 287, 375 endotoxic shock, 39, 45, 257, 277

endotoxin, 65, 77, 121, 128, 129, 133, 234, 242, 243, 256, 307, 324, 344, 346, 349, 364 energy metabolism, 249, 320, 321, 323, 325, 370 energy transfer, 13 eNOS, 319 environmental factors, 188 environmental tobacco, 195 enzymatic activity, 33, 97, 164, 189, 199, 318 enzyme induction, 101 enzymes, vii, 5, 6, 7, 8, 9, 11, 12, 13, 17, 22, 25, 26, 27, 31, 33, 36, 40, 42, 47, 53, 54, 60, 64, 66, 67, 73, 82, 88, 91, 92, 93, 95, 98, 103, 112, 123, 124, 130, 135, 136, 137, 143, 149, 151, 154, 159, 162, 169, 170, 179, 180, 182, 186, 187, 188, 190, 193, 195, 197, 201, 202, 203, 204, 206, 208, 209, 217, 219, 222, 244, 245, 248, 249, 250, 251, 255, 258, 259, 265, 266, 283, 284, 294, 299, 301, 302, 303, 308, 317, 318, 319, 322, 323, 324, 326, 327, 329, 337, 339, 340, 342, 343, 344, 356, 367, 368, 372, 377 eosinophilia, 293, 308 eosinophils, 184 epidemiological studies, 68, 79, 98, 186, 187, 188, 190, 197, 206, 207, 209, 224, 230, 243, 244, 254, 265, 271, 272, 287, 288, 290, 297, 304, 305, 306, 315, 327, 328, 329 epidermis, 51, 56, 133, 154, 158, 167, 235, 293 epididymal GPx, 74 epididymal lumen, 74, 78 epididymidis, 78 epididymis, 74 epigallocatechin-3-gallate, 316, 317, 319, 324, 326, 328, 329, 331, 333, 335 epigenetic modifications, 151, 156 epigenetic regulation, 35 epigenetic silencing, 35 epirubicin, 187 epistasis, 56 epithelia, 203, 212, 224 epithelial cells, 55, 82, 107, 109, 145, 164, 185, 194, 301, 302, 308, 310, 332, 333, 356, 362 epithelial-mesenchymal transition, 167, 175, 222, 232 epithelium, 74, 153, 157, 250, 382, 387 EPR, 367, 368, 370, 373 equipment, 135 erdosteine, 354, 362 erectile dysfunction, 45, 209, 216, 287 erythrocuprein, 3, 33 erythrocyte, 47, 80, 83, 84, 133 erythrocytes, 44, 56, 67, 74 ES936, 182, 186, 191

Escherichia coli, 104, 108, 111, 118, 213

ester, 64, 65, 66, 144, 146, 162, 354, 355, 356, 361, 362, 363, 364 esterases, 209, 354 estradiol, 247, 248, 258 estriol, 247 estrogen, 183, 184, 187, 192, 194, 199, 247, 248, 258, 319, 333 estrogen carcinogenesis, 183 estrogen quinones, 187 estrogen receptor, 248, 258 estrogens, 183, 239, 247, 248, 249, 254 estrone, 247 ethanol, 49, 50, 65, 77, 96, 115, 120, 128, 133, 252, 270, 356 ethanol-induced liver injury, 50, 252 ethyl pyruvate, 251, 252, 259 etiology, 358 EUK series, 340 EUK-134, 340, 344, 346, 349, 351 EUK-189, 340, 345 EUK-207, 340 EUK-418, 340 EUK-423, 340 EUK-8, 340, 344, 379 eukaryotic cell, 143 eukaryotic translation initiation factor, 181, 190 excision, 326, 332 excitation, vii, 3, 12, 13 excited nitric dioxide, 10 excitotoxicity, 164, 173, 227, 370, 374 excretion, 180, 194 exencephaly, 114, 118 exercise, 52, 56, 254, 260, 305, 312 exercise performance, 52, 56 exons, 161, 169 experimental autoimmune encephalomyelitis, 234, 242 experimental condition, 39, 105, 124, 135, 151, 268, 303, 321, 346, 381 exploration, 254 exposure, 6, 17, 64, 77, 86, 120, 128, 163, 185, 187, 195, 213, 228, 234, 237, 308, 350, 370, 382 extracellular matrix, 307 extracellular space, 34, 111, 113, 124 extracellular superoxide dismutase, 44, 45 eye diseases, 311

F

F₂-isoprostane, 185 FAD, 86, 87, 136 falciparum malaria, 89 familial amyotrophic lateral sclerosis, 40, 41, 42, 45, 138, 141, 381 farnesyl diphosphate, 244 Fas-Fas ligand, 165, 173 fasting, 172 fat, 127, 155, 158, 173, 201, 211, 226, 227, 233, 234, 297, 308, 323, 331, 332, 356 fatty liver, 16, 38, 43, 66, 184, 224, 232, 307, 325, 332, 356 fava beans, 87 favism, 85, 87 Fenton reaction, 6, 8, 9, 161, 220 Fenton-type reaction, 6, 9, 217, 226, 378 ferric ion, 268 ferrioxidase, 218, 222 ferriprotoporphyrin IX, 48 ferritin, 22, 25, 31, 161, 217, 218, 219, 220, 221, 222, 223, 224, 230, 231, 232, 256 ferritin heavy subunit, 25, 218, 219, 220, 221, 222, 223 ferritin light subunit, 219, 221, 222, 223 ferritin nanocage, 218 ferrous ion, 160, 239, 268 fertility, 37, 75 FeSOD, 34 FeTMPS, 342 FeTMPvP, 342 FeTPPS, 342 fetus, 37 fibers, 6, 246 fibrinolysis, 173 fibroblasts, 78, 82, 106, 116, 121, 126, 131, 156 fibrosis, 43, 51, 55, 56, 68, 71, 114, 120, 165, 167, 173, 175, 212, 228, 233, 235, 242, 249, 256, 258, 259, 270, 276, 292, 301, 308, 324, 332, 344, 348, 356, 359, 362, 365, 366 filament, 172 first-pass effect, 377 fisetin, 320 flaky skin mouse model, 326, 333 flavonoids, 29, 312, 315, 316, 325, 326, 328, 329, 331, 333 flavoprotein, 6, 85 flavoprotein dehydrogenase, 6 fluid, 75, 115, 267, 273, 276, 278 foam cell, 202, 321 folate, 260 follicles, 154, 326, 333 forkhead transcription factor, 49, 125 free radical, 3, 4, 7, 10, 11, 12, 13, 34, 38, 94, 212, 218, 254, 367, 368, 369, 370, 372, 373, 374, 379, 380, 381, 382, 384, 385, 386

free radicals, 3, 4, 12, 17, 218, 367, 368, 369, 370, 373, 379, 380 frequency distribution, 169 fruits, 265, 271, 297, 299, 315, 317, 327, 329 FT-Raman spectroscopy, 154 fucoxanthin, 301, 308 fudosteine, 354 fullerene, 6, 270, 277, 340, 344, 347, 377, 379, 380, 381, 382, 383, 384, 385, 386, 387, 388 fullerenols, 379 fulminant hepatitis, 370, 375 fusion, 37, 43

G

GABAergic, 269 gait, 153 γ -glutamyl transpeptidase, 97 y-glutamylcysteine, 67 γ -glutamylcysteine ligase, 25, 59, 60, 61, 87 y-glutamylcysteine synthetase, 60 γ -glutamylcysteinylglycine, 60 γ -irradiation, 229, 326 γ-tocopherol, 282, 285, 286, 287, 288, 289 gamma-tocopherol, 293, 295 gangrene, 47, 53 gap junctional communication, 303 gastric adenocarcinoma, 170, 177 gastric cancer, 188 gastric mucosa, 38, 115, 120, 270, 286, 293, 349 gastric noncardia adenocarcinoma, 289 gastritis, 115, 121, 273 gastrointestinal diseases, 165, 302, 332, 345 gastrointestinal GPx, 74 gastrointestinal stromal tumor, 170 gastrointestinal tract, 74, 77, 325 GCL, 60, 61, 63, 64, 67 GCLC, 60, 61, 63, 65, 67 GCLM, 60, 61, 63, 64, 65, 67 GCS, 60 gene delivery, 49, 114, 118, 126, 176 gene expression, 23, 24, 25, 29, 35, 49, 61, 75, 76, 87, 93, 125, 131, 137, 151, 154, 155, 161, 181, 182, 185, 189, 191, 193, 194, 200, 210, 211, 225, 234, 235, 257, 258, 284, 300, 330, 360, 369, 370, 374 gene knockdown, 200 gene knockout, 36, 37, 39, 47, 49, 51, 63, 64, 73, 76, 79, 86, 88, 95, 96, 97, 106, 113, 126, 137, 139, 151, 162, 164, 169, 185, 200, 204, 206, 225 gene mutations, 67, 88, 243

gene polymorphism, 40, 42, 47, 52, 53, 57, 67, 79, 91, 98, 99, 107, 117, 130, 138, 158, 169, 170, 179, 184, 186, 187, 188, 189, 190, 204, 205, 206, 207, 208, 213, 214, 215, 236, 237, 243 gene promoter, 151, 169, 176, 177, 181, 182, 189, 191, 193 gene therapy, 28, 30, 40, 42, 106, 108, 119, 235 gene transfer, 39, 128, 162, 164, 165, 166, 173, 174, 200, 201, 202, 211, 228 genes, 23, 24, 25, 28, 29, 30, 34, 35, 36, 40, 42, 43, 54, 56, 60, 67, 71, 74, 76, 82, 83, 92, 94, 117, 118, 135, 137, 141, 150, 161, 177, 179, 186, 188, 193, 194, 195, 213, 214, 218, 219, 223, 224, 225, 226, 227, 228, 229, 233, 236, 245, 246, 257, 258, 291, 332 genetic alteration, 128 genetic defect, 247, 283 genetic disorders, 217 genetic manipulations, 343 genetic variation, 40, 52, 99, 117, 118, 139, 153, 206, 223, 224, 229, 243, 287 genistein, 315, 316, 317 genome, 36, 155 genotoxicity, 183, 184 genotype, 83, 89, 98, 101, 102, 177, 186, 187, 188, 189, 194, 206, 207, 208, 215, 288, 295 gentamicin, 203, 302, 310 germline, 36 gestational diabetes, 224, 232 GI-GPx, 74, 80 gland, 250, 269 glial cells, 127 glioblastoma, 277 glioblastoma multiforme, 277 glioma, 371, 373, 375 gliomas, 188 Gln192Arg, 205, 214, 216 global ischemia, 64, 127 globins, 47 glucocorticoid, 161, 200, 225 glucocorticoid response element, 161, 200, 225 glucokinase, 323 gluconeogenetic enzyme, 323 glucose, 64, 69, 76, 86, 127, 132, 172, 184, 188, 200, 202, 207, 210, 212, 214, 232, 269, 275, 301, 302, 310, 322, 323, 334, 345, 355, 359, 361, 365 glucose homeostasis, 127, 269, 275 glucose intolerance, 127, 184, 345 glucose regulation, 323 glucose tolerance, 132, 188 glucose transporter GLUT2, 322 glucose-6-phosphate dehydrogenase, 86 glucoside, 276

glucuronidation, 240, 243 glutamate, 30, 59, 60, 68, 69, 70, 71 glutamatergic, 164, 269 glutamine, 123, 205 glutamine synthase, 123 glutaredoxin, 62, 75, 103, 108, 109 glutathione, 11, 12, 13, 22, 23, 25, 28, 31, 47, 50, 55, 59, 60, 61, 62, 63, 68, 69, 70, 71, 73, 75, 80, 81, 82, 83, 84, 85, 86, 87, 88, 89, 91, 92, 93, 94, 99, 100, 101, 102, 103, 104, 105, 108, 112, 123, 128, 135, 136, 139, 140, 143, 144, 151, 195, 199, 239, 240, 248, 249, 255, 267, 276, 332, 337, 339, 342, 347, 353, 361, 362, 363, 364, 365 glutathione conjugation, 94 glutathione disulfide, 85, 86, 93, 105, 136 glutathione peroxidase, 11, 47, 55, 62, 73, 80, 81, 82, 83, 84, 87, 88, 93, 123, 128, 139, 339, 342, 347 glutathione precursors, 337 glutathione reductase, 88, 89, 105, 135, 136 glutathione S-transferase, 28, 62, 87, 88, 92, 99, 100, 101, 102, 195 glutathione synthetase, 59, 60, 61, 70, 87 glutathione transferase, 91, 99, 100, 101 glutathionylated proteins, 103, 104 glutathionylation, 95, 103, 104, 113, 146 glycerol, 293 glycine, 59, 60, 67 glycogenesis, 323 glycol, 41, 165, 173 glycolysis, 251 glycosylation, 137 gold compound, 140 gout, 253 gp91^{phox}, 5 GPx, 73, 74, 75, 76, 77, 78, 79, 80, 83, 339, 341, 342, 343, 344, 346 GPx mimetics, 341, 342 GPx1, 73, 74, 75, 76, 77, 78, 79 GPx2, 74, 75, 77, 78 GPx3, 74, 75, 76, 77, 78, 79 GPx4, 74, 75, 76, 77, 78, 80, 81, 82 GPx5, 73, 74, 75, 78, 79 GPx6, 73, 74, 75 GR, 85, 86, 87, 88 Gr1^{a1Neu}, 86, 87, 88 graft dysfunction, 38 granulocytes, 184 growth factor, 122, 126, 127, 165, 167, 222, 226, 277, 300, 328, 334 growth rate, 252 Grx, 103, 104, 105, 106, 107 Grx1, 104, 106, 107 Grx2, 104, 107

Grx3, 104, 106, 107 Grx4, 104 Grx5, 103, 104, 107, 108 GSH, 59, 60, 61, 62, 63, 64, 65, 66, 67, 68, 69, 70, 73, 75, 77, 85, 86, 87, 88, 91, 92, 93, 94, 96, 97, 103, 104, 105, 124, 136, 144, 353, 354, 355, 356, 357, 358, 359, 360, 364 GSH diethyl ester, 354 GSH ethyl ester, 64, 65, 66 GSH monoethyl ester, 354, 355, 356 GSH precursors, 64, 65, 66, 67, 68, 353, 354, 355, 356, 359, 360 GSH-dependent oxidoreductases, 103, 104 GSS, 60, 61, 67 GSSG, 75, 85, 86, 105, 136 GST, 91, 92, 93, 94, 95, 96, 97, 98, 102 GSTA1, 92 GSTA2, 92 GSTA3, 92, 96 GSTA4, 92, 96, 98 GSTA5, 92 **GSTK. 93** GSTM1, 91, 92, 95, 98, 101, 102, 194, 195 GSTM2, 92 GSTM3, 92 GSTM4, 92 GSTM5, 92 GSTO1, 92 GSTP, 94, 96, 97 GSTP1, 92, 96, 215 GSTS1, 92 GSTT1, 91, 92, 98, 101, 102, 194 GSTT2, 92 GSTZ, 95, 96 GSTZ1, 92, 96, 97 GTPase, 5 GULO, 268, 269 gulonolactone oxidase, 265, 268 Gunn rat, 241, 242, 243, 255, 256, 257

Η

H1N1, 358, 364 H₂O₂, 8, 9, 12, 34, 48, 75, 125, 131, 132, 145, 154, 378 Haber-Weiss reaction, 8 hair, 154, 158, 326, 333, 383, 388 hair follicle, 154, 326, 333 hair follicles, 154, 326, 333 hair graying, 154, 158 hairless, 310, 311 half-life, 42 halogen, 69, 94, 99 halogen-containing compounds, 69, 94, 99 haploid, 82 haplotypes, 45, 141, 176, 215 haptoglobin, 288, 295 Harderian gland, 250 harmful effects, 381 HDL, 18, 197, 199, 201, 202, 204, 213 hearing loss, 65, 69, 358, 364 heart disease, 68, 98, 101, 152, 155, 188, 194, 206, 207, 209, 214, 224, 226, 232, 243, 244, 268, 272, 304. 327. 333 heart failure, 36, 45, 49, 54, 64, 76, 103, 107, 113, 126, 127, 162, 246, 278, 300, 321, 322, 331, 343, 355 heat stress, 116 heavy metal, 137, 217, 225, 229 heavy metals, 66, 161, 166, 217, 224, 225, 227 Heinz body, 129 Helicobacter pylori, 209, 215, 228, 235, 273, 279, 286, 293, 302, 306, 312, 325, 356, 360 hematocrit, 345 hematopoiesis, 137, 139 hematotoxicity, 183 heme, 11, 18, 25, 47, 48, 66, 70, 107, 114, 118, 136, 159, 160, 161, 163, 165, 166, 170, 171, 172, 173, 174, 175, 176, 177, 219, 221, 239, 241, 255, 256, 341, 371, 373 heme oxygenase, 18, 25, 66, 70, 114, 118, 136, 159, 170, 171, 172, 173, 174, 175, 176, 177, 221, 239, 241, 255, 256, 371, 373 heme ring, 239 hemin, 162, 163, 165, 242, 256, 349 hemodialysis, 274, 279 hemoglobin, 74, 80, 129, 159, 240, 257, 345, 369 hemoglobin glycation, 369 hemolysis, 56, 243 hemolytic anemia, 39, 59, 67, 70, 87, 129, 169 hemorrhage, 163, 172, 222, 266, 346, 350 hemorrhagic shock, 252, 253, 260 hepatic cell death, 202, 222 hepatic diseases, 324 hepatic failure, 358 hepatic fibrosis, 120, 249, 258 hepatic injury, 38, 115, 201, 227, 275 hepatic lipid accumulation, 37, 38, 43, 133 hepatitis, 18, 96, 120, 154, 158, 176, 208, 209, 215, 302, 309, 325, 370, 375 hepatitis a, 96, 370 hepatitis B, 154, 158, 209 hepatitis B virus, 154, 158 hepatitis c, 120 hepatitis C virus, 18, 176, 208

hepatocarcinogenesis, 39, 44, 70, 309, 350, 358, 363, 371 hepatocellular cancer, 155 hepatocellular carcinoma, 30, 45, 154, 158, 168, 229, 236, 371, 375 hepatocuprein, 33 hepatocyte growth factor, 328, 334 hepatocyte regeneration, 228 hepatocytes, 165, 166, 174, 176, 200, 210, 240, 361 hepatosplenomegaly, 107 hepatotoxicity, 38, 39, 50, 65, 69, 77, 82, 96, 100, 165, 166, 203, 235, 270, 286, 302, 325, 344, 370, 374, 382, 386, 387 hepatotoxins, 50 herbicide, 370 hereditary catalase deficiency, 47, 52 hereditary ferritinopathies, 223 hereditary GR deficiency, 85 hereditary hyperferritinemia cataract syndrome, 223 heterogeneity, 359 heteropolymers, 218 heterozygous, 36, 38, 63, 97, 137, 163, 183, 220, 221, 223 hexasulfonated C₆₀ fullerene, 381 H-ferritin mRNA, 223 HHCS, 223 HIF-10, 271 high blood pressure, 162, 250 high fat, 323 high-density lipoprotein, 152, 197, 210 high-fat diet, 127, 201, 211, 226, 227, 233, 308, 323, 332, 356 high-sucrose diet, 355, 361 hippocampal dentate gyrus, 163 hippocampus, 77, 153 histology, 243, 289 histone, 145, 146, 268, 322, 331 histone acetylation, 145, 146 histone deacetylase, 145 histone demethylation, 268 HIV, 67, 140, 209, 215, 287, 289, 294, 295, 327, 333, 358, 360 HIV infection, 67, 209, 287, 289, 327, 333, 360 HIV-1, 140, 294, 295 HO, 159, 160, 161, 162, 163, 164, 165, 166, 167, 169, 170 HO-1, 136, 159, 160, 161, 162, 163, 164, 165, 166, 167, 168, 169, 170, 175, 177, 255, 371 HO-2, 159, 160, 161, 162, 163, 166, 167, 170, 171, 176 HO-3, 159 HOBr, 13 HOCl, 9, 12

homeostasis, 11, 14, 17, 64, 77, 85, 87, 103, 105, 106, 107, 108, 127, 143, 144, 166, 174, 217, 219, 230, 232, 269, 275 homocysteine, 184, 260 homodimer, 34, 86, 124 homolysis, 6 homolytic, 9 homolytic cleavage, 9 homopolymers, 218 homotetramer, 48 homozygosity, 98 homozygous, 36, 37, 63, 65, 76, 88, 91, 97, 98, 106, 113, 118, 137, 154, 186, 187, 200, 220, 221, 380 HONO, 10 HOSCN, 13 host, 7, 11, 185, 212 host defense, 11 human brain, 134 human subjects, 16, 26, 33, 40, 52, 53, 79, 96, 98, 99, 107, 111, 117, 130, 135, 139, 146, 153, 155, 159, 170, 186, 190, 204, 205, 208, 209, 217, 223, 244, 250, 254, 288, 289, 290, 299, 303, 305, 306, 317, 327, 328, 329, 339, 346, 359, 372, 373 humanized mouse, 267, 275 humanized mouse model, 267, 275 humoral immunity, 185, 193 hydrogen, 3, 4, 5, 6, 8, 9, 11, 12, 13, 21, 28, 33, 34, 35, 47, 48, 49, 50, 51, 52, 53, 54, 56, 62, 73, 75, 99, 112, 123, 124, 125, 126, 129, 133, 145, 154, 158, 180, 219, 230, 251, 252, 267, 277, 283, 298, 339, 340, 341, 342, 378, 379, 380, 385 hydrogen abstraction, 283 hydrogen atoms, 298 hydrogen peroxide, 3, 4, 5, 6, 8, 9, 12, 13, 21, 28, 33, 34, 35, 47, 48, 49, 50, 51, 52, 53, 54, 56, 62, 73, 75, 99, 112, 123, 124, 125, 126, 129, 133, 145, 154, 158, 180, 219, 230, 251, 252, 267, 277, 339, 340, 341, 342, 378, 379, 380, 385 hydrolysis, 60, 198, 199, 204, 205 hydronephrosis, 38, 44 hydroperoxides, 18, 73, 75, 77, 78, 93, 94, 112, 123, 124, 125, 198, 214, 218, 267, 275, 342 hydroquinone, 181 hydroxycarboxylic acid, 199 hydroxyl, 4, 6, 8, 9, 10, 13, 18, 161, 180, 217, 219, 220, 225, 226, 251, 252, 266, 315, 317, 318, 371, 378, 379, 380, 385 hydroxyl groups, 266, 315, 317, 379 hydroxyl radical, 4, 8, 9, 10, 13, 18, 180, 217, 225, 251 hydroxylation, 244, 268 hyperbilirubinemia, 241, 242, 243, 255 hyperferritinemia, 223, 224, 231

hyperglycemia, 37, 127, 162, 172, 301, 310, 322, 355.369 hyperleptinemia, 227, 234 hyperlipidemia, 98, 211, 304, 311, 327 hypermethylation, 35 hyperoxia, 37, 50, 55, 65, 77, 81, 87, 88, 108, 127, 132, 144, 145, 146, 152, 161, 164, 173, 185, 242, 256, 356, 362, 370, 374 hyperoxidation, 126, 143, 146 hyperoxidized peroxiredoxins, 143 hyperplasia, 120, 183, 185, 191, 192, 211, 256, 293 hypersensitivity, 70, 115, 121, 175 hypertension, 36, 38, 44, 45, 49, 52, 53, 56, 59, 64, 68, 69, 76, 95, 100, 113, 118, 162, 176, 242, 246, 248, 254, 255, 268, 272, 284, 300, 301, 304, 311, 321, 322, 327, 343, 349, 355, 361, 369 hypertriglyceridemia, 117 hypertrophy, 49, 51, 103, 107, 108, 109, 114, 118, 261, 300, 307, 321, 322, 331, 355, 361 hypobromous acid, 13 hypocatalasemic mice, 47 hypochlorous acid, 4, 13 hypoplasia, 138 hyporeflexia, 287 hypotension, 79, 243 hypothesis, 231, 258, 260, 278 hypothiocyanous acid, 13 hypoxanthine, 6, 252, 253 hypoxia, 64, 130, 140, 152, 156, 161, 233, 259, 268, 275 hypoxia-inducible factor, 64, 140, 161, 275 hypoxia-inducible factor- 1α , 64

I

I-29 allele, 189 IC₅₀, 182 idebenone, 245, 246 idiopathic, 195, 209, 215 idiopathic chronic pancreatitis, 209, 215 ifosfamide, 66 IgE, 381 illumination, 12 imatinib, 182, 191 immune dysfunction, 74, 383 immune regulation, 78 immune response, 158, 185, 203, 236, 269, 312 immunity, 153, 185, 193, 203, 287 immunomodulation, 300 immunomodulatory, 256 immunoprecipitation, 143 immunosuppression, 326, 332

immunosuppressive agent, 38 inflammatory mediators, 319 immunosurveillance, 287 inflammatory responses, 6, 37, 66, 95, 97, 129, 167, in utero, 77, 380 175, 185, 226, 242, 248, 249, 250, 252, 271, 283, in vivo, 11, 18, 25, 35, 62, 64, 78, 81, 82, 86, 87, 96, 292, 294, 300, 319, 344, 345, 369, 372, 380 107, 129, 132, 143, 151, 153, 154, 160, 162, 163, inflammatory stress, 15, 42, 113, 116, 123, 130, 155, 164, 168, 174, 182, 183, 184, 185, 193, 199, 200, 162, 164, 165, 166, 169, 217, 230, 242, 246, 247, 203, 204, 205, 222, 228, 231, 236, 243, 248, 252, 251, 302, 322, 359, 380, 381 256, 259, 260, 276, 277, 281, 283, 293, 299, 300, influenza virus, 115, 120, 269, 276 302, 304, 308, 310, 317, 318, 319, 321, 326, 330, ingestion, 87, 335 339, 343, 344, 347, 353, 356, 362, 367, 374, 378, inhibition, 28, 36, 63, 65, 69, 87, 95, 97, 113, 138, 380, 381, 382, 387, 388 145, 162, 164, 165, 166, 167, 168, 185, 191, 199, incidence, 39, 44, 67, 78, 101, 116, 117, 170, 192, 201, 203, 211, 212, 222, 230, 233, 235, 243, 257, 204, 206, 208, 215, 271, 272, 312, 345, 348, 358, 258, 269, 271, 283, 286, 290, 300, 303, 307, 318, 359.363 319, 321, 322, 324, 328, 343, 345, 347, 354, 369, indinavir, 241 371, 372, 373, 379, 380, 383, 388 indole, 48 inhibitor, 36, 42, 49, 54, 63, 87, 88, 95, 96, 100, 107, indomethacin, 115, 120 109, 140, 145, 163, 165, 173, 182, 186, 191, 201, inducible nitric oxide synthase, 7, 14, 173, 185, 270, 255, 286, 324, 334 276, 360, 380 initiation, 15, 39, 181, 190, 192, 259, 303, 327, 363, induction, 23, 25, 26, 27, 29, 30, 36, 40, 54, 61, 66, 370 70, 77, 82, 95, 97, 101, 115, 128, 139, 140, 145, innate immunity, 203 161, 162, 165, 166, 167, 168, 172, 174, 176, 182, iNOS, 270, 277 184, 191, 192, 193, 200, 219, 221, 228, 230, 231, insecticide, 197 241, 243, 245, 248, 255, 281, 284, 290, 300, 302, insertion, 189, 195 303, 318, 319, 326, 327, 330, 343, 367, 368, 371, insulin, 54, 64, 69, 76, 81, 98, 131, 162, 172, 184, 372, 382, 386 194, 207, 221, 224, 232, 233, 269, 275, 277, 285, ineffectiveness, 290 289, 292, 301, 312, 322, 323, 334, 343, 345, 349, infancy, 383 355, 361, 374 infantile multisystemic disease, 246 insulin resistance, 76, 81, 98, 162, 184, 207, 221, infarct, 81, 106, 107, 163, 269, 355, 386 224, 232, 233, 285, 289, 301, 323, 343, 345, 349, infarction, 41, 45, 70, 81, 114, 118, 127, 131, 152, 355, 361 206, 213, 275, 278, 359, 364, 386 insulin sensitivity, 64, 172, 194, 285, 292, 334 infertility, 74, 78, 80, 116, 138, 246 interface, 136 inflammation, 7, 10, 11, 12, 13, 14, 38, 44, 49, 50, interference, 17, 145, 321 65, 66, 69, 71, 77, 78, 79, 82, 96, 100, 103, 106, interferon, 165 107, 108, 111, 114, 115, 116, 120, 126, 128, 161, interleukin-1, 174, 185, 200, 210 164, 165, 166, 167, 170, 171, 173, 175, 176, 185, interleukin-6, 82, 185, 188, 200, 210 188, 190, 197, 201, 202, 208, 209, 211, 212, 223, interleukin-8, 185, 362 225, 227, 228, 229, 230, 234, 235, 236, 242, 245, interstitial fibrosis, 51, 56, 228 246, 247, 248, 249, 250, 251, 253, 257, 259, 265, intervention, 1, 21, 26, 27, 28, 29, 39, 40, 41, 42, 53, 269, 270, 271, 283, 285, 286, 289, 290, 292, 293, 67, 68, 76, 99, 127, 130, 138, 170, 184, 190, 202, 297, 300, 301, 302, 303, 304, 306, 307, 308, 309, 203, 204, 221, 243, 245, 246, 247, 250, 251, 252, 310, 312, 319, 321, 323, 324, 325, 326, 327, 329, 254, 255, 260, 265, 271, 272, 273, 274, 281, 287, 330, 332, 333, 343, 344, 346, 355, 356, 357, 358, 288, 290, 297, 304, 305, 306, 327, 328, 329, 331, 359, 360, 361, 365, 367, 369, 370, 372, 380, 382, 337, 339, 342, 344, 345, 346, 353, 355, 359, 360, 383, 386, 387 364, 367, 371, 372, 373, 380, 381, 383, 384 inflammatory arthritis, 176, 383 interventional trials, 265, 272, 288, 290, 297, 304, inflammatory bowel disease, 77, 166, 189, 228, 235, 305, 306, 327, 328, 329 252, 302, 325, 356 intestinal ischemia-reperfusion injury, 242, 270, 302, inflammatory cells, 7, 9, 14, 222, 227, 246, 271 325, 344, 382 inflammatory cytokines, 64, 113, 161, 162, 184, 219, intestine, 256, 309, 344, 387 225, 246, 380 intoxication, 293 inflammatory disease, 18 intracerebral hemorrhage, 163, 172

intratracheal instillation, 38, 227 intra-uterine growth retardation, 224, 232 intravenously, 273 introns, 161 invasive cancer, 333 iodine, 363 iodothyronine deiodinases, 73 ionizing radiation, 6, 9, 371 ions, 6, 8, 9, 11, 137, 217, 218, 219, 225, 249, 250, 252, 267, 283, 318, 319, 323, 354 IRE, 218, 223 iron, 8, 9, 11, 47, 48, 103, 105, 106, 107, 108, 109, 159, 160, 161, 164, 169, 173, 176, 217, 218, 219, 220, 221, 222, 223, 224, 230, 231, 232, 239, 252, 268, 270, 283, 319, 323, 331, 341, 342, 369, 374, 380, 381, 386 iron deficiency anemia, 223 iron homeostasis, 103, 105, 106, 107, 108, 219, 230, 232 iron overload, 103, 107, 108, 109, 223, 224, 231, 232 iron regulatory element, 218, 223 iron response proteins, 218 iron-chelating agent, 270 iron-dithiocarbamate, 369 iron-protoporphyrin IX-containing protein, 47 iron-sulfur clusters, 8, 219 iron-sulfur proteins, 105 IRP1, 218 IRP2, 218 IRPs, 218, 223 irradiated C3H mice, 47 irradiation, 13, 17, 78, 82, 116, 128, 157, 228, 229, 236, 270, 286, 289, 294, 303, 305, 326, 329, 333, 346, 350, 358, 383 ischaemic heart disease, 232 ischemia, 5, 6, 16, 17, 36, 38, 44, 49, 50, 55, 64, 65, 66, 70, 76, 78, 81, 82, 95, 103, 106, 107, 108, 114, 115, 118, 119, 120, 121, 126, 127, 128, 131, 133, 152, 156, 162, 164, 165, 166, 167, 172, 173, 174, 175, 217, 220, 222, 226, 227, 232, 233, 234, 242, 246, 248, 249, 252, 253, 256, 258, 269, 270, 276, 284, 286, 292, 293, 300, 301, 302, 307, 308, 309, 310, 321, 324, 325, 343, 344, 345, 348, 349, 355, 356, 357, 361, 363, 369, 371, 373, 374, 381, 382, 384, 386, 387 ischemia-reperfusion injury, 6, 16, 38, 50, 65, 66, 76, 78, 95, 106, 114, 115, 120, 128, 131, 164, 165, 166, 173, 220, 222, 226, 242, 252, 256, 258, 269, 270, 276, 286, 293, 301, 302, 309, 324, 325, 343, 344, 345, 356, 357, 363, 369, 371, 381, 384 ischemic brain injury, 37, 43, 114, 260, 308 ischemic heart disease, 68, 152, 206, 224, 226, 268, 272

ischemic injury, 70, 114, 127, 132, 163, 227, 248, 324.344 ischemic stroke, 79, 83, 224, 254, 261, 269, 370, 372, 373 islet allografts, 242, 256 isoflavone, 334, 335 isolated perfused hearts, 64, 373 isoliquiritigenin, 320 isomers, 294 isopentenyl diphosphate, 244 isoprenoid, 244 isozymes, 25, 33, 34, 36, 37, 40, 62, 73, 74, 75, 76, 77, 78, 79, 85, 91, 93, 94, 96, 97, 98, 103, 104, 106, 111, 123, 124, 126, 127, 135, 136, 139, 149, 150, 151, 153, 155, 159, 160, 161, 164, 166, 198, 200, 203, 206, 209, 224

I

Janus kinase/STAT, 97, 101 jaundice, 370, 375 JB6 cell, 145 JNK pathway, 140 joint destruction, 116, 121 joint swelling, 116 Jun D, 219

К

- kainic acid-induced seizure, 227, 234 kale, 297 Kaposi sarcoma, 175 kappa GST, 93 Keap1, 24, 25 Kelch-like ECH-associated protein 1, 24 keratinocytes, 51, 56, 116, 121, 128, 167, 228, 286, 294, 326, 333 ketone, 12 kidney, 39, 41, 45, 55, 48, 51, 66, 78, 96, 97, 101, 115, 135, 145, 150, 153, 155, 166, 167, 184, 193, 200, 224, 229, 236, 245, 254, 261, 270, 275, 293, 310, 325, 329, 345, 357, 359, 360, 373
 - kidney transplantation, 41, 166, 270, 329, 357, 360 knockout mouse, 81, 109, 153, 157, 182, 226, 268, 284

L

- L. Pauling, 33, 271, 272
- LA, 249, 250, 254, 255
- lactonase, 199, 203, 205, 206, 208, 213

lactonization, 199 lanthanide, 378 L-arginine, 71, 359, 365 latency, 345 lateral sclerosis, 16, 40, 41, 42, 45, 138, 141, 207, 208, 381 LDL, 64, 69, 87, 197, 199, 201, 202, 205, 209, 210, 268, 300, 307 LDL cholesterol, 87, 201 LDL receptor, 202, 307 leakage, 7 learning, 55, 138, 140, 275, 344, 348, 382 Leigh syndrome encephalopathy, 247 lens, 78, 107, 109, 153, 157, 223, 250, 267, 275, 294, 302 lens crystallins, 267, 275 leptin, 202, 214 lesions, 38, 41, 114, 118, 167, 175, 184, 193, 201, 234, 235, 326, 328, 333, 334, 343, 371 lettuce, 265 leucine, 191, 205 leukemia, 120, 182, 191, 278 leukotrienes, 319 lifespan, 52, 85, 87, 97, 129, 138, 152, 156, 320, 330, 344, 348, 380, 382, 387 ligand, 165, 173, 340, 343, 346 light scattering, 83 lignans, 315 lipid hydroperoxide, 4, 11, 81 lipid metabolism, 184, 323 lipid peroxidation, 11, 12, 13, 14, 18, 28, 38, 62, 77, 78, 94, 99, 163, 172, 181, 199, 201, 202, 213, 215, 221, 245, 248, 254, 283, 285, 294, 309, 369, 374, 379 lipid peroxides, 205, 351 lipid peroxyl radical, 28, 62, 63, 340 lipids, 10, 11, 13, 14, 17, 22, 197, 198, 199, 201, 211.283 lipopolysaccharide, 114, 115, 116, 120, 129, 175, 227, 235, 270, 301, 349 lipoproteins, 12, 13, 140, 199, 212, 213, 240, 245 lipoxygenases, 319 liquid chromatography, 232 liver, 6, 16, 17, 33, 38, 43, 44, 48, 50, 55, 65, 67, 68, 69, 70, 71, 74, 77, 82, 96, 98, 99, 100, 102, 115, 117, 120, 122, 128, 130, 133, 137, 150, 153, 155, 159, 165, 166, 174, 177, 179, 181, 184, 188, 193, 194, 197, 200, 202, 208, 210, 212, 215, 221, 222, 224, 227, 231, 232, 235, 240, 242, 243, 245, 249, 252, 253, 256, 258, 259, 260, 269, 275, 276, 281, 282, 286, 287, 289, 293, 301, 302, 303, 307, 322, 324, 329, 332, 344, 345, 348, 349, 354, 356, 357, 358, 360, 362, 363, 364, 370

liver cells, 96, 130, 200, 357 liver cirrhosis, 208 liver damage, 222, 228, 231 liver disease, 16, 38, 65, 68, 69, 71, 165, 166, 174, 202, 208, 212, 215, 243, 270, 286, 289, 302, 324, 344, 349, 356 liver dysfunction, 102, 165, 188, 194, 258 liver enzymes, 222, 344, 356 liver failure, 50, 65, 70, 344, 348, 349, 358, 363, 364 liver fibrosis, 228, 235, 259, 270, 276 liver ischemia-reperfusion injury, 38, 65, 128, 165, 242, 249, 252, 276, 356 liver transplant, 38, 177, 242, 276, 286 liver transplantation, 38, 177, 242, 276, 286 LO', 4, 11, 12 localization, 17, 74, 113, 124, 128, 156, 157, 159, 161 locomotor, 153 locus, 155 longevity, 114, 117, 119, 127, 129, 133, 154, 158, 206, 213, 229, 237, 330, 350, 382 LOO', 4, 11, 12, 63, 245 LOOH, 4, 11, 12, 63, 75, 93, 125, 144, 245 lovastatin, 199 low-density lipoprotein, 64, 87, 89, 140, 197, 201, 213, 268, 300, 321, 333 LPS, 129, 133 lumen, 74, 78 lung adenocarcinoma, 170, 177 lung adenomas, 97 lung cancer, 101, 130, 134, 188, 193, 209, 215, 229, 304, 363 lung disease, 40, 43, 69, 71, 285, 344 lung fibrosis, 165, 301, 344, 348, 362 lung function, 312 lung injury, 16, 38, 45, 50, 55, 65, 77, 87, 89, 120, 127, 132, 138, 140, 164, 165, 173, 183, 185, 189, 193, 195, 227, 234, 242, 243, 256, 285, 292, 294, 324, 344, 356, 362, 370, 374, 387 lung surfactant, 128 lupus, 57 lutein, 297, 299, 300, 301, 302, 303, 305, 307, 309, 310 luteolin, 316 lycopene, 13, 297, 299, 300, 301, 302, 304, 305, 307, 308, 309, 310, 311, 312 lymphocytes, 185, 250 lymphoid, 116, 121, 185, 345 lymphoid tissue, 185 lymphoma, 116, 121, 229, 345, 350, 358, 363 lymphomas, 128, 184, 188 lymphosarcoma, 345 lysine, 266

lysine hydroxylase, 266 lysis, 203

Μ

M40401, 340, 343 M40403, 340, 343, 344, 347, 349, 350 M55L, 205, 208 mAb, 116 machinery, 297 macrophage cholesterol efflux, 201 macrophage migration inhibitory factor, 115 macrophages, 86, 87, 88, 89, 129, 140, 185, 200, 201, 202, 277, 321 macrosteatosis, 222 macula lutea, 305 macular degeneration, 16, 157, 209, 216, 302, 305, 310.311 macular pigments, 302, 305 magnetic resonance, 375 magnetic resonance imaging, 375 Maillard reaction, 267, 275 malaria, 87, 89, 176 malarial infection, 287, 294 male fertility, 75 male infertility, 74, 78, 80, 116, 138, 246 maleylacetoacetate, 96, 97, 101 maleylacetoacetate isomerase, 97, 101 maleylacetone, 98 malignancies, 128, 209, 223, 229, 272, 345 malignancy, 194, 278 malignant melanoma, 170 malignant transformation, 14 malignant tumors, 252 malondialdehyde, 14 mammalian tissues, 47, 73, 91, 104, 111, 124, 127, 135, 136, 143, 150, 159, 161, 179, 182, 197, 224, 225 mammals, vii, 1, 4, 9, 12, 14, 21, 22, 23, 29, 31, 33, 34, 63, 65, 74, 104, 107, 111, 123, 124, 135, 149, 152, 155, 156, 159, 179, 197, 217, 218, 220, 224, 239, 250, 252, 266, 267, 353 management, 26, 274, 279, 343, 345, 353, 356, 357, 360, 362 manganese, 5, 18, 33, 42, 43, 340, 341, 342, 350, 355 manganese superoxide dismutase, 5, 18, 42, 43, 350, 355 mangoes, 265 manipulation, 29, 79, 168, 221 MAPEG, 91, 93 markers, 129, 188, 194, 259, 273, 310, 328, 343

marrow, 88, 183, 184, 185, 191, 192, 222, 229, 236, 250 mast cells, 286, 294, 381 matrix, 5, 34, 82, 116, 307, 322 matrix metalloproteinase, 82, 116, 322 matrix metalloproteinase-1, 82, 116 maze learning, 382 MCP-1, 114, 386 medulla, 325, 332 MEK, 307 melanin, 186 melanocytes, 154, 158, 186 melanogenesis, 154, 186 melanoma, 168, 170, 175, 177, 186, 222, 232, 364, 383 melatonin, 179, 180, 181, 186, 190, 191, 239, 250, 251, 254, 259, 276, 293 membrane permeability, 41 membranes, 74, 75, 78, 181, 240, 245 memory, 43, 65, 138, 140, 275, 382 menadione, 183, 191 menopausal, 329, 334, 335 mercapturic acid, 92, 94, 99 mercury, 217, 224, 302 mercury chloride, 302 mesalamine, 357, 363, 365, 371, 375 mesangial cells, 302, 310 mesoderm, 137 mesothelioma, 277 Met55Leu, 205 meta-analysis, 155, 158, 206, 208, 213, 214, 274, 278, 279, 333, 334, 335, 359, 365 metabolic disorder, 331 metabolic disturbances, 215, 355 metabolic syndrome, 118, 155, 162, 171, 184, 188, 197, 201, 202, 207, 212, 221, 227, 246, 249, 250, 254, 259, 269, 270, 273, 279, 285, 301, 306, 308, 312, 323, 328, 331, 339, 343, 355, 360 metabolism, 25, 53, 62, 65, 68, 69, 77, 88, 91, 94, 131, 132, 172, 183, 184, 188, 194, 199, 221, 225, 232, 239, 241, 249, 252, 253, 255, 266, 287, 300, 301, 320, 321, 323, 325, 331, 343, 370, 378, 382 metabolites, 62, 94, 95, 96, 97, 183, 186, 240, 283, 291, 318, 319, 345 metal response element, 225 metalloproteinase, 82, 116 metallothionein, 217, 226, 233, 234, 235, 236, 237 metal-response element-binding transcription factor-1,225 metals, 66, 161, 166, 217, 224, 225, 227, 233, 236, 319 metastasis, 16, 51, 54, 78, 83, 117, 122, 130, 154, 158, 168, 187, 303, 326, 358, 363

metastatic cancer, 346 metformin, 366 methamphetamine, 189, 195 methamphetamine psychosis, 189, 195 methionine, 22, 73, 112, 149, 150, 151, 152, 153, 154, 155, 156, 157, 158, 205, 356, 362 methionine oxidation, 152, 153, 156, 158 methionine R-sulfoxide, 149, 150, 151 methionine S-sulfoxide, 149, 150, 151 methionine sulfoxide, 22, 73, 112, 149, 150, 153, 154, 155, 156, 157, 158 methionine sulfoxide reductase, 22, 73, 149, 150, 155, 156, 157, 158 methylation, 151, 244 mevalonate, 244 mice, 36, 37, 38, 39, 43, 44, 47, 50, 51, 52, 54, 55, 56, 63, 64, 65, 67, 69, 70, 73, 76, 77, 78, 80, 81, 82, 83, 87, 88, 89, 96, 97, 98, 100, 101, 106, 107, 108, 113, 114, 115, 116, 118, 119, 120, 121, 122, 126, 127, 128, 129, 131, 132, 133, 137, 138, 140, 145, 146, 151, 152, 153, 156, 157, 162, 163, 164, 165, 167, 169, 171, 172, 173, 174, 175, 176, 182, 183, 184, 185, 186, 191, 192, 193, 201, 202, 203, 204, 211, 212, 213, 220, 221, 222, 224, 226, 227, 228, 229, 231, 232, 233, 234, 235, 236, 248, 255, 258, 260, 268, 269, 270, 275, 276, 277, 284, 285, 286, 287, 291, 293, 294, 301, 302, 307, 308, 309, 310, 311, 320, 321, 324, 325, 326, 330, 332, 333, 343, 344, 345, 347, 348, 349, 350, 355, 358, 361, 362, 363, 369, 370, 375, 380, 381, 382, 387, 388 microarrays, 154 microglia, 164, 308, 386 microorganisms, 9, 13, 203 midbrain, 221, 232 midgestation, 77 migration, 95, 100, 115, 184, 193, 256, 310, 319 military, 204 minocycline pigment, 270, 277 mitochondria, 4, 5, 7, 17, 21, 37, 43, 48, 49, 50, 52, 54, 56, 60, 74, 85, 104, 107, 111, 113, 124, 131, 135, 143, 145, 146, 149, 159, 219, 220, 244, 245, 249, 250, 251, 257, 284, 302, 340, 356, 362, 380 mitochondrial complex I, 379 mitochondrial damage, 69, 87, 284 mitochondrial depolarization, 64 mitochondrial DNA, 127 mitochondrial dysfunction, 98, 101, 107, 133, 222, 260, 292, 349 mitochondrial ferritin, 218, 219, 222, 224, 232 mitochondrial intermembrane space, 34 mitochondrial matrix, 5, 34 mitogen, 25, 161, 307 mitogen-activated protein kinase, 25, 307

mitomycin C, 183, 192 MitoQ, 245, 246 MitoVitE, 284 Mn(II) pentaazamacrocyclic ligand-based complexes, 340 Mn(III) metalloporphyrins, 340, 341 Mn(III) salen complexes, 340, 341 MnSOD, 33, 34, 36, 37, 38, 39, 43, 44, 248, 380, 382 MnTBAP, 37, 340, 341, 342, 344, 345, 348 MnTE-2-PvP⁵⁺, 340, 341, 345 MnTMPyP, 340, 341, 342, 343, 344, 349 molecular oxygen, vii, 1, 3, 4, 5, 6, 7, 8, 11, 12, 18, 21, 28, 33, 34, 35, 47, 48, 160, 341, 379 monitoring, 208 monoclonal antibody, 116 monocyte chemoattractant protein, 114, 212 monocyte chemoattractant protein-1, 114, 212 monoglucuronide, 240 mono-O-methylated flavanols, 21 monophenol, 315 monothiol Grxs, 104 morbidity, 273 Morris water maze learning and memory task, 382 mortality risk, 213 motif, 104, 223 motor activity, 374 motor neuron disease, 107 motor neurons, 40, 43 MPO, 9, 367, 368 MPP⁺, 164, 276, 381, 387 MPTP, 96, 132, 231, 276, 292, 361, 374, 381, 387 MRE, 225 MRI, 374 mRNA, 35, 40, 61, 154, 188, 189, 203, 210, 223, 232 mRNA stabilization, 61 Msr, 149, 150, 151, 152, 154, 155 MsrA, 149, 150, 151, 152, 153, 154, 155, 156, 157 MsrA (-402C/T), 154 MsrB, 149, 150, 152, 153, 154, 156 MsrB1, 73, 149, 150, 151, 152, 153, 156 MsrB2, 149, 150, 151 MsrB3, 149, 150, 151 MT, 217, 224, 225, 226, 227, 228, 229 MT-1, 224, 225, 226, 227, 228, 229 MT-2, 224, 225, 226, 227, 228, 229 MT-3, 224, 225, 226, 227 MT-4, 224, 225 MTF-1, 225 mucin, 362 mucosa, 41 multidrug resistant protein efflux pumps, 94 multiple organ dysfunction, 346 multiple sclerosis, 50, 207, 208, 214, 253, 260

multistage carcinogenesis, 39, 78, 118, 128, 222, 229, 243, 303, 327, 345 muscle atrophy, 39, 44 mutagenesis, 18 mutagens, 243 mutant, 40, 41, 43, 86, 104, 129, 153, 186, 221, 222, 232, 268, 383 mutation, 70, 87, 103, 107, 194, 213, 223, 232 myelogenous hyperplasia, 183, 185, 192 myeloperoxidase, 9, 13, 18, 56 myeloproliferative disease, 183, 192 myocardial infarction, 41, 45, 70, 81, 114, 127, 131, 152, 206, 213, 275, 278, 359, 364 myocardial ischemia, 36, 49, 64, 76, 95, 103, 106, 107, 113, 126, 127, 152, 162, 172, 217, 242, 246, 248, 252, 258, 284, 292, 300, 307, 321, 343, 355, 369, 373 myocardial ischemia-reperfusion injury, 36, 49, 64, 76, 95, 103, 106, 107, 113, 126, 127, 152, 162, 217, 242, 246, 248, 252, 258, 284, 300, 307, 321, 343, 355, 369, 373 myocarditis, 76, 81, 113, 119 myocardium, 106, 114, 119, 134, 162, 172, 220, 226, 227 myoglobinuria, 246 myopathy, 246, 247 myosin, 114, 119 Ν

N(G)-nitroarginine methyl ester, 162 N,N-bis(2-chloroethyl)-N-nitrosourea, 86 N₂, 378 N-3-oxododecanoyl homoserine lactone, 203 N-acetyl-5-methoxytryptamine, 250 N-acetylcysteine, 63, 64, 66, 68, 70, 71, 293, 337, 342, 353, 354, 355, 356, 357, 358, 359, 360, 361, 362, 363, 364, 365, 366 N-acetyl-p-benzoquinoneimine, 96, 356, 357, 358 N-acystelyn, 354 NAD, v, 4, 5, 7, 14, 21, 22, 23, 25, 29, 179, 180, 181, 184, 190, 191, 192, 193, 194, 195, 202, 243, 245, 249, 257, 267, 270, 283, 285, 300, 318, 319, 342, 370 NAD(P)H quinone oxidoreductase, 23, 25, 179, 180, 190, 191, 192, 193, 194, 195, 245 quinone oxidoreductase 1, 23, 25, 179, 190, 191, 192, 193, 194, 195, 245 NAD(P)H oxidase, 4, 5, 7, 14, 21, 202, 243, 249, 257, 267, 270, 283, 285, 300, 318, 319, 342, 370 NADH, 184, 190, 193, 379, 385 ubiquinone oxidoreductase, 379, 385

- NADPH cytochrome P450 reductase, 160
- nanocage, 218
- nanocapsules, 377
- nanoceria, 378, 384
- nanocomposites, 380, 385
- nanocrystalline C60 fullerene, 383
- nanomaterials, viii, 6, 337, 377, 380, 381, 382, 383, 384, 385, 386, 387, 388
- nanomedicine, 377, 383, 384, 385
- nanoparticles, 6, 18, 22, 337, 340, 377, 378, 379,
- 380, 381, 382, 383, 384, 385, 386, 387
- Nano-Se, 380, 385
- nanoshells, 377
- nanospheres, 377
- nanotechnology, 377
- nanotoxicology, 381, 383
- necrosis, 50, 107, 108, 184, 185, 193, 200, 210, 219, 230.370
- negative outcomes, 272
- nematode, 382, 387
- neoplasm, 386
- neovascularization, 167, 176
- nephronectomy, 51
- nephropathy, 16, 51, 78, 82, 114, 115, 119, 162, 166, 167, 209, 215, 246, 257, 270, 286, 293, 302, 309, 325, 345, 353, 357, 359, 363, 364, 365
- nephrotoxicants, 66
- nephrotoxicity, 38, 44, 51, 56, 66, 97, 101, 166, 175, 228, 243, 257, 270, 286, 293, 302, 310, 325, 345, 357, 363
- nerve, 199, 201, 204, 288
- nerve agents, 199, 201, 204
- nervous system, 127, 132, 138, 153, 185, 246, 324, 381
- neural function, 80
- neural tube defects, 111, 117
- neuroblastoma, 193
- neurodegeneration, 6, 33, 37, 50, 52, 64, 69, 77, 91, 95, 98, 103, 111, 117, 127, 129, 132, 135, 138, 153, 169, 186, 209, 217, 221, 223, 232, 234, 249,
 - 251, 285, 331, 344, 370
- neurodegenerative diseases, 77, 98, 109, 157, 247, 331, 348, 356
- neurodegenerative disorders, 40, 50, 64, 95, 152, 163, 164, 185, 221, 246, 323, 370
- neuroendocrine tissues, 269
- neuroferritinopathy, 221
- neurogenesis, 234
- neurohormone, 179, 186, 251
- neurological disease, 37, 149, 154, 208, 370, 381
- neurological diseases, 37, 149, 154, 208, 381
- neuromotor, 350
- neuronal apoptosis, 132

neuronal cells, 37, 107, 127, 163, 164, 173, 242 neurons, 37, 40, 43, 65, 69, 96, 127, 145, 163, 164, 173, 186, 221, 222, 232, 256, 260, 288, 308, 310, 324, 356, 381, 382, 386 neuropathy, 36, 43, 50, 56, 285, 288, 292, 355 neuroprotection, 37, 50, 76, 81, 163, 186, 232, 242, 269, 301, 308, 323, 324, 331, 361, 386 neuroprotective agents, 373, 374, 386 neurotoxicity, 114, 119, 127, 164, 173, 231, 234, 241, 242, 256, 387 neurotoxin. 65 neurotransmission, 11, 269 neurotransmitter, 48 neutrophil, 38, 121 neutrophils, 13, 18, 184, 185 NF-Y, 75, 231 NF-KB, 38, 125, 219, 245, 300, 354, 369, 370 nickel, 227, 380 nickel oxide nanoparticles, 380 nigrostriatal, 381 N-isobutyrylcysteine, 354 nitrate, 11, 18, 228 nitration, 15, 113 nitrative stress, 15 nitric oxide, 4, 5, 6, 7, 8, 10, 11, 14, 15, 18, 62, 112, 161, 162, 173, 185, 243, 255, 268, 270, 276, 285, 291, 300, 317, 319, 321, 328, 330, 354, 360, 364, 369, 370, 373, 374, 378, 380 nitric oxide synthase, 5, 6, 7, 11, 14, 173, 185, 243, 270, 276, 291, 319, 330, 360, 380 nitrite, 11 nitroaromatics, 180 nitrogen, 1, 3, 4, 5, 7, 10, 11, 16, 17, 21, 31, 59, 62, 85, 111, 115, 161, 226, 240, 255, 259, 267, 283, 299, 300, 318, 337, 339, 373, 377, 378 nitrogen dioxide, 4, 10, 11, 378 nitrogen oxides, 7, 11, 299 nitroglycerin, 41 nitrones, 368, 369, 375 nitrosative stress, 15, 99, 131, 134, 292 nitroso (-NO) group, 15, 112 nitroso compounds, 369 nitrosoperoxycarbonate, 10 nitrosylation, 15, 113 nitroxide, 340, 349 NO, 4, 8, 11, 378 'NO₂*, 10 ·NO₂, 4, 10, 378 Nobel Laureate, 271 Nobel Prize, 379 NOD mice, 114, 119, 172 noise, 65, 69, 358, 364 noise-induced hearing loss, 65, 69, 358, 364

nonalcoholic fatty liver disease, 38, 65, 165, 166, 203, 208, 286, 289, 302, 324, 344, 349, 356 nonalcoholic steatohepatitis, 70, 289, 295, 309, 360, 362 non-enzymatic antioxidants, 59, 62 non-small cell lung cancer, 130, 134 NOX1.5 NOX2, 5 NOX3, 5 NOX4, 5 NOX5.5 NQO, 179, 180, 182, 186, 187, 188, 189, 190 NQO1, 179, 180, 181, 182, 183, 184, 185, 186, 187, 188, 189, 190, 191, 192, 193, 194, 195, 245 NOO1*1, 186, 189 NQO1*2, 186, 187, 188, 189, 194 NQO1*2/*2, 186, 187 NQO1*3, 186, 187 NQO2, 179, 180, 181, 182, 183, 184, 185, 186, 187, 188, 189, 191, 192, 193, 195, 196 Nrf1, 23, 29, 61, 225, 233 Nrf2, 23, 24, 25, 27, 29, 30, 35, 42, 49, 54, 61, 75, 77, 80, 82, 86, 88, 93, 113, 125, 130, 137, 139, 144, 145, 146, 161, 181, 182, 191, 219, 225, 230, 233, 248, 258, 299, 318, 324, 329, 330, 331, 332 Nrf3, 23, 29 NSC66041, 182 N-terminal Cys, 124 nuclear factor-erythroid 2 p45-related factor 2, 23 nuclear ferritin, 219 nuclear respiratory factor 1, 125 nuclear respiratory factor 2, 125 nuclei, 24, 34, 48, 60, 74, 104, 111, 113, 124, 135, 149, 159, 161, 218, 219, 220 nucleic acid, vii, 10, 11, 13, 14, 17, 22, 268 nucleotide excision repair, 326, 332 nucleus, 171, 244 null genotype, 91, 98, 102, 186 nuts, 281 NXY-059, 367, 368, 370, 371, 372, 374, 376

0

 O_2 , 4, 7, 8, 9, 34, 35, 53, 180, 378 O_3 , 4, 13 obese Zucker rats, 343, 348 obesity, 76, 81, 98, 149, 154, 155, 184, 188, 193, 202, 207, 212, 227, 234, 253, 292, 301, 323, 331, 334, 343, 355, 361 obstruction, 51, 55, 167 obstructive kidney disease, 166, 167 obstructive lung disease, 344 occlusion, 163, 172, 269, 350, 355 Oct-1, 137 octanoic acid, 249 'OH, 4, 10, 11, 180 OKN007, 371, 375 olfactory epithelium, 74 oltipraz, 184 omeprazole, 279 oncogenesis, 236, 271 one-electron reduction, 6, 7, 180, 340 ONOO⁻, 4, 8, 10 ONOOCO₂, 10 ONOOH, 10 optic neuropathy, 50 oral cancer, 328 organ, 16, 38, 65, 96, 208, 227, 229, 252, 270, 273, 346, 349, 364, 383 organic peroxides, 126 organophosphate poisoning, 201, 204, 209, 213 organophosphates, 198, 199, 201, 216 osteoarthritis, 364 osteonecrosis, 52, 56 osteopenia, 114, 119 ovalbumin, 96, 227, 276, 285, 301, 308, 324 ovarian cancer, 215, 289 ovarian epithelial carcinoma, 209 overweight, 333 oxidants, 12, 13, 18, 96, 137, 167, 181, 184, 211, 219, 225 oxidation, 8, 10, 11, 13, 18, 35, 44, 64, 69, 82, 113, 123, 129, 130, 131, 140, 149, 152, 153, 154, 156, 158, 184, 193, 197, 199, 205, 209, 210, 218, 219, 251, 276, 300, 319, 331, 365 oxidation products, 11 oxidative cardiomyopathy, 64, 380 oxidative cytotoxicity, 184 oxidative damage, 9, 13, 14, 15, 22, 44, 52, 64, 65, 78, 80, 83, 120, 127, 132, 152, 153, 157, 219, 220, 222, 226, 227, 230, 231, 240, 251, 253, 258, 267, 269, 287, 303, 356, 378, 380, 382 oxidative stress, 1, 3, 14, 15, 16, 17, 25, 26, 27, 28, 29, 30, 31, 37, 38, 39, 41, 42, 43, 44, 45, 49, 50, 51, 52, 54, 55, 59, 61, 64, 65, 66, 67, 69, 76, 78, 79, 81, 82, 85, 86, 87, 88, 91, 93, 94, 95, 96, 98, 100, 101, 103, 105, 106, 107, 109, 111, 113, 114, 115, 116, 118, 119, 120, 121, 122, 126, 127, 128, 129, 131, 132, 133, 134, 143, 144, 145, 146, 149, 152, 153, 155, 156, 157, 158, 161, 162, 163, 164, 165, 166, 167, 169, 170, 171, 172, 177, 185, 186, 188, 190, 193, 194, 197, 199, 201, 202, 206, 208, 209, 210, 211, 213, 217, 219, 220, 221, 222, 223, 225, 227, 228, 229, 230, 231, 232, 235, 240, 242, 246, 247, 248, 249, 250, 251, 253, 254, 256, 259, 260, 265, 268, 269, 270, 272, 273, 274, 275, 276,

279, 285, 286, 288, 289, 290, 291, 292, 294, 295, 297, 300, 301, 302, 303, 304, 305, 306, 307, 310, 312, 319, 321, 323, 324, 325, 326, 327, 329, 332, 337, 339, 343, 344, 345, 346, 348, 349, 350, 355, 356, 357, 358, 359, 360, 361, 362, 364, 367, 368, 369, 370, 374, 380, 381, 382, 383, 384, 386 oxidative stress injury, 1, 3, 52, 169, 242, 256, 368, 380, 381 oxidative tissue injury, 25, 86, 219, 340 oxidative/inflammatory stress, 15 oxide nanoparticles, 6, 22, 337, 340, 377, 378, 380, 381, 382, 383, 384, 385, 386, 387 oxidized LDL, 87, 202 oxyferryl species, 48 oxygen, 1, 3, 4, 5, 6, 7, 8, 11, 12, 13, 16, 17, 18, 21, 28, 31, 33, 34, 35, 47, 48, 50, 51, 54, 55, 59, 62, 82, 85, 91, 106, 111, 112, 120, 126, 145, 149, 153, 160, 161, 179, 180, 183, 219, 226, 229, 232, 233, 236, 240, 241, 252, 259, 267, 283, 297, 299, 300, 303, 318, 337, 339, 341, 344, 348, 349, 350, 364, 368, 369, 370, 373, 375, 377, 378, 379, 380, 382, 385, 387 oxygen free radicals, 4, 17, 369, 370 oxygen toxicity, 3 ozone, 4, 7, 13, 185, 193, 227, 234, 252, 285, 293, 310

Р

p22^{phox}, 5 p300 histone acetyltransferase, 322, 331 p32^{TrxL}, 111 p33, 181, 190 p40^{phox}, 5 P4502E1, 6, 7, 302, 309 p47^{phox}, 5, 274 p53, 67, 70, 75, 101, 126, 138, 140, 179, 180, 181, 187, 190, 192, 194, 219, 231, 271 p63,75 $p67^{phox}, 5$ pain, 370, 375 pancreas, 258, 322 pancreatic cancer, 168, 175, 289 pancreatic fibrosis, 114 pancreatic β-cells, 50, 76, 78, 127, 162, 227, 242, 322, 343, 355 pancreatitis, 114, 119, 209, 215, 252 papaya, 297 para-hydroxybenzoate, 244 paraoxon, 197, 198, 199, 204, 205 paraoxonase, 197, 198, 210, 211, 212, 213, 214, 215, 216

paraquat, 55, 77, 81, 82, 127, 132, 152, 221, 231, 349.370 parasitic infection, 169, 203 parathion, 197 Parkin, 114, 119 Parkinson's disease, 16, 37, 40, 76, 95, 96, 107, 114, 164, 185, 186, 188, 189, 207, 208, 246, 254, 269, 285, 301, 323, 344, 356, 371, 379 Parkinsonian, 188, 195 parkinsonism, 96, 100, 127, 153, 164, 186, 221, 227, 253, 269, 301, 370, 381 pathogenesis, 13, 41, 42, 87, 95, 163, 172, 176, 189, 208, 217, 221, 227, 234, 324, 332, 363 pathogens, 17, 158, 203 pathology, 52, 70, 167, 189, 234, 276, 362 pathophysiology, 3, 14, 15, 16, 25, 27, 33, 35, 40, 49, 52, 62, 64, 66, 106, 107, 118, 153, 154, 155, 162, 166, 169, 188, 204, 209, 214, 221, 230, 247, 252, 268, 284, 286, 287, 300, 315, 318, 320, 321, 322, 327, 337, 339, 342, 343, 353, 354, 359, 367, 368, 369, 373, 380, 381, 384 pathways, 5, 11, 12, 63, 97, 100, 101, 113, 123, 126, 128, 131, 137, 145, 153, 161, 173, 174, 200, 224, 225, 248, 259, 291, 300, 310, 318, 322, 323, 325, 326, 333, 373 Pax5, 125, 130 PBN, 367, 368, 369, 370, 371, 372, 374 PDGF, 127, 131, 310 PDGF signaling, 127 penis, 39, 44 peptic ulcer, 302, 356 peptic ulcer disease, 356 peptide amidation, 266 peptides, 164, 381 percutaneous coronary intervention, 359, 364 performance, 52, 56, 107, 232, 350, 374, 382 perfusion, 45 peripheral blood, 183, 184, 185 peritoneal, 51, 55 peritoneal dialysis, 51 peritoneal fibrosis, 55 peritonitis, 252 permeability, 41, 349 peroxidases, 47, 80, 82, 83, 94, 123 peroxidation, 11, 12, 13, 14, 18, 28, 38, 62, 77, 78, 81, 94, 99, 163, 172, 181, 199, 201, 202, 213, 215, 221, 245, 248, 254, 283, 285, 294, 309, 369, 374, 379 peroxide, 3, 4, 5, 6, 8, 9, 12, 13, 21, 28, 33, 34, 35, 47, 48, 49, 50, 51, 52, 53, 54, 55, 56, 62, 73, 75, 99, 112, 123, 124, 125, 126, 129, 133, 145, 154, 158, 180, 219, 230, 251, 252, 267, 277, 339, 340, 341, 342, 378, 379, 380, 385

peroxides, 62, 75, 77, 79, 123, 126, 128, 144, 205, 208, 342, 351, 387 peroxiredoxin, 11, 25, 123, 124, 130, 131, 132, 133, 134, 140, 144, 146, 267 peroxisomal diseases, 52 peroxisome, 6, 49, 200, 233, 245, 257, 270, 276, 285.292 peroxisome proliferator-activated receptor, 49, 245, 257 peroxyl radical, 4, 11, 240, 243, 245 peroxynitrite, 4, 7, 8, 10, 11, 15, 18, 35, 42, 62, 75, 80, 112, 123, 124, 126, 240, 251, 252, 253, 260, 267, 283, 299, 319, 339, 340, 342, 344, 347 peroxynitrite scavengers, 339, 342, 344 peroxynitrous acid, 10 pesticide, 145 pesticides, 197, 198, 199, 201, 204 pH, 35, 252, 267 phagocytes, 5 phagocytosis, 87, 164 pharmacodynamics, 274 pharmacokinetics, 274, 290 pharmacology, 36, 274, 347, 365 phase 2 biotransformation, 94, 317 phase 2 enzymes, 88, 193, 259, 303, 327 PHB, 244 PHB-polyprenyl transferase, 244 phenol, 315 phenolic acids, 315 phenolic compounds, 181, 200, 219, 263, 315, 316, 317, 318, 319, 320, 321, 322, 323, 324, 325, 326, 327, 328, 329 phenotype, 43, 138, 171, 187, 214, 222, 269, 276 phenyl acetate, 199 phenylalanine, 98, 244 phenylephrine, 321 PHGPx, 80 phorbol-12-myristate-13-acetate, 383 phosphatase type 2A, 270, 277 phosphatases, 8, 62 phosphates, 291 phosphatidylcholine, 82 phospholipase A2, 128, 319 phospholipid hydroperoxide GPx, 74, 75 phosphorylation, 49, 50, 103, 126, 132, 137, 153, 233, 246, 277, 291 photocarcinogenesis, 303, 310, 311 photochemical reaction, 13 photodamage, 293, 297, 303, 305 photons, 382 photoprotection, 295, 303, 332 photoreceptor, 382 photosensitivity, 12

photosensitivity reaction, 12 photosensitizer, 12 physical quenching, 13 physicochemical properties, 384 physiology, 18, 39, 59, 62, 65, 66, 77, 108, 155, 274, 330 phytochemicals, 306, 329 π -cationic porphyrin radical, 48 piceatannol, 320 PICOT, 104, 108, 109 PICOT homologous domains, 104 pigmentation, 186, 193 pigments, 243, 244, 257, 297, 302, 303, 305 pigs, 73, 165, 173, 265, 362 pilot study, 260, 289, 295, 311, 334, 365 pineal gland, 250 pinpoint hemorrhage, 266 pioglitazone, 289, 295 PKC-B2, 355 placebo, 28, 41, 247, 254, 259, 272, 273, 278, 289, 295, 311, 312, 334, 350, 359, 364, 365, 372 placenta, 250 plant pigments, 297 plants, 297, 303, 315 plaque, 202, 212, 268, 275, 355 plasma, 34, 60, 64, 74, 76, 77, 80, 83, 112, 159, 167, 197, 199, 207, 208, 210, 214, 216, 218, 219, 240, 241, 242, 243, 245, 246, 249, 250, 252, 253, 254, 266, 268, 272, 282, 287, 299, 305, 317, 333 plasma GPx, 74 plasma levels, 76, 252, 253, 305 plasma membrane, 34, 159, 266 plasminogen, 254, 261 platelet aggregation, 158, 354 platelets, 184, 361 platinum, 377, 378, 379, 381, 382, 383, 384, 385, 387 platinum nanoparticles, 377, 378, 379, 381, 382, 383, 384.385 PMA, 383 pneumonia, 115, 120, 364, 381 pneumonitis, 387 p-nitrophenol, 198 point mutation, 206, 223 pollution, 13, 56 polychlorinated alkenes, 66 polycyclic aromatic hydrocarbon, 215 polyethylene glycol-SOD, 41 polyhydroxylated C₆₀ fullerenes, 379 polyhydroxylated C₈₂ fullerene, 383 polymeric materials, 377 polymorphism, 40, 45, 52, 56, 57, 67, 68, 70, 83, 84, 101, 102, 117, 138, 154, 158, 169, 170, 176, 177,

184, 186, 187, 188, 189, 193, 194, 195, 205, 206, 207, 208, 213, 214, 215, 229, 236, 243 polymorphisms, 40, 42, 45, 47, 52, 53, 56, 57, 67, 70, 71, 74, 79, 83, 91, 98, 99, 101, 102, 107, 117, 122, 130, 138, 169, 170, 176, 179, 186, 187, 188, 189, 190, 194, 195, 204, 205, 206, 207, 208, 209, 213, 214, 215, 216, 229, 236, 237 polypectomy, 328 polypeptide, 232 polyphenols, 22, 26, 200, 210, 211, 306, 315, 321, 323, 324, 325, 326, 327, 328, 329, 330, 331, 332, 333.334 polyps, 360 polyunsaturated fat, 283 PON, 197, 200, 201, 202, 203, 204, 205, 206, 207, 208.209 PON1, 197, 198, 199, 200, 201, 202, 203, 204, 205, 206, 207, 208, 209, 210, 211, 213, 214, 215, 216 PON1 L55M, 207 PON1 Q192R, 207, 208 PON2, 197, 198, 199, 200, 201, 202, 203, 205, 206, 207PON3, 197, 198, 199, 200, 201, 202, 203, 206 population group, 98 porphyrin, 12, 37, 48, 340, 341, 347, 350 positive correlation, 254 postmenopausal, 224, 232 postmenopausal women, 224, 232 post-transcriptional regulation, 61, 93 preclinical studies, 346, 367, 371, 373, 384 premature death, 128 pressure overload, 107 prevention, 79, 140, 184, 206, 213, 233, 277, 278, 289, 291, 295, 305, 309, 311, 321, 328, 330, 331, 333, 334, 346, 349, 350, 351, 360, 363, 365, 374, 384 primary function, 103 primate, 275, 372 primates, 370 primitive mesoderm, 137 Pro187Ser, 186, 194 Pro197Leu, 79 prodrugs, 70, 353 prognosis, 117, 122, 129, 134, 207, 236 progressive IgA nephropathy, 209, 215 pro-inflammatory, 246, 380 proinflammatory cytokine, 15, 38, 54, 120, 127, 166, 227, 252, 276, 283, 300, 302, 319, 354, 356, 370 proliferation, 51, 95, 97, 100, 109, 116, 126, 129, 130, 136, 143, 145, 147, 168, 175, 184, 193, 222, 228, 235, 242, 255, 290, 319 proline hydroxylase, 266

- promoter, 49, 56, 70, 75, 79, 83, 94, 105, 117, 125,
- 140, 144, 145, 151, 156, 161, 169, 176, 177, 181, 182, 189, 191, 193, 200, 205, 207, 214, 219, 225, 229, 231
- pro-oxidant, 38, 40, 41, 161, 217, 266, 267, 272, 283, 304, 317, 318, 319, 326, 330
- propagation, 11
- prophylaxis, 359, 365
- prostaglandin E1, 41
- prostaglandins, 140, 319
- prostate cancer, 78, 79, 83, 84, 168, 175, 188, 209, 215, 295, 305, 311, 328, 334, 345, 350
- prostate carcinogenesis, 78, 83
- prostate-specific antigen, 328, 334
- protective factors, 209
- protective role, 25, 36, 37, 38, 39, 42, 47, 50, 55, 59, 67, 76, 78, 94, 95, 98, 106, 107, 113, 114, 115, 116, 118, 123, 126, 127, 128, 129, 138, 152, 154, 155, 162, 163, 175, 181, 184, 185, 187, 188, 200, 202, 206, 217, 220, 224, 226, 227, 229, 231, 234, 241, 244, 254, 287, 320, 323, 328, 337, 345, 367, 369, 377, 383
- protein carbonylation, 98, 101
- protein deglutathionylation, 59, 105, 143
- protein denitrosylation, 112, 118
- protein dephosphorylation, 62
- protein disulfide, 86, 112, 136
- protein disulfide isomerase, 136
- protein glutathionylation, 103, 105, 145, 199
- protein glycation, 267
- protein kinase C, 25, 104, 200, 210, 284, 347
- protein kinase C interacting cousin of thioredoxin, 104
- protein kinases, 25, 28, 161, 307
- protein oxidation, 44, 82, 129, 152, 156
- protein sequence, 223
- protein-protein interactions, 112
- proteins, 10, 11, 12, 13, 14, 15, 17, 18, 21, 22, 26, 27, 28, 31, 36, 47, 52, 63, 74, 91, 103, 104, 105, 106, 108, 111, 112, 113, 115, 117, 118, 123, 130, 134, 137, 138, 144, 149, 150, 153, 154, 164, 171, 172, 173, 190, 191, 200, 217, 218, 223, 224, 225, 226, 227, 228, 229, 230, 245, 246, 248, 301, 303, 308, 348 proteoglycan deglycanation, 268 proteolysis, 145 proteome, 191 pro-vitamin A activity, 297, 299 pruritis, 41 Prx, 123, 124, 125, 126, 127, 128, 133, 144
- Prx1, 123, 124, 125, 126, 127, 128, 159, 111 Prx1, 123, 124, 125, 126, 127, 128, 129, 130
- Prx6, 124, 125, 127, 128, 129
- T 1X0, 124, 125, 127, 120, 127
- Pseudomonas aeruginosa, 203, 210, 212, 213

psoriasis, 326 psychiatric disorders, 189 psychosis, 189, 195 PTEN, 128, 133 pulmonary diseases, 65, 169, 242, 285, 324, 344, 360, 381 pulmonary edema, 273 pulmonary endothelium, 50, 55 pulmonary fibrosis, 43, 165, 173, 242, 256, 292, 308, 324, 332 pulmonary toxicity, 115, 164 pumps, 94 purification, 108, 130, 191 Purkinje cells, 159 pyridine nucleotide disulfide oxidoreductase, 135 pyruvate, 239, 251, 252, 254, 259 pyruvic acid, 251

Q

QQ192, 206, 207 QR192, 206 quality of life, 272, 359 quantum dots, 6 quartile, 206 quercetin, 200, 293, 294, 316, 320, 324, 325, 334 quinone, 5, 7, 23, 25, 29, 93, 95, 179, 180, 181, 182, 183, 184, 186, 187, 190, 191, 192, 193, 194, 195, 244, 245, 257 quinoneimines, 180 quinones, 179, 180, 183, 184, 186, 187, 188, 192, 218 quorum sensing, 199, 203, 212

R

Rac, 5

radiation, 3, 6, 9, 37, 70, 80, 183, 192, 236, 252, 335, 344, 345, 346, 348, 350, 371, 381, 382, 387, 388 radiation therapy, 345 radical formation, 9, 38, 369 radical mechanism, 3 radicals, 3, 4, 9, 11, 12, 17, 19, 69, 180, 218, 283, 297, 299, 303, 340, 350, 364, 367, 368, 369, 370, 373, 374, 378, 379, 380 radiocontrast agents, 167, 270 radiotherapy, 345, 351 ragged-red fibers, 246 Raman spectroscopy, 154 randomized controlled clinical trials, 271, 274, 281, 287, 306 Ras, 128 reaction rate, 8, 10, 33, 34, 35, 283, 378 reaction rate constant, 8, 10, 33, 34, 35, 378 reactions, 8, 9, 11, 13, 21, 35, 59, 60, 61, 62, 75, 91, 93, 94, 104, 160, 180, 217, 225, 244, 252, 267, 268, 317, 358, 378, 385 reactive nitrogen species, 4 reactive oxygen, 1, 5, 16, 17, 21, 31, 50, 51, 54, 55, 59, 62, 85, 91, 106, 111, 112, 120, 126, 145, 149, 161, 179, 180, 226, 232, 233, 240, 241, 259, 267, 283, 300, 318, 337, 339, 349, 350, 373, 375, 377, 378, 385, 387 reactive oxygen species, 17, 51, 54, 55, 91, 106, 112, 120, 145, 149, 180, 232, 241, 259, 349, 350, 373, 375, 385, 387 reactive species, 3, 4, 7, 8, 10, 14, 26, 62, 96, 161, 218, 225, 243, 253, 299, 303, 318, 384 reactivity, 7, 11, 13, 17, 99 receptors, 151, 186, 258, 285, 292, 319, 320 recognition, 165, 339, 344, 380 recombinant tissue plasminogen activator, 254, 261 recommendations, iii recurrence, 130, 134, 170, 177 red blood cell count, 345 red blood cells, 39, 48, 67, 85, 87, 129 redox cycling, 5, 180, 181 redox homeostasis, 85, 87, 103, 105, 143, 144, 166, 174 redox proteins, 111 redox regulation, 51, 82, 91, 131, 156, 161 redox signaling, 62, 69, 85, 95, 118, 199, 277, 330, 361 reepithelialization, 167, 228 reflexes, 374 regenerate, 48, 96, 245, 267 regeneration, 63, 86, 228, 235, 381 regression, 371, 373 rejection, 41, 45, 369, 374 remodelling, 131 renal allograft nephropathy, 286, 293 renal cell carcinoma, 130 renal diseases, 39, 286, 325, 371 renal dysfunction, 167, 172, 302, 310, 343 renal failure, 249, 258, 293, 345, 349, 371, 375 renal inner medullary cells, 159 renal medulla, 325, 332 renal proximal tubular cells, 38, 56 reoxygenation, 130, 152, 156 repair, 21, 22, 28, 29, 70, 136, 146, 154, 155, 157, 158, 326, 332 replication, 67, 140, 176 repression, 191 repressor, 218 reproduction, 78, 135, 281, 291

reproductive organs, 224 residues, 22, 25, 73, 103, 104, 112, 113, 124, 126, 149, 150, 151, 153, 154, 219, 226 resistance, 16, 37, 38, 66, 76, 81, 96, 97, 98, 100, 101, 113, 116, 117, 118, 127, 128, 132, 145, 156, 157, 162, 164, 168, 176, 183, 184, 187, 203, 207, 212, 221, 224, 227, 229, 231, 232, 233, 242, 285, 287, 289, 294, 301, 320, 323, 343, 345, 348, 349, 355, 361 respiration, 5, 17 respiratory burst, 5, 12, 13, 291 respiratory distress syndrome, 176, 344 respiratory dysfunction, 273 responsiveness, 44, 258, 277 restenosis, 127, 184, 193 resveratrol, 182, 191, 200, 210, 315, 316, 319, 321, 323, 324, 325, 326, 327, 328, 334 retardation, 137, 167, 169, 224, 232, 247, 321 reticulum, 48, 124, 150, 159, 210, 233, 380 retina, 43, 78, 83, 153, 157, 250, 299, 301, 302, 305, 382 retina-associated carotenoids, 301 retinal degeneration, 371, 375, 382, 387 retinal ischemia, 302, 310 retinal ischemia-reperfusion injury, 302 retinal photic injury, 116, 121 retinitis, 287 retinitis pigmentosa, 287 retinoic acid receptors, 151 retinopathy, 36, 301, 302 rheumatoid arthritis, 134, 177 rhinitis, 285, 293 rhinosinusitis, 285, 293 rhinovirus infection, 301, 308 rHuPON1k192, 204 riboflavin, 87, 89 ribonucleotide reductase, 104, 111, 112 risk assessment, 68, 138 risk factors, 87, 101, 117, 208, 232 RNA, 69, 87, 145, 173, 176, 187 RNS, 1, 3, 4, 6, 7, 14, 15, 16, 17, 21, 22, 24, 25, 26, 27, 28, 29, 62, 87, 163, 240, 245, 248, 249, 251, 252, 254, 270, 283, 285, 318, 337, 339, 342, 343, 354, 378, 384 rodents, 224, 370 ROS, 4, 5, 6, 7, 17, 51, 66, 94, 106, 112, 126, 127, 128, 129, 145, 157, 163, 181, 186, 189, 219, 221, 242, 246, 249, 267, 270, 283, 285, 300, 302, 318, 319, 343, 349, 379, 382, 383 rotenone, 145, 153 RR192, 206, 208 RRR-α-tocopherol, 282, 289 rt-PA, 260

senescence, 14, 39, 44, 116, 121, 129, 133, 138, 140, S 152, 270, 276, 308 senescence marker protein 30, 270, 276 Saccharomyces cerevisae, 123 sensing, 199, 203, 210, 212 SAINT I, 372, 376 sensitivity, 38, 44, 55, 64, 77, 81, 87, 97, 100, 101, SAINT II, 372 107, 109, 128, 129, 152, 153, 157, 164, 168, 172, salmon, 297 175, 183, 191, 192, 194, 222, 232, 257, 285, 292, salt-induced hypertension, 242 323.334 sarcomas, 128 sensorimotor deficits, 269, 275 sarin, 199 sensors, 25 S-carboxymethylcysteine, 356, 359, 362 sepsis, 78, 116, 129, 169, 226, 233, 243, 246, 249, scattering, 83 251, 252, 270, 273, 277, 289, 295, 327, 333, 346, scavengers, 42, 80, 243, 297, 303, 339, 341, 342, 358, 371 344, 386 Sept15, 73 schizophrenia, 68, 71, 154, 158, 189, 195, 207, 208 septic shock, 50, 116, 129 schizophrenic patients, 214 sequencing, 86, 88 sclerosis, 16, 41, 50, 207, 208, 214, 253, 260, 289, serine, 186, 205 295 serum, 169, 185, 188, 199, 201, 203, 204, 205, 206, scorbutic gums, 266 207, 208, 210, 211, 213, 214, 215, 222, 223, 224, screening, 47, 145 232, 254, 257, 260, 261, 273, 289, 295, 311, 312, scurvy, 67, 266, 271 328, 334 seaweed, 301 serum ferritin, 222, 223, 224, 232 secoisolariciresinol, 315, 316 sex, 189, 320 second messenger, 62 sex hormones, 320 secretion, 43, 203, 276, 282, 322, 356, 362 SGLT1, 322 seeds, 281, 317 S-glutathionylation, 62, 95, 108, 146, 199 seizure, 227, 370 sheer force, 184 Sel I, 73 shock, 16, 39, 45, 50, 116, 129, 133, 164, 252, 253, Sel K. 73 257, 258, 260, 277, 350, 364 Sel M, 73 sickle cell, 169, 176 Sel N, 73 sickle cell disease, 169, 176 Sel O. 73 sideroblastic-like microcytic anemia, 107, 109 Sel P, 73 sideroblasts, 107 Sel S, 73 sigma state, 4 Sel T, 73 signal transduction, 3, 7, 14, 15, 91, 103, 109, 111, Sel V, 73 251, 284, 290, 320 Sel W, 73 signaling pathway, 63, 123, 126, 153, 161, 200, 259, selectivity, 36, 162, 340, 343, 346, 347 293, 307, 318, 321, 323, 324, 325, 326 selenenic acid, 151 signalling, 17, 131, 307 seleneylsulfide, 151 signals, 199, 203 selenium, 73, 81, 82, 84, 138, 295, 341, 380 silica, 6 selenium deficiency, 138 silica dusts, 6 selenocystamine, 151 silicon, 6 selenocysteine, 73, 150, 151 single-nucleotide polymorphism, 138 selenoenzyme, 74, 80, 141 singlet oxygen, 4, 12, 13, 18, 283, 297, 299, 380 selenophophate synthetase 2 siRNA, 87, 165, 200, 220 selenoprotein, 29, 73, 78, 80, 139, 140, 149, 150, Sirt1, 49, 320, 321, 323, 324, 325, 331, 332 155, 156 Sirt3, 49, 54 selenoprotein H, 73 sirtinol, 324 selenoprotein R, 73, 150 sirtuins, 319, 330 semen, 333 skeletal muscle, 44, 150, 221, 231, 258, 277, 374 semidehydroascorbate, 267 skin, 16, 39, 41, 44, 51, 66, 78, 116, 121, 128, 133, semiquinone radicals, 180 145, 147, 149, 153, 154, 159, 167, 175, 183, 186,

192, 228, 229, 235, 236, 246, 250, 251, 266, 270, 273, 277, 286, 289, 294, 303, 305, 310, 311, 312, 317, 325, 329, 332, 333, 335, 346, 350, 351, 358, 383, 388 skin aging, 39, 41, 78, 116 skin cancer, 78, 145, 286, 294 skin carcinogenesis, 39, 116, 121, 183, 192, 229, 236, 303, 311, 326, 350 skin ischemic injury, 39 Smad, 75 small intestine, 309 small Maf, 24, 30 S-methyl-L-cysteine, 153, 157 smoke exposure, 195 smoking, 56, 83, 101, 102, 312 smooth muscle, 95, 100, 152, 156, 184, 193, 226, 233, 242, 255 smooth muscle cells, 152, 156, 193 S-nitrosoglutathione, 62, 64 S-nitrosylation, 15, 19, 64, 68, 132 soccer, 379 SOD, 33, 34, 35, 36, 37, 38, 39, 40, 41, 42, 43, 44, 53, 340, 341, 342, 343, 344, 345, 346, 348, 349, 361, 378, 379, 382, 384, 387 SOD/catalase mimetics, 341 SOD1, 44 SOD2, 43, 45 sodium, 302, 309, 322, 345, 349, 360, 370, 375 sodium bicarbonate, 360 sodium-dependent glucose transporter, 322 solid tumors, 176 solubility, 240, 284, 377, 379 soman, 199 soy flavonoids, 326, 328 SP1, 35, 61, 75, 112, 113, 137, 151, 182, 200, 225 SP3, 75, 137, 182 spectroscopy, 154, 367, 368, 370, 373 sperm, 74, 75, 80, 83, 111, 136, 139 sperm maturation, 75, 136 sperm Trx, 111 spermatocytes, 78, 80 spermatozoa, 74, 75, 78 spin traps, 22, 337, 367, 368, 369, 370, 371, 372, 374 spina bifida, 117, 118, 122 spinach, 297 spinal cord, 163, 172, 227, 234, 252, 253, 260, 356, 361, 386 spinal cord injury, 163, 227, 234, 253, 260, 356, 361 spironolactone, 199 spleen, 117, 159, 218, 222 splenic atrophy, 96 spontaneous dismutation, 8, 35

spontaneous hypertensive rats, 248

Sprague-Dawley rats, 192 squamous cell, 78, 82, 170, 177 squamous cell carcinoma, 78, 82, 170, 177 Srx, 143, 144, 145 stabilization, 61, 64, 179, 259, 271 stable asthma, 312 statin, 200 statin drugs, 200 steatohepatitis, 166, 174, 215, 289, 366 steatosis, 65, 70, 362, 366 stenosis. 236 steroids, 359 sterol regulatory element-binding protein-2, 200, 210 stilbenes, 315 strawberries, 265 streptococci, 158 streptozotocin, 78, 82, 114, 115, 119, 131, 162, 172, 202, 227, 233, 242, 348, 361, 369 stressors, 166, 168 stroke, 79, 83, 100, 163, 207, 213, 224, 232, 252, 254, 260, 261, 269, 285, 323, 331, 346, 350, 356, 361, 367, 370, 372, 374, 376 structural protein, 78 ST-segment elevation myocardial infarction, 359, 364 subacute, 387 subarachnoid hemorrhage, 346, 350 substantia nigra, 96, 164, 188, 195, 221, 381 substitution, 79, 186, 187, 205 substitutions, 206 substrates, 18, 28, 48, 94, 97, 104, 124, 136, 181, 198, 199, 353 succinvlated catalase, 50 sucrose, 355, 361 sulfenic acid, 125, 144, 151 sulfhydryl, 8, 25, 60, 86, 112, 125, 199, 210, 362 sulfinic acid, 126, 144, 146, 147 sulfinic acid phosphoric ester, 144 sulfinic peroxiredoxins, 143 sulfiredoxin, 144, 146 sulfur, 8, 55, 73, 82, 105, 107, 109, 151, 219, 249 sunburn cells, 228 superoxide, 3, 4, 5, 6, 7, 8, 9, 10, 11, 18, 21, 22, 25, 33, 34, 35, 37, 38, 39, 42, 43, 44, 45, 49, 50, 51, 53, 54, 55, 87, 167, 172, 180, 181, 190, 195, 219, 248, 276, 318, 319, 337, 339, 340, 341, 342, 343, 344, 347, 348, 349, 351, 361, 370, 378, 379, 380, 384, 385, 387 superoxide dismutase, 3, 8, 18, 21, 22, 25, 33, 34, 42, 43, 44, 45, 49, 50, 51, 53, 54, 55, 87, 195, 248, 337, 339, 341, 347, 348, 349, 351, 378, 379, 384, 385, 387 superoxide dismutase mimetics, 347, 348, 349

suppression, 38, 70, 82, 116, 133, 165, 166, 167, 168, 183, 186, 201, 242, 259, 271, 294, 300, 303, 310, 318, 322, 326, 350, 354, 358, 370 surface structure, 379 surfactant, 128 survival, 36, 37, 38, 39, 41, 43, 44, 51, 63, 65, 69, 79, 87, 88, 114, 117, 119, 130, 134, 164, 165, 166, 168, 170, 174, 175, 177, 187, 204, 213, 222, 233, 242, 252, 257, 269, 271, 272, 277, 278, 292, 320, 324, 330, 345, 356, 358, 361, 369, 371, 381 survival rate, 358 susceptibility, 35, 51, 53, 55, 79, 84, 86, 88, 97, 98, 101, 107, 109, 118, 122, 128, 131, 133, 139, 153, 177, 179, 183, 185, 187, 189, 190, 192, 193, 207, 208, 215, 229, 236, 237, 269, 363 swelling, 116 symptoms, 77, 87, 157, 189, 195, 223, 287, 306, 312, 372 synchronization, 186 syndrome, 118, 138, 155, 162, 165, 166, 167, 171, 174, 184, 188, 197, 201, 202, 207, 208, 212, 215, 221, 223, 227, 243, 246, 247, 249, 250, 254, 257, 259, 269, 270, 273, 274, 276, 279, 285, 301, 306, 308, 312, 323, 328, 331, 339, 343, 355, 360 synergistic effect, 67, 100 syngeneic recipients, 222 synthesis, 59, 60, 61, 63, 65, 68, 69, 71, 99, 108, 109, 111, 202, 208, 220, 245, 249, 266, 268, 284, 285, 332, 354 synthetic antioxidants, viii, 22, 337 systemic inflammatory response syndrome, 165, 166, 167, 174 systemic lupus erythematosus, 57 systemic sclerosis, 41, 289, 295 systolic blood pressure, 71, 333, 359, 365 Т T cell, 70, 156, 159, 167, 173, 175 T regulatory cells, 242, 256 tamoxifen, 258 Tartars, 154 Tat-Cu,ZnSOD fusion protein, 37 tau, 153, 164 tau phosphorylation, 153

tea catechins, 322, 323, 324, 325, 326, 328, 331, 334

tea polyphenols, 324, 325, 326, 328, 329, 330, 332

tau protein, 164

taurine, 13

taxifolin, 316

telangiectasia, 345

telomerase, 49

TBP-2, 113

Tempol, 343, 344, 345, 346, 347, 348, 351 tension, 344 testes, 159, 250 tetrachlorodibenzo-p-dioxin, 182 tetrapyrrole biliverdin, 160 TGR, 135, 136, 137 Th1/Th2 balance, 269, 324, 332 therapeutic agents, 331 therapeutic goal, 68 therapeutic index, 51, 243 therapeutic intervention, 272, 274, 287 therapeutic targets, 127 therapeutics, 18, 113, 118, 347, 348, 373, 376, 377, 387 therapy, 28, 30, 40, 42, 55, 69, 70, 101, 106, 108, 117, 119, 121, 140, 168, 170, 175, 176, 207, 231, 235, 256, 259, 273, 278, 279, 286, 289, 294, 295, 345, 357, 363, 364, 365, 387 thioacetamide, 50, 115, 120 thiocyanate, 13 thiol, 15, 59, 62, 68, 104, 123, 126, 130, 143, 144, 149, 151, 274, 355 thiol-specific antioxidant, 123, 274 thionein, 151, 233 thioredoxin, 25, 73, 75, 87, 104, 111, 112, 113, 117, 118, 119, 120, 121, 122, 123, 125, 131, 132, 135, 139, 140, 141, 144, 149, 150, 151, 248, 258, 342, 347 thioredoxin binding protein-2, 113 thioredoxin glutathione reductase, 135 thioredoxin interacting protein, 113, 117, 122 thioredoxin peroxidase, 123, 131, 132 thioredoxin reductase, 25, 73, 87, 111, 112, 135, 139, 140, 141, 342, 347 third ventricle, 355 thrombogenesis, 321 thrombosis, 79, 83, 158 thymic lymphoma, 116, 121, 229, 345 thymine, 187 thymocytes, 185 thymus, 222, 250 tin, 162, 163, 167 tin chloride, 162 tin protoporphyrin-IX, 167 tissue, 5, 6, 7, 8, 9, 13, 14, 15, 17, 25, 26, 27, 35, 37, 38, 49, 50, 51, 53, 54, 55, 62, 63, 64, 65, 66, 77, 79, 86, 98, 108, 115, 116, 117, 118, 127, 129, 137, 139, 145, 152, 153, 154, 161, 164, 166, 167, 169, 181, 183, 184, 185, 186, 188, 193, 194, 209, 217,219, 220, 221, 222, 223, 225, 228, 234, 240, 242, 248, 249, 254, 261, 269, 270, 282, 285, 286, 299, 301, 303, 305, 308, 318, 322, 323, 324, 325, 340, 344, 353, 354, 360, 369, 370, 371

tissue injury, 7, 8, 9, 10, 13, 14, 15, 17, 25, 26, 37, 86, 115, 116, 183, 217, 219, 242, 269, 324, 340 tissue plasminogen activator, 254, 261 tissue remodeling, 37, 51 TNF, 82, 260, 277 tobacco, 38, 96, 170, 195, 308 tobacco smoke, 38, 96, 195, 308 tocopherols, 281, 283, 284, 286, 294, 329 tocotrienols, 281, 282, 283, 284, 285, 286, 287, 290, 292, 294, 329 tomato, 265, 297, 301, 307, 308, 309, 311 total cholesterol, 87 toxicity, 3, 11, 16, 42, 44, 54, 69, 76, 77, 82, 94, 96, 98, 101, 115, 116, 121, 132, 145, 153, 163, 164, 166, 179, 183, 185, 186, 191, 192, 204, 211, 217, 219, 227, 228, 229, 235, 237, 293, 309, 333, 344, 361, 368, 369, 382, 387 toxicology, 69, 99 toxification, 62 toxin, 227, 301, 370 TPA, 294 transaldolase, 358, 363 transcription, 23, 24, 25, 30, 35, 47, 49, 61, 75, 86, 93, 112, 113, 125, 126, 131, 140, 144, 151, 161, 171, 191, 200, 210, 219, 225, 230, 268, 271, 283, 300, 307, 318, 319, 373 transcription factors, 23, 35, 47, 61, 75, 93, 112, 113, 125, 126, 131, 140, 161, 171, 191, 219, 225, 373 transcripts, 161 transduction, 3, 7, 14, 15, 91, 103, 109, 111, 251, 284, 290, 320 transformation, 14, 129, 145, 147, 383 transforming growth factor, 165, 167, 222 transforming growth factor- β 1, 167 transgene, 167 transgenic overexpression, 26, 29, 36, 37, 38, 47, 49, 50, 51, 73, 76, 81, 86, 87, 88, 106, 107, 113, 114, 115, 116, 126, 127, 128, 139, 151, 162, 200, 201, 202, 203, 204, 206, 212, 219, 225, 371 transition metal, 8, 9, 385 transition metal ions, 8, 9 translation, 35, 40, 49, 181, 190, 219 translational research, 53, 79 translocation, 143, 145, 146, 291 transmission, 269 transplant recipients, 334 transplantation, 38, 41, 43, 50, 55, 114, 120, 162, 164, 166, 174, 177, 222, 242, 269, 270, 276, 286, 329, 334, 357, 360, 363, 369 transport, 5, 7, 18, 145, 152, 156, 182, 245, 246, 265, 266, 274, 322, 369, 373 trauma, 50, 371, 376, 381 traumatic brain injury, 163, 172, 234, 361

traumatic injury, 163, 172, 273 tremor. 138 trial, 41, 252, 273, 278, 279, 288, 289, 294, 295, 305, 306, 311, 312, 313, 328, 334, 346, 350, 359, 365, 372 triazoloacridin-6-ones, 182 tricarboxylic acid, 251 tricarboxylic acid cycle, 251 trichostatin A, 145 triggers, 13, 140 triglycerides, 101, 184, 202, 212 trinitrobenzene sulfonic acid, 174, 371 tripeptide, 60, 67, 239 triphenyl phosphine gold chloride, 137 triphenylphosphonium, 245, 284 Trn, 111 Trx, 111, 112, 113, 114, 115, 116, 118, 150 Trx1, 111, 112, 113, 114, 115, 116, 117, 118 Trx2, 111, 112, 113, 114, 115, 117 Trx2 haploinsufficiency, 115 TrxR, 112, 135, 136, 137, 139 TrxR1, 73, 135, 137, 138, 139, 140 TrxR2, 135, 137, 138, 139 TrxR3, 135, 136 Trypanosoma congolense, 203, 213 trypanosome infection, 204 trypanosome lysis factor, 203 tryptophan, 48, 187 tubular cell vaculolization, 228 tumor, 51, 54, 66, 67, 75, 78, 97, 101, 113, 116, 117, 128, 129, 130, 133, 138, 145, 168, 170, 175, 176, 177, 179, 180, 181, 183, 184, 185, 190, 192, 193, 200, 210, 219, 222, 229, 230, 236, 243, 257, 259, 267, 271, 275, 287, 300, 307, 319, 345, 348, 349, 350, 363, 371, 373, 383 tumor cells, 51, 66, 97, 168, 175, 183, 319 tumor growth, 51, 116, 117, 168, 176, 259, 275, 307, 345.383 tumor invasion, 177, 307 tumor metastasis, 51, 54 tumor necrosis factor- α , 184, 185, 193, 200, 210, 219, 230 tumor progression, 168 tumor promoter, 145 tumor promoting activity, 229 tumor suppressor, 75, 113, 129, 133, 138, 179, 180, 181, 190, 229, 236, 271 tumor xenografts, 267, 275 tumorigenesis, 70, 97, 101, 128, 133, 181, 320, 358, 383 tumors, 57, 67, 94, 97, 98, 101, 130, 168, 171, 176, 187, 192, 215, 252, 303, 333

turmeric, 317

turnover, 128, 159, 166 two-electron reduction, 6, 179, 180, 181, 183, 186, 245 Txnip, 113 Txnrd1, 135, 137, 138 Txnrd2, 135, 137, 138 Txnrd3, 135 type 1 diabetes, 37, 71, 78, 102, 114, 115, 122, 172, 195, 202, 242, 369 type 2 diabetes, 50, 54, 69, 71, 79, 83, 101, 114, 117, 119, 127, 155, 172, 176, 188, 194, 214, 236, 257, 276, 288, 295, 302, 312, 322, 331, 359, 365, 370 type II cells, 87, 89 typical 2-Cys Prxs, 124 tyrosinase, 154, 186 tyrosine, 35, 49, 54, 75, 80, 98, 113, 161, 233, 244, 266 tyrosine kinases, 49, 54, 75, 80, 161 tyrosine nitration, 35, 113

U

ubiquinol, 245 ubiquinone, 136, 180, 181, 244, 247, 257 ubiquitin, 181, 190, 268 ubiquitination, 24, 49, 54 UDP-glucuronosyltransferase, 241, 243 UDP-glucuronosyltransferase 1A1, 241 UGT1A1, 241, 243 UGT1A1*28, 243, 257 ulcer, 289, 295, 302, 356 ulcerative colitis, 195, 289, 295, 302, 357, 360, 365, 371 ultraviolet, 6, 39, 51, 66, 70, 78, 107, 116, 128, 161, 228, 236, 286, 303, 310, 325, 335, 346, 358, 383 ultraviolet B, 70, 228, 236 ultraviolet irradiation, 116, 346, 358, 383 unconjugated bilirubin, 240, 243 uncoupled nitric oxide synthase, 6 uncoupling protein, 245, 246, 257, 301, 308, 324, 332 underlying mechanisms, 241, 254, 325, 383 unesterified cholesterol, 200, 211 urate, 252, 253, 254, 260 urate oxidase, 252, 253 urban areas, 13 urea, 115, 345, 371 uremia, 345 ureteral obstruction, 51, 55, 167 uric acid, 6, 239, 252, 253, 254, 260, 261 uric acid levels, 254, 260 urinary bladder, 97, 101 urinary bladder injury, 97

urolithiasis, 189, 195 UV, 9, 51, 66, 67, 78, 82, 128, 157, 228, 270, 274, 277, 286, 289, 295, 303, 305, 325, 329, 332, 364, 388 UV irradiation, 78, 128, 157, 228, 270, 286, 289, 303, 305, 326, 329 UV light, 9, 51, 66, 67, 270, 274, 305 UVA irradiation, 82 UVB, 121, 293

V

vancomycin, 349 vascular cell adhesion molecule-1, 242 vascular diseases, 204 vascular dysfunction, 80, 113, 118, 127, 250, 306, 321, 355 vascular homeostasis, 11 vascular inflammation, 201, 202, 211, 304, 355 vascular smooth muscle cells, 152, 156, 193 vascular wall, 355 vasculature, 162, 181, 184, 321 vasodilation, 321, 348 vasomotor, 70, 201, 211 VCAM-1, 242 vector, 36, 37, 38, 49, 50, 51, 52, 87, 95, 106, 114, 119, 126, 129, 162, 184, 186, 200, 201 vegetable oil, 281 vegetables, 265, 271, 297, 299, 315, 327, 329 ventricle, 355 vesicle, 386 vessels, 64, 163, 168, 266, 291 vibration, 287 viral hepatitis, 302, 325 viral infection, 138, 209, 324 viral vector-mediated gene delivery, 36, 50 virus, 18, 56, 76, 174, 215 virus replication, 176 viruses, 5, 7 virus-induced myocarditis, 76 vision, 311, 382 visual acuity, 305 vitamin A, 274, 297, 299 vitamin A deficiency, 299 vitamin C, 12, 22, 26, 44, 62, 263, 265, 266, 267, 268, 269, 270, 271, 272, 273, 274, 275, 276, 277, 278, 279, 285, 286, 295, 327 vitamin C deficiency, 266, 268, 269, 270 vitamin E, 12, 22, 26, 30, 263, 269, 270, 272, 273, 281, 282, 283, 284, 285, 286, 287, 288, 289, 290, 291, 292, 293, 294, 295, 327 vitamin E deficiency, 283, 284, 287, 291, 292 vitamin supplementation, 265, 269, 271, 272

vitamins, 271, 275, 276, 278, 279, 294, 305, 311, 312 vitiligo, 52, 154, 158 VLDL, 201 vulnerability, 37, 169, 233

W

watermelon, 297 weight loss, 188, 194, 343 wheat, 115, 121 wheeze, 274 white blood cells, 98, 189 white matter, 163, 172 withdrawal, 189, 195 workers, 204, 236 working memory, 138, 140 wound healing, 51, 56, 66, 167, 175, 228, 235, 266, 326, 332, 358, 364

X

xanthine, 6, 7, 34, 252, 253, 318 xanthine dehydrogenase, 6 xanthine oxidase, 6, 34 xanthine oxidoreductase, 6, 7, 252, 253, 318 xanthophylls, 298 xenobiotic, 6, 36, 62, 65, 93, 95, 165, 166, 181, 200, 284, 286, 291, 302, 321, 325, 326, 343, 344, 345, 356, 357 xenobiotic biotransformation, 62, 284, 326 xenobiotic response element, 181, 200 xenograft, 78, 166 xenografts, 243, 267, 275 xeroderma pigmentosum, 51, 56 X-irradiation, 17, 229, 236 X-ray, 364 XRE, 182

Y

yAP-1, 86, 88 yeast, 86, 143, 151, 152, 155 young adults, 83

Ζ

zeaxanthin, 297, 299, 300, 301, 302, 303, 305, 307, 309, 310 zebrafish, 109, 186 zinc, 3, 33, 34, 156, 164, 165, 217, 224, 225, 228, 233, 235, 236, 305, 311, 386 zinc protoporphyrin-IX, 164 zinc sulfate, 228 zinc-deuteroporphyrin IX-2,4-bisethylene glycol, 165 Zutphen Elderly Study, 327, 333 zymosan, 370, 375